Confidential draft registration statement no. 1 submitted to the Securities and Exchange Commission on August 28, 2020. This draft registration statement has not been filed publicly with the Securities and Exchange Commission and all information contained herein remains strictly confidential.

Registration No. 333-

# UNITED STATES SECURITIES AND EXCHANGE COMMISSION

Washington, D.C. 20549

FORM S-1
REGISTRATION STATEMENT
UNDER
THE SECURITIES ACT OF 1933

## ContextLogic Inc.

(Exact Name of Registrant as Specified in its Charter)

**Delaware** (State or Other Jurisdiction of Incorporation or Organization) 5961

(Primary Standard Industrial Classification Code Number)

One Sansome Street 40<sup>th</sup> Floor San Francisco, CA 94104 (415) 432-7323

(Address, including zip code, and telephone number, including area code, of registrant's principal executive offices)

Rajat Bahri Chief Financial Officer ContextLogic Inc. One Sansome Street 40<sup>th</sup> Floor San Francisco, CA 94104 415-432-7323

(Name, address, including zip code, and telephone number, including area code, of agent for service)

Copies to:

Devang S. Shah Renee Jackson Jilliana Wong ContextLogic Inc. One Sansome Street 40<sup>th</sup> Floor San Francisco, CA 94104 (415) 432-7323 David Peinsipp Eric Jensen Calise Cheng Cooley LLP 101 California Street, 5<sup>th</sup> Floor San Francisco, CA 94111 (415) 693-2000

27-2930953

(I.R.S. Employer

Identification Number)

Ilan Lovinsky
Jeffrey R. Vetter
Ryan J. Gunderson
Colin G. Conklin
Gunderson Dettmer Stough
Villeneuve Franklin & Hachigian, LLP
One Bush Plaza, Suite 1200
San Francisco, CA 94104
(415) 978-9803

Approximate date of commencement of proposed sale to the public: As soon as practicable after the effective date of this Registration Statement.

If any of the securities being registered on this Form are to be offered on a delayed or continuous basis pursuant to Rule 415 under the Securities Act of 1933, check the following box.

If this Form is filed to register additional securities for an offering pursuant to Rule 462(b) under the Securities Act, please check the following box and list the Securities Act registration statement number of the earlier effective registration statement for the same offering.

If this Form is a post-effective amendment filed pursuant to Rule 462(c) under the Securities Act, check the following box and list the Securities Act registration statement number of the earlier effective registration statement for the same offering.

If this Form is a post-effective amendment filed pursuant to Rule 462(d) under the Securities Act, check the following box and list the Securities Act registration statement number of the earlier effective registration statement for the same offering.

Indicate by check mark whether the registrant is a large accelerated filer, an accelerated filer, a non-accelerated filer, a smaller reporting company or an emerging growth company. See the definitions of "large accelerated filer," "accelerated filer," "smaller reporting company," and "emerging growth company" in Rule 12b-2 of the Exchange Act.:

Large accelerated filer □
Non-accelerated filer ⊠

Accelerated filer □
Smaller reporting company □

Emerging growth company, indicate by check mark if the registrant has elected not to use the extended transition period for complying with any new or revised financial accounting standards provided pursuant to Section 7(a)(2)(B) of the Securities Act. 

□

## **CALCULATION OF REGISTRATION FEE**

Title of Each Class of Securities to be Registered	Proposed Maximum Aggregate Offering Price <sup>(1)(2)</sup>	Amount of Registration Fee
Class A Common Stock, \$0,0001 par value		

- 1) Estimated solely for the purpose of calculating the amount of the registration fee in accordance with Rule 457(o) under the Securities Act of 1933, as amended.
- (2) Includes the aggregate offering price of additional shares that the underwriters have the option to purchase.

The Registrant hereby amends this Registration Statement on such date or dates as may be necessary to delay its effective date until the Registrant shall file a further amendment which specifically states that this Registration Statement shall thereafter become effective in accordance with Section 8(a) of the Securities Act of 1933, as amended, or until the Registration Statement shall become effective on such date as the Securities and Exchange Commission, acting pursuant to such Section 8(a), may determine.

## **EXPLANATORY NOTE**

Pursuant to the applicable provisions of the Fixing America's Surface Transportation Act, we are omitting unaudited financial statements as of and for the six months ended June 30, 2020 and 2019. While this financial information is otherwise required by Regulation S-X, we reasonably believe that it will not be required to be included in the prospectus at the time of the contemplated offering. We intend to amend this registration statement to include all financial information required by Regulation S-X at the date of such amendment before distributing a preliminary prospectus to investors.

The information in this preliminary prospectus is not complete and may be changed. We and the selling stockholders may not sell these securities until the registration statement filed with the Securities and Exchange Commission is effective. This preliminary prospectus is not an offer to sell nor does it seek an offer to buy these securities in any jurisdiction where the offer or sale is not permitted.

**Subject to Completion, Dated** 

. 2020



**Shares** 

## Class A Common Stock

This is the initial public offering of shares of Class A common stock of ContextLogic Inc. (d/b/a "Wish"). We are offering shares of Class A common stock, and the selling stockholders identified in this prospectus are offering shares of Class A common stock. We will not receive any of the proceeds from the sale of shares by the selling stockholders.

We have two classes of common stock, Class A common stock and Class B common stock. The rights of the holders of Class A common stock and Class B common stock are identical, except voting, transfer, and conversion rights. Each share of Class A common stock is entitled to one vote. Each share of Class B common stock is entitled to 20 votes and is convertible at any time into one share of Class A common stock. See the section titled "Description of Capital Stock" herein for additional information on our capital stock. The holders of our outstanding shares of Class B common stock will hold approximately % of the voting power of our outstanding capital stock immediately following this offering, and our founder, Chief Executive Officer, and Chairperson, Peter Szulczewski, will hold, or have the ability to control, approximately % of the voting power of our outstanding capital stock immediately following this offering. See the section titled "Risk Factors—Risks Related to this Offering and Our Class A Common Stock" herein for additional information.

Prior to this offering, there has been no public market for our Class A common stock. It is currently estimated that the initial public offering price will be between \$ and \$ per share.

We intend to apply to list our Class A common stock on

under the symbol "

See the section titled "<u>Risk Factors</u>" beginning on page 19 to read about factors you should consider before deciding to invest in shares of our Class A common stock.

Neither the Securities and Exchange Commission nor any other regulatory body has approved or disapproved of these securities or passed upon the accuracy or adequacy of this prospectus. Any representation to the contrary is a criminal offense.

	Per Share	Total
Initial public offering price	\$	\$
Underwriting discounts and commissions <sup>(1)</sup>	\$	\$
Proceeds, before expenses, to Wish	\$	\$
Proceeds, before expenses, to selling stockholders	\$	\$

<sup>1)</sup> See the section titled "Underwriting" on page 165 for a description of the compensation payable to the underwriters.

To the extent that the underwriters sell more than shares of Class A common stock, the underwriters have an option to purchase up to an additional shares of Class A common stock from us at the initial public offering price, less the underwriting discounts and commissions.

The underwriters expect to deliver the shares of Class A common stock against payment in New York, New York on , 2020.

Goldman Sachs & Co. LLC

J.P. Morgan

**BofA Securities** 

Prospectus dated

, 2020

#### **TABLE OF CONTENTS**

Prospectus Summary	1
Summary Consolidated Financial and Other Data	16
<u>Risk Factors</u>	19
Cautionary Note Regarding Forward-Looking Statements	59
Industry Data and User Metrics	60
<u>Use Of Proceeds</u>	62
<u>Dividend Policy</u>	63
<u>Capitalization</u>	64
<u>Dilution</u>	67
Selected Consolidated Financial and Other Data	70
Management's Discussion and Analysis of Financial Condition and Results of Operations	75
<u>Business</u>	98
<u>Management</u>	120
Executive Compensation	128
Certain Relationships and Related Person Transactions	144
Principal and Selling Stockholders	147
Description of Capital Stock	151
Shares Eligible for Future Sale	158
Material U.S. Federal Income Tax Considerations for Non-U.S. Holders of Our Class A Common Stock	161
<u>Underwriting</u>	165
<u>Legal Matters</u>	172
Experts Experts	172
Where You Can Find More Information	172
Index to Consolidated Financial Statements	F-1

Through and including , (the 25th day after the date of this prospectus), all dealers effecting transactions in these securities, whether or not participating in this offering, may be required to deliver a prospectus. This is in addition to a dealer's obligation to deliver a prospectus when acting as an underwriter and with respect to an unsold allotment or subscription.

Neither we, the selling stockholders, nor any of the underwriters have authorized anyone to provide any information or to make any representations other than those contained in this prospectus or in any free writing prospectuses prepared by us or on our behalf or to which we have referred you. Neither we, the selling stockholders, nor any of the underwriters take any responsibility for, and can provide no assurance as to the reliability of, any other information that others may give you. This prospectus is an offer to sell only the shares our Class A common stock offered hereby, but only under circumstances and in jurisdictions where it is lawful to do so. The information contained in this prospectus is current only as of the date on the front cover of the prospectus. Our business, financial condition, results of operations, and prospects may have changed since that date.

For investors outside the United States: Neither we, the selling stockholders, nor any of the underwriters have done anything that would permit this offering or possession or distribution of this prospectus in any jurisdiction where action for that purpose is required, other than in the United States. Persons outside the United States who come into possession of this prospectus must inform themselves about, and observe any restrictions relating to, the offering of the shares of our Class A common stock and the distribution of this prospectus outside of the United States.

#### **PROSPECTUS SUMMARY**

This summary highlights information appearing elsewhere in this prospectus and is qualified in its entirety by the more detailed information and financial statements included elsewhere in this prospectus. This summary does not contain all the information you should consider before investing in our Class A common stock. You should read this entire prospectus carefully, including the sections titled "Risk Factors" and "Management's Discussion and Analysis of Financial Condition and Results of Operations" and our consolidated financial statements and related notes included elsewhere in this prospectus, before making any investment decision. Unless the context otherwise requires, we use the terms "Wish," the "company," "we," "us," and "our" in this prospectus to refer to ContextLogic Inc. (d/b/a "Wish").

#### Overview

We launched Wish with a simple mission—to bring an affordable and entertaining mobile shopping experience to billions of consumers around the world. Since our founding in 2010, our vision has been to unlock ecommerce for consumers and merchants, by providing consumers access to a vast selection of affordable products and by providing merchants access to hundreds of millions of consumers globally. We have become one of the largest and fastest growing global ecommerce platforms, connecting more than 100 million monthly active users ("MAUs" or "monthly active users") in over 100 countries to over 500,000 merchants offering approximately 150 million items. Our platform combines technology and data science capabilities, an innovative and discovery-based mobile shopping experience, a comprehensive suite of indispensable merchant services, and a massive scale of users, merchants, and items. This combination has allowed us to become the most downloaded global shopping app for each of the last three years according to a report from Sensor Tower.<sup>1</sup>

We are focused on democratizing mobile commerce by making it affordable and accessible to anyone. The global mobile commerce market was \$2.1 trillion in 2019 and is expected to more than double to reach \$4.5 trillion by 2024. While ecommerce grew from 3% of global commerce in 2010 to 14% in 2019, ecommerce companies have largely focused on serving affluent consumers by offering branded goods and prioritizing convenience over price. However, 44% of U.S. consumers and 85% of European consumers have a household income of less than \$75,000³ and cannot afford many traditional ecommerce offerings. Additionally, in the emerging economies of Africa, the Middle East, Latin America, and Eastern Europe, where the average household income is approximately \$18,000, affordability will be the key element for users shopping online. We believe that the next billion ecommerce customers will be these value-conscious consumers. According to a survey we conducted in 2020 across 2,850 consumers in select countries, approximately 75% of those responding prioritize the price of an item over brand and delivery time. We built Wish to serve these consumers who favor affordability over brand and convenience, and are being underserved by traditional ecommerce platforms.

We are revolutionizing mobile commerce with a user experience that is mobile-first, discovery-based, deeply-personalized, and entertaining. Over 90% of our user activity and purchases occur on our mobile app. Our data science capabilities allow us to mirror how consumers have shopped for decades in brick-and-mortar stores by offering a discovery-based shopping experience on a mobile device. Our highly-personalized product feed enables our users to discover products they want to purchase by simply scrolling through our mobile app and browsing. Over 70% of the sales on our platform do not involve a search query and instead come from personalized browsing. Wish users

- Sensor Tower, analysis of store intelligence platform data, November 2019.
- eMarketer, Global eCommerce 2020, June 2020.
- Euromonitor International Limited, Economies and Consumers, updated August 2020.

engage with our app in a similar manner to how they engage with social media: scrolling through image-rich, highly-engaging, and interactive content. To enhance user engagement, we incorporate fresh gamified features, rich user-generated content including photos, videos, and reviews, and a wide range of relevant products to make shopping more entertaining. Our differentiated user experience has driven superior engagement with Wish users spending on average 10 minutes per day on our platform in 2019.

We also built Wish to empower merchants around the world. Today, most of our merchants are based in China. We initially grew our platform focusing on merchants in China, the world's largest exporter of goods for the last decade,<sup>4</sup> due to these merchants' strength in selling quality products at competitive prices. We continue to expand our merchant base around the world. The number of merchants on our platform in North America, Europe, and Latin America has grown approximately 130% since 2019. In particular, the number of merchants on our platform in the United States has grown approximately 150% since 2019. Through our diversified and global merchant base, we are able to offer greater depth and breadth of categories and products. For example, in 2019, four out of the top 10 selling merchants on our platform were located in the United States selling refurbished electronics, beauty products, and hobby items, which illustrates the ongoing diversification of our merchant base and product categories. We give our merchants immediate access to our global base of over 100 million monthly active users and a comprehensive suite of indispensable services, including demand generation and engagement, user-generated content creation, data intelligence, promotional and logistics capabilities, and business operations support, all in a cost-efficient manner.

Local brick-and-mortar stores worldwide are struggling to attract consumers and compete in a retail world being transformed by ecommerce and industry consolidation. We launched Wish Local in 2019 to help these merchants increase their online reach and discovery, gain foot traffic, and drive additional sales. Today, we have over 40,000 Wish Local partners in over 40 countries who have signed up to our Sell on Wish feature to upload their in-store inventory on Wish for local pick-up or delivery. Our Wish Local partners also serve as Wish Pickup locations for online Wish orders, which effectively gives us a local warehousing and fulfillment footprint around the world without owning any real estate.

Our scale, combined with our extensive data science capabilities, provides us with a unique competitive advantage and is core to our business operations. We collect, analyze, and utilize data across over 100 million monthly active users, over 500,000 merchants, and approximately 2 million items sold per day to improve the shopping experience for users and the selling experience for merchants. Our proprietary algorithms analyze a rich and growing data set of transactions and historical behaviors of both users and merchants to drive continuous optimization on the platform and inform key business decisions on a daily basis. Our data science enables personalization at the individual user level at a massive scale and drives significant advantages across all aspects of our business operations, including user acquisition, user experience, pricing strategies, user-generated content, merchant insights, and user and merchant support. For example, our user acquisition strategies utilize our data science capabilities to make decisions on what to show to whom, when, and through which acquisition channel, with a focus on maximizing our return on marketing investment and user conversion. We also leverage our data and unique insights to extend our platform outside of our core business and drive additional growth opportunities, including new services to merchants and new categories for users.

Our proprietary data and technology also fuel our powerful network effects. As Wish grows, we accumulate more data across user and merchant activities, we strengthen our data advantage, and we create an even better experience for everyone on our platform, which in turn attracts more users and

The World Factbook 2020. Washington, DC: Central Intelligence Agency, 2020

merchants. As more users come to Wish, driven by the affordable value proposition and differentiated shopping experience, we drive more sales to our merchants. Adding more users also reinforces our user-generated feedback loop of ratings, reviews, photos, and videos, which drives greater user engagement. As more merchants succeed on Wish, more merchants join the platform to grow their businesses, broadening our product selection, which in turn improves the user experience. This flywheel effect has driven tremendous value to both users and merchants and has made Wish one of the largest ecommerce marketplaces in the world.

We have experienced substantial growth since our founding in 2010. We grew our revenue from \$1.1 billion in 2017 to \$1.9 billion in 2019 at a compound annual growth rate of 31%. Our revenue is diversified and global. In 2019, 93% of our revenue was from marketplace services and 7% was from logistics services. Within our marketplace, we built an advertising service with annual run-rate revenue of nearly \$180 million, based on the three months ended June 30, 2020. In terms of geographic diversification by users, in 2019, 49% of our marketplace revenue came from Europe, 38% from North America, 5% from South America, and 8% from the rest of the world. Our growth has been highly capital efficient. We have been able to achieve this massive growth and scale while only using \$8 million of cumulative cash flow from operations from 2017 to 2019, aided by our positive working capital dynamics. In 2019, we generated a net loss of \$129 million and Adjusted EBITDA of \$(127) million, compared to a net loss of \$208 million and Adjusted EBITDA of \$(211) million in 2018, and a net loss of \$207 million and Adjusted EBITDA of \$(135) million in 2017, which improved from (12)% of revenue in 2017 to (7)% of revenue in 2019. See "Selected Consolidated Financial and Other Data—Non-GAAP Financial Measures" for more information and for a reconciliation of Adjusted EBITDA to net loss, the most directly comparable financial measure calculated and presented in accordance with U.S. generally accepted accounting principles ("U.S. GAAP").

## **Our Market Opportunity**

Global ecommerce is a massive and growing market. In 2019, global ecommerce was a \$3.4 trillion market expected to nearly double to reach \$6.3 trillion by 2024. Within ecommerce, mobile is the clear dominant force, comprising 63% of global ecommerce in 2019, and is expected to grow to 71% by 2024<sup>5</sup>. While the market is large and rapidly growing, modern ecommerce has not evolved to fit the expectations and affordability needs of the global population.

### **Billions of Value-Conscious Consumers Are Underserved**

Value-conscious consumers represent a large and growing portion of the global consumer population, and they have been historically underserved by traditional ecommerce. For this segment of the global population, price is often the single most important determinant when making a purchase. Forty-four percent of U.S. consumers and 85% of European consumers have a household income of less than \$75,0006, and we estimate that there are over 1 billion households with income of less than \$75,000 around the world, excluding China and India. Additionally, in the emerging economies of Africa, the Middle East, Latin America, and Eastern Europe, where the average household income is approximately \$18,000, affordability will be the key element for users shopping online for the first time. We believe that the next billion ecommerce customers will be these value-conscious consumers.

<sup>5</sup> eMarketer, Global eCommerce, 2020.

<sup>6</sup> Euromonitor - Households by Disposable Income Band in Latin America, Eastern Europe, Western Europe and USA 2019-2020 and Number of Households in Latin America, Eastern Europe, Western Europe, Western Europe and USA 2019-2020

#### **Traditional Ecommerce Does Not Meet Evolving Consumer Behavior**

Shopping in store allows consumers to browse and discover new products that they want, driving many consumers to purchase items beyond their planned purchases. This type of navigational browsing often creates purchase intent for new products.

The largest ecommerce companies in the world were created on desktop, and their consumer experiences are predominantly search-driven rather than discovery-based. When porting that search-driven experience to mobile, consumers can shop for what they know they need, but struggle to browse, engage, and discover new products.

#### Merchants Around the World Lack Access to Global Consumers

Ecommerce represents a massive opportunity for global merchants, but they often struggle to access and serve global ecommerce consumers. IDC estimates that there are approximately 348 million small- and medium-sized businesses ("SMBs") around the world as of 2019,<sup>7</sup> and the success of these businesses is imperative to local, national, and global economies. SMBs make up over 90% of all enterprises globally and employ over 50% of the global workforce.<sup>8</sup> Despite their scale and economic power collectively, individual SMBs are often challenged, as evidenced by approximately 20% of U.S. SMBs failing in their first year and approximately 50% failing after five years in business, according to 2019 data.<sup>9</sup> For merchants around the world to grow their businesses, the ability to reach and target consumers at scale is critical.

### Merchants Lack Technology-Driven Tools to Operate Their Businesses

Many merchants lack the tools to operate their businesses efficiently, including expertise and resources to acquire users, a shipping and logistics platform, payment capabilities, access to credit, effective user support, and comprehensive data insights. For example, based on an independent survey, almost half (45%) of SMBs in the United States have reported a lack of any online presence, demonstrating challenges these merchants face in running their businesses in an increasingly digital world.<sup>10</sup>

#### The Wish Platform

Our global ecommerce platform connects over 100 million monthly active users in over 100 countries to over 500,000 global merchants. We seek to democratize ecommerce by making the Wish platform affordable, open, and accessible to all users and merchants worldwide. We do this through our relentless focus on product, technology, and data science. For our users, we are revolutionizing the mobile shopping experience by making it affordable, personalized, and entertaining. For our merchants, we offer immediate, cost-efficient access to our global user base, scaled data, and technology platform, as well as a comprehensive suite of indispensable services to help run their businesses and drive sales. To serve our global and diversified user and merchant base, we approach our platform development with a specific geographic focus, tailoring key features to solve for the needs of that locality, and enabling an authentic, localized experience.

Small Business Administration (SBA) Office of Advocacy, Frequently Asked Questions, 2018.
 CNBC Small Business Survey, 2017, updated May 2019.

IDC, Understanding the Needs of the Global Small and Medium-Sized Business Market, US46393020, May 2020.

United Nations Conference on Trade and Development, The International Day of Micro, Small and Medium Enterprises (MSMEs), June 2020.

#### **Value Proposition to Wish Users**

We have democratized ecommerce by making it:

- Affordable. Price is the single most important determinant when making a purchase for a substantial portion of the global population, and we aim to serve the affordability needs of these consumers. The merchants on our platform offer primarily unbranded products that can be discounted in excess of 85% as compared to branded alternatives across a number of categories<sup>11</sup>. We have a number of policies which, in combination with our robust user-generated content, promote higher quality merchants and products on our platform. This allows us to offer a vast selection of high-quality items at competitive prices, a value proposition that attracts more than 100 million monthly active users to our platform.
- Accessible. Within ecommerce, mobile is the clear dominant force, comprising 63% of global ecommerce in 2019, and is expected to grow to 71% by 2024. We built Wish to be mobile-first so any consumer around the world can easily access our shopping platform on a mobile device. We also do not have membership fees, understanding that millions of consumers are value-conscious and/or would not be able to afford such fees.
- Everywhere. To better serve our global user base, we localize various features on our online platform and tailored our experience to each respective market through, for example, making it accessible in 40 different languages and providing country-specific payment methods. This localization improves the engagement of our large, diverse user base, in addition to connecting our users with more than 40,000 local brick-and-mortar stores.

We have re-invented the online shopping experience to be:

- **Mobile-First.** Wish was built for mobile. Our application is image-rich, with minimal search input or text-based interactions. Over 90% of our user activity and purchases occur on our mobile app.
- **Discovery-Based.** Our platform is designed to make it easy to navigate a vast selection of products when users do not have a specific item or brand in mind. On average, our users see over 500 distinct products across many different categories on a daily basis. Unlike other ecommerce platforms where consumers often visit with a predetermined purchase intent on specific items, our navigational and entertaining shopping experience gives us the ability to create purchase intent in our users across a diverse set of products and categories. Over 70% of the sales on our platform do not involve a search query and instead come from personalized browsing.
- Personalized. No two users' Wish interfaces and product feeds look the same. Utilizing big data technology, we enable customization on a massive scale. We deliver personalized and curated products to our users and help them discover desired products quickly. Over 65% of our users click on a product detail page from the main feed.
- Entertaining. We have transformed the user experience to make shopping on Wish as engaging and entertaining as browsing social media. We utilize highly-personalized feeds with a wide range of relevant products as well as gamified, interactive, and social features to increase the length and frequency of a user's sessions and drive increased engagement. Wish users spent on average 10 minutes per day on our platform in 2019.
- Based on current internal research.
- eMarketer, Global eCommerce, 2020

### **Value Proposition to Wish Merchants**

Accessible and Cost-Efficient Ecommerce Platform. We give our merchants immediate access to our global base of over 100 million monthly active users and a comprehensive suite of indispensable services in a cost-efficient manner to help them run their businesses and grow sales. We seek to empower these highly capable merchants offering quality products at compelling values and unlock this supply of goods to consumers globally.

In addition, we give our merchants the following indispensable services:

Demand Generation and Engagement

- Global Reach for Online Merchants. Wish gives merchants immediate access to over 100 million monthly active users across more than 100 countries, with a significant user footprint in the United States and Europe. We help our merchants reach these users in a highly targeted and cost-efficient manner.
- Digital Presence for Brick-and-Mortar Stores. Through Wish Local, we partner with a global network of local brick-and-mortar stores and provide them with access to our global online user base and help enable online discovery of their stores and products.
- **Promotion.** Wish merchants can amplify their reach and sales by utilizing our native advertising tool, ProductBoost. We utilize data science to optimize ad placement, target users, and maximize the merchant's return on ad spend.

**User-Generated Content Creation.** User-generated content, in particular, authentic, localized content, can meaningfully improve user engagement and increase the purchases of products on our platform. Our value-conscious users rely on user-generated content such as reviews, ratings, photos, and videos, rather than brand recognition, when making purchase decisions. Our data science prioritizes items with favorable reviews, higher ratings and shipping history, connecting buyers with high-quality merchants and enhancing both the user and merchant experience.

#### Data Intelligence

- **Data Insight.** We provide our merchants a comprehensive data set to run their businesses through the Wish Merchant Dashboard. This dashboard helps our merchants improve their performance in terms of total impressions, overall sales, product assortment, service quality, fulfillment, shipping needs, and refunds, among others.
- Revenue Impact. Our proprietary, state-of-the-art data science capabilities are designed to display products to users who are most likely to buy them, in turn driving more revenue to merchants.

### Logistics

- Shipping Logistics. With ongoing changes to the global postal regulations and increase in cross-border sales volume, logistics
  has become paramount for small merchants to succeed in ecommerce. We have developed a number of logistics programs to
  provide a set of reliable cross-border logistics solutions at competitive costs for our merchants. We believe that ensuring a
  consistent delivery experience for our users increases value to our merchants by boosting sales volumes and minimizing
  returns
- Wish Local Pick-Up. Through our Wish Local partnerships, we enable our online merchants to send their inventory to our over 40,000 partner pick-up locations in close proximity to users to

allow for quicker, localized pickup. These Wish Local stores effectively give us a local warehousing and fulfillment footprint around the world without owning any real estate.

#### **Business Operations**

- Optimization Tools, Services, and Education. We provide tools, services, and ongoing education to our merchant base to help them improve their business operations and drive greater success.
- Merchant Support. Wish assists merchants with international compliance, payment processing, user support, and certain other services

## Data Science

Our technology and data science capabilities drive all aspects of our business. Our data advantage comes from our rich and growing data set of historical and recent user and merchant behaviors and transactions, and deep understanding of our more than 100 million monthly active users, over 500,000 merchants, and approximately 2 million items sold per day. All of this feeds into a proprietary data science algorithm that we continuously optimize for more intelligent insights and decision making.

Key examples of how we use data science to drive our business include:

- User Acquisition: We leverage the power of our proprietary data to make decisions on what to show to whom, when, and through which acquisition channel, with a focus on maximizing our return on marketing investment and conversion.
- **Lifetime Value Maximization:** Data science plays a critical role in maximizing lifetime value ("LTV") of our users and optimizing the initial user acquisition investment. We use data science to determine the allocation of marketing investment across different users, marketing channels, and user acquisition and re-engagement strategies.
- **Dynamic Pricing:** We utilize our data to dynamically vary prices across products to optimize conversion as well as our margin. This type of shopping environment is well suited to gauge user demand and conduct price discovery on a global level as well as on an individual user level. We take into account characteristics of the product and the user to estimate price sensitivity and vary pricing to achieve a margin target at the user and basket level, as opposed to on individual items.
- **User Personalization:** Our proprietary algorithms utilize a rich and growing data set of historical and recent user behaviors that includes browsing data, past transactions, reviews, and preferences noted on Wish, to display only the most relevant and personalized content. This data-driven approach enables efficient and enjoyable navigation and discovery on a mobile screen, creates purchase intent across a diverse set of products, and increases conversion to sales. For every 100 user sessions on our platform, an average of approximately 40 items get added to cart.
- **User-Generated Content:** Our platform offers mostly unbranded goods today. Our value-conscious users rely on user-generated content such as reviews, ratings, photos, and videos, rather than brand recognition, when making purchase decisions. This makes the user-generated content on our platform an important source of trust and quality for our largely unbranded product selection.

- Merchant Insights: Our platform includes a merchant dashboard with built-in analytics to help merchants sell more products and track their performance. Our data capabilities help merchants better understand user behavior and preferences, enabling them to operate more intelligently and efficiently.
- **ProductBoost:** ProductBoost is our native advertising tool for merchants, which helps them promote their products on our platform. We utilize data science to improve the performance of ProductBoost and help maximize the merchants' return on their ad spend. Approximately 28% of our merchants have used ProductBoost in 2020 to date.
- Logistics: We leverage data science to improve transparency and logistics operational efficiency for our merchants, while also aiming at reducing shipping time and improving delivery reliability for our buyers.
- User and Merchant Support: We use data to understand what a user or merchant is likely to need help with in order to improve the quality of support and maximize cost efficiency of providing such support.

### **Our Growth Strategy**

#### **Grow Our Base of Users**

- Continue to Acquire New Users. We are focused on growing our user base around the world. We currently serve users in over 100 countries. We estimate that there are over 1 billion households with income of less than \$75,000 around the world, excluding China and India.
- **Drive User Conversion.** We have over 13 million monthly active buyers<sup>13</sup> and over 100 million monthly active users on the platform. We will continue to drive greater user engagement and convert more active users on our platform to become active buyers by utilizing our data science and introducing more interactive and entertaining features.
- **Drive Profitable Lifetime Value from Existing Users.** We plan to continue to improve the engagement and monetization of our users on our platform to maximize their lifetime value. We plan to achieve this goal by:
  - Using our data science to drive personalization of our platform;
  - Continuing to offer attractive discounts and value, an entertaining user experience, and robust user-generated content;
  - Improving our ease of use by investing in our user support and logistics infrastructure to enable faster deliveries and localization of our platform.
- Expand Geographically. We intend to continue to expand our global footprint and enter new geographies and acquire new users in those markets.

### **Grow Our Base of Merchants and Offerings**

• Diversify Our Merchant Base and Expand Product Categories. As we grow and diversify our merchant base, we partner with other merchant platforms, such as PayPal, ShipStation and PlentyMarkets, to acquire additional merchants outside of China and expand our geographic reach and diversity across the merchant base. We will continue to expand product selection to provide more diversified products at competitive price points to our users.

Based on available internal data from January 2020 to June 2020.

- Broaden Merchant Services. We intend to add additional services to help our merchants grow their businesses and sell more on Wish. These services include educational content and resources as well as tools enabling merchants to promote their products in a variety of ways both on and off the Wish platform, efficiently manage working capital, and fulfill and ship their orders.
- Expand Logistics Platform. We plan to continue to expand our logistics platform and optimize our proprietary logistics programs. Through strategic partnerships with selected regional postal networks and commercial logistics partners, we aim to become an integrated part of the cross-border logistics value chain with our own logistics services and provide faster and more reliable delivery to our buyers. We are also exploring the opportunity to open up our logistics programs to merchants outside of Wish.
- **Grow Our Wish Local Offering.** We aim to continue to expand our Wish Local program to drive both online and offline commerce. Wish Local enables local brick-and-mortar stores to digitize their storefronts by uploading their in-store inventory on our platform and selling to our global user base. These stores in turn give us access to local warehousing and fulfillment footprint at more than 40,000 stores around the world by serving as Wish Pickup locations for online Wish orders.

#### Continue to Innovate and Extend Our Platform

- Monetize Brick-and-Mortar Stores. As we continue to grow our Wish Local program, we plan to explore different ways in which we can further enhance our value proposition and monetize our offering, whether in the form of additional traffic or sales to the store.
- Add New Product Categories. We plan to continue to diversify product categories and service offerings on Wish. These new product categories include consumer packaged goods ("CPG") products and in-person services primarily offered through our Wish Local merchants as well as affordable brands and off-price branded inventory.
- Expand to New Advertising Partners. We believe our user base presents a diverse and global audience that is attractive for advertisers across many end markets beyond ecommerce, including financial services, subscription services, and travel. We see a significant opportunity to connect these advertisers with our users to expand our advertising partners beyond our merchant base.
- **Grow First-Party Sales.** We leverage the data across our platform to identify key selling trends and consumer preferences to source branded and unbranded inventory directly from manufacturers and sell on our platform. While this revenue is a modest amount today, we will continue to experiment with and optimize first-party sales including our private label strategy.
- Open Our Commerce Platform to Additional Businesses. We plan to open the Wish commerce platform and capabilities to additional businesses not currently on our platform. For example, we intend to offer access to our logistics offering to any merchant who engages in ecommerce and wants a reliable, affordable, and global shipping solution. Similarly, we plan to offer our digital performance marketing solutions to any advertiser, regardless of their vertical or end market.

#### **Risks Associated With Our Business**

Investing in our Class A common stock involves substantial risk. These risks are described in the section titled "Risk Factors" immediately following this summary. Some of these risks include the following:

- Our efforts to acquire new users and engage existing users may not be successful or may be more costly than we expect, which could prevent us from maintaining or increasing our revenue.
- If we are unable to promote, maintain, and protect our brand and reputation and offer a compelling user experience, our ability
  to attract new users and engage with our existing base of users will be impaired.
- If we lose the services of Peter Szulczewski, our founder, Chief Executive Officer ("CEO"), and Chairperson, or other members of our senior management team, we may not be able to execute our business strategy.
- We rely on the Apple App Store and the Google Play Store to offer and promote our app. If we are unable to maintain a good relationship with such platform providers, if their terms and conditions change to our detriment, if we violate, or if a platform provider believes that we have violated, the terms and conditions of its platform, our business will suffer.
- Our brand, reputation, and business may be harmed if our merchants use unethical or illegal business practices, including the sale of counterfeit or fraudulent products or if our policies and practices with respect to such sales are perceived or found to be inadequate, and we may be impacted by the unlawful activity of merchants on our platform.
- We face intense competition, the market in which we operate is rapidly evolving and if we do not compete effectively, our results
  of operations and financial condition could be harmed.
- The COVID-19 pandemic may adversely affect our business and results of operations.
- Economic tension between the United States and China, or between other countries, may intensify and the United States, China, or other countries may adopt drastic measures in the future that impact our business.
- Any significant disruption in service on our platform or in our computer systems, some of which are currently hosted by thirdparty providers, could damage our reputation and result in a loss of users, which would harm our business and results of operations.
- The dual class structure of our common stock has the effect of concentrating voting control with our founder, CEO and Chairperson, Peter Szulczewski, and also with our employees; this will limit or preclude your ability to influence corporate matters. The holders of our outstanding Class B common stock will hold approximately % of the voting power of our outstanding capital stock following this offering, with % held by Mr. Szulczewski.
- · We may be involved in litigation matters or other legal proceedings that are expensive and time consuming.
- Our certificate of incorporation will provide that the Court of Chancery of the State of Delaware and the federal district courts of
  the United States of America will be the exclusive forums for substantially all disputes between us and our stockholders, which
  could limit our stockholders' ability to obtain a favorable judicial forum for disputes with us or our directors, officers or
  employees.

See "Risk Factors" immediately following this prospectus summary for a more thorough discussion of these and other risks and uncertainties we face.

## **Our Corporate Information**

We were incorporated in the state of Delaware in June 2010 as ContextLogic Inc., d/b/a "Wish." Our principal executive offices are located at One Sansome Street 40th Floor, San Francisco, California 94104. Our telephone number is (415) 432-7323. Our website address is www.wish.com. The information contained in, or accessible through, our website is not part of, and is not incorporated into, this prospectus, and investors should not rely on any such information in deciding whether to invest in our Class A common stock.

#### **Trademarks**

We use various trademarks, trade names, and design marks in our business, including Wish<sup>™</sup>, Wish Shopping Made Fun<sup>™</sup>, W<sup>™</sup>, WishPost<sup>™</sup>, Wish Pickup<sup>™</sup>, Wish Local<sup>™</sup>, ProductBoost<sup>™</sup>, and ContextLogic<sup>™</sup>. This prospectus also contains trademarks and trade names of other businesses that are the property of their respective holders. We do not intend our use or display of other companies' trademarks and trade names to imply a relationship with, or endorsement or sponsorship of us by, these other companies.

#### The Offering

Class A common stock offered by us

shares.

Class A common stock offered by the selling stockholders

shares.

Option to purchase additional shares

shares.

Class A common stock to be outstanding after this offering

shares (or shares if the underwriters exercise in full their option to purchase additional shares).

Class B common stock to be outstanding after this offering

shares.

Total Class A and Class B common stock to be outstanding after this offering

shares.

Use of proceeds

We estimate that our net proceeds from this offering will be approximately \$\\$\text{million}, or approximately \$\\$\text{million} if the underwriters exercise in full their option to purchase additional shares, assuming an initial public offering price of \$\\$\text{per share}, the midpoint of the price range set forth on the cover page of this prospectus, and after deducting underwriting discounts and commissions and estimated offering expenses payable by us.

We will not receive any of the proceeds from the sale of shares of Class A common stock by the selling stockholders.

The principal purposes of this offering are to increase our capitalization and financial flexibility and create a public market for our Class A common stock, obtain additional working capital, and facilitate our future access to the public equity markets to allow us to implement our business plan. We currently intend to use the net proceeds received by us from this offering for working capital, operating expenses, sales and marketing expenses to fund the growth of our business, and capital expenditures. In addition, we may use a portion of the net proceeds to acquire complementary businesses, products, services, or technologies. However, we have no current understandings, agreements or commitments for any specific material acquisitions at this time. For a more complete description of our intended use of the proceeds from this offering, see the section of this prospectus titled "Use of Proceeds."

## Voting rights

Each share of our Class A common stock entitles its holder to one vote on all matters to be voted on by stockholders generally. Each share of our Class B common stock entitles its holder to 20 votes on all matters to be voted on by stockholders generally.

Holders of our Class A common stock and Class B common stock will generally vote together as a single class, unless otherwise required by law or our amended and restated certificate of incorporation.

The holders of our outstanding Class B common stock will hold approximately % of the voting power of our outstanding capital stock following this offering and will have the ability to control the outcome of matters submitted to our stockholders for approval, including the election of our directors and the approval of any change in control transaction. For additional information, see the sections of this prospectus titled "Principal and Selling Stockholders" and "Description of Capital Stock."

All shares of Class B common stock will automatically convert, on a one-for-one basis, into shares of Class A common stock on the earliest of (i) the 10-year anniversary of the closing date of this offering, (ii) the date on which the number of outstanding shares of Class B common stock represents less than 5% of the aggregate combined number of outstanding shares of Class A common stock and Class B common stock, (iii) the date specified by a vote of the holders of a majority of the then outstanding shares of Class B common stock, and (iv) a date that is between 90 and 270 days, as determined by the board of directors, after the death or permanent incapacity of Peter Szulczewski, our founder, CEO, and Chairperson.

Following this offering, the holders of our outstanding Class B common stock will beneficially own % of our outstanding shares and % of the voting power of our outstanding shares and our directors, executive officers, greater than 5% stockholders and their respective affiliates will hold in the aggregate approximately % of the voting power of our outstanding capital stock following this offering, assuming no exercise of the underwriters' option

Concentration of ownership

to purchase additional shares of our Class A common stock from the selling stockholders in this offering. In addition, Peter Szulczewski, our founder, CEO, and Chairperson, will be able to exercise voting rights with respect to an aggregate of of Class B common stock, which will represent approximately % of the voting power of our outstanding capital stock following this offering.

See the section of this prospectus titled "Risk Factors" and the other information included in this prospectus for a discussion of factors you should consider before deciding to invest in our Class A common stock.

#### Risk factors

on

#### Proposed trading symbol

The number of shares of our Class A common stock and Class B common stock that will be outstanding after this offering is based shares of our Class A common stock (including our redeemable convertible preferred stock on an as-converted basis) and shares of our Class B common stock outstanding as of September 30, 2020, and excludes:

- shares of our Class B common stock issuable upon the exercise of options outstanding as of September 30, 2020 under our 2010 Stock Plan, with a weighted-average exercise price of approximately \$ per share:
- shares of Class B common stock issuable upon the exercise of options granted under our 2010 Stock Plan after September 30, 2020, with a weighted-average exercise price of approximately \$ per share:
- shares of our Class B common stock issuable under restricted stock units ("RSUs") that were outstanding as of September 30, 2020 for which the service condition was not satisfied as of September 30, 2020;
- shares of our Class B common stock issuable under RSUs that were granted after September 30, 2020;
- shares of our Class B common stock issuable upon exercise of a warrant at an exercise price of \$ per share: and
- shares of our common stock reserved for issuance under our equity compensation plans, consisting of:
  - shares of our Class A common stock that will be reserved for issuance under our 2020 Equity Incentive Plan,
  - shares of our Class B common stock reserved for issuance under our 2010 Stock Plan as of September 30, 2020, which will become available for issuance under our 2020 Equity Incentive Plan on the date of this prospectus,
  - shares of our Class A common stock that will be reserved for issuance under our 2020 Employee Stock Purchase Plan.

On the date of this prospectus we will cease granting awards under our 2010 Stock Plan. Our 2020 Equity Incentive Plan and 2020 Employee Stock Purchase Plan also provide for automatic

annual increases in the number of shares reserved thereunder (evergreen provisions), as more fully described in "Executive Compensation—2020 Equity Incentive Plan" and "Executive Compensation—2020 Employee Stock Purchase Plan."

Except as otherwise indicated, all information in this prospectus assumes or gives effect to the following:

- the filing and effectiveness of our amended and restated certificate of incorporation in Delaware and the adoption of our amended and restated bylaws, each of which will occur immediately prior to the completion of this offering;
- the reclassification of all outstanding shares of our common stock into an equivalent number of shares of our Class B common stock, as if such reclassification had occurred immediately prior to the completion of this offering;
- the automatic conversion of all outstanding shares of redeemable convertible preferred stock into an aggregate of shares of our Class A common stock, the conversion of which will occur immediately prior to the completion of this offering;
- the automatic conversion of shares of our Class B common stock into an equal number of shares of our Class A common stock upon their sale by the selling stockholders in our initial public offering;
- no exercise of the outstanding options or settlement of the RSUs described above;
- the exercise of an outstanding warrant to purchase outstanding as of September 30, 2020 into shares of our Class A common stock on a net-exercise basis (assuming an initial public offering price of \$ per share, the midpoint of the price range set forth on the cover page of this prospectus), which we expect to occur immediately prior to the completion of this offering;
- the issuance of shares of our Class B common stock subject to RSUs for which the service condition was satisfied as of September 30, 2020, and for which we expect the performance condition to be satisfied in connection with this offering;
- the effectiveness of a -for-1 stock split of our common stock to be effected on ; and
- no exercise by the underwriters of their option to purchase up to an additional shares of our Class A common stock.

### **Summary Consolidated Financial and Other Data**

You should read the summary consolidated financial and other data in conjunction with the sections of this prospectus caption titled "Management's Discussion and Analysis of Financial Condition and Results of Operations" and "Selected Consolidated Financial and Other Data" and the consolidated financial statements and related notes found elsewhere in this prospectus.

The summary consolidated statements of operations and comprehensive loss data for the years ended December 31, 2017, 2018, and 2019, and the summary consolidated balance sheet data as of December 31, 2019, are derived from our audited consolidated financial statements and related notes included elsewhere in this prospectus. The following tables also show certain operational metrics and non-GAAP financial measures. Our historical results and key metrics are not necessarily indicative of future results.

	Year Ended December 31,					
	2017 <sup>(3)</sup> 2018 <sup>(3)</sup>				2019	
Summary Consolidated Statements of Operations	(in millions, except share and per share data)					
and Comprehensive Loss Data:	•	4.404	Φ.	4 700	Φ.	4.004
Revenue	\$	1,101	\$	1,728	\$	1,901
Cost of revenue <sup>(1)</sup>		205		278		443
Gross profit		896		1,450		1,458
Operating expenses: <sup>(1)</sup>						
Sales and marketing		989		1,576		1,463
Product development		28		45		74
General and administrative		26		52		65
Total operating expenses		1,043		1,673		1,602
Loss from operations		(147)		(223)		(144)
Other income (expense), net:						
Interest and other income (expense), net		10		15		19
Remeasurement of redeemable convertible preferred stock						
warrant liability		(70)				(3)
Loss before provision for income taxes		(207)		(208)		(128)
Provision for income taxes		_		_		1
Net loss and comprehensive loss		(207)		(208)		(129)
Deemed dividend to redeemable convertible preferred stockholders		(40)				(7)
Net loss attributable to common stockholders	\$	(247)	\$	(208)	\$	(136)
Net loss per share attributable to common stockholders, basic and						
diluted <sup>(2)</sup>	\$	(24.60)	\$	(20.20)	\$	(13.07)
Weighted-average shares used in computing net loss per share						
attributable to common stockholders, basic and diluted(2)	10	,038,841	10	,296,604	10	,404,562
Pro forma net loss per share attributable to common stockholders,					' <u></u>	
basic and diluted (unaudited)(2)					\$	
Weighted-average shares used in computing pro forma net loss per						
share attributable to common stockholders, basic and diluted (unaudited) <sup>(2)</sup>						
(undultou)						

- Cost of revenue and operating expenses include stock-based compensation expense of \$8 million, \$2 million, and \$2 million for the year ended December 31, 2017, 2018, (1) and 2019, respectively. Following this offering, our future operating expenses, particularly in the quarter in which this offering is completed, will include substantial stock-based compensation expense with respect to our RSUs as well as any other stock-based awards we may grant in the future.
- See Note 11 and Note 12 of the notes to our consolidated financial statements included in this prospectus for a description of how we compute basis and diluted net loss (2) per share attributable to common stockholders and pro forma basic and diluted net loss per share attributable to common stockholders.

  On January 1, 2018, we adopted Accounting Standards Update No. 2014-09, Revenue from Contracts with Customers ("Topic 606"), on a modified retrospective basis.
- (3) Accordingly, our audited consolidated financial statements for 2018 and 2019 were reported under Topic 606. Our audited consolidated financial statements for 2017 were reported under Accounting Standards Codification ("ASC") Topic 605, Revenue Recognition ("Topic 605"). See Note 2 of the notes to our consolidated financial statements included in this prospectus.

		As of December 31, 2019		
	Actual	Pro Forma <sup>(1)</sup>	Pro Forma As Adjusted <sup>(2)(3)</sup>	
Summary Consolidated Balance Sheet Data:		(in millions)		
Cash, cash equivalents and marketable securities	\$ 1,078	\$	\$	
Working capital	134			
Total assets	1,366			
Redeemable convertible preferred stock warrant liability	127			
Redeemable convertible preferred stock	1,536			
Total stockholders' deficit	\$(1,439)			

- (1) Reflects, on a pro forma basis, (i) the automatic conversion of all of our outstanding shares of redeemable convertible preferred stock into an aggregate of shares of Class A common stock as of December 31, 2019, (ii) the issuance of shares of Class A common stock purchase shares of Series B redeemable convertible preferred stock outstanding at December 31, 2019 at \$ shares of Class A common stock upon the cashless exercise of a warrant to per share (assuming an initial public offering price of \$ per share, the midpoint of the price range set forth on the cover page of this prospectus) and (iii) stock-based compensation expense of approximately \$ million associated with RSUs subject to service-based and performance-based vesting conditions, which we will recognize upon the completion of this offering, as further described in Note 2 to our consolidated financial statements included elsewhere in this prospectus, which is reflected as an increase to additional paid-in capital and accumulated deficit.
- (2) Reflects, on a pro forma as adjusted basis, (i) the pro forma adjustments described in footnote (1) above and (ii) the sale and issuance by us of shares of Class A common stock in this offering at the assumed initial public offering price of \$ per share, the midpoint of the price range set forth on the cover page of this prospectus, after deducting underwriting discounts and commissions and estimated offering expenses payable by us. A \$1.00 increase or decrease in the assumed initial public offering price of \$ per share, the midpoint of the offering estimated price range set forth on the cover page of this prospectus, would increase or decrease, as applicable, each of cash, cash equivalents and marketable securities, total stockholders' equity and total capitalization by \$ million, assuming that the number of shares offered by us, as set forth on the cover page of this prospectus, remains the same, and after deducting underwriting discounts and commissions payable by us. We may also increase or decrease the number of shares we are offering. Each increase or decrease of 1.0 million in the number of shares offered by us would increase or decrease, as applicable, each of our cash, cash equivalents and marketable securities, total stockholders' equity and total capitalization by \$ that the assumed initial public offering price remains the same, and after deducting underwriting discounts and commissions payable by us.

  The pro forma as adjusted summary consolidated balance sheet data above is illustrative only and will be adjusted based on the actual initial public offering price and other
- (3) terms of our initial public offering determined at pricing.

#### Other Financial Information and Data

In addition to the measures presented in our summary consolidated financial statements, we use the following key metrics and other financial information to measure our performance, identify trends affecting our business, and make strategic decisions.

		Year Ended December 31,				
	2015	2016	2017	2018	2019	
		(in millions, except percentages)				
Monthly Active Users	21	30	49	73	90	
LTM Active Buyers	19	32	54	66	64	
Adjusted EBITDA	\$ (502)	\$(132)	\$ (135)	\$(211)	\$ (127)	
Adjusted EBITDA Margin	(349)%	(30)%	(12)%	(12)%	(7)%	
Free Cash Flow	\$ (305)	\$ 15	\$ 134	\$(114)	\$ (71)	

See the section titled "Selected Consolidated Financial and Other Data—Other Financial Information and Data" for additional information about our key business metrics.

#### **Non-GAAP Financial Measures**

#### Adjusted EBITDA and Adjusted EBITDA Margin

In this prospectus, we provide Adjusted EBITDA, a non-GAAP financial measure that represents our net loss before interest and other income (expense), net (which includes foreign exchange gain or loss, gain or loss on one time transactions recognized), income tax expense, and depreciation and amortization, adjusted to eliminate stock-based compensation expense, remeasurement of redeemable convertible preferred stock warrant liability, and add back certain recurring other items. Additionally, in this prospectus, we provide Adjusted EBITDA Margin, a non-GAAP financial measure that represents Adjusted EBITDA divided by revenue. See "Selected Consolidated Financial and Other Data—Non-GAAP Financial Measures" for more information and for a reconciliation of Adjusted EBITDA to net loss, the most directly comparable financial measure calculated and presented in accordance with U.S. GAAP.

### Free Cash Flow

In this prospectus, we also provide Free Cash Flow, a non-GAAP financial measure that represents net cash provided by (used in) operating activities less purchases of property and equipment. See "Selected Consolidated Financial and Other Data—Non-GAAP Financial Measures" for more information and for a reconciliation of Free Cash Flow to net cash provided by (used in) operating activities, the most directly comparable financial measure calculated and presented in accordance with U.S. GAAP.

#### **RISK FACTORS**

Investing in our Class A common stock involves a high degree of risk. Before deciding whether to purchase shares of our Class A common stock, you should consider carefully the risks and uncertainties described below, our consolidated financial statements and related notes, and all of the other information in this prospectus. If any of the following risks actually occurs, our business, financial condition, results of operations, and prospects could be adversely affected. As a result, the price of our Class A common stock could decline and you could lose part or all of your investment.

#### Risks Related to Our Business and Industry

Our efforts to acquire new users and engage existing users may not be successful or may be more costly than we expect, which could prevent us from maintaining or increasing our revenue.

Our success depends on our ability to attract new users and engage existing users in a cost-effective manner. In order to acquire and engage users, we must, among other things, promote and sustain our platform and provide high-quality products, user experiences, and service. Our marketing efforts currently include various initiatives and consist primarily of digital marketing on a variety of social media channels, such as Facebook, search engine optimization on websites, such as Google, Bing, and Yahoo!, various branding strategies, such as our relationship with the Los Angeles Lakers and social influencers, and mobile "push" notifications, text messaging and email. For the year ended December 31, 2019, we spent \$1.5 billion on sales and marketing, representing 77% of our revenue. We anticipate that sales and marketing expenses will continue to comprise a substantial majority of our overall operating costs for the foreseeable future. We have historically acquired a significant number of our users through digital advertising on platforms and websites owned by Facebook and Google, which may terminate their agreements with us anytime. Our investments in sales and marketing may not effectively reach potential users, potential users may decide not to buy through us, or user spend on our platform may not yield the intended return on investment, any of which could negatively affect our financial results.

Many factors, some of which are beyond our control, may reduce our ability to acquire and engage with users, including those described in this "Risk Factors" section and the following:

- system updates to app stores and advertising platforms such as Facebook and Google, including adjustments to algorithms that may decrease user engagement or negatively affect our ability to target a broad audience;
- · changes in advertising platforms' pricing, which could result in higher advertising costs;
- changes in digital advertising platforms" policies, such as those of Facebook and Google, that may delay or prevent us from advertising through these channels, which could result in reduced traffic to and sales on our platform;
- changes in search algorithms by search engines;
- inability of our email marketing messages to reach the intended recipients' inbox;
- · decline in popularity of, or governmental restrictions on, social media platforms where we advertise;
- the development of new search engines or social media sites that reduce traffic on existing search engines and social media sites;
- · consumer behavior changes as a result of COVID-19; and
- products listed by merchants on our platform that are the subject of adverse media reports, regulatory investigations, or other negative publicity.

If we are unable to promote, maintain, and protect our brand and reputation, and offer a compelling user experience, our ability to attract new users and engage with our existing base of users will be impaired.

We believe that maintaining our brand and reputation will be critical to attracting new users and encouraging users to transact on our platform. In addition to targeted online marketing, we spend a considerable amount of resources on promoting our brand and reputation. For example, we have recently begun to invest in additional off-line marketing activities. Our brand promotion activities may not be successful or cost effective, and to the extent that these activities yield increased revenue, the increased revenue may not offset the expenses we incur. If we do not successfully drive brand awareness, we may fail to attract new users or increase engagement with existing users and our business may not grow or may decline, all of which could harm our business, financial condition, results of operations, and prospects.

Our ability to provide a high-quality user experience is also highly dependent on external factors over which we may have little or no control, including, without limitation, the reliability and performance of our merchants and third-party carriers. If our users are dissatisfied with the quality of the products sold on our platform, the customer service they receive or their overall user experience, or if our merchants or third-party carriers cannot deliver products to our users in a timely manner or at all, our users may stop purchasing products on our platform. Our users may also become dissatisfied with their user experience if they are unable to receive timely customer service, and because we rely in large part on an automated customer service system, it is possible our users could become dissatisfied with our customer service. We also rely on merchants for information, including product characteristics, descriptions, images, and availability that may be inaccurate or misleading. Our failure to provide our users with high-quality products and high-quality user experiences for any reason could substantially harm our reputation and adversely impact our efforts to develop Wish as a trusted brand, which could have a material adverse effect on our business, financial condition, results of operations, and prospects.

In addition, we may be subject to unfavorable publicity that would create a public perception that non-authentic, counterfeit, dangerous, illegal, or defective goods are sold on our platform, or that our policies and practices are insufficient to deter or respond to such conduct. Even if these claims are factually incorrect or based on isolated incidents, it could damage our reputation, diminish the value of our brand, draw governmental or regulatory scrutiny or action, undermine our trust and credibility, or have a negative impact on our ability to attract new users, or discourage our existing users from continuing to transact on our platform. We may also be subject to negative media regarding our privacy or cyber security practices, terms of service, product quality, litigation or regulatory activity, the sale of illicit or dangerous goods, other unauthorized actions by merchants on our platform, or the actions of other companies that provide similar services to ours, which may adversely affect our reputation, business, and financial results.

If we are unable to offer features and attract merchants to list products that keep pace with changing consumer preferences, our business, financial condition, and results of operations may be materially and adversely affected.

Constantly changing consumer preferences have affected and will continue to affect the ecommerce industry. We must stay ahead of emerging consumer preferences and anticipate product trends that will appeal to existing and potential users. Our users choose to purchase products due in part to the attractive prices that we offer, and they may choose to shop elsewhere if we cannot match the prices offered by other websites and platforms or by brick-and-mortar stores. If our users do not find our platform entertaining and are not shown desired products on our platform at attractive prices, they may lose interest in us, which in turn may materially and adversely affect our business, financial condition, and results of operations.

We rely on the Apple App Store and the Google Play Store to offer and promote our app. If we are unable to maintain a good relationship with such platform providers, if their terms and conditions change to our detriment, if we violate, or if a platform provider believes that we have violated, the terms and conditions of its platform, our business will suffer.

A significant portion of our users download our mobile app through the Apple App Store and the Google Play Store, and over 90% of our user activity and purchases occur on our mobile app.

We are subject to the policies and terms of service of these third-party platforms, which govern the promotion, distribution, content, and operation of our app on the platform. Each platform provider has broad discretion to change and interpret its terms of service and other policies with respect to us and other developers, and those changes may be unfavorable to us. A platform provider may also add fees associated with access to and use of its platform, alter how we are able to advertise on the platform, prevent our app from being offered on their platform, change how the personal information of its users is made available to application developers on the platform, or limit the use of personal information for advertising purposes.

If we violate, or a platform provider believes we have violated, its terms of service (or if there is any change or deterioration in our relationship with these platform providers), that platform provider could limit or discontinue our access to the platform. A platform provider may also object to content created by our merchants, such as drug paraphernalia or adult content, and our perceived distribution or advertisement of such content may cause a platform provider to view us in a negative light or take other adverse actions against us. For example, platform providers have warned application developers on their platform, including Wish, that providing content related to drug paraphernalia or adult content could cause such platform providers to remove the apps from their platforms. While we believe that we have complied with platform providers' requirements, they may introduce additional requirements in the future. If a platform provider establishes more favorable relationships with one or more of our competitors or such platform provider determines that we are a competitor, our access to a platform may be limited or discontinued entirely. Any limit or discontinuation of our access to any platform could adversely affect our business, financial condition, and results of operations.

In the past, some of these platforms have been unavailable for short periods of time. This and other changes, bugs, or technical system issues could degrade the user experience on our platform. There may also be changes to mobile hardware or software technology that make it more difficult for our users to access and use our platform on their mobile devices, which could adversely affect our user growth and user engagement. If any of these events recurs on a prolonged, or even short-term basis, or other similar issues arise that impact users' ability to access our app or use mobile devices, our business, financial condition, results of operations, or reputation may be harmed.

#### Our quarterly and annual operating results may fluctuate, which could cause our stock price to decline.

Our quarterly and annual operating results may fluctuate for a variety of reasons, many of which are beyond our control. These reasons include those described in this "Risk Factors" section as well as the following:

- · the amount and timing of our sales and marketing costs;
- · our user acquisition strategies;
- · traffic on our platform;
- selling prices on our platform and the percentage of revenue we retain from the sale of products;

- mix of products listed on our platform;
- fraud, including the sale of counterfeit goods, and refunds, including our response to these areas;
- continued impact from COVID-19, including the effects of increased online activity and government stimulus programs;
- · the level of merchant advertising on our platform;
- disruptions in supply or shipment of products listed on our platform, especially from China where most of our merchants are currently located;
- the actions of app stores and advertising platforms such as Facebook and Google;
- seasonality;
- · fluctuations in exchange rates;
- · the amount and timing of our other operating expenses;
- the impact of competitive developments and our response to those developments;
- · changes in carrier policies and pricing and resulting higher logistics costs;
- · actual or perceived disruptions or defects in our platform, such as data security breaches or outages;
- · changes in laws and regulations that impact our business;
- · changes in tax laws in the jurisdictions in which we operate; and
- · general political, economic, and market conditions, particularly those affecting our industry.

Fluctuations in our quarterly and annual operating results may cause those results to fall below the expectations of analysts or investors, which could cause the price of our Class A common stock to decline. Fluctuations in our results could also cause a number of other problems. For example, analysts or investors might change their models for valuing our Class A common stock, we could experience short-term liquidity issues, our ability to retain or attract key personnel may diminish, and other unanticipated issues may arise.

In addition, we believe that our quarterly and annual operating results may vary in the future and that period-to-period comparisons of our operating results may not be meaningful. For example, our historical growth may have overshadowed the seasonal effects on our historical operating results. These seasonal effects may become more pronounced over time, which could also cause our operating results to fluctuate. You should not rely on the results of one quarter or one year as an indication of future performance.

If we lose the services of Peter Szulczewski, our founder, CEO, and Chairperson, or other members of our senior management team, we may not be able to execute our business strategy.

Our success depends in a large part upon the continued service of our founder, CEO, and Chairperson, Peter Szulczewski, who is critical to our vision, strategic direction, culture, products, and technology. The loss of our founder and CEO, even temporarily, could harm our business.

Additionally, we rely on the continued service of our senior management team, key employees, and other highly skilled personnel. The failure to properly manage succession plans, develop leadership talent, and/or the loss of services of senior management or other key employees, could

significantly delay or prevent the achievement of our objectives. From time to time, there may be changes in our senior management team resulting from the hiring or departure of executives, which could disrupt our business. We do not have long-term employment agreements with any of our key personnel, and do not maintain any "key person" life insurance policies. The loss of the services of one or more of our senior management or other key employees for any reason could adversely affect our business, financial condition, and results of operations, and require significant amounts of time, training and resources to find suitable replacements and integrate them within our business, and could affect our corporate culture.

#### We rely on our merchants to provide a good experience to our users.

Negative publicity or sentiment as a result of complaints about merchants selling on our platform could reduce our ability to attract users, discourage users from making additional purchases on our platform, or otherwise damage our reputation. A perception that our levels of responsiveness and support for our users are inadequate could have similar results. In some situations, we may choose to reimburse users for their purchases to help avoid harm to our reputation, but we may not be able to recover the funds we expend for those reimbursements.

Disruptions in the operations of a substantial number of our merchants, to the extent they are caused by events that are beyond their control, such as interruptions in order or payment processing, transportation disruptions, natural disasters, pandemics, inclement weather, terrorism, public health crises, or political unrest, could result in negative experiences for a substantial number of our users, which could harm our reputation and adversely affect our business. For example, during the initial outbreak of COVID-19, our merchants based in China experienced supply interruptions and delivery delays. If there are subsequent or further increases in the number of COVID-19 outbreaks in China or elsewhere, our merchants may experience additional disruptions to their supply and restrictions on their ability to deliver products to our users in a timely manner, which could harm our business.

Our brand, reputation, and business may be harmed if merchants on our platform use unethical or illegal business practices, including the sale of hazardous, counterfeit, fraudulent, or illegal products, or if our policies and practices with respect to such sales are perceived or found to be inadequate, and we may be impacted by the unlawful activities of merchants on our platform.

It is important that both merchants and users have confidence in the transactions they are completing on our platform. Merchants on our platform have in the past, and may in the future, engage in illegal or unethical business practices. Allegations or findings of such illegal or unethical business practices by merchants on our platform could harm our brand, reputation, and business. Our policies promote legal and ethical business practices, such as prohibiting false or misleading selling information and the sale of hazardous, counterfeit, fraudulent, or illegal products. However, we do not control merchants or their business practices and cannot ensure their compliance with our policies. If merchants on our platform engage in illegal or unethical business practices or are perceived to do so, we may receive negative publicity and our brand and reputation may be harmed. If our policies are violated by merchants, or if our policies and practices or responses to such conduct are perceived as or found to be inadequate by regulators or law-enforcement agencies, it could harm our brand, reputation, and business, including by subjecting us to government inquiries, investigations, or enforcement actions, as well as potential civil or criminal liabilities, or requiring changes to our policies and practices with respect to illegal or unethical business practices that could lower our revenue, increase our costs, make our platform less user-friendly, or otherwise adversely impact our business. For example, during the initial outbreak of COVID-19, a small number of merchants created listings of personal protective equipment and other health-related products that regulators deemed to violate consumer protections related to pricing and advertising. Though these listings were posted by

merchants in violation of our policies, Wish has received and expects to continue to receive inquiries and demands from regulators regarding these listings. We believe that we have responded to and resolved these particular inquiries and demands, but we frequently receive and respond to inquiries and demands from regulators and law-enforcement agencies, and we expect to continue to receive more inquiries and demands in the future.

Merchants on our platform have in the past, and may in the future, engage in fictitious transactions or collaborate with third parties in order to artificially inflate their sales records and search results rankings. Such activity may frustrate other merchants by enabling the perpetrating merchants to be favored over legitimate merchants, and may harm users by misleading them to believe that a merchant is more reliable or trustworthy than the merchant actually is. This activity may also result in inflated MAUs and other key metrics by which we measure our performance. Although we have implemented policies and practices to detect and penalize merchants who engage in fraudulent activities on our platform, there can be no assurance that such policies and practices will be effective in preventing fraudulent transactions. Any of these activities may adversely affect our brand, reputation, and business. If a governmental authority determines that we have aided and abetted the infringement or sale of counterfeit goods or if legal changes result in us potentially being liable for actions by merchants on our platform, we could face regulatory, civil, or criminal penalties. Successful claims by third-party rights owners could require us to pay substantial damages or refrain from permitting any further listing of the relevant items. These types of claims could force us to modify our business practices, which could lower our revenue, increase our costs, or make our platform less user friendly. Moreover, public perception that counterfeit or other unauthorized items are common in our platform, even if factually incorrect, could result in negative publicity and damage to our reputation and brand.

# Our merchants rely on third-party carriers and transportation providers as part of the fulfillment process, and these third parties may fail to adequately serve our users.

We rely on merchants to properly and promptly prepare products ordered by our users for shipment and our logistics program relies on third-party carriers to deliver products as well as third parties to consolidate packages for shipping. Any failure by merchants to timely prepare such products for shipment or any delay by third-party carriers to deliver the products will have an adverse effect on the fulfillment of user orders, which could negatively affect the user experience and harm our business and results of operations. Any increase in shipping costs, any significant shipping difficulties, disruptions or delays, or any failure by our merchants to deliver products in a timely manner or to otherwise adequately serve our users, could damage our reputation and brand, and may harm our business. For example, in the first quarter of 2018, PostNord, the postal service in Sweden, suspended delivery of packages coming from outside the European Union as it evaluated imposing processing fees and collection of taxes, which resulted in a decrease in sales in Sweden. In addition, during the initial outbreak of COVID-19, our cross-border logistics function was severely impacted in terms of both disrupted processing capabilities and increased costs, which resulted in a decrease in sales due to higher logistics costs and higher refund rates due to poor performance. Additionally, during the initial outbreak of COVID-19, our merchants based in China experienced supply interruptions and delivery delays.

Historically, our merchants in China have benefitted from lower shipping costs due to the Universal Postal Union Treaty ("UPU"). Certain expected changes to UPU postal rates that went into effect in July 2020 are likely to increase the shipping rates our merchants incur to ship products from China. The actions we have taken in our logistics program to mitigate these increased costs may not be successful over the long term. If there are increases in shipping costs, the sales price of products on our platform could increase, which could reduce the volume of transaction activity on our platform to decrease and may consequently have a negative impact on our results of operations.

If merchants on our platform experience any recalls, product liability claims, or government or user concerns about product safety with respect to products sold on our platform, our reputation and sales could be harmed.

Our merchants are subject to regulation by the U.S. Consumer Product Safety Commission and similar state and international regulatory authorities, and their products sold on our platform could be subject to involuntary recalls and other actions by these authorities. Concerns about product safety including concerns about the safety of products manufactured in developing countries, could lead to recalls of selected products sold on our platform. Recalls and government or user concerns about product safety could harm our reputation and reduce sales, either of which could have a material adverse effect on our business, results of operations, financial condition, and prospects.

Additionally, laws and regulations relating to platform liability, such as the EU's Market Surveillance Regulation, have been passed or implemented or are currently being considered by policymakers and regulators, which will place additional responsibilities on marketplaces to screen and monitor certain content or products. If these laws are passed or regulations implemented, we could be subject to product liability claims for products that are listed on our platform. Additionally, we may need to implement additional screening and monitoring systems in order to assess the products listed on our platform. These laws and regulations could cause our expenses to increase and require us to respond to product liability claims, which could have a material impact on our business and results of operations. Further, we may be subject to product liability claims where merchants lack sufficient assets or are not reachable, which could be costly to defend in the aggregate.

## We generate a portion of our revenue from merchant advertising on our platform. A reduction in advertising spend by merchants could harm our business.

We have implemented new features on our platform, such as ProductBoost, which allow merchants to promote their listings to our users. In addition to generating revenue from merchants, these advertisements may also result in increased purchases by users. However, not all merchants on our platform may agree with us on the value of these new features and may not use ProductBoost, and some of our merchants could react negatively to these new features. During the initial outbreak of COVID-19, merchant advertising declined due to the shutdown of business activity in China. If we are unable to monetize existing and new features for merchants, it could have a significant impact on our business, financial condition, and prospects.

## We plan to continue our efforts to improve our logistics programs and enable faster and more reliable delivery in order to help grow our business and generate revenue, but those efforts may not be effective.

We have worked to improve our logistics programs and to streamline our processes in order to provide a more consistent and reliable experience for our users through programs such as Wish Express and Wish Local. However, we still rely on third-party carriers for delivery and we are still in the process of establishing reliable long term agreements with such carriers both in the United States and worldwide. If we are not able to negotiate acceptable pricing, service level requirements, and other terms with these carriers, or these carriers experience performance problems or other issues, it could negatively impact our results of operations and our users' experience. For example, due to COVID-19, global logistics has experienced longer delivery times.

We have also recently developed and experimented with different logistics programs in order to monetize our logistics platform. This is a relatively new business initiative for us. If we are unable to consistently generate revenue from our logistics platform or offer logistics services that are appealing to merchants and users, or if changes in carrier policies and pricing result in higher logistics costs, it could have a material impact on our business, financial condition, and prospects.

#### Building out our Wish Local program may be costly and time intensive and we may not receive the expected benefits.

In 2019, we introduced Wish Local, a program that develops partnerships with local brick-and-mortar stores that can serve as local pickup and delivery locations for users' orders. In addition, Wish Local merchants can sign up to our Sell on Wish feature and upload their in-store inventory on Wish for local pick-up or delivery. The process of reaching out to and entering into relationships with these retailers can be time intensive and costly because we must evaluate and approve each retailer individually prior to them becoming part of the Wish Local program. Therefore, growing Wish Local may be more expensive and time consuming than we have estimated. Also, users may not use the Wish Local program as much as we expect, which would delay or prevent any expected benefits. For example, users may curtail their use of Wish Local due to health concerns regarding COVID-19 or stay-at-home mandates, which could slow growth of this program.

Our terms of service require Wish Local retailers to meet certain service level requirements with respect to holding and delivering Wish orders to users. If these retailers do not comply with these service level requirements, our reputation may be harmed. Additionally, we may need to implement monitoring systems to confirm that the Wish Local retailers are complying with service level requirements and to prevent fraudulent activities by these retailers. Implementing these systems may prove to be costly and time intensive.

#### Seasonality may cause fluctuations in our operating results.

Our operating results are seasonal in nature because our transaction volume is affected by traditional retail selling periods that impact sales on our platform. Our historical growth may have reduced or outweighed seasonal effects on our past financial results. However, seasonal effects may become more pronounced over time, which could cause fluctuations in our financial results. For example, sales on our platform have historically peaked in the fall and user activity begins to slow down in December as it may be too late to place orders for holiday delivery. Additionally, we have experienced some slowdown in merchant activity in late January or early February due to our China-based merchants celebrating the Chinese New Year holiday.

## We face intense competition, the market in which we operate is rapidly evolving, and if we do not compete effectively, our results of operations and financial condition could be harmed.

Our market is highly competitive and characterized by rapid changes in technology and consumer sentiment. Competition in our industry has intensified, and we expect this trend to continue as the list of our competitors grows. This competition, among other things, affects our ability to attract new users and engage our existing users.

We compete with ecommerce platforms and other retailers for merchants on our platform and merchants can list their goods on a number of ecommerce platforms, such as Amazon.com, Alibaba, and Shopify.

There are various factors that affect how merchants engage with our platform, including:

- the number and engagement of users on our platform;
- · our fees:
- · our brand awareness;
- · our reputation;
- · the quality of our services; and
- · the functionality of our platform.

We also compete with retailers for the attention of users. A user has the choice of shopping with any online or offline retailer, whether large marketplaces, such as Amazon.com, Alibaba, and Shopify, as well as more traditional discount retailers, such as Walmart and Target, and discount retailers that offer heavily discounted and off-season merchandise, such as Dollar Store and TJMaxx, or local stores or other venues or marketplaces. Many of these competitors offer low-cost or free shipping, fast shipping times, favorable return policies, and other features that may be difficult or impossible for our merchants to match.

There are various factors that affect how users engage with our platform, including:

- · our brand awareness and recognition;
- · our reputation;
- · the prices of goods sold on our platform;
- · the functionality of our platform;
- · ease of payment;
- · shipping terms: and
- · the breadth of the products sold on our platform.

Some of our competitors have, and potential competitors may have, longer operating histories, greater financial, technical, marketing, institutional and other resources, faster shipping times, lower-cost shipping, larger databases, greater name and brand recognition, or a larger base of users or merchants than we do. For example, Google or Facebook could enter the ecommerce space and they have significantly more resources and users than we do. They may devote greater resources to the development, marketing, and promotion of their services than we do, and they may offer lower pricing or free shipping to the users on their platforms. These factors may allow our competitors to derive greater revenue and profits from their existing user and merchant bases, acquire users at lower costs or respond more quickly than we can to new or emerging technologies and changes in trends and consumer shopping behavior. If we are unable to compete successfully, or if competing successfully requires us to expend greater resources, our financial condition and results of operations could be adversely affected.

## We have a limited operating history at our current scale, which may make it difficult to evaluate our business and future prospects.

We began commercial operations in 2010 and have a limited history of generating revenue at our current scale. As a result of our relatively short operating history at our current scale, we have limited financial data that can be used to evaluate our business and future prospects. Any evaluation of our business and prospects must be considered in light of our limited operating history, which may not be indicative of future performance. Because of our limited operating history, we face increased risks, uncertainties, expenses, and difficulties, including the risks and uncertainties discussed in this section.

### We have a history of operating losses and we may not achieve or maintain profitability in the future.

Since our inception in 2010, we have incurred net losses each year. We incurred net losses of \$207 million, \$208 million, and \$129 million for the years ended December 31, 2017, 2018, and 2019, respectively. As of December 31, 2019, we had an accumulated deficit of approximately \$1.4 billion. We may not achieve or maintain profitability in the future. Our operating expenses may continue to increase in the future as we increase our efforts to expand our user base, continue to invest in the research and development of our technologies and service offerings and begin to operate as a public company. These efforts may be more costly than we expect and we may not be able to

increase our revenue to offset our operating expenses. Our revenue growth slowed for the year ended December 31, 2019 and may slow again, or our revenue may decline for a number of other possible reasons, including increased competition, a decrease in the growth or reduction in size of our overall market, or if we fail for any reason to capitalize on growth opportunities.

### If we fail to effectively manage our growth, our business, financial condition, and operating results could be harmed.

We have experienced rapid growth in our business, in the number of merchants and users, and the number of countries in which we have merchants and users, and we plan to continue to grow in the future, both in the United States and abroad. Our recent and historical growth should not be considered indicative of our future performance. We have encountered in the past, and will encounter in the future, risks and uncertainties frequently experienced by growing companies in rapidly changing industries. If our assumptions regarding these risks and uncertainties, which we use to plan and operate our business, are incorrect or change, or if we do not address these risks adequately, our financial condition and operating results could differ materially from our expectations, our growth rates may slow, and our business would be adversely impacted.

Additionally, the growth of our business places significant demands on our management team and pressure to expand our operational and financial infrastructure. For example, we may need to continue to develop and improve our operational, financial and management controls and enhance our reporting systems and procedures. If we do not manage our growth effectively, the increases in our operating expenses could outpace any increases in our revenue and our business could be harmed.

### Use of social media, emails, and text messages may adversely impact our reputation or subject us to fines or other penalties.

We use social media, emails, and text messages as part of our omnichannel approach to marketing. As laws and regulations rapidly evolve to govern the use of these channels, the failure by us, our employees or third parties acting on our behalf or at our direction to abide by applicable laws and regulations in the use of these channels could adversely affect our reputation or subject us to fines, other penalties, or lawsuits. Although we continue to update our practices as these laws change over time, we may be subject to lawsuits alleging our failure to comply with such laws. In addition, our employees or third parties acting on our behalf or at our direction may knowingly or inadvertently use social media, including through advertisements, in ways that could lead to the loss or infringement of intellectual property, as well as the public disclosure of proprietary, confidential, or sensitive personal information of our business, employees, users, merchants, or others. Any such inappropriate use of social media, emails, and text messages could also cause reputational damage.

Our users may engage with us online through social media platforms, including Facebook, Instagram, and Twitter, by providing feedback and public commentary about all aspects of our business. Information concerning us or our merchants, whether accurate or not, may be posted on social media platforms at any time and may have a disproportionately adverse impact on our brand, reputation, or business. The harm may be immediate without affording us an opportunity for redress or correction and could have a material adverse effect on our business, results of operations, financial condition, and prospects.

#### We are subject to payment-related risks.

Our users can pay for purchases using a variety of methods, including through credit cards or through various third-party payment processors, and we pay our merchants through a variety of methods. If these service providers do not perform adequately or if our relationships with these service

providers were to terminate, our users' ability to place orders, and our merchants' ability to receive orders or payment could be adversely affected and our business could be harmed. For example, in 2014, PayPal temporarily suspended processing payments on our platform as a result of concerns related to products listed on our platform. If a third-party payment processor suspends service or has significant outages in the future and we do not have alternative payment processors in place or are unable to provide our own solution, our business could be harmed. In addition, if our third-party providers increase the fees they charge us, our operating expenses could increase. If we respond by increasing the fees we charge to our merchants, some merchants may stop listing new items for sale or even close their accounts altogether.

The laws and regulations related to payments are complex, evolving, subject to change and vary across different jurisdictions in the United States and globally. Any failure or claim of our failure to comply, or any failure by our third-party payment processors to comply, could cost us substantial resources and could result in liabilities. Further, through our agreements with our third-party payment processors, we are indirectly subject to payment card association operating rules, and certification requirements, including the Payment Card Industry Data Security Standard, which are subject to change. Failure to comply with these rules and certification requirements could impact our ability to meet our contractual obligations with our third-party payment processors and could result in potential fines. We are also subject to rules governing electronic funds transfers. Any change in these rules and requirements could make it difficult or impossible for us to comply. In addition, similar to a potential increase in costs from third-party providers described above, any increased costs associated with compliance with payment card association rules could lead to increased fees for our merchants, which may negatively impact our markets.

We track certain performance metrics with internal tools and do not independently verify such metrics. Certain of our performance metrics are subject to inherent challenges in measurement, and real or perceived inaccuracies in such metrics may harm our reputation and negatively affect our business.

We calculate and track performance metrics with internal tools, which are not independently verified by any third-party. While we believe our metrics are reasonable estimates of our user or merchant base for the applicable period of measurement, the methodologies used to measure these metrics require significant judgment and may be susceptible to algorithm or other technical errors. For example, user accounts are based on email addresses, and a user could use multiple email addresses to establish multiple accounts, and merchants in many instances will have multiple accounts. As a result, the data we report may not be accurate. Our internal tools and processes we use to identify multiple accounts or fraudulent accounts have a number of limitations, and our methodologies for tracking key metrics may change over time, which could result in unexpected changes to our metrics, including historical metrics. Our ability to recalculate our historical metrics may be impacted by data limitations or other factors that require us to apply different methodologies for such adjustments and we generally do not intend to update previously disclosed metrics for any such changes. Though we regularly review our processes for calculating metrics and may adjust our processes for calculating metrics to improve their accuracy, limitations or errors with respect to how we measure data (or the data that we measure) may affect our understanding of certain details of our business, which could affect our longer term strategies. If our performance metrics are not accurate representations of our business, user or merchant base, or traffic levels; if we discover material inaccuracies in our metrics; or if the metrics we rely on to track our performance do not provide an accurate measurement of our business, user or merchant base or traffic levels, we may not be able to effectively implement our business strategy, our reputation may be harmed, and our operating and financial results could be adversely affected.

Our merchants, platform partners, and investors rely on our key metrics as a representation of our performance. If these third parties do not perceive our user metrics to be accurate representations

of our user base or user engagement, or if we discover material inaccuracies in our user metrics, our reputation may be harmed and merchants may be less willing to sell on our platform, which could negatively affect our business, financial condition, or results of operations.

#### We must develop new offerings to respond to our users' and merchants' changing needs.

Our industry is characterized by rapidly changing technology, new service and product introductions, and changing user and merchant demands.

Our users and merchants may not be satisfied with our new platform offerings or perceive that the new offerings do not respond to their needs. Developing new offerings is complex, and the timetable for commercial release is difficult to predict and may vary from our historical experience. As a result, the introduction of new offerings may occur after anticipated or announced release dates. In addition, new offerings could require us to comply with additional governmental regulations. Our new offerings also may bring us more directly into competition with companies that are better established or have greater resources than we do.

If we do not continue to cost-effectively develop new offerings that satisfy our users or merchants, then our competitive position and growth prospects may be harmed. In addition, new offerings may have lower margins than existing offerings and our revenue may not grow enough as a result of the new offerings to offset the cost of developing them.

## If we fail to maintain, expand, and diversify our relationships with merchants, our revenue and results of operations will be harmed.

We rely on our merchants to offer products that appeal to our existing and potential users at attractive prices. Our ability to provide popular products on our platform at attractive prices depends on our ability to develop mutually beneficial relationships with our merchants. For example, we rely on our merchants, most of whom are based in China, to make available sufficient inventory and fulfill large volumes of orders in an efficient and timely manner to ensure a positive user experience. Merchants can leave our platform at any time, so we may experience merchant attrition in the ordinary course of business resulting from several factors, such as losses to competitors, perception that marketing on our platform is ineffective, reduction in merchants' marketing budgets, and the penalties we impose on merchants for failing to comply with our policies. We have had, and may continue to have, disputes with merchants with respect to their compliance with our delivery requirements, quality control policies and measures, and the penalties imposed by us for violation of these policies or measures from time to time, which may cause them to be dissatisfied with our platform or to legally challenge the enforceability of our terms. If we experience significant merchant attrition, or if we are unable to attract new and geographically-diverse merchants, our revenue and results of operations may be materially and adversely affected. For example, during the initial outbreak of COVID-19, many of our merchants based in China were adversely impacted, which had a negative impact on the supply of inventory on our marketplace. In addition, our agreements with merchants also typically do not restrict them from establishing or maintaining business relationships with our competitors.

## Failure to deal effectively with fraudulent activities on our platform would increase our loss rate and harm our business, and could severely diminish merchant and user confidence in and use of our services.

We have in the past incurred and may in the future incur losses from various types of fraud, including stolen credit card numbers, claims that a user did not authorize a purchase, merchant fraud, and users who have closed bank accounts or have insufficient funds in open bank accounts to satisfy payments. We face risks with respect to fraudulent activities on our platform and periodically receive

complaints from users who may not have received the products that they had contracted to purchase. In some European and Asian jurisdictions, users may also have the right to cancel a sale made by a merchant within a specified time period and for any reason. Although we have implemented measures to detect and reduce the occurrence of fraudulent activities, combat bad user experiences, and increase user satisfaction, including evaluating merchants on the basis of their transaction history and restricting or suspending their activity, there can be no assurance that these measures will be effective in combating fraudulent transactions or improving overall satisfaction among merchants, users, and other participants. Additional measures to address fraud could negatively affect the attractiveness of our services to users or merchants, resulting in a reduction in our ability to attract new users or continue to engage current users, damage to our reputation, or a diminution in the value of our brand.

Additionally, under current credit card practices, we are liable for fraudulent credit card transactions because we do not obtain a cardholder's signature, which results in chargebacks made to our users that we are not able to collect from our merchants. We do not currently carry insurance against this risk. We face the risk of significant losses from this type of fraud as our net sales increase and as we continue to expand globally. Our failure to adequately control fraudulent credit card transactions could damage our reputation and brand and substantially harm our business, results of operations, financial condition, and prospects.

We also accept payments for many of our sales through credit and debit card transactions, which are handled through third-party payment processors. As a result, we are subject to a number of risks related to credit and debit card payments, including that we pay interchange and other fees, which may increase over time and could require us to either increase the prices for products or absorb an increase in our costs and expenses. In addition, as part of payment processing, our users' credit and debit card information is transmitted to our third-party credit card payment processors. We may in the future become subject to lawsuits or other proceedings for purportedly fraudulent transactions arising out of the actual or alleged theft of our users' credit or debit card information if the security of our third-party credit card payment processors is breached. We and our third-party credit card payment processors are also subject to payment card association operating rules, certification requirements and rules governing electronic funds transfers, which could change or be reinterpreted to make it difficult or impossible for us to comply. If we or our third-party credit card payment processors fail to comply with these rules or requirements, we may be subject to fines and higher transaction fees and lose our ability to accept credit and debit card payments from our users in addition to the consequences that could arise from such action or inaction violating applicable privacy, data protection, data security and other laws as outlined above, and there may be an adverse impact on our business, results of operations, financial condition, and prospects.

#### The COVID-19 pandemic may adversely affect our business and results of operations.

On March 12, 2020, the World Health Organization declared COVID-19 to be a pandemic. In an effort to contain and mitigate the spread of COVID-19, many countries, including the United States, have imposed unprecedented restrictions on travel and business operations, and there have been business closures and a substantial reduction in economic activity in countries that have had significant outbreaks of COVID-19.

Our operations and performance depend significantly on global and regional economic conditions, and the outbreak of COVID-19 has had a significant negative effect on global and regional economies. Further, the ability of our merchants to offer products and make available sufficient inventory in an efficient manner may be adversely affected by the health impacts, travel restrictions, required social distancing, and other governmental mandates due to COVID-19, which could negatively impact our users' experience and cause our revenue and reputation to decline. Additionally, due to the economic impacts caused by COVID-19, consumer discretionary spending has been adversely

affected, which may cause the demand for the products available on our platform to be reduced and our revenue to decline.

We are conducting business with substantial restrictions, such as remote working and limited to no employee travel, among other modifications. While our business occurs over an online platform, which allows us to support our merchants and users virtually, we cannot be certain that our ability to service our merchants and users will not be adversely affected by COVID-19.

We saw increased delivery times due to COVID-19 as the supply chain was disrupted and we experienced a reduction in the number of merchants on our platform. During this time, we temporarily shifted from air to ocean transport, which increased delivery times and also led to an increase in refunds. While the number of merchants on our platform has recovered due to the reopening of the economy in China, we continue to see moderate disruption in the supply chain that is affecting delivery times.

We benefited from greater mobile usage and less competition from physical retail as a result of shelter-in-place mandates. We also benefited from increased user spending due to U.S. government stimulus programs. However, such stimulus programs have been reduced recently, which may adversely affect user spending. We cannot assure you that increased levels of mobile commerce will continue when COVID-19 has subsided or otherwise, or that the U.S. government will offer additional stimulus programs.

The global outbreak of COVID-19 continues to rapidly evolve, especially as COVID-19 cases and corresponding government actions responsive to COVID-19 have recently increased again in certain parts of the world. The extent to which COVID-19 may impact our business will depend on future developments, which are highly uncertain and cannot be predicted with confidence, the duration of the pandemic, travel restrictions, social distancing requirements, changes to consumer ecommerce activity in response to evolving governmental mandates that impact brick-and-mortar stores such as business closures or other governmental or business disruptions, global unemployment rates, and the effectiveness of actions taken in the United States and other countries to prevent, contain and treat the disease.

# Our pricing strategies may not meet users' price expectations or result in net income (loss), and laws and regulations could negatively impact the effectiveness of our model.

Our pricing strategies have had, and may continue to have, a significant impact on our revenue and net income (loss). In addition to offering discounted prices and shipping as a means of attracting users and encouraging repeat purchases, we use dynamic pricing, where pricing varies depending on factors such as user location and demand, which is intended to maximize volume and revenue and allows us to offer a variety of promotions and discounts. Such offers and discounts, however, may reduce our revenue and margins. In the future, laws applicable to data protection, consumer protection, and artificial intelligence may change in a manner that limits our ability to employ dynamic pricing. In addition, our competitors' pricing and marketing strategies are beyond our control and can significantly affect the results of our pricing strategies. If our pricing strategies, which may evolve over time, fail to meet our users' price expectations or fail to result in increased margins, or if we are unable to compete effectively with our competitors if they engage in aggressive pricing strategies or other competitive activities, it could have a material adverse effect on our business.

#### Our refund policies may adversely affect our results of operations.

We have adopted user-friendly refund policies that make it convenient and easy for users to receive a refund after completing purchases. These policies are designed to improve users' shopping

experience and promote user loyalty, which in turn help us acquire and engage our existing users. However, these policies also subject us to additional costs and expenses which we may not recoup through increased revenue. We may also be required by law to adopt new or amend existing refund policies from time to time. These policies also make us more susceptible to misuse and if our refund policy is misused by a significant number of users, our costs may increase significantly and our results of operations may be materially and adversely affected. If we revise these policies to reduce our costs and expenses, our users may be dissatisfied, which may result in loss of existing users or failure to acquire new users at a desirable pace or cost, which may materially and adversely affect our results of operations.

## Our ability to recruit and retain employees is important to our success.

Our future performance depends on our employees, including key engineering and product development personnel. Competition for key personnel is intense, especially in the San Francisco Bay area where our corporate headquarters are located, and we may be unable to successfully attract, integrate, or retain sufficiently qualified key personnel. In making employment decisions, particularly in the Internet and high-technology industries, job candidates often consider the value of the equity awards they would receive in connection with their employment and fluctuations in our stock price may make it more difficult to attract, retain, and motivate employees.

# The forecasts of market opportunity and market growth included in this prospectus may prove to be inaccurate, and, even if these forecasts materialize, we cannot assure you our business will grow at similar rates, if at all.

Estimates of market opportunity and forecasts of market growth are subject to significant uncertainty and are based on assumptions and estimates that may not prove to be accurate, including due to the recent impacts from COVID-19. The estimates in this prospectus of the size of the markets that we may be able to address and the forecasts in this prospectus relating to the expected growth in ecommerce and other markets are subject to many assumptions and may prove to be inaccurate. These markets may not grow at the rates that we forecast. We may not grow our business at similar rates, or at all. Our growth is subject to many factors, including our success in implementing our business strategy, which is subject to many risks and uncertainties. Accordingly, the estimates of market opportunity and forecasts of market growth included in this prospectus should not be taken as indicative of our future growth.

# We rely on consumer discretionary spending and may be adversely affected by economic downturns and other macroeconomic conditions or trends.

Macroeconomic conditions may adversely affect our business. If general economic conditions deteriorate globally or in specific markets where we operate, consumer discretionary spending may decline and demand for products available in our platform may be reduced. For example, we benefited from increased user spending due to U.S. government stimulus programs. However, such stimulus programs have been reduced recently, which may adversely affect user spending. We cannot assure you that the U.S. government will offer additional stimulus programs once the COVID-19 pandemic has subsided or otherwise. A decrease in consumer discretionary spending would cause sales in our platform to decline and adversely impact our business.

## We face risks relating to the inventory we purchase ourselves.

We directly purchase, on a very limited basis, some of the products that we sell on our platform. We assume the inventory damage, theft, and obsolescence risks, as well as product safety and price erosion risks for products that we purchase directly. These risks could become more significant in the

future if we increase the amount of inventory that we purchase. These risks could also become more significant depending on the types of product we hold in our inventory, such as products that are subject to seasonality, changes in consumer preferences, rapid technological change, obsolescence, and price erosion. If we choose to carry significant levels of inventory in the future, any one of these inventory risks could adversely affect our operating results.

## Unfavorable changes or failure by us to comply with evolving internet and ecommerce regulations could substantially harm our business and operating results.

We are subject to general business regulations and laws as well as regulations and laws specifically governing the internet and ecommerce. These regulations and laws may involve taxes, privacy and data security, consumer protection, the ability to collect and/or share necessary information that allows us to conduct business on the internet, marketing communications and advertising, content protection, electronic contracts, or gift cards. Furthermore, the regulatory landscape impacting internet and ecommerce businesses is constantly evolving. Any failure, or perceived failure, by us to comply with any of these laws or regulations could result in damage to our reputation, a loss in business and proceedings, or actions against us by governmental entities or others, which could impact our operating results.

## Our business could suffer if we are unsuccessful in making, integrating, and maintaining any future acquisitions and investments.

We may acquire businesses or technologies in the future. Integrating an acquired business or technology is difficult and can be risky. These potential and completed transactions create risks such as:

- · disruption of our ongoing business, including loss of management focus on existing businesses;
- · the difficulty of integrating new businesses and technologies into our infrastructure; and
- the risks associated with assuming liabilities related to the activities of the acquired business before and after the acquisition, including liabilities for violations of laws and regulations, commercial disputes, cyberattacks, taxes, and other matters.

Moreover, acquisitions may divert management's time and focus from operating our business. Acquisitions also may require us to spend a substantial portion of our available cash, issue stock, incur debt or other liabilities, amortize expenses related to intangible assets, or incur write-offs of goodwill or other assets. Finally, acquisitions could be viewed negatively by analysts, investors, merchants, or our users.

## We may be involved in litigation matters or other legal proceedings that are expensive and time consuming.

We may become involved in litigation matters, including class action lawsuits and lawsuits relating to intellectual property and product liability, whether for our own products or those offered by merchants. Any lawsuit to which we are a party, with or without merit, may result in an unfavorable judgment. We also may decide to settle lawsuits on unfavorable terms. Any such negative outcome could result in payments of substantial damages or fines, damage to our reputation, loss of rights, or adverse changes to our offerings or business practices. Any of these results could adversely affect our business. In addition, defending claims is costly and can impose a significant burden on our management.

From time to time, we are subject to investigations, demands, litigation and other proceedings involving consumer protection and data protection authorities or other regulatory agencies, including, in

particular, in Europe and Asia. These proceedings can result in orders, fines and penalties. For example, as a result of the initial outbreak of COVID-19, consumer protection authorities demanded rapid and decisive changes in the way that we screen and handle product listings that potentially violate various laws, including emergency price caps on certain items. Implementing these requests or defending against any associated fines could prove expense and time consuming and negatively affect our results of operations and financial condition.

Additionally, the market price of our Class A common stock may be volatile and, in the past, companies that have experienced volatility in the market price of their stock have been subject to securities class action litigation. We may be the target of this type of litigation in the future. Securities litigation against us could result in substantial costs and divert our management's attention from other business concerns, which could seriously harm our business.

## Risks Related to our International Operations

Economic tension between the United States and China, or between other countries, may intensify and the United States, China, or other countries may adopt drastic measures in the future that impact our business.

Our merchants source a large percentage of the products we list on our platform from China and other countries outside the United States. Additionally, most of our merchants, and some of our operations, are located in China, making the price and availability of products on our platform susceptible to international trade risks and other international economic conditions.

If the U.S. government imposes tariffs or other economic measures that directly or indirectly increase the price of products its imports and that we list on our platform, the increased prices could have a material adverse effect on our financial results and business. For example, the United States has recently threatened more restrictive trade terms with China and other countries, leading to the imposition, or announcement of future imposition, of substantially higher U.S. Section 301 tariffs on roughly \$500 billion of imports from China. In response, China imposed or proposed new or higher tariffs on U.S. exports. The effects of the recently imposed and proposed tariffs are uncertain because of the dynamic nature of governmental action, relations and responses. Further escalation of trade tensions between the United States and its trading partners, especially China, could result in long-term changes to global trade, including retaliatory trade restrictions that restrict the international flow of products. We cannot predict what actions may ultimately be taken with respect to tariffs or trade relations between the United States and China or other countries, what products may be subject to such actions, or what actions may be taken by the other countries in retaliation. Any alterations to our business strategy or operations made in order to adapt to or comply with any such changes would be time-consuming and expensive, and certain of our competitors may be better suited to withstand or react to these changes.

Additionally, the U.S. government has recently indicated that it may restrict the operation and access of certain China-based companies, which include TikTok and WeChat, in the United States and other countries have indicated taking similar actions. In response, government authorities in China, or elsewhere, may seek to restrict access and operation of U.S. companies. Most of our merchants and some of our operations are located in China, and if our activities were restricted in China, our platform, our business, financial condition, and results of operations would be adversely affected.

We are not able to predict future economic policy of the United States, China, or of any foreign countries in which we operate. The adoption and expansion of restrictions, including restrictions on access to apps and other platforms, cross-border data transfers, tariffs, or other governmental action related to economic policies, has the potential to adversely impact our business, operational results and financial position.

## Certain aspects of our business relating to the provision of financial services are subject to government regulation and oversight.

Many jurisdictions in which we operate have laws that govern financial services activities. Regulators in certain jurisdictions may determine that certain aspects of our business are subject to these laws and could require us to obtain licenses to continue to operate in such jurisdictions. For example, if we are deemed to be a money transmitter as defined by applicable regulation, we could be subject to certain laws, rules and regulations enforced by multiple authorities and governing bodies in the United States and abroad. If we are found to be a money transmitter under any applicable regulation and we are not in compliance with such regulations, we may be subject to fines or other penalties in one or more jurisdictions levied by federal or state or local regulators, including state Attorneys General, as well as those levied by foreign regulators. In addition to fines, penalties for failing to comply with applicable rules and regulations could include criminal and civil proceedings or other enforcement actions. We could also be required to make changes to our business practices or compliance programs as a result of regulatory scrutiny.

One of our subsidiaries, ContextLogic B.V., has applied for a payments institution license with its competent authority, De Nederlandsche Bank. This license would permit ContextLogic B.V. to provide payment services (including acquiring and executing payment transactions, as referred to in the Revised Payment Services Directive (2015/2366/EU)) in the Netherlands. In addition, ContextLogic B.V., once licensed, intends to provide such services on a cross-border passport basis into other countries within the European Economic Area (the "EEA").

We continue to evaluate our options for seeking additional licenses in several other jurisdictions to optimize our payment solutions and support the future growth of our business. We could be denied such licenses, have existing licenses revoked, or be required to make significant changes to our business operations before being granted such licenses. If we are denied licenses or such licenses are revoked, we may be forced to cease or limit business operations in certain jurisdictions, including in the EEA, and even if we are able to obtain such licenses, we could be subject to fines or other enforcement action, or stripped of such licenses, if we are found to violate the requirements of such licenses. Such regulatory actions, or the need to obtain regulatory approvals, could impose significant costs and involve substantial delay in payments we make in certain local markets, any of which could adversely affect our business, financial condition, or operating results.

In addition, laws related to money transmission and online payments are evolving, and changes in such laws could affect our ability to provide payment processing on our platform in the same form and on the same terms as we have historically, or at all. For example, changes to the EU Payment Services Directive caused aspects of our payment operations in the EEA to fall within the scope of European payments regulation. As a result, ContextLogic B.V. is directly subject to financial services regulations (including those relating to anti-money laundering, terrorist financing, and sanctioned or prohibited persons) in the Netherlands and in other countries in the EEA where it conducts business. In addition, as we evolve our business or make changes to our business structure, we may be subject to additional laws or requirements related to money transmission, online payments and financial regulation. These laws govern, among other things, money transmission, prepaid access instruments, lending, electronic funds transfers, anti-money laundering, counter-terrorist financing, banking, systemic integrity risk assessments, cyber security of payment processes, and import and export restrictions. Our business operations may not always comply with these financial laws and regulations. Historical or future non-compliance with these laws or regulations could result in significant criminal and civil lawsuits, penalties, forfeiture of significant assets or other enforcement actions. Costs associated with fines and enforcement actions, as well as reputational harm, changes in compliance requirements or limits on our ability to expand our product offerings, could harm our business.

Further, our payment system is susceptible to illegal and improper uses, including money laundering, terrorist financing, fraudulent sales of goods or services and payments to sanctioned parties. We have invested and will need to continue to invest substantial resources to comply with applicable anti-money laundering, counter-terrorism, and sanctions laws, and in the EEA to conduct appropriate risk assessments and implement appropriate controls once licensed as a regulated payments institution. Government authorities may seek to bring legal action against us if our payment system is used for improper or illegal purposes or if our enterprise risk management or controls in the EEA are not adequately assessed, updated, or implemented, and any such action could result in financial or reputational harm to our business. Additionally, some of our merchants use applications, such as WeChat, for transmitting payments and communicating with us. If any of these payments applications were limited or banned by governmental authorities, certain payments could be delayed or our communications with merchants could be adversely impacted.

We are subject to customs and international trade laws that could require us to incur increased costs or could result in a delay in getting products to users, which may limit our growth and cause us to suffer reputational damage.

Our business is conducted worldwide, with goods imported from and exported to a substantial number of countries. The vast majority of products sold on our platform are shipped internationally. We are subject to numerous regulations, including customs and international trade laws that govern the importation, exportation, and sale of goods. In addition, we face risks associated with trade protection laws, policies and measures and other regulatory requirements affecting trade and investment, including loss or modification of exemptions for taxes and tariffs, imposition of new tariffs and duties, and import and export licensing requirements in the countries in which we operate. If these laws or regulations were to change or were violated by our management, employees or merchants, we could experience delays in shipments of our goods, be subject to fines or penalties, or suffer reputational harm, which could reduce demand for our services and negatively impact our results of operations.

Legal requirements are frequently changed and subject to interpretation, and we are unable to predict the ultimate cost of compliance with these requirements or their effects on our operations. We may be required to make significant expenditures to comply with existing or future laws and regulations, which may increase our costs and materially limit our ability to operate our business. In addition, changes to legal requirements can create delays in the introduction and sale of our products and services, or in some cases, prevent the export or our products and services to certain countries, governments, or persons altogether.

We rely on the free flow of goods through open and operational ports worldwide. Labor disputes or other disruptions at ports create significant risks for our business, particularly if work slowdowns, lockouts, strikes, or other disruptions occur. Any of these factors could result in reduced sales or canceled orders, which may limit our growth and damage our reputation and may have a material adverse effect on our business, results of operations, financial condition, and prospects.

## Our international operations are subject to increased risks.

There are inherent risks in doing business internationally, including:

- expenses associated with localizing our products and services and user data, including offering our users and merchants the ability to transact business in the local currency and language, and adapting our platform to local preferences;
- challenges to enforceability in some foreign jurisdictions of so-called "clickwrap" contracts with our customers, merchants and Wish Local retailers;
- trade barriers and changes in trade regulations;

- difficulties in developing, staffing, and simultaneously managing a large number of varying foreign operations as a result of distance, language, and cultural differences;
- · stringent local labor laws and regulations;
- · credit risk and higher levels of payment fraud;
- laws or regulations related to the import or export of goods alleged to violate third-party intellectual property rights;
- political or social unrest, economic instability, repression, or human rights issues;
- · geopolitical events, including natural disasters, public health issues, acts of war, and terrorism;
- compliance with U.S. laws such as the Foreign Corrupt Practices Act and foreign laws prohibiting corruption, U.S. and foreign economic and trade sanctions laws, and U.S. and foreign laws designed to combat money laundering and the financing of terrorist activities:
- · antitrust and competition regulations;
- potentially adverse tax developments and consequences;
- economic uncertainties relating to sovereign and other debt;
- different, uncertain, or more stringent user protection, data protection, data collection, privacy, payments, advertising, pricing, and other laws;
- limitations by governmental authorities on transmission of privacy information and other data between countries, whether from the United States or other jurisdictions:
- restrictions on sales or distribution of certain products or services and uncertainty regarding liability for products, services, content, including uncertainty as a result of less internet-friendly legal systems, local laws, lack of legal precedent, and varying rules, regulations, and practices;
- risks related to other government regulation or required compliance with local laws;
- · national or regional differences in macroeconomic growth rates; and
- · local licensing and reporting obligations.

Violations of the complex foreign and U.S. laws and regulations that apply to our international operations may result in litigation, fines, criminal actions, or sanctions against us, our officers, or our employees; prohibitions on the conduct of our business; and damage to our reputation. Although we have implemented policies and procedures designed to promote compliance with these laws, there can be no assurance that our employees, contractors, or agents will not violate our policies. These risks inherent in our international operations and expansion increase our costs of doing business internationally and could harm our business.

## We face exposure to foreign currency exchange rate fluctuations.

Our sales to users are denominated in local currencies, primarily in U.S. dollars and Euros, and we pay our merchants in China for products sold on our platform in Renminbi ("RMB"), a few weeks after sales are completed, which creates exposure to currency rate fluctuations during that time. Additionally, our operating expenses are denominated in the currencies of the countries in which our operations are located, and may be subject to fluctuations due to changes in foreign currency exchange rates, particularly changes in the RMB. We do not generally hedge foreign exchange risk, and therefore, our results of operations have in the past, and will in the future, fluctuate due to movements in exchange rates. In addition, a weakening U.S. dollar will typically adversely affect the volume of goods being sold by foreign merchants to users in the United States more than it positively affects the volume of goods being sold by merchants in the United States to users in foreign countries, thereby further negatively impacting our financial results.

#### Any factors that reduce cross-border trade or make such trade more difficult could harm our business.

Cross-border trade is an important source of revenue for us. The shipping of goods across national borders is often more expensive and complicated than domestic shipping. Customs and duty procedures and reviews, including duty-free thresholds in various key markets, the interaction of national postal systems, and security related governmental processes at international borders, may increase costs, discourage cross-border purchases, delay transit, and create shipping uncertainties. Any factors that increase the costs of cross-border trade or restrict, delay, or make cross-border trade more difficult or impractical would lower our revenue and profits and could harm our business.

# Changes in China's economic, political or social conditions or government policies could have a material adverse effect on our business and operations.

Most of our merchants, and some of our operations, are located in China. Accordingly, our business, financial condition, results of operations, and prospects may be influenced to a significant degree by political, economic and social conditions in China generally. The Chinese economy differs from the economies of most developed countries in many respects, including the level of government involvement, level of development, and growth rate. The Chinese government continues to play a significant role in regulating industry development by imposing industrial policies. The Chinese government also exercises significant control over China's economic growth through allocating resources, controlling payment of foreign currency-denominated obligations, setting monetary policy, and providing preferential treatment to particular industries or companies.

# Uncertainties with respect to the People's Republic of China's ("PRC") legal system and changes in laws and regulations in China could adversely affect us.

Our operations in China are governed by PRC laws and regulations. Our Chinese subsidiaries are subject to laws and regulations applicable to foreign investment in China. The PRC legal system is a civil law system based on written statutes. Unlike the common law system, prior court decisions under the civil law system may be cited for reference but have limited precedential value. In addition, any new or changes in PRC laws and regulations related to foreign investment in China could affect the business environment and our ability to operate our business in China.

From time to time, we may have to resort to administrative and court proceedings to enforce our legal rights. Any administrative and court proceedings in China may be protracted, resulting in substantial costs and diversion of resources and management attention. Since PRC administrative and court authorities have significant discretion in interpreting and implementing statutory provisions and contractual terms, it may be more difficult to evaluate the outcome of administrative and court proceedings and the level of legal protection we enjoy than in more developed legal systems. These uncertainties may impede our ability to enforce the contracts we have entered into and could materially and adversely affect our business and results of operations.

Furthermore, the PRC legal system is based in part on government policies and internal rules, some of which are not published on a timely basis or at all and may have retroactive effect. As a result, we may not be aware of our violation of any of these policies and rules until sometime after the violation. Such unpredictability towards our contractual, property and procedural rights could adversely affect our business and impede our ability to continue our operations.

We may be adversely affected by the complexity, uncertainties and changes in PRC regulation of internet-related businesses and companies, and any lack of requisite approvals, licenses or permits applicable to our business may have a material adverse effect on our business and results of operations.

The Chinese government extensively regulates the internet industry, including foreign ownership of, and the licensing and permit requirements pertaining to, companies in the internet industry. These internet-related laws and regulations are relatively new and evolving, and their interpretation and enforcement involve significant uncertainties. As a result, in certain circumstances it may be difficult to determine what actions or omissions may be deemed to be in violation of applicable laws and regulations.

Because our business may be subject to governmental supervision and regulation by the relevant Chinese governmental authorities in many aspects of the operation of online retailing, we may be required to hold a number of licenses and permits in connection with our business operation. New laws and regulations may be adopted from time to time to require additional licenses and permits other than those we currently have. As a result, substantial uncertainties exist regarding the interpretation and implementation of current and any future PRC laws and regulations applicable to online retail businesses. If the Chinese government considers that we were operating without the proper approvals, licenses or permits, or promulgates new laws and regulations that require additional approvals or licenses or impose additional restrictions on the operation of any part of our business, it has the power, among other actions, to levy fines, confiscate our income, revoke our business licenses, or require us to discontinue our relevant business or impose restrictions on the affected portion of our business. Any of these or other regulatory actions by the Chinese governmental authorities, including issuance of official notices, change of policies, promulgation of regulations and imposition of sanctions, may adversely affect our business and have a material and adverse effect on our results of operations.

#### Risks Related to Network and Infrastructure

Any significant disruption in service on our platform or in our computer systems, some of which are currently hosted by third-party providers, could damage our reputation and result in a loss of users, which would harm our business and results of operations.

Our brand, reputation and ability to deliver a positive user experience depends upon the reliability of our infrastructure. We have experienced interruptions in these systems in the past, including server failures that temporarily slowed down or interfered with the performance of our websites and apps, or particular features of our websites and apps, and we may experience interruptions in the future. For example, in June 2020, we experienced a full platform outage for more than one hour due to the release of a software update that did not follow proper internal protocols. We have since updated our processes for following such protocols. Interruptions, whether due to system failures, human errors, computer viruses, physical or electronic break-ins, denial-of-service attacks, and capacity limitations, could prevent or inhibit the ability of merchants to access, or users from completing purchases on, our websites and apps. Volume of traffic and activity on our platform spikes on certain days, and any such interruption would be particularly problematic if it were to occur at such a high volume time. Problems with the reliability of our systems could prevent us from earning revenue and could harm our reputation. Damage to our reputation, any resulting loss of user confidence and the cost of remedying these problems could negatively affect our business, results of operations, financial condition, and prospects.

We either lease or own our servers and have service agreements with data center providers. Our systems and operations are vulnerable to damage or interruption from fire, flood, power loss, telecommunications failure, terrorist attacks, acts of war, electronic and physical break-ins, computer viruses, earthquakes, and similar events. The occurrence of any of the foregoing events could result in

damage to our systems and hardware or could cause them to fail completely, and our insurance may not cover such events or may be insufficient to compensate us for losses that may occur. Our systems are not completely redundant, so a system failure at one site could result in reduced platform functionality for our users, and a total failure of our systems could cause our websites or apps to be inaccessible by some or all of our users. A significant portion of our data storage, data processing and other computing services and systems is hosted by Amazon Web Services ("AWS"). AWS provides us with computing and storage capacity pursuant to an agreement that continues until terminated by either party. Problems with our third-party service providers, including AWS, or with their network providers or with the systems allocating capacity among their users, including us, could adversely affect our users' experience. Our third-party service providers could decide to close their facilities without adequate notice. Any financial difficulties, such as bankruptcy or reorganization, faced by our third-party service providers or any of the service providers with whom they contract may have negative effects on our business, the nature and extent of which are difficult to predict. If our third-party service providers are unable to keep up with our needs for capacity, this could have an adverse effect on our business. In the event that our agreement with AWS, or other third-party service providers, is terminated, or we add additional cloud infrastructure service providers. Any of the above circumstances or events may harm our reputation and brand, reduce the availability or usage of our platform, lead to a significant short-term loss of revenue, increase our costs, and impair our ability to attract new users or merchants, any of which could adversely affect our business, financial condition, and results of operations.

Our failure or the failure of third parties to protect our sites, networks and systems against security breaches, or otherwise to protect our confidential information, could damage our reputation and brand and substantially harm our business and operating results.

We collect, maintain, transmit, and store data about our users, merchants and others, including personally identifiable information and personal data, as well as other confidential information.

We also engage third parties that store, process, and transmit these types of information on our behalf. We rely on technology licensed from third parties in an effort to securely transmit confidential and sensitive information, including credit card numbers. Advances in computer capabilities, new technological discoveries, or other developments may result in the whole or partial failure of this technology to protect transaction data or other confidential and sensitive information from being breached or compromised. In addition, ecommerce websites are often attacked through compromised credentials, including those obtained through phishing, credential stuffing, and password spraying. Our security measures, and those of our third-party service providers, may not detect or prevent all attempts to breach our systems, viruses, malicious software, break-ins, phishing attacks, accidental actions or omissions to act that create vulnerabilities, social engineering, security breaches, or other attacks and similar disruptions that may jeopardize the security of information stored in or transmitted by our websites, networks and systems or that we or such third parties otherwise maintain, including payment card systems, which may subject us to fines or higher transaction fees or limit or terminate our access to certain payment methods. We and such third parties may not be able to anticipate or prevent all types of attacks, and we may not detect such attacks until after they have already been launched. Further, techniques used to obtain unauthorized access to or sabotage systems change frequently and may not be known until launched against us or our third-party service providers. In addition, security breaches can also occur as a result of non-technical issues, including intentional or inadvertent actions by our employees or by third parties. These risks may increase over time as the complexity and number of technical systems and applications we use also increases.

Cyber security incidents or breaches of our security measures or those of our third-party service providers or cyber security incidents could result in unauthorized access to our websites, networks and

systems; unauthorized access to and misappropriation of our data, including user information, personally identifiable information, or other confidential or proprietary information of ours or of third parties; viruses, worms, spyware, or other malware being served from our websites, networks, or systems; deletion or modification of content or the display of unauthorized content on our sites; interruption, disruption, or malfunction of operations; costs relating to breach remediation, deployment of additional personnel and protection technologies, response to governmental investigations and media inquiries and coverage; engagement of third-party experts and consultants; litigation, regulatory action and other potential liabilities. Social engineering, phishing, malware, and similar attacks and threats of denial-of-service attacks could have a material adverse effect on our operations. Additionally, from time to time, our merchants' and users' accounts have been subject to unauthorized access by third parties, including through illicit purchase of usernames and passwords by bad actors. If any of these breaches of security should occur, our reputation and brand could be damaged, our business may suffer, we could be required to expend significant capital and other resources to alleviate problems caused by such breaches, and we could be exposed to a risk of loss, litigation or regulatory action and possible liability. We cannot guarantee that recovery protocols and backup systems will be sufficient to prevent data loss. Actual or anticipated attacks may cause us to incur increasing costs, including costs to deploy additional personnel and protection technologies, train employees and engage third-party experts and consultants. In addition, any party who is able to illicitly obtain a user or merchant password could access the user or merchant's transaction data or personal information, resulting in the perception that our systems are insecure.

Any compromise or breach of our security measures, or those of our third-party service providers, could violate applicable privacy, data protection, data security, network and information systems security and other laws and cause significant legal and financial exposure (including costs for technical teams to investigate and remediate such incidents), adverse publicity and a loss of confidence in our security measures, which could have a material adverse effect on our business, results of operations, financial condition, and prospects. We devote significant resources to protect against security breaches and we may need to devote more resources in the future to address problems caused by breaches, including notifying affected users and responding to any resulting litigation, which in turn, diverts resources from the growth and expansion of our business.

## Catastrophic events may disrupt our business.

Natural disasters or other catastrophic events may cause damage or disruption to our operations, international commerce, and the global economy, and thus could harm our business. In the event of a major earthquake, hurricane or catastrophic event such as fire, power loss, telecommunications failure, cyber-attack, war, or terrorist attack, we may be unable to continue our operations and may endure reputational harm, delays in developing our platform and solutions breaches of data security and loss of critical data, all of which could harm our business, results of operations and financial condition.

Additionally, we rely on our network and third-party infrastructure and applications, internal technology systems, and our websites for our development, marketing, operational support, hosted services, and marketing activities. If these systems were to fail or be negatively impacted as a result of a natural disaster or other event, our ability to deliver a positive user and merchant experience would be impaired.

As we grow our business, the need for business continuity planning and disaster recovery plans will grow in significance. If we are unable to develop adequate plans to ensure that our business functions continue to operate during and after a disaster, and successfully execute on those plans in the event of a disaster or emergency, our business and reputation would be harmed.

We are subject to governmental regulation and other legal obligations related to privacy, data protection, information security, and consumer protection. If we are unable to comply with these, we may be subject to governmental enforcement actions, litigation, fines and penalties, or adverse publicity.

We collect personally identifiable information and other data from users and prospective users. We use this information to provide services and relevant products to our users, to support, expand and improve our business, and to tailor our marketing and advertising efforts. We may also share users' personal data with certain third parties as authorized by the user or as described in our privacy policy.

As a result, we are subject to governmental regulation and other legal obligations related to the protection of confidential and sensitive data (including personally identifiable information and personal data), privacy, information security and consumer protection in certain countries where we do business and there has been and will continue to be a significant increase globally in such laws that restrict or control the use of personal data.

In Europe, where the data privacy and information security regime underwent a significant change in 2018, the legal environment related to personal data continues to evolve and companies like us that process personal data from large numbers of individuals are subject to increasing regulatory scrutiny. The General Data Protection Regulation ("GDPR"), implemented more stringent operational requirements for our use of personal data. These more stringent requirements include expanded disclosures to tell our users about how we may use their personal data, increased controls on profiling users and increased rights for users to access, control and delete their personal data. In addition, there are mandatory data breach notification requirements and significantly increased penalties of the greater of €20 million or 4% of global turnover for the preceding financial year.

Although there are legal mechanisms to allow for the transfer of personal data from the United Kingdom, EEA and Switzerland to the United States, uncertainty about compliance with such data protection laws remains and such mechanisms may not be available or applicable with respect to the personal data processing activities necessary to research, develop and market our products and services. For example, legal challenges in Europe to the mechanisms allowing companies to transfer personal data from the EEA to the United States could result in further limitations on the ability to transfer personal data across borders, particularly if governments are unable or unwilling to reach new or maintain existing agreements that support cross-border data transfers, such as the EU-U.S. and Swiss-U.S. Privacy Shield Frameworks. Specifically, on July 16, 2020, the Court of Justice of the European Union invalidated Decision 2016/1250 on the adequacy of the protection provided by the EU-U.S. Privacy Shield Framework. To the extent that we or our service providers were to rely on the EU-U.S. Privacy Shield Framework, we may not be able to do so in the future, which could increase our costs and limit our ability to process personal data from the European Union. The same decision also cast doubt on the ability to use one of the primary alternatives to the Privacy Shield, namely, the European Commission's Standard Contractual Clauses, to lawfully transfer personal data from Europe to the United States and most other countries. At present, there are few if any viable alternatives to the Privacy Shield and the Standard Contractual Clauses.

In recent years, U.S. and European lawmakers and regulators have expressed concern over the use of third-party cookies and similar technologies for online behavioral advertising, and laws in this area are also under reform. In the European Union, current national laws that implement the ePrivacy Directive will be replaced by an EU regulation known as the ePrivacy Regulation. The draft ePrivacy Regulation retains existing informed consent conditions and also imposes the strict opt-in marketing rules on direct marketing that is "presented" on a web page rather than sent by email, alters rules on third-party cookies and similar technology and significantly increases penalties for breach of the rules. Regulation of cookies and similar technologies may lead to broader restrictions on our marketing and

personalization activities and may negatively impact the effectiveness of our marketing. Such regulations may also increase regulatory scrutiny and increase potential civil liability under data protection or consumer protection laws. The ePrivacy Regulations draft also advocates the development of browsers that block cookies by default. These developments could impair our ability to collect user information, including personal data and usage information, that helps us provide more targeted advertising to our current and prospective users, which could adversely affect our business, given our use of cookies and similar technologies to target our marketing and personalize the user experience.

Further, Brexit has created uncertainty with regard to data protection regulation in the United Kingdom. In particular, while the Data Protection Act of 2018, that "implements" and complements the GDPR achieved Royal Assent on May 23, 2018 and is now effective in the United Kingdom, it is still unclear whether transfer of data from the EEA to the United Kingdom will remain lawful under GDPR. During the period of "transition" (i.e., until December 31, 2020), EU law will continue to apply in the UK, including the GDPR, after which the GDPR will be converted into U.K. law. Beginning in 2021, the UK will be a "third country" under the GDPR. We may, however, incur liabilities, expenses, costs, and other operational losses under GDPR and applicable EU Member States and the U.K. privacy laws in connection with any measures we take to comply with them.

As interpretation of both the ePrivacy Regulation and GDPR develop, we could incur substantial costs to comply with these regulations. The changes could require significant systems changes, limit the effectiveness of our marketing activities, adversely affect our margins, increase costs and subject us to additional liabilities.

In the United States, federal and various state governments have adopted or are considering, laws, guidelines or rules for the collection, distribution, use and storage of information collected from or about users or their devices. For example, California recently passed the California Consumer Privacy Act (the "CCPA"), which became effective on January 1, 2020 and introduced substantial changes to privacy law for businesses that collect personal information from California residents. The CCPA creates individual privacy rights for California consumers and increases the privacy and security obligations of entities handling certain personal information. The CCPA provides for civil penalties for violations, as well as a private right of action for data breaches that is expected to increase data breach litigation. Additionally, the U.S. Federal Trade Commission and many state attorneys general are applying federal and state consumer protection laws, to impose standards for the online collection, use and dissemination of data. Furthermore, these obligations may be interpreted and applied inconsistently from one jurisdiction to another and may conflict with other requirements or our practices.

Many data protection regimes apply based on where a user is located, and as we expand our platform and new laws are enacted or existing laws change, we may be subject to new laws, regulations or standards or new interpretations of existing laws, regulations or standards, including those in the areas of data security, data privacy and regulation of email providers and those that require localization of certain data, which could require us to incur additional costs and restrict our business operations. Any failure or perceived failure by us to comply with rapidly evolving privacy or security laws policies (including our own stated privacy policies), legal obligations or industry standards or any security incident that results in the unauthorized release or transfer of personally identifiable information or other user data may result in governmental enforcement actions, litigation (including user class actions), fines and penalties or adverse publicity and could cause our users to lose trust in us, which could have a material adverse effect on our business, results of operations, financial condition, and prospects.

A failure to comply with current laws, rules and regulations or changes to such laws, rules and regulations and other legal uncertainties may adversely affect our business, financial performance, results of operations or business growth.

Our business and financial performance could be adversely affected by unfavorable changes in or interpretations of existing laws, rules, and regulations or the promulgation of new laws, rules, and regulations applicable to us and our business, including those relating to the internet and ecommerce, internet advertising and price display, consumer protection, anti-corruption, economic and trade sanctions, tax, payments, banking, data security, network and information systems security, data protection and privacy. As a result, regulatory authorities could prevent or temporarily suspend us from carrying on some or all of our activities or otherwise penalize us if our practices were found not to comply with applicable regulatory or licensing requirements or any binding interpretation of such requirements. Unfavorable changes or interpretations could decrease demand for our services, limit marketing methods and capabilities, affect our margins, increase costs or subject us to additional liabilities. For example, as a result of the initial outbreak of COVID-19, consumer protection authorities demanded rapid and decisive changes in the way that Wish screens and handles product listings that potentially violate various laws, including emergency price caps on certain items. We believe we have legal grounds to satisfy current requests or prevail against associated fines and penalties, and we intend to vigorously defend such fines and penalties.

Additionally, there are, and will likely continue to be, an increasing number of laws and regulations pertaining to the internet and ecommerce that may relate to liability for information retrieved from or transmitted over the internet, display of certain taxes and fees, online editorial and user-generated content, user privacy, data security, network and information systems security, behavioral targeting and online advertising, taxation, liability for third-party activities, quality of services and consumer protection. For example, the European Union's Market Surveillance Regulation, which takes effect in July 2021, places new obligations on marketplaces and is designed to reduce the availability of unsafe products in Europe when offered by sellers outside of the region. Denmark has passed a law placing new burdens on marketplaces and giving its regulators the right to request fines and shutdowns where marketplaces are consistently unsuccessful in screening products that are unsafe or unlawful. Furthermore, the growth and development of ecommerce may prompt calls for more stringent consumer protection laws and more aggressive enforcement efforts, which may impose additional burdens on online businesses generally.

Likewise, the Securities and Exchange Commission (the "SEC"), the U.S. Department of Justice, the U.S. Treasury Department's Office of Foreign Assets Controls, the U.S. Department of State, as well as other foreign regulatory authorities continue to enforce economic and trade regulations and anti-corruption laws, across industries. U.S. trade sanctions relate to transactions with designated foreign countries and territories, including Cuba, Iran, North Korea, Syria and the Crimea region of Ukraine, as well as specifically targeted individuals and entities that are identified on United States' and other blacklists, and those owned by them or those acting on their behalf. Anti-corruption laws, including the U.S. Foreign Corrupt Practices Act and the U.K. Bribery Act, generally prohibit direct or indirect corrupt payments to government officials and, under certain laws, to private persons to obtain or retain business or an improper business advantage. Although we have policies and procedures in place designed to promote compliance with these laws and regulations, our employees, partners, or agents could take actions in contravention of our policies and procedures, or violate applicable laws or regulations. In the event our controls should fail, or we are found to be not in compliance for other reasons, we could be subject to monetary damages, civil and criminal monetary penalties, withdrawal of business licenses or permits, litigation and damage to our reputation and the value of our brand.

Additionally, the law relating to liability of online service providers is currently unsettled. Lawmakers and governmental agencies have in the past and could in the future require changes in the

way our business is conducted, including with explicit obligations to inspect and screen content and products or implicit obligations that might stem from increased legal liability for online service providers. Unfavorable regulations, laws, decisions, or interpretations by government or regulatory authorities applying those laws and regulations, or inquiries, investigations, or enforcement actions threatened or initiated by them, could cause us to incur substantial costs, expose us to unanticipated civil and criminal liability or penalties (including substantial monetary fines), increase our cost of doing business, require us to change our business practices in a manner materially adverse to our business, damage our reputation, impede our growth, or otherwise have a material effect on our operations.

## **Risk Related to Our Intellectual Property**

## We may be unable to protect our intellectual property adequately.

Our intellectual property is an essential asset of our business and our success is dependent, in part, upon protecting our intellectual property. To establish and protect our intellectual property rights, we rely on a combination of trade secret, copyright, trademark and, to a lesser extent, patent laws, as well as confidentiality protection procedures and contractual provisions. The efforts we have taken to protect our intellectual property may not be sufficient or effective. We generally do not elect to register our copyrights, relying instead on the laws protecting unregistered intellectual property, which may not be sufficient. We rely on both registered and unregistered trademarks, which may not always be comprehensive in scope. In addition, our copyrights and trademarks, whether or not registered, and patents may be held invalid or unenforceable if challenged, and may be of limited territorial reach. Moreover, effective trademark, copyright, patent and trade secret protection may not be available or commercially feasible in every country in which we conduct business. Further, intellectual property law, including statutory and case law, particularly in the United States, is constantly developing, and any changes in the law could make it harder for us to enforce our rights. While we have obtained or applied for patent protection with respect to some of our intellectual property, we generally do not rely on patents as a principal means of protecting intellectual property. We make business decisions about when to seek patent protection for a particular technology and when to rely upon trade secret protection, and the approach we select may ultimately prove to be inadequate. To the extent we do seek patent protection, any U.S. or other patents issued to us may not be sufficiently broad to protect our proprietary technologies.

## We may be subject to claims that items listed on our platform are counterfeit, infringing or illegal, which may harm our business.

We frequently receive communications alleging that items listed on our platform infringe third-party copyrights, trademarks, or other intellectual property rights. We have intellectual property complaint and take-down procedures in place to address these communications, and we believe such procedures are important to promote confidence in our platform and provide users reassurance in the products they are purchasing. We follow these procedures to review complaints and relevant facts to determine the appropriate action, which may include removal of the item from our platform and, in certain cases, prohibiting merchants from participating in our platform who repeatedly violate our policies.

Our procedures may not effectively reduce or eliminate our liability. In particular, we may be subject to civil or criminal liability for activities carried out by merchants on our platform, especially outside the United States where laws may offer less protection for intermediaries and platforms than the United States. Under current U.S. copyright law and the Communications Decency Act, we may benefit from statutory safe harbor provisions that protect us from copyright liability for content posted on our platform by our merchants and users. However, trademark and patent laws do not include similar statutory provisions, and liability for these forms of intellectual property is often determined by court decisions. These safe harbors and court rulings may change unfavorably. In that event, we may

be held secondarily liable for the intellectual property infringement of merchants on our platform. We may also be directly liable for the inventory we purchase ourselves that we sell on our platform.

In addition, allegations of infringement of intellectual property rights, including but not limited to counterfeit items, have resulted in actual litigation from time to time by rights owners against merchants. These and similar suits have resulted in the freezing of merchant accounts or the shutdown of merchant storefronts on our platform, which can adversely impact revenue in the short-term, and may require us to spend substantial resources to comply with court orders. We may also incur costs responding to subpoenas from governmental authorities regarding illegal or counterfeit products listed for sale on our platform. In addition, we may receive media attention relating to the listing or sale of illegal or counterfeit goods, which could damage our reputation, diminish the value of our brand, and make users and merchants reluctant to use our platform.

Regardless of the validity of any claims made against us, we may incur significant costs and efforts to defend against or settle them.

Under our standard form agreements, we require our merchants to indemnify us for any losses we suffer or any costs that we incur due to any products sold by these merchants. However, we may not be able to successfully enforce our contractual rights and may need to initiate costly and lengthy legal proceedings to protect our rights.

We may be subject to intellectual property claims, which are extremely costly to defend, could require us to pay significant damages and could limit our ability to use certain technologies in the future.

Companies in the internet and technology industries are frequently subject to litigation based on allegations of infringement or other violations of intellectual property rights. We periodically receive communications that claim we have infringed, misappropriated or misused others' intellectual property rights. To the extent we gain greater public recognition, we may face a higher risk of being the subject of intellectual property claims. Third parties may have intellectual property rights that cover significant aspects of our technologies or business methods and prevent us from expanding our offerings. Any intellectual property claim against us, with or without merit, could be time consuming and expensive to settle or litigate and could divert the attention of our management. Litigation regarding intellectual property rights is inherently uncertain due to the complex issues involved, and we may not be successful in defending ourselves in such matters. In addition, some of our competitors have extensive portfolios of issued patents. In a patent infringement claim against us, we may assert, as a defense, that we do not infringe the relevant patent claims, that the patent is invalid, or both. The strength of our defenses will depend on the patents asserted, the interpretation of these patents, and our ability to invalidate the asserted patents. However, we could be unsuccessful in advancing non-infringement and/or invalidity arguments in our defense. In the United States, issued patents enjoy a presumption of validity, and the party challenging the validity of a patent claim must present clear and convincing evidence of invalidity, which is a high burden of proof. Conversely, the patent owner need only prove infringement by a preponderance of the evidence, which is a lower burden of proof. We may be unaware of the intellectual property rights that others may claim cover some or all of our technology or services. Because patent applications can take years to issue and are often afforded confidentiality for some period of time, there may currently be pending applications, unknown to us, that later result in issued patents that could cover one or more of our products. Many potential litigants, including some of our competitors and patent holding companies, have the ability to dedicate substantial resources to enforcing their intellectual property rights. Moreover, our patents may provide little or no deterrence in litigation involving patent holding companies or other adverse patent owners that have no relevant product revenue. Any claims successfully brought against us could subject us to significant liability for damages and we may be required to stop using technology or other intellectual property alleged to be

in violation of a third-party's rights in jurisdictions where we do business. We also might be required to enter into costly settlement agreements or seek a license for third-party intellectual property. Even if a license is available, we could be required to pay significant royalties or submit to unreasonable terms, which would increase our operating expenses. We may also be required to develop alternative non-infringing technology, which could require significant time and expense. If we cannot license or develop technology for any allegedly infringing aspect of our business, we would be forced to limit our service and may be unable to compete effectively. Any of these results could harm our business.

## Our software is highly complex and may contain undetected errors.

The software and code underlying our platform is highly interconnected and complex and may contain undetected errors or vulnerabilities, some of which may only be discovered after the code has been released. We typically release software code daily and this practice may result in the more frequent introduction of errors or vulnerabilities into the software underlying our platform, which can impact the user and merchant experience on our platform. Additionally, due to the interconnected nature of the software underlying our platform, updates to certain parts of our code, including changes to our website or mobile app or third-party APIs on which our website and mobile app rely, could have an unintended impact on other sections of our code, which may result in errors or vulnerabilities to our platform. Any errors or vulnerabilities discovered in our code after release could result in damage to our reputation, loss of our merchants or users, loss of revenue or liability for damages, any of which could adversely affect our growth prospects and our business.

## Our use of open source software may pose particular risks to our proprietary software and systems.

We use open source software in our proprietary software and systems and will use open source software in the future. The licenses applicable to our use of open source software may require that source code that is developed using open source software be made available to the public and that any modifications or derivative works to certain open source software continue to be licensed under open source licenses. From time to time, we may face claims from third parties claiming infringement of their intellectual property rights, or demanding the release or license of the open source software or derivative works that we developed using such software (which could include our proprietary source code) or otherwise seeking to enforce the terms of the applicable open source license. These claims could result in litigation and could require us to purchase a costly license, publicly release the affected portions of our source code, be limited in or cease using the implicated software unless and until we can re-engineer such software to avoid infringement or change the use of, or remove, the implicated open source software. Our use of open source software may also present additional security risks because the source code for open source software is publicly available, which may make it easier for hackers and other third parties to determine how to breach our website and systems that rely on open source software. Any of these risks could be difficult to eliminate or manage, and, if not addressed, could have a material adverse effect on our business, results of operations, financial condition, and prospects.

## Risks Related to Our Taxes and Financial Position

#### Our business and our merchants and users may be subject to sales and other taxes.

The application of indirect taxes, such as sales and use tax, value-added tax, provincial taxes, goods and services tax, business tax and gross receipt tax to our business and to our merchants is a complex and evolving issue. Many of the statutes and regulations that govern these taxes were established before the expansion of the internet and ecommerce. In many cases, it is not clear how existing statutes apply to ecommerce. In addition, governments are increasingly looking for ways to increase revenue, which has resulted in discussions about new legislative action to increase tax revenue, including through indirect taxes.

Significant judgment and expertise is required to evaluate applicable tax obligations. As a result, amounts recorded may be subject to adjustments by the relevant tax authorities. In many cases, the ultimate tax determination is uncertain because it is not clear how new and existing statutes might apply to our business or to the businesses of our merchants. One or more states, the federal government or other countries may seek to impose additional reporting, recordkeeping or indirect tax collection obligations on businesses like ours that facilitate ecommerce.

State and local taxing authorities in the United States have identified ecommerce platforms as a means to calculate, collect and remit indirect taxes for transactions taking place over the internet. Multiple U.S. states have enacted related legislation and other states are now considering such legislation. Furthermore, the U.S. Supreme Court recently held in South Dakota v. Wayfair that a U.S. state may require an online retailer to collect sales taxes imposed by the state in which the buyer is located, even if the retailer has no physical presence in that state, thus permitting a wider enforcement of such sales tax collection requirements.

Outside of the United States, the application of value added tax or other indirect taxes on ecommerce providers continues to evolve. An increasing number of jurisdictions are legislating or have adopted laws that impose new taxes, including revenue-based taxes that target online commerce and the remote selling of goods. These laws include new obligations to collect sales, consumption, value added, or other taxes on online marketplaces and remote sellers, or other requirements that may result in liability for third-party obligations. Several countries, in particular the European Union, have proposed or enacted taxes on marketplace facilitators and online advertising. Our business could be adversely affected by additional taxes that focus on marketplace service revenue.

Additionally, existing and new tax laws and legislation could require us or our merchants to incur substantial costs in order to comply, including costs associated with legal advice, tax calculation, collection, remittance and audit requirements, which could make selling in such markets less attractive and could adversely affect our business. Further, these laws can be applied prospectively or retroactively. Noncompliance with new laws may result in fines or penalties. It is possible we may not have sufficient notice to create and adopt processes to properly comply with new reporting or collection obligations by the effective date.

Our results of operations and cash flows could be adversely effected by additional taxes or increasing taxes of this nature imposed on us prospectively or retroactively, or additional taxes or penalties resulting from the failure to comply with any collection obligations or failure to provide information about our users, merchants or other third parties for tax reporting purposes to various government agencies.

## We may experience fluctuations in our tax obligations.

We are subject to taxation in the United States and in numerous other jurisdictions. We record tax expense based on current tax payments and our estimates of future tax payments, which may include reserves for estimates of probable settlements of tax audits. The determination of these liabilities requires estimation and significant judgment and the ultimate determination is uncertain. At any one time, multiple tax years could be subject to audit by various taxing jurisdictions. As a result, we could be subject to higher than anticipated tax liabilities as well as ongoing variability in our quarterly tax rates as audits close and exposures are re-evaluated. While we have estimated reserves that we believe are reasonable to cover potential exposures, the reserves may ultimately not be sufficient and additional cash outflows may result. Fluctuations in our tax obligations could adversely affect our business.

## We may be subject to tax controversies.

We may also be subject to tax controversies in the United States and in foreign jurisdictions that can result in tax assessments against us. Developments in an audit, investigation, or other tax controversy can have a material effect on our operating results or cash flows. We regularly assess the likelihood of an adverse outcome resulting from these proceedings to determine the adequacy of our tax accruals and while we believe our tax estimates are reasonable, the final outcome of audits, investigations, and any other tax controversies could be materially different from our historical tax accruals.

## We may not be able to utilize a significant portion of our net operation loss carryforwards, and other tax attributes, which could adversely affect our profitability.

As of December 31, 2019, we had federal net operating loss carryforwards ("NOLs") available to reduce future taxable income, if any, of \$885 million that begin to expire in 2030 and continue to expire through 2037 and \$324 million that have an unlimited carryover period. As of December 31, 2019, we had state NOLs available to reduce future taxable income, if any, of \$1.4 billion that begin to expire in 2030 and continue to expire through 2039. Under legislation enacted in 2017, informally titled the Tax Cuts and Jobs Act (the "Tax Act"), as modified by the Coronavirus Aid, Relief, and Economic Security Act (the "CARES Act"), unused U.S. federal NOLs generated in tax years beginning after December 31, 2017, will not expire and may be carried forward indefinitely, but the deductibility of such federal NOLs in tax years beginning after December 31, 2020, is limited to 80% of taxable income. It is uncertain if and to what extent various states will conform to the Tax Act or the CARES Act. In addition, the utilization of NOLs and other tax attributes to offset future taxable income or taxes may be subject to limitations under Sections 382 and 383 of the Internal Revenue Code of 1986, as amended (the "Code"), and similar state statutes as a result of ownership changes that have occurred or could occur in the future. Additionally, portions of these NOLs could expire unused and be unavailable to offset future income tax liabilities. In addition, at the state level, there may be periods during which the use of net operating losses is suspended or otherwise limited. For example, California recently imposed limits on the usability of California state NOLs to offset taxable income in tax years beginning after 2019 and before 2023\_As a result, even if we attain profitability, we may be unable to use a material portion of our NOLs and other tax attributes, which could adversely affect our future cash flows, which could adversely affect our profitability.

## We may need additional capital, which may not be available to us on acceptable terms or at all.

We believe that our existing cash, cash equivalents and current marketable securities, together with cash generated from our operations, will be enough to meet our anticipated cash needs for at least the next 12 months. However, we may require additional cash resources due to changes in business conditions or other developments, such as acquisitions or investments we may decide to pursue. We may seek to borrow funds under a credit facility that we decide to enter into or sell additional equity or debt securities. The sale of additional equity or convertible debt securities could result in dilution to our existing stockholders. Any debt financing that we may secure in the future could result in additional operating and financial covenants that would limit or restrict our ability to take certain actions, such as incurring additional debt, making capital expenditures or declaring dividends. It is also possible that financing may not be available to us in amounts or on terms acceptable to us, if at all.

Operating as a public company will require us to incur substantial costs and will require substantial management attention. In addition, our management team has limited experience managing a public company and the requirements of being a public company may strain our resources, divert management's attention and affect our ability to attract and retain additional executive management and qualified board members.

As a public company, we will incur substantial legal, accounting and other expenses that we did not incur as a private company. For example, we will be subject to the reporting requirements of the Securities Exchange Act of 1934 as amended (the "Exchange Act"), the applicable requirements of the Sarbanes-Oxley Act of 2002 (the "Sarbanes-Oxley Act"), the Dodd-Frank Wall Street Reform and Consumer Protection Act and the rules and regulations of will also apply to us following this offering. As part of the new requirements, we will need to establish and maintain effective disclosure and financial controls and make changes to our corporate governance practices. We expect that compliance with these requirements will increase our legal and financial compliance costs and will make some activities more time-consuming or costly, and increase demand on our systems and resources.

Most of our management and other key personnel have little experience managing a public company and preparing public filings. In addition, we expect that our management and other key personnel will need to divert attention from other business matters to devote substantial time to the reporting and other requirements of being a public company. In particular, we expect to incur significant expense and devote substantial management effort to complying with the requirements of Section 404 of the Sarbanes-Oxley Act. We will need to hire additional accounting and financial staff with appropriate public company experience and technical accounting knowledge.

As a result of disclosure of information in this prospectus and in filings required of a public company, our business and financial condition will become more visible, which may result in threatened or actual litigation, including by stockholders and competitors. If such claims are successful, our business and operating results could be adversely affected, and even if the claims do not result in litigation or are resolved in our favor, these claims, and the time and resources necessary to resolve them, could divert the resources of our management and adversely affect our business and operating results.

In addition, as a result of our disclosure obligations as a public company, we will have reduced flexibility and will be under pressure to focus on short-term results, which may adversely affect our ability to achieve long-term profitability.

Our management will be required to evaluate the effectiveness of our internal control over financial reporting. If we are unable to maintain effective internal control over financial reporting, investors may lose confidence in the accuracy of our financial reports.

As a public company, we will be required to maintain internal control over financial reporting and to report any material weaknesses in such internal control. Section 404 of the Sarbanes-Oxley Act requires that we evaluate and determine the effectiveness of our internal control over financial reporting. Additionally, our auditor will be required to deliver an attestation report on the effectiveness of our disclosure controls and internal control over financial reporting starting with the second annual report that we file with the SEC after the completion of this offering.

If we have a material weakness in our internal control over financial reporting, we may not detect errors on a timely basis and our financial statements may be materially misstated. In addition, our internal control over financial reporting will not prevent or detect all errors and fraud. Because of the inherent limitations in all control systems, no evaluation can provide absolute assurance that misstatements due to error or fraud will not occur or that all control issues and instances of fraud will be detected.

If there are material weaknesses or failures in our ability to meet any of the requirements related to the maintenance and reporting of our internal control, investors may lose confidence in the accuracy and completeness of our financial reports and that could cause the price of our Class A common stock to decline. In addition, we could become subject to investigations by the SEC or other regulatory authorities, which could require additional management attention and which could adversely affect our business.

If our estimates or judgments relating to our critical accounting policies prove to be incorrect or financial reporting standards or interpretations change, our operating results could be adversely affected.

The preparation of financial statements in conformity with U.S. GAAP requires management to make estimates, judgments, and assumptions that affect the amounts reported in our consolidated financial statements and accompanying notes. We base our estimates on historical experience and on various other assumptions that we believe to be reasonable under the circumstances, as provided in "Management's Discussion and Analysis of Financial Condition and Results of Operations." The results of these estimates form the basis for making judgments about the carrying values of assets, liabilities, and equity as of the date of the financial statements, and the amount of revenue and expenses, during the periods presented, that are not readily apparent from other sources. Significant assumptions and estimates used in preparing our consolidated financial statements include those related to the fair value of financial instruments, useful lives of long-lived assets, fair value of common stock, fair value of redeemable convertible preferred stock and related redeemable convertible preferred stock warrants and equity awards and other equity issuances, incremental borrowing rate applied to lease accounting, contingent liabilities, allowances for returns and refunds and uncertain tax positions. Our operating results may be adversely affected if our assumptions change or if actual circumstances differ from those in our assumptions, which could cause our operating results to fall below the expectations of industry or financial analysts and investors, resulting in a decline in the trading price of our Class A common stock.

## Risks Related to this Offering and Our Class A Common Stock

The price of our Class A common stock could be volatile and you may not be able to resell your shares at or above our initial public offering price. Declines in the price of Class A common stock could subject us to litigation.

There has not been a public market for our Class A common stock prior to this offering and an active trading market may not develop following this offering. Even if such a market does develop, it may not be sustainable. If trading in our Class A common stock is not active, you may not be able to sell your shares quickly, at the market price or at all. The initial public offering price for the shares was determined by negotiations between us and the representative of the underwriters and may not be indicative of prices that will prevail in the trading market following this offering. In addition, the trading prices of the securities of technology companies have historically been highly volatile. Accordingly, the price of our Class A common stock could be subject to wide fluctuations for many reasons, many of which are beyond our control, including those described in this "Risk Factors" section and others such as:

- variations in our operating results and other financial and operational metrics, including the key financial and operating metrics disclosed in this prospectus, as well as how those results and metrics compare to analyst and investor expectations;
- speculation about our operating results in the absence of our own financial projections;
- failure of analysts to initiate or maintain coverage of our company, changes in their estimates of our operating results or changes in recommendations by analysts that follow our Class A common stock;

- announcements of new services or enhancements, strategic alliances or significant agreements or other developments by us or our competitors;
- announcements by us or our competitors of mergers or acquisitions or rumors of such transactions involving us or our competitors;
- · changes in our senior management or other key personnel;
- disruptions in our platform due to hardware, software or network problems, security breaches or other issues;
- the strength of the global economy or the economy in the jurisdictions in which we operate, and market conditions in our industry and those affecting our merchants and users;
- trading activity by our principal stockholders, including upon the expiration of contractual lock-up agreements, and other market participants, in whom ownership of our Class A common stock may be concentrated following this offering;
- · changes in legal or regulatory requirements relating to our business;
- · litigation or other claims against us;
- the number of shares of our Class A common stock that are available for public trading; and
- · any other factors discussed in this prospectus.

In addition, if the market for technology stocks or the stock market in general experiences a loss of investor confidence, the price of our Class A common stock could decline for reasons unrelated to our business, results of operations or financial condition. The price of our Class A common stock might also decline in reaction to events that affect other companies, even if those events do not directly affect us. Some companies that have experienced volatility in the trading price of their stock have been the subject of securities class action litigation. If we are the subject of such litigation, it could result in substantial costs and could divert our management's attention and resources, which could adversely affect our business.

## We have broad discretion in the use of the net proceeds from this offering and may not use them effectively.

We currently intend to use the net proceeds from this offering for working capital, operating expenses, sales and marketing expenses to fund the growth of our business, and capital expenditures. In addition, we may use a portion of the net proceeds to acquire complementary businesses, products, services, or technologies. However, we have no current understandings, agreements or commitments for any specific material acquisitions at this time. We have not yet determined the manner in which we will allocate the net proceeds we receive from this offering and as a result, our management will have broad discretion in the allocation and use of the net proceeds. See "Use of Proceeds."

The failure by our management to allocate or use these funds effectively could harm our business. Pending their use, we may invest the net proceeds we receive from this offering in a manner that does not produce income or that loses value. Our ultimate use of the net proceeds from this offering may vary substantially from the currently intended use.

Our directors, executive officers and principal stockholders beneficially own a substantial percentage of our stock and will be able to exert significant control over matters subject to stockholder approval.

Our directors, executive officers, greater than 5% stockholders and their respective affiliates will hold in the aggregate approximately % of the voting power of our outstanding capital stock

following this offering, assuming no exercise of the underwriters' option to purchase additional shares of our Class A common stock from the selling stockholders in this offering. In addition, Peter Szulczewski, our founder, CEO, and Chairperson, will be able to exercise voting rights with respect to an aggregate of shares of Class B common stock, which will represent approximately % of the voting power of our outstanding capital stock following this offering. Therefore, these stockholders will continue to have the ability to influence us through their ownership position, even after this offering. If these stockholders act together, they may be able to determine all matters requiring majority stockholder approval. For example, these stockholders will be able to control elections of directors, amendments of our charter documents or approval of any merger, sale of assets or other major corporate transaction. This may prevent or discourage unsolicited acquisition proposals or offers for our capital stock that other stockholders may feel are in their best interests.

The dual class structure of our common stock has the effect of concentrating voting control with certain stockholders, in particular, our Chief Executive Officer, which will limit your ability to influence the outcome of important transactions, including a change in control.

Our Class B common stock has 20 votes per share, and our Class A common stock, which is the stock we are offering in our initial public offering, has one vote per share. Stockholders who hold shares of Class B common stock, including our executive officers, employees, and directors and their affiliates, will together hold approximately % of the voting power of our outstanding capital stock following our initial public offering. Because of the 20-to-one voting ratio between our Class B and Class A common stock, the holders of our Class B common stock collectively will continue to control a majority of the combined voting power of our common stock and therefore be able to control all matters submitted to our stockholders for approval so long as the shares of Class B common stock represent at least 50% of all outstanding shares of our Class A and Class B common stock. This concentrated control will limit or preclude your ability to influence corporate matters for the foreseeable future.

Future transfers by holders of Class B common stock will generally result in those shares converting to Class A common stock, subject to limited exceptions, such as certain transfers effected for estate planning or charitable purposes. The conversion of Class B common stock to Class A common stock will have the effect, over time, of increasing the relative voting power of those holders of Class B common stock who retain their shares in the long term. If, for example, Mr. Szulczewski retains a significant portion of his holdings of Class B common stock for an extended period of time, he could, in the future, continue to control a majority of the combined voting power of our Class A common stock and Class B common stock. For a description of the dual class structure, see "Description of Capital Stock—Anti-Takeover Provisions."

Because we qualify as a "controlled company" under the corporate governance rules for required to have a majority of our board of directors be independent, nor are we required to have a compensation committee or an independent nominating function. Our board of directors has not made a determination about whether or not to take advantage of the "controlled company" exemption. However, if we were to take advantage of such exemptions in the future, our Class A common stock could be less attractive to some investors or otherwise our stock price could be otherwise harmed. Additionally, should the interests of our controlling stockholder differ from those of other stockholders, the other stockholders may not have the same protections afforded to stockholders of companies that are subject to all of the corporate governance rules for -listed companies. Our status as a controlled company could make our Class A common stock less attractive to some investors or otherwise harm our stock price.

In addition, in July 2017, FTSE Russell and Standard & Poor's announced that they would cease to allow most newly public companies utilizing dual or multi-class capital structures to be included in

their indices. Affected indices include the Russell 2000 and the S&P 500, S&P MidCap 400, and S&P SmallCap 600, which together make up the S&P Composite 1500. Under the announced policies, our multi-class capital structure would make us ineligible for inclusion in any of these indices, and as a result, mutual funds, exchange-traded funds, and other investment vehicles that attempt to passively track these indices will not be investing in our stock. These policies are relatively new and it is as of yet unclear what effect, if any, they will have on the valuations of publicly traded companies excluded from the indices, but it is possible that they may depress these valuations compared to those of other similar companies that are included.

Sales of a substantial number of shares of our Class A common stock in the public market by our existing stockholders following this offering could cause the price of our Class A common stock to decline.

The price of our Class A common stock could decline if there are substantial sales of our Class A common stock, particularly sales by our directors, executive officers, employees, and significant stockholders, or when there is a large number of shares of our Class A common shares of our Class A common stock and stock available for sale. After our initial public offering, we will have outstanding shares of our Class B common stock, based on the number of shares outstanding as of September 30, 2020. This includes shares that we are selling in this offering, which shares may be resold in the public market immediately following date of this prospectus. Shares of our Class B common stock are convertible into an equivalent number of shares of our Class A common stock and generally convert into shares of our Class A common stock upon transfer. In addition, we expect to issue shares of our Class B common stock upon the settlement of RSUs on March 14, 2021. The shares of our Class A common stock and shares of our Class B common stock that are not offered and sold in our initial public offering (as well as the shares underlying outstanding RSUs) will be available for sale in the public market 181 days after the date of this prospectus or, in the case of the shares issuable upon settlement of RSUs, on March 14, 2021, subject in some cases to the volume and other restrictions of Rule 144 as described below. We also expect holders of such shares of Class B common stock issued upon settlement of RSUs to sell a portion of such shares in the public market to cover tax obligations that become due upon vesting and settlement of such RSUs.

In addition, as of September 30, 2020, options to purchase shares of Class B common stock were outstanding and fully vested and the Class B common stock underlying such options will be eligible for sale approximately six months after the date of this prospectus.

After our initial public offering, certain holders of our Class A common stock and Class B common stock will have rights, subject to some conditions, to require us to file registration statements covering their shares or to include their shares in registration statements that we may file for ourselves or our stockholders. All of these shares are subject to market standoff or lock-up agreements restricting their sale for specified periods of time after the date of this prospectus. We also intend to register shares of common stock that we have issued and may issue under our employee equity incentive plans. Once we register these shares, they will be able to be sold freely in the public market upon issuance, subject to existing market standoff or lock-up agreements.

Goldman Sachs & Co., LLC may permit our executive officers, our directors, and all of our stockholders to sell shares prior to the expiration of the restrictive provisions contained in the "lock-up" agreements with the underwriters. See the section titled "Underwriting" for a more complete description of the lock-up agreements that we and our directors, executive officers, and all of our stockholders have entered into with the underwriters. In addition, we may, in our sole discretion, permit our employees and current stockholders who are subject to market standoff agreements or arrangements with us and who are not subject to a lock-up agreement with the underwriters to sell shares prior to the expiration of the restrictive provisions contained in those market standoff agreements or arrangements.

The market price of the shares of our Class A common stock could decline as a result of the sale of a substantial number of our shares of common stock in the public market or the perception in the market that the holders of a large number of shares intend to sell their shares.

## If you purchase our Class A common stock in this offering, you will incur immediate and substantial dilution.

The initial public offering price is substantially higher than the pro forma net tangible book value per share of our Class A common stock of \$ per share as of September 30, 2020. Investors purchasing Class A common stock in this offering will pay a price per share that substantially exceeds the net tangible book value per share. As a result, investors purchasing Class A common stock in this offering will incur immediate dilution of \$ per share, based on the initial public offering price of \$ per share.

This dilution is due to the substantially lower price paid by our investors who purchased shares prior to this offering as compared to the price offered to the public in this offering. In addition, as of September 30, 2020, there were outstanding options to purchase shares of our common stock with a weighted-average exercise price of approximately \$ per share and warrants to purchase shares of our common stock (including redeemable convertible preferred stock on an as-converted basis) with a weighted-average exercise price of approximately \$ per share. The exercise of any of these options or warrants would result in additional dilution. As a result of the dilution to investors purchasing shares in this offering, investors may receive less than the purchase price paid in this offering in the event of our liquidation. See the section titled "Dilution."

# Future sales and issuances of our Class A common stock or rights to purchase Class A common stock could result in additional dilution to our stockholders and could cause the price of our Class A common stock to decline.

We may issue additional Class A common stock, convertible securities or other equity following the completion of this offering. We also expect to issue Class A common stock to our employees, directors and other service providers pursuant to our equity incentive plans. Such issuances could be dilutive to investors and could cause the price of our Class A common stock to decline. New investors in such issuances could also receive rights senior to those of holders of our Class A common stock.

# If analysts do not publish research about our business or if they publish inaccurate or unfavorable research, our stock price and trading volume could decline.

The trading market for our Class A common stock will depend in part on the research and reports that analysts publish about our business. We do not have any control over these analysts. If one or more of the analysts who cover us downgrade our Class A common stock or publish inaccurate or unfavorable research about our business, the price of our Class A common stock would likely decline. If few analysts cover us, demand for our Class A common stock could decrease and our Class A common stock price and trading volume may decline. Similar results may occur if one or more of these analysts stop covering us in the future or fail to publish reports on us regularly.

Anti-takeover provisions in our charter documents and under Delaware law could make an acquisition of our company more difficult, could limit attempts to make changes in our management and could depress the price of our Class A common stock.

Provisions in our certificate of incorporation and bylaws may have the effect of delaying or preventing a change in control of our company or limiting changes in our management. Among other things, they provide:

- for a dual class common stock structure, which provides Mr. Szulczewski with the ability to control the outcome of matters requiring stockholder approval, even if he owns significantly less than a majority of the shares of our outstanding Class A and Class B common stock:
- when the outstanding shares of our Class B common stock represent less than a majority of the combined voting power of our common stock, our board of directors will be classified into three classes of directors with staggered three-year terms and directors will only be able to be removed from office for cause;
- when the outstanding shares of our Class B common stock represent less than a majority of the combined voting power of our common stock, directors will be able to be removed only for cause and vacancies on our board of directors will be able to be filled only by our board of directors and not by stockholders;
- certain amendments to our restated certificate of incorporation or bylaws will require the approval of two-thirds of the combined vote of our then-outstanding shares of Class A and Class B common stock;
- authorization of the issuance of "blank check" preferred stock that our board of directors could use to implement a stockholder rights plan;
- when the outstanding shares of our Class B common stock represent less than a majority of the combined voting power of our common stock, our stockholders will only be able to take action at a meeting of stockholders and not by written consent and will not be able to call a special meeting of stockholders;
- · our board of directors is expressly authorized to amend or repeal any provision of our bylaws;
- · that the forum for certain litigation against us must be Delaware or the U.S. federal district courts; and
- advance notice requirements for nominations for election to our board of directors or for proposing matters that can be acted upon by stockholders at annual stockholder meetings.

These provisions may delay or prevent attempts by our stockholders to replace members of our management by making it more difficult for stockholders to replace members of our board of directors, which is responsible for appointing the members of our management. In addition, Section 203 of the Delaware General Corporation Law (the "DGCL") may delay or prevent a change in control of our company. Section 203 imposes certain restrictions on mergers, business combinations and other transactions between us and holders of 15% or more of our common stock. Anti-takeover provisions could depress the price of our common stock by acting to delay or prevent a change in control of our company.

For information regarding these and other provisions, see "Description of Capital Stock."

Our certificate of incorporation will provide that the Court of Chancery of the State of Delaware and the federal district courts of the United States of America will be the exclusive forums for substantially all disputes between us and our stockholders, which could limit our stockholders' ability to obtain a favorable judicial forum for disputes with us or our directors, officers or employees.

Our certificate of incorporation will provide that the Court of Chancery of the State of Delaware and the federal district courts of the United States of America will be the exclusive forum for substantially all disputes between us and our stockholders, which could limit our stockholders' ability to obtain a favorable judicial forum for disputes with us or our directors, officers or employees. Our certificate of incorporation will provide that the Court of Chancery of the State of Delaware is the exclusive forum for: (i) any derivative action or proceeding brought on behalf of us; (ii) any action asserting a claim of breach of a fiduciary duty; (iii) any action arising pursuant to any provision of the DGCL, our certificate of incorporation or bylaws (as either may be amended from time to time); or (iv) any action asserting a claim against us that is governed by the internal affairs doctrine.

This provision would not apply to suits brought to enforce a duty or liability created by the Exchange Act. Furthermore, Section 22 of the Securities Act creates concurrent jurisdiction for federal and state courts over all such Securities Act actions. Accordingly, both state and federal courts have jurisdiction to entertain such claims. To prevent having to litigate claims in multiple jurisdictions and the threat of inconsistent or contrary rulings by different courts, among other considerations, our certificate of incorporation will further provide that the U.S. federal district courts will be the exclusive forum for resolving any complaint asserting a cause of action arising under the Securities Act. While the Delaware courts have determined that such choice of forum provisions are facially valid, a stockholder may nevertheless seek to bring a claim in a venue other than those designated in the exclusive forum provisions. In such instance, we would expect to vigorously assert the validity and enforceability of the exclusive forum provisions of our certificate of incorporation. This may require significant additional costs associated with resolving such action in other jurisdictions and there can be no assurance that the provisions will be enforced by a court in those other jurisdictions.

These exclusive-forum provisions may limit a stockholder's ability to bring a claim in a judicial forum that it finds favorable for disputes with us or our directors, officers, or other employees. If a court were to find either exclusive forum provision of our certificate of incorporation to be inapplicable to or unenforceable in an action, we may incur further significant additional costs associated with resolving the dispute in other jurisdictions, all of which could seriously harm our business.

We do not intend to pay dividends on our capital stock, so any returns will be limited to increases in the value of our Class A common stock.

We have never declared or paid any cash dividends on our capital stock. We currently anticipate that we will retain future earnings for the operation and expansion of our business. Accordingly, we do not anticipate declaring or paying any cash dividends for the foreseeable future. In addition, our ability to pay cash dividends on our capital stock is likely to be restricted by any future debt financing arrangement we may enter into. Any return to stockholders will therefore be limited to increases in the price of our Class A common stock, if any.

We may be subject to securities litigation, which is expensive and could divert management attention.

The market price of our Class A common stock may be volatile and, in the past, companies that have experienced volatility in the market price of their stock have been subject to securities class action litigation. We may be the target of this type of litigation in the future. Securities litigation against us could result in substantial costs and divert our management's attention from other business concerns, which could seriously harm our business.

## **CAUTIONARY NOTE REGARDING FORWARD-LOOKING STATEMENTS**

This prospectus contains forward-looking statements that are based on our beliefs and assumptions and on information currently available to us. The forward-looking statements are contained principally in "Prospectus Summary," "Risk Factors," "Management's Discussion and Analysis of Financial Condition and Results of Operations" and "Business." Forward-looking statements include all statements that are not historical facts such as information concerning our possible or assumed future results of operations and expenses, business strategies and plans, competitive position, business environment and potential growth strategies and opportunities. In some cases, forward-looking statements can be identified by terms such as "anticipates," "believes," "could," "estimates," "expects," "intends," "may," "plans," "potential," "predicts," "projects," "seeks," "should," "will," "would" or similar expressions and the negatives of those terms.

Forward-looking statements involve known and unknown risks, uncertainties and other factors that may cause our actual results, performance or achievements to be materially different from any future results, performance or achievements expressed or implied by the forward-looking statements. Those risks include those described in "Risk Factors" and elsewhere in this prospectus. The inclusion of forward-looking information should not be regarded as a representation by us, the underwriters or any other person that the future plans, estimates or expectations contemplated by us will be achieved. Given these uncertainties, you should not place undue reliance on any forward-looking statements in this prospectus. Also, forward-looking statements represent our beliefs and assumptions only as of the date of this prospectus. You should read this prospectus and the documents that we have filed as exhibits to the registration statement, of which this prospectus is a part, and any related free writing prospectus, completely and with the understanding that our actual future results may be materially different from what we expect.

In addition, statements that "we believe" and similar statements reflect our beliefs and opinions on the relevant subject. These statements are based on information available to us as of the date of this prospectus. While we believe such information provides a reasonable basis for these statements, such information may be limited or incomplete. Our statements should not be read to indicate that we have conducted an exhaustive inquiry into, or review of, all relevant information. These statements are inherently uncertain, and investors are cautioned not to unduly rely on these statements.

Any forward-looking statement made by us in this prospectus speaks only as of the date on which it is made. Except as required by law, we disclaim any obligation to update these forward-looking statements publicly, or to update the reasons actual results could differ materially from those anticipated in these forward-looking statements, even if new information becomes available in the future. All forward-looking statements are expressly qualified in their entirety by the foregoing cautionary statements.

## **INDUSTRY DATA AND USER METRICS**

This prospectus contains estimates and information concerning our industry, including market position, market size, and growth rates of the markets in which we participate, that are based on industry publications and reports. This information involves a number of assumptions and limitations, and you are cautioned not to give undue weight to these estimates. We have not independently verified the accuracy or completeness of the data contained in these industry publications and reports. The industry in which we operate is subject to a high degree of uncertainty and risk due to variety of factors, including those described in the "Risk Factors" section. These and other factors could cause results to differ materially from those expressed in these publications and reports.

In addition, industry publications, studies and surveys generally state that they have been obtained from sources believed to be reliable, although they do not guarantee the accuracy or completeness of such information. While we believe that each of these studies and publications is reliable, we have not independently verified market and industry data from third parties. Further, in August 2020 we commissioned a third-party vendor to conduct a study (the "Study") involving 2,850 participants across 10 countries. All participants received the same 50 questions. Responses were analyzed to identify demographic and consumer preference data, which is referenced in this document. While we believe our internal company research and commissioned studies are reliable and the market definitions are appropriate, neither such research nor these definitions have been verified by any independent source. The third-party industry publications, studies and surveys contained in this prospectus are identified by footnotes and are provided below:

- eMarketer, Global eCommerce 2020, June 2020.
- eMarketer, US Mobile Time Spent 2020, June 2020.
- eMarketer, US Private Label vs. Branded Retail Sales Share of CPG Products, by Channel, March 2019, March 2018, and August 2019.
- Euromonitor International Limited, Economies and Consumers, updated August 2020.
- Euromonitor International Limited, Retailing 2020 edition, published July 2020.
- Sensor Tower, analysis of store intelligence platform data, November 2019.
- IDC, Understanding the Needs of the Global Small and Medium-Sized Business Market, US46393020, May 2020.
- The World Factbook 2020. Washington, DC: Central Intelligence Agency, 2020.
- · United Nations Conference on Trade and Development, June 2020.
- Small Business Administration (SBA) Office of Advocacy, Frequently Asked Questions, 2018.
- CNBC, Small Business Survey, 2017, updated May 2019.
- First Insight, The State of Consumer Spending, published March 2019.
- United Nations Conference on Trade and Development, The International Day of Micro, Small and Medium Enterprises (MSMEs), June 2020.
- U.S. Census Bureau, analysis of population data from, 2019.

The numbers of monthly active users, LTM active buyers, items, and merchants presented in this prospectus are based on internal company data, and we use these numbers in managing our business. We believe that these numbers are reasonable estimates, and we take measures to improve their accuracy, such as eliminating known fictitious or "bot" accounts. There are inherent challenges in measuring usage across large mobile populations around the world. For example, our data regarding

the geographic location of our users and merchants is based on a number of factors, which may not always accurately reflect user location. We define a merchant as of a particular date as a unique merchant account on our platform, and a seller or selling entity may have several unique merchant accounts. In addition, our MAUs and LTM active buyers are based on unique email addresses, so it is possible for a single individual with multiple email addresses to open multiple accounts. We regularly review and may adjust our processes for calculating these metrics to improve their accuracy, and therefore it is possible that in future periods our reported metrics could change as a result of these adjustments. In addition, our MAU estimates will differ from estimates published by third parties due to differences in methodology. See "Risk Factors—Risks Related to our Business and Industry—We track certain performance metrics with internal tools and do not independently verify such metrics. Certain of our performance metrics are subject to inherent challenges in measurement, and real or perceived inaccuracies in such metrics may harm our reputation and negatively affect our business."

## **USE OF PROCEEDS**

We estimate that our net proceeds from this offering will be approximately \$\frac{1}{2}\$ million, or approximately \$\frac{1}{2}\$ million if the underwriters exercise in full their option to purchase additional shares, assuming an initial public offering price of \$\frac{1}{2}\$ per share, the midpoint of the price range set forth on the cover page of this prospectus, and after deducting underwriting discounts and commissions and estimated offering expenses payable by us.

We will not receive any of the proceeds from the sale of shares our Class A common stock by the selling stockholders.

Each \$1.00 increase or decrease in the assumed initial public offering price of \$ per share, the midpoint of the price range set forth on the cover page of this prospectus, would increase or decrease, as applicable, net proceeds to us by \$ million, assuming that the number of shares offered by us on the cover page of this prospectus remains the same, and after deducting underwriting discounts and commissions and estimated offering expenses payable by us. We may also increase or decrease the number of shares we are offering. Each 1.0 million increase or decrease, as applicable, in the number of shares of our Class A common stock offered by us would increase or decrease net proceeds to us by approximately \$ million, assuming an initial public offering price of \$ per share, the midpoint of the price range set forth on the cover page of this prospectus, and after deducting underwriting discounts and commissions payable by us.

The principal purposes of this offering are to increase our capitalization and financial flexibility and create a public market for our Class A common stock, obtain additional working capital, and facilitate our future access to the public equity markets to allow us to implement our business plan. We currently intend to use the net proceeds received by us from this offering for working capital, operating expenses, sales and marketing expenses to fund the growth of our business, and capital expenditures. In addition, we may use a portion of the net proceeds to acquire complementary businesses, products, services, or technologies. However, we have no current understandings, agreements or commitments for any specific material acquisitions at this time. We cannot specify with certainty the particular uses of the net proceeds that we received from this offering. Accordingly, we will have broad discretion in using these proceeds.

Pending our use of the net proceeds received by us from this offering, we intend to invest the net proceeds in short and intermediate term, interest-bearing obligations, investment grade instruments, certificates of deposit or direct or guaranteed obligations of the U.S. government.

## **DIVIDEND POLICY**

We have never declared or paid cash dividends on our capital stock. We currently intend to retain all available funds and any future earnings for use in the operation of our business and do not anticipate paying any cash dividends in the foreseeable future. Any future decision to declare cash dividends will be made at the discretion of our board of directors, subject to applicable laws.

## **CAPITALIZATION**

The following table sets forth our cash, cash equivalents, and marketable securities, and capitalization as of September 30, 2020 on:

- · an actual basis;
- a pro forma basis to give effect to: (i) the automatic conversion of all outstanding shares of our redeemable convertible preferred stock into shares of Class A common stock, (ii) the exercise of an outstanding warrant to purchase shares of our Series B redeemable convertible preferred stock and the conversion of that Series B redeemable convertible preferred stock into shares of our Class A common stock on a net-exercise basis (assuming an initial public offering price of \$ per share, the midpoint of the price range set forth on the cover page of this prospectus), (iii) the reclassification of all outstanding shares of our common stock into an equivalent number of shares of our Class B common stock; (iv) stock-based compensation expense of approximately \$ million associated with RSUs subject to service-based and performance-based vesting conditions, which we will recognize upon the completion of this offering, as further described in Note 2 to our consolidated financial statements included elsewhere in this prospectus, which is reflected as an increase to additional paid-in capital and accumulated deficit; and (v) the filing and effectiveness of our amended and restated certificate of incorporation in Delaware; and
- a pro forma as adjusted basis to give effect to: (i) the pro forma adjustments set forth above; and (ii) our sale and issuance in this offering of shares of Class A common stock at the assumed initial public offering price of \$ per share, the midpoint of the price range set forth on the cover page of this prospectus, after deducting the underwriting discounts and commissions and estimated offering expenses payable by us.

The pro forma and pro forma as adjusted information below is illustrative only, and cash, cash equivalents and marketable securities, total stockholders' equity and total capitalization after this offering will be adjusted based on the actual initial public offering price and other terms of this offering determined at pricing. You should read this table in conjunction with the sections titled "Management's Discussion and Analysis of Financial Condition and Results of Operations" and "Description of Capital Stock" and our consolidated financial statements and related notes included elsewhere in this prospectus.

	As of September 30, 2020		
	Actual	Pro Forma (unaudited) lions, except share	Pro Forma As Adjusted <sup>(1)</sup> and per
Cash, cash equivalents, and marketable securities	\$	share data) \$	\$
Redeemable convertible preferred stock warrant liability	\$	\$ -	\$
Redeemable convertible preferred stock, \$0.0001 par value; 44,346,293 shares authorized, shares issued and outstanding, actual; no shares authorized, issued and			
outstanding, pro forma and pro forma as adjusted	\$	\$	\$
Stockholders' equity (deficit):			
Preferred stock, \$0.0001 par value; no shares authorized, no shares issued and outstanding, actual; shares authorized, no shares issued and outstanding, pro forma and pro forma as adjusted			
Common stock, \$0.0001 par value; 73,000,000 shares authorized, shares issued and outstanding, actual; no shares authorized, no shares issued and outstanding, pro forma and pro forma as adjusted	_	_	_
Class A common stock, \$0.0001 par value; no shares authorized, no shares issued and outstanding, actual; shares authorized, shares issued and outstanding, pro forma and pro forma as adjusted			
Class B common stock, \$0.0001 par value; no shares authorized, no shares issued and outstanding, actual; shares authorized, shares issued and outstanding, pro forma and pro forma as adjusted			
Additional paid-in capital	<u> </u>		
Accumulated deficit			
Total stockholders' equity (deficit)			
Total capitalization	\$	\$	\$

<sup>(1)</sup> A \$1.00 increase or decrease in the assumed initial public offering price of \$ per share, the midpoint of the price range set forth on the cover page of this prospectus, would increase or decrease, as applicable, each of cash, cash equivalents, and marketable securities, additional paid-in capital, total stockholders' equity, and total capitalization by \$ million, assuming that the number of shares offered by us, as set forth on the cover page of this prospectus, remains the same, and after deducting underwriting discounts and commissions and estimated offering expenses payable by us. We may also increase or decrease the number of shares we are offering. Each increase or decrease or 1.0 million shares in the number of shares offered by us would increase or decrease, as applicable, each of cash, cash equivalents and marketable securities, additional paid-in capital, total stockholders' equity, and total capitalization by \$ million, assuming that the assumed initial public offering price of \$ per share, the midpoint of the price range set forth on the cover page of this prospectus, remains the same, and after deducting underwriting discounts and commissions payable by us. The pro forma and pro forma as adjusted information discussed above is illustrative only and will change based on the actual initial public offering price and other terms of this offering determined at pricing.

The number of shares of our Class A common stock and Class B common stock issued and outstanding, pro forma and pro forma adjusted, in the table above is based on shares of our Class A common stock (including our redeemable convertible preferred stock on an as-converted basis) and shares of our Class B common stock outstanding as of September 30, 2020, and excludes:

- shares of our Class B common stock issuable upon the exercise of options outstanding as of September 30, 2020 under our 2010 Stock Plan, with a weighted-average exercise price of approximately \$ per share;
- shares of Class B common stock issuable upon the exercise of options granted under our 2010 Stock Plan after September 30, 2020, with a weighted-average exercise price of approximately \$ per share;
- shares of our Class B common stock issuable under RSUs that were outstanding as of September 30, 2020 for which the service condition was not satisfied as of September 30, 2020;
- shares of our Class B common stock issuable under RSUs that were granted after September 30, 2020;
- shares of our Class B common stock issuable upon exercise of a warrant at an exercise price of \$ per share; and
- shares of our common stock reserved for issuance under our equity compensation plans, consisting of:
  - shares of our Class A common stock that will be reserved for issuance under our 2020 Equity Incentive Plan,
  - shares of our Class B common stock reserved for issuance under our 2010 Stock Plan as of September 30, 2020, which will become available for issuance under our 2020 Equity Incentive Plan on the date of this prospectus, and
  - shares of our Class A common stock that will be reserved for issuance under our 2020 Employee Stock Purchase Plan.

## DILUTION

If you invest in our Class A common stock, your ownership interest will be diluted to the extent of the difference between the offering price per share of our Class A common stock and the pro forma as adjusted net tangible book value per share of our Class A common stock immediately after this offering. Dilution results from the fact that the per share offering price of our Class A common stock is substantially higher than the book value per share attributable to our existing stockholders. Pro forma net tangible book value per share represents the amount of stockholders' equity (deficit) excluding intangible assets, divided by the number of shares of common stock outstanding at that date, after giving effect to the automatic conversion of our redeemable convertible preferred stock, including the redeemable convertible preferred stock issuable upon the automatic net exercise of an outstanding warrant, immediately prior to the completion of this offering.

Our pro forma net tangible book value as of September 30, 2020 was \$ million, or \$ per share of common stock. Our pro forma net tangible book value per share represents the amount of our total tangible assets reduced by the amount of our total liabilities and divided by the total number of shares of our common stock outstanding as of September 30, 2020, after giving effect to (i) the automatic conversion of all outstanding shares of our redeemable convertible preferred stock into shares of Class A common stock, (ii) the exercise of an outstanding warrant to purchase shares of our Series B redeemable convertible preferred stock into shares of our Class A common stock on a net-exercise basis (assuming an initial public offering price of \$ per share, the midpoint of the price range set forth on the cover page of this prospectus), (iii) the reclassification of all outstanding shares of our common stock into an equivalent number of shares of our Class B common stock; and (iv) the filing and effectiveness of our amended and restated certificate of incorporation in Delaware.

After giving effect to our sale in this offering of shares of Class A common stock at an assumed initial public offering price of \$\) per share, the midpoint of the price range set forth on the cover page of this prospectus, after deducting underwriting discounts and commissions and estimated offering expenses payable by us, our pro forma as adjusted net tangible book value as of September 30, 2020 would have been approximately \$\) million, or \$\) per share of common stock. This represents an immediate pro forma as adjusted dilution of \$\) per share to investors purchasing shares in this offering.

The following table illustrates this per share dilution.

Assumed initial offering price per share	\$
Historical net tangible book value (deficit) per share as of September 30, 2020	\$
Increase per share attributable to the pro forma adjustments described above	 _
Pro forma net tangible book value per share as of September 30, 2020	 _
Increase in pro forma net tangible book value per share attributable to investors purchasing shares in this offering	
Pro forma as adjusted net tangible book value per share after giving effect to this offering	 -
Dilution in pro forma as adjusted net tangible book value per share to investors in this offering	\$

A \$1.00 increase or decrease in the assumed initial public offering price of \$ forth on the cover page of this prospectus, would increase or

per share, the midpoint of the price range set

decrease, as applicable, our pro forma as adjusted net tangible book value per share after this offering by \$ , assuming that the number of shares offered by us, as set forth on the cover page of this prospectus remains the same, and after deducting underwriting discounts and commissions, and estimated offering expenses payable by us. We may also increase or decrease the number of shares we are offering. Each increase of 1.0 million in the number of shares offered by us would increase our pro forma as adjusted net tangible book value million. or \$ per share, and the dilution per share to investors participating in this offering would be \$ share, assuming that the assumed initial public offering price remains the same, and after deducting underwriting discounts and commissions payable by us. Similarly, each decrease of 1.0 million shares in the number of shares offered by us would decrease our pro forma as adjusted net tangible book value by \$ million, or \$ per share, and the dilution per share to investors participating in this offering per share, assuming that the assumed initial public offering price remains the same, and after deducting underwriting would be \$ discounts and commissions payable by us. The pro forma as adjusted information discussed above is illustrative only and will change based on the actual initial public offering price and other terms of this offering determined at pricing.

If the underwriters exercise their option to purchase additional shares in full, the pro forma as adjusted net tangible book value per share after giving effect to this offering would be approximately \$\frac{1}{2}\text{ representing an immediate increase in pro forma as adjusted net tangible book value per share of \$\frac{1}{2}\text{ to existing stockholders and immediate dilution in pro forma as adjusted net tangible book value per share of \$\frac{1}{2}\text{ to new investors purchasing common stock in this offering, assuming an initial public offering price of \$\frac{1}{2}\text{ per share, the midpoint of the price range set forth on the cover page of this prospectus, and after deducting estimated underwriting discounts.

The following table summarizes, on a pro forma as adjusted basis described above as of September 30, 2020, the total cash consideration paid and the average price per share paid by our existing stockholders and by our new investors purchasing shares in this offering at the assumed offering price of our Class A common stock of \$ per share, the midpoint of the price range set forth on the cover page of this prospectus, before deducting underwriting discounts and commissions, and estimated offering expenses payable by us:

	Shares Purchased		Total Consideration		Average Price
	Number	Percent	Amount	Percent	Per Share
Existing stockholders		<del></del> %	\$	<del></del> %	\$
New investors					\$
Total		100%	\$	100%	

A \$1.00 increase or decrease in the assumed initial public offering price of \$ per share, the midpoint of the price range set forth on the cover page of this prospectus, would increase or decrease, as applicable, total consideration paid by new investors by \$ million and, in the case of an increase, would increase the percentage of total consideration paid by new investors by percentage points and, in case of a decrease, would decrease the percentage of total consideration paid by new investors by percentage points, assuming that the number of shares offered by us on the cover page of this prospectus remains the same, and after deducting underwriting discounts and commissions. We may also increase or decrease the number of shares we are offering. An increase or decrease of 1.0 million in the number of shares offered by us would increase or decrease, as applicable, total consideration paid by new investors by \$ million, assuming that the assumed initial public offering price remains the same, and after deducting underwriting discounts and commissions payable by us.

The table above assumes no exercise of the underwriters' option to purchase additional shares in this offering. If the underwriters exercise their option to purchase additional shares in full, our existing stockholders would own % and our new investors would own % of the total number of shares outstanding after this offering.

The foregoing tables and calculations (other than the historical net tangible book value calculation) are based on the our Class A common stock and Class B common stock that will be outstanding after this offering is based on shares of our Class A common stock (including our redeemable convertible preferred stock on an as-converted basis) and shares of our Class B common stock outstanding as of September 30, 2020, and excludes:

- shares of our Class B common stock issuable upon the exercise of options outstanding as of September 30, 2020 under our 2010 Stock Plan, with a weighted-average exercise price of approximately \$ per share;
- shares of Class B common stock issuable upon the exercise of options granted under our 2010 Stock Plan after September 30, 2020, with a weighted-average exercise price of approximately \$ per share;
- shares of our Class B common stock issuable under RSUs that were outstanding as of September 30, 2020 for which the service condition was not satisfied as of September 30, 2020;
- shares of our Class B common stock issuable under RSUs that were granted after September 30, 2020;
- shares of our Class B common stock issuable upon exercise of a warrant at an exercise price of \$ per share; and
- shares of our common stock reserved for issuance under our equity compensation plans, consisting of:
  - shares of our Class A common stock that will be reserved for issuance under our 2020 Equity Incentive Plan,
  - shares of our Class B common stock reserved for issuance under our 2010 Stock Plan as of September 30, 2020, which will become available for issuance under our 2020 Equity Incentive Plan on the date of this prospectus, and
  - shares of our Class A common stock that will be reserved for issuance under our 2020 Employee Stock Purchase Plan.

To the extent that any outstanding options or warrants are exercised, outstanding RSUs settle, new options or RSUs are issued under our stock-based compensation plans, or we issue additional shares of common stock in the future, there will be further dilution to investors participating in this offering. If we raise additional capital through the sale of equity or convertible debt securities, the issuance of these securities could result in further dilution to our stockholders.

#### SELECTED CONSOLIDATED FINANCIAL AND OTHER DATA

The following tables show selected consolidated financial data. The selected consolidated statements of operations and comprehensive loss data for the years ended December 31, 2017, 2018, and 2019, and the selected consolidated balance sheet data as of December 31, 2018 and 2019 are derived from our audited consolidated financial statements and related notes included elsewhere in this prospectus. The selected consolidated statements of operations and comprehensive loss data for the years ended December 31, 2015 and 2016 and the selected consolidated balance sheet data as of December 31, 2015, 2016, and 2017 are derived from our audited consolidated financial statements that are not included in this prospectus. The following tables also show certain operational and non-GAAP financial measures. See the accompanying footnotes and "Selected Consolidated Financial and Other Data—Non-GAAP Financial Measures" below for more information. Our historical results and key metrics are not necessarily indicative of future results.

The following selected consolidated financial data and key metrics should be read together with "Management's Discussion and Analysis of Financial Condition and Results of Operations" and our consolidated financial statements and related notes included elsewhere in this prospectus.

						ded Decembe	r 31,			
	2015 <sup>(3)</sup> 2016 <sup>(3)</sup> 2017 <sup>(3)</sup> 2018 <sup>(3)</sup> (in millions, except share and per share data)					2019 <sup>(3)</sup>				
Consolidated Statements of Operations and Comprehensive Loss Data:				(in millioi	ıs, exce	ept snare and	per sna	re data)		
Revenue	\$	144	\$	445	\$	1,101	\$	1,728	\$	1,901
Cost of revenue <sup>(1)</sup>		78		131		205		278		443
Gross profit		66		314		896		1,450		1,458
Operating expenses:(1)										
Sales and marketing		560		428		989		1,576		1,463
Product development		6		10		28		45		74
General and administrative		22		17		26		52		65
Total operating expenses		588		455		1,043		1,673		1,602
Loss from operations		(522)		(141)		(147)		(223)		(144)
Other income (expense), net:		, ,		, ,		, ,		, ,		, ,
Interest and other income (expense), net		(5)		(5)		10		15		19
Remeasurement of redeemable convertible										
preferred stock warrant liability		(10)		<u>(5</u> )		(70)				(3)
Loss before provision for income taxes		(537)		(151)		(207)		(208)		(128)
Provision for income taxes						<u> </u>				1
Net loss and comprehensive loss		(537)		(151)		(207)		(208)		(129)
Deemed dividend to redeemable convertible preferred stockholders		_		_		(40)		<u>-</u>		(7)
Net loss attributable to common stockholders	\$	(537)	\$	(151)	\$	(247)	\$	(208)	\$	(136)
Net loss per share attributable to common stockholders, basic and ${\sf diluted}^{(2)}$	\$	(56.10)	\$	(15.48)	\$	(24.60)	\$	(20.20)	\$	(13.07)
Weighted-average shares used in computing net loss per share attributable to common stockholders, basic and diluted <sup>(2)</sup>	9,	571,398	9.	,753,940	10	),038,841	10	,296,604	10	0,404,562
Pro forma net loss per share attributable to common stockholders, basic and diluted (unaudited) <sup>(2)</sup>									\$	
Weighted-average shares used in computing pro forma net loss per share attributable to common stockholders, basic and diluted (unaudited) <sup>(2)</sup>										

- (1) Cost of revenue and operating expenses include stock-based compensation expense of \$19 million, \$7 million, \$8 million, \$2 million, and \$2 million for the years ended December 31, 2015, 2016, 2017, 2018, and 2019, respectively. Following this offering, our future operating expenses, particularly in the quarter in which this offering is completed, will include substantial stock-based compensation expense with respect to our RSUs as well as any other stock-based awards we may grant in the future.
- (2) See Note 11 and Note 12 of the notes to our consolidated financial statements included in this prospectus for a description of how we compute basis and diluted net loss per share attributable to common stockholders and pro forma basic and diluted net loss per share attributable to common stockholders.
- (3) On January 1, 2018, we adopted Topic 606, on a modified retrospective basis. Accordingly, our audited consolidated financial statements for 2018 were adjusted to conform to Topic 606 and our consolidated financial statements for 2019 were reported under Topic 606. Our audited consolidated financial statements for 2015, 2016, and 2017 were reported under Topic 605. See Note 2 of the notes to our consolidated financial statements included in this prospectus.

	As of December 31,						
	2015	2016	2017	2018	2019		
Consolidated Balance Sheet Data:			(in millions)	)			
Cash, cash equivalents and marketable securities	\$ 359	\$ 752	\$ 1,122	\$ 1,014	\$ 1,078		
Working capital	57	289	352	134	134		
Total assets	434	862	1,301	1,193	1,366		
Redeemable convertible preferred stock warrant liability	51	67	124	124	127		
Redeemable convertible preferred stock	744	1,107	1,376	1,376	1,536		
Total stockholders' (deficit)	\$(692)	\$ (830)	\$(1,076)	\$(1,287)	\$(1,439)		

	Year Ended December 31,						
	2015	2016	2017	2018	2019		
		(in millio	ns, except percent	ages)			
Other Operational and Financial Data:							
Monthly Active Users ("MAUs")(1)	21	30	49	73	90		
LTM Active Buyers <sup>(1)</sup>	19	32	54	66	64		
Adjusted EBITDA <sup>(1)</sup>	\$ (502)	\$(132)	\$(135)	\$(211)	\$(127)		
Adjusted EBITDA Margin <sup>(1)</sup>	(349)%	(30)%	(12)%	(12%)	(7)%		
Free Cash Flow <sup>(1)</sup>	\$ (305)	\$ 15	\$ 134	\$ (114)	\$ (71)		

<sup>(1)</sup> See the section titled "—Other Financial Information and Data" below for more information.

#### Other Financial Information and Data

In addition to the measures presented in our consolidated financial statements, we use the following key metrics and other financial information to measure our performance, identify trends affecting our business, and make strategic decisions.

#### Monthly Active Users

We define MAUs as the number of unique users that visited the Wish platform, either on our mobile app, mobile web, or on a desktop, during the month. 14 MAUs for a given reporting period equal the average of the MAUs for that period. An active user is identified by a unique email-address; a single person can have multiple user accounts via multiple email addresses. The change in MAUs in a reported period captures both the inflow of new users as well as the outflow of existing users who did not visit the platform in a given month. We view the number of MAUs as key driver of revenue growth as well as a key indicator of user engagement and awareness of our brand.

Wish apps include Wish, Wish Local for Partner Stores, Geek, Home Design & Décor Shopping, Mama, and Cute.

#### LTM Active Buyers

As of the last date of each reported period, we determine our number of unique active buyers by counting the total number of individual users who have placed at least one order on the Wish platform, either on our mobile app, mobile web, or on desktop, during the preceding 12 months, regardless of whether the products are actually sold, delivered or returned. The number of Active Buyers is an indicator of our ability to attract and monetize a large user base to our platform and of our ability to convert visits into purchases. We believe that increasing our Active Buyers will be a significant driver to our future revenue growth.

#### Non-GAAP Financial Measures

#### Adjusted EBITDA and Adjusted EBITDA Margin

In this prospectus, we provide Adjusted EBITDA, a non-GAAP financial measure that represents our net income (loss) before interest and other income (expense), net (which includes foreign exchange gain or loss, gain or loss on one time transactions recognized), income tax expense, and depreciation and amortization, adjusted to eliminate stock-based compensation expense and remeasurement of redeemable convertible preferred stock warrant liability and to add back certain recurring other items. Additionally, in this prospectus, we provide Adjusted EBITDA Margin, a non-GAAP financial measure that represents Adjusted EBITDA divided by revenue. Below is a reconciliation of Adjusted EBITDA to net loss, the most directly comparable GAAP financial measure.

We have included Adjusted EBITDA and Adjusted EBITDA Margin in this prospectus because they are key measures used by our management and board of directors to understand and evaluate our operating performance and trends and how we are allocating internal resources, to prepare and approve our annual budget and to develop short- and long-term operating plans. We also believe that the exclusion of certain items in calculating Adjusted EBITDA can provide a useful measure for period-to-period comparisons of our business as it removes the impact of non-cash items and certain variable charges.

Adjusted EBITDA has limitations as an analytical measure, and you should not consider it in isolation or as a substitute for analysis of our results as reported under GAAP. Some of these limitations are:

- although depreciation and amortization are non-cash charges, the assets being depreciated and amortized may have to be replaced
  in the future, and Adjusted EBITDA does not reflect cash capital expenditure requirements for such replacements or for new capital
  expenditure requirements;
- Adjusted EBITDA does not reflect changes in, or cash requirements for, our working capital needs;
- Adjusted EBITDA does not consider the potentially dilutive impact of stock-based compensation;
- · Adjusted EBITDA does not reflect tax payments that may represent a reduction in cash available to us; and
- other companies, including companies in our industry, may calculate Adjusted EBITDA differently, which reduces its usefulness as a comparative measure.

Because of these limitations, you should consider Adjusted EBITDA and Adjusted EBITDA Margin alongside other financial performance measures, including various cash flow metrics, net income (loss) and our other GAAP results.

The following table reflects the reconciliation of net loss to Adjusted EBITDA and net loss as a percentage of revenue to Adjusted EBITDA margin for each of the periods indicated:

	Year Ended December 31,						
	2015	2016	2017	2018	2019		
	(in millions, except percentages)						
Revenue	\$ 144	\$ 445	\$1,101	\$1,728	\$1,901		
Net loss	(537)	(151)	(207)	(208)	(129)		
Net loss as a percentage of revenue	(373)%	(34)%	(19)%	(12)%	(7)%		
Excluding:							
Interest and other income (expense), net	5	5	(10)	(15)	(19)		
Provision for income taxes	_	_	_	_	1		
Depreciation and amortization	1	2	4	8	10		
Stock-based compensation expense	19	7	8	2	2		
Remeasurement of redeemable convertible preferred stock							
warrant liability	10	5	70	_	3		
Recurring other items				2	5		
Adjusted EBITDA	\$(502)	\$(132)	\$ (13 <del>5</del> )	\$ (211)	\$ (127)		
Adjusted EBITDA margin	(349)%	(30)%	(12)%	(12)%	(7)%		

The following table summarizes our cash flows for the periods presented:

		Year ended December 31,						
	2015	2016	2017	2018	2019			
Net Cash provided by (used in) operating activities	\$(298)	\$ 21	\$ 146	\$(94)	\$ (60)			
Net Cash used in investing activities	(113)	(23)	(192)	(16)	(40)			
Net cash provided by (used in) financing activities	508	379	212	(5)	132			

#### Free Cash Flow

In this prospectus, we also provide Free Cash Flow, a non-GAAP financial measure that represents net cash provided by (used in) operating activities less purchases of property and equipment. We believe that Free Cash Flow is an important measure since we use third parties to host our services and therefore we do not incur significant capital expenditures to support revenue generating activities.

Free cash flow has limitations as an analytical measure, and you should not consider it in isolation or as a substitute for analysis of our results as reported under GAAP. Some of these limitations are:

- it is not a substitute for net cash provided by (used in) operating activities;
- other companies may calculate free cash flow or similarly titled non-GAAP measures differently or may use other measures to evaluate their performance, all of which could reduce the usefulness of free cash flow as a tool for comparison; and
- the utility of free cash flow is further limited as it does not reflect our future contractual commitments and does not represent the total increase or decrease in our cash balance for any given period.

Because of these limitations, you should consider Free Cash Flow alongside other financial performance measures, such as net cash provided by (used in) operating activities, net income (loss) and our other GAAP results.

The following table reflects the reconciliation of net cash provided by (used in) operating activities to Free Cash Flow for each of the periods indicated:

		Year Ended December 31,					
	2015	2015 2016 2017 2018					
		(	in millions)				
Net cash provided by (used in) operating activities	\$(298)	\$21	\$146	\$ (94)	\$(60)		
Less:							
Purchases of property and equipment	7	6	12	20	11		
Free Cash Flow	<u>\$ (305)</u>	\$15	\$134	<u>\$(114</u> )	\$(71)		

### MANAGEMENT'S DISCUSSION AND ANALYSIS OF FINANCIAL CONDITION AND RESULTS OF OPERATIONS

You should read the following discussion and analysis of our financial condition and results of operations in conjunction with the section titled "Selected Consolidated Financial Data" and our consolidated financial statements and the related notes included elsewhere in this prospectus. This discussion contains forward-looking statements that involve risks and uncertainties about our business and operations. Our actual results and the timing of selected events may differ materially from those anticipated in these forward-looking statements as a result of various factors, including those we describe under the section titled "Risk Factors" and elsewhere in this prospectus.

#### Overview

We launched Wish with a simple mission—to bring an affordable and entertaining mobile shopping experience to billions of consumers around the world. Since our founding in 2010, we have become one of the largest and fastest growing global ecommerce platforms, connecting more than 100 million monthly active users in over 100 countries to over 500,000 merchants offering approximately 150 million items. Our platform combines technology and data science capabilities, an innovative and discovery-based mobile shopping experience, a comprehensive suite of indispensable merchant services, and a massive scale of users, merchants, and items. This combination has allowed us to become the most downloaded global shopping app for each of the last three years according to a report from Sensor Tower.<sup>15</sup>

We are revolutionizing mobile with a user experience that is mobile-first, discovery-based, deeply-personalized and entertaining. Over 90% of our user activity and purchases occur on our mobile app. Our data science capabilities allow us to mirror how consumers have shopped for decades in brick-and-mortar stores by offering a discovery-based shopping experience on a mobile device. Our highly-personalized product feed enables our users to discover products they want to purchase by simply scrolling through our mobile app and browsing. Over 70% of the sales on our platform do not involve a search query and instead come from personalized browsing. Wish users engage with our app in a similar manner to how they engage with social media; scrolling through image-rich, highly-engaging, and interactive content. To enhance user engagement, we incorporate fresh gamified features, rich user-generated content including photos, videos, and reviews, and a wide range of relevant products to make shopping more entertaining. Our differentiated user experience has driven superior engagement with Wish users spending on average 10 minutes per day on our platform in 2019.

We also built Wish to empower merchants around the world. We define a merchant as of a particular date as a unique merchant account on our platform. Today, most of our merchants are based in China. We initially grew our platform focusing on merchants in China, the world's largest exporter of goods for the last decade, <sup>16</sup> due to these merchants' strength in selling quality products at competitive prices. We continue to expand our merchant base around the world. The number of merchants on our platform in North America, Europe, and Latin America has grown approximately 130% since 2019. In particular, the number of merchants on our platform in the U.S. has grown approximately 150% since 2019. Through our diversified and global merchant base, we are able to offer greater depth and breadth of categories and products. We give our merchants immediate access to our global base of over 100 million monthly active users and a comprehensive suite of indispensable services, including demand generation and engagement, user-generated content creation, data intelligence, promotional and logistics capabilities, and business operations support, all in a cost-efficient manner.

Sensor Tower, analysis of store intelligence platform data, November 2019.

The World Factbook 2020. Washington, DC: Central Intelligence Agency, 2020.

Local brick-and-mortar stores worldwide are struggling to attract consumers and compete in a retail world being transformed by ecommerce and industry consolidation. We launched Wish Local in 2019 to help these merchants increase their online reach and discovery, gain foot traffic, and drive additional sales. Today, we have over 40,000 Wish Local partners in over 40 countries who have signed up to our Sell on Wish feature to upload their in-store inventory on Wish for local pick-up or delivery. Our Wish Local partners also serve as Wish Pickup locations for online Wish orders, which effectively gives us local warehousing and fulfillment footprint around the world without owning any real estate.

Our data science capabilities provide us with a unique competitive advantage and are core to our business operations. We collect, analyze, and utilize data across over 100 million monthly active users, over 500,000 merchants, and approximately 2 million items sold per day to improve the shopping experience for users and the selling experience for merchants. Our proprietary algorithms analyze a rich and growing data set of transactions and historical behaviors of both users and merchants to drive continuous optimization on the platform and inform key business decisions on a daily basis. Our data science enables personalization at the individual user level at a massive scale and drives significant advantages across all aspects of our business operations, including user acquisition, user experience, pricing strategies, user-generated content, merchant insights, and user and merchant support. For example, our user acquisition strategies utilize our data science capabilities to make decisions on what to show to whom, when, and through which acquisition channel, with a focus on maximizing our return on marketing investment and user conversion. We also leverage our data and unique insights to extend our platform outside of our core business and drive additional growth opportunities, including new services to merchants and new categories for users.

Our proprietary data and technology also fuel our powerful network effects. As Wish grows, we accumulate more data across user and merchant activities, we strengthen our data advantage, and we create an even better experience for everyone on our platform, which in turn attracts more users and merchants. As more users come to Wish, driven by the affordable value proposition and differentiated shopping experience, we drive more sales to our merchants. Adding more users also reinforces our user-generated feedback loop of ratings, reviews, photos, and videos, which drives greater user engagement. As more merchants succeed on Wish, more merchants join the platform to grow their businesses, broadening our product selection which in turn improves the user experience. This flywheel effect has driven tremendous value to both users and merchants and has made Wish one of the largest ecommerce marketplaces in the world.

We have experienced substantial growth since our founding in 2010. We grew our revenue from \$1.1 billion in 2017 to \$1.9 billion in 2019 at a compound annual growth rate of 31%. Our revenue is diversified and global. In 2019, 93% of our revenue was from marketplace services and 7% was from logistics services. Within our marketplace, we built an advertising service with annual run-rate revenue of nearly \$180 million based on the three months ended June 30, 2020. In terms of geographic diversification by users, in 2019, 49% of our marketplace revenue came from Europe, 38% from North America, 5% from South America, and 8% from the rest of the world. Our growth has been highly capital efficient. We have been able to achieve this massive growth and scale while only using \$8 million of cumulative cash flow from operations from 2017 to 2019, aided by our positive working capital dynamics. In 2019, we generated a net loss of \$129 million and Adjusted EBITDA of \$(127) million, compared to a net loss of \$208 million and Adjusted EBITDA of \$(211) million in 2018, and a net loss of \$207 million and Adjusted EBITDA of \$(135) million in 2017, which improved from (12)% of revenue in 2017 to (7)% of revenue in 2019. See "Selected Consolidated Financial and Other Data—Non-GAAP Financial Measures" for more information and for a reconciliation of Adjusted EBITDA to net loss, the most directly comparable financial measure calculated and presented in accordance with U.S. GAAP.

#### **Key Milestones**

Since our founding in 2010, we have scaled our marketplace connecting millions of users with our merchants, improved monetization of our platform, and made substantial investments to further extend our platform to new services, channels, and categories.

#### Scaling Marketplace

In 2015 and 2016, we focused on scaling our marketplace and building an affordable and entertaining mobile shopping platform.

- · On the consumer platform:
  - We utilized our data science capabilities to acquire users worldwide and grew our monthly active user base to over 48 million by 2017.
  - In 2015, we launched Blitz Buy, one of our first gamified features on the platform, which offers users once-a-day sales on a selection of extra discounted products. In 2016, we began to introduce on a limited basis, first-party sales, where we source and procure merchandise to sell on our own behalf.
- · On the merchant platform:
  - To supplement our merchant dashboard and data insights, we introduced additional features on our platform to help our merchants increase user exposure. We launched Wish Express, which helps merchants increase impressions to users who want faster delivery by designating certain eligible products as "Wish Express." We also launched ProductBoost, a native advertising tool that allows merchants to pay a fee to move their products higher up within the product rankings.

#### Improving Monetization

Starting in 2017, we increased our focus on further improving engagement and monetization of our users and merchants.

- · On the consumer platform:
  - In 2017, we began to deploy dynamic pricing at scale, which utilizes our rich data set to instantly adjust prices across
    products to optimize user conversion as well as increase our margin.
  - In 2018, we diversified our marketing efforts to include sports partnerships, TV campaigns, and our Wish Stars program.
     These additional marketing programs enabled us to acquire more users in a cost-efficient manner.
- · On the merchant platform:
  - Throughout 2017 and 2018, we continued to expand our merchant base globally.
  - In 2018, to serve our merchants' different advertising needs, we introduced, as part of our ProductBoost offerings, MaxBoost, which enables additional product impressions outside of Wish including on social media platforms, and IntenseBoost, which provides increased product impressions in a shorter period of time.
  - In 2018, we launched our logistics platform that includes a number of proprietary logistics service programs aimed at reducing cost and operational friction associated with cross-border fulfillment for our merchant base.
  - In 2018, we launched the early payment program to accelerate payment to select qualified merchants and aid their working capital management.

#### **Extending Platform**

Since 2019, we have continued to develop new services that improve the Wish user experience by continuing to offer an affordable and entertaining mobile shopping platform as well as helping our merchants succeed and grow their businesses.

- · On the consumer platform:
  - In 2019, we began to offer affordable brands and off-price branded inventory to provide our users with a larger selection of products.
  - In 2019, we launched Limited Quantity Deals ("LQD") which price higher cost items at under a dollar for a limited time only. Only one user can win the LQD, and this gamified experience drives higher engagement and overall spend on our platform.
  - o In 2019, we launched Pay Later program where users can defer payment for up to 30 days.
- · On the merchant platform:
  - In 2019, we continued to grow and diversify our merchant base.
  - o In 2019, in response to certain expected changes to the Universal Postal Union Treaty and overall increasing logistics costs globally, particularly costs related to ChinaPost services, we have invested in new logistics offerings and related technology and data science to further diversify our logistics operations. The percentage of packages shipped through our proprietary logistics platform has grown substantially from 0% in 2016 to over 90% in July 2020. Of this volume, we perform all logistic services on behalf of our merchants for approximately 50% of the packages. For the remaining 50%, merchants can choose the carrier and the desired service level based on what is available on our logistics platform. As a result, we also grew our logistics revenue from over \$6 million in 2018 to approximately \$415 million on an annual run-rate based on the three months ended June 30, 2020.
  - In 2019, we launched Wish Local to help local brick-and-mortar stores reach our massive user base, leverage our technology platform and digitize their storefronts. We have since grown our Wish Local program to over 40,000 stores in over 40 countries.
  - In 2019, we launched our tax engine that assists merchants with local tax requirements compliance.

#### **Our Financial Model**

Our business benefits from powerful network effects, fueled by our data advantage and massive scale. As more users join Wish, attracted by our affordable value proposition and shopping experience, we are able to increase revenue potential for our merchants. As more merchants succeed on Wish, more merchants join the platform and grow their businesses with Wish, broadening our product selection which in turn improves user experience. By developing a strategy focused on users and merchants, we align user and merchant success with the success of our financial model. The growth in users and merchants generates more data, further strengthens our data advantage and creates an even better experience for everyone on our platform, in turn attracting more users and high-quality merchants.

Our model relies on cost-effectively adding new users, converting those users into buyers and improving engagement and monetization of those buyers on Wish over time as well as adding new merchants, delivering economic success for those merchants, and having those merchants use more of our end-to-end platform.

The following are key elements of our financial model that drive our growth:

- Increase the scale and growth of our user and buyer base. Attracting and engaging users have been our key areas of focus since our founding. We measure our effectiveness in attracting and engaging users through metrics such as our MAUs, which increased over 40% between June 30, 2019 and June 30, 2020, as well as the average minutes spent per visit by user. These increases have primarily been driven by the growing popularity and recognition of our brand and platform, the user preferences for our differentiated mobile shopping experience, wide selection of attractively priced products, and the success of our promotional and marketing campaigns. This growth has contributed to making Wish the most downloaded global shopping app for each of the last three years according to a report from Sensor Tower. As a result, we have experienced significant revenue growth in recent periods. We will need to continue to invest in our marketing efforts to attract additional users and buyers and to enhance our brand and drive user engagement, and over time we will need to do this cost-effectively in order to achieve profitability.
- Invest in our sales and marketing engine. We have made significant investments in our data science which underpins all aspects of our operations including marketing and user acquisition. By leveraging our unique and scaled data set and algorithms, our goal is to execute cost-effective and successful digital marketing strategies to acquire new users and re-engage existing users on the Wish platform. We consider our user acquisition expertise a strong competitive advantage and have invested in this capability over time to continue to drive user and revenue growth. We will need to continue to invest in our marketing efforts to attract new users and increase user engagement.
- Increase the lifetime value of our users. Our ability to successfully drive improvements in user engagement and monetization on the Wish platform will be an important driver in expanding the lifetime value of our users and allow us to achieve profitability. We utilize data science to estimate lifetime value of users and accordingly adjust our user acquisition strategies to maximize our return on marketing investment. We also deploy dynamic pricing to drive effective monetization of our user base. Dynamic pricing utilizes our data to instantly adjust prices across products to optimize user conversion and has increased our margins.
- Continue to grow and diversify our merchant base and product categories. As of June 30, 2020, we had over 500,000 merchants on our platform. While most of our merchants today are based in China and sell unbranded goods, we plan to continue to grow our merchant base globally and diversify our product categories. While we initially grew our platform focusing on merchants in China due to their production strength, we have since started to diversify our merchant base around the world; the number of merchants on our platform in North America, Europe, and Latin America has grown approximately 130% since 2019. We believe the increase in the number and the diversity of merchants leads to more competitive pricing and broader product categories, which will in turn help us attract more users, generating powerful network effects.
- Continue to innovate and expand our platform. We generate revenue primarily from commissions earned on the sales by our merchants on the Wish marketplace, as well as from fees from offering our logistics platform to our merchants. We aim to enhance the value of our platform by broadening our marketplace service offerings, expanding the size and engagement of our user base, and improving data insights and services available to our merchants across digital marketing, payments, logistics, user support, and operations. We believe that expanding our platform will attract more merchants to Wish and drive our revenue.

In recent periods, our logistics and advertising offerings provided critical capabilities to our merchants to improve their operations and sales, which has driven their success on Wish and

Sensor Tower, analysis of store intelligence platform data, November 2019.

in turn, our growth. We have been able to drive strong adoption of these merchant offerings and intend to continue to invest to increase this adoption. For example, approximately 28% of our merchants have utilized ProductBoost in 2020 to date. This has enabled our logistics revenue and advertising revenue to reach approximately \$415 million and \$180 million, respectively, on an annual run-rate based on the three months ended June 30, 2020. We aim to continue to expand our platform so that we can grow and diversify our future revenue streams.

Our ability to manage our costs by leveraging the scale of our business. Our results of operations depend on our ability to manage our costs. While we expect our costs to increase as we grow our business, we do not expect our costs to grow at the same pace as our revenue because we do not require a proportional increase in the size of our workforce and infrastructure to support our revenue growth. This operating leverage resulted in revenue per employee of \$2.8 million in 2019 which has increased from \$2.6 million in 2017. We believe that our platform model has significant operating leverage and enables us to realize structural cost savings.

As we focused on driving the growth of our marketplace, we invested significantly in marketing activities to target and acquire users we believe will have a high propensity to transact on our platform, and promote our brand and our platform. Our sales and marketing expenses were \$1.0 billion in 2017, \$1.6 billion in 2018 and \$1.5 billion in 2019, representing 90%, 91% and 77% of revenue, respectively. While our sales and marketing expenses comprises the majority of our operating expenses, we have the flexibility to scale them up or down on a daily basis depending on the unit economics of the users. These marketing spend decisions are informed by deep data science optimized to maximize return. We anticipate that sales and marketing expenses will continue to be our most significant operating expense in the future and our overall profitability will depend on the success of our investments in sales and marketing.

Our data-driven sales and marketing strategy and lean and highly-efficient operations have allowed us to reduce our net losses from \$207 million in 2017 compared to \$129 million in 2019. As our business continues to grow, we believe we will be able to take advantage of economies of scale to further improve our operational efficiency.

• Capital efficiency of our business model. We operate an asset-light business model and have achieved substantial revenue growth with limited capital requirements. Our business model enables us to avoid the costs, risks and capital requirements associated with sourcing merchandise or holding inventory. As a result, our growth has been highly capital efficient. We have been able to achieve this massive growth and scale while only using \$8 million of cumulative cash flow from operations from 2017 to 2019, aided by our positive working capital dynamics.

#### Other Financial Information and Data

In addition to the measures presented in our consolidated financial statements, we use the following key metrics and other financial information to measure our performance, identify trends affecting our business, and make strategic decisions.

		Year Ended December 31,								
	2015	2016	2017	2018	2019					
	· · · · · · · · · · · · · · · · · · ·	(in millions, except percentages)								
MAUs	21	30	49	73	90					
LTM Active Buyers	19	32	54	66	64					
Adjusted EBITDA	\$ (502)	\$(132)	\$ (135)	\$(211)	\$(127)					
Adjusted EBITDA Margin	(349)%	(30)%	(12)%	(12)%	(7)%					
Free Cash Flow	\$ (305)	\$ 15	\$ 134	\$(114)	\$ (71)					

#### **Monthly Active Users**

We define MAUs as the number of unique users that visited the Wish platform, either on our mobile app, mobile web, or on a desktop, during the month. <sup>18</sup> MAUs for a given reporting period equal the average of the MAUs for that period. An active user is identified by a unique email-address; a single person can have multiple user accounts via multiple email addresses. The change in MAUs in a reported period captures both the inflow of new users as well as the outflow of existing users who did not visit the platform in a given month. We view the number of MAUs as key driver of revenue growth as well as a key indicator of user engagement and awareness of our brand.

#### LTM Active Buyers

As of the last date of each reported period, we determine our number of unique active buyers by counting the total number of individual users who have placed at least one order on the Wish platform, either on our mobile app, mobile web, or on a desktop, during the preceding 12 months, regardless of whether the products are actually sold, delivered or returned. The number of Active Buyers is an indicator of our ability to attract and monetize a large user base to our platform and of our ability to convert visits into purchases. We believe that increasing our Active Buyers will be a significant driver to our future revenue growth.

#### **Non-GAAP Financial Measures**

#### Adjusted EBITDA and Adjusted EBITDA Margin

In this prospectus, we provide Adjusted EBITDA, a non-GAAP financial measure that represents our net loss before interest and other income (expense), net (which includes foreign exchange gain or loss, gain or loss on one time transactions recognized), income tax expense, and depreciation and amortization, adjusted to eliminate stock-based compensation expense, remeasurement of redeemable convertible preferred stock warrant liability, and add back certain recurring other items. Additionally, in this prospectus, we provide Adjusted EBITDA Margin, a non-GAAP financial measure that represents Adjusted EBITDA divided by revenue. See "Selected Consolidated Financial and Other Data—Non-GAAP Financial Measures" for more information and for a reconciliation of Adjusted EBITDA to net loss, the most directly comparable financial measure calculated and presented in accordance with U.S. GAAP.

Wish apps include Wish, Wish Local for Partner Stores, Geek, Home Design & Décor Shopping, Mama, and Cute.

#### Free Cash Flow

In this prospectus, we also provide Free Cash Flow, a non-GAAP financial measure that represents net cash provided by (used in) operating activities less purchases of property and equipment. See "Selected Consolidated Financial and Other Data—Non-GAAP Financial Measures" for more information and for a reconciliation of Free Cash Flow to net cash provided by (used in) operating activities, the most directly comparable financial measure calculated and presented in accordance with U.S. GAAP.

#### **Components of Results of Operations**

#### Revenue

Our revenue consists of marketplace and logistics revenue.

Marketplace revenue: We provide a mix of marketplace services to merchants. We provide merchants access to our marketplace where merchants display and sell their products to users. We also provide ProductBoost services to help merchants promote their products within our marketplace. Marketplace revenue includes commission fees collected in connection with user purchases of the merchants' products. The commission fees vary depending on factors such as user location, demand, product type, and dynamic pricing. We recognize revenue when a user's order is processed and the related order information has been made available to the merchant. Commission fees are recognized net of estimated refunds and chargebacks. Marketplace revenue also includes ProductBoost fees for displaying a merchant's selected products in preferential locations within our marketplace. We recognize revenue when the merchants' selected products are displayed.

Logistics revenue: Our logistics offering for merchants, introduced in 2018, is designed for direct end-to-end single order dropship from a merchant's location to the user. Logistics services include warehousing, transportation, and delivery of the merchant's products to the user. We recognize revenue over time as the logistics services are provided.

#### Cost of Revenue and Operating Expenses

Cost of revenue. Cost of revenue includes colocation and data center charges, interchange and other fees for credit card processing services, fraud and chargeback prevention service charges, costs of refunds and chargebacks made to our users that we are not able to collect from our merchants, depreciation and amortization of property and equipment, shipping charges, tracking costs, and warehouse fees, and employee-related costs, including salaries, benefits and stock-based compensation expense for our infrastructure, merchant support, and logistics teams. Cost of revenue also includes an allocation of general IT and facilities overhead expenses.

Sales and marketing. Our sales and marketing expenses are primarily driven by the cost of acquiring and engaging users by targeting social media and search engine digital advertisements, outsourced user support services, sponsorships and local marketing campaigns, and employee-related costs, including salaries, benefits, and stock-based compensation, for our employees involved in marketing, user support, and business development functions. Sales and marketing spend also includes an allocation of general and facilities overhead expenses as well as business development expenses for attracting merchants and conducting ongoing merchant education. We expect our sales and marketing expenses to decrease as a percentage of our revenue over the long term, although our expenses may fluctuate from period to period due to the timing of expenses related to our sales and marketing campaigns.

*Product development.* Our product development expenses consist primarily of employee-related costs, including salaries, benefits, and stock-based compensation for our engineers and other employees involved in product development activities. Product development costs have historically been expensed as incurred. Product development costs also include the cost of IT used by the product development team as well as an allocation of general IT and facilities overhead expenses. We expect our product development expenses to continue to increase in absolute dollars for the foreseeable future as we continue to invest in the development of our marketplace and merchant offerings.

General and administrative. Our general and administrative expenses consist primarily of employee-related costs, including salaries, benefits, and stock-based compensation for our executives, finance, legal, information technology, human resources, and other administrative teams. General and administrative expenses also include outside consulting, legal, tax, and accounting services, and facilities and other supporting overhead costs. We expect our general and administrative expenses to continue to increase in absolute dollars for the foreseeable future as we continue to invest in our corporate infrastructure to support our revenue growth. Further, we expect to incur additional general and administrative expenses in connection with our becoming a public company.

#### Remeasurement of Redeemable Convertible Preferred Stock Warrant Liability

Remeasurement of our redeemable convertible preferred stock warrant liability is as a result of the change to the underlying redeemable convertible preferred stock value at the end of each reporting period. After the date of this offering, we will no longer incur expenses related to the change in fair value of our redeemable convertible preferred stock warrant liability.

#### Interest and Other Income (expense), net

Interest and other income (expense), net consists primarily of interest income earned on our cash, cash equivalents, and marketable securities, interest expense and foreign exchange gain or loss.

#### Income Tax

Income taxes consist primarily of income taxes in certain U.S. state and foreign jurisdictions in which we conduct business. As we have expanded our global operations, we have incurred increased foreign tax expense and we expect this to continue.

#### The Impact of COVID-19

The COVID-19 pandemic and resulting global disruptions have affected our businesses, as well as those of our users and third-party merchants. To serve our users while also providing for the safety of our employees and service providers, we have adapted numerous aspects of our supply chain, purchasing, and third-party merchant processes. We saw increased delivery times as the supply chain was disrupted and experienced a reduction in the number of merchants on our platform. While the number of merchants on our platform has recovered due to the reopening of the economy in China, we continue to see moderate disruption in the supply chain that is affecting delivery times. During this time, we temporarily shifted from air to ocean transport, which increased delivery times led to an increase in refunds. However, we benefited from greater mobile usage, online shopping, and less competition from physical retail as a result of shelter-in-place mandates. We also benefited from increased user spending due to U.S. government stimulus programs. However, such stimulus programs have recently been reduced, which may adversely impact user spending. We cannot assure that increased levels of mobile commerce will continue when the COVID-19 pandemic has subsided or otherwise, or that the U.S. government will offer additional stimulus programs.

We have also implemented a number of measures to protect the health and safety of our workforce. These measures include substantial modifications to employee travel, employee work locations, and virtualization or cancellation of meetings, among other modifications. Currently, the vast majority of our employees are working remotely. For the employees who are in the office, we are following the guidance from public health officials and government agencies, including implementation of enhanced cleaning measures, social distancing guidelines and wearing of masks.

We continue to monitor the rapidly evolving situation and expect to continue to adapt our operations to address federal, state, and local standards as well as to implement standards or processes that we determine to be in the best interests of our employees, users, and merchants.

The global outbreak of COVID-19 continues to rapidly evolve. The extent to which COVID-19 may impact our business will depend on future developments related to the geographic spread of the disease, the duration and severity of the outbreak, travel restrictions, required social distancing, governmental mandates, business closures or governmental or business disruptions, and the effectiveness of actions taken in the United States and other countries to prevent, contain and treat the virus. These impacts are highly uncertain and cannot be predicted with confidence.

#### Comparison of Years Ended December 31, 2018, and 2019

The following table summarizes our consolidated results of operations for the year ended December 31, 2018 as compared to the year ended December 31, 2019:

	Ye	Year Ended December 31,			Chan	ge
		2018		2019	\$	%
		(in millions, except perce				
Revenue	\$	1,728	\$	1,901	\$ 173	10%
Cost of revenue		278		443	165	59
Gross profit		1,450		1,458	8	1
Operating expenses:						
Sales and marketing		1,576		1,463	(113)	(7)
Product development		45		74	29	64
General and administrative		52		65	13	25
Total operating expenses		1,673		1,602	(71)	(4)
Loss from operations		(223)		(144)	79	(35)
Other income (expense), net:						
Interest and other income (expense), net		15		19	4	27
Remeasurement of redeemable convertible preferred stock warrant liability		_		(3)	(3)	n/m
Loss before provision for income taxes		(208)		(128)	80	(38)
Provision for income taxes				1	1	n/m
Net loss	\$	(208)	\$	(129)	\$ 79	(38)%

The following table presents the components of our consolidated statements of operations as a percentage of revenue:

	Year Ende December	
	2018	2019
Revenue	100%	100%
Cost of revenue	<u>16</u>	23
Gross profit	84	77
Operating expenses:		
Sales and marketing	91	77
Product development	3	4
General and administrative	3	3
Total operating expenses	97	_ 84
Loss from operations	(13)	(8)
Other income (expense), net:		
Interest and other income (expense), net	1	1
Remeasurement of redeemable convertible preferred stock warrant liability	<u>-</u> _	
Loss before provision for income taxes	(12)	(7)
Provision for income taxes	<u> </u>	
Net loss	(12)%	(7)%

#### Revenue

Y	ear Ended	d Decemb	er 31,	Chan	ige
	2018 2019		\$	%	
	(in	n millions,	, except pe	rcentages)	<u> </u>
\$	1,728	\$	1,901	\$173	10%

Revenue increased \$173 million, or 10%, to \$1,901 million for the year ended December 31, 2019 as compared to \$1,728 million for the year ended December 31, 2018. This increase was primarily attributable to increased logistics revenue and marketplace revenue.

Logistics revenue increased \$131 million, to \$137 million for the year ended December 31, 2019, as compared to \$6 million for the year ended December 31, 2018. This growth was the result of a full year of revenue and growth associated with widespread merchant adoption of our logistics platform in 2019, after launching our logistics platform in July 2018.

Marketplace revenue increased \$42 million to \$1,764 million for the year ended December 31, 2019, as compared to \$1,722 million for the year ended December 31, 2018. This growth was driven primarily by the increased adoption of ProductBoost.

#### Cost of Revenue and Gross Margin

	Year Ended December 31,			Change		
	 2018		2019	\$	%	
		rcentages)				
Cost of revenue	\$ 278	\$	443	\$165	59%	
Percentage of revenue	16%		23%			
Gross margin	84%		77%			

Cost of revenue increased \$165 million, or 59%, to \$443 million for the year ended December 31, 2019, as compared to \$278 million for the year ended December 31, 2018, primarily due to costs related to the increased volume of logistics services provided.

The gross margin decreased to 77% for the year ended December 31, 2019 from 84% in the prior year, primarily due to costs related to the increased volume of logistics services provided.

#### Sales and Marketing

	Year Ended	December 31,	Change				
	2018	2018 2019		%			
	(ii	(in millions, except percentages)					
Sales and marketing	\$ 1,576	\$ 1,463	\$(113)	(7)%			
Percentage of revenue	91%	77%					

Sales and marketing expense decreased \$113 million, or 7%, to \$1,463 million for the year ended December 31, 2019, compared to \$1,576 million for the year ended December 31, 2018. The decrease in sales and marketing expenses consisted primarily of reductions to advertising spend, due to our intentional moderation of user acquisition spend resulting from our focus on the launch of the logistics platform. We also realized \$28 million in savings from automating some of our user support.

#### **Product Development**

	Year Ended December 31,			Change		
	 2018	2	019	\$	%	
	 (in millions, except percentages)					
Product development	\$ 45	\$	74	\$29	64%	
Percentage of revenue	3%		4%			

Product development expense increased \$29 million, or 64%, to \$74 million for the year ended December 31, 2019, as compared to \$45 million for the year ended December 31, 2018, primarily as a result of an increase in employee-related costs driven by an increase in headcount in our product and engineering teams.

#### General and Administrative

		Year Ended December 31,			Change		
	_	2018		2019		%	
	_	(in millions, except percentages)					
General and administrative	\$	52	\$	65	\$13	25%	
Percentage of revenue		39	%	3%			

General and administrative expense increased \$13 million, or 25%, to \$65 million for the year ended December 31, 2019, as compared to \$52 million for the year ended December 31, 2018, primarily as a result of an increase in employee-related costs driven by an increase in headcount in our general and administrative teams.

#### Interest and Other Income (Expense), net

	<u> Y</u>	ear End	ed December 3	1,	Change			
	20	018	2019		\$	%		
		(in millions, except percentages)						
Interest and other income (expense), net	\$	15	\$	19	\$4	27%		
Percentage of revenue		1%		1%				

Interest and other income (expense), net increased \$4 million, to \$19 million of net income in 2019 compared to \$15 million of net income in 2018, primarily as a result of increased foreign exchange losses.

#### Remeasurement of Redeemable Convertible Preferred Stock Warrant Liability

	Year Ended D	ecember 31,	Change		
	2018	2019	\$	%	
	ii)	millions, except per	centages)		
Remeasurement of redeemable convertible preferred stock warrant liability	\$ -	\$ (3)	\$(3)	n/m	
Percentage of revenue	-%	-%			

The change in fair value of redeemable convertible preferred stock warrant liability for the year ended December 31, 2019 was an expense of \$3 million as compared with insignificant income for the year ended December 31, 2018. The increase is primarily due to changes in the fair value of the underlying redeemable convertible preferred stock.

#### **Provision for Income Taxes**

	Year	Year Ended December 31,			Change	
	2018	20	)19	\$	%	
		(in millions, e	xcept perce	entages)		
Provision for income taxes	\$ -	- \$	1	\$1	n/m	
Percentage of revenue	_	-%	-%			

Our provision for income taxes was due to international operations. At December 31, 2019, we had NOLs available to reduce future taxable income, if any, of \$885 million that begin to expire in 2030 and continue to expire through 2037 and \$324 million that have an unlimited carryover period. As of December 31, 2019, we had state NOLs available to reduce future taxable income, if any, of \$1.4 billion that begin to expire in 2030 and continue to expire through 2039. The utilization of NOLs to offset future taxable income may be subject to limitations under Section 382 of the Code and similar state statutes as a result of ownership changes that have occurred or could occur in the future. In addition, at the state level, there may be periods during which the use of NOLs is suspended or otherwise limited. For example, California recently imposed limits on the usability of California state NOLs to offset taxable income in tax years beginning after 2019 and before 2023. If necessary, the deferred tax assets will be reduced by any carryforward that expires prior to utilization as a result of such limitations, with a corresponding reduction of the valuation allowance.

#### Comparison of Years Ended December 31, 2017 and 2018

The following table summarizes our consolidated results of operations for the year ended December 31, 2017 as compared to the year ended December 31, 2018:

	Ye	Year Ended December 31,			Cha	nge
	2	017		2018	\$	%
		(in ı	millions	s, except pe	rcentages)	
Revenue	\$	1,101	\$	1,728	\$627	57%
Cost of revenue		205		278	73	36
Gross profit		896		1,450	554	62
Operating expenses:						
Sales and marketing		989		1,576	587	59
Product development		28		45	17	61
General and administrative		26		52	26	100
Total operating expenses		1,043		1,673	630	60
Loss from operations		(147)		(223)	(76)	52
Other income (expense), net:						
Interest and other income (expense), net		10		15	5	50
Remeasurement of redeemable convertible preferred stock warrant liability		(70)			70	(100)
Loss before provision for income taxes		(207)		(208)	(1)	_
Provision for income taxes						n/m
Net loss	\$	(207)	\$	(208)	\$ (1)	-%

The following table presents the components of our consolidated statements of operations as a percentage of revenue:

	Year En Decemb	
	2017	2018
Revenue	100%	100%
Cost of revenue	19	16
Gross profit	81	84
Operating expenses:		
Sales and marketing	90	91
Product development	3	3
General and administrative	2	3
Total operating expenses	95	3 97 (13)
Loss from operations	(13)	(13)
Other income (expense), net:		
Interest and other income (expense), net	1	1
Remeasurement of redeemable convertible preferred stock warrant liability	(6)	
Loss before provision for income taxes	(19)	(12)
Provision for income taxes		
Net loss	(19)%	(12)%

#### Revenue

	Y	ear Ended	Decem	ber 31,	Chan	ge	
		2017		2018	\$	%	
•		(in m	illions,	, except per	centages)	<del></del>	
:	\$	1.101	\$	1,728	\$627	57%	

Revenue increased \$627 million, or 57%, to \$1,728 million for the year ended December 31, 2018, as compared to \$1,101 million for the year ended December 31, 2017. The increase was primarily due to increased marketplace revenue which increased \$621 million, or 56%, to \$1,722 million for the year ended December 31, 2018, as compared to \$1,101 million for the year ended December 31, 2017, which was driven by an increase in our global user base and, to a lesser extent, increased adoption of ProductBoost.

#### Cost of Revenue and Gross Margin

		Year Ended December 31,			Change		
	<del>- :</del>	2017 2018			\$	%	
			rcentages)	<u>.</u>			
Cost of revenue	\$	205	\$	278	\$73	36%	
Percentage of revenue		19%	·	16%			
Gross margin		81%		84%			

Cost of revenue increased \$73 million, or 36%, to \$278 million for the year ended December 31, 2018 compared to \$205 million for the year ended December 31, 2017. The increase was primarily due to hosting services and other costs required to support increased revenue.

The gross margin for the year ended December 31, 2018 increased to 84%, from 81% in the prior year primarily due to the impact of ProductBoost revenue.

#### Sales and Marketing

	•	Year Ende	d December 31,	Change		
		2017 2018		\$	%	
			entages)			
Sales and marketing	\$	989	\$ 1,576	\$587	59%	
Percentage of revenue		90%	91%			

Sales and marketing expense increased \$587 million, or 59%, to \$1,576 million for the year ended December 31, 2018, as compared to \$989 million for the year ended December 31, 2017, due to an increase in digital marketing expense to expand our user base.

#### **Product Development**

	•	Year End	ded December 31,	С	Change	
	2	017	2018	\$	%	
			(in millions, except	percentages)		
Product development	\$	28	\$ 45	\$17	61%	
Percentage of revenue		3%	3	%		

Product development expense increased \$17 million, or 61%, to \$45 million for the year ended December 31, 2018, as compared to \$28 million for the year ended December 31, 2017.

The increase was primarily due to an increase in employee-related costs resulting from increased headcount in our product and engineering teams.

#### General and Administrative

	`	Year Ended December 31,			Change		
	2	2017		018	\$	%	
	·		(in millions,	except pe	rcentages)	<u>.</u>	
General and administrative	\$	26	\$	52	\$26	100%	
Percentage of revenue		2%		3%			

General and administrative expense increased \$26 million, or 100%, to \$52 million for the year ended December 31, 2018, as compared to \$26 million for the year ended December 31, 2017. The increase in general administrative expenses was primarily due to an increase in employee-related costs resulting from increased headcount in our general and administrative teams to support our growth.

#### Interest and Other Income (Expense), net

	·	Year Ende	ed December 3	Change					
	2	2017	2018		\$	%			
		(in millions, except percentages)							
Interest income (expense), net	\$	10	\$	15	\$5	50%			
Percentage of revenue		1%		1%					

Interest income (expense), net increased \$5 million to \$15 million for the year ended December 31, 2018, as compared to \$10 million for the year ended December 31, 2017, primarily due to increased interest rates on higher cash balances.

#### Remeasurement of Redeemable Convertible Preferred Stock Warrant Liability

	Year Ended Dec	Year Ended December 31, Change				
	2017	2018	\$	%		
	(in m	(in millions, except percentages)				
Remeasurement of redeemable convertible preferred stock warrant liability	\$ (70)	\$ -	\$70	(100)%		
Percentage of revenue	(6)%	-%				

The change in fair value of redeemable convertible preferred stock warrant liability for the year ended December 31, 2018 was insignificant income as compared with expense of \$70 million for the year ended December 31, 2017. The nominal income in 2018 is due to minimal change in the fair value of the underlying redeemable convertible preferred stock in 2018.

#### **Liquidity and Capital Resources**

As of December 31, 2019, we had cash, cash equivalents and current marketable securities of \$1.0 billion, a majority of which were held in cash deposits and money market funds, and were held for working capital purposes. We believe that our existing cash, cash equivalents and current marketable securities will be sufficient to meet our anticipated cash needs for at least the next 12 months.

#### Sources of Liquidity

Since our inception, we have financed our operations and capital expenditures primarily through sales of redeemable convertible preferred stock and cash flows generated by operations. Since

inception through December 31, 2019, we have raised a total of \$1.5 billion from the sale of redeemable convertible preferred stock (including redeemable convertible preferred stock warrant exercises), net of costs and expenses associated with such financings, and net of repurchases.

#### Cash Flows

	Year	r Ended December 31,			
	2017	2018	2019		
		(in millions)			
Cash provided by (used in):					
Operating activities	\$ 146	\$ (94)	\$ (60)		
Investing activities	(192)	(16)	(40)		
Financing activities	212	(5)	132		

#### Net Cash Provided by (Used in) Operating Activities

Our cash flows from operations are largely dependent on the amount of revenue we generate. Net cash provided by operating activities in each period presented has been influenced by changes in funds receivable, prepaid expenses, and other current and noncurrent assets, accounts payable, merchants payable, accrued and refund liabilities, lease liabilities, and other current and noncurrent liabilities.

Net cash used in operating activities was \$60 million in the year ended December 31, 2019, as a result of net loss of \$129 million, off-set by non-cash expense of \$21 million, which was offset by favorable net working capital. Favorable working capital movement was driven by accounts payable and accrued and refund liabilities due to increased operations.

Net cash used in operating activities was \$94 million in the year ended December 31, 2018, as a result of net loss of \$208 million, off-set by non-cash expense of \$1 million, which was further offset by favorable net working capital. Favorable working capital movement was driven by accounts payable and accrued and refund liabilities due to increased operations.

Net cash provided by operating activities was \$146 million in the year ended December 31, 2017 as a result of net loss of \$207 million, off-set by non-cash expense of \$85 million (primarily remeasurement of redeemable convertible preferred stock warrant liability of \$70 million) and favorable net working capital. Favorable working capital movement was driven by accounts payable and accrued and refund liabilities due to increased operations and as a function of our business model, in which cash is generally collected within a few days and remitted to merchants a few weeks later.

#### Net Cash Used in Investing Activities

Our primary investing activities have consisted of investing excess cash balances in marketable securities and also have consisted of capital expenditures which are primarily purchases of property and equipment.

Net cash used in investing activities was \$40 million in the year ended December 31, 2019. This was primarily due to purchases of marketable securities, net of sales and maturities, of \$29 million and capital expenditures of \$11 million.

Net cash used in investing activities was \$16 million in the year ended December 31, 2018. This was primarily due to capital expenditures of \$20 million and \$5 million in purchases of marketable securities, net of sales and maturities, partially offset by \$9 million proceeds from sale of a cost method investment.

Net cash used in investing activities was \$192 million in the year ended December 31, 2017. This was primarily due to capital expenditures of \$12 million and \$180 million in purchases of marketable securities, net of sales and maturities.

#### Net Cash Provided by (Used in) Financing Activities

Net cash provided by financing activities was \$132 million in the year ended December 31, 2019. This was primarily due to net proceeds of \$160 million from the sale of our Series H redeemable convertible preferred stock. This was partially off-set by \$28 million in repurchases of our capital stock.

Net cash used in financing activities was \$5 million in the year ended December 31, 2018, primarily due to repurchases of our capital stock in 2018.

Net cash provided by financing activities was \$212 million in the year ended December 31, 2017. This was primarily due to proceeds of \$226 million from the sale of our Series G redeemable convertible preferred stock and proceeds of \$35 million from the exercise of redeemable convertible preferred stock warrants. This was partially off-set by \$48 million in repurchases of our capital stock.

#### **Off Balance Sheet Arrangements**

We did not have any off balance sheet arrangements, as defined in Item 303(a)(4)(ii) of Regulation S-K, in the years ended December 31, 2017, 2018, or 2019.

#### Contingencies

We are involved in claims, lawsuits, government investigations, and proceedings arising from the ordinary course of our business. We record a provision for a liability when we believe that it is both probable that a liability has been incurred, and the amount can be reasonably estimated. Significant judgment is required to determine both probability and the estimated amount. Such legal proceedings are inherently unpredictable and subject to significant uncertainties, some of which are beyond our control. Should any of these estimates and assumptions change or prove to be incorrect, it could have a material impact on our results of operations, financial position, and cash flows.

#### **Contractual Obligations**

The following table summarizes our future fixed contractual obligations as of December 31, 2019:

		Payments Due by Period							
	Total	Less than al 1 year					3 to 5 years		than ears
				(in m	illions)				
Purchase obligations <sup>(1)</sup>	\$129	\$	64	\$	65	\$	_	\$	_
Operating lease obligations <sup>(2)</sup>	58		11		22		20		5
Total contractual obligations	\$187	\$	75	\$	87	\$	20	\$	5

1) Purchase obligations are for purchases of colocation and cloud services and marketing services under non-cancelable contracts.

#### **Critical Accounting Policies and Estimates**

Our management's discussion and analysis of our financial condition and results of operations is based on our consolidated financial statements, which have been prepared in accordance with GAAP.

<sup>(2)</sup> Operating leases include office space and equipment under non-cancelable operating leases, primarily due to our headquarters in San Francisco, California and for our other offices in the U.S., Canada, Europe and China.

The preparation of these consolidated financial statements requires us to make estimates and assumptions that affect the reported amounts of assets, liabilities, revenue, expenses and related disclosures. We evaluate our estimates and assumptions on an ongoing basis. Our estimates are based on historical experience and various other assumptions that we believe to be reasonable under the circumstances. Our actual results could differ from these estimates.

We believe that the assumptions and estimates associated with revenue recognition, operating lease obligations, stock-based compensation, valuation of common stock, redeemable convertible preferred stock, redeemable convertible preferred stock warrants, and income taxes have the greatest potential impact on our consolidated financial statements. Therefore, we consider these to be our critical accounting policies and estimates. For further information on all of our significant accounting policies, see Note 1 of the accompanying notes to our consolidated financial statements.

#### Revenue Recognition

Through the fiscal year ended December 31, 2017, in accordance with Accounting Standards Codification, ("ASC"), Topic 605, Revenue Recognition ("Topic 605"), we recognized revenue when all of the following conditions were satisfied: (1) there was persuasive evidence of an arrangement; (2) delivery has occurred or the service has been rendered; (3) collectability was reasonably assured; and (4) the selling price or fee was fixed or determinable. We evaluated whether it was appropriate to recognize revenue on a gross or net basis based upon its evaluation of whether it was the primary obligor in a transaction, had inventory risk and had latitude in establishing pricing and selecting suppliers, among other factors.

On January 1, 2018, we adopted Accounting Standards Update No. 2014-09, *Revenue from Contracts with Customers* ("Topic 606"), which supersedes the revenue recognition requirements in Topic 605, using the modified retrospective transition method. We now recognize revenue upon transfer of control of promised products or services to customers in an amount that reflects the consideration we expect to receive in exchange for those products or services. Under the new revenue recognition standard, we apply the following five step approach: (1) identify the contract with a customer, (2) identify the performance obligations in the contract, (3) determine the transaction price, (4) allocate the transaction price to the performance obligations in the contract, and (5) recognize revenue when a performance obligation is satisfied.

Results for the years ended December 31, 2018 and 2019 are presented under Topic 606, while the prior year results have not been adjusted and continue to be reported in accordance with our historic accounting under Topic 605. Upon the adoption of Topic 606, we recorded the costs of free products to users and costs of certain refunds as reduction of revenue in the consolidated statements of operations and comprehensive loss. These costs were previously recorded in cost of revenue under Topic 605. There was no adjustment to accumulated deficit on January 1, 2018.

We generate revenue from marketplace and logistics services provided to merchants. Revenue is recognized as we transfer control of promised goods or services to our customers, in an amount that reflects consideration we expect to be entitled to in exchange for those goods or services. We consider both the merchant and the user to be customers. We evaluate whether it is appropriate to recognize revenue on a gross or net basis based upon our evaluation of whether we obtain control of the specified goods or services by considering if it is primarily responsible for fulfillment of the promise, has inventory risk and has latitude in establishing pricing and selecting suppliers, among other factors. Based on these factors, marketplace revenue is generally recorded on a net basis and logistics revenue is generally recorded on a gross basis. Revenue excludes any amounts collected on behalf of third parties, including indirect taxes.

#### Marketplace Revenue

We provide a mix of marketplace services to merchants. We provide merchants access to our marketplace where merchants display and sell their products to users. We also provide ProductBoost services to help merchants promote their products within our marketplace.

Marketplace revenue includes commission fees collected in connection with user purchases of the merchants' products. The commission fees vary depending on factors such as user location, demand, product type, and dynamic pricing. We recognize revenue when a user's order is processed and the related order information has been made available to the merchant. Commission fees are recognized net of estimated refunds and chargebacks. Marketplace revenue also includes ProductBoost fees for displaying a merchant's selected products in preferential locations within our marketplace. We recognize revenue when the merchants' selected products are displayed.

#### Logistics Revenue

Our logistics offering for merchants, introduced in 2018, is designed for direct end-to-end single order dropship from a merchant's location to the user. Logistics services include warehousing, transportation, and delivery of the merchant's products to the user. We recognize revenue over time as the logistics services are provided.

#### Deferred Revenue

Deferred revenue consists of amount received related to unsatisfied performance obligations at the end of the period. Due to the short-term duration of contracts, all of the performance obligations will be satisfied in the following reporting period.

#### Refunds and Chargebacks

Refunds and chargebacks are associated with marketplace revenue. Returns are not material to our business. Estimated refunds and chargebacks are recorded on the consolidated balance sheets as refunds liability. The merchants' share of the refunds are recorded as a reduction to the amount due to merchants. The revenue recognized on transactions subject to refunds and chargebacks is reversed. We estimate future refunds and chargebacks using a model that incorporates historical experience, and considering recent business trends and market activity.

#### Incentive Discount Offers

We provide incentive discount offers to our users to encourage purchases of goods through our marketplace. Such offers include current discount offers of a certain percentage off current purchases, and inducement offers, such as set percentage offers off future purchases subject to a minimum current purchase. We generally record the related discounts taken as a reduction of revenue when the offer is redeemed. We also offer free products to encourage users to make purchases on our marketplace. The resulting discount is recorded as a reduction of revenue when the offer for free product is redeemed.

#### Lease Obligations

We lease facilities and data center co-locations in multiple locations under non-cancelable lease agreements through 2025. We determine if an arrangement is a lease at inception. For leases where we are the lessee, ROU assets represent our right to use the underlying asset for the term of the lease and the lease liabilities represent an obligation to make lease payments arising from the lease. Certain

lease agreements contain tenant improvement allowances, rent holidays and rent escalation provisions, all of which are considered in determining the ROU assets and lease liabilities. We begin recognizing rent expense when the lessor makes the underlying asset available for use by us. Lease liabilities are recognized at the lease commencement date based on the present value of the future lease payments over the lease term. Lease renewal periods are considered on a lease-by-lease basis in determining the lease term. The interest rate we use to determine the present value of future lease payments is our incremental borrowing rate, which is the estimated rate we would be required to pay for a collateralized borrowing equal to the total lease payments over the term of the lease. We estimate our incremental borrowing rate based on an analysis of publicly-traded debt securities of companies with credit and financial profiles similar to our own. The ROU asset is determined based on the lease liability initially established and adjusted for any prepaid lease payments and any lease incentives received. The lease term to calculate the ROU asset and related lease liability includes options to extend or terminate the lease when it is reasonably certain that we will exercise the option. Certain leases contain variable costs, such as common area maintenance, real estate taxes, or other costs. Variable lease costs are expensed as incurred on the consolidated statements of operations and comprehensive loss.

#### Stock-Based Compensation

We have granted stock options and RSUs to employees and nonemployees. We measure stock options based on their estimated grant date fair values, which we determine using the Black-Scholes option-pricing model. We record the resulting expense in the consolidated statements of operations and comprehensive loss over the period of service required to vest in the award, which is generally four or five years. We account for forfeitures as they occur.

RSUs that we grant to employees vest upon the satisfaction of both a service condition and a liquidity condition. The service condition for these awards is satisfied over four or five years. The liquidity condition is satisfied upon the occurrence of a qualifying event, defined as a change of control transaction or an IPO. We measure RSUs granted to employees based on the fair market value of its common stock on the grant date. We have not recorded any stock-based compensation expense for RSUs as of December 31, 2019, because a qualifying event has not occurred. If a qualifying event occurs in the future, we will record cumulative stock-based compensation expense using the accelerated attribution method for those RSUs for which the service condition has been satisfied prior to the qualifying event, and we will record the remaining unrecognized stock-based compensation expense over the remainder of the requisite service period. If the qualifying event had occurred on December 31, 2019, we would have recorded cumulative stock-based compensation expense of approximately \$283 million for the year ended December 31, 2019, and would recognize the remaining \$120 million of unrecognized stock-based compensation expense over a weighted-average period of approximately 1.82 years.

#### Valuation of Common Stock, Redeemable Convertible Preferred Stock, and Redeemable Convertible Preferred Stock Warrants

We determined the fair value of common stock, redeemable convertible preferred stock, and redeemable convertible preferred stock warrant in 2017, 2018, and 2019 based on the market approach under which our equity value is estimated by applying valuation multiples derived from the observed valuation multiples of comparable public companies to management forecasted financial results.

We used the Probability Weighted Expected Return Method ("PWERM") to allocate our equity value among outstanding common stock, redeemable convertible preferred stock, warrants, and equity awards. We applied the PWERM by first defining the range of potential future liquidity

outcomes, such as an IPO, and then allocating its value to outstanding stock, warrants and equity awards based on the relative probability that each outcome will occur. The valuation and allocation approaches involve the use of estimates, judgments, and assumptions that are highly complex and subjective, such as those regarding expected future revenue, expenses and cash flows, as well as discount rates, valuation multiples, the selection of comparable public companies and the probability of future events. Changes in any or all of these estimates and assumptions, or the relationships between these assumptions, impact our valuation as of each valuation date and may have a material impact on the valuation of our common stock, redeemable convertible preferred stock, and redeemable convertible preferred stock warrants

#### Income Taxes

We account for income taxes using the asset and liability method, under which deferred tax liabilities and assets are recognized for the expected future tax consequences of temporary differences between consolidated financial statement carrying amounts and the tax basis of assets and liabilities and net operating loss and tax credit carryforwards. Valuation allowances are established when necessary to reduce deferred tax assets to the amount expected to be realized.

We determine whether it is more likely than not that a tax position will be sustained upon examination. If it is not more likely than not that a position will be sustained, no amount of benefit attributable to the position is recognized. The tax benefit to be recognized for any tax position that meets the more likely than not recognition threshold is calculated as the largest amount that is more than 50% likely of being realized upon resolution of the contingency.

It is our policy to include penalties and interest expense related to income taxes as a component of interest and other income (expense), net as necessary.

#### **Recent Accounting Pronouncements**

Refer to Note 2 to our consolidated financial statements included elsewhere in this prospectus for accounting pronouncements recently adopted and recent accounting pronouncements not yet adopted as of the date of this prospectus.

#### **Quantitative and Qualitative Disclosures About Market Risk**

We have operations both within the United States and internationally, and we are exposed to market risks in the ordinary course of our business, including the effects of interest rate changes and foreign currency fluctuations. Information relating to quantitative and qualitative disclosures about these market risks is described below.

#### Interest Rate Sensitivity

Cash, cash equivalents and marketable securities as of December 31, 2019 were held primarily in cash deposits and money market funds. The fair value of our cash, cash equivalents, and investments would not be significantly affected by either an increase or decrease in interest rates due mainly to the short-term nature of these instruments. A 100 basis point increase or decrease in our current interest rates could increase or decrease our interest income by \$11 million.

#### Foreign Currency Risk

Our sales to users are denominated in local currencies, primarily in U.S. dollars and Euros. We believe we have minimal exposure to Euros as we convert Euros into U.S. dollars a few days after

receipt. We do pay our merchants in China in Renminbi and that creates an exposure for a short period of time. Fluctuations in foreign currency exchange rates may cause us to recognize transaction gains and losses in our statement of operations. A 10% increase or decrease in current exchange rates could result in additional income or expense of \$29 million.

#### Inflation Risk

We do not believe that inflation has had a material effect on our business, financial condition, or results of operations.

#### **BUSINESS**

#### Overview

We launched Wish with a simple mission — to bring an affordable and entertaining mobile shopping experience to billions of consumers around the world. Since our founding in 2010, our vision has been to unlock ecommerce for consumers and merchants, by providing consumers access to a vast selection of affordable products and by providing merchants access to hundreds of millions of consumers globally. We have become one of the largest and fastest growing global ecommerce platforms, connecting more than 100 million monthly active users in over 100 countries to over 500,000 merchants offering approximately 150 million items. Our platform combines technology and data science capabilities, an innovative and discovery-based mobile shopping experience, a comprehensive suite of indispensable merchant services, and a massive scale of users, merchants, and items. This combination has allowed us to become the most downloaded global shopping app for each of the last three years according to a report from Sensor Tower 19

We are focused on democratizing mobile commerce by making it affordable and accessible to anyone. The global mobile commerce market was \$2.1 trillion in 2019 and is expected to more than double to reach \$4.5 trillion by 2024.20 While ecommerce grew from 3% of global commerce in 2010 to 14% in 2019, ecommerce companies have largely focused on serving affluent consumers by offering branded goods and prioritizing convenience over price. However, 44% of U.S. consumers and 85% of European consumers have a household income of less than \$75,00021 and cannot afford many traditional ecommerce offerings. Additionally, in the emerging economies of Africa, the Middle East, Latin America, and Eastern Europe, where the average household income is approximately \$18,000, affordability will be the key element for users shopping online. We believe that the next billion ecommerce customers will be these value-conscious consumers. According to a survey we conducted in 2020 across 2,850 consumers in select countries, approximately 75% of those responding prioritize the price of an item over brand and delivery time. We built Wish to serve these consumers who favor affordability over brand and convenience, and are being underserved by traditional ecommerce platforms.

We are revolutionizing mobile commerce with a user experience that is mobile-first, discovery-based, deeply-personalized and entertaining. Over 90% of our user activity and purchases occur on our mobile app. Our data science capabilities allow us to mirror how consumers have shopped for decades in brick-and-mortar stores by offering a discovery-based shopping experience on a mobile device. Our highly-personalized product feed enables our users to discover products they want to purchase by simply scrolling through our mobile app and browsing. Over 70% of the sales on our platform do not involve a search query and instead come from personalized browsing. Wish users engage with our app in a similar manner to how they engage with social media; scrolling through image-rich, highly-engaging, and interactive content. To enhance user engagement, we incorporate fresh gamified features, rich user-generated content including photos, videos, and reviews, and a wide range of relevant products to make shopping more entertaining. Our differentiated user experience has driven superior engagement with Wish users spending on average 10 minutes per day on our platform in 2019.

We also built Wish to empower merchants around the world. Today, most of our merchants are based in China. We initially grew our platform focusing on merchants in China, the world's largest exporter of goods for the last decade, 22 due to these merchants' strength in selling quality products at

20 eMarketer, Global eCommerce 2020, June 2020.

Sensor Tower, analysis of store intelligence platform data, November 2019.

Euromonitor International Limited, Economies and Consumers, updated August 2020. The World Factbook 2020. Washington, DC: Central Intelligence Agency, 2020.

competitive prices. We continue to expand our merchant base around the world. The number of merchants on our platform in North America, Europe, and Latin America has grown approximately 130% since 2019. In particular, the number of merchants on our platform in the U.S. has grown approximately 150% since 2019. Through our diversified and global merchant base, we are able to offer greater depth and breadth of categories and products. We give our merchants immediate access to our global base of over 100 million monthly active users and a comprehensive suite of indispensable services, including demand generation and engagement, user-generated content creation, data intelligence, promotional and logistics capabilities, and business operations support, all in a cost-efficient manner.

Local brick-and-mortar stores worldwide are struggling to attract consumers and compete in a retail world being transformed by ecommerce and industry consolidation. We launched Wish Local in 2019 to help these merchants increase their online reach and discovery, gain foot traffic, and drive additional sales. Today, we have over 40,000 Wish Local partners in over 40 countries who have signed up to our Sell on Wish feature to upload their in-store inventory on Wish for local pick-up or delivery. Our Wish Local partners also serve as Wish Pickup locations for online Wish orders, which effectively gives us local warehousing and fulfillment footprint around the world without owning any real estate.

Our scale, combined with our extensive data science capabilities, provides us with a unique competitive advantage and is core to our business operations. We collect, analyze, and utilize data across over 100 million monthly active users, over 500,000 merchants, and approximately 2 million items sold per day to improve the shopping experience for users and the selling experience for merchants. Our proprietary algorithms analyze a rich and growing data set of transactions and historical behaviors of both users and merchants to drive continuous optimization on the platform and inform key business decisions on a daily basis. Our data science enables personalization at the individual user level at a massive scale and drives significant advantages across all aspects of our business operations, including user acquisition, user experience, pricing strategies, user-generated content, merchant insights, and user and merchant support. For example, our user acquisition strategies utilize our data science capabilities to make decisions on what to show to whom, when, and through which acquisition channel, with a focus on maximizing our return on marketing investment and user conversion. We also leverage our data and unique insights to extend our platform outside of our core business and drive additional growth opportunities, including new services to merchants and new categories for users.

Our proprietary data and technology also fuel our powerful network effects. As Wish grows, we accumulate more data across user and merchant activities, we strengthen our data advantage, and we create an even better experience for everyone on our platform, which in turn attracts more users and merchants. As more users come to Wish, driven by the affordable value proposition and differentiated shopping experience, we drive more sales to our merchants. Adding more users also reinforces our user-generated feedback loop of ratings, reviews, photos, and videos, which drives greater user engagement. As more merchants succeed on Wish, more merchants join the platform to grow their businesses, broadening our product selection which in turn improves the user experience. This flywheel effect has driven tremendous value to both users and merchants and has made Wish one of the largest ecommerce marketplaces in the world.

We have experienced substantial growth since our founding in 2010. We grew our revenue from \$1.1 billion in 2017 to \$1.9 billion in 2019 at a compound annual growth rate of 31%. Our revenue is diversified and global. In 2019, 93% of our revenue was from marketplace services and 7% was from logistics services. Within our marketplace, we built an advertising service with annual run-rate revenue of nearly \$180 million based on the three months ended June 30, 2020. In terms of geographic diversification by users, in 2019, 49% of our marketplace revenue came from Europe, 38% from North

America, 5% from South America, and 8% from the rest of the world. Our growth has been highly capital efficient. We have been able to achieve this massive growth and scale while only using \$8 million of cumulative cash flow from operations from 2017 to 2019, aided by our positive working capital dynamics. In 2019, we generated a net loss of \$129 million and Adjusted EBITDA of \$(127) million, compared to a net loss of \$208 million and Adjusted EBITDA of \$(211) million in 2018, and a net loss of \$207 million and Adjusted EBITDA of \$(215) million in 2017, which improved from (12)% of revenue in 2017 to (7)% of revenue in 2019. See "Selected Consolidated Financial and Other Data—Non-GAAP Financial Measures" for more information and for a reconciliation of Adjusted EBITDA to net loss, the most directly comparable financial measure calculated and presented in accordance with U.S. GAAP.

#### **Our Market Opportunity**

Global ecommerce is a massive and growing market. In 2019, global ecommerce was a \$3.4 trillion market expected to nearly double to reach \$6.3 trillion by 2024. Excluding China, the global ecommerce market was \$1.6 trillion in 2019 and is expected to grow to \$2.7 trillion by 2024.23 Within ecommerce, mobile is the clear dominant force, comprising 63% of global ecommerce in 2019, and is expected to grow to 71% by 2024.24 While the market is large and rapidly growing, modern ecommerce has not evolved to fit the expectations and affordability needs of the global population.

#### **Billions of Value-Conscious Consumers Are Underserved**

Value-conscious consumers represent a large and growing portion of the global consumer population, and they have been historically underserved by traditional ecommerce. For this segment of the global population, price is often the single most important determinant when making a purchase. Forty-four percent of U.S. consumers and 85% of European consumers have a household income of less than \$75,000,25 and we estimate that there are over 1 billion households with income of less than \$75,000 around the world, excluding China and India. Additionally, in the emerging economies of Africa, the Middle East, Latin America, and Eastern Europe, where the average household income is approximately \$18,000, affordability will be the key element for users shopping online for the first time. We believe that the next billion ecommerce customers will be these value-conscious consumers.

This value-conscious population has significant and highly resilient buying power which can be evidenced by the success of discount retailers. While traditional brick-and-mortar retail continues to face challenges, discount retailers remain resilient as bargain-hunting consumers continue to shop at value-focused stores. Between 2019 and 2021, the number of variety store retailer locations in the U.S. is expected to increase.<sup>26</sup> Despite their sizable presence, these discount retailers have achieved limited success in moving their low-price point business models online.

#### Traditional Ecommerce Does Not Meet Evolving Consumer Behavior

Discovery is a foundational aspect of the brick-and-mortar shopping experience. Shopping in store allows consumers to browse and discover new products that they want, driving many consumers to purchase items beyond their planned purchases. This type of navigational browsing often creates purchase intent for new products. A study conducted in 2019 by First Insight, a provider of data analytics for the retail industry, found that 89% of women and 78% of men who visit physical stores frequently add additional items to their cart beyond their identified need.27

eMarketer, Global eCommerce 2020, June 2020. eMarketer, Global eCommerce, 2020.

Euromonitor International Limited, Economies and Consumers, updated August 2020.

Euromonitor International Limited, Retailing 2020 edition, published July 2020.

First Insight, The State of Consumer Spending, March 2019.

driven experience to mobile, consumers can shop for what they know they need, but struggle to browse, engage, and discover new products. On average, people spend 3 hours per day<sup>28</sup> on their mobile device, primarily on social and video mobile apps.<sup>29</sup> As consumers spend more and more of their time on mobile, they expect visually rich, engaging, and personalized experiences. Mobile has become an effective platform for discovery in other verticals outside of ecommerce such as social media, gaming, music, and video, but we believe that online shopping has lagged in offering mobile discovery.

#### Merchants Around the World Lack Access to Global Consumers

Ecommerce represents a massive opportunity for global merchants, but they often struggle to access and serve global ecommerce consumers. IDC estimates that there are approximately 348 million SMBs around the world as of 2019,30 and the success of these businesses is imperative to local, national, and global economies. SMBs make up over 90% of all enterprises globally and employ over 50% of the global workforce.31 Despite their scale and economic power collectively, individual SMBs are often challenged, as evidenced by approximately 20% of U.S. SMBs failing in their first year and approximately 50% failing after five years in business, according to 2019 data. For merchants around the world to grow their businesses, the ability to reach and target consumers at scale is critical.

#### Merchants Lack Technology-Driven Tools to Operate Their Businesses

Many merchants lack the tools to operate their businesses efficiently, including expertise and resources to acquire users, shipping and logistics platform, payment capabilities, access to credit, effective user support, and comprehensive data insights. For example, based on an independent survey, almost half (45%) of SMBs in the United States have reported a lack of any online presence, demonstrating challenges these merchants face in running their businesses in an increasingly digital world. 33

#### The Wish Platform

Our global ecommerce platform connects over 100 million monthly active users in over 100 countries to over 500,000 global merchants. We seek to democratize ecommerce by making the Wish platform affordable, open, and accessible to all users and merchants worldwide. We do this through our relentless focus on product, technology, and data science. For our users, we are revolutionizing the mobile shopping experience by making it affordable, personalized, and entertaining. For our merchants, we offer immediate, cost-efficient access to our global user base, scaled data, and technology platform, as well as a comprehensive suite of indispensable services to help run their businesses and drive sales. To serve our global and diversified user and merchant base, we approach our platform development with a specific geographic focus, tailoring key features to solve for the needs of that locality, and enabling an authentic, localized experience.

#### Value Proposition to Wish Users

Today, we serve more than 100 million monthly active users and more than 60 million annual buyers. Our users are global and diverse, representing a wide range of age demographics and

- eMarketer, US Mobile Time Spent 2020, June 2020.
- eMarketer, US Mobile Time Spent 2020, June 2020. IDC, Understanding the Needs of the Global Small and Medium-Sized Business Market, US46393020, May 2020.
- United Nations Conference on Trade and Development, The International Day of Micro, Small and Medium Enterprises (MSMEs), June 2020. Small Business Administration (SBA) Office of Advocacy, Frequently Asked Questions, 2018. CNBC, Small Business Survey, 2017, updated May 2019.

household incomes. Forty-three percent of our users are millennials and the median household income of our users is approximately \$50,000, as of 2020 based on our Survey.

We have democratized ecommerce by making it:

- Affordable. Price is the single most important determinant when making a purchase for a substantial portion of the global population, and we aim to serve the affordability needs of these consumers. According to our Survey of 2,850 consumers in select countries, approximately 75% of those responding prioritize the price of an item over brand and delivery time. Additionally, 95% of the Survey participants who shop on Wish stated that they find items on Wish to be at a discount to branded alternatives. The merchants on our platform offer primarily unbranded products that can be discounted in excess of 85% as compared to branded alternatives across a number of categories, such as wireless ear buds and air fryers. For example, based on our internal research, the average price of the top 20 wireless ear buds sold on Wish is approximately \$20, compared to the online retail average price of a select set of recognized branded alternatives of approximately \$165, and the average price of the top 20 air fryers sold on Wish is approximately \$85, compared to approximately \$173 on average for a select set of recognized branded alternatives.<sup>34</sup> We have a number of policies which, in combination with our robust user-generated content, promote higher quality merchants and products on our platform. This allows us to offer a vast selection of high-quality items at competitive prices, a value proposition that attracts more than 100 million monthly active users to our platform.
- Accessible. Making Wish open and accessible to everyone is core to our strategy. Within ecommerce, mobile is the clear dominant force, comprising 63% of global ecommerce in 2019, and is expected to grow to 71% by 2024.35 We built Wish to be mobile-first so any consumer around the world can easily access our shopping platform on a mobile device. We also do not have membership fees, understanding that millions of consumers are value-conscious and/or would not be able to afford such fees.
- Since our founding we have continuously expanded our global footprint, with the result that we have users in over Everywhere. 100 countries today. To better serve our global user base, we localize various features on our online platform and tailored our experience to each respective market through, for example, making it accessible in 40 different languages and providing countryspecific payment methods. This localization improves the engagement of our large, diverse user base, in addition to connecting our users with our online merchants, through the launch of Wish Local in 2019, we started connecting users with local brick-and-mortar stores. Wish Local allows our users to shop online and also stop by our more than 40,000 Wish Local partner stores to pick up products that they purchase through our online platform. Our users can support their local businesses through increased foot traffic. which drives additional sales.

We have re-invented the online shopping experience to be:

- Mobile-First. Wish was built for mobile. Our application is deliberately designed to be image-rich, with minimal search input or text-based interactions. Over 90% of our user activity and purchases occur on our mobile app.
- Our mobile shopping experience mirrors how consumers have shopped for decades in brick-and-mortar stores. Our platform is designed to make it easy to navigate a vast selection of products when users do not have a specific item or brand in mind. On

Screening criteria for these products sold on Wish includes the top selling unbranded items (based on total sales) sold on Wish achieving an average rating of 4.0 out of 5.0 stars or higher from our users. eMarketer, Global eCommerce, 2020.

average, our users see over 500 distinct products across many different categories on a daily basis. Unlike other ecommerce platforms where consumers often visit with a predetermined purchase intent on specific items, our navigational and entertaining shopping experience gives us the ability to create purchase intent in our users across a diverse set of products and categories. Over 70% of the sales on our platform do not involve a search query and instead come from personalized browsing.

- **Personalized.** No two users' Wish interfaces and product feeds look the same. Utilizing big data technology, we enable customization on a massive scale. We deliver personalized and curated products to our users and help them discover desired products quickly. Our goal is to create an experience that is responsive to our users' preferences so that the products that they see are relevant to them. As a result, over 65% of our users click on a product detail page from the main feed. Every single one of our more than 100 million monthly active users receives a unique experience tailored to their individual user profiles, history and other contextual data in our data engine. Our focus on personalization has resulted in improving engagement and monetization.
- Entertaining. We have transformed the user experience to make shopping on Wish as engaging and entertaining as browsing social media. For example, the product assortment on the user's feed is a source of entertainment as this personalized feed allows the user to scroll through and explore a wide range of relevant products. In addition, our Blitz Buy feature engages users to spin a wheel in the app and creates a sense of exclusivity by generating a once-a-day sale on a limited number of extra discounted products. By incorporating interactive and social features, we increase the length and frequency of a user's sessions and drive increased engagement. Wish users spent on average 10 minutes per day on our platform in 2019.

#### **Value Proposition to Wish Merchants**

We built Wish to empower merchants around the world. Today, we serve more than 500,000 merchants and we partner with over 40,000 Wish Local stores across more than 40 countries. Some of our merchants sell exclusively online, others from their brick-and-mortar stores, and some sell both online and offline through their stores.

Accessible and Cost Efficient Ecommerce Platform. We give our merchants immediate access to our global base of over 100 million monthly active users and a comprehensive suite of indispensable services in a cost-efficient manner to help them run their businesses and grow sales. Most of our merchants sell unbranded goods, while approximately 80% of U.S retail is estimated to be branded.<sup>36</sup> Merchants selling unbranded goods have historically been underserved and lacked effective, direct access to a scaled global user base. In order to make our platform accessible to every merchant globally, we do not charge any monthly subscription fees, nor do we require deposits. Merchants pay us only when we have helped them generate a sale. Through our merchant platform, we seek to empower these highly capable merchants offering quality products at compelling values and unlock this supply of goods to consumers globally.

In addition, we give our merchants the following indispensable services:

#### **Demand Generation and Engagement**

• Global Reach for Online Merchants. Wish gives merchants immediate access to over 100 million monthly active users across more than 100 countries, with a significant user

eMarketer, US Private Label vs. Branded Retail Sales Share of CPG Products, by Channel, March 2019, March 2018, and August 2019.

footprint in the U.S. and Europe. We help our merchants reach these users in a highly targeted and cost-efficient manner. Our global reach is supported by our proprietary logistics platform which enables reliable and affordable delivery to users around the world.

- **Digital Presence for Brick-and-Mortar Stores.** Through Wish Local, we partner with a global network of local brick-and-mortar stores and provide them with access to our global online user base and help enable online discovery of their stores and products. Wish Local stores can sign up to our Sell on Wish feature to upload their in-store inventory on Wish for local pick-up or delivery. They can also serve as Wish Pickup locations for online Wish orders. These features help Wish Local stores drive additional foot traffic and increase sales.
- **Promotion.** Wish merchants can amplify their reach and sales by utilizing our native advertising tool, ProductBoost. Merchants can select which products they would like to advertise and create a campaign for them including setting a budget and timeline to display the ad. We utilize data science to optimize placement of the product on a user's feed, target users who have a higher propensity to engage with our platform and buy from our merchants, and maximize the merchant's return on ad spend. In addition, for products that receive exceptional buyer feedback, we offer our merchants the opportunity to earn Verified by Wish badges, which are shown to users and increase the visibility of the product on our platform.

**User-Generated Content Creation.** User-generated content, in particular, authentic, localized content, can meaningfully improve user engagement and increase purchases of products on our platform. Our value-conscious users rely on user-generated content such as reviews, ratings, photos, and videos in local languages, rather than brand recognition, when making purchase decisions. In the first seven months of 2020, our users contributed approximately 70 million product ratings and 60 million store ratings as well as approximately 9 million images and 1.5 million videos. Our user-generated content repository also contains guides and how-to's that can add instructional value to our product listings. This makes the content on our platform an important source of trust and quality for our largely unbranded product selection. Our data science prioritizes items with favorable reviews, higher ratings and shipping history, connecting buyers with high-quality merchants and enhancing both the user and merchant experience.

#### **Data Intelligence**

- Data Insight. We provide our merchants a comprehensive data set to run their businesses through the Wish Merchant Dashboard. This dashboard of key valuable metrics and analytics helps our merchants monitor and improve their performance in terms of total impressions, overall sales, product assortment, service quality, fulfillment, shipping needs, and refunds, among others. These data insights help our merchants drive more sales on Wish.
- Revenue Impact. Our proprietary, state-of-the-art data science capabilities are designed to display products to users who are most likely to buy them, in turn driving more revenue to merchants. We take into account a variety of factors, including our users' demographics and past purchase behaviors, to ensure that merchants reach the most relevant population of users.

#### Logistics

• Shipping Logistics. With ongoing changes to the global postal regulations and increase in cross-border sales volume, logistics has become paramount for small merchants to succeed in ecommerce. We facilitated the shipment of over 680 million items in 2019 to buyers across more than 100 countries. To support this immense volume, we have built and continue to invest in our robust global logistics platform. We have developed a number of logistics programs

including WishPost, Export Processing Center (EPC), A+, Fulfillment by Wish, and Fulfillment by Store. We have also established relationships with national postal networks and formed partnerships with third-party carriers to provide a set of reliable cross-border logistics solutions at competitive costs for our merchants. We believe that ensuring a consistent delivery experience for our users increases value to our merchants by boosting sales volumes and minimizing returns.

Given the UPU's recent postal rate reform and overall increasing logistics costs globally, particularly costs related to ChinaPost services, we have invested in new logistics offerings and related technology and data science to further diversify our logistics operations. The percentage of packages shipped through our proprietary logistics platform has grown substantially from 0% in 2016 to over 90% in 2020. Of this volume, we perform all logistic services on behalf of our merchants for approximately 50% of the packages. For the remaining 50%, merchants can choose the carrier and the desired service level based on what is available on our logistics platform. In July 2020, the U.S. was granted permission from the UPU members to set its own inbound postage rates, subject to certain caps, which led to higher postage rates for cross-border shipments. Despite this development, based on our diversified logistics offering, we were able to continue operations seamlessly.

• Wish Local Pick-Up. Through our Wish Local partnerships, we enable our online merchants to send their inventory to our over 40,000 partner pick-up locations in close proximity to users to allow for quicker, localized pickup. These Wish Local stores effectively give us a local warehousing and fulfillment footprint around the world without owning any real estate. Pick-up enables faster fulfillment as well as additional savings for our users. We believe that additional fulfillment options and savings to users increase conversion and sales to our merchants.

## **Business Operations**

- Optimization Tools, Services, and Education. We provide tools, services, and ongoing education to our merchant base to help them improve their business operations and drive greater success. At onboarding, we equip new merchants with resources and best practices to help them list their products, launch their stores, and upload, manage, and edit product listings. We also provide ongoing education and seminars for our merchants on broader topics relating to merchandising, logistics, and ecommerce, among others, to further promote merchant success.
- International Compliance. Wish assists merchants with local compliance requirements globally. For example, Wish has a tax engine that helps merchants comply with local tax requirements.
- Payment Processing. Wish optimizes global payment acceptance, installment payments and fraud prevention on behalf of our users and merchants by leveraging numerous third-party Payment Service Providers. We work with a broad network of regional and local payment providers and customize currency on our app to ensure optimal engagement locally.
- Early Payment Program. We provide an early payment program to our merchants where we accelerate payment to select qualified merchants by leveraging our data on merchants' sales activities on our platform, consumer reviews, shipping times, and inventory values, among others.
- **User Support.** Our largely outsourced global user support team offers localized support in over 20 languages, and handles user inquiries at all stages of the shopping experience. Hundreds of agents assist users with questions regarding delivery, item particulars, and payment and checkout. The team also focuses on ensuring quality interactions, supporting users through social media, training new agents, and processing vendor escalations. In 2018,

we launched the Wish Support Bot, an automated user support platform to supplement our team of agents, further increasing operational efficiency and quality of user support. From January to July 2020, our Wish Support Bot has been responsible for resolving over 65% of inbound support requests. As the user experience remains our primary focus, we will continue to expand and improve the user support platform.

• **Merchant Support.** We offer our merchants with 24x7 support so they can fully leverage our merchant platform. In many geographies, we have native language support and can support over 20 languages.

#### **Data Science**

Our technology and data science capabilities drive all aspects of our business. Our data advantage comes from our rich and growing data set of historical and recent user and merchant behaviors and transactions, and deep understanding of our more than 100 million monthly active users, over 500,000 merchants, and approximately 2 million items sold per day. All of this feeds into a proprietary data science algorithm that we continuously optimize for more intelligent insights and decision making.

Key examples of how we use data science to drive our business include:

- **User Acquisition:** We have extensive experience and have invested heavily in cost-effective, data-driven digital marketing and user acquisition strategies. We use a diverse range of social media channels and other online channels to market to potential Wish users. Additionally, our massive scale has resulted in significant growth in our organic reach which supplements our paid marketing efforts. We leverage the power of our proprietary data to make decisions on what to show to whom, when, and through which acquisition channel, with a focus on maximizing our return on marketing investment and conversion. We also design innovative marketing programs that help increase brand awareness by targeting people who have a higher propensity to engage with our platform and buy from our merchants.
- Lifetime Value Maximization: We are intently focused on maximizing lifetime value ("LTV") of our users and optimizing the initial user acquisition investment. Data science plays a critical role in achieving this goal as it informs key decisions we make on a continuous basis including an estimation of potential long-term value of an individual user, a group of users as well as specific countries. This determines how we allocate marketing investment across different users, marketing channels, and geographies, and drives our user acquisition and re-engagement strategies as well as longer-term investment priorities.
- Dynamic Pricing: We utilize our data to dynamically vary prices across products to optimize conversion as well as our margin. Given the wide array of unbranded items available on Wish, we can easily substitute products in our discovery-based shopping environment. This type of shopping environment is well suited to gauge user demand and conduct price discovery on a global level as well as on an individual user level. We take into account characteristics of the product and the user to estimate price sensitivity and vary pricing to achieve a margin target at the user and basket level, as opposed to on individual items. Our users typically buy multiple items in a single transaction, which allows us to use low cost items to maximize conversion while still optimizing basket-level margin. Price variability and constant changes on our platform also create a sense of urgency and add to the gamification element
- **User Personalization:** We leverage our data science and technology to place the right product in front of the right user at the right time and at the right price. Our proprietary algorithms utilize a rich and growing data set of historical and recent user behaviors that includes browsing data, past transactions, reviews, and preferences noted on Wish, to display

only the most relevant and personalized content. This data-driven approach enables efficient and enjoyable navigation and discovery on a mobile screen, creates purchase intent across a diverse set of products, and increases conversion to sales. For every 100 user sessions on our platform, an average of approximately 40 items get added to cart.

- **User-Generated Content:** Our platform offers mostly unbranded goods today. Our value-conscious users rely on user-generated content such as reviews, ratings, photos, and videos, rather than brand recognition, when making purchase decisions. This makes the user-generated content on our platform an important source of trust and quality for our largely unbranded product selection. Our data science prioritizes items with favorable reviews, higher ratings and shipping history, connecting buyers with high-quality merchants and enhancing both the user and merchant experience.
- Merchant Insights: Our platform includes a merchant dashboard with built-in analytics to help merchants sell more products and track their performance. Our data capabilities help merchants better understand user behavior and preferences, enabling them to operate more intelligently and efficiently. We give our merchants access to their products' sales, impressions, refund ratios, ratings, and reviews, as well as insight into their products' shipping performance including time to door. We regularly rank merchants based on their performance related to user experience and policy compliance. Merchants with high rankings receive greater product impressions, eligibility for faster payment, as well as discounted access to our native advertising tool ProductBoost.
- **ProductBoost:** ProductBoost is our native advertising tool for merchants, which helps them promote their products. Approximately 28% of our merchants used ProductBoost in 2020 to date. Merchants can select which products they would like to advertise and create a campaign for them including setting a budget and timeline to display the ad. Merchants receive various key metrics that track the performance of their Boosted products, including return on ad spend, number of impressions, and number of conversions. We utilize data science to optimize placement of the product on a user's feed, target users who have a higher propensity to engage with our platform and buy from our merchants, and maximize the merchant's return on ad spend.
- Logistics: We leverage data science to power our logistics platform and continuously improve our logistics offering to our merchants. We combine this global network with data science to improve transparency and logistics operational efficiency for our merchants, while also aiming at reducing shipping time and improving delivery reliability for our buyers.
- **User and Merchant Support:** We use data to understand what a user or merchant is likely to need help with in order to improve the quality of support and maximize cost efficiency of providing such support. Utilizing our vast data set drives efficiency for us by allowing us to track various success metrics including spend per ticket, resolution rates, and user and merchant satisfaction scores.

#### **Our Network Effects**

Our platform benefits from powerful network effects, fueled by our proprietary data and technology. We collect, analyze, and utilize data across our over 100 million monthly active users, over 500,000 merchants, and approximately 2 million items sold per day to improve the shopping experience for users and the selling experience for merchants. Our proprietary algorithms analyze a rich and growing data set of transactions and historical behaviors of both users and merchants to drive continuous optimization on the platform and inform key business decisions on a daily basis. Our data science enables personalization at the individual user level at a massive scale and drives significant advantages across all aspects of our business operations, including user acquisition, user experience, pricing strategies, user-generated content, merchant insights, and user and merchant support.

As Wish grows, we accumulate more data across user and merchant activities, we strengthen our data advantage, and we create an even better experience for everyone on our platform, which in turn attracts more users and merchants. As more users come to Wish, driven by the affordable value proposition and differentiated shopping experience, we drive more sales to our merchants. Adding more users also reinforces our user-generated feedback loop of ratings, reviews, photos, and videos, which drives greater user engagement. As more merchants succeed on Wish, more merchants join the platform and grow their businesses with Wish, broadening our product selection which in turn improves the user experience. This flywheel effect has driven tremendous value to both users and merchants and made Wish one of the largest ecommerce marketplaces in the world.

## **Our Growth Strategy**

We leverage our data and unique insights on merchants and users to extend our platform outside of our core business and drive additional growth opportunities. Our data science capabilities are a unique advantage and core to the operations of our business.

#### **Grow Our Base of Users**

- Continue to Acquire New Users. We are focused on growing our user base around the world. We currently serve users in over 100 countries. We estimate that there are over 1 billion households with income of less than \$75,000 around the world, excluding China and India. We intend to grow our user base by cost effectively reaching new users through a combination of data-driven and targeted advertising campaigns, word of mouth and other marketing efforts, as well as increasing our brand awareness.
- **Drive User Conversion.** We have over 13 million monthly active buyers<sup>37</sup> and over 100 million monthly active users on the platform. We will continue to drive greater user engagement and convert more active users on our platform to become active buyers by utilizing our data science and introducing more interactive and entertaining features.
- Drive Profitable Lifetime Value from Existing Users. We will continue to improve the engagement and monetization of our users on our platform to maximize their lifetime value. We plan to achieve this goal using a number of strategies. We will use our data science to drive personalization of our platform, so that we can continue to offer a differentiated mobile shopping experience and continue to drive higher user engagement. We will also continue to offer deeply discounted promotions, including LQD, and an overall gamified experience that further incentivizes users to make a purchase. In addition, we will promote robust, user-generated content which will continue to serve as a source of trust and quality for our largely unbranded product selection. We will also seek to continuously improve our ease of use by investing in our user support and logistics platform to enable faster deliveries and localization of our platform, ensuring optimal engagement by our global user base.
- Expand Geographically. We will continue to expand our global footprint and enter new geographies and acquire new users in those markets. In particular, countries in Africa, Latin America, and Eastern Europe are key growth markets with large and growing populations as well as a significant portion of the population with lower average household incomes. Within these markets, the proliferation of consumer mobile use and improvements in internet infrastructure will help us establish and grow our business. In addition, our investments in our own logistics network will help us achieve better breadth and density of coverage in our newer regions and countries, enabling a better user experience and localized, high-quality delivery and customer service.

Based on available internal data from January 2020 to June 2020.

#### **Grow Our Base of Merchants and Offerings**

- Diversify Our Merchant Base and Expand Product Categories. As we grow and diversify our merchant base, we partner with other merchant platforms, such as PayPal, ShipStation, and PlentyMarkets, to acquire additional merchants outside of China and expand our geographic reach and diversity across the merchant base. In 2019, four out of the top 10 selling merchants on our platform were located in the United States selling refurbished electronics, beauty products, and hobby items, which illustrates the ongoing diversification of our merchant base and product categories. We will continue to expand product selection to provide more diversified products at competitive price points to our users.
- Broaden Merchant Services. We intend to add additional services to help our merchants grow their businesses and sell more on Wish. These services include educational content and resources as well as tools enabling merchants to promote their products in a variety of ways both on and off the Wish platform, efficiently manage working capital, and fulfill and ship their orders. For example, we have recently expanded our native advertising tool ProductBoost by launching two products MaxBoost, which enables additional product impressions outside of Wish including on social media platforms, and IntenseBoost, which provides increased product impressions in a shorter period of time.
- Expand Logistics Platform. We plan to continue to expand our logistics platform and optimize our proprietary logistics programs. Through strategic partnerships with selected regional postal networks and commercial logistics partners, we aim to become an integrated part of the cross-border logistics value chain with our own logistics services. We intend to leverage our own dedicated channels that we developed to provide faster and more reliable delivery to our buyers. Given the recent changes to the Universal Postal Union Treaty and overall increasing logistics costs globally, particularly costs related to ChinaPost services, we have invested in new logistics offerings and related data science to further diversify our logistics operations. The percentage of packages shipped through our proprietary logistics platform has grown substantially from 0% in 2016 to over 90% in 2020. Of this volume, we perform all logistic services on behalf of our merchants for approximately 50% of the packages. For the remaining 50%, merchants can choose the carrier and the desired service level based on what is available on our logistics platform. By investing in logistics, we will also help our merchants reach a larger and more global user base and increase demand for their products. We believe that ensuring a consistent delivery experience for our users increases value to our merchants by boosting sales volumes and minimizing returns. We are also exploring the opportunity to open up our logistics programs to merchants outside of Wish.
- Grow Our Wish Local Offering. We will continue to expand our Wish Local program to drive both online and offline commerce. Through the Sell on Wish feature, Wish Local enables local brick-and-mortar stores to digitize their storefronts by uploading their in-store inventory on our platform and selling to our global user base. These stores in turn give us access to local warehousing and fulfillment footprint at more than 40,000 stores around the world by serving as Wish Pickup locations for online Wish orders. Over time, we will expand our offerings to Wish Local stores and grow our footprint by leveraging our digital marketing expertise as well as partnerships with small business platforms to identify and onboard new Wish Local stores.

## Continue to Innovate and Extend Our Platform

• Monetize Brick-and-Mortar Stores. We are still in the early stages of building out and monetizing our Wish Local offering. We do not currently charge our Wish Local stores for any incremental traffic or revenue that we help bring to their store. As we continue to grow our Wish Local program, we plan to explore different ways in which we can further enhance our value proposition and monetize our offering, whether in the form of additional traffic or sales to the store.

- Add New Product Categories. We plan to continue to diversify product categories and service offerings on Wish. We believe a more comprehensive catalog on our platform will drive superior user engagement. First, as our Wish Local footprint continues to grow, we see an opportunity to expand CPG offerings sold through our Wish Local stores. Many of our Wish Local stores carry CPG products which they can list on our platform through Sell on Wish feature, and the location of Wish Local stores within the community makes this option attractive for CPG products. Second, we aim to expand our platform to include in-person services offered through our Wish Local stores, such as salon, repair, tailor, and event planning. We believe expansion into in-person services will drive increased foot traffic, recurring visits, and increased sales to our merchants. Lastly, we are also expanding our offerings to include affordable brands and off-price branded inventory given the significant consumer demand. Products such as branded refurbished electronics, closeout branded apparel, and off-price branded beauty are some of the most popular categories on our platform. We believe that a larger selection of affordable, branded items will increase Wish's appeal to both new and existing users.
- Expand to New Advertising Partners: Historically, the majority of the advertising on our platform has comprised of our merchants advertising their products. We see a significant opportunity to connect other types of advertisers to our over 100 million monthly active users around the world as we believe our user base presents a diverse and global audience for advertisers across many end markets beyond ecommerce, including financial services, subscription services, and travel. We help these advertisers target individual users using our data science and can support advertisements that show up digitally within a user's scrolling feed, natively among user-generated content as well as physical advertisements that can be inserted into our shipments. Leveraging our scale and data science, we plan to continue to expand our advertising partners.
- **Grow First-Party Sales.** We leverage the data across our platform to identify key selling trends and consumer preferences to source branded and unbranded inventory directly from manufacturers and sell on our platform. For these direct sales, we act as the principal and owner of the inventory, and for some, we develop and sell under our own private brand. While this revenue is a modest amount today, we will continue to experiment with and optimize first-party sales including our private label strategy.
- Open Our Commerce Platform to Additional Businesses. Over time, we plan to open the Wish commerce platform and capabilities to additional businesses. We believe we have been able to attract some of the best engineering talent and have developed best-in-class capabilities across data science, digital performance marketing, logistics, and others functions that support the Wish platform. These capabilities address some of the most enduring challenges that many businesses both within ecommerce and beyond face today. We see a significant opportunity in providing access to these capabilities to additional businesses not currently on the Wish platform. For example, we plan to offer access to our logistics offering to any merchant who engages in ecommerce and wants a reliable, affordable, and global shipping solution. Similarly, we plan to offer our digital performance marketing solutions to any advertiser, regardless of their vertical or end market.

# **Our Platform Offerings and Features**

We have developed a powerful two-sided marketplace that drives significant value to both users and merchants. By connecting our users directly to the merchants who directly source their products, we can offer a vast selection of high-quality items at competitive prices.

#### For Our Users

We designed the Wish platform to be affordable and entertaining by providing relevant products that are determined through data science and personalization. Over 65% of our users click on a product detail page from the main feed, and over 70% of the sales on our platform do not involve a search query and instead come from personalized browsing. Our mobile app and website make it easy for our users to browse and discover new products they want to purchase.

We are focused on enhancing user experience and driving conversion through the use of data. The user experience begins on our main feed, where users scroll through, and the products displayed are determined by our growing repository of historical and recent user behaviors. The relevancy of products displayed is gauged by the number of scrolls and new page downloads by a particular user. Similarly, the appearance of specific app features are tailored to individual users and guided by their prior engagement with and response to such features. The app features that we tailor to individual users include feed banners, feed collection tiles, ordering of top level categories, highlighting of features such as Blitz Buy and Referral Rewards, pricing, notifications outside of the app, and payment options.

We also customize currency and language of our app by region to ensure optimal engagement locally. We enable features on our online platform to be viewed in 40 different languages, which improves engagement of our large, diverse user base.

In order to optimize user conversion, we track user product journey–starting with the time spent on the main feed, followed by the number of detailed product page views, the addition of product to cart, and ultimately check-out. We continue to improve the user experience on the Wish app to drive continued improvements in user conversion.

The following products help us to provide a seamless experience for our users across our platform.

## Affordability and Accessibility

**Pricing.** The merchants on our platform offer primarily unbranded products that can be discounted in excess of 85% as compared to branded alternatives across a number of categories, such as wireless ear buds and air fryers. For example, based on our internal research, the average price of the top 20 wireless ear buds sold on Wish is approximately \$20, compared to the online retail average price of a select set of recognized branded alternatives of approximately \$165, and the average price of the top 20 air fryers sold on Wish is approximately \$85, compared to approximately \$173 on average for a select set of recognized branded alternatives. Additionally, competition within our merchant base to generate sales keeps price points for items low. We also leverage data science to help our users find attractive value, and we ingest signals and data from user engagement with specific products and advertisements on our platform in order to dynamically determine the most optimal price for the individual user.

**Pay Later.** Through our Pay Later program eligible users can defer payment of portions of the purchase price for up to 30 days. Pay Later does not include any fees or interest and makes many products more affordable for our users.

**Wish Cash.** Users can also build their Wish Cash balances by getting rewards through the application in various ways including referring friends to make a purchase on Wish. Users can in turn use their Wish Cash balance to make purchases, as well as get refunds faster than waiting for

<sup>38</sup> Screening criteria for these products sold on Wish includes the top selling unbranded items (based on total sales) sold on Wish achieving an average rating of 4.0 out of 5.0 stars or higher from our users.

traditional banking delays. In some countries such as the United States, users are also able to use cash deposits at certain Wish Local partner stores to pay for items on Wish without needing a credit card or traditional online form of payment.

Add to Cart Offer. Add to Cart Offer unlocks discounts as users browse and put items into their carts, which gamifies the shopping experience with surprise offers.

**Promotions**. Through weekly promotional events and promo codes, users can receive significant savings off of our already value-driven pricing.

**Wish Local Product Pick-Up.** Our over 40,000 Wish Local partners around the world serve as Wish Pickup locations for online Wish orders, providing an additional mode of fulfillment and enabling additional savings for our users. Users can select to have products they purchase shipped to one of our Wish Local stores for a pick-up and save on average of over 25%. For items that are already available in our Wish Local store and do not require shipping, users can save even more, on average of over 50%, for selecting pick-up.

**Rewards.** Users can not only earn reward points on Wish by making purchases but also by taking various actions throughout the application. Submitting a review or uploading a photo or video can add to a user's reward points balance. These reward points can be redeemed for discounts on future orders. Periodically, we may include other offers in the reward redemption such as items that are purchasable directly with reward points.

**In-App User Support.** Our largely outsourced global user support team offers localized support in over 20 languages, and handles user inquiries at all stages of the shopping experience. Hundreds of agents assist users with questions regarding delivery, item particulars, and payment and checkout. Users can exchange messages 24x7 with Wish's support assistant to immediately solve item and delivery issues. We also provide support to users to help identify product listings that infringe on intellectual property and take prompt action against accounts known to sell counterfeit listings.

Feedback and User-Generated Content. With millions of product ratings, store ratings, images, and videos uploaded monthly, users are able to make informed purchase decisions. To encourage users to continue to leave feedback, Wish prompts them to review their purchases immediately upon receipt, such that easy information gets submitted first and any incremental images or videos can be contributed later. As additional incentive, users receive reward points for contributing images and videos. In the first seven months of 2020, our users contributed approximately 70 million product ratings and 60 million store ratings as well as approximately 9 million images and 1.5 million videos.

#### **Entertainment**

In addition to making Wish affordable and accessible to all, we have built in features that make our platform entertaining to users. Our testing of gamified features against passive discount codes has demonstrated higher engagement and greater conversion, and we continue to enhance and refine these features in order to keep users engaged. The following features gamify our platform while further incentivizing users to transact.

**Blitz Buy.** Our Blitz Buy feature offers once a day sales on a selection of extra discounted products. Users can spin a wheel to find out the number of products that will be in their daily Blitz Buy selection.

**Daily Login Bonus.** With our daily bonus feature, users receive a stamp for every day they visit the Wish app. If users acquire seven stamps within a one-month period, they receive an offer for up to

50% off a future order. Since January 2020, approximately 21% of our active users have received an offer, and 11% have applied that discount in a subsequent purchase.

**Limited Quantity Deals.** Our LQD feature offers users deals throughout the day pricing higher cost items at under a dollar for a limited time only. The first user who successfully pays for and completes the transaction gets the item. This risk-free gamification mechanism drives engagement on the platform with approximately 30% increase in unique users initiating a transaction in any given day. The feature also drives higher user engagement and approximately 5% increase in daily spend on non-LQD items.

Add to Cart Offer. Add to Cart Offer unlocks discounts as users browse and put items into their carts, which gamifies the shopping experience with surprise offers.

**Periodic Sales.** Users are offered personalized sale events on a periodic basis. The available offer and the duration of the offer have elements of gamification added to them by requiring the user to claim a code or take an action before revealing their final discount amount. Users are made aware of the sale event when they open the app, creating a sense of excitement that they stumbled upon an opportunity to save even more.

**Community TV.** To help our users get a more entertaining and dynamic experience with our platform, we built the Community TV section to highlight the vast amount of video content created by our community. This section offers an alternate way to browse highly relevant and entertaining content.

**Referral Program.** We built our referral program to help our users get rewarded for their efforts to grow our user base. Users have the ability to earn up to \$100 of Wish Cash by inviting friends to the platform who ultimately make a purchase. Periodic increases to this amount may appear, incentivizing users to continue to refer friends.

## For Our Merchants

**Our Platform.** Our online marketplace provides merchants immediate and direct access to users all around the world serving as a highly effective demand generation engine. Merchants have the unique ability to bypass retailers or other intermediaries and sell directly to a global consumer base. In addition, our platform seamlessly integrates with third-party partners (such as ERP and channel systems) to ensure efficient execution and enable a unified view of the merchant's business. We also manage all sales processing and support logistics so that our merchants can focus on their core business and selling rather than operational details.

**Data Analytics.** We provide metrics and analytics to help our merchants understand store performance, shipping and refund metrics. Our dashboard displays data including sales, impressions, refund ratios, ratings, and reviews, as well as insight into their product's shipping performance including time to door. Visual trends of various metrics help assess historical performance.

**Promotions**. We have designed innovative programs such as Wish Express, ProductBoost, Trusted Store and Verified by Wish to help merchants promote their products within our platform and increase sales.

• Wish Express. This program allows merchants who can deliver the product quickly to designate that product as "Wish Express." Such products will be noted with the Wish Express logo of an orange truck on the product listing page and will also be listed in Wish Express dedicated sections in the app.

- ProductBoost. This is Wish's native advertising tool that allows merchants to pay a fee to move their products higher up within the product rankings if they are relevant to Wish users and increase product impressions and sales. As part of our ProductBoost tool, we have rolled out MaxBoost, which enables additional product impressions outside of Wish including on social media platforms, and IntenseBoost, which provides increased product impressions in a shorter period of time at a higher spend.
- Merchant Standing. This program allows merchants with good delivery performance and high product quality to access additional tools and benefits. Our merchants are ranked and assigned a category platinum, gold or silver based on user experience and policy compliance. The highest-ranked merchants are eligible for incremental impressions on new products.
- Verified by Wish. For products that receive exceptional buyer feedback, we offer our merchants the opportunity to earn Verified by Wish badges, which are shown to users and increase the visibility of the product on our platform.

**Payments.** Wish supports its merchants with a variety of local payment providers. We have also designed an incentive program, where for orders that meet qualifications related to delivery compliance, merchants are provided early payment. Our payment processing optimization and financial services provide our merchants with access to more value-conscious consumers who often use credit for their purchases.

**Logistics**. We facilitated shipment of over 680 million items globally in 2019 via a robust global logistics platform developed internally for our Wish merchants as well as our third-party logistics partners. We have built a number of proprietary logistics service programs aimed at reducing cost and operational friction associated with cross-border fulfillment for our merchant base. Our logistics platform and programs comprise the following:

- WishPost. This is our proprietary logistics engine designed for direct end-to-end single order dropship from a merchant's warehouse to an end-user. It aggregates over 20 China-based carriers and hundreds of logistics service levels from which our merchants can choose. Our merchants often connect their own ERP systems with WishPost and select corresponding service levels for their cross-border fulfillment needs.
- Export Processing Center ("EPC"). This is a program created based on our own proprietary consolidation algorithm that we use to combine different orders from different or the same merchants to send in one single parcel to the user for her multiple orders. This program allows Wish to take advantage of a higher combined starting weight for better cost and performance efficiency with improved user experience, which in turn benefits our merchants with lower refund rates and better logistics costing structure.
- A+. As we began to develop our own expertise in cross-border logistics, we designed an A+ program that we implement in chosen destination countries. Through close partnerships with logistics and warehousing operations vendors around the world, we are able to assume the end-to-end logistics responsibilities ranging from first mile collection from merchants, to warehousing operations where EPC or WishPost-like activities take place, and ultimately to last mile delivery. With our volume advantage, we negotiate competitive cross-border, last mile pricings from participating partners to complete delivery. Through our A+ program, we are able to ensure improved delivery performance for our end users in A+ designated countries. Our merchants are also able to take advantage of a lowered logistics pricing at the improved service levels.
- Fulfillment by Wish ("FBW"). This is a forward deployment and fulfillment program that our merchants can opt into, in which they can ship their inventory based on our recommendation engine selection to our warehouse partners in advance. Our warehouse partners will take care

of order fulfillment including pick, pack, and ship. When merchants opt into the FBW program, their FBW-eligible listings by default earn the Wish Express badge. Wish Express items often enjoy higher prices, better conversion and faster payment from Wish.

• Fulfillment by Store ("FBS"). This program enables our online merchants to fulfill their products in local offline stores, increasing both the online and offline exposure and sales of their products. Based on forecasting data and insights, Wish strategically recommends certain high-potential products for merchants to select from for FBS. After merchants ship selected products to FBW warehouses, some or all of them will be available for sale in stores in high demand areas across North America and Europe. This means that the inventory is not only stored in our FBW warehouses, but also in our Wish Local partner stores around North America and Europe in close proximity to users. Merchants simply ship their selected inventory to the chosen FBW warehouses, and the warehouses will take care of order fulfillment (pick, pack, and ship) including selecting and sending inventory to offline stores across the globe.

Wish Local Offering. Through Wish Local, we partner with a global network of local brick-and-mortar stores and provide them with access to our global online users and help enable online discovery of their stores and products. We identify zip and postal codes of interest based on where our uses are clustered and acquire merchants predominantly through Facebook ads. Wish Local partners can sign up to utilize the Sell on Wish feature, which allows them to upload their in-store inventory on our platform for immediate access to our global base of over 100 million monthly active users. Wish Local stores can also serve as Wish Pickup locations for online Wish orders, which drive additional foot traffic and increase sales.

**Intellectual Property Guidelines**. Wish has a zero-tolerance policy against intellectual property infringement and counterfeits. To help support merchants aligned with this policy, Wish offers a brand partner program that allows brand owners to accelerate the takedown-request process, view and report infringing items, review historical takedown requests, and provide them with confidence that their takedown requests are reviewed in a timely fashion. Wish also requires each merchant that wants to have its items listed in the "Brands" section of the Wish platform to declare that it is able to list and sell its branded products without infringing on one's intellectual property rights.

Tax Compliance. Wish has a tax engine that assists merchants with local tax requirements compliance.

### Competition

We compete for both users and merchants. For consumers, we compete on the basis of affordability and user experience. For merchants, we compete on the basis of providing profitable distribution, an end to end platform and global reach.

Our online competitors include large, global ecommerce platforms such as Amazon, Alibaba, and Shopify as well as more traditional discount retailers such as Walmart and Target. Our offline competitors also include scaled discount retailers that offer heavily discounted and off-season products, such as Dollar General and TJ Maxx. We are able to compete for Wish users based on our massive product selection, low prices and daily discounts, deeply-personalized and differentiated shopping experience powered by our data science and optimized for the mobile device, and entertainment derived from various engaging and interactive features of our platform.

#### **Research and Development**

We have a technology and data-driven research and development culture that allows us to deliver a high-quality experience for our users and merchants. Our research and development talent is

responsible for the design, development, testing, and delivery of our platform and user experience. The vast majority of our research and development talent is located in San Francisco and Toronto. We strive to create an environment that utilizes our employees' talents and satisfies their intellectual curiosities while promoting the development of impactful and transformative technologies.

As a company, we invest substantial resources in research and development to drive core technology innovation and bring new products to market. As of December 31, 2019, 45% of our headcount was involved in research and development and related activities.

# **Sales and Marketing**

Our sales and marketing capabilities represent a core competency that is essential to the success of the Wish platform. We are focused on continuing to acquire new users efficiently, and building brand awareness and a demand generation engine. We have extensive experience in cost-effective, data-driven digital marketing and user acquisition. We also design innovative marketing programs that help increase brand awareness by targeting people who we believe have a higher propensity to engage with our platform and buy from our merchants. We continue to partner with various social-media platforms to ensure we gain exposure with broader audiences.

We currently acquire new users through a variety of marketing channels including social media, search engine optimization and brandoriented marketing campaigns. We rely on our data to understand consumer behavior and long term value of the consumer which guide our acquisition strategy. We also utilize data in determining how best to engage our users and seek to optimize the mode, timing and frequency of interactions across our mobile app, SMS text, mobile notifications, and emails.

Over the last four years, we have invested heavily in building a talented in-house marketing team, while also developing proprietary technologies that enable us to build data-driven and highly-personalized campaigns that can scale globally on digital platforms including Facebook, Instagram, and various Google properties. Our marketing efforts also focus on re-targeting of existing users, building our brand, generating awareness, and cultivating the Wish community. In 2017, we announced a partnership with the Los Angeles Lakers, which involved placing the Wish logo on the jersey of Lakers players.

We also utilize our Wish Stars community, a loyalty program of approximately 10,000 Wish power users and influencers across over 25 countries, as an additional marketing tool as well as to gather feedback and continue improving our platform. Targeting for this loyalty program is based on users' social following and their purchase activities on Wish.

Additionally, our massive scale has resulted in significant growth in our organic reach which supplements our paid marketing efforts.

Finally, we also invest in efficient sales and marketing activities to identify and onboard high potential merchants around the world both for our online marketplace as well as Wish Local. Similar to our user acquisition strategies, we deploy digital performance marketing campaigns on social media and other channels to acquire new merchants. We supplement this effort with offline marketing activities that includes attending trade shows and conducting seminars. We also leverage partnerships with third-party platforms for merchant referrals.

#### **Data Security and Protection**

We have an information security team whose responsibilities include securing the data provided through our platform by our users and merchants. The team employs various controls to meet this goal, from technical measures that run on the networks and systems we maintain, to processes and procedures that define how data should be classified and handled.

The team also looks to identify and mitigate security risks across our platform, whether they involve the data we handle or the services we use. Some measures we take here include running regular internal anomaly scans, performing security assessments of any third-party vendors that we may wish to engage, and limiting employee access to data based on business need. We also build mechanisms that identify and can mitigate automated attacks against our site. To try and verify the effectiveness of our controls, we have external security firms perform penetration tests against our platform, as well as run bug bounty programs that pay hackers to find weaknesses in our systems.

Another facet of information security involves responding to reports (from external parties or through internal sources) of any data incidents or potential breaches, with the team driving the process from initial validation, through investigation of the root cause, and finally to mitigation and resolution. To minimize the likelihood of these issues occurring in the first place, the team also creates training material to educate staff on various security best practices.

#### **Intellectual Property**

We rely on federal, state, common law, and international rights, as well as contractual restrictions, to protect our intellectual property. We control access to our proprietary technology through a combination of trademarks, domain names, copyrights, trade secrets, patents, and confidentiality agreements with employees and third parties. We pursue the registration of our copyrights, trademarks, service marks, and domain names in the United States and in certain locations outside the United States.

Despite our efforts to protect our proprietary technology and our intellectual property rights, unauthorized parties may attempt to copy or obtain and use our technology to develop applications with the same functionality as our applications.

## **Government Regulations**

As with any company operating on the Internet, we grapple with a growing number of local, national and international laws and regulations. These laws are often complex, sometimes contradict other laws, and are frequently still evolving. Laws may be interpreted and enforced in different ways in various locations around the world, posing a significant challenge to our global business. For example, U.S. federal and state laws, EU directives, and other national laws govern the processing of payments, consumer protection and the privacy of consumer information; other laws define and regulate unfair and deceptive trade practices. Still other laws dictate when and how sales or other taxes must be collected. The growing regulation of ecommerce worldwide could impose additional compliance burdens and costs on us or on Wish merchants, and could subject us to significant liability for any failure to comply. Additionally, because we operate internationally, we need to comply with various laws associated with doing business outside of the United States, including anti-money laundering, anti-corruption and export control laws. Recent trends globally toward increased protectionism and trade barriers can result in actions by governments around the world that may be disruptive to our businesses. See the sections titled "Risk Factors—We may be adversely affected by the complexity,

uncertainties and changes in PRC regulation of internet-related businesses and companies, and any lack of requisite approvals, licenses or permits applicable to our business may have a material adverse effect on our business and results of operations," "—We are subject to governmental regulation and other legal obligations related to privacy, data protection, information security, and consumer protection. If we are unable to comply with these, we may be subject to governmental enforcement actions, litigation, fines and penalties or adverse publicity," and "—A failure to comply with current laws, rules and regulations or changes to such laws, rules and regulations and other legal uncertainties may adversely affect our business, financial performance, results of operations or business growth" herein for additional information.

## **Employees**

As of June 30, 2020, we had 752 full-time employees. We also engage temporary employees and consultants as needed to support our operations. None of our employees in the United States are represented by a labor union or subject to a collective bargaining agreement. In certain countries in which we operate, we are subject to, and comply with, local labor law requirements which may automatically make our employees subject to industry-wide collective bargaining agreements. We may be required to comply with the terms of these collective bargaining agreements. We have not experienced any work stoppages, and we consider our relations with our employees to be good.

### **Facilities**

Our corporate headquarters occupies approximately 106,000 square feet in San Francisco, California under leases that expire between 2022 and 2025. We also lease offices in California and Washington, as well as locations internationally, including in China and the Netherlands.

Currently, our employees are working remotely in accordance with stay-at-home mandates enacted due to COVID-19. We believe that when our employees return to the office, our existing facilities will be sufficient for our current needs. In the future, we may need to add new facilities and expand our existing facilities as we add employees, grow our infrastructure and evolve our business, and we believe that suitable additional or substitute space will be available on commercially reasonable terms to meet our future needs.

## **Legal Proceedings**

We are currently involved in, and may in the future be involved in, actual and threatened legal proceedings, claims, investigations and government inquiries arising in the ordinary course of our business, including legal proceedings, claims, investigations and government inquiries involving intellectual property, data privacy and data protection, torts, consumer protection, securities, employment, contractual rights or false or misleading advertising. We are also regularly subject to proceedings, claims, investigations and government inquiries seeking to hold us liable for the actions of merchants on our platform.

Although the results of the actual and threatened legal proceedings, claims, investigations and government inquiries in which we currently are involved cannot be predicted with certainty, we do not believe that there is a reasonable possibility that the final outcome of these matters will have a material adverse effect on our business or financial results. Regardless of the final outcome, however, litigation can have an adverse impact on us because of defense and settlement costs, diversion of management resources, harm to our reputation and brand, and other factors.

For additional information on risks relating to litigation, see the sections titled "Risk Factors—Risks Related to Our Business and Industry—We may be involved in litigation matters or other legal proceedings that are expensive and time consuming" and "Risk Factors—Risks Related to this Offering and Our Class A Common Stock—We may be subject to securities litigation, which is expensive and could divert management attention."

#### **MANAGEMENT**

#### **Executive Officers and Directors**

Our executive officers and directors and their positions, are listed below, as well as their ages as of July 31, 2020:

Name	Age	Position(s)
Executive Officers:		
Peter Szulczewski	38	Founder, Chief Executive Officer, and Chairperson
Rajat Bahri	56	Chief Financial Officer
Devang Shah	48	General Counsel
Thomas Chuang	43	Vice President of Operations
Pai Liu	35	Vice President of Data Science
Non-Employee Directors:		
Ari Emanuel	59	Director
Joe Lonsdale	37	Director
Tanzeen Syed	38	Director
Stephanie Tilenius	53	Director
Hans Tung	49	Director

The following is a brief biography of each of our executive officers and directors:

#### **Executive Officers**

Peter Szulczewski is our founder and has served as our Chief Executive Officer since July 2010 and as our Chairperson of the Board since August 2020. Prior to founding the Company, Mr. Szulczewski served in various positions at Google, Inc., a technology company, including as a technical lead and senior software engineer, from April 2005 until January 2009. Mr. Szulczewski holds a B.S. in mathematics and computer science from University of Waterloo. We believe that Mr. Szulczewski is qualified to serve as a member of our board based on the perspective and experience he brings as our founder and Chief Executive Officer.

Rajat Bahri has served as our Chief Financial Officer since December 2016. Prior to joining the Company, Mr. Bahri served as Chief Financial Officer at Jasper Technologies, Inc., a technology company, from June 2013 until October 2016. Prior to Jasper Technologies, Mr. Bahri spent over eight years as Chief Financial Officer of Trimble Navigation, a publicly traded company. Mr. Bahri began his career at Kraft Foods Inc., where he spent 18 years in various positions including Chief Financial Officer of Kraft Canada, Inc. and Kraft Pizza Company. Since July 2020, Mr. Bahri has served on the board of directors of Pacific Gas and Electric Company. Mr. Bahri holds a Bachelor in Commerce from the University of Delhi and an M.B.A. from the Fugua School of Business at Duke University.

Devang Shah has served as our General Counsel since February 2018. From August 2010 until April 2017, Mr. Shah served in various positions at Zynga, Inc., a game development company, most recently as General Counsel, Secretary and Senior Vice President. Prior to Zynga, Mr. Shah served as General Counsel and Associate General Counsel at UTStarcom, a telecommunications company, Senior Corporate Counsel at Longs Drugs Inc., and Corporate Counsel at Chiron Corporation, a biotechnology company. Mr. Shah began his career as an attorney at Skadden Arps Slate Meagher & Flom, LLP. Mr. Shah holds a B.S. in engineering from Cornell University, an M.S. in engineering from Stanford University and a J.D. from the University of California, Berkeley School of Law.

Thomas Chuang has served as our Vice President of Operations since December 2016 and prior to that served as our Head of Finance beginning in June 2014. Prior to joining the Company, Mr. Chuang was a senior analyst at GGV Capital, a venture capital firm and began his career at PricewaterhouseCoopers. Mr. Chuang holds a B.S. in Business Administration from the University of California, Santa Cruz and an M.B.A. from the Walter A. Haas School of Business at the University of California, Berkeley.

Pai Liu has served as our Vice President of Data Science and Engineering since August 2020. He previously served as Director of Data Science September 2019. Prior to joining the Company, Mr. Liu served as Data Science Manager at Airbnb, an online rental company, from August 2015 until August 2017 and Data Science Lead until September 2019. Earlier in his career, Mr. Liu was a research scientist at PARC, a Xerox company. Mr. Liu holds a Bachelor's degree from Tsinghua University, a Masters in Computer Science from the University of Southern California, and a Ph.D. in Philosophy – Industrial and System Engineering from the University of Southern California.

## Non-Employee Directors

Ari Emanuel. Mr. Emanuel has served on of our board of directors since November 2019. Mr. Emanuel has served as the CEO of Endeavor Group Holdings, a talent agency, since 2009 following the merger of the William Morris Agency and Endeavor Talent Agency, which he founded in 1995. Prior to founding Endeavor, Mr. Emanuel was a Partner at InterTalent, a talent agency, and a senior agent at ICM Partners, a talent agency. He began his entertainment industry career at Creative Artists Agency (CAA), a talent agency. He has served as a member of the board of directors of Live Nation Entertainment, Inc. since 2007. Mr. Emanuel holds a bachelor's degree from Macalester College. We believe that Mr. Emanuel is qualified to serve as a member of our board of directors due to his extensive leadership experience.

Joe Lonsdale. Mr. Lonsdale has served on our board of directors since May 2012. Mr. Lonsdale co-founded and is a General Partner at Eight Partners VC, LLC ("8VC"), a venture capital firm. Prior to joining 8VC in 2015, he was a founding partner or Formation8 GP LLC, the precursor venture fund to 8VC, and co-founded Palantir Technologies, a global software company, in 2003. Mr. Lonsdale also founded Addepar, an investment management technology company, and OpenGov, a software company that offers solutions for the public sector. He previously served as a Principal at Clarium Capital Management LLC, an investment management and hedge fund company. Mr. Lonsdale holds a B.S. in Computer Science from Stanford University. We believe that Mr. Lonsdale is qualified to serve as a member of our board of directors due to his extensive experience with technology companies.

Tanzeen Syed. Mr. Syed has served on our board of directors since October 2016. Mr. Syed is a Managing Director at General Atlantic and focuses on investments in General Atlantic's Technology sector. He rejoined General Atlantic in June 2018 after working there from 2006 to September 2013. Prior to rejoining General Atlantic, Mr. Syed was a Director at Temasek, an investment company, from July 2015 until June 2018, where he led U.S. technology growth investments, and prior to that, Mr. Syed was a Vice President at Great Hill Partners from October 2013 to June 2015, where he focused on Internet and software growth investing. Mr. Syed currently serves on the boards of directors of Kiwi.com, s.r.o., an online travel booking platform, and Riskified Ltd., a payment fraud management solution for online merchants, marketplaces, e-travel retailers, digital goods, and services providers, both of which are private portfolio companies of General Atlantic. Mr. Syed holds a B.A. in Economics and Political Science from Macalester College. We believe that Mr. Syed is qualified to serve as a member of our board of directors due to his extensive experience with e-commerce and technology companies.

Stephanie Tilenius. Ms. Tilenius has served on our board of directors since August 2020. She is currently the CEO of Vida Health, Inc., a mobile continuous healthcare platform, which she founded in January 2014. Ms. Tilenius was an Executive in Residence at Kleiner Perkins Caufield & Byers, a venture capital firm, from June 2012 until October 2014, primarily focusing on companies within its Digital Growth Fund. From February 2010 until June 2012, Ms. Tilenius was Vice President of Global Commerce and Payments at Google, Inc., a technology company, where she oversaw digital commerce, product search and payments. Prior to joining Google, she served in various positions at eBay Inc.from March 2001 until October 2009, ultimately as Senior Vice President of eBay.com and Global Products. Ms. Tilenius was also a co-founder of PlanetRx.com, an online healthcare provider. She currently serves on the board of directors of SeaGate Technology PLC and Tradesy, a privately-held ecommerce company focused on women's fashion, and within the past five years, served on the board of directors of Coach Inc. and Redbubble Limited. Ms. Tilenius holds a B.A. and an M.A from Brandeis University and an M.B.A. from the Harvard Business School. We believe that Ms. Tilenius is qualified to serve as a member of our board of directors due to her senior executive experience in the consumer internet and ecommerce sectors, as a company founder and as a board member for other public and private companies.

Hans Tung. Mr. Tung has served on our board of directors since January 2014. Mr. Tung is a Managing Partner at GGV Capital, a venture capital firm, where he has worked since October 2013, in Menlo Park, CA. Prior to joining GGV, Mr. Tung was a managing partner at Qiming Venture Partners, a venture capital firm, in Shanghai, China. Mr. Tung began his VC career with Bessemer Venture Partners, a venture capital firm, in Menlo Park, CA, and was a former entrepreneur himself and a former technology banker at Merrill Lynch in New York and Hong Kong. Mr. Tung has served on the board of directors of private companies including musical.ly, Poshmark, StockX, Misfit, Xiaomi, eHi Car Service, Vedantu, and Rupeek. He holds a B.S. in Management Science and Engineering from Stanford University. We believe that Mr. Tung is qualified to serve as a member of our board of directors due to his extensive experience with ecommerce companies, as well as his experience as a venture capitalist investing in technology companies.

#### **Family Relationships**

Our directors hold office until their successors have been elected and qualified or appointed, or the earlier of their death, resignation or removal. There are no family relationships among any of our directors or executive officers.

#### **Board Leadership Structure**

Our Board of Directors has combined the roles of Chairperson and Chief Executive Officer, who is Peter Szulczewski. Our Board has determined that we would be best served by having a

Chairperson with deep operational and strategic knowledge of our business. Our Board has also appointed as our Lead Independent Director. Our Board has determined that we would be best served by also having a lead independent director to be responsible for conducting sessions with the independent directors as part of every Board meeting, calling special meetings of the independent directors and chairing all meetings of the independent directors.

#### **Director Independence**

We intend to apply to have our Class A common stock listed on . The listing rules of this stock exchange generally require that a majority of the members of a listed company's board of directors be independent within 12 months following the closing of an initial public offering. Our board of directors has determined that none of our non-employee directors has a relationship that would interfere with the exercise of independent judgment in carrying out the responsibilities of a director and that each of these directors is "independent" as that term is defined under the rules of . The independent members of our board of directors will hold separate regularly scheduled executive session meetings at which only independent directors are present.

Audit committee members must also satisfy the independence rules in SEC Rule 10A-3 adopted under the Exchange Act. In order to be considered independent for purposes of Rule 10A-3, a member of an audit committee of a public company may not, other than in his or her capacity as a member of the audit committee, the board of directors, or any other board committee, accept, directly or indirectly, any consulting, advisory, or other compensatory fee from the listed company or any of its subsidiaries or be an affiliated person of the listed company or any of its subsidiaries. Each of qualify as an independent director pursuant to Rule 10A-3.

# **Controlled Company**

Because we qualify as a "controlled company" under the corporate governance rules for required to have a majority of our board of directors be independent, nor are we required to have a compensation committee or an independent nominating function. Our board of directors has not made a determination about whether or not to take advantage of the "controlled company" exemption. Additionally, as described in the section titled "Description of Capital Stock—Anti-Takeover Provisions—Amended and Restated Certificate of Incorporation and Amended and Restated Bylaw Provisions," so long as the outstanding shares of our Class B common stock represent a majority of the combined voting power of our common stock, Mr. Szulczewski will be able to effectively control all matters submitted to our stockholders for a vote, as well as the overall management and direction of our company.

# **Board Composition**

So long as the outstanding shares of our Class B common stock represent of the combined voting power of common stock, we will not have a classified board of directors, and all directors will be elected for annual terms.

When the outstanding shares of our Class B common stock represent less than of the combined voting power of common stock, we will have a classified board of directors consisting of three classes of approximately equal size, each serving staggered three-year terms. Our directors will be assigned by the then-current board of directors to a class.

Upon expiration of the term of a class of directors, directors for that class will be elected for three-year terms at the annual meeting of stockholders in the year in which that term expires. As a result, only one class of directors will be elected at each annual meeting of our stockholders, with the other

classes continuing for the remainder of their respective three-year terms. Each director's term continues until the election and qualification of his or her successor, or his or her earlier death, resignation, or removal.

So long as our board of directors is classified, only our board of directors may fill vacancies on our board. Any additional directorships resulting from an increase in the number of directors will be distributed among the three classes so that, as nearly as possible, each class will consist of one-third of the total number of directors.

The classification of our board of directors may have the effect of delaying or preventing changes in our control or management. See "Description of Capital Stock—Anti-Takeover Provisions—Certificate of Incorporation and Bylaw Provisions."

## **Board Oversight of Risk**

One of the key functions of our board of directors is informed oversight of our risk management process. In particular, our board of directors is responsible for monitoring and assessing strategic risk exposure. Our executive officers are responsible for the day-to-day management of the material risks we face. The risk oversight process includes receiving regular reports from board committees and members of senior management to enable our board of directors to understand our risk identification, risk management and risk mitigation strategies with respect to areas of potential material risk, including operations, finance, legal, regulatory, cyber security, strategic and reputational risk. Our board of directors administers its oversight function directly as a whole, as well as through various standing committees of our board of directors that address risks inherent in their respective areas of oversight. For example, our audit committee is responsible for overseeing the management of risks associated with our financial reporting, accounting and auditing matters; our compensation committee oversees the management of risks associated with our compensation policies and programs; and our nominating and corporate governance committee oversees the management of risks associated with director independence, conflicts of interest, composition and organization of our board of directors and director succession planning.

### **Board Committees**

Our board of directors has established an audit committee, a compensation committee and a nominating and corporate governance committee, to be effective after this offering. Our board of directors may establish other committees to facilitate the management of our business. Our board of directors and its committees set schedules for meeting throughout the year and can also hold special meetings and act by written consent from time to time, as appropriate. Our board of directors has delegated various responsibilities and authority to its committees as generally described below. The committees will regularly report on their activities and actions to the full board of directors. Each member of each committee of our board of directors qualifies as an independent director in accordance with listing standards. Each committee of our board of directors has a written charter approved by our board of directors. After this offering, copies of each charter will be posted on our website at www.wish.com under the Investor Relations section. The inclusion of our website address in this prospectus does not include or incorporate by reference the information on our website into this prospectus. Members serve on these committees until their resignation or until otherwise determined by our board of directors.

## **Audit Committee**

The members of our audit committee will be after this offering, each of whom can read and understand fundamental financial statements. are each independent under the rules and

regulations of the SEC and the listing standards of Our board of directors has determined that and meets the financial sophistication requirements of

applicable to audit committee members. will chair the audit committee. qualifies as an audit committee financial expert within the meaning of SEC regulations

Our audit committee assists our board of directors' oversight of the integrity of our financial statements, our compliance with legal and regulatory requirements, the qualifications, independence and performance of the independent registered public accounting firm, the design and implementation of our internal audit function and risk assessment and risk management. Among other things, our audit committee is responsible for reviewing and discussing with our management the adequacy and effectiveness of our disclosure controls and procedures. The audit committee also discusses with our management and independent registered public accounting firm the annual audit plan and scope of audit activities, scope and timing of the annual audit of our financial statements, and the results of the audit, quarterly reviews of our financial statements and, as appropriate, initiates inquiries into aspects of our financial affairs. Our audit committee is responsible for establishing and overseeing procedures for the receipt, retention and treatment of any complaints reporting accounting, internal accounting controls or auditing matters, as well as for the confidential and anonymous submissions by our employees of concerning questionable accounting or auditing matters. In addition, our audit committee has direct responsibility for the appointment, compensation, retention and oversight of the work of our independent registered public accounting firm. Our audit committee has sole authority to approve the hiring and discharging of our independent registered public accounting firm, all audit engagement fees and terms and all permissible non-audit engagements with the independent auditor. Our audit committee will review and oversee all related person transactions in accordance with our policies and procedures.

## Compensation Committee

The members of our compensation committee will be after this offering. will chair the compensation committee. Each member of our compensation committee is independent under the rules and regulations of the SEC and the listing standards of applicable to compensation committee members, is a "non-employee director," as defined in Rule 16b-3 adopted under Section 16 of the Exchange Act and an "outside director" under Regulation Section 1.162-27 adopted under Section 162(m) of the Code of 1986, as amended. Our compensation committee assists our board of directors with its oversight of the forms and amount of compensation for our executive officers, and the administration of our incentive plans for employees and other service providers, including our equity incentive plans, and certain other matters related to our compensation programs.

#### Nominating and Corporate Governance Committee

The members of our nominating and corporate governance committee will be after this offering. will chair the nominating and corporate governance committee. Our nominating and corporate governance committee assists our board of directors with its oversight of and identification of individuals qualified to become members of our board of directors, consistent with criteria approved by our board of directors, and selects, or recommends that our board of directors select, director nominees; develops and recommends to the board of directors a set of corporate governance guidelines; and oversees the evaluation of our board of directors.

## **Board Diversity**

Under our corporate governance guidelines, which will become effective upon the closing of this offering, diversity is one of several critical factors that the nominating and corporate governance committee considers when evaluating the composition of our board of directors, amongst other critical

selection criteria. We consider various diversity factors when considering director candidates, including include race, ethnicity, gender, national origin, and geography. Our board of directors currently includes directors with a range of diversity. We believe each director contributes to the board's overall diversity by providing a variety of perspectives based on distinct personal and professional experiences and backgrounds. We are committed to maintaining and enhancing the diversity of our board of directors and in furtherance of this, the nominating and corporate governance committee will conduct annual self-evaluations to assess its performance and effectiveness, which we expect will include its consideration of diversity and other selection criteria.

#### **Code of Conduct**

Our board of directors has adopted a code of conduct that will be effective after this offering. The code of conduct will apply to all of our employees, officers, and directors. We also expect our contractors, consultants, suppliers, agents and other third parties to follow our code of conduct in connection with their work for us. The full text of our code of conduct will be posted on our website after this offering at www.wish.com under the Investor Relations section. We intend to disclose future amendments to, or waivers of, our code of conduct as and to the extent required by SEC regulations, at the same location on our website identified above and in public filings. Information contained on our website is not incorporated by reference into this prospectus, and you should not consider information contained on our website to be part of this prospectus or in deciding whether to invest in our common stock. Our code of conduct represents the standards by which we operate and reflects that we are an ethical, mindful and transparent business. The purpose of our code of conduct is to promote honesty and integrity, including with respect to actual or apparent conflicts of interest between personal and professional relationships, to promote full, fair, accurate, timely and understandable disclosure in periodic reports to be filed by us and to promote compliance with all applicable rules and regulations that apply to us and our employees.

## **Compensation Committee Interlocks and Insider Participation**

As noted above, the compensation committee of our board of directors will consist of . None of the members of our compensation committee is or has been an officer or employee of our company. None of our executive officers currently serves, or in the past year has served, as a member of the board of directors or compensation committee of any entity that has one or more executive officers serving on our board of directors or compensation committee. See the section titled "Certain Relationships and Related Person Transactions" for information about related party transaction involving members of our compensation committee or their affiliates.

#### **Director Compensation**

Our directors play a critical role in guiding our strategic direction and overseeing the management of our Company. Ongoing developments in corporate governance and financial reporting have resulted in an increased demand for highly qualified and productive public company directors. The varied responsibilities, the substantial time commitment and the potential risks of serving as a director for a public company require that we provide adequate compensation for the continued service of our non-employee directors by offering them compensation that is commensurate with the workload and the demands we place on them.

In fiscal 2019, we did not have a formal policy with respect to compensation payable to our non-employee directors for service as directors. The table below shows the total compensation that we paid to Mr. Emanuel, our only non-employee director who received compensation, during 2019:

	Fees Earned					
	or Paid in	Stock	Option	All Other		
<u>Name</u>	Cash (\$)	Awards (\$) <sup>(1)</sup>	Awards (\$) <sup>(2)</sup>	Compensation (\$)	Total (\$)	
Ari Emanuel		\$1,132,413 <sup>(3)</sup>			\$1,132,413	

(1) In accordance with SEC rules, this column reflects the grant date fair value of RSUs calculated in accordance with ASC Topic 718 for stock-based compensation transactions. As of December 31, 2019, only one of our non-employee directors, Mr. Emanuel, held RSUs, which are described in further detail in the footnote below.

(2) As of December 31, 2019, only one of our non-employee directors, Mr. Lonsdale, held outstanding options to purchase shares of our common stock (for 62,500 shares).

Mr. Emanuel received a grant of 10,000 RSUs on November 22, 2019, which are subject to both a service-based vesting condition and a liquidity-based vesting condition. The

Mr. Szulczewski receives no additional compensation for his service as a director, nor is it intended that he will receive additional compensation for such role following the completion of this offering. All of our non-employee directors are entitled to reimbursement for their reasonable travel and lodging expenses for attending board and board committee meetings.

Historically, we have not compensated Messrs. Tung or Syed for their services as directors. Mr. Emanuel became a member of our board of directors in November 2019. Ms. Tilenius received a grant of 11,111 RSUs on August 5, 2020, which are subject to both a service-based vesting condition and a liquidity-based vesting condition. The service-based vesting condition will be satisfied for one fourth of the total number of RSUs on each of the first four anniversaries of her appointment, subject to her continued service through each such date. All RSUs granted to Ms. Tilenius shall vest in full immediately prior to, but conditioned upon, the consummation of a sale of our company.

Prior to the completion of this offering, we intend to adopt a non-employee director compensation program, which will become effective on the date of this offering.

<sup>(3)</sup> Mr. Emanuel received a grant of 10,000 RSUs on November 22, 2019, which are subject to both a service-based vesting condition and a liquidity-based vesting condition. The service-based vesting condition applicable to his RSUs will be satisfied in four equal annual installments on each anniversary of the commencement of his service with us. The service-based vesting condition applicable to Mr. Emanuel's RSUs will also be satisfied in its entirety upon the consummation of a sale of our company.

#### **EXECUTIVE COMPENSATION**

## **Compensation Discussion and Analysis**

This Compensation Discussion and Analysis describes the compensation for our principal executive officer, our principal financial officer and the three most highly compensated executive officers (other than our chief executive officer and chief financial officer) who were serving as executive officers at the end of 2019, whom we refer to herein as our "named executive officers." Our named executive officers for 2019 were:

- · Peter Szulczewski, our Founder, CEO, and Chairperson of our Board of Directors;
- · Rajat Bahri, our Chief Financial Officer;
- · Devang Shah, our General Counsel;
- · Thomas Chuang, our Vice President of Operations; and
- · Pai Liu, our Vice President of Data Science.

This Compensation Discussion and Analysis describes the material elements of our executive compensation program for our named executive officers during 2019 and discusses the key factors that were considered in determining their compensation.

### Compensation Philosophy and Design

Our general compensation philosophy has been to compensate our named executive officers using competitive compensation packages that reward the achievement of our short-term and long-term business objectives and strategies and align their interests with the interests of our stockholders. To focus our executive officers on the fulfillment of our business objectives, a significant portion of their compensation has been and continues to be equity-based.

To date, the compensation packages of our named executive officers have reflected our stage of development as a privately-held company. Accordingly, we have emphasized the use of equity compensation in the form of RSU awards to motivate our named executive officers to focus on the growth of our overall enterprise value and, correspondingly, to create sustainable long-term value for our stockholders. We believe that RSU awards offer our named executive officers a valuable long-term incentive that aligns their interests with the interests of our stockholders. In addition, to maintain a competitive compensation program, we also offer cash compensation in the form of base salaries to reward individual contributions and compensate our employees for their day-to-day responsibilities.

Currently, we do not provide any form of short-term incentive (such as performance bonuses or other incentives) to our named executive officers. We expect our named executive officers to perform at a level deserving of a bonus, however, and have taken this into consideration in establishing their current total direct compensation opportunities. We believe that having a substantial portion of our named executive officers' compensation tied to equity awards aligns more closely with our business strategy to focus on long-term growth and innovation. We believe that our current compensation structure promotes a focus on long-term retention and stockholder value creation.

# **Compensation-Setting Process**

During 2019, our CEO and board of directors were involved in determining the compensation of our named executive officers. Generally, our CEO would determine the annual base salary and size of the equity award for our named executive officers (other than his own compensation) in connection with conducting their annual performance review. Our CEO would present his proposed equity awards for our named executive officers to our board of directors for their consideration and approval.

With respect to cash compensation, our Vice President of Human Resources gathered data from the Radford technology company surveys to develop an understanding of the competitive market for various executive positions and formulated recommendations for our CEO with respect to the cash compensation, including any base salary adjustment, for our named executive officers.

With respect to equity compensation, our CEO would determine the appropriate size of each named executive officer's award in connection with the performance review process and after giving consideration to each named executive officer's anticipated future contributions, the criticality of his position, and individual performance. Our CEO would then present his proposed equity awards to our board of directors for its consideration. In approving the equity awards, our board of directors took into account such factors as company and individual performance, current and anticipated business conditions, the criticality of the position, the scope of the role and responsibilities, the individual's tenure, the existing retention hold of the individual's existing equity position, internal pay parity, and the recommendations of our CEO.

In May 2019, our board of directors formed a compensation committee comprised of three independent directors and delegated to such committee responsibility for overseeing our executive compensation program. The compensation committee held its first meeting in June 2019, at which time it approved a committee charter, which was then presented to our board of directors for approval. Among other things, the charter authorized the committee to review and recommend to our board of directors the compensation payable to our CEO, executive vice presidents and their direct reports, including the adjustment of base salaries each year and all bonus and other incentive compensation programs for such executives. Since most of the compensation actions and decisions for 2019 had already taken place by the time the compensation committee was formed, it took no formal actions in 2019.

For 2020 and going forward, the compensation committee will be responsible for our executive compensation program, including establishing our executive compensation philosophy, policies, and practices, recommending to the independent members of our board of directors the specific compensation, including cash and equity for our CEO, and determining the specific compensation, including cash and equity, for our other named executive officers. The members of the compensation committee have only recently begun to discuss our overall executive compensation philosophy. Therefore, the approach that we will use to compensate our named executive officers in the future may not be the same as how they have been compensated previously. We expect that the compensation committee will continue to review, evaluate and modify the executive compensation framework as a result of our becoming a publicly-traded company after this offering. Consequently, our compensation program following this offering may, over time, vary significantly from our historical practices.

In May 2019, we engaged Compensia, a national compensation consulting firm, to provide executive compensation advisory services for 2019, including the following:

- development of a compensation peer group for purposes of conducting a series of competitive market assessments;
- an analysis of our executive officers' base salaries and equity compensation levels and plan structures for 2019;
- assistance with a review of our equity compensation strategy, including the development of award guidelines and an aggregate spending budget;
- · a review of considerations and market practices related to short-term cash incentive plans; and
- a review of board of director compensation market practices among late-stage pre-IPO and recently-public technology companies.

Subsequently, in August 2020, the compensation committee retained Compensia to provide executive compensation advisory services for 2020.

#### **Compensation Elements**

In 2019, we used cash and equity compensation to provide an overall competitive total compensation and benefits package to our named executive officers that is tied to creating value, commensurate with our results and aligns with our business strategy. Set forth below are the key elements of the compensation program for our named executive officers for 2019.

# Base Salary

As part of his annual performance review of his direct reports in March 2019, our CEO also reviewed their base salaries for the year. In addition to considering their past performance and expected future contributions, our CEO reviewed an analysis of competitive market data drawn from Radford technology company compensation surveys prepared by our Vice President of Human Resources. Based on these considerations, as well the fact that we place greater emphasis on providing compensation in the form of equity awards in order to motivate our named executive officers and foster long-term growth for the benefit of our stockholders, our CEO determined to increase the base salaries of our other incumbent named executive officers at that time. At the same time, our Board of Directors determined to maintain the annual base salary of our CEO at its fiscal 2018 level.

The base salaries of our incumbent named executive officers for fiscal 2019 were as follows:

Named Executive Officer	Fisc	al 2018 Base Salary	Base Salary <sup>(1)</sup>	Percentage Increase
Mr. Szulczewski	\$	450,000	\$ 450,000	<del></del> %
Mr. Bahri		295,000	380,000	29
Mr. Shah		284,000	340,000	20
Mr. Chuang		210,650	252,780	20

<sup>(1)</sup> Base salary increases were effective April 1, 2019.

Subsequently, our CEO determined to increase Mr. Chuang's annual base salary to \$290,000 effective August 1, 2019 in order to further recognize his contributions to the Company during 2018 and ensure internal parity with the other individuals in the Company holding similar positions.

In connection with Mr. Liu's appointment as our Director, Data Science effective September 23, 2019, we agreed to pay him an annual base salary of \$275,000. In addition, we agreed to pay him a signing bonus in the amount of \$50,000, subject to repayment of the full net amount of the bonus if he voluntarily leaves employment before the first anniversary of his employment start date. Mr. Liu was promoted and appointed our Vice President of Data Science on August 12, 2020.

## Long-Term Incentive Compensation

As a privately-held company, we have used RSU awards as the principal component of our executive compensation program. Consistent with our compensation objectives, we believe this approach aligned our named executive officers' contributions with our long-term interests and allowed our named executive officers to be accountable for and participate in any future appreciation in our common stock. Historically, our RSU awards have generally included both a multi-year service-based vesting requirement (generally four years) and a liquidity event vesting requirement (that is, the effectiveness of either a sale event or an initial public offering) (the "Liquidity Event"), allowing them to serve as an effective retention tool while also motivating our named executive officers to work toward corporate objectives that provide a meaningful return to our stockholders.

In granting equity awards, our board of directors generally considers, among other things, the named executive officer's cash compensation, the need to create a meaningful opportunity for reward predicated on the creation of long-term stockholder value, our financial results, our total annual equity budget, an evaluation of the expected and actual performance of each named executive officer, his individual contributions and responsibilities, the retention hold of his existing equity awards and how that lapses over time as awards vest, and the recommendations of our CEO (except with respect to his own equity award).

In February 2019, our board of directors determined to grant RSU awards to certain employees who had worked as part of a series of teams involved in several projects that were considered to have had a significant positive impact on our business. The number of shares of our common stock subject to each award varied by project, the various milestones assigned to that project, and the employment level of the employee. Recipients of the awards included two of the named executive officers, Messrs. Bahri and Chuang, who were each granted RSU awards in the following amounts:

	RSU Award	F	RSU Award
Named Executive Officer	(number of shares)	(grant	date fair value)
Mr. Bahri	12,637	\$	1,471,958
Mr. Chuang	12.637	\$	1.471.958

The "Special Impact" awards vest (i) over a four-year period, with one-quarter of the shares of our common stock subject to the award vesting on the first anniversary of the vesting commencement date, and the remaining shares vesting monthly thereafter over 36 months in equal monthly increments, contingent upon the named executive officer remaining continuously employed by us through each applicable vesting date and (ii) subject to the occurrence of a Liquidity Event.

As part of our annual performance review cycle, in May 2019 our board of directors determined to grant RSU awards to our incumbent named executive officers other than our CEO. The number of shares of our common stock subject to each award was determined by our board of directors after considering the factors described above. Consistent with our CEO's recommendation, our board of directors determined that his existing vested and unvested equity holdings (including his 2018 RSU award covering 648,689 shares in October 2018) provided the necessary motivation and retention incentive and therefore did not grant any equity grants to him in 2019.

The RSU awards granted to our other named executive officers in the following amounts:

Named Executive Officer Mr. Szulczewski	RSU Award (number of shares)	RSU Award t date fair value) —
Mr. Bahri	147,429	\$ 17,117,803
Mr. Shah	23,589	2,738,890
Mr. Chuang	2,949	342,405

The annual "Refresh" awards vest (i) over a four-year period, with 10% of the shares of our common stock subject to the award vesting monthly for the first year from the vesting commencement date, 20% of the shares subject to the award vesting monthly for the one-year period between the first and second anniversaries of the vesting commencement date, 30% of the shares subject to the award vesting monthly for the one-year period between the second and third anniversaries of the vesting commencement date, and 40% of the shares subject to the award vesting monthly for the one-year period between the third and fourth anniversaries of the vesting commencement date, contingent upon the named executive officer remaining continuously employed by us through each applicable vesting date and (ii) subject to the occurrence of a Liquidity Event.

In connection with Mr. Liu's appointment as our Director, Data Science effective September 23, 2019, on November 22, 2019 our board of directors granted him an RSU award for 11,794 shares with a grant date fair value of \$1,335,568. This RSU award vests (i) over a four-year period, with one-quarter of the shares of our common stock subject to the award vesting on the first anniversary of the vesting commencement date, and the remaining shares vesting monthly thereafter over 36 months in equal monthly increments, contingent upon his remaining continuously employed by us through each applicable vesting date and (ii) subject to the occurrence of a Liquidity Event.

The offering currently being contemplated by this prospectus will satisfy the Liquidity Event vesting requirement. As a result, if we successfully complete this offering, the named executive officers will vest in their "Special Impact" and "Refresh" RSU awards upon their satisfaction of the service-based vesting requirement.

The RSU award agreements between us and our named executive officers provide that vested RSU awards will settle (that is, we will issue the vested shares to our named executive officer) once the named executive officer has satisfied the vesting conditions applicable to such shares, provided that shares that become vested upon our completion of an initial public offering will settle on the earlier of (i) the 185th day following the initial public offering or (ii) two and one-half months following the end of the year in which the vesting conditions are satisfied.

Information with respect to the potential accelerated vesting applied to the equity awards held by certain of our named executive officers in the event of a change of control of the Company or the termination of the named executive officer's services in connection with or following a change of control of the Company is discussed under "Potential Payments upon Termination or Change in Control" below.

#### Health and Welfare Benefits

Our named executive officers are eligible to participate in the same employee benefit plans, and on the same terms and conditions, as all other full-time, salaried U.S. employees. These benefits include medical, dental, and vision insurance, business travel insurance, an employee assistance program, health and dependent care flexible spending accounts, basic life insurance, accidental death and dismemberment insurance, short-term and long-term disability insurance and commuter benefits.

We design our employee benefits programs to be affordable and competitive in relation to the market as well as compliant with applicable laws and practices. We adjust our employee benefits programs as needed based upon regular monitoring of applicable laws and practices and the competitive market.

#### Retirement Benefits

We maintain a Section 401(k) plan for our employees, including our named executive officers. The Section 401(k) plan is intended to qualify under Section 401(k) of the Code, so that contributions to the plan by employees or by us, and the investment earnings thereon, are not taxable to the employees until withdrawn, and so that contributions made by us, if any, will be deductible by us when made. Currently, we do not provide company matching contributions to participants in the Section 401(k) plan.

We do not provide pension arrangements for our named executive officers or other employees, nor do we provide any nonqualified defined contribution or other deferred compensation plans to any of our employees.

#### Perquisites and other Personal Benefits

Currently, we do not view perquisites or other personal benefits as a significant component of our executive compensation program. Accordingly, we do not provide significant perquisites or other personal benefits to our named executive officers except as generally made available to our employees or in situations where we believe it is appropriate to assist an individual in the performance of his or her duties, to make him or her more efficient and effective, and for recruitment and retention purposes. During 2019, none of our named executive officers received perquisites or other personal benefits that were, in the aggregate, \$10,000 or more for each individual.

In the future, we may provide perquisites or other personal benefits in limited circumstances, such as those described in the preceding paragraph. All future practices with respect to perquisites or other personal benefits will be approved and subject to periodic review by the compensation committee.

## **Employment Arrangements**

We have entered into written employment offer letters with each of our named executive officers (other than our CEO). We believe that these arrangements were necessary to secure the service of these individuals in a highly competitive job market.

Each of these employment offer letters does not have a specific term, provides for "at will" employment (meaning that either we or the named executive officer may terminate the employment relationship at any time without cause) and generally set forth the named executive officer's initial base salary, eligibility to participate in our standard employee benefit plans and programs and includes a recommendation for an equity award to be approved by our board of directors. In addition, each of these employment offer letters required the named executive officer to execute our standard Proprietary (Confidential) Information and Inventions Agreement.

The employment offer letter we entered into with Mr. Bahri provides for 100% accelerated vesting of his RSU award described in his employment offer letter if we are subject to a "sale event" (as defined in his employment offer letter) while he is providing services to us. Mr. Shah's employment offer letter provides for 50% accelerated vesting of any then unvested RSUs under his RSU award described in his employment offer letter if we are subject to a "sale event" (as defined in his employment offer letter) before his service with us terminates and he is subject to an "involuntary termination" (as defined in his employment offer letter) within 12 months following the sale event. For additional information regarding the termination and change in control benefits payable to our named executive officers, see "—Potential Payments upon Termination or Change in Control" below.

For detailed descriptions of the employment offer letters of our named executive officers, see "—Potential Payments upon Termination or Change in Control" below.

#### **Tax and Accounting Considerations**

As a general matter, we review and consider the various tax and accounting implications of the compensation vehicles that we use.

#### **Deductibility of Executive Compensation**

In approving the amount and form of compensation for our named executive officers, our board of directors will consider all elements of our cost of providing such compensation, including the potential impact of Section 162(m) of the Internal Revenue Code.

Our board of directors believes that our stockholders' interests are best served if its discretion and flexibility in awarding compensation is not restricted, even though some portion may result in non-deductible compensation expense. Thus, we expect that in future years some portion of the compensation of our named executive officers may not be fully deductible by us for federal income tax purposes.

#### Accounting Implications

We follow FASB ASC Topic 718, Compensation—Stock Compensation, for our stock-based compensation awards. FASB ASC Topic 718 requires us to measure the compensation expense for all share-based payments made to our employees and the members of our board of directors, including options to purchase shares of our common stock and other stock-based awards, based on the grant date "fair value" of these awards. This calculation is performed for financial accounting purposes and reported in the compensation tables below, even though recipients may never realize any value from their awards. FASB ASC Topic 718 also requires us to recognize the compensation cost of our share-based compensation awards in our income statements over the period that a recipient is required to render services in exchange for the option or other award.

#### Elements of Total Compensation—Risks and Mitigating Factors

We believe that the structure of our executive compensation program provides a mix of cash and equity compensation that balances short- and long-term incentives. We believe that the different time horizons and metrics used in the annual and long-term elements of compensation provide incentives to build our business prudently and profitably over time, while encouraging retention of our top talent. In addition, each element of compensation has been designed and is administered in a manner intended to minimize potential risks to us. The result is a program that we believe mitigates inappropriate risk taking and aligns the interests of our executive officers with those of our stockholders. Moreover, we have determined that any risks arising from our compensation policies and practices for all of our employees are not reasonably likely to have a material adverse effect on us.

#### **Summary Compensation Table for 2019**

The following table sets forth information concerning the total compensation awarded to, earned by, or paid to our named executive officers for the year ended December 31, 2019.

	Salary	Bonus	Stock Awa <u>r</u> ds	Non-Equity Incentive Plan Compensation	All Other Compensation	Total
Name and Principal Position	(\$)	(\$)	(\$) <sup>(1)</sup>	(\$)	(\$)	(\$)
Peter Szulczewski						
Founder, CEO, Chairperson of the Board	450,000	_	_	_	_	450,000
Rajat Bahri						
Chief Financial Officer	358,750	_	18,589,761	_	_	18,948,511
Devang Shah						
General Counsel	326,000	_	2,738,890	_	_	3,064,890
Thomas Chuang						
Vice President of Operations	257,756	_	1,814,363	_	_	2,072,119
Pai Liu						
Vice President of Data Science	75,000(2)	50,000(3)	1,335,568	_	_	1,460,568

<sup>(1)</sup> The amounts reported in this column reflect the accounting value for these equity awards and may not correspond to the actual economic value that may be received by our named executive officers from the equity awards. In accordance with SEC rules, this column reflects the grant date fair value of RSUs calculated in accordance with ASC Topic 718 for stock-

based compensation transactions. Our RSUs are subject to both a service-based vesting condition and a liquidity-based vesting condition. The grant date fair values do not take into account any estimated forfeitures related to the service-based vesting condition.

Represents Mr. Liu's prorated annual salary of \$275,000 following his commencement of employment in September 2019. Represents a signing bonus paid to Mr. Liu following his commencement of employment with us.

#### **Grants of Plan-Based Awards Table for 2019**

The following table provides information on the grants of plan-based awards made to each named executive officer during the year ended December 31, 2019.

Name	Grant Date	Estimated Future Payouts Under Equity Incentive Plan Awards Target (#)	All Other Stock Awards: Number of Shares of Stock or Units (#)	All Other Option Awards: Number of Securities Underlying Options (#)	Grant Date Fair Value of Stock and Option Awards (\$) <sup>(1)</sup>
Peter Szulczewski	Orani Bate			<u> </u>	
Rajat Bahri	2/5/2019	_	12,637(2)	_	1,471,958
	5/2/2019	_	147,429(3)	_	17,117,803
Devang Shah	5/2/2019	_	23,589(3)	_	2,738,890
Thomas Chuang	2/5/2019	-	12,637(2)	=	1,471,958
	5/2/2019	_	2,949(3)	_	342,405
Pai Liu	11/22/2019	-	11,794 <sup>(2)</sup>	_	1,335,568

<sup>(1)</sup> This column reflects the grant date fair value of RSU awards, calculated in accordance with ASC Topic 718 for stock-based compensation. Our RSUs are subject to both a service-based vesting condition and a liquidity-based vesting condition.

The service-based vesting condition is satisfied as to 1/4th of the total shares of Class B common stock underlying the RSU award on the 12-month anniversary of the Vesting

## **Outstanding Equity Awards at 2019 Year-End**

The following table provides information regarding outstanding equity awards held by our named executive officers as of December 31, 2019. The number of shares subject to each award and, where applicable, the exercise price per share. The vesting schedule applicable to each outstanding award is described in the footnotes to the table below.

		Option Awards				Stock Awards		
<u>Name</u>	Vesting Commencement Date	Number of Securities Underlying Unexercised Options Exercisable (#)	Number of Securities Underlying Unexercised Options Unexercisable (#)	Option Exercise Price (\$)	Option Expiration Date	Number of Shares or Units of Stock That Have Not Vested (#) <sup>(1)</sup>	Market Value of Shares or Units of Stock That Have Not Vested (\$)(*)	
Peter Szulczewski	11/17/2013	837,500 <sup>(2)</sup>		1.49	4/15/2024			
	6/9/2014	$3,500,000^{(3)}$	_	2.38	8/11/2024	_	_	
	1/1/2015	_	<del>-</del>	<del>-</del>	_	238,886(4)		
	6/1/2016	_	_	_	_	68,832(4)		
	4/24/2017	_	_	_	_	258,155 <sup>(4)</sup>		
	1/1/2018	_	_	_	_	81,764(5)		
	9/23/2018	_	-	_	_	648,689 <sup>(6)</sup>		

Commencement Date, and the service-based condition is satisfied as to 1/36th of the remaining shares on a monthly basis thereafter, subject to continued service to us through

The service-based vesting condition is satisfied on a monthly basis over a period of four years from the Vesting Commencement Date, with 10% of the total shares of Class B common stock underlying the RSU award vesting over the first year, 20% of the total shares vesting over the second year, 30% of the total shares vesting over the third year, and 40% of the total shares vesting over the fourth year, subject to continued service to us through each vesting date.

			Option Award		Stock Awards		
Name	Vesting Commencement Date	Number of Securities Underlying Unexercised Options Exercisable (#)	Number of Securities Underlying Unexercised Options Unexercisable (#)	Option Exercise Price (\$)	Option Expiration Date	Number of Shares or Units of Stock That Have Not Vested (#) <sup>(1)</sup>	Market Value of Shares or Units of Stock That Have Not Vested (\$)(*)
Rajat Bahri	12/7/2016					248,399 <sup>(7)</sup>	
	1/1/2018	_	_	_	_	29,732(5)	
	1/1/2019	_	_	_	_	12,637 <sup>(4)</sup>	
	4/1/2019	_	_	_	_	147,429 <sup>(8)</sup>	
Devang Shah	2/5/2018	-	=	_	_	57,367 <sup>(9)</sup>	
	4/1/2019	_	_	_	_	23,589(8)	
Thomas Chuang	7/1/2014	27,900(10)	_	2.38	8/11/2024	_	_
	1/1/2015	_	_	_	_	1,620 <sup>(9)</sup>	
	4/24/2017	_	_	_	_	2,004 <sup>(9)</sup>	
	6/1/2016	_	_	_	_	1,620 <sup>(9)</sup>	
	5/1/2018	_	_	_	_	2,081(9)	
	1/1/2019	_	_	_	_	12,637(4)	
	4/1/2019	-	-	=	_	2,949(8)	
Pai Liu	9/23/2019	_	_	_	_	11.794 <sup>(4)</sup>	

(\*) Market value is based on the fair market value of our Class B common stock on . As there was no public market for our common stock on , we have assumed , which represents the midpoint of the price range set forth on the cover page of this prospectus. that the fair market value on such date was \$

(3) (4)

- The shares subject to this option were fully vested as of June 9, 2018. The service-based vesting condition is satisfied as to 1/4th of the total shares of Class B common stock underlying the RSU award on the 12-month anniversary of the Vesting Commencement Date, and the service-based condition is satisfied as to 1/36<sup>th</sup> of the remaining shares on a monthly basis thereafter, subject to continued service to us through
- each vesting date.

  The service-based vesting condition is satisfied as to 1/60<sup>th</sup> of the total shares of Class B common stock underlying the RSU award on a monthly basis from the Vesting (5)
- Commencement Date, subject to continued service to us through each vesting date.

  The service-based vesting condition is satisfied as to 1/48<sup>th</sup> of the total shares of Class B common stock underlying the RSU award on a monthly basis from the Vesting (6) Commencement Date, subject to continued service to us through each vesting date.
- The service-based vesting condition is satisfied as to 1/5th of the total shares of Class B common stock underlying the RSU award on the 12-month anniversary of the Vesting Commencement Date, and the service-based condition is satisfied as to 1/20<sup>th</sup> of the total shares on a quarterly basis thereafter, subject to continued service to us through each vesting date.
- The service-based vesting condition is satisfied on a monthly basis over a period of four years from the Vesting Commencement Date, with 10% of the total shares of Class B common stock underlying the RSU award vesting over the first year, 20% of the total shares vesting over the second year, 30% of the total shares vesting over the third year, and 40% of the total shares vesting over the fourth year, subject to continued service to us through each vesting date.

  The service-based vesting condition is satisfied as to 1/5th of the total shares of Class B common stock underlying the RSU award on the 12-month anniversary of the Vesting Commencement Date, and the service-based condition is satisfied as to 1/48<sup>th</sup> of the remaining shares on a monthly basis thereafter, subject to continued service to us through (8)
- (9) each vesting date.
- (10)The shares subject to this option were fully vested as of July 1, 2019.

RSUs granted to our executive officers only vest upon the satisfaction of both (i) a service-based vesting condition and (ii) a liquidity-based vesting condition. The liquidity-based vesting condition is (i) the effective date of our initial public offering or (ii) a sale event (as defined in our RSU award agreements).

The shares subject to this option were fully vested as of November 17, 2017. (1)

## Stock Vested in 2019

The following table sets forth the number of shares acquired and the value realized upon exercise of stock options during 2019 by each of our named executive officers:

	C	Option Awards
Name	Number of Shares Acquired on Exercise (#)	Value Realized on Exercise (\$) <sup>(1)</sup>
Peter Szulczewski		
Rajat Bahri	<del>-</del>	<del>-</del>
Devang Shah	_	_
Thomas Chuang	4,600	548,872
Pai Liu	_	_

<sup>(1)</sup> The aggregate dollar amount realized upon the exercise of option awards represents the amount by which (x) the aggregate market price of the shares of our Class B common stock on the date of exercise, which is an assumed fair value as of the date of exercise, exceeds (y) the aggregate exercise price of the option.

#### **Pension Benefits**

We do not have any defined benefit pension plans.

## **Nonqualified Deferred Compensation**

We do not offer any nonqualified deferred compensation plans.

## Potential Payments upon Termination or Change in Control

We do not have employment agreements with our named executive officers. Pursuant to the terms of his employment offer letter and his RSU agreements, Mr. Bahri will receive 100% accelerated vesting of his outstanding RSU awards if we are subject to a sale event while he is providing services to us. Pursuant to the terms of his employment offer letter and his RSU agreements, Mr. Shah will receive 50% accelerated vesting of any then unvested RSUs if we are subject to a sale event while he is providing services to us and he is terminated without cause or resigns for good reason within 12 months following the sale event.

The following table sets forth the amounts payable to each of our current named executive officers based on an assumed termination or change of control of us as of December 31, 2019:

Name	Cash Severance (\$)	Health and Other Insurance Benefits (\$)	Stock Options (Unvested and Accelerated) (\$)	Restricted Stock Units (Unvested and Accelerated) (\$) <sup>(1)</sup>	Total (\$)
Peter Szulczewski					
Termination for reasons other than Cause, death or Disability, or for Good Reason					
Change of Control or Termination in connection with a Change of Control					
Rajat Bahri					
Termination for reasons other than Cause, death or Disability, or for Good Reason					
Change of Control or Termination in connection with a Change of Control <sup>(2)</sup>					
Devang Shah					
Termination for reasons other than Cause, death or Disability, or for Good Reason					
Change of Control or Termination in connection with a Change of Control <sup>(3)</sup>					
Thomas Chuang					
Termination for reasons other than Cause, death or Disability, or for Good Reason					
Change of Control or Termination in connection with a Change of Control					
Pai Liu					
Termination for reasons other than Cause, death or Disability, or for Good Reason					
Change of Control or Termination in connection with a Change of Control					
(1) The value of accelerated DSI is were determined by multiplying the	number of universal am	ad accelerated DCI is by	the commend initial much	lia affaring price of C	

<sup>(1)</sup> The value of accelerated RSUs were determined by multiplying the number of unvested and accelerated RSUs by the assumed initial public offering price of \$ share, the midpoint of the price range set forth on the coverage page of this prospectus.

share, the midpoint of the price range set forth on the coverage page of this prospectus.

Mr. Bahri is entitled to receive 100% accelerated vesting of his outstanding RSUs if we are subject to a sale event while he is providing services to us

# 2020 Equity Incentive Plan

#### General

Our board of directors intends to adopt our 2020 Equity Incentive Plan (the "2020 Plan") and submit it to our stockholders for approval prior to the offering. We expect that our 2020 Plan will become effective immediately on adoption although no awards will be made under it until the effective date of the registration statement of which this prospectus is a part. Our 2020 Plan is intended to replace our 2010 Stock Plan (the "2010 Plan"). However, awards outstanding under our 2010 Plan will continue to be governed by their existing terms. Although not yet adopted, we expect that our 2020 Plan will have the features described below.

Mr. Shah is entitled to receive 50% accelerated vesting of any then unvested RSUs upon a qualifying termination of employment within 12 months of the change of control.

#### Share Reserve

The number of shares of our Class A common stock available for issuance under our 2020 Plan will equal the sum of shares plus up to shares remaining available for issuance under, or issued pursuant to or subject to awards granted under, our 2010 Plan. The number of shares reserved for issuance under our 2020 Plan will be increased automatically on the first business day of each of our fiscal years, commencing in 2021 and ending in 2030, by a number equal to the smallest of:

- shares:
- 5% of the shares of common stock outstanding on the last business day of the prior fiscal year; or
- · the number of shares determined by our board of directors.

In general, to the extent that any awards under our 2020 Plan are forfeited, terminate, expire or lapse without the issuance of shares, or if we repurchase the shares subject to awards granted under our 2020 Plan, those shares will again become available for issuance under our 2020 Plan, as will shares applied to pay the exercise or purchase price of an award or to satisfy tax withholding obligations related to any award.

#### Administration

The compensation committee of our board of directors will administer our 2020 Plan. The compensation committee will have complete discretion to make all decisions relating to our 2020 Plan and outstanding awards, including repricing outstanding options and modifying outstanding awards in other ways.

#### Eligibility

Employees, non-employee directors, consultants and advisors will be eligible to participate in our 2020 Plan.

Under our 2020 Plan, the aggregate grant date fair value of awards granted to our non-employee directors may not exceed \$ in any one fiscal year, except that the grant date fair value of awards granted to newly appointed non-employee directors may not exceed \$ in the fiscal year in which such non-employee director is initially appointed to our board of directors.

#### Types of Awards

Our 2020 Plan will provide for the following types of awards:

- · incentive and nonstatutory stock options;
- stock appreciation rights;
- · restricted shares; and
- · restricted stock units.

#### Options and Stock Appreciation Rights

The exercise price for options granted under our 2020 Plan may not be less than 100% of the fair market value of our Class A common stock on the grant date. Optionees will be permitted to pay the exercise price in cash or, with the consent of the compensation committee:

· with shares of common stock that the optionee already owns;

- by an immediate sale of shares through a broker approved by us;
- by instructing us to withhold a number of shares having an aggregate fair market value that does not exceed the exercise price; or
- · by other methods permitted by applicable law.

An optionee who exercises a stock appreciation right receives the increase in value of our Class A common stock over the base price. The base price for stock appreciation rights may not be less than 100% of the fair market value of our Class A common stock on the grant date. The settlement value of a stock appreciation right may be paid in cash, shares of our Class A common stock or a combination.

Options and stock appreciation rights vest as determined by the compensation committee. In general, they will vest over a four-year period following the date of grant or, in the case of grants made to new hires, over the four-year period following their first day of employment. Options and stock appreciation rights expire at the time determined by the compensation committee but in no event more than 10 years after they are granted. These awards generally expire earlier if the participant's service terminates earlier.

#### Restricted Shares and Stock Units

Restricted shares and stock units may be awarded under our 2020 Plan in return for any lawful consideration, and participants who receive restricted shares or stock units generally are not required to pay cash for their awards. In general, these awards will be subject to vesting. Vesting may be based on length of service or upon satisfaction of other conditions determined by the compensation committee.

Settlement of vested stock units may be made in the form of cash, shares of Class A common stock or a combination.

#### Corporate Transactions

In the event we are a party to a merger, consolidation or certain change in control transactions, outstanding awards granted under our 2020 Plan, and all shares acquired under our 2020 Plan, will be subject to the terms of the definitive transaction agreement (or, if there is no such agreement, as determined by our compensation committee). Unless an award agreement provides otherwise, such treatment may include any of the following with respect to each outstanding award:

- · the continuation, assumption or substitution of an award by a surviving entity or its parent;
- the cancellation of an award without payment of any consideration;
- the cancellation of the vested portion of an award (and any portion that becomes vested as of the effective time of the transaction) in exchange for a payment equal to the excess, if any, of the value that the holder of each share of our Class A common stock receives in the transaction over (if applicable) the exercise price otherwise payable in connection with the award; or
- the assignment of any reacquisition or repurchase rights held by us in respect of an award of restricted shares to the surviving entity or its parent (with proportionate adjustments made to the price per share to be paid upon exercise of such rights).

Each award held by a participant who remains a service provider with us as of the effective time of a merger or change in control, will become fully vested and, if applicable, exercisable immediately

prior to the effective time of the transaction, unless the applicable award agreement provides otherwise or the award is continued, assumed or substituted (as provided above). The compensation committee is not required to treat all awards, or portions thereof, in the same manner.

The vesting of an outstanding award may be accelerated by the compensation committee upon the occurrence of a change in control, whether or not the award is to be assumed or replaced in the transaction, or in connection with a termination of service following a change in control transaction.

#### A change in control includes:

- any person acquiring beneficial ownership of more than 50% of our total voting power;
- · the sale or other disposition of all or substantially all of our assets;
- our merger or consolidation after which our voting securities represent 50% or less of the total voting power of the surviving or acquiring entity; or
- the members of our board cease to constitute a majority of the members of our board over a period of 12 months, excluding any new members appointed or elected by the then incumbent board.

### Changes in Capitalization

In the event of certain changes in our capital structure without our receipt of consideration, such as a stock split, reverse stock split or dividend paid in Class A common stock, proportionate adjustments will automatically be made to:

- the maximum number and kind of shares available for issuance under our 2020 Plan, including the maximum number and kind of shares that may be issued upon the exercise of incentive stock options;
- the maximum number and kind of shares covered by, and exercise price, base price or purchase price, if any, applicable to each outstanding stock award.; and
- · the maximum number and kind of shares by which the share reserve may increase automatically each year.

In the event that there is a declaration of an extraordinary dividend payable in a form other than our Class A common stock in an amount that has a material effect on the price of our common stock, a recapitalization, a spin-off or a similar occurrence, the compensation committee may make such adjustments to any of the foregoing as it deems appropriate, in its sole discretion.

#### Amendments or Termination

Our board of directors may amend or terminate our 2020 Plan at any time. If our board of directors amends our 2020 Plan, it does not need stockholder approval of the amendment unless required by applicable law, regulation or rules. Our 2020 Plan will terminate automatically 10 years after the later of the date when our board of directors adopts our 2020 Plan or it approves a share increase that is also approved by our stockholders.

### 2010 Stock Plan

#### General

Our board of directors and stockholders adopted our 2010 2010 Plan in July 2010, which was subsequently amended from time to time.

As of September 30, 2020, we have reserved shares of our common stock for issuance under our 2010 Plan, there were outstanding options to purchase shares of common stock, at exercise prices ranging from \$ to \$ per share, or a weighted-average exercise price of \$ per share, shares of common stock issuable upon the vesting and settlement of RSUs, and shares of common stock remained available for future issuance. Unissued shares subject to awards that expire or are cancelled and shares reacquired by us will again become available for issuance under our 2010 Plan or, following consummation of this offering, under our 2020 Plan.

#### 2020 Employee Stock Purchase Plan

#### General

We expect that our board of directors will adopt a 2020 Employee Stock Purchase Plan (the "2020 ESPP") prior to this offering. If adopted, our 2020 ESPP will be subsequently approved by our stockholders. We expect that our 2020 ESPP will become effective as of the effective date of the registration statement of which this prospectus is a part. Our 2020 ESPP is intended to qualify under Section 423 of the Code. Although not yet adopted, we expect that our 2020 ESPP will have the features described below.

#### Share Reserve

shares of our Class A common stock will be reserved for issuance under our 2020 ESPP. The number of shares reserved for issuance under our 2020 ESPP will automatically be increased on the first business day of each of our fiscal years, commencing in 2021 and ending in 20 , by a number equal to the least of:

- shares:
- % of the shares of common stock outstanding on the last business day of the prior fiscal year; or
- · the number of shares determined by our board of directors.

The number of shares reserved under our 2020 ESPP will automatically be adjusted in the event of a stock split, stock dividend or a reverse stock split (including an adjustment to the per-purchase period share limit).

#### Administration

The compensation committee of our board of directors will administer our 2020 ESPP.

#### Eligibility

All of our employees will be eligible to participate if we employ them for more than hours per week and for more months per year. Eligible employees may begin participating in our 2020 ESPP at the start of any offering period.

## Offering Periods

Each offering period will last a number of months determined by the compensation committee, not to exceed 27 months. A new offering period will begin periodically, as determined by the compensation committee. Offering periods may overlap or may be consecutive. Unless otherwise determined by the compensation committee, offering periods of months' duration will begin in each year on and . However, if so determined by the compensation

committee, the first offering period may start on the effective date of the registration statement related to this offering and will end on , 2021, with the first purchase date occurring on , 2021.

#### Amount of Contributions

Our 2020 ESPP will permit each eligible employee to purchase common stock through payroll deductions. Each employee's payroll deductions may not exceed of the employee's cash compensation. Each participant may purchase up to the number of shares determined by our board of directors on any purchase date, not to exceed shares. The value of the shares purchased in any calendar year may not exceed \$25,000. Participants may withdraw their contributions at any time before stock is purchased.

#### Purchase Price

The price of each share of common stock purchased under our 2020 ESPP will not be less than 85% of the lower of the fair market value per share of common stock on the first day of the applicable offering period (or, in the case of the first offering period, the price at which one share of common stock is offered to the public in this offering) or the fair market value per share of common stock on the purchase date.

#### Other Provisions

Employees may end their participation in our 2020 ESPP at any time. Participation ends automatically upon termination of employment with us. If we experience a change in control, our 2020 ESPP will end and shares will be purchased with the payroll deductions accumulated to date by participating employees. Our board of directors or our compensation committee may amend or terminate our 2020 ESPP at any time.

#### **CERTAIN RELATIONSHIPS AND RELATED PERSON TRANSACTIONS**

In addition to the compensation arrangements with directors and executive officers described under "Executive Compensation" and "Management" and the registration rights described elsewhere in this prospectus, the following is a description of each transaction since January 1, 2017 and each currently proposed transaction in which:

- · we have been or are to be a participant;
- · the amount involved exceeds or will exceed \$120,000; and
- any of our directors, executive officers or beneficial holders of more than 5% of our capital stock, or any immediate family member
  of, or person sharing the household with, any of these individuals (other than tenants or employees), had or will have a direct or
  indirect material interest.

#### **Equity Financings**

Series H Redeemable Convertible Preferred Stock Financing

In March 2019, we entered into a closing of a Series H redeemable convertible preferred stock financing, pursuant to which we sold 943,548 shares of Series H redeemable convertible preferred stock at a purchase price of \$169.573 per share for aggregate proceeds of \$160 million. General Atlantic (WI), L.P. purchased 884,577 shares of Series H redeemable convertible preferred stock. Mr. Syed, a member of our board of directors, is a Managing Partner of General Atlantic LLC, which is an affiliate of General Atlantic (WI), L.P.

#### Series G Redeemable Convertible Preferred Stock Financing

From September 2017 until October 2017, we entered into various closings of a Series G redeemable convertible preferred financing, pursuant to which we sold 1,687,319 shares of Series G redeemable convertible preferred stock at a purchase price of \$134.533 per share for aggregate proceeds of approximately \$227 million. The following table summarizes purchases of our Series G redeemable convertible preferred stock by beneficial holders of more than 5% of our outstanding capital stock and entities managed by certain of our directors:

	Snares of Series G Redeemable Convertible	
Name of Stockholder	Preferred Stock (#)	Total Purchase Price (\$)
Entities affiliated with The Founders Fund <sup>(1)</sup>	37,166	5,000,052
DST Investments XVI, L.P.(2)	460,854	62,000,071
Entities affiliated with 8VC(3)	14,866	1,999,968

(1) Affiliates of The Founders Fund holding our securities, whose shares are aggregated for purposes of determining holders of 5% or more of our outstanding capital stock as of the Series G Stock redeemable convertible preferred stock financing and for purposes of reporting the above share ownership information, are, The Founders Fund V, LP, The Founders Fund V Principals Fund LP and FF Wish VI LLC

Founders Fund V Entrepreneurs Fund, LP, The Founders Fund V Principals Fund, LP and FF Wish VI, LLC.

(2) Affiliates of DST Investments holding our securities, whose shares are aggregated for purposes of determining holders of 5% or more of our outstanding capital stock as of the Series G redeemable convertible preferred stock financing, are, DST Global IV, L.P., DST Global IV Co-Invest, L.P., DST Global V, L.P., DST Investments XVI, L.P., and DST Team Fund Limited.

(3) Affiliates of 8VC holding our securities, whose shares are aggregated for purposes of determining holders of 5% or more of our outstanding capital stock as of the Series G redeemable convertible preferred stock financing are Formation8 Partners, 8VC Co-Invest Fund, I, L.P., F8 StarLight SPV, L.P, F8 StarLight SPV II, L.P., Anduin I, L.P. and CL SPV. Mr. Lonsdale, a member of our board of directors holds ultimate voting and investment power with regard to the securities held directly by the trusts and entities listed above.

#### Certain Transactions with our Chief Executive Officer, President, and Chairperson.

In June 2019, we repurchased 142,974 shares of common stock from Mr. Szulczewski for \$121.70 per share for an aggregate purchase price of approximately \$17.4 million.

In August 2020, Mr. Szulczewski repaid a loan we previously issued to him in the amount of approximately \$1.1 million to reimburse certain expenses incurred by the Company on behalf of Mr. Szulczewski.

In August 2020, we paid a fee in the amount of \$280,000 to the United States Treasury Department in connection with a change in the composition of our board of directors, which resulted in an increase in Mr. Szulczewski's voting rights on our board of directors, as required by the Hart-Scott-Rodino Antitrust Improvements Act of 1976.

#### Right of First Refusal and Repurchases

Pursuant to certain agreements with our stockholders, including our amended and restated first refusal and co-sale agreement dated originally dated May 11, 2012 and amended and restated on August 25, 2020, we or our assignees have a right to purchase shares of our capital stock that stockholders propose to sell to other parties. Since January 1, 2017, we have (i) waived our right of first refusal in connection with the sale of 651,303 shares of our Class B common stock in 15 separate transactions and (ii) exercised our right of first refusal in one transaction that involved the repurchase of 142,974 shares of our Class B common stock, that involved the sale or purchase or repurchase of such shares by our directors and officers, entities with which certain of our directors are affiliated and certain holders of more than 5% of our capital stock. Each of Messrs. Szulczewski, Lonsdale, Zhang, and Chuang, as well as 8VC Co-Invest Fund I, L.P. and FF Wish VI, LLC, were parties in one or more of these transactions.

#### **Amended and Restated Voting Agreement**

On March 18, 2019, we entered into an amended and restated voting agreement, with certain holders of our common stock and the holders of our redeemable convertible preferred stock, including our founder and chief executive officer, holders of more than 5% of our capital stock and entities with which certain of our directors are affiliated, with respect to the election of our directors and certain other matters. All of our current directors were elected pursuant to the terms of this agreement. The amended and restated voting agreement will terminate upon the completion of this offering.

#### Amended and Restated First Refusal and Co-Sale Agreement

On August 25, 2020, we entered into an amended and restated first refusal and co-sale agreement with holders of our common stock and redeemable convertible preferred stock, including our founder and chief executive officer, holders of more than 5% of our capital stock and entities with which certain of our directors are affiliated. This agreement provides the holders of redeemable convertible preferred stock a right of purchase and of co-sale in respect of sales of securities by certain holders of our common stock. The rights of purchase and co-sale will terminate upon the completion of this offering.

#### Amended and Restated Investors' Rights Agreement

On March 18, 2019, we entered into an amended and restated investors' rights agreement with holders of our redeemable convertible preferred stock, including holders of more than 5% of our capital stock and entities with which certain of our directors are affiliated. These stockholders are entitled to

rights with respect to the registration of their shares following this offering under the Securities Act. For a description of these registration rights, see "Description of Capital Stock—Registration Rights."

#### **Management Rights Letters**

In connection with our sale of our redeemable convertible preferred stock, we entered into management rights letters with certain purchasers of our redeemable convertible preferred stock, including holders of more than 5% of our capital stock and entities with which certain of our directors are affiliated, pursuant to which such entities were granted certain management rights, including the right to consult with and advise our management on significant business issues, review our operating plans, examine our books and records and inspect our facilities. These management rights will terminate upon completion of this offering.

#### **Indemnification Agreements**

Our amended and restated certificate of incorporation, which will be effective upon the completion of this offering, will contain provisions limiting the liability of directors, and our amended and restated bylaws, which will be effective upon the completion of this offering, will provide that we will indemnify each of our directors to the fullest extent permitted under Delaware law. Our certificate of incorporation and bylaws will also provide our board of directors with discretion to indemnify our officers and employees when determined appropriate by the board of directors.

We also intend to enter into indemnification agreements with each of our directors and executive officers. The indemnification agreements will provide that we will indemnify each such person against any and all expenses incurred by such person because of his or her status as one of our directors or executive officers, to the fullest extent permitted by Delaware law, our amended and restated certificate of incorporation and our amended and restated bylaws. In addition, the indemnification agreements will provide that, to the fullest extent permitted by Delaware law, we will advance all expenses incurred by our directors and executive officers in connection with a legal proceeding involving his or her status as a director, executive officer or employee.

#### **Policies and Procedures for Related Person Transactions**

Our audit committee has the primary responsibility for the review, approval and oversight of any "related person transaction," which is any transaction, arrangement, or relationship (or series of similar transactions, arrangements, or relationships) in which we are, were, or will be a participant and the amount involved exceeds \$120,000, and in which the related person had, has, or will have a direct or indirect material interest. We intend to adopt a written related person transaction policy to be effective upon the completion of this offering. Under our related person transaction policy, our management will be required to submit any related person transaction not previously approved or ratified by our audit committee to our audit committee. In approving or rejecting the proposed transactions, our audit committee will take into account all of the relevant facts and circumstances available. Our audit committee will approve only those transactions that, as determined by our audit committee, are in, or are not inconsistent with, our best interests and the best interests of our stockholders.

Although we have not had a written policy prior to this offering for the review and approval of transactions with related persons, our board of directors has historically reviewed and approved any transaction where a director or officer had a financial interest, including the transactions described above. Prior to approving such a transaction, the material facts as to a director's or officer's relationship or interest as to the agreement or transaction were disclosed to our board of directors. Our board of directors would take this information into account when evaluating the transaction and in determining whether such transaction was fair to us and in the best interest of all of our stockholders.

#### PRINCIPAL AND SELLING STOCKHOLDERS

The following table sets forth certain information with respect to the beneficial ownership of our common stock as of August 27, 2020, and as adjusted to reflect the sale of Class A common stock offered by us and the selling stockholders in this offering, for:

- each stockholder known by us to be the beneficial owner of more than 5% of our outstanding shares of Class A common stock or Class B common stock:
- · each of our directors;
- · each of our named executive officers:
- all of our directors and executive officers as a group; and
- · each selling stockholder.

We have determined beneficial ownership in accordance with the rules of the SEC. Except as indicated by the footnotes below, we believe, based on the information furnished to us, that the persons and entities named in the table below have sole voting and investment power with respect to all shares of Class A common stock or Class B common stock that they beneficially own, subject to applicable community property laws.

We have based our calculation of the percentage of beneficial ownership prior to this offering on shares of our Class A shares of our Class B common stock outstanding as of August 27, 2020, assuming the reclassification of all outstanding shares of our common stock into an equivalent number of shares of our Class B common stock, the automatic conversion and reclassification of our redeemable convertible preferred stock into shares of our Class A common stock, the exercise of an shares of our Series B redeemable convertible preferred stock outstanding as of August 27, 2020 outstanding warrant to purchase shares of our Class A common stock on a net-exercise basis (assuming an initial public offering price of \$ into per share, the midpoint of the price range set forth on the cover page of this prospectus), which we expect to occur immediately prior to the completion of this offering. For purposes of calculating the percentage of beneficial ownership prior to this offering, we did not include the effect of any voting agreements or voting proxies that terminate upon the offering. We have based our calculation of the percentage of shares of our Class B common stock beneficial ownership after this offering on shares of our Class A common stock and outstanding immediately after the completion of the offering and assuming no exercise by the underwriters of their option to purchase additional shares. We have deemed shares of our Class B common stock subject to stock options that are currently exercisable or exercisable within 60 days of August 27, 2020 or issuable pursuant to RSUs which are subject to vesting and settlement conditions expected to occur within 60 days of August 27, 2020 (assuming the satisfaction of the performance-based vesting condition) to be outstanding and to be beneficially owned by the person holding the stock option or RSU for the purpose of computing the percentage ownership of that person. We did not deem these shares outstanding, however, for the purpose of computing the percentage ownership of any other person.

Unless otherwise indicated, the address of each beneficial owner listed in the table below is One Sansome Street, 40th Floor, San Francisco, California 94104.

	Shares Beneficially Owned Prior to this Offering		% of Total Voting Power Before Our Initial	Number of Shares	Shares Beneficially Owned After this Offering				% of Total Voting Power After Our Initial		
	Class A		Class B		Public Offering <sup>(1)</sup>	Being Offered	Class		Class		Public Offering <sup>(1)</sup>
Name of Beneficial Owner Named Executive Officers and Directors:	Shares	<u>%</u>	Shares	%			Shares	<u>%</u>	<u>Shares</u>	<u>%</u>	
Peter Szulczewski <sup>(2)</sup>	_	_	10,904,956								
Ari Emanuel <sup>(3)</sup>	37,166		-	_							
Joe Lonsdale <sup>(4)</sup>	6,944,338		893,003								
Tanzeen Syed <sup>(5)</sup>	1,635,923		140,000								
Stephanie Tilenius	_	_	_	_							
Hans Tung <sup>(6)</sup>	3,380,576		469,660								
Rajat Bahri <sup>(7)</sup>	_	-	237,652								
Devang Shah <sup>(8)</sup>	_	-	35,299								
Thomas Chuang <sup>(9)</sup>	_	-	39,442								
Pai Liu <sup>(10)</sup>	_	-	3,194								
All executive officers and directors as a											
group (10 persons) <sup>(10)</sup>	11,998,003		12,723,206								
Other 5% Stockholders:											
Entities affiliated with DST Global (11)	10,719,509		_	_							
Entities affiliated with The Founders Fund <sup>(12)</sup>	6,247,712		_	-							
Formation8 Partners <sup>(13)</sup>	6,935,480		657,320								
Entities affiliated with GGV Capital (14)	3,380,576		469,660								
Republic Technologies <sup>(15)</sup>	2,683,488		_	_							
Sheng (Danny) Zhang <sup>(16)</sup>	_	_	2,683,373								

# Other Selling Stockholders: **Total Selling Stockholders**

<sup>(1)</sup> Percentage of total voting power represents voting power with respect to all shares of our Class A and Class B common stock, as a single class. The holders of our Class B

common stock are entitled to 20 votes per share, and holders of our Class A common stock are entitled to one vote per share. For more information about the voting rights of our Class A and Class B common stock, see "Description of Capital Stock—Common Stock."

Consists of (i) 5,651,025 shares of Class B common stock, (ii) 4,337,500 shares of Class B common stock issuable upon exercise of options exercisable within 60 days of August 27, 2020 and (iii) 916,431 shares of Class B common stock issuable upon vesting and settlement of restricted stock units within 60 days of August 27, 2020.

Mr. Szulczewski also holds 379,895 restricted stock units which are subject to vesting conditions not expected to occur within 60 days of August 27, 2020.

Consists of 37,166 shares of Class A common stock held by Mr. Emanuel.

Consists of (i) 70,570 shares of Class B common stock and 25,000 shares of Class B common stock issuable upon exercise of options exercisable within 60 days of August 27, 2020 held by Mr. Lonsdale, (ii) 886 shares of Class A common stock and 109,227 shares of Class B common stock held by the Children's Trust Created Under The Brunswick Liberty Trust Dated February 15, 2018, (iii) 3,986 shares of Class A common stock and 15,443 shares of Class B common stock held by the Trust for Jonathan Lonsdale created under The Brunswick Liberty Trust dated February 15, 2018, (iv) 3,986 shares of Class A common stock and 15,443 shares of Class B common stock held by the Trust for Jonathan Lonsdale created under The Brunswick Liberty Trust dated February 15, 2018, (v) 5,640,750 shares of Class A common stock, 580,063 shares of Class B common stock and 986,640 shares of Class B common stock issuable upon exercise of a Series B redeemable

convertible preferred stock warrant held by Formation8 Partners, (vi) 204,996 shares of Class A common stock and 77,257 shares of Class B common stock held by 8VC Co-Invest Fund, I, L.P., (vii) 539,955 shares of Class A common stock held by F8 StarLight SPV, L.P., (viii) 147,094 shares of Class A common stock held by F8 StarLight SPV, I.P., (viii) 147,094 shares of Class A common stock held by F8 StarLight SPV II.P., (viii) 147,094 shares of Class A common stock held by F8 StarLight SPV II.P., (viii) 147,094 shares of Class A common stock held by F8 StarLight SPV II.P., (viii) 147,094 shares of Class A common stock held by F8 StarLight SPV II.P., (viii) 147,094 shares of Class A common stock held by F8 StarLight SPV II.P., (viii) 147,094 shares of Class A common stock held by F8 StarLight SPV II.P., (viii) 147,094 shares of Class A common stock held by F8 StarLight SPV II.P., (viii) 147,094 shares of Class A common stock held by F8 StarLight SPV II.P., (viii) 147,094 shares of Class A common stock held by F8 StarLight SPV II.P., (viii) 147,094 shares of Class A common stock held by F8 StarLight SPV II.P., (viii) 147,094 shares of Class A common stock held by F8 StarLight SPV II.P., (viii) 147,094 shares of Class A common stock held by F8 StarLight SPV II.P., (viii) 147,094 shares of Class A common stock held by F8 StarLight SPV II.P., (viii) 147,094 shares of Class A common stock held by F8 StarLight SPV II.P., (viii) 147,094 shares of Class A common stock held by F8 StarLight SPV II.P., (viii) 147,094 shares of Class A common stock held by F8 StarLight SPV II.P., (viii) 147,094 shares of Class A common stock held by F8 StarLight SPV II.P., (viii) 147,094 shares of Class A common stock held by F8 StarLight SPV II.P., (viii) 147,094 shares of Class A common stock held by F8 StarLight SPV II.P., (viii) 147,094 shares of Class A common stock held by F8 StarLight SPV II.P., (viii) 147,094 shares of Class A common stock held by F8 StarLight SPV II.P., (viii) 147,094 shares of Class A common stock held by F8 StarLight SPV II.P., (viii) 147,094 shares of Class A common stock held by F8 StarLight SPV II.P., (viii) 147,094 shares of Class A common stock held by F8 StarLight SPV I board of directors holds ultimate voting and investment power with regard to the securities held directly by the trusts and entities listed above. The address for Mr. Lonsdale is c/o 8VC, Pier 5, Suite 101, San Francisco, CA 94111.

- (5)
- 8VC, Pier 5, Suite 101, San Francisco, CA 94111.

  Consists of (i) 140,000 shares of Class B common stock and (ii) 1,635,923 shares of Class A common stock held by General Atlantic (WI), L.P. ("GA WI"). The limited partners that share beneficial ownership of the shares held by GA WI are the following General Atlantic investment funds (the "GA Funds"): General Atlantic Partners 100, L.P. ("GAP 100"), General Atlantic Partners (Bermuda) EU, L.P. ("GAP EU"), General Atlantic Partners (Lux) SCSp ("GAP Lux"), GAP Coinvestments III, LLC ("GAPCO III"), GAP Coinvestments IV, LLC ("GAPCO IV") and GAP Coinvestments CDA, L.P. ("GAPCO CDA").

  The general partner of GA WI is General Atlantic (Sev) GP, LLC ("GA SPV"). The general partner of GAP Lux is General Atlantic GenPar Lux") and the general partner of GA GenPar Lux is General Atlantic (Lux) S.à r.I. ("GA Lux"). The general partner of GAP Bermuda EU and GA Lux is General Atlantic GenPar (Bermuda), L.P. ("GenPar Bermuda"). GAP (Bermuda) Limited ("GAP (Bermuda) Limited") is the general partner of GenPar Bermuda. The general partner of GAP 100 is General Atlantic GenPar, L.P. ("GA GAPCO CDA and is the sole member of GAP SPV. There are eight members of the management committee of GAP LIC (the "GAP Management). V, the general partner of GAPCO CDA and is the sole member of GA SPV. There are eight members of the management committee of GA LLC (the "GA Management Committee"). The members of the GA Management Committee are also the members of the management committee of GAP (Bermuda) Limited. GA LLC, GA GenPar, GAP Committee ). The members of the GA Management Committee are also the members of the management committee of GAP (Bermuda) Limited. GA LLC, GA GenPar Lux, GA SPV and the GA Funds (collectively, the "GA Group") are a "group" within the meaning of Rule 13d-5 of the Securities Exchange Act of 1934, as amended. The mailing address of the foregoing General Atlantic entities (other than GAP Bermuda EU, GAP Lux, GA GenPar Lux, GA Lux, GenPar Bermuda and GAP (Bermuda) Limited) is c/o General Atlantic Service Company, L.P., 55 East 52nd Street, 33rd Floor, New York, NY 10055. The mailing address of GAP Bermuda EU, GenPar Bermuda, and GAP (Bermuda) Limited is Clarendon House, 2 Church Street, Hamilton HM 11, Bermuda. The mailing address for GAP Lux, GA GenPar Lux and GA Lux is Luxembourg is 412F, Route d'Esch, L-2086 Luxembourg.

  Each of the members of the GA Management Committee disclaims ownership of the shares except to the extent that he has a pecuniary interest therein. Mr. Syed is a Managing

- Director of GA LLC and serves as a director of the Company. Mr. Syed disclaims ownership of all such shares except to the extent that he has a pecuniary interest therein. Consists of (i) 56,580 shares of Class A common stock and 9,752 shares of Class B common stock held by GGV Capital IV Entrepreneurs Fund L.P., (ii) 2,668,467 shares of Class A common stock and 459,908 shares of Class B common stock held by GGV Capital IV L.P. and (iii) 655,529 shares of Class A common stock held by GGV Capital Select (6) L.P. GGV Capital IV L.L.C. is the general partner of GGV Capital IV Enterpreneurs Fund L.P. and GGV Capital IV L.L.C. and GGV Capital Select L.P. is GGV Capital Select L.L.C. and GGV Capital Select L.L.C. and therefore, may be deemed to share voting and investment power with regard to the securities held directly such entities. The address for Mr. Tung is c/o GGV Capital, 3000 Sand Hill Road, Building 4, Suite 230, Menlo Park, California 94025.
- Consists of 237,652 shares of Class B common stock issuable upon vesting and settlement of restricted stock units within 60 days of August 27, 2020. Mr. Bahri also holds 200,545 restricted stock units which are subject to vesting conditions not expected to occur within 60 days of August 27, 2020. (7)
- (8)
- 200,343 restricted stock units which are subject to vesting conditions not expected to occur within 60 days of August 27, 2020. Mr. Shah also holds 45,657 restricted stock units which are subject to vesting conditions not expected to occur within 60 days of August 27, 2020. Mr. Shah also holds 45,657 restricted stock units which are subject to vesting conditions not expected to occur within 60 days of August 27, 2020. Consists of 27,900 shares of Class B common stock issuable upon exercise of options exercisable within 60 days of August 27, 2020 and 11,542 shares of Class B common stock issuable upon vesting and settlement of restricted stock units which are subject to vesting conditions not expected to occur within 60 days of August 27, 2020.
- Consists of 3,194 shares of Class B common stock issuable upon vesting and settlement of restricted stock units within 60 days of August 27, 2020. Mr. Liu also holds 8,600 (10)
- restricted stock units which are subject to vesting conditions not expected to occur within 60 days of August 27, 2020.

  Consists of (i) 9,263,488 shares of Class A common stock beneficially owned by our directors and named executive officers, (ii) 6,780,249 shares of Class B common stock (11)beneficially owned by our directors and named executive officers and (iii) 5,605,547 shares of Class B common stock issuable to our directors and named executive officers upon exercise of outstanding stock options and vesting and settlement of RSUs exercisable or issuable within 60 days of August 27, 2020.

  Represents (i) 3,440,340 shares of Class A common stock held by DST Global IV, L.P., (iii) 647,825 shares of Class A common stock held by DST Global IV Co-Invest, L.P., (iii)
- (11)Class A common stock held by DST Team Fund Limited. We refer to DST

- Global IV, L.P., DST Global IV Co-Invest, L.P., DST Global V, L.P., DST Investments XI, L.P., DST Investments XV, L.P., DST Investments XVI, L.P. and DST Team Fund Limited as DST Global. Yuri Milner holds ultimate voting and investment power over the shares held by the DST Global. The address for DST Global is c/o Tulloch & Co., 4 Hill Street, London W1J 5NE, United Kingdom.
- Represents of (i) 66,151 shares of Class A common stock held of record by The Founders Fund V Entrepreneurs Fund, LP (FF-VE), (ii) 1,257,282 shares of Class A common stock held of record by The Founders Fund V Principals Fund, LP (FF-VP), (iii) 4,673,907 shares of Class A common stock held of record by The Founders Fund V, LP (FF-V) and (iv) 70,254 shares of Class B common stock and 180,118 shares of Class A common stock held of record by FF Wish VI, LLC (FF-Wish). The Founders Fund V Management, LLC (FF-VIM) is the general partner of each of FF-VE, FF-VP, and FF-V. The Founders Fund VI Management, LLC (FF-VIM) is the managing member of FF-VIM and FF-VIM are Peter Thiel and Brian Singerman. As a result, and by virtue of the relationships described in this footnote, each of the managing members of FF-VM and FF-VIM may be deemed to share beneficial ownership of the shares held by FF-VE, FF-VP, FF-V, and FF-Wish. The address of each of the entities identified in this footnote is One Letterman Drive, Building D, 5th Floor, San Francisco, California 94129.
- managing members of FF-VM and FF-VIM may be deemed to share beneficial ownership of the shares held by FF-VE, FF-VP, FF-V, and FF-Wish. The address of each of the entitites identified in this footnote is One Letterman Drive, Building D, 5th Floor, San Francisco, California 94129.

  (13) Represents (i) 5,640,750 shares of Class A common stock, 580,063 shares of Class B common stock and 986,640 shares of Class B common stock issuable upon exercise of a Series B redeemable convertible preferred stock warrant held by Formation8 Partners, (ii) 204,996 shares of Class A common stock and 77,257 shares of Class B common stock held by 8VC Co-Invest Fund, I, L.P., (iii) 539,955 shares of Class A common stock held by F8 StarLight SPV, I.P., (v) 147,094 shares of Class A common stock held by F8 StarLight SPV II, L.P., (v) 147,094 shares of Class A common stock held by Anduin I, L.P. and (vi) 244,777 shares of Class A common stock held by E8 StarLight SPV II, L.P., (v) 147,094 shares of Class A common stock held by Anduin I, L.P. and (vi) 244,777 shares of Class A common stock held by E8 StarLight SPV II, L.P., (v) 147,094 shares of Class A common stock held by Co-Invest Funding III, L.P., (vi) 147,094 shares of Class A common stock held by E8 StarLight SPV II, L.P., (vi) 147,094 shares of Class A common stock held by E8 StarLight SPV II, L.P., (vi) 147,094 shares of Class A common stock held by E8 StarLight SPV II, L.P., (vi) 157,908 shares of Class A common stock held by E8 StarLight SPV II, L.P., (vi) 147,094 shares of Class A common stock held by E8 StarLight SPV II, L.P., (vi) 147,094 shares of Class A common stock held by E8 StarLight SPV II, L.P., (vii) 157,908 shares of Class A common stock held by E8 StarLight SPV II, L.P., (vii) 157,908 shares of Class A common stock held by E8 StarLight SPV, L.P., (vii) 157,908 shares of Class A common stock held by E8 StarLight SPV, L.P., (vii) 157,908 shares of Class A common stock held by E8 StarLight SPV, L.P., (vii) 157,908 shares of Class A common stock held by E8 StarLight
- (14) Consists of (i) 56,580 shares of Class A common stock and 9,752 shares of Class B common stock held by GGV Capital IV Entrepreneurs Fund L.P., (iii) 2,668,467 shares of Class A common stock and 459,908 shares of Class B common stock held by GGV Capital IV L.P. and (ii) 655,529 shares of Class A common stock held by GGV Capital Select L.P. GGV Capital IV L.L.C. is the general partner of GGV Capital IV Entrepreneurs Fund L.P. and GGV Capital IV L.P. The general partner of GGV Capital Select L.P. is GGV Capital Select L.L.C. Mr. Tung, a member of our board of directors, is a Managing Director of GGV Capital IV L.L.C. and GGV Capital Select L.L.C. and therefore, may be deemed to share voting and investment power with regard to the securities held directly such entities. Mr. Tung disclaims beneficial ownership of such shares except to the extent of his pecuniary interest in such shares. The address for Mr. Tung is c/o GGV Capital, 3000 Sand Hill Road, Building 4, Suite 230, Menlo Park, California 94025.
- (15) Represents 2,683,488 shares of Class A common stock held by Republic Technologies Pte. Ltd. The address for Republic Technologies Pte. Ltd. is 60B Orchard Road, #06-18 Tower 2, TheAtrium@Orchard, Singapore 233891.
- (16) Consists of (i) 76,052 shares of Class B common stock held by Mr. Zhang, (ii) 2,131,089 shares of Class A common stock issuable upon exercise of options exercisable within 60 days of August 27, 2020, (ii) 394,129 shares of Class B common stock issuable upon vesting and settlement of restricted stock units exercisable within 60 days of August 27, 2020 and (iii) 82,103 shares of Class B common stock held by Sheng Zhang, trustee of the ZLZ Trust. Mr. Zhang also holds 35,534 restricted stock units which are subject to vesting conditions not expected to occur within 60 days of August 27, 2020.

#### **DESCRIPTION OF CAPITAL STOCK**

This section contains a description of our capital stock and the material provisions of our amended and restated certificate of incorporation and amended and restated bylaws that will be in effect upon the completion of this offering and is qualified by reference to the forms of our amended and restated certificate of incorporation and our amended and restated bylaws filed as exhibits to the registration statement relating to this prospectus, and by the applicable provisions of Delaware law.

#### General

Upon the completion of this offering, our amended and restated certificate of incorporation will authorize shares of undesignated preferred stock, the rights, preferences and privileges of which may be designated from time to time by our board of directors.

Upon the completion of this offering, our authorized capital stock will consist of shares, all with a par value of \$0.0001 per share, of which:

- shares are designated as Class A common stock;
- shares are designated as Class B common stock; and
- shares are designated as preferred stock.

As of September 30, 2020, and after giving effect to the automatic conversion of all of our outstanding redeemable convertible preferred stock into Class A common stock in connection with this offering and the reclassification of all outstanding shares of our common stock into an equivalent number of shares of our Class B common stock, as if such reclassification had occurred immediately prior to the completion of this offering, there were outstanding:

- shares of our Class A common stock held of record by stockholders;
- shares are designated as Class B common stock held of record by stockholders;
- shares of our Class B common stock issuable upon exercise of outstanding stock options;
- shares of our Class B common stock subject to outstanding RSUs;
- a warrant to purchase shares of our Class A common stock; and
- a warrant to purchase shares of our Class B common stock.

#### **Common Stock**

We have two classes of authorized common stock: Class A common stock and Class B common stock. The rights of the holders of Class A common stock and Class B common stock are identical, except with respect to voting, conversion and transfer rights.

#### **Dividend Rights**

Subject to preferences that may apply to shares of preferred stock outstanding at the time, the holders of outstanding shares of our common stock are entitled to receive dividends out of funds legally available if our board of directors, in its discretion, determines to issue dividends and only then at the times and in the amounts that our board of directors may determine. Under Delaware law, we can only pay dividends either out of "surplus" or out of the current or the immediately preceding year's

net profits. Surplus is defined as the excess, if any, at any given time, of the total assets of a corporation over its total liabilities and statutory capital. The value of a corporation's assets can be measured in a number of ways and may not necessarily equal their book value. See "Dividend Policy" for more information.

#### **Voting Rights**

The holders of our Class B common stock are entitled to 20 votes per share, and holders of our Class A common stock are entitled to one vote per share. The holders of our Class A common stock and Class B common stock vote together as a single class, unless otherwise required by law. Delaware law or our amended and restated certificate of incorporation could require either holders of our Class A common stock or our Class B common stock to vote separately as a single class in the following circumstances:

- if we were to seek to amend our amended and restated certificate of incorporation to increase the authorized number of shares of a class of stock, or to increase or decrease the par value of a class of stock, then that class would be required to vote separately to approve the proposed amendment; and
- if we were to seek to amend our amended and restated certificate of incorporation in a manner that alters or changes the powers, preferences or special rights of a class of stock in a manner that affected its holders adversely, then that class would be required to vote separately to approve the proposed amendment.

The holders of common stock will not have cumulative voting rights in the election of directors. Accordingly, holders of a majority of the voting power of the shares of common stock entitled to vote in any election of directors may elect all of the directors standing for election. Except for the election of directors, if a quorum is present, an action on a matter is approved if it receives the affirmative vote of the holders of a majority of the voting power of the shares of capital stock present in person or represented by proxy at the meeting and entitled to vote on the matter, unless otherwise required by applicable law, the DGCL, our amended and restated certificate of incorporation or our amended and restated bylaws. The election of directors will be determined by a plurality of the votes cast in respect of the shares present in person or represented by proxy at the meeting and entitled to vote, meaning that the nominees with the greatest number of votes cast, even if less than a majority, will be elected. The rights, preferences and privileges of holders of common stock are subject to, and may be impacted by, the rights of the holders of shares of any series of preferred stock that we may designate and issue in the future.

#### No Preemptive or Similar Rights

Except for the conversion provisions with respect to our Class B common stock described below, holders of our common stock have no preemptive or conversion rights or other subscription rights, and there are no redemption or sinking fund provisions applicable to our common stock.

#### Right to Receive Liquidation Distributions

Upon our dissolution, liquidation or winding-up, the assets legally available for distribution to our stockholders are distributable ratably among the holders of our common stock, subject to prior satisfaction of all outstanding debt and liabilities and the preferential rights and payment of liquidation preferences, if any, on any outstanding shares of preferred stock.

#### Conversion of our Class B Common Stock

Each share of our Class B common stock is convertible at any time at the option of the holder into one share of our Class A common stock. Each share of our Class B common stock will convert

automatically into one share of our Class A common stock upon any transfer, whether or not for value, except certain transfers to entities, including certain charities and foundations, to the extent the transferor retains sole dispositive power and exclusive voting control with respect to the shares of Class B common stock, and certain other transfers described in our amended and restated certificate of incorporation. Upon the death or permanent incapacity of each holder of Class B common stock who is a natural person, the Class B common stock held by that person or his or her permitted estate planning entities will convert automatically into Class A common stock. However, shares of Class B common stock held by Mr. Szulczewski or his permitted estate planning entities or other permitted transferees will not convert automatically into Class A common stock until a time that is between 90 and 270 days after his death or permanent incapacity, as determined by the board of directors.

In addition, all shares of Class B common stock will automatically convert into shares of Class A common stock on the earlier of (i) the 10-year anniversary of the closing date of this offering, (ii) the date on which the number of outstanding shares of Class B common stock represents less than 5% of the aggregate combined number of outstanding shares of Class A common stock and Class B common stock, (iii) the date specified by a vote of the holders of a majority of the then outstanding shares of Class B common stock and (iv) a date that is between 90 and 270 days, as determined by the board of directors, after the death or permanent incapacity of Mr. Szulczewski.

Once transferred and converted into Class A common stock, the Class B common stock will not be reissued.

#### **Preferred Stock**

Upon the completion of this offering, no shares of preferred stock will be outstanding, but we will be authorized, subject to limitations prescribed by Delaware law, to issue preferred stock in one or more series, to establish from time to time the number of shares to be included in each series, and to fix the designation, powers, preferences and rights of the shares of each series and any of its qualifications, limitations or restrictions. Our board of directors also can increase or decrease the number of shares of any series, but not below the number of shares of that series then outstanding, without any further vote or action by our stockholders. Our board of directors may authorize the issuance of preferred stock with voting or conversion rights that could adversely affect the voting power or other rights of the holders of the common stock. The issuance of preferred stock, while providing flexibility in connection with possible acquisitions and other corporate purposes, could, among other things, have the effect of delaying, deferring or preventing a change in control of our company and may adversely affect the market price of our common stock and the voting and other rights of the holders of common stock. We have no current plan to issue any shares of preferred stock.

#### **Options**

As of September 30, 2020, we had options to purchase shares of our Class B common stock outstanding under our 2010 Stock Plan with a weighted-average price per share of \$

#### **RSUs**

As of September 30, 2020, we had shares of Class B common stock subject to RSUs outstanding pursuant to our 2010 Stock Plan. Subsequent to September 30, 2020, we awarded RSUs under our 2010 Stock Plan.

#### **Warrants**

As of September 30, 2020, we had outstanding an immediately exercisable warrant to purchase shares of our Series B redeemable convertible preferred stock at an exercise price of

\$0.0001 per share, (the "Series B Warrant"). The Series B Warrant is only exercisable one day prior to the earliest to occur of (a) the consummation of our first public offering (including this offering) pursuant to an effective registration statement under the Securities Act, or (b) the consummation of a Liquidation Event as defined in our currently existing restated certificate of incorporation. The warrant is subject to a cashless exercise mechanism

Additionally, as of September 30, 2020, we had an outstanding immediately exercisable warrant to purchase shares of our Class B common stock at an exercise price of \$ per share. The warrant is subject to a cashless exercise mechanism.

### **Registration Rights**

Following the completion of this offering, holders of an aggregate of shares of our common stock will have registration rights. These shares are referred to as registrable securities. The holders of these registrable securities possess registration rights pursuant to the terms of our amended and restated investors' rights agreement dated March 18, 2019, as amended (the "investors' rights agreement"), which terms are described in additional detail below.

#### **Demand Registration Rights**

Under our investors' rights agreement, at any time commencing on the earlier of (i) June 9, 2021 and (ii) the date that is 180 days following the effective date of our first registration statement, upon the written request of the holders of not less than 50% of the registrable securities then outstanding that we file a registration statement under the Securities Act with an anticipated aggregate price to the public of at least \$15 million, we will be obligated to use our commercially reasonable efforts to register the sale of all registrable securities that holders may request in writing to be registered within 20 days of the mailing of a notice by us to all holders of such registration. We are required to effect no more than two registration statements that are declared or ordered effective, subject to certain exceptions. We may postpone the filling of a registration statement for up to 90 days no more than once in any 12-month period if in the good faith judgment of our board of directors such registration would be seriously detrimental to us, and we do not file another registration statement on our account or that of any other stockholder during such 90 day period.

### Piggyback Registration Rights

If we register any of our securities for public sale, we will be obligated to use all commercially reasonable efforts to register all registrable securities that the holders of such securities request in writing be registered within 20 days of mailing of notice by us to all holders of the proposed registration. However, this right does not apply to a registration relating solely to employee benefit plans, a registration relating to the offer and sale of debt securities or a registration relating to a corporate reorganization or other transaction under Rule 145 of the Securities Act. The managing underwriter of any underwritten offering will have the right to limit, due to marketing reasons, the number of shares registered by these holders to 30% of the total shares covered by the registration statement, except for in this offering, in which these holders may be excluded entirely if the underwriters determine that the sale of their shares may jeopardize the success of the offering.

#### Form S-3 Registration Rights

At any time commencing on the date that is 180 days following the effective date of our first registration statement, the holders of the registrable securities can request that we register all or a portion of their shares on Form S-3 if we are eligible to file a registration statement on Form S-3 and the aggregate price to the public of the shares offered is at least \$5,000,000. We are required to file no

more than one registration statement on Form S-3 per 12-month period upon exercise of these rights, subject to certain exceptions. We may postpone the filing of a registration statement for up to 90 days once in any 12-month period if in the good faith judgment of our board of directors such registration would be seriously detrimental to us, and we do not register any other securities for our account or the account of any other stockholder during such 90-day period.

Additionally, we are required, once we become eligible to register securities on Form S-3, to use commercially reasonable efforts to qualify the registrable securities for registration on a delayed or continues basis on Form S-3 pursuant to Rule 415 under the Securities Act. Holders of registrable securities may, no more than twice in a 12-month period, elect to sell registrable securities pursuant to such registration on a delayed or continuous basis, including up to once in a 12-month period through an underwritten offering.

#### Registration Expenses

We will pay all expenses (other than underwriting discounts, selling commissions and stock transfer taxes) of the holders incurred in connection with each of the registrations described above, subject to certain limitations. However, we will not pay for any expenses of any demand or Form S-3 registration if the request is subsequently withdrawn at the request of the holders of a majority of the registrable securities to be registered, subject to limited exceptions.

#### Termination of Registration Rights

The registration rights described above will survive this offering and will terminate upon a liquidation event or as to any stockholder at such time as all of such stockholder's securities (together with any affiliate of the stockholder with whom such stockholder must aggregate its sales) could be sold pursuant to Rule 144 of the Securities Act, but in any event no later than the third-year anniversary of this offering.

#### **Anti-Takeover Provisions**

#### Section 203 of the Delaware General Corporation Law

Upon the completion of this offering, we will be governed by the provisions of Section 203 of the Delaware General Corporation Law regulating corporate takeovers. This section prevents some Delaware corporations from engaging, under some circumstances, in a business combination, which includes a merger or sale of at least 10% of the corporation's assets with any interested stockholder, meaning a stockholder who, together with affiliates and associates, owns or, within three years prior to the determination of interested stockholder status, did own 15% or more of the corporation's outstanding voting stock, unless:

- the transaction is approved by the board of directors prior to the time that the interested stockholder became an interested stockholder; or
- subsequent to such time that the stockholder became an interested stockholder the business combination is approved by the board
  of directors and authorized at an annual or special meeting of stockholders by at least two-thirds of the outstanding voting stock
  which is not owned by the interested stockholder.

A Delaware corporation may "opt out" of these provisions with an express provision in its original certificate of incorporation or an express provision in its certificate of incorporation or amended and restated bylaws resulting from a stockholders' amendment approved by a majority of the outstanding voting shares. We have not opted out of these provisions. As a result, mergers or other takeover or change in control attempts of us may be discouraged or prevented.

#### Certificate of Incorporation and Bylaw Provisions

Upon the completion of this offering, our amended and restated certificate of incorporation and our amended and restated bylaws will include a number of provisions that may have the effect of deterring hostile takeovers or delaying or preventing changes in control of our management team, including the following:

- Dual Class Stock. As described above in "—Common Stock—Voting Rights," our amended and restated certificate of incorporation will provide for a dual class common stock structure pursuant to which holders of our Class B common stock will have the ability to control the outcome of matters requiring stockholder approval, even if they own significantly less than a majority of the shares of our outstanding Class A and Class B common stock, including the election of directors and significant corporate transactions, such as a merger or other sale of our company or its assets. Current investors, executives, and employees will have the ability to exercise significant influence over those matters.
- Separate Class B Vote for Certain Transactions. Any transaction that would result in a change in control of our company will require the approval of a majority of our outstanding Class B common stock voting as a separate class. This provision could delay or prevent the approval of a change in control that might otherwise be approved by a majority of outstanding shares of our Class A and Class B common stock voting together on a combined basis.
- Supermajority Approvals. Our amended and restated certificate of incorporation and amended and restated bylaws do provide that certain amendments to our amended and restated certificate of incorporation or amended and restated bylaws by stockholders will require the approval of two-thirds of the combined vote of our then-outstanding shares of Class A and Class B common stock. This will have the effect of making it more difficult to amend our amended and restated certificate of incorporation or amended and restated bylaws to remove or modify certain provisions.
- Board of Directors Vacancies. Our amended and restated certificate of incorporation and amended and restated bylaws authorize
  only our board of directors to fill vacant directorships. In addition, the number of directors constituting our board of directors is set
  only by resolution adopted by a majority vote of our entire board of directors. These provisions restricting the filling of vacancies will
  prevent a stockholder from increasing the size of our board of directors and gaining control of our board of directors by filling the
  resulting vacancies with its own nominees.
- Classified Board. Our board of directors will not initially be classified. When the outstanding shares of our Class B common stock
  represent less than a majority of the combined voting power of our common stock, our board of directors will be classified into three
  classes of directors with staggered three-year terms and directors will only be able to be removed from office for cause. The
  existence of a classified board could delay a successful tender offeror from obtaining majority control of our board of directors, and
  the prospect of that delay might deter a potential offeror.
- Stockholder Action; Special Meeting of Stockholders. Our restated certificate of incorporation provides that stockholders will be able
  to take action by written consent. When the outstanding shares of our Class B common stock represent less than a majority of the
  combined voting power of our common stock, our stockholders will no longer be able to take action by written consent, and will only
  be able to take action at annual or special meetings of our stockholders. Stockholders will not be permitted to cumulate their votes
  for the election of directors. Our restated bylaws further provide that special meetings of our stockholders may be called by a
  majority of our board of directors, the chairperson of our board of directors, our chief executive officer or our president and a majority
  of the outstanding common stock. When the outstanding

shares of our Class B common stock represent less than a majority of the combined voting power of our common stock, our stockholders will not be able to call a special meeting.

- Advance Notice Requirements for Stockholder Proposals and Director Nominations. Our amended and restated bylaws will provide
  advance notice procedures for stockholders seeking to bring business before our annual meeting of stockholders or to nominate
  candidates for election as directors at any meeting of stockholders. Our amended and restated bylaws will also specify certain
  requirements regarding the form and content of a stockholder's notice. These provisions may preclude our stockholders from
  bringing matters before our annual meeting of stockholders or from making nominations for directors at our meetings of
  stockholders.
- Issuance of Undesignated Preferred Stock. Our board of directors will have the authority, without further action by the holders of common stock, to issue up to shares of undesignated preferred stock with rights and preferences, including voting rights, designated from time to time by the board of directors. The existence of authorized but unissued shares of preferred stock will enable our board of directors to render more difficult or to discourage an attempt to obtain control of us by means of a merger, tender offer, proxy contest or otherwise.

#### Choice of Forum

Upon the completion of this offering, our amended and restated certificate of incorporation will provide that the Court of Chancery of the State of Delaware will be the exclusive forum for any derivative action or proceeding brought on our behalf, any action asserting a breach of fiduciary duty, any action asserting a claim against us arising pursuant to the Delaware General Corporation Law, our amended and restated certificate of incorporation or our amended and restated bylaws or any action asserting a claim against us that is governed by the internal affairs doctrine. Our certificate of incorporation will also provide that the U.S. federal district courts will be the exclusive forum for resolving any complaint asserting a cause of action arising under the Securities Act. Some companies that adopted a similar federal district court forum selection provision were subject to a suit in the Chancery Court of Delaware by stockholders who asserted that the provision is not enforceable. While the Delaware Supreme Court held that such federal district court forum selection provision was in fact valid, there can be no assurance that federal courts or other state courts will follow the holding of the Delaware Supreme Court or determine that the our federal district court forum selection provision should be enforced in a particular case.

These choice of forum provisions do not apply to actions brought to enforce a duty or liability created by the Exchange Act. We intend for the choice of forum provision regarding claims arising under the Securities Act to apply despite the fact that Section 22 of the Securities Act creates concurrent jurisdiction for federal and state courts over all actions brought to enforce any duty or liability created by the Securities Act or the rules and regulations thereunder. There is uncertainty as to whether a court would enforce such provision with respect to claims under the Securities Act, and our stockholders will not be deemed to have waived our compliance with the federal securities laws and the rules and regulations thereunder.

#### **Transfer Agent and Registrar**

Upon the completion of this offering, the transfer agent and registrar for our Class A common stock will be agent's address is address is , and its telephone number is . The transfer

#### Listing

We intend to apply to have our Class A common stock listed on under the symbol "..."

#### SHARES ELIGIBLE FOR FUTURE SALE

Before this offering, there has not been a public market for shares of our Class A common stock. Future sales of substantial amounts of shares of our Class A common stock, including shares issued upon the settlement of RSUs and exercise of outstanding options, in the public market following this offering or the possibility of these sales occurring, could cause the prevailing market price for our Class A common stock to fall or impair our ability to raise equity capital in the future.

Following this offering, we will have outstanding shares of our Class A common stock and shares of our Class B common stock, based on the number of shares outstanding as of September 30, 2020. Of these outstanding shares, all of the shares of Class A common stock sold in this offering will be freely tradable, except that any shares purchased in this offering by our affiliates, as that term is defined in Rule 144 under the Securities Act, can only be sold in compliance with Rule 144. Shares of our Class B common stock are convertible into an equivalent number of shares of our Class A common stock and generally convert into shares of our Class A common stock upon transfer. In addition, we expect to issue shares of our Class B common stock upon the settlement of RSUs on March 14, 2020.

The remaining shares of common stock that are not sold in this offering will be "restricted securities," as that term is defined in Rule 144 under the Securities Act. These restricted securities are eligible for public sale only if they are registered under the Securities Act or if they qualify for an exemption from registration under Rule 144 or Rule 701 under the Securities Act, which are summarized below.

In addition, substantially all of our security holders have entered into market standoff agreements with us or lock-up agreements with the underwriters under which they have agreed, subject to specific exceptions, not to sell any of our capital stock until at least 181 days after the date of this prospectus, as described below. As a result of these agreements and the provisions of our investors' rights agreement described above under "Description of Capital Stock—Registration Rights", subject to the provisions of Rule 144 or Rule 701, shares will be available for sale in the public market as follows:

- beginning on the date of this prospectus, the shares of Class A common stock sold in this offering will be immediately available for sale in the public market;
- beginning 181 days after the date of this prospectus, up to an aggregate of in the public market, of which as described below; and
   additional shares will become eligible for sale shares will be held by affiliates and subject to the volume and other restrictions of Rule 144, as described below; and
- the remainder of the shares will be eligible for sale in the public market from time to time thereafter, subject in some cases to the volume and other restrictions of Rule 144, as described below.

Of the shares of our Class B common stock that were subject to stock options outstanding as of September 30, 2020, options to purchase shares of Class B common stock were vested as of September 30, 2020 and the Class B common stock underlying such options will be eligible for sale approximately six months after the date of this prospectus. We expect an additional shares of Class B common stock to be delivered upon the settlement of RSUs between the date that is approximately six months after the date of this prospectus and 2020, which shares would be eligible for sale in the public market immediately following settlement.

#### **Lock-Up Agreements**

Our officers, directors and holders of substantially all of our outstanding securities have agreed with the underwriters not to, among other things, dispose of any of our common stock or securities

convertible into or exchangeable for shares of our common stock during the 180-day period following the date of this prospectus, except with the prior written consent of Goldman Sachs & Co. LLC, subject to certain exceptions. In addition, substantially all other holders of our common stock, RSUs, options and warrants have previously entered into market stand-off agreements with us not to sell or otherwise transfer any of their common stock or securities convertible into or exchangeable for shares of common stock for a period that extends until 181 days after the date of this prospectus.

#### **Rule 144**

In general, under Rule 144 as currently in effect, a person who has beneficially owned shares of our restricted common stock for at least six months would be entitled to sell their securities provided that such person is not deemed to have been one of our affiliates at the time of, or at any time during the 90 days preceding, a sale, and we are subject to the periodic reporting requirements of the Exchange Act, for at least 90 days before the sale. In addition, under Rule 144, any person who is not an affiliate of ours and has held their shares for at least one year, including the holding period of any prior owner other than one of our affiliates, would be entitled to sell an unlimited number of shares immediately upon the completion of this offering without regard to whether current public information about us is available. Persons who have beneficially owned shares of our restricted common stock for at least six months but who are our affiliates at the time of, or any time during the 90 days preceding, a sale, would be subject to additional restrictions, by which such person would be entitled to sell within any three-month period only a number of securities that does not exceed the greater of either of the following:

- 1% of the number of common shares then outstanding, which will equal approximately shares immediately after this offering assuming no exercise of the underwriters' option to purchase additional shares, based on the number of common shares outstanding as of September 30, 2020; or
- the average weekly trading volume of our common shares during the four calendar weeks preceding the filing of a notice on Form 144 with respect to the sale;

provided, in each case, that we are subject to the Exchange Act periodic reporting requirements for at least 90 days, before the sale. Such sales both by affiliates and by non-affiliates must also comply with the manner of sale, current public information and notice provisions of Rule 144.

#### **Rule 701**

Any of our service providers who purchased shares under a written compensatory plan or contract prior to this offering may be entitled to rely on the resale provisions of Rule 701. Rule 701, as currently in effect, permits resales of shares, including by affiliates, in reliance upon Rule 144 but without compliance with certain restrictions, including the holding period requirement, of Rule 144. Rule 701 further provides that non-affiliates may sell such shares in reliance on Rule 144 without having to comply with the public information, volume limitation or notice provisions of Rule 144. All holders of Rule 701 shares are required to wait until 90 days after the date of this prospectus before selling such shares if such resale is done under Rule 701. All Rule 701 shares are, however, subject to lock-up agreements and will only become eligible for sale upon the expiration of these lock-up agreements.

#### **Registration Rights**

Upon completion of this offering, the holders of shares of our common will be entitled to rights with respect to the registration of the sale of common stock under the Securities Act. See "Description of Capital Stock—Registration Rights". All such shares are covered by lock-up agreements. Following the expiration of the lock-up period, registration of these shares under the

Securities Act would result in the shares becoming freely tradable without restriction under the Securities Act immediately upon the effectiveness of the registration, except for shares purchased by our affiliates.

## Form S-8 Registration Statements

We intend to file a registration statement on Form S-8 under the Securities Act covering all of the shares of common stock subject to options outstanding and RSUs, as well as reserved for future issuance, under our stock plans. We expect to file this registration statement as soon as practicable after this offering. However, none of the shares registered on Form S-8 will be eligible for resale until the expiration of the lock-up agreements to which they are subject.

# MATERIAL U.S. FEDERAL INCOME TAX CONSIDERATIONS FOR NON-U.S. HOLDERS OF OUR CLASS A COMMON STOCK

The following is a general discussion of the material U.S. federal income tax considerations applicable to non-U.S. holders (as defined below) with respect to their ownership and disposition of shares of our Class A common stock issued pursuant to this offering. For purposes of this discussion, a non-U.S. holder means a beneficial owner of our Class A common stock (other than an entity or arrangement that is treated as a partnership or pass-through entity for U.S. federal income tax purposes) that is not, for U.S. federal income tax purposes, any of the following:

- an individual who is a citizen or resident of the United States;
- a corporation (or other entity treated as a corporation for U.S. federal income tax purposes) created or organized in or under the laws of the United States, any state thereof or the District of Columbia;
- an estate, the income of which is includable in gross income for U.S. federal income tax purposes regardless of its source; or
- a trust if (i) a court within the United States is able to exercise primary supervision over the administration of the trust and one or more "U.S. persons," as defined under the Code, have the authority to control all substantial decisions of the trust or (ii) such trust has made a valid election to be treated as a U.S. person for U.S. federal income tax purposes under applicable U.S. Treasury Regulation.

This discussion is based on current provisions of the Code, existing, temporary and proposed Treasury Regulations promulgated thereunder, judicial opinions, published positions of the Internal Revenue Service (the "IRS") and other applicable authorities, each as in effect as of the date of this prospectus, and all of which are subject to change or to differing interpretation, possibly with retroactive effect. Any such change or different interpretation could alter the tax considerations to non-U.S. holders described in this prospectus. In addition, there can be no assurance that the IRS will not challenge one or more of the tax considerations described in this prospectus. This discussion assumes that a non-U.S. holder holds shares of our Class A common stock as a capital asset within the meaning of Section 1221 of the Code (generally, property held for investment) for U.S. federal income tax purposes. This discussion does not address all aspects of U.S. federal income taxation that may be important to a particular non-U.S. holder in light of that non-U.S. holder's individual circumstances, nor does it address any aspects of the Medicare contribution tax on net investment income, any U.S. non-income taxes, such as estate or gift taxes, any U.S. alternative minimum taxes or any state, local or non-U.S. taxes. This discussion may not apply, in whole or in part, to particular non-U.S. holders in light of their individual circumstances or to holders subject to special treatment under the U.S. federal income tax laws (such as insurance companies, tax-exempt organizations, government organizations, financial institutions, brokers or dealers in securities, "controlled foreign corporations," "passive foreign investment companies," corporations that accumulate earnings to avoid U.S. federal income tax, tax-qualified retirement plans and "qualified foreign pension funds" as defined in Section 897(1)(2) of the Code and entities all of the interests of which are held by qualified foreign pension funds, non-U.S. holders that hold our Class A common stock as part of a straddle, hedge, conversion transaction, synthetic security or other integrated investment and certain U.S. expatriates). If a partnership (or other entity or arrangement treated as a partnership for U.S. federal income tax purposes) holds our Class A common stock, the tax treatment of a partner therein will generally depend on the status of the partner and the activities of the partnership. Partners of a partnership holding our Class A common stock should consult their tax advisor as to the particular U.S. federal income tax consequences applicable to them

INVESTORS CONSIDERING THE PURCHASE OF OUR CLASS A COMMON STOCK SHOULD CONSULT THEIR OWN TAX ADVISORS REGARDING THE APPLICATION OF THE U.S.

FEDERAL INCOME TAX LAWS TO THEIR PARTICULAR SITUATIONS AND THE CONSEQUENCES OF U.S. FEDERAL NON-INCOME, STATE, LOCAL, OR NON-U.S. TAX LAWS AND TAX TREATIES.

#### **Dividends**

We have no present intention to make distributions on our Class A common stock. If we do pay dividends on shares of our Class A common stock, however, such distributions will constitute dividends for U.S. federal income tax purposes to the extent paid from our current or accumulated earnings and profits, as determined under U.S. federal income tax principles. Distributions in excess of our current and accumulated earnings and profits will constitute a return of capital that is applied against and reduces, but not below zero, a non-U.S. holder's adjusted tax basis in shares of our Class A common stock. Any amount distributed in excess of basis will be treated as capital gain and will be subject to the treatment described below under "—Gain on Sale or Other Disposition of Class A Common Stock." Any distributions will also be subject to the discussion below under "—Backup Withholding and Information Reporting" and "—Foreign Account Tax Compliance Act."

Any dividend paid to a non-U.S. holder on our Class A common stock that is not effectively connected with a non-U.S. holder's conduct of a trade or business in the United States will generally be subject to U.S. withholding tax at a 30% rate. The withholding tax might apply at a reduced rate, however, under the terms of an applicable income tax treaty between the United States and the non-U.S. holder's country of residence. You should consult your own tax advisors regarding your entitlement to benefits under a relevant income tax treaty. Generally, in order for us or our paying agent to withhold tax at a lower treaty rate, a non-U.S. holder must certify its entitlement to treaty benefits. A non-U.S. holder generally can meet this certification requirement by providing an appropriate and properly completed IRS Form W-8BEN, W-8BEN-E or other appropriate form (or any successor or substitute form thereof) to us or our paying agent. If the non-U.S. holder holds the stock through a financial institution or other agent acting on the holder's behalf, the holder will be required to provide appropriate documentation to the holder's agent. The holder's agent will then be required to provide certification to us or our paying agent, either directly or through other intermediaries. If you are eligible for a reduced rate of U.S. federal withholding tax under an income tax treaty, you may obtain a refund or credit of any excess amounts withheld by filing an appropriate claim for a refund with the IRS in a timely manner.

Dividends received by a non-U.S. holder that are effectively connected with a U.S. trade or business conducted by the non-U.S. holder, and if required by an applicable income tax treaty between the United States and the non-U.S. holder's country of residence, are attributable to a permanent establishment or fixed base maintained by the non-U.S. holder in the United States, are generally not subject to U.S. withholding tax if the non-U.S. holder satisfies applicable certification and disclosure requirements. To obtain this exemption, a non-U.S. holder must provide us or our paying agent with an IRS Form W-8ECI (or any successor form) properly certifying that the dividends are effectively connected with the non-U.S. holder's conduct of a trade or business within the United States. Such effectively connected dividends, although not subject to withholding tax, are taxed at the same rates applicable to U.S. persons, net of certain deductions and credits. In addition, dividends received by a corporate non-U.S. holder that are effectively connected with a U.S. trade or business of the corporate non-U.S. holder may also be subject to a branch profits tax at a rate of 30% or such lower rate as may be specified by an applicable tax treaty.

#### Gain on Sale or Other Disposition of Class A Common Stock

Subject to the discussion below under "—Backup Withholding and Information Reporting" and "—Foreign Account Tax Compliance Act," non-U.S. holders will generally not be subject to U.S. federal

income tax on any gains realized on the sale, exchange or other disposition of our Class A common stock unless:

- the gain (i) is effectively connected with the conduct by the non-U.S. holder of a U.S. trade or business and (ii) if required by an applicable income tax treaty between the United States and the non-U.S. holder's country of residence, is attributable to a permanent establishment or fixed base maintained by the non-U.S. holder in the United States (in which case the special rules described below apply);
- the non-U.S. holder is an individual who is present in the United States for 183 days or more in the taxable year of the sale, exchange or other disposition of our Class A common stock, and certain other requirements are met (in which case the gain would be subject to a flat 30% tax, or such reduced rate as may be specified by an applicable income tax treaty, which may be offset by U.S. source capital losses of such non-U.S. holder, if any, provided the non-U.S. holder has timely filed U.S. federal income tax returns with respect to such losses, even though the individual is not considered a resident of the United States); or
- the rules of the Foreign Investment in Real Property Tax Act ("FIRPTA") treat the gain as effectively connected with a U.S. trade or business.

The FIRPTA rules may apply to a sale, exchange or other disposition of our Class A common stock if we are, or were within the shorter of the five-year period preceding the disposition and the non-U.S. holder's holding period, a "U.S. real property holding corporation" ("USRPHC"). In general, we would be a USRPHC if interests in U.S. real estate comprised at least half of the value of our business assets. We do not believe that we are a USRPHC and we do not anticipate becoming one in the future. Even if we are or were to become a USRPHC, gain arising from the sale or other taxable disposition by a non-U.S. holder will not be subject to U.S. federal income tax if our Class A common stock is "regularly traded," as defined by applicable Treasury Regulations, on an established securities market, and such non-U.S. holder owned, actually and constructively, 5% or less of our common stock throughout the shorter of the five-year period ending on the date of the sale or other taxable disposition or the non-U.S. holder's holding period.

If any gain from the sale, exchange or other disposition of our Class A common stock, (i) is effectively connected with a U.S. trade or business conducted by a non-U.S. holder and (ii) if required by an applicable income tax treaty between the United States and the non-U.S. holder's country of residence, is attributable to a permanent establishment or fixed base maintained by such non-U.S. holder in the United States, then the gain generally will be subject to U.S. federal income tax at the same rates applicable to U.S. persons, net of certain deductions and credits. If the non-U.S. holder is a corporation, under certain circumstances, that portion of its earnings and profits that is effectively connected with its U.S. trade or business, subject to certain adjustments, generally would be subject also to a "branch profits tax" at a rate of 30% or such lower rate as may be specified by an applicable income tax treaty.

#### **Backup Withholding and Information Reporting**

We must report annually to the IRS and to each non-U.S. holder the amount of dividends paid to, and the tax withheld with respect to, each non-U.S. holder. These reporting requirements apply regardless of whether withholding was reduced or eliminated by an applicable tax treaty. Copies of this information reporting may also be made available under the provisions of a specific tax treaty or agreement with the tax authorities in the country in which the non-U.S. holder resides or is established.

A non-U.S. holder will generally be subject to backup withholding for dividends on our Class A common stock paid to such holder unless such holder certifies under penalties of perjury that, among

other things, it is a non-U.S. holder (and the payer does not have actual knowledge or reason to know that such holder is a U.S. person) or otherwise establishes an exemption. Generally, a non-U.S. holder will comply with such procedures if it provides a properly executed IRS Form W-8BEN, W-8BEN-E, or W-8ECI (or successor form) or otherwise meets documentary evidence requirements for establishing that it is a non-U.S. holder. Dividends paid to non-U.S. holders subject to withholding of U.S. federal income tax, as described above under the heading "—Dividends," will generally be exempt from U.S. backup withholding.

Information reporting and backup withholding generally are not required with respect to the amount of any proceeds from the sale or other disposition of our Class A common stock by a non-U.S. holder outside the United States through a foreign office of a foreign broker that does not have certain specified connections to the United States. However, if a non-U.S. holder sells or otherwise disposes of its shares of Class A common stock through a U.S. broker or the U.S. offices of a foreign broker, the broker will generally be required to report the amount of proceeds paid to the non-U.S. holder to the IRS and impose backup withholding on that amount unless such non-U.S. holder provides appropriate certification to the broker of its status as a non-U.S. holder (and the payer does not have actual knowledge or reason to know that such holder is a U.S. person) or otherwise establishes an exemption.

Backup withholding is not an additional income tax. Any amounts withheld under the backup withholding rules from a payment to a non-U.S. holder generally may be credited against the non-U.S. holder's U.S. federal income tax liability, if any, or refunded, provided that the required information is furnished to the IRS in a timely manner. Non-U.S. holders should consult their tax advisors regarding the application of the information reporting and backup withholding rules to them.

## **Foreign Account Tax Compliance Act**

Under the Foreign Account Tax Compliance Act ("FATCA"), withholding tax of 30% applies to certain payments to foreign financial institutions, investment funds and certain other non-U.S. persons that fail to comply with certain information reporting and certification requirements pertaining to their direct and indirect U.S. securityholders and/or U.S. accountholders and do not otherwise qualify for an exemption. Under applicable Treasury Regulations and IRS guidance, this withholding currently applies to payments of dividends, if any, on, and, subject to the proposed Treasury Regulations discussed below, gross proceeds from the sale or other disposition of, our Class A common stock. An intergovernmental agreement between the United States and a foreign country may modify the requirements described in this paragraph.

While, beginning on January 1, 2019, withholding under FATCA would have applied to payments of gross proceeds from the sale or other disposition of our Class A common stock, proposed Treasury Regulations eliminate FATCA withholding on payments of gross proceeds entirely. Taxpayers generally may rely on these proposed Treasury Regulations until final Treasury Regulations are issued.

Prospective investors are encouraged to consult with their own tax advisors regarding the possible implications of FATCA on their investment in our Class A common stock.

PROSPECTIVE INVESTORS SHOULD CONSULT THEIR OWN TAX ADVISORS REGARDING THE POTENTIAL APPLICATION OF WITHHOLDING UNDER FATCA TO THEIR INVESTMENT IN OUR CLASS A COMMON STOCK. THE PRECEDING DISCUSSION OF U.S. FEDERAL TAX CONSIDERATIONS IS FOR GENERAL INFORMATION PURPOSES ONLY. IT IS NOT TAX ADVICE. EACH PROSPECTIVE INVESTOR SHOULD CONSULT ITS OWN TAX ADVISOR REGARDING THE PARTICULAR U.S. FEDERAL, GIFT, ESTATE, STATE, LOCAL, AND NON-U.S. TAX CONSEQUENCES OF PURCHASING, HOLDING, AND DISPOSING OF OUR CLASS A COMMON STOCK, INCLUDING THE CONSEQUENCES OF ANY PROPOSED CHANGE IN APPLICABLE LAWS.

#### **UNDERWRITING**

We and the underwriters named below have entered into an underwriting agreement with respect to the shares of our Class A common stock being offered. Subject to certain conditions, each underwriter has severally agreed to purchase the number of shares indicated in the following table. Goldman Sachs & Co. LLC, J.P. Morgan Securities LLC, and BofA Securities, Inc. are the representatives of the underwriters.

Underwriters	Number of Shares
Goldman Sachs & Co. LLC	
J.P. Morgan Securities LLC	
BofA Securities, Inc	
Total	

The underwriters are committed to take and pay for all of the shares being offered, if any are taken, other than the shares covered by the option described below unless and until this option is exercised.

The underwriters have an option to buy up to an additional shares of our Class A common stock from us to cover sales by the underwriters of a greater number of shares than the total number set forth in the table above. They may exercise that option for 30 days. If any shares are purchased pursuant to this option, the underwriters will severally purchase shares in approximately the same proportion as set forth in the table above.

The following table shows the per share and total underwriting discounts and commissions to be paid to the underwriters by us. Such amounts are shown assuming both no exercise and full exercise of the underwriters' option to purchase up to an additional shares from us.

	No Exercise	Full Exercise
Per Share	\$	\$
Total	\$	\$

Shares sold by the underwriters to the public will initially be offered at the initial public offering price set forth on the cover of this prospectus. Any shares sold by the underwriters to securities dealers may be sold at a discount of up to \$ per share from the initial public offering price. After the initial offering of the shares, the representatives may change the offering price and the other selling terms. The offering of the shares by the underwriters is subject to receipt and acceptance and subject to the underwriters' right to reject any order in whole or in part.

We and our officers, directors, and holders of substantially all of our common stock and securities convertible into or exchangeable for our common stock have agreed or will agree with the underwriters, subject to certain exceptions, not to dispose of or hedge any of our or their common stock or securities convertible into or exchangeable for shares of common stock during the period from the date of this prospectus continuing through the date 180 days after the date of this prospectus, subject to certain exceptions, except with the prior written consent of Goldman Sachs & Co. LLC. This agreement does not apply to any existing employee benefit plans. See the section titled "Shares Eligible for Future Sale" for a discussion of certain transfer restrictions.

Prior to the offering, there has been no public market for the shares of our Class A common stock. The initial public offering price has been negotiated between the company and the

representatives. Among the factors to be considered in determining the initial public offering price of the shares, in addition to prevailing market conditions, will be the company's historical performance, estimates of the business potential and earnings prospects of the company, an assessment of our management and the consideration of the above factors in relation to market valuation of companies in related businesses.

We intend to apply to list our Class A common stock on under the symbol "

In connection with the offering, the underwriters may purchase and sell shares of our Class A common stock in the open market. These transactions may include short sales, stabilizing transactions and purchases to cover positions created by short sales. Short sales involve the sale by the underwriters of a greater number of shares than they are required to purchase in the offering, and a short position represents the amount of such sales that have not been covered by subsequent purchases. A "covered short position" is a short position that is not greater than the amount of additional shares for which the underwriters' option described above may be exercised. The underwriters may cover any covered short position by either exercising their option to purchase additional shares or purchasing shares in the open market. In determining the source of shares to cover the covered short position, the underwriters will consider, among other things, the price of shares available for purchase in the open market as compared to the price at which they may purchase additional shares pursuant to the option described above. "Naked" short sales are any short sales that create a short position greater than the amount of additional shares for which the option described above may be exercised. The underwriters must cover any such naked short position by purchasing shares in the open market. A naked short position is more likely to be created if the underwriters are concerned that there may be downward pressure on the price of the Class A common stock in the open market after pricing that could adversely affect investors who purchase in the offering. Stabilizing transactions consist of various bids for or purchases of Class A common stock made by the underwriters in the open market prior to the completion of the offering.

The underwriters may also impose a penalty bid. This occurs when a particular underwriter repays to the underwriters a portion of the underwriting discount received by it because the representatives have repurchased shares sold by or for the account of such underwriter in stabilizing or short covering transactions.

Purchases to cover a short position and stabilizing transactions, as well as other purchases by the underwriters for their own accounts, may have the effect of preventing or retarding a decline in the market price of our Class A common stock, and together with the imposition of the penalty bid, may stabilize, maintain or otherwise affect the market price of our Class A common stock. As a result, the price of our Class A common stock may be higher than the price that otherwise might exist in the open market. The underwriters are not required to engage in these activities and may end any of these activities at any time. These transactions may be effected on , in the over-the-counter market or otherwise.

We estimate that our share of the total expenses of the offering, excluding underwriting discounts and commissions, will be approximately \$\,\) . We have agreed to reimburse the underwriters for certain of their expenses in an amount up to \$\,\).

We have agreed to indemnify the several underwriters against certain liabilities, including liabilities under the Securities Act.

The underwriters and their respective affiliates are full service financial institutions engaged in various activities, which may include sales and trading, commercial and investment banking, advisory, investment management, investment research, principal investment, hedging, market making,

brokerage and other financial and non-financial activities and services. Certain of the underwriters and their respective affiliates may in the future provide, a variety of these services to the issuer and to persons and entities with relationships with the issuer, for which they will receive customary fees and expenses.

In the ordinary course of their various business activities, the underwriters and their respective affiliates, officers, directors and employees may purchase, sell or hold a broad array of investments and actively trade securities, derivatives, loans, commodities, currencies, credit default swaps and other financial instruments for their own account and for the accounts of their customers, and such investment and trading activities may involve or relate to assets, securities and/or instruments of the issuer (directly, as collateral securing other obligations or otherwise) and/or persons and entities with relationships with the issuer. The underwriters and their respective affiliates may also communicate independent investment recommendations, market color or trading ideas and/or publish or express independent research views in respect of such assets, securities or instruments and may at any time hold, or recommend to clients that they should acquire, long and/or short positions in such assets, securities and instruments.

#### European Economic Area and United Kingdom

In relation to each Member State of the European Economic Area and the United Kingdom (each a "Relevant State"), no common shares (the "Shares") have been offered or will be offered pursuant to the offering to the public in that Relevant State prior to the publication of a prospectus in relation to the Shares which has been approved by the competent authority in that Relevant State or, where appropriate, approved in another Relevant State and notified to the competent authority in that Relevant State, all in accordance with the Prospectus Regulation), except that offers of Shares may be made to the public in that Relevant State at any time under the following exemptions under the Prospectus Regulation:

- (a) to any legal entity which is a qualified investor as defined under the Prospectus Regulation;
- (b) to fewer than 150 natural or legal persons (other than qualified investors as defined under the Prospectus Regulation), subject to obtaining the prior consent of the representatives for any such offer; or
- (c) in any other circumstances falling within Article 1(4) of the Prospectus Regulation,

provided that no such offer of Shares shall require the company or any representatives to publish a prospectus pursuant to Article 3 of the Prospectus Regulation or supplement a prospectus pursuant to Article 23 of the Prospectus Regulation.

For the purposes of this provision, the expression an "offer to the public" in relation to any Shares in any Relevant State means the communication in any form and by any means of sufficient information on the terms of the offer and any Shares to be offered so as to enable an investor to decide to purchase or subscribe for any Shares, and the expression "Prospectus Regulation" means Regulation (EU) 2017/1129.

#### United Kingdom

Each Underwriter has represented and agreed that:

(a) it has only communicated or caused to be communicated and will only communicate or cause to be communicated an invitation or inducement to engage in investment activity (within the meaning of Section 21 of the Financial Services and Markets Act 2000 (as amended, the "FSMA")) received by it in connection with the issue or sale of the shares in circumstances in which Section 21(1) of the FSMA does not apply to the company; and

(b) it has complied and will comply with all applicable provisions of the FSMA with respect to anything done by it in relation to the shares in, from or otherwise involving the United Kingdom.

#### Canada

The securities may be sold in Canada only to purchasers purchasing, or deemed to be purchasing, as principal that are accredited investors, as defined in National Instrument 45-106 Prospectus Exemptions or subsection 73.3(1) of the Securities Act (Ontario), and are permitted clients, as defined in National Instrument 31-103 Registration Requirements, Exemptions, and Ongoing Registrant Obligations. Any resale of the securities must be made in accordance with an exemption form, or in a transaction not subject to, the prospectus requirements of applicable securities laws.

Securities legislation in certain provinces or territories of Canada may provide a purchaser with remedies for rescission or damages if this prospectus (including any amendment thereto) contains a misrepresentation, provided that the remedies for rescission or damages are exercised by the purchaser within the time limit prescribed by the securities legislation of the purchaser's province or territory. The purchaser should refer to any applicable provisions of the securities legislation of the purchaser's province or territory of these rights or consult with a legal advisor.

Pursuant to section 3A.3 of National Instrument 33-105 Underwriting Conflicts (NI 33-105), the underwriters are not required to comply with the disclosure requirements of NI 33-105 regarding underwriter conflicts of interest in connection with this offering.

#### Hong Kong

The shares may not be offered or sold in Hong Kong by means of any document other than (i) in circumstances which do not constitute an offer to the public within the meaning of the Companies (Winding Up and Miscellaneous Provisions) Ordinance (Cap. 32 of the Laws of Hong Kong) ("Companies (Winding Up and Miscellaneous Provisions) Ordinance") or which do not constitute an invitation to the public within the meaning of the Securities and Futures Ordinance (Cap. 571 of the Laws of Hong Kong) ("Securities and Futures Ordinance"), or (ii) to "professional investors" as defined in the Securities and Futures Ordinance and any rules made thereunder, or (iii) in other circumstances which do not result in the document being a "prospectus" as defined in the Companies (Winding Up and Miscellaneous Provisions) Ordinance, and no advertisement, invitation or document relating to the shares may be issued or may be in the possession of any person for the purpose of issue (in each case whether in Hong Kong or elsewhere), which is directed at, or the contents of which are likely to be accessed or read by, the public in Hong Kong (except if permitted to do so under the securities laws of Hong Kong) other than with respect to shares which are or are intended to be disposed of only to persons outside Hong Kong or only to "professional investors" in Hong Kong as defined in the Securities and Futures Ordinance and any rules made thereunder.

#### Singapore

This prospectus has not been registered as a prospectus with the Monetary Authority of Singapore. Accordingly, this prospectus and any other document or material in connection with the offer or sale, or invitation for subscription or purchase, of the shares may not be circulated or distributed, nor may the shares be offered or sold, or be made the subject of an invitation for subscription or purchase, whether directly or indirectly, to persons in Singapore other than (i) to an institutional investor (as defined under Section 4A of the Securities and Futures Act, Chapter 289 of Singapore (the "SFA")) under Section 274 of the SFA, (ii) to a relevant person (as defined in Section 275(2) of the SFA) pursuant to Section 275(1) of the SFA, or any person pursuant to

Section 275(1A) of the SFA, and in accordance with the conditions specified in Section 275 of the SFA or (iii) otherwise pursuant to, and in accordance with the conditions of, any other applicable provision of the SFA, in each case subject to conditions set forth in the SFA.

Where the shares are subscribed or purchased under Section 275 of the SFA by a relevant person which is a corporation (which is not an accredited investor (as defined in Section 4A of the SFA)) the sole business of which is to hold investments and the entire share capital of which is owned by one or more individuals, each of whom is an accredited investor, the securities (as defined in Section 239(1) of the SFA) of that corporation shall not be transferable for 6 months after that corporation has acquired the shares under Section 275 of the SFA except: (1) to an institutional investor under Section 274 of the SFA or to a relevant person (as defined in Section 275(2) of the SFA), (2) where such transfer arises from an offer in that corporation's securities pursuant to Section 275(1A) of the SFA, (3) where no consideration is or will be given for the transfer, (4) where the transfer is by operation of law, (5) as specified in Section 276(7) of the SFA, or (6) as specified in Regulation 32 of the Securities and Futures (Offers of Investments) (Shares and Debentures) Regulations 2005 of Singapore ("Regulation 32").

Where the shares are subscribed or purchased under Section 275 of the SFA by a relevant person which is a trust (where the trustee is not an accredited investor (as defined in Section 4A of the SFA)) whose sole purpose is to hold investments and each beneficiary of the trust is an accredited investor, the beneficiaries' rights and interest (howsoever described) in that trust shall not be transferable for 6 months after that trust has acquired the shares under Section 275 of the SFA except: (1) to an institutional investor under Section 274 of the SFA or to a relevant person (as defined in Section 275(2) of the SFA), (2) where such transfer arises from an offer that is made on terms that such rights or interest are acquired at a consideration of not less than S\$200,000 (or its equivalent in a foreign currency) for each transaction (whether such amount is to be paid for in cash or by exchange of securities or other assets), (3) where no consideration is or will be given for the transfer, (4) where the transfer is by operation of law, (5) as specified in Section 276(7) of the SFA, or (6) as specified in Regulation 32.

Solely for the purposes of its obligations pursuant to Section 309B of the SFA, we have determined, and hereby notify all relevant persons (as defined in the CMP Regulations 2018), that the shares are "prescribed capital markets products" (as defined in the CMP Regulations 2018) and Excluded Investment Products (as defined in MAS Notice SFA 04-N12: Notice on the Sale of Investment Products and MAS Notice FAA-N16: Notice on Recommendations on Investment Products).

#### Japan

The securities have not been and will not be registered under the Financial Instruments and Exchange Act of Japan (Act No. 25 of 1948, as amended) (the "FIEA"). The securities may not be offered or sold, directly or indirectly, in Japan or to or for the benefit of any resident of Japan (including any person resident in Japan or any corporation or other entity organized under the laws of Japan) or to others for reoffering or resale, directly or indirectly, in Japan or to or for the benefit of any resident of Japan, except pursuant to an exemption from the registration requirements of the FIEA and otherwise in compliance with any relevant laws and regulations of Japan.

#### Australia

No placement document, prospectus, product disclosure statement or other disclosure document has been lodged with the Australian Securities and Investments Commission ("ASIC"), in relation to the offering. This offering document does not constitute a prospectus, product disclosure statement or

other disclosure document under the Corporations Act 2001 (the "Corporations Act"), and does not purport to include the information required for a prospectus, product disclosure statement or other disclosure document under the Corporations Act.

Any offer in Australia of the shares may only be made to persons (the "Exempt Investors") who are "sophisticated investors" (within the meaning of section 708(8) of the Corporations Act), "professional investors" (within the meaning of section 708(11) of the Corporations Act) or otherwise pursuant to one or more exemptions contained in section 708 of the Corporations Act so that it is lawful to offer the shares without disclosure to investors under Chapter 6D of the Corporations Act.

The shares applied for by Exempt Investors in Australia must not be offered for sale in Australia in the period of 12 months after the date of allotment under the offering, except in circumstances where disclosure to investors under Chapter 6D of the Corporations Act would not be required pursuant to an exemption under section 708 of the Corporations Act or otherwise or where the offer is pursuant to a disclosure document which complies with Chapter 6D of the Corporations Act. Any person acquiring shares must observe such Australian on-sale restrictions.

This offering document contains general information only and does not take account of the investment objectives, financial situation or particular needs of any particular person. It does not contain any securities recommendations or financial product advice. Before making an investment decision, investors need to consider whether the information in this offering document is appropriate to their needs, objectives and circumstances, and, if necessary, seek expert advice on those matters.

#### Dubai International Financial Centre

This offering document relates to an Exempt Offer in accordance with the Offered Securities Rules of the Dubai Financial Services Authority ("DFSA"). This offering document is intended for distribution only to persons of a type specified in the Offered Securities Rules of the DFSA. It must not be delivered to, or relied on by, any other person. The DFSA has no responsibility for reviewing or verifying any documents in connection with Exempt Offers. The DFSA has not approved this prospectus nor taken steps to verify the information set forth in this prospectus and has no responsibility for the offering document. The securities to which this offering document relates may be illiquid and/or subject to restrictions on their resale. Prospective purchasers of the securities offered should conduct their own due diligence on the securities. If you do not understand the contents of this offering document you should consult an authorized financial advisor.

#### Switzerland

We have not and will not register with the Swiss Financial Market Supervisory Authority ("FINMA") as a foreign collective investment scheme pursuant to Article 119 of the Federal Act on Collective Investment Scheme of 23 June 2006, as amended ("CISA"), and accordingly the securities being offered pursuant to this prospectus have not and will not be approved, and may not be licenseable, with FINMA. Therefore, the securities have not been authorized for distribution by FINMA as a foreign collective investment scheme pursuant to Article 119 CISA and the securities offered hereby may not be offered to the public (as this term is defined in Article 3 CISA) in or from Switzerland. The securities may solely be offered to "qualified investors," as this term is defined in Article 10 CISA, and in the circumstances set out in Article 3 of the Ordinance on Collective Investment Scheme of 22 November 2006, as amended ("CISO"), such that there is no public offer. Investors, however, do not benefit from protection under CISA or CISO or supervision by FINMA. This prospectus and any other materials relating to the securities are strictly personal and confidential to each offeree and do not constitute an offer to any other person. This prospectus may only be used by those qualified investors to whom it has been handed out in connection with the offer described in this

prospectus and may neither directly or indirectly be distributed or made available to any person or entity other than its recipients. It may not be used in connection with any other offer and shall in particular not be copied and/or distributed to the public in Switzerland or from Switzerland. This prospectus does not constitute an issue prospectus as that term is understood pursuant to Article 652a and/or 1156 of the Swiss Federal Code of Obligations. We have not applied for a listing of the securities on the SIX Swiss Exchange or any other regulated securities market in Switzerland, and consequently, the information presented in this prospectus does not necessarily comply with the information standards set out in the listing rules of the SIX Swiss Exchange and corresponding prospectus schemes annexed to the listing rules of the SIX Swiss Exchange.

#### **LEGAL MATTERS**

The validity of the shares of Class A common stock offered by this prospectus will be passed upon for us by Gunderson Dettmer Stough Villeneuve Franklin & Hachigian, LLP, San Francisco, California. The underwriters have been represented by Cooley LLP, San Francisco, California.

#### **EXPERTS**

Ernst & Young LLP, independent registered public accounting firm, has audited our consolidated financial statements at December 31, 2018 and 2019, and for each of the three years in the period ended December 31, 2019, as set forth in their report. We have included our financial statements in the prospectus and elsewhere in the registration statement in reliance on Ernst & Young LLP's report, given on their authority as experts in accounting and auditing.

#### WHERE YOU CAN FIND MORE INFORMATION

We have filed with the SEC a registration statement on Form S-1 under the Securities Act with respect to the shares of Class A common stock offered hereby. This prospectus, which constitutes a part of the registration statement, does not contain all of the information set forth in the registration statement or the exhibits filed therewith. For further information about us and the Class A common stock offered hereby, reference is made to the registration statement and the exhibits filed therewith. Statements contained in this prospectus regarding the contents of any contract or any other document that is filed as an exhibit to the registration statement are not necessarily complete, and in each instance we refer you to the copy of such contract or other document filed as an exhibit to the registration statement.

Upon the completion of this offering, we will be subject to the information reporting requirements of the Exchange Act and we will file reports, proxy statements and other information with the SEC. You can read these reports, proxy statements and other information at the SEC's website at www.sec.gov. We also maintain a website at www.wish.com, at which you may access these materials free of charge as soon as reasonably practicable after they are electronically filed with, or furnished to, the SEC. Information contained on our website is not incorporated by reference into this prospectus, and you should not consider information contained on our website to be part of this prospectus or in deciding whether to invest in our Class A common stock.

# CONTEXTLOGIC INC.

# INDEX TO CONSOLIDATED FINANCIAL STATEMENTS

	Page
Report of Independent Registered Public Accounting Firm	F-2
Consolidated Financial Statements	
Consolidated Balance Sheets	F-3
Consolidated Statements of Operations and Comprehensive Loss	F-4
Consolidated Statements of Redeemable Convertible Preferred Stock and Stockholders' Deficit	F-5
Consolidated Statements of Cash Flows	F-6
Notes to Consolidated Financial Statements	F-7

#### Report of Independent Registered Public Accounting Firm

To the Stockholders and the Board of Directors of ContextLogic Inc.

#### **Opinion on the Financial Statements**

We have audited the accompanying consolidated balance sheets of ContextLogic Inc. (the Company) as of December 31, 2018 and 2019, the related consolidated statements of operations and comprehensive loss, redeemable convertible preferred stock and stockholders' deficit, and cash flows for each of the three years in the period ended December 31, 2019, and the related notes (collectively referred to as the "consolidated financial statements"). In our opinion, the consolidated financial statements present fairly, in all material respects, the financial position of the Company at December 31, 2018 and 2019, and the results of its operations and its cash flows for each of the three years in the period ended December 31, 2019 in conformity with U.S. generally accepted accounting principles.

#### **Basis for Opinion**

These financial statements are the responsibility of the Company's management. Our responsibility is to express an opinion on the Company's financial statements based on our audits. We are a public accounting firm registered with the Public Company Accounting Oversight Board (United States) (PCAOB) and are required to be independent with respect to the Company in accordance with the U.S. federal securities laws and the applicable rules and regulations of the Securities and Exchange Commission and the PCAOB.

We conducted our audits in accordance with the standards of the PCAOB and in accordance with auditing standards generally accepted in the United States of America. Those standards require that we plan and perform the audit to obtain reasonable assurance about whether the financial statements are free of material misstatement, whether due to error or fraud. The Company is not required to have, nor were we engaged to perform, an audit of its internal control over financial reporting. As part of our audits we are required to obtain an understanding of internal control over financial reporting but not for the purpose of expressing an opinion on the effectiveness of the Company's internal control over financial reporting. Accordingly, we express no such opinion.

Our audits included performing procedures to assess the risks of material misstatement of the financial statements, whether due to error or fraud, and performing procedures that respond to those risks. Such procedures included examining, on a test basis, evidence regarding the amounts and disclosures in the financial statements. Our audits also included evaluating the accounting principles used and significant estimates made by management, as well as evaluating the overall presentation of the financial statements. We believe that our audits provide a reasonable basis for our opinion.

#### **Adoptions of New Accounting Standards**

As discussed in Note 2 to the consolidated financial statements, the Company changed its methods of accounting for revenue in 2018 and for leases in 2019 as a result of the adoption of Accounting Standards Update (ASU) No. 2014-09, Revenue from Contracts with Customers (Topic 606) and ASU No. 2016-02, Leases (Topic 842), and related amendments, respectively. Our opinion is not modified with respect to this matter.

/s/ Ernst & Young LLP

We have served as the Company's auditor since 2015.

San Francisco, California August 28, 2020

# CONTEXTLOGIC INC. CONSOLIDATED BALANCE SHEETS (in millions, except share and per share data)

	As of December 31, 2018 2019		Pro Forma December 31, 2019 (unaudited)
Assets			,
Current assets:			
Cash and cash equivalents	\$ 712	\$ 744	
Marketable securities	262	300	
Funds receivable	93	95	
Prepaid expenses and other current assets	37	95	
Total current assets	\$ 1,104	\$ 1,234	
Property and equipment, net	33	34	
Right-of-use assets	_	41	
Marketable securities	40	34	
Restricted cash	10	10	
Other assets	6	13	
Total assets	\$ 1,193	\$ 1,366	
Liabilities, Redeemable Convertible Preferred Stock and Stockholders' Deficit			
Current liabilities:			
Accounts payable	\$ 131	\$ 171	
Merchants payable	653	620	
Refunds liability	61	97	
Accrued liabilities	125	212	
Total current liabilities	\$ 970	\$ 1,100	
Redeemable convertible preferred stock warrant liability	124	127	
Lease liabilities, non-current	_	42	
Deferred rent	10	_	
Total liabilities	\$ 1,104	\$ 1,269	
Commitments and contingencies (Note 6)  Redeemable convertible preferred stock, \$0.0001 par value:  43,352,566 and 44,346,293 shares authorized as of December 31, 2018 and 2019,			
respectively; 41,295,644 and 42,169,192 shares issued and outstanding as of December 31, 2018 and 2019, respectively; aggregate liquidation preference of \$1,380 and \$1,619 as of December 31, 2018 and 2019, respectively; shares authorized, issued and outstanding as of December 31, 2019, pro forma (unaudited)	1,376	1,536	
Stockholders' deficit:			
Common stock, \$0.0001 par value: 70,000,000 and 73,000,000 shares authorized as of December 31, 2018 and 2019; 10,445,075 and 10,369,485 shares issued and outstanding as of December 31, 2018 and 2019; shares issued and outstanding as of December 31, 2019, pro forma (unaudited)	_	_	
Accumulated deficit	(1,287)	(1,439)	
Total stockholders' deficit	\$(1,287)	\$(1,439)	
Total liabilities, redeemable convertible preferred stock and stockholders' deficit	\$ 1,193	\$ 1,366	

The accompanying notes are an integral part of these consolidated financial statements

# CONTEXTLOGIC INC. CONSOLIDATED STATEMENTS OF OPERATIONS AND COMPREHENSIVE LOSS (in millions, except share and per share data)

Year Ended December 31,					
2017		2018 (As Adjusted)		2019	
\$	1,101	\$	1,728	\$	1,901
	205		278		443
	896		1,450		1,458
			1,576		1,463
	28		45		74
	26		52		65
	1,043		1,673		1,602
	(147)		(223)		(144)
	10		15		19
	(70)		<u></u>		(3)
	(207)		(208)		(128)
	<u> </u>		<u> </u>		1
	(207)		(208)		(129)
	(40)		<u> </u>		(7)
\$	(247)	\$	(208)	\$	(136)
<del></del>			<del></del>		
\$	(24.60)	\$	(20.20)	\$	(13.07)
10,038,841		10,296,604		10,404,562	
				\$	
					_
				_	
	\$	\$ 1,101 205 896 989 28 26 1,043 (147) 10 (70) (207) ————————————————————————————————————	2017 2018 (a \$ 1,101 \$ 205 896  989 28 26 1,043 (147)  10  (70) (207) ————————————————————————————————————	2017         2018 (As Adjusted)           \$ 1,101         \$ 1,728           205         278           896         1,450           989         1,576           28         45           26         52           1,043         1,673           (147)         (223)           10         15           (70)         —           (207)         (208)           (40)         —           \$ (247)         \$ (208)           \$ (24.60)         \$ (20.20)	2017     2018 (As Adjusted)       \$ 1,101     \$ 1,728       205     278       896     1,450       989     1,576       28     45       26     52       1,043     1,673       (147)     (223)       10     15       (70)     —       (207)     (208)       (40)     —       \$ (247)     \$ (208)       \$ (2460)     \$ (20.20)

The accompanying notes are an integral part of these consolidated financial statements

### CONTEXTLOGIC INC.

### CONSOLIDATED STATEMENTS OF REDEEMABLE CONVERTIBLE PREFERRED STOCK AND STOCKHOLDERS' DEFICIT (In millions, except share and per share amounts)

	Redeema Convertible Pi Stock	eferred	Common S		Additional Paid-in	Accumulated	Sto	Total ockholders'
Balances as of January 1, 2017	Shares 39,926,498	\$1,107	Shares 9,947,001	Amount \$ -	\$ 38	Deficit \$ (868)	\$	Deficit (830)
Issuance of Series G redeemable convertible preferred stock for cash, at \$134.53 per			3,347,001	Ψ –	Ψ 30	ψ (000)	Ψ	(030)
share, net of issuance costs	1,687,319	226	_	_	_	_		_
Repurchase of Series C redeemable convertible preferred stock	(857,601)	(5)	_	_	(39)	(1)		(40)
Reclassification of Series F redeemable convertible preferred stock warrant liability upon exercise	_	13	_	_	_	_		_
Exercise of Series F redeemable convertible preferred stock warrant	539,428	35	_	-	-	_		-
Issuance of common stock upon exercise of options for cash	-	_	147,538	_	_	_		_
Repurchase of common stock	_	-	(8,766)	_	(3)	-		(3)
Stock-based compensation	_	-	-	_	6	-		6
Loans to employees secured by shares and options	_	-	_	-	(2)	_		(2)
Net loss and comprehensive loss						(207)		(207)
Balances as of December 31, 2017	41,295,644	\$1,376	10,085,773	\$ -	\$ -	\$ (1,076)	\$	(1,076)
Issuance of common stock upon exercise of options for cash	_	_	422,748	_	1	_		1
Repurchase of common stock.	_	_	(63,446)	_	(3)	(3)		(6)
Stock-based compensation	_	_	<u> </u>	_	2	<u> </u>		2
Net loss and comprehensive loss.	_	_	_	_	_	(208)		(208)
Balances as of December 31, 2018	41,295,644	\$1,376	10,445,075	\$ -	\$ -	\$ (1,287)	\$	(1,287)
Issuance of Series H redeemable convertible preferred stock for cash, at \$169.5729 per share, net of issuance costs	943,548	160		_	_	_		_
Issuance of common stock upon exercise of options for cash	940,040	-	75,563	_	_	_		_
Repurchase of Series A redeemable convertible preferred stock	(70,000)		_	_	(3)	(4)		(7)
Issuance of common stock for services	(10,000)	_	27,534	_	3	(1)		3
Repurchase of common stock.	_	_	(178,687)	_	(2)	(19)		(21)
Stock-based compensation	_	_	(170,007)		2	(13)		2
Net loss and comprehensive loss	_	_		_	_	(129)		(129)
Balances as of December 31, 2019	42,169,192	\$1,536	10.369.485	\$ -	\$ -	\$ (1,439)	\$	(1,439)
Dalatices as Of December 31, 2018	72,103,132	ψ 1,000	10,309,403	ψ –	Ψ –	<u>ψ (1,439</u> )	φ	(1, <del>4</del> 39)

The accompanying notes are an integral part of these consolidated financial statements

# CONTEXTLOGIC INC. CONSOLIDATED STATEMENTS OF CASH FLOWS (in millions)

	Year Er		
Cash flows from operating activities:	2017	2018	2019
Net loss	\$(207)	\$(208)	\$(129)
Adjustments to reconcile net loss to net cash used in operating activities:	Ψ(201)	Ψ(200)	ψ(123)
Depreciation and amortization	4	8	10
Noncash lease expense		_	9
Stock-based compensation expense	8	2	2
Remeasurement of redeemable convertible preferred stock warrant liability	70	_	3
Other	3	(9)	(3)
Changes in operating assets and liabilities:		(-)	(-)
Funds receivable	(59)	13	(2)
Prepaid expenses and other current assets	(24)	(1)	(58)
Other noncurrent assets	(5)	1	(7)
Accounts payable	85	2	40
Merchants payable	235	15	(33)
Accrued and refund liabilities	37	82	94
Lease liabilities	_	_	(10)
Current and noncurrent liabilities	(1)	1	24
Net cash provided by (used in) operating activities	146	(94)	(60)
Cash flows from investing activities:			
Purchases of property and equipment	(12)	(20)	(11)
Purchases of marketable securities	(343)	(366)	(485)
Sales of marketable securities	(343)	(300)	53
Maturities of marketable securities	157	360	403
Other	157	9	403
			(40)
Net cash used in investing activities	(192)	(16)	(40)
Cash flows from financing activities:			
Proceeds from issuance of redeemable convertible preferred stock, net of issuance cost	226	_	160
Proceeds from exercise of redeemable convertible preferred stock warrants	35	_	_
Proceeds from issuance of common stock upon exercise of stock options	_	1	-
Payments to repurchase common and redeemable convertible preferred stock	(48)	(6)	(28)
Other	(1)		
Net cash provided by (used in) financing activities	212	(5)	132
Net increase (decrease) in cash, cash equivalents and restricted cash	166	(115)	32
Cash, cash equivalents and restricted cash at beginning of period	671	837	722
Cash, cash equivalents and restricted cash at end of period	\$ 837	\$ 722	\$ 754
Reconciliation of cash, cash equivalents, and restricted cash to the consolidated balance sheets:			
Cash and cash equivalents	\$ 827	\$ 712	\$ 744
Restricted cash	10	10	10
Total cash, cash equivalents and restricted cash	\$ 837	\$ 722	\$ 754
Supplemental noncash financing activities:	<del></del>	<del></del>	<del></del>
Reclassification of Series F redeemable convertible preferred stock warrant liability upon exercise	\$ 13	\$ -	\$ -
Issuance of common stock for services	_	_	3

The accompanying notes are an integral part of these consolidated financial statements

### CONTEXTLOGIC INC.

### NOTES TO CONSOLIDATED FINANCIAL STATEMENTS

### 1. DESCRIPTION OF BUSINESS

ContextLogic Inc. ("Wish" or the "Company") is a mobile ecommerce company that provides a shopping experience that is mobile-first and discovery-based, which connects merchants' products to users based on user preferences. The Company generates revenue from marketplace and logistics services provided to merchants.

The Company was incorporated in the state of Delaware in June 2010 and is headquartered in San Francisco, California, with operations in Canada, China and the Netherlands.

The Company has incurred net losses from operations since inception. For the year ended December 31, 2019, the Company incurred a net loss of \$129 million and had an accumulated deficit of \$1.4 billion as of December 31, 2019. The Company expects losses from operations to continue for the foreseeable future as it incurs costs and expenses related to brand development, expansion of market share, continued development of the Company's mobile shopping marketplace infrastructure and development of other businesses. The Company believes that its cash and cash equivalents and current marketable securities balances of \$1.0 billion as of December 31, 2019 are sufficient to fund its operations through at least the next 12 months from the issuance of these financial statements.

### 2. SUMMARY OF SIGNIFICANT ACCOUNTING POLICIES

#### Basis of Presentation and Consolidation

The accompanying consolidated financial statements have been prepared in accordance with U.S. generally accepted accounting principles ("U.S. GAAP"). The consolidated financial statements include the accounts of the Company and its wholly owned subsidiaries. All intercompany balances and transactions have been eliminated in consolidation.

### Unaudited Pro Forma Balance Sheet

Immediately prior to the completion of the Company's planned initial public offering ("IPO"), all of the Company's outstanding redeemable convertible preferred stock will automatically convert into shares of the Company's common stock, and all of the Company's outstanding redeemable convertible preferred stock warrant will be exercised and will also convert into shares of the Company's common stock. These conversions are reflected into an aggregate of shares of the Company's common stock in the Company's unaudited pro forma balance sheet as of , which also reflects an increase to additional paid-in capital and accumulated deficit related to the recognition of \$ of stock-based compensation expense related to the Company's restricted stock units ("RSUs"), which are subject to both a service condition, which is typically satisfied over four or five years, and a performance condition, which will be satisfied upon completion of this IPO. These RSUs are excluded from the pro forma disclosures on the consolidated balance sheet because the underlying shares of common stock will be issued subsequent to the completion of this IPO. The shares of common stock issuable and the proceeds expected to be received in connection with an IPO are excluded from such unaudited pro forma financial information.

### **Deferred Offering Costs**

Deferred offering costs consist primarily of accounting, legal and other fees related to the Company's proposed IPO. Upon consummation of the IPO, the deferred offering costs will be reclassified to stockholders' deficit and recorded against the proceeds from the offering. In the event the offering is aborted, deferred offering costs will be expensed. There were no offering costs capitalized as of December 31, 2018 and 2019.

### Use of Estimates

The preparation of consolidated financial statements in accordance with U.S. GAAP requires management to make estimates and assumptions that affect the reported amounts of assets and liabilities, disclosures of contingent assets and liabilities at the date of the consolidated financial statements, and the reported amounts of revenue and expenses during the reporting period. These estimates form the basis for judgments the Company makes about the carrying values of its assets and liabilities that are not readily apparent from other sources. These estimates include, but are not limited to, fair value of financial instruments, useful lives of long-lived assets, fair value of common stock, fair value of redeemable convertible preferred stock and related redeemable convertible preferred stock warrant and equity awards and other equity issuances, incremental borrowing rate applied to lease accounting, contingent liabilities, allowances for refunds and chargebacks and uncertain tax positions. These estimates are based on management's knowledge about current events and expectations about actions the Company may undertake in the future. Actual results could differ from these estimates.

### Segments

The Company manages its operations and allocates resources as a single operating segment. Further, the Company manages, monitors and reports its financials as a single reporting segment. The Company's chief operating decision-maker ("CODM") is its Chief Executive Officer who makes operating decisions, assesses financial performance and allocates resources based on consolidated financial information. As such, the Company has determined that it operates in one reportable segment.

### Revenue Recognition

Through the fiscal year ended December 31, 2017, in accordance with Accounting Standards Codification ("ASC") Topic 605, *Revenue Recognition* ("Topic 605"), the Company recognized revenue when all of the following conditions were satisfied: (1) there was persuasive evidence of an arrangement; (2) delivery has occurred or the service has been rendered; (3) collectability was reasonably assured; and (4) the selling price or fee was fixed or determinable. The Company evaluated whether it was appropriate to recognize revenue on a gross or net basis based upon its evaluation of whether it was the primary obligor in a transaction, had inventory risk and had latitude in establishing pricing and selecting suppliers, among other factors.

On January 1, 2018, the Company adopted Accounting Standards Update ("ASU") No. 2014-09, Revenue from Contracts with Customers ("Topic 606"), which supersedes the revenue recognition requirements in Topic 605, using the modified retrospective transition method. Revenue for the years ended December 31, 2018 and 2019 was presented under Topic 606, and revenue for the year ended December 31, 2017 was not adjusted and continues to be presented under Topic 605. Upon adoption of Topic 606, the Company recognizes revenue upon transfer of control of promised products or services to customers in an amount that reflects the consideration the Company expects to receive in exchange for those products or services. Under the new revenue recognition standard, the Company applies the following five step approach: (1) identify the contract with a customer, (2) identify the performance obligations in the contract, (3) determine the transaction price, (4) allocate the transaction price to the performance obligations in the contract and (5) recognize revenue when a performance obligation is satisfied.

Upon the adoption of Topic 606, the Company recorded the costs of free products provided to users and the costs of certain refunds as a reduction of revenue in the consolidated statements of operations and comprehensive loss. These costs were previously recorded in cost of revenue under Topic 605. There was no adjustment to accumulated deficit on January 1, 2018.

The following table reflects the impact of the Topic 606 adoption on the relevant line items of the consolidated statement of operations and comprehensive loss for the initial year of adoption had the Company reported under Topic 605 (in millions):

		Year Ended December 31, 2018					
	As Reported	Free Product Incentives	Cost of Refunds	If Reported under Topic 605			
Revenue	\$ 1,728	\$ 22	\$ 4	\$ 1,754			
Cost of revenue	278	22	4	304			
Gross profit	<u>\$ 1,450</u>	\$ _	\$ _	\$ 1,450			

The Company generates revenue from marketplace and logistics services provided to merchants. Revenue is recognized as the Company transfers control of promised goods or services to its customers, in an amount that reflects the consideration the Company expects to be entitled to in exchange for those goods or services. The Company considers both the merchant and the user to be customers. The Company evaluates whether it is appropriate to recognize revenue on a gross or net basis based upon its evaluation of whether the Company obtains control of the specified goods or services by considering if it is primarily responsible for fulfillment of the promise, has inventory risk and has latitude in establishing pricing and selecting suppliers, among other factors. Based on these factors, marketplace revenue is generally recorded on a net basis and logistics revenue is generally recorded on a gross basis. Revenue excludes any amounts collected on behalf of third parties, including indirect taxes.

### Marketplace Revenue

The Company provides a mix of marketplace services to merchants. The Company provides merchants access to its marketplace where merchants display and sell their products to users. The Company also provides ProductBoost services to help merchants promote their products within the Company's marketplace.

Marketplace revenue includes commission fees collected in connection with user purchases of the merchants' products. The commission fees vary depending on factors such as user location, demand, product type, and dynamic pricing. The Company recognizes revenue when a user's order is processed and the related order information has been made available to the merchant. Commission fees are recognized net of estimated refunds and chargebacks. Marketplace revenue also includes ProductBoost fees for displaying a merchant's selected products in preferential locations within the Company's marketplace. The Company recognizes revenue when the merchants' selected products are displayed.

Marketplace revenue was approximately 100%, 100% and 93% of total revenue in 2017, 2018 and 2019, respectively.

#### Logistics Revenue

The Company's logistics offering for merchants, introduced in 2018, is designed for direct end-to-end single order dropship from a merchant's location to the user. Logistics services include warehousing, transportation, and delivery of the merchant's products to the user. The Company recognizes revenue over time as the logistics services are provided.

Logistics revenue was approximately 0% and 7% of total revenue in 2018 and 2019, respectively.

### Deferred revenue

Deferred revenue consists of amount received related to unsatisfied performance obligations at the end of the period. The deferred revenue balances as of December 31, 2018 and 2019 are disclosed in Note 4. Due to the short-term duration of contracts, all of the performance obligations will be satisfied in the following reporting period.

### Refunds and Chargebacks

Refunds and chargebacks are associated with marketplace revenue. Returns are not material to the Company's business. Estimated refunds and chargebacks are recorded on the consolidated balance sheets as refunds liability. The merchant's share of the refunds are recorded as a reduction to the amount due to merchants. The revenue recognized on transactions subject to refunds and chargebacks is reversed. The Company estimates future refunds and chargebacks using a model that incorporates historical experience and considering recent business trends and market activity.

### Incentive Discount Offers

The Company provides incentive discount offers to its users to encourage purchases of products through its marketplace. Such offers include current discount offers of a certain percentage off current purchases and inducement offers, such as set percentage offers off future purchases subject to a minimum current purchase. The Company generally records the related discounts taken as a reduction of revenue when the offer is redeemed. The Company also offers free products to encourage users to make purchases on its marketplace. The resulting discount is recorded as a reduction of revenue when the offer for free product is redeemed.

### Cost of Revenue

Cost of revenue includes colocation and data center charges, interchange and other fees for credit card processing services, fraud and chargeback prevention service charges, costs of refunds and chargebacks made to users that the Company is not able to collect from merchants, depreciation and amortization of property and equipment, shipping charges, tracking costs, warehouse fees, and employee-related costs, including salaries, benefits, and stock-based compensation expense, for the Company's infrastructure, merchant support and logistics. Cost of revenue also includes an allocation of general IT and facilities overhead expenses.

### Advertising Expense

Advertising expenses are included in sales and marketing expenses within the consolidated statements of operations and comprehensive loss and are expensed as incurred. Advertising expenses were \$0.9 billion, \$1.5 billion and \$1.4 billion in 2017, 2018 and 2019, respectively.

### Software Development Costs

The Company capitalizes costs to develop its mobile application and website when preliminary development efforts are successfully completed, management has authorized and committed project funding, and it is probable that the project will be completed and the software will be used as intended. Due to the iterative process by which the Company performs upgrades and the relatively short duration of its development projects, development costs meeting capitalization criteria were not material during the periods presented.

#### Cash, Cash Equivalents and Restricted Cash

The Company considers all highly liquid investments with an original maturity of three months or less to be cash equivalents. As of December 31, 2018, and 2019, cash and cash equivalents consisted of cash deposited with banks and money market funds for which their cost approximates their fair value. The Company held 84% and 82% of its cash and cash equivalents in the United States as of December 31, 2018 and 2019, respectively.

Restricted cash consists of mandatory deposits as collateral for standby letters of credit related to the Company's primary office space lease and is classified as non-current.

#### Marketable Securities

Marketable securities consist of short-term and long-term debt securities classified as available-for-sale and have original maturities greater than 90 days. Marketable securities are carried at fair value based upon quoted market prices or pricing models for similar securities. Unrealized gains and losses on available-for-sale securities are excluded from earnings and are not material. Realized gains or losses on the sale of all such securities are reported in interest and other income (expense), net, and computed using the specific identification method. For declines in fair market value below the cost of an individual marketable security, the Company assesses whether the decline in value is other than temporary based on the length of time the fair market value has been below cost, the severity of the decline and the Company's intent and ability to hold or sell the investment. If an investment is impaired, the Company writes it down through earnings to its recoverable value and establishes that as a new cost basis for the investment.

#### Funds Receivable

The Company uses several third-party Payment Service Providers ("PSPs") to process user transactions on its marketplace. Transactions on the Company's marketplace are mainly credit and debit card-based transactions that convert to cash on a regular basis and are net settled against refunds and chargebacks, with little default risk. Funds receivable represent the amounts expected to be received from PSPs for purchases on the Company's marketplace and are recorded net of processing fees.

#### Concentrations of Risk

Credit Risk — Financial instruments that potentially subject the Company to concentrations of credit risk consist primarily of cash and cash equivalents, restricted cash, funds receivable and marketable securities. The Company's cash and cash equivalents, including restricted cash, are held on deposit with creditworthy institutions. Although the Company's deposits exceed federally insured limits, the Company has not experienced any losses in such accounts. The Company invests its excess cash in money market accounts, U.S. Treasury notes, U.S. Treasury bills, commercial paper and corporate bonds. The Company is exposed to credit risk in the event of a default by the financial institutions holding its cash, cash equivalents, restricted cash and marketable securities for the amounts reflected on the consolidated balance sheets. The Company's investment policy limits investments to certain types of debt securities issued by the U.S. government, its agencies and institutions with investment-grade credit ratings and places restrictions on maturities and concentration by type and issuer.

As of December 31, 2018 and 2019, the following PSPs each represented 10% or more of the Company's funds receivable balance:

	Decen	nber 31,
	2018	2019
PSP 1	49%	2019 50%
PSP 2	23%	18%
PSP 3	14%	13%
PSP 4	9%	10%

The Company does not generate revenue from PSPs. Significant changes in the Company's relationship with its PSPs could adversely affect users' ability to process transactions on the Company's marketplaces, thereby impacting the Company's operating results.

Services Risk — The Company serves all of its users using third-party data center and hosting providers. The Company has disaster recovery protocols at the third-party service providers. Even with these procedures for disaster recovery in place, access to the Company's service could be significantly interrupted, resulting in an adverse effect on its operating results and financial position. No significant interruptions of service were known to have occurred during 2018 and 2019.

### Property and Equipment, Net

Property and equipment are stated at historical cost less accumulated depreciation. Depreciation and amortization are computed using the straight-line method over the estimated useful lives. Expenditures for repairs and maintenance are charged to expense as incurred.

The estimated useful lives of the Company's property and equipment are generally as follows:

Computers, equipment, software Furniture and fixtures, servers, networking equipment Leasehold improvements

3 years 5 years Shorter of the estimated useful life or remaining lease term

### Impairment of Long-Lived Assets

The Company reviews long-lived assets, including intangible assets, for impairment whenever events or changes in circumstances indicate that the carrying amount of an asset may not be recoverable. Recoverability of assets to be held and used is measured first by a comparison of the carrying amount of an asset to future undiscounted net cash flows expected to be generated by the asset. If such assets are considered to be impaired, an impairment loss would be recognized based on the excess of the carrying amount of the asset above the fair value of the asset. No impairment charges were recorded on long-lived assets in 2017, 2018 or 2019.

### Merchants Payable

Merchants payable represents the amount of funds due to merchants and are recorded net of commission fees earned by the Company for marketplace transactions and other fees due from merchants. Merchants payable is adjusted for actual and estimated refunds the Company is expected to recover from merchants. The Company remits funds to merchants on a regular basis.

### Lease Obligations

On January 1, 2019, the Company adopted ASU 2016-02, *Leases (Topic 842)* ("Topic 842"), using the modified retrospective method, which resulted in the recognition of right-of-use ("ROU") assets and lease liabilities for operating leases on the Company's consolidated balance sheet as of December 31, 2019, with no impact to its consolidated statements of operations and comprehensive loss for the year ended December 31, 2019. The prior period amounts have not been adjusted and continue to be reported in accordance with the Company's historical accounting under Topic 840, *Leases* ("Topic 840"). See further discussion of the impact of the adoption below in *Accounting Pronouncements Recently Adopted*.

The Company determines if an arrangement is a lease at inception. For leases where the Company is the lessee, ROU assets represent the Company's right to use the underlying asset for the term of the lease and the lease liabilities represent an obligation to make lease payments arising from the lease. Certain lease agreements contain tenant improvement allowances, rent holidays and rent escalation provisions, all of which are considered in determining the ROU assets and lease liabilities. The Company begins recognizing rent expense when the lessor makes the underlying asset available for use by the Company. Lease liabilities are recognized at the lease commencement date based on the present value of the future lease payments over the lease term. Lease renewal periods are considered on a lease-by-lease basis in determining the lease term. The interest rate the Company uses to determine the present value of future lease payments is the Company's incremental borrowing rate because the rate implicit in the Company's leases is not readily determinable. The incremental borrowing rate is a hypothetical rate for collateralized borrowings in economic environments where the leased asset is located based on credit rating factors. The ROU asset is determined based on the lease liability initially established and adjusted for any prepaid lease payments and any lease incentives received. The lease term to calculate the ROU asset and related lease liability includes options to extend or terminate the lease when it is reasonably certain that the Company will exercise the option. Certain leases contain variable costs, such as common area maintenance, real estate taxes or other costs. Variable lease costs are expensed as incurred on the consolidated statements of operations and comprehensive loss.

Operating leases are included in the ROU assets, accrued liabilities and lease liabilities, non-current on the consolidated balance sheets. The Company has no finance leases.

### Loss Contingencies

The Company is involved in various lawsuits, claims and proceedings that arise in the ordinary course of business. The Company records a liability for these when it believes it is probable that it has incurred a loss, and the Company can reasonably estimate the loss. If the Company determines that a material loss is reasonably possible and the loss or range of loss can be estimated, the Company discloses the possible loss in the notes to the consolidated financial statements. The Company regularly evaluates current information to determine whether it should adjust a recorded liability or record a new one. Significant judgment is required to determine both the probability and the estimated amount.

### Redeemable Convertible Preferred Stock Warrant Liability

The Company classifies the redeemable convertible preferred stock warrant as a liability on the consolidated balance sheets. The redeemable convertible preferred stock warrant liability is subject to remeasurement at each balance sheet date, and any change in fair value is recognized as gain or loss within earnings. The Company will continue to adjust the liability for the changes in fair value until the earlier of the exercise or expiration of the warrant.

Immediately prior to the completion of the Company's planned IPO, all of the Company's outstanding redeemable convertible preferred stock warrants will be exercised and the fair value of the liability at that time will also be reclassified into the Company's common stock and additional paid-in capital.

### Stock-Based Compensation

The Company grants stock options to employees and nonemployees. The Company measures stock options based on their estimated grant date fair values, which the Company determines using the Black-Scholes option-pricing model. The Company records the resulting expense in the consolidated statements of operations and comprehensive loss over the period of service required to vest in the award, which is generally four or five years. The Company accounts for forfeitures as they occur.

The Company's RSU grants to employees vest upon the satisfaction of both a service condition and a liquidity condition. The service condition for these awards is satisfied over four or five years. The liquidity condition is satisfied upon the occurrence of a qualifying event, defined as a change of control transaction or an IPO. The Company measures RSUs granted to employees based on the fair market value of its common stock on the grant date. The Company has not recorded any stock-based compensation expense for RSUs as of December 31, 2019, because a qualifying event has not occurred. If a qualifying event occurs in the future, the Company will record cumulative stock-based compensation expense using the accelerated attribution method for those RSUs for which the service condition has been satisfied prior to the qualifying event, and the Company will record the remaining unrecognized stock-based compensation expense over the remainder of the requisite service period.

### Valuation of Common Stock, Redeemable Convertible Preferred Stock and Redeemable Convertible Preferred Stock Warrant

The Company determined the fair value of common stock, redeemable convertible preferred stock and redeemable convertible preferred stock warrant in 2017, 2018 and 2019 based on the market approach under which the Company's equity value is estimated by applying valuation multiples derived from the observed valuation multiples of comparable public companies to management forecasted financial results.

The Company used the Probability Weighted Expected Return Method ("PWERM") to allocate the Company's equity value among outstanding common stock, redeemable convertible preferred stock, redeemable convertible preferred stock warrant and equity awards. The Company applied the PWERM by first defining the range of potential future liquidity outcomes, such as an IPO, and then allocating its value to outstanding stock, warrant and equity awards based on the relative probability that each outcome will occur. The valuation and allocation approaches involve the use of estimates, judgments and assumptions that are highly complex and subjective, such as those regarding expected future revenue and discount rates, valuation multiples, the selection of comparable public companies and the probability of future events. Changes in any or all of these estimates and assumptions, or the relationships between these assumptions, impact the Company's valuation as of each valuation date and may have a material impact on the valuation of the Company's common stock, redeemable convertible preferred stock and redeemable convertible preferred stock warrant.

### Fair Value Measurement

The Company applies fair value accounting for all financial assets and liabilities and non-financial assets and liabilities that are recognized or disclosed at fair value in the financial statements on a recurring basis. The Company defines fair value as the price that would be received from selling an asset or paid to transfer a liability in an orderly transaction between market participants at the

measurement date. When determining fair value measurements for assets and liabilities, the Company considers the principal or most advantageous market in which it would transact and the market-based risk measurements or assumptions that market participants would use in pricing the asset or liability, such as risks inherent in valuation techniques, transfer restrictions and credit risk. Fair value is estimated by applying the following hierarchy, which prioritizes the inputs used to measure fair value into three levels and bases the categorization within the hierarchy upon the lowest level of input that is available and significant to the fair value measurement:

Level 1— Quoted	prices in active markets for identical assets or liabilities.
Level 2— Observable	inputs other than quoted prices in active markets for identical assets and liabilities, quoted prices for identical or similar assets or liabilities in inactive markets, or other inputs that are observable or can be corroborated by observable market data for substantially the full term of the assets or liabilities.
Level 3— Inputs	that are generally unobservable and typically reflect management's estimate of assumptions that market participants would use in pricing the asset or liability.

#### Income Taxes

The Company accounts for income taxes using the asset and liability method, under which deferred tax liabilities and assets are recognized for the expected future tax consequences of temporary differences between consolidated financial statement carrying amounts and the tax basis of assets and liabilities and net operating loss and tax credit carryforwards. Valuation allowances are established when necessary to reduce deferred tax assets to the amount expected to be realized.

The Company determines whether it is more likely than not that a tax position will be sustained upon examination. If it is not more likely than not that a position will be sustained, no amount of benefit attributable to the position is recognized. The tax benefit to be recognized of any tax position that meets the more likely than not recognition threshold is calculated as the largest amount that is more than 50% likely of being realized upon resolution of the contingency.

It is the Company's policy to include penalties and interest expense related to income taxes as a component of interest and other income (expense), net as necessary.

### Foreign Currency Risk

The functional currency of the Company's foreign subsidiaries is the local currency for operating entities with employees and is the U.S. dollar for holding companies and pass-through entities. The assets and liabilities of its non-U.S. dollar functional currency subsidiaries are translated into U.S. dollars using exchange rates in effect at the end of each period. Revenue and expenses for its foreign subsidiaries are translated using rates that approximate those in effect during the period. Foreign currency translation adjustments are not material.

Transactions on the Company's marketplace occur in various foreign currencies that are processed by its PSPs. These transactions are collected on a regular basis and are converted to U.S. dollars or euros within the short period of time between the recognition of revenue and cash collection on a regular basis, which limits the Company's exposure to foreign currency risk.

Amounts payable to merchants are denominated in USD or in merchants' local currencies, which increases the Company's exposure to foreign currency risk. As of December 31, 2019, the Company only supported Chinese yuan as a local currency and 47% of the merchants payable amount was denominated in Chinese yuan.

Transaction gains and losses, including intercompany transactions denominated in a currency other than the functional currency of the entity involved are included in interest and other income (expense), net on the consolidated statements of operations and comprehensive loss. For the years ended December 31, 2018 and 2019, the Company recorded immaterial cumulative translation gains and losses and recognized net losses resulting from foreign exchange transactions of \$7 million and \$8 million, respectively.

### Comprehensive Loss

Comprehensive loss is comprised of two components: net loss and other comprehensive income (loss). Other comprehensive income (loss) consists of foreign currency translation and unrealized gain or loss on marketable securities. The Company's other comprehensive income (loss) was not material for all periods presented.

### Net Loss Per Share Attributable to Common Stockholders

The Company follows the two-class method when computing net loss per share as the Company has issued shares that meet the definition of participating securities. All series of redeemable convertible preferred stock are participating securities. The two-class method determines net loss per share for each class of common stock and participating securities according to dividends declared or accumulated and participation rights in undistributed earnings. The Company did not allocate net loss to redeemable convertible preferred stock because the holders of such shares are not contractually obligated to share in losses.

Basic net loss per share attributable to common stockholders is computed by dividing the net loss attributable to common stockholders by the weighted-average number of common stock outstanding for the period.

Diluted net loss per share attributable to common stockholders is computed by dividing the net loss attributable to common stockholders by the weighted-average number of common stocks, including potential dilutive common stocks assuming the dilutive effect of outstanding common stock options, RSUs, redeemable convertible preferred stock and warrants to purchase common stock and redeemable convertible preferred stock.

For periods in which the Company has reported net losses, diluted net loss per share attributable to common stockholders is the same as basic net loss per share attributable to common stockholders, as inclusion of all potential common stocks outstanding would have an antidilutive effect.

### Accounting Pronouncements Recently Adopted

In May 2014, the FASB issued Topic 606 which amended the existing accounting standards for revenue recognition. Topic 606 establishes principles for recognizing revenue upon the transfer of promised goods or services to customers in an amount that reflects the expected consideration to be received in exchange for those goods or services. In March 2016, the FASB issued ASU No. 2016-08, *Revenue from Contracts with Customers (Topic 606): Principal versus Agent Considerations (Reporting Revenue Gross versus Net)* ("ASU 2016-08"), which clarifies the implementation guidance on principal versus agent considerations. The guidance includes indicators to assist an entity in determining whether it controls a specified good or service before it is transferred to the customer. The new standards further require new disclosures about contracts with customers, including the significant judgments the company has made when applying the guidance. The Company adopted the new guidance as of January 1, 2018, utilizing the modified retrospective method of transition. Upon the adoption of Topic 606, the Company recorded the costs of free products provided to users and the

costs of certain refunds as a reduction of revenue in the consolidated statements of operations and comprehensive loss. These costs were previously recorded in cost of revenue under Topic 605. There was no adjustment to accumulated deficit on January 1, 2018.

In February 2016, the FASB issued Topic 842, which generally requires lessees to recognize operating and finance lease liabilities and corresponding right-of-use assets on the balance sheet and to provide enhanced disclosures surrounding the amount, timing and uncertainty of cash flows arising from leasing arrangements. The Company adopted the new guidance as of January 1, 2019, using the modified retrospective transition approach by applying the new standard to all leases existing at the date of initial application and not restating comparative periods. Results and disclosure requirements for reporting periods beginning after January 1, 2019 are presented under Topic 842, while prior period amounts have not been adjusted and continue to be reported in accordance with the Company's historical accounting under Topic 840.

The Company elected the package of practical expedients permitted under the transition guidance, which allowed for the carry-forward of the Company's historical lease classification and assessment on whether a contract is or contains a lease. The Company also elected to combine lease and non-lease components and to keep leases with an initial term of 12 months or less off the balance sheet and recognize the associated lease payments in the consolidated statements of operations and comprehensive loss on a straight-line basis over the lease term.

Upon adoption, the Company recognized right-of-use assets of approximately \$49 million and lease liabilities for operating leases of approximately \$59 million for operating leases on the Company's consolidated balance sheet, with no impact to its consolidated statements of operations and comprehensive loss. The Company did not have finance leases.

In January 2016, the FASB issued ASU No. 2016-01, *Recognition and Measurement of Financial Assets and Financial Liabilities* ("ASU 2016-01"). The amendments in ASU 2016-01 supersede the guidance to classify equity securities with readily determinable fair values into different categories and require equity securities to be measured at fair value with changes in the fair value recognized through net income. The amendments allow equity investments that do not have readily determinable fair values to be remeasured at fair value either upon the occurrence of an observable price change or upon identification of an impairment. The Company adopted ASU 2016-01 as of January 1, 2018. The adoption did not have a material impact on the Company's consolidated financial statements.

In June 2018, the FASB issued ASU No. 2018-07, *Improvements to Nonemployee Share-Based Payment Accounting* ("ASU 2018-07"), which simplifies the accounting for share-based payments to nonemployees by aligning it with the accounting for share-based payments to employees, with certain exceptions. The Company elected to early adopt ASU 2018-07 on January 1, 2019. The adoption had no impact on the Company's consolidated financial statements.

### Recent Accounting Pronouncements Not Yet Adopted

In June 2016, the FASB issued ASU No. 2016-13, *Financial Instruments—Credit Losses (Topic 326): Measurement of Credit Losses on Financial Instruments* ("ASU 2016-13"), which provides new guidance on the measurement of credit losses for financial assets measured at amortized cost, which includes accounts receivable and available-for-sale debt securities. The new guidance replaces the existing incurred loss impairment model with an expected loss methodology, which will result in more timely recognition of credit losses. ASU 2016-13 is effective for annual periods, and interim periods within those years, beginning after December 15, 2019 for public companies. The adoption of this standard will not have a material impact on the Company's consolidated financial statements.

In August 2018, the FASB issued ASU No. 2018-13, Fair Value Measurement (Topic 820)— Disclosure Framework—Changes to the Disclosure Requirements for Fair Value Measurement ("ASU 2018-13"). The amendments on changes in unrealized gains and losses, the range and weighted-average of significant unobservable inputs used to develop Level 3 fair value measurements, and the narrative description of measurement uncertainty should be applied prospectively for only the most recent interim or annual period presented in the initial fiscal year of adoption. All other amendments should be applied retrospectively to all periods presented upon their effective date. ASU 2018-13 is effective for annual periods, and interim periods within those years, beginning after December 15, 2019 for all entities. The adoption of this standard will not have a material impact on the Company's consolidated financial statements.

In August 2018, the FASB issued ASU No. 2018-15, *Internal-Use Software* (Subtopic 350-40) ("ASU 2018-15"), which clarifies the accounting for implementation costs in a cloud computing arrangement. ASU 2018-15 aligns the requirements for capitalizing implementation costs incurred in a hosting arrangement that is a service contract with the requirements for capitalizing implementation costs incurred to develop or obtain internal-use software. The standard is effective for annual periods, and interim periods within those years, beginning after December 15, 2019 for public companies. The adoption of this standard will not have a material impact on the Company's consolidated financial statements.

In December 2019, the FASB issued ASU No. 2019-12, *Income Taxes (Topic 740): Simplifying the Accounting for Income Taxes (*"ASU 2019-12"), which simplifies the accounting for income taxes. The standard is effective for annual periods, and interim periods within those years, beginning after December 15, 2020 for public companies. Early adoption is permitted. The Company does not expect the adoption of this standard to have a material impact on the Company's consolidated financial statements.

The Company has reviewed other recent accounting pronouncements and concluded they are either not applicable to the business or no material impact is expected on the consolidated financial statements as a result of future adoption.

#### 3. FINANCIAL INSTRUMENTS AND FAIR VALUE MEASUREMENT

The Company's financial instruments consist of cash equivalents, marketable securities, funds receivable, accounts payable, accrued liabilities, merchants payable and redeemable convertible preferred stock warrant liability. Cash equivalents' carrying value approximates fair value at the balance sheet dates, due to the short period of time to maturity. Marketable securities are recorded at fair value. Funds receivable, accounts payable, accrued liabilities and merchants payable carrying values approximate fair value due to the short time to the expected receipt or payment date. The redeemable convertible preferred stock warrant liability is carried at fair value.

Assets and liabilities recorded at fair value on a recurring basis in the consolidated balance sheets consisting of cash equivalents, marketable securities and the redeemable convertible preferred stock warrant liability are categorized based upon the level of judgment associated with the inputs used to measure their fair values. Fair value is defined as the price that would be received from selling an asset or paid to transfer a liability in an orderly transaction between market participants at the measurement date.

Financial assets and liabilities subject to fair value measurements on a recurring basis and the level of inputs used in such measurements are as follows (in millions):

	December 31, 2018							
		Total		Level 1		Level 2		Level 3
Financial assets:								
Cash equivalents:								
Money market funds	\$	133	\$	133	\$	_	\$	<u> </u>
Total cash equivalents	\$	133	\$	133	\$		\$	<u> </u>
Marketable securities:								
U.S. Treasury notes	\$	60	\$	_	\$	60	\$	_
U.S. Treasury bills		35		_		35		_
Commercial paper		69		_		69		_
Corporate bonds		138		_		138		_
Total marketable securities	\$	302	\$	<u> </u>	\$	302	\$	<u> </u>
Total financial assets	\$	435	\$	133	\$	302	\$	_
Financial liabilities:								
Redeemable convertible preferred stock warrant								
liability	\$	124	\$	_	\$	_	\$	124
Total financial liabilities	\$	124	\$	_	\$	_	\$	124

	December 31, 2019						
	Total	Level 1	Level 2	Level 3			
Financial assets:							
Cash equivalents:							
Money market funds	\$ 2	\$ 2	\$ -	\$ -			
U.S. Treasury bills	16	_	16	_			
Commercial paper	<u>      6                              </u>	<del></del>	6				
Total cash equivalents	\$ 24	\$ 2	\$ 22	\$ -			
Marketable securities:							
U.S. Treasury bills	\$140	\$ -	\$ 140	\$ -			
Commercial paper	65	_	65	_			
Corporate bonds	_129		129				
Total marketable securities	\$334	\$ -	\$ 334	\$ -			
Total financial assets	\$358	\$ 2	\$ 356	<del>\$ -</del>			
Financial liabilities:	<del></del>	<del></del>	<del></del>	<del></del>			
Redeemable convertible preferred stock warrant							
liability	\$127	\$ -	\$ -	\$ 127			
Total financial liabilities	\$127	<u>\$ -</u>	<del>\$ -</del>	\$ 127			

The Company classifies cash equivalents and marketable securities within Level 1 or Level 2 because the Company uses quoted market prices or alternative pricing sources and models utilizing market observable inputs to determine their fair value.

The following table summarizes the contractual maturities of the Company's marketable securities (in millions):

	December 31,							
	2018			2019				
	Estimated Fair				i1 04		timated	
	Amortize	d Cost		/alue	Amort	ized Cost	<u>Fa</u>	ir Value
Due within one year	\$	262	\$	262	\$	300	\$	300
Due after one year through five years		40		40		34		34
Total marketable securities	\$	302	\$	302	\$	334	\$	334

All of the Company's available-for-sale marketable securities are subject to a periodic impairment review. The Company did not identify any of its available-for-sale marketable securities as other-than- temporarily impaired in any of the periods presented. Additionally, the unrealized net gain and net loss on available-for-sale marketable securities as of December 31, 2018 and 2019 were immaterial.

The following table sets forth a summary of the changes in the estimated fair value of the Company's Level 3 financial liabilities, consisting solely of redeemable convertible preferred stock warrant liability (see Note 7), which is measured at fair value on a recurring basis (in millions):

	December 31,			
		2018		2019
Balance at beginning of period	\$	124	\$	124
Remeasurement of redeemable convertible preferred stock warrant liability		<u> </u>		3
Balance at end of period	\$	124	\$	127

The significant unobservable inputs used in the fair value measurement of the redeemable convertible preferred stock warrant liability is the fair value of the underlying redeemable convertible preferred stock at the valuation date. Generally, increases (decreases) in the fair value of the underlying redeemable convertible preferred stock would result in a directionally similar impact to the fair value measurement.

### 4. BALANCE SHEET COMPONENTS

Property and Equipment, Net

Property and equipment, net consisted of the following (in millions):

		December 31,		
	20	)18	20	019
Computers, software, servers and network equipment	\$	23	\$	25
Furniture and fixtures		2		3
Leasehold improvements		22		28
Total property and equipment		47		56
Accumulated depreciation and amortization		(14)		(22)
Total property and equipment, net	\$	33	\$	34

Depreciation and amortization expense related to property and equipment for the years ended December 31, 2017, 2018 and 2019 was \$4 million, \$8 million and \$10 million, respectively.

### Accrued Liabilities

Accrued liabilities consist of the following (in millions):

	Dec	December 31,			
	2018		2019		
Vendor services	\$ 56	\$	68		
Deferred revenue	<del>-</del>		24		
Other	69	ı	120		
Total accrued liabilities	\$ 125	\$	212		

### 5. OPERATING LEASES

The Company leases its facilities and data center co-locations under operating leases with various expiration dates through 2025.

The components of the Company's lease costs for the year ended December 31, 2019 were as follows (in millions):

Operating lease costs	\$ 12
Short-term lease costs	1
Variable costs	1
Total	\$ 14

Rent expense under Topic 840 was \$10 million for the year ended December 31, 2018. As of December 31, 2019, the Company's consolidated balance sheet included right-of-use assets in the amount of \$41 million and lease liabilities in the amount of \$8 million in accrued liabilities and \$42 million in lease liabilities, non-current.

As of December 31, 2019, the weighted-average remaining lease term was 5 years and the weighted-average discount rate used to determine the net present value of the lease liabilities was 6%.

Supplemental cash flow information for the Company's operating leases for the year ended December 31, 2019 were as follows (in millions):

Cash paid for amounts included in the measurement of lease liabilities:	
Operating cash flows from operating leases	\$ 13
Right-of-use assets obtained in exchange for new lease liabilities	1

The maturities of the Company's operating lease liabilities as of December 31, 2019 are as follows (in millions):

\$ 11
11
11
10
10
 5
58
(8)
\$ 50
\$

The aggregate future minimum lease obligations under Topic 840 for all operating lease agreements with a minimum term of one year as of December 31, 2018, are as follows (in millions):

Year ending December 31,	
2019	\$ 13
2020	12
2021	11
2022	11
2023	10
Thereafter	15
Total	\$ 72

### **6. COMMITMENTS AND CONTINGENCIES**

### Letter of Credit

The Company has a standby letter of credit with a bank in the amount of \$10 million in conjunction with the second amendment to the lease of the Company's corporate headquarters in San Francisco, California. As of December 31, 2019, no amounts had been drawn down and the Company was in compliance with the covenants under this letter of credit.

### **Purchase Obligations**

During 2019, the Company amended a marketing arrangement which increased the total commitment under the agreement and extended the arrangement to July 31, 2021. As of December 31, 2019, the remaining commitment under the amended agreement is \$18 million.

Effective September 1, 2019, the Company entered into an amendment to a colocation and cloud services arrangement committing the Company to make payments of \$120 million for services over 3 years. As of December 31, 2019, the remaining commitment under this amended agreement is \$99 million.

In August 2018, the Company entered into a separate agreement with a different vendor for colocation and cloud services committing the Company to make payments of \$28 million for services over 3 years. As of December 31, 2019, the remaining commitment under this agreement is \$12 million.

### Legal Contingencies

As of December 31, 2019, there were no legal contingency matters, either individually or in aggregate, that would have a material adverse effect on the financial position, results of operations, or cash flows of the Company. Given the unpredictable nature of legal proceedings, the Company bases its estimate on the information available at the time of the assessment. As additional information becomes available, the Company reassesses the potential liability and may revise the estimate.

### 7. REDEEMABLE CONVERTIBLE PREFERRED STOCK WARRANT

In August 2016, the Company issued a warrant to purchase 986,640 shares of the Company's Series B redeemable convertible preferred stock ("Series B warrant") at an exercise price of \$0.0001 per share. The Company accounted for the Series B warrant as a liability upon issuance. The Series B warrant expires on the day preceding a liquidity event, which is defined as an IPO or change of control event. The fair value of Series B warrant liability as of December 31, 2018 and 2019 was \$124 million and \$127 million, respectively. The Series B warrant remains outstanding as of December 31, 2019.

### 8. REDEEMABLE CONVERTIBLE PREFERRED STOCK

In March 2019, the Company's Board of Directors ("Board of Directors") approved an amendment and restatement to the Company's Certificate of Incorporation, as amended and restated, which increased the total authorized shares of redeemable convertible preferred stock to 44,346,293 shares, decreased the authorized shares of Series C redeemable convertible preferred stock ("Series C") to 7,108,490, decreased the authorized shares of Series G redeemable convertible preferred stock ("Series G") to 1,687,319, and authorized 2,064,009 shares for a new series of convertible preferred stock to be designated as Series H redeemable convertible preferred stock ("Series H").

In March 2019, the Company raised gross proceeds of approximately \$160 million from the sale of 943,548 shares of Series H at \$169.5729 per share. Furthermore, the Company granted the lead investor the right to purchase up to an additional 884,575 shares of Series H for \$169.5729 per share. The right is considered an embedded feature of the Series H shares sold to this lead investor and, therefore, does not have a standalone value. The right expired unexercised on March 31, 2020.

Redeemable convertible preferred stock consists of the following (in millions, except share amounts):

		December 31, 2018			December 31, 2019		
	Shares Authorized	Shares Issued and Outstanding	Liquidation Preference	Shares Authorized	Shares Issued and Outstanding	Liquidation Preference	
Series A	6,437,367	6,437,367	\$ 23	6,437,367	6,367,367	\$ 22	
Series B	6,883,083	5,896,443	23	6,883,083	5,896,443	23	
Series C	7,966,091	7,108,490	45	7,108,490	7,108,490	45	
Series D	5,519,533	5,519,533	138	5,519,533	5,519,533	138	
Series E	8,324,560	8,324,560	514	8,324,560	8,324,560	514	
Series F	6,321,932	6,321,932	410	6,321,932	6,321,932	410	
Series G	1,900,000	1,687,319	227	1,687,319	1,687,319	227	
Series H	_	_	_	2,064,009	943,548	240	
Total redeemable convertible preferred stock	43,352,566	41,295,644	\$ 1,380	44,346,293	42,169,192	\$ 1,619	

Series A — In June 2019, the Company entered into an agreement with one of its stockholders to repurchase 70,000 shares of the Company's Series A redeemable convertible preferred stock ("Series A") for a total of \$7 million in cash. The amount paid in excess of the carrying value of Series A is considered a deemed dividend and is reflected as such in the calculation of net loss attributable to common stockholders. The repurchased shares were constructively retired and have been removed from both the issued and outstanding number of shares on the 2019 consolidated balance sheet and consolidated statement of stockholders' deficit.

Series C — In August 2017, the Company entered into agreements with two stockholders to repurchase a combined 857,601 shares of the Company's Series C for a total of \$46 million in cash. The amount paid in excess of the carrying value of Series C is considered a deemed dividend and is reflected as such in the calculation of net loss attributable to common stockholders. The repurchased shares were constructively retired and have been removed from both the issued and outstanding number of shares in the consolidated statement of stockholders' deficit as of December 31, 2017.

The holders of redeemable convertible preferred stock have various significant rights and preferences as follows:

Dividend Provisions — Holders of Series A, Series B redeemable convertible preferred stock ("Series B"), Series C, Series D redeemable convertible preferred stock ("Series E"), Series F redeemable convertible preferred stock ("Series E"), Series G and Series H are entitled to receive noncumulative dividends, prior and in preference to any declaration of dividends on common stock, at the per annum rate of \$0.280464, \$0.316, \$0.502128, \$2.000168, \$4.9396, \$5.190688, \$10.76264 and \$13.565832 per share, respectively, when and if declared by the Board of Directors, except for dividends on common stock payable solely in common stock. The holders of redeemable convertible preferred stock are also entitled to participate in dividends on common stock, when and if declared by the Board of Directors, based on the number of shares of common stock that would be held on an as-if converted basis, except for dividends on common stock payable solely in common stock. No dividends on redeemable convertible preferred stock or common stock have been declared by the Board of Directors from inception through December 31, 2019.

Liquidation Preference — In the event of any liquidation, dissolution, or winding-up of the Company, including a merger, acquisition of the Company, or sale of assets, the stockholders of redeemable convertible preferred stock are entitled to receive, on a pari passu basis, their liquidation preference prior to any distributions to common stockholders, plus any declared but unpaid dividends. Upon the completion of the distribution to the stockholders of the redeemable convertible preferred stock, the remaining assets of the corporation available for distribution to stockholders will be distributed among the holders of common stock pro rata based on the number of shares of common stock held by each such holder.

Series A, Series B, Series C, Series D, Series E, Series F and Series G each have a liquidation preference of \$3.5058, \$3.95, \$6.2766, \$25.0021, \$61.7450, \$64.8836 and \$134.533 per share, respectively. The liquidation preference of Series H is contingent on the date of the liquidation event. If the liquidation event occurs within three calendar years from the date of issuance of Series H, the holders of Series H will be entitled to receive \$254.3594 per share; if the event occurs during the fourth calendar year from the date of issuance of Series H, the holders of Series H will be entitled to receive \$296.7526 per share; and, if the event occurs after four calendar years from the date of issuance of Series H, the holders of Series H will be entitled to receive \$339.1458 per share.

If the liquidity event consists of an IPO and occurs within three calendar years from the date of issuance of Series H, then, if the value of the common stock issued upon conversion of Series H is

less than 150% of the aggregate consideration paid for the Series H pursuant to the Series H purchase agreement, the holders of Series H receive an additional number of shares of common stock such that the value of the common stock issued upon conversion of Series H in the IPO (at the IPO price) plus the value of the additional shares of common stock shall equal 150% of the aggregate consideration paid for the Series H pursuant to the Series H purchase agreement. If the liquidity event consists of an IPO and occurs during the fourth calendar year from the date of issuance of Series H, then, if the value of the common stock issued upon conversion of Series H is less than 175% of the aggregate consideration paid for the Series H pursuant to the Series H purchase agreement, the holders of Series H receive an additional number of shares of common stock such that the value of the common stock issued upon conversion of Series H in the IPO (at the IPO price) plus the value of the additional shares of common stock shall equal 175% of the aggregate consideration paid for Series H pursuant to the Series H purchase agreement. If the liquidity event consists of an IPO and occurs after the fourth calendar year from the date of issuance of Series H, then, if the value of the common stock issued upon conversion of Series H is less than 200% of aggregate consideration paid for the Series H pursuant to the Series H purchase agreement, the holders of Series H receive an additional number of shares of common stock such that the value of the common stock issued upon conversion of Series H in the IPO (at the IPO price) plus the value of the additional shares of common stock shall equal 200% of the aggregate consideration paid for the Series H pursuant to the Series H purchase agreement.

Redemption — The holders of the redeemable convertible preferred stock have no voluntary rights to redeem the shares. The sale, transfer or other disposition of substantially all of the Company's assets would constitute a redemption event as would other deemed liquidation events such as a change in control, whether by merger, consolidation or otherwise. If such a deemed liquidation event occurs, all of the holders of equally and more subordinated equity instruments are not explicitly entitled to receive the same form of consideration. Although the redeemable convertible preferred stock is not mandatorily or currently redeemable, any of these events would constitute a redemption event that is outside of the control of the Company.

The Company classifies its redeemable convertible preferred stock as temporary equity because it may become redeemable due to certain change in control events that are outside the Company's control, including a merger, acquisition, or sale of assets of the Company. The Company has not adjusted the carrying value of the redeemable convertible preferred stock to its redemption value because redemption was not probable as of the balance sheet dates presented. The Company will adjust the carrying value of the redeemable convertible preferred stock to its redemption value if redemption becomes probable in the future.

Conversion Rights — Each share of redeemable convertible preferred stock is convertible, after notice to the Company one business day after the applicable Hart-Scott-Rodino Antitrust Improvements Act of 1976, as amended ("HSR"), waiting period has expired or been terminated; provided, however, that no notice needs to be sent if the redeemable convertible preferred stockholder has verified that the conversion is not subject to HSR. As of December 31, 2019, each share of redeemable convertible preferred stock would convert into common stock on a one-for-one basis. Each share of redeemable convertible preferred stock automatically converts into the number of shares of common stock into which such shares are convertible at the then-effective conversion ratio upon the closing of a firm commitment underwritten public offering in which the pre-money valuation is at least \$8.75 billion and the gross cash proceeds are no less than \$500 million (a "Qualified Public Offering"). Prior to a Qualified Public Offering, each share of Series A, Series B, Series C and Series D automatically converts into common stock at the then-effective conversion ratio upon vote or written consent or agreement of the holders of a majority of the Series A, Series B, Series C and Series D, voting together as a single class and on an as-converted basis; each share of Series E automatically converts into shares of common stock at the then-effective conversion ratio upon vote or written consent or agreement of the holders of a majority of Series E, voting as a separate series and on an

as-converted basis; each share of Series F automatically converts into shares of common stock at then effective conversion ratio upon vote or written consent or agreement of the holders of a majority of the Series F, voting as a separate series and on an as-converted basis, which such vote must include one or more particular redeemable convertible preferred stockholders; each share of Series G automatically converts into shares of common stock at the then-effective conversion ratio upon vote or written consent or agreement of the holders of a majority of Series G, voting as a separate series, which such vote must include one or more particular redeemable convertible preferred stockholders; and, each share of Series H automatically converts into shares of common stock at the then-effective conversion ratio upon vote or written consent or agreement of the holders of a majority of Series H, voting as a separate series and on an as-converted basis.

Antidilution Provisions — The conversion price of the redeemable convertible preferred stock is subject to adjustment to mitigate dilution in the event that the Company issues additional shares at a purchase price less than the applicable conversion price.

Voting Rights — The holder of each share of redeemable convertible preferred stock has voting rights equal to the number of shares of common stock into which it is convertible and votes together as one class with the common stock; provided, however, that the holders of shares of Series D, Series E, Series F, Series G and Series H stock do not vote in any elections of directors. Each share of common stock is entitled to one vote.

As long as at least 1 million shares (as adjusted for any stock splits, stock dividends, combinations, subdivision or recapitalizations) of each class of Series A and Series B, respectively, and as long as any shares of Series C, remain outstanding, the holders of each such class of Series A, Series B and Series C shall be entitled to elect (voting as separate classes) one director of the Company for each respective class of redeemable convertible preferred stock. The holders of outstanding common stock (voting as a separate class) shall be entitled to elect two directors of the Company. The holders of outstanding common stock together with the holders of Series A, Series B and Series C (voting together as a single class using an as-converted basis for the redeemable convertible preferred stock) shall be entitled to elect one additional director.

Protective Provisions — So long as at least 10,839,913 shares of all redeemable convertible preferred stock remain outstanding (as adjusted for any stock splits, stock dividends, combinations, subdivision or recapitalizations), the Company must obtain approval from a majority of the then outstanding shares of redeemable convertible preferred stock in order to (i) consummate a liquidation event or effect any other merger or consolidation; (ii) amend, alter, repeal, or waive any provision of the Company's amended and restated certificate of incorporation or bylaws; (iii) authorize or issue any equity security having a preference over, or being on a parity with, any series of outstanding redeemable convertible preferred stock with respect to dividends, liquidation or redemption (other than authorized but unissued shares of Series H); (iv) redeem, purchase or otherwise acquire any shares of redeemable convertible preferred stock or common stock, subject to certain exceptions; (v) change the authorized number of members of the Board of Directors; (vii) pay or declare any dividend on any shares of common stock or redeemable convertible preferred stock, subject to certain exceptions; (vii) effect an IPO of common stock or (viii) create, or hold capital stock in, any subsidiary that is not wholly owned by the Company, unless approved by the Board of Directors.

So long as any shares of Series E are outstanding, the Company must obtain approval from a majority of the then outstanding shares of Series E in order to (i) amend, alter, repeal, waive or change the powers, preferences or rights of the Series E, or (ii) increase or decrease (other than by redemption or conversion) the number of authorized shares of Series E.

So long as any shares of Series F are outstanding, the Company must obtain approval from a majority of the then outstanding shares of Series F, which such vote must include one or more

particular redeemable convertible preferred stockholders, in order to (i) amend, alter, repeal, waive or change the powers, preferences or rights of the Series F, (ii) increase or decrease (other than by redemption or conversion) the number of authorized shares of Series F, or (iii) reclassify, alter or amend the Series G, if such reclassification, alteration or amendment would render Series G senior to Series F in respect of any dividend, liquidation or redemption right, preference or privilege or reclassify, alter or amend any of the Series A, Series B, Series C, Series D, Series E, Series F or Series G (the "prior preferred series"), if such reclassification, alteration or amendment would render any of the prior redeemable convertible preferred series senior to or pari passu with Series F in respect of any dividend, liquidation or redemption right, preference or privilege.

So long as any shares of Series G are outstanding, the Company must obtain approval from a majority of the then outstanding shares of Series G, which such vote must include one or more particular redeemable convertible preferred stockholders, in order to (i) amend, alter, repeal, waive or change the powers, preferences or rights of the Series G, (ii) increase or decrease (other than by redemption or conversion) the number of authorized shares of Series G, or (iii) reclassify, alter or amend Series F, if such reclassification, alteration or amendment would render Series F senior to Series G in respect of any dividend, liquidation or redemption right, preference or privilege or reclassify, alter or amend any of the prior preferred series, if such reclassification, alteration or amendment would render any of the prior preferred series senior to or pari passu with Series G in respect of any dividend, liquidation or redemption right, preference or privilege.

So long as any shares of Series H are outstanding, the Company must obtain approval from a majority of the then outstanding shares of Series H in order to (i) amend, alter, repeal, waive or change the powers, preferences or rights of the Series H, (ii) increase or decrease (other than by redemption or conversion) the number of authorized shares of Series H, or (iii) issue additional shares of Series H, unless such issuance is effected pursuant to the Series H purchase agreement.

### 9. COMMON STOCK AND STOCK-BASED COMPENSATION

The Company's Certificate of Incorporation, as amended and restated, authorizes the Company to issue up to 73,000,000 shares of common stock with par value of \$0.0001 per share. Each share of common stock is entitled to one vote. The holders of common stock are also entitled to receive dividends whenever funds are legally available and when declared by the Board of Directors, subject to the priority rights of holders of all classes of redeemable convertible preferred stock outstanding.

In 2014, the Company issued warrants to purchase 55,000 shares of common stock at an exercise price of \$1.49 per share. The common stock warrants expire in May 2024. The fair value of these warrants was recorded in additional paid in capital upon issuance. These warrants remain outstanding as of December 31, 2018 and 2019.

As of December 31, 2019, the Company had reserved shares of common stock for future issuance as follows:

	December 31, 2019
Redeemable convertible preferred stock, all series	42,169,192
Series B warrants.	986,640
Warrants to purchase common stock	55,000
Common stock options outstanding	8,032,296
Restricted stock units outstanding	4,383,735
Shares available for future grant of equity awards	443,634
Total	56,070,497

#### Related Party Transaction

In June 2019, the Company repurchased 142,974 shares of common stock from its Chief Executive Officer at \$121.70 per share. The incremental value between the repurchase price and the fair value of the common stock at the time of the transaction resulted in stock-based compensation expense of \$2 million for the year ended December 31, 2019.

### **Equity Incentive Plan**

In 2010, the Board of Directors approved the adoption of the 2010 Stock Plan (the "Plan"). As of December 31, 2019, the total shares reserved for issuance under the Plan is 14,577,861. The Plan provides for the grant of incentive and nonstatutory stock options and RSUs to employees, directors and consultants of the Company. Options granted under the Plan to employees continue to vest until the last day of employment and generally will vest over four or five years and expire 10 years from the date of grant. Employees generally forfeit their rights to exercise vested options after three months following their termination of employment or 6 or 12 months in the event of termination of services by reason of disability or death, respectively. Stock options to consultants are generally granted for services performed and vest according to an award-specific schedule as approved by the Board of Directors. When options are exercised subject to a repurchase right, the Company may buy back any unvested shares at their original exercise price in the event of an employee's termination prior to full vesting. As of December 31, 2019, there were no shares subject to this repurchase right. Under the terms of the Plan, any shares that expire, are forfeited, or are otherwise reacquired by the Company shall be added back to the number of shares then available for issuance.

The exercise price of incentive stock options granted under the Plan must be at least equal to 100% of the fair market value of the Company's common stock at the date of grant, as determined by the Board of Directors. The exercise price must not be less than 110% of the fair market value of the Company's common stock at the date of grant for incentive stock options granted to an employee that owns greater than 10% of the Company stock. Through December 31, 2019, no options have been granted to purchase stock at a price less than fair value as determined by the Board of Directors at the time of the grant.

The Company grants RSUs to employees, which generally expire 7 years from the date of grant and vest upon the achievement of both a service condition and a liquidity condition. The service condition for these awards is satisfied over four or five years. The liquidity condition is satisfied upon the occurrence of a qualifying event, defined as a change of control transaction or an IPO. Upon employee termination, RSUs that have satisfied the service condition remain outstanding until the earlier of a liquidity event, the expiration date, or the third anniversary of the service termination.

### Stock Option and RSU Activity

A summary of stock option and RSU activity under the Plan and related information is as follows (in millions, except share, term and per share data):

		0	ptions Outstand	RSUs Out	standing	
	Shares Available for Grant	Number of Options	Weighted- Average Exercise Pric	Weighted- Average Remaining Contractual Term (In e Years)	Number of RSUs	Weighted- Average Remaining Contractual Term (In Years)
Balances at December 31, 2018	638,570	8,112,859	\$ 2.4	5.16	3,335,076	5.32
Authorized	836,344	_	-	=	_	
Granted	(1,184,570)	_	-	=	1,184,570	
Common stock issued for services	(23,334)	_	-	_	_	
Exercised		(75,563)	4.29	9	_	
Forfeited or cancelled	140,911	(5,000)	0.0	1	(135,911)	
Repurchased common stock	35,713	_	-	=	_	
Balances at December 31, 2019	443,634	8,032,296	\$ 2.42	4.16	4,383,735	4.83

There were no options granted during the year ended December 31, 2017, 2018 or 2019. All options outstanding as of December 31, 2019 were fully vested. The aggregate intrinsic value of options exercised during the years ended December 31, 2017, 2018 and 2019 was \$12 million, \$48 million and \$8 million, respectively. The intrinsic value is the difference between the estimated fair value of the Company's common stock at the date of exercise and the exercise price for in-the-money options. The weighted-average grant date fair value of RSUs granted during the years ended December 31, 2017, 2018 and 2019 was \$82.84, \$113.16 and \$115.53 per share, respectively. The aggregate intrinsic value of options and RSUs outstanding as of December 31, 2019 was \$910 million and \$507 million, respectively.

### Stock-Based Compensation Expense

As of December 31, 2019, no stock-based compensation expense has been recognized for RSUs, because a qualifying event, defined as a change of control transaction or an IPO, had not yet occurred. In the year in which a qualifying event is completed, the Company will record cumulative stock-based compensation expense using the accelerated attribution method for those RSUs for which the service condition has been satisfied prior to the occurrence of the qualifying event. If the qualifying event had occurred on December 31, 2019, the Company would have recorded cumulative stock-based compensation expense of approximately \$283 million for the year ended December 31, 2019, and would recognize the remaining \$120 million of unrecognized stock-based compensation expense over a weighted-average period of approximately 1.82 years.

For the years ended December 31, 2017, 2018 and 2019, the Company recorded stock-based compensation expense of \$8 million, \$2 million and \$2 million, respectively.

### 10. INCOME TAXES

The components of loss before provision for income taxes are as follows (in millions):

	Year Ended		
	D	1,	
	2017	2018	2019
Domestic	\$204	\$205	\$130
Foreign	3	3	(2)
Loss before provision for income taxes	\$207	\$208	\$128

There has historically been no federal or state provision for income taxes because the Company has historically incurred operating losses and maintains a full valuation allowance against its net deferred tax assets. For the year ended December 31, 2019, the Company recognized an immaterial provision related to foreign income taxes.

The table below details the activity of the deferred tax asset valuation allowance (in millions):

	Beg	ance at inning Period	Ado	litions	Ded	uctions	E	ance at nd of eriod
Year ended December 31, 2017								
Deferred tax assets valuation allowance	\$	276	\$	2	\$	(73)	\$	205
Year ended December 31, 2018								
Deferred tax assets valuation allowance	\$	205	\$	50	\$	(1)	\$	254
Year ended December 31, 2019								
Deferred tax assets valuation allowance	\$	254	\$	29	\$	(1)	\$	282

The difference between income taxes computed at the statutory federal income tax rate and the provision for income taxes is attributable to the following (in millions):

	Years	Years Ended December 31			
	2017	2018	2019		
Federal tax (benefit) at statutory rate	\$ (70)	\$ (44)	\$ (27)		
Stock-based compensation	(1)	(5)	(1)		
Foreign rate differential	1	1	_		
Change in tax rate for the Tax Act	120	_	_		
Non-deductible warrant valuation	24	-	1		
Other	(1)	_	1		
Change in valuation allowance	(73)	48	27		
Total provision for income taxes	<del>\$ -</del>	\$ -	\$ 1		

The tax provision differs from the benefit that would result from applying statutory rates to losses before income taxes primarily due to the valuation allowance provided on the net deferred tax assets. Deferred income taxes reflect the net tax effects of (a) temporary differences between the carrying amounts of assets and liabilities for financial reporting purposes and (b) operating losses and tax credit carryforwards.

Deferred tax assets and liabilities are as follows (in millions):

	Decem	ber 31,
	2018	2019
Deferred tax assets:		
Net operating loss carryforwards	\$ 239	\$ 266
Lease liabilities	-	11
Reserves and accruals not currently deductible	9	8
Stock-based compensation	6	7
Total gross deferred tax assets	254	292
Less: valuation allowance	(254)	(282)
Total deferred tax assets, net of valuation allowance	_	10
Deferred tax liabilities:		
Property and equipment, including right-of-use assets		<u>(9</u> )
Total gross deferred tax liabilities	-	(9)
Net deferred tax (liabilities) assets	<u>\$</u> _	\$ 1

Due to a history of losses, the Company believes it is not more likely than not that its net deferred tax assets will be realized as of December 31, 2018 or 2019. Accordingly, the Company has established a full valuation allowance on its domestic net deferred tax assets. The Company's valuation allowance increased by \$27 million during the year ended December 31, 2019.

On December 22, 2017, the U.S. government enacted comprehensive tax legislation commonly referred to as the Tax Cuts and Jobs Act (the "Tax Act"). The Tax Act made broad and complex changes to the U.S. tax code, including, but not limited to, (1) reducing the U.S. federal corporate tax rate from 35 percent to 21 percent; (2) requiring companies to pay a one-time transition tax on certain unrepatriated earnings of foreign subsidiaries; (3) generally eliminating U.S. federal income taxes on dividends from foreign subsidiaries; (4) requiring a current inclusion of global intangible low-taxed income ("GILTI") in U.S. federal taxable income of certain earnings of controlled foreign corporations; (5) eliminating the corporate alternative minimum tax ("AMT") and changing how existing AMT credits can be realized; (6) creating the base erosion anti-abuse tax ("BEAT"), a new minimum tax; (7) creating a new limitation on deductible interest expense; and (8) changing rules related to uses and limitations of net operating loss carryforwards created in tax years beginning after December 31, 2017 changing classification of certain deferred tax assets as indefinite-lived.

The Securities and Exchange Commission ("SEC") staff issued Staff Accounting Bulletin No. 118, Income Tax Accounting Implications of the Tax Cuts and Jobs Act ("SAB 118"), which provides guidance on accounting for the tax effects of the Tax Act. SAB 118 provides a measurement period that should not extend beyond one year from the Tax Act enactment date for companies to complete the accounting under ASC 740, Income Taxes ("ASC 740"). During the fourth quarter of 2018, the Company completed its accounting for the Tax Act with no adjustments to the 2017 provisional amounts:

- Reduction of U.S. federal corporate tax rate to 21 percent: in 2017, the Company recorded a provisional income tax benefit of \$120 million from the reduction in its net deferred tax liabilities resulting from the revaluation, which was offset by a corresponding amount of valuation allowance.
- One-time mandatory transition tax: in 2017, the Company recorded a provisional transition tax obligation of \$0.

As of December 31, 2019, the Company had federal net operating loss carryforwards available to reduce future taxable income, if any, of \$885 million that begin to expire in 2030 and continue to expire through 2037 and \$324 million that have an unlimited carryover period. As of December 31, 2019, the Company had state net operating loss carryforwards available to reduce future taxable income, if any, of \$1.4 billion that begin to expire in 2030 and continue to expire through 2039.

Utilization of net operating loss carryforwards may be subject to future annual limitations due to the ownership change limitations provided by Section 382 of the Internal Revenue Code and similar state provisions. The federal net operating loss carryforwards and the state net operating loss carryforwards both begin expiring in 2030.

The Company had immaterial unrecognized tax benefits as of December 31, 2019, fully offset by a valuation allowance. These unrecognized tax benefits, if recognized, would not affect the effective tax rate. There were no unrecognized tax benefits at December 31, 2018. No interest or penalties were incurred during the years ended December 31, 2017, 2018 or 2019.

The Company files income tax returns in the U.S. federal jurisdiction and various state and foreign jurisdictions. The Company is not currently under examination by income tax authorities in federal, state or other jurisdictions. All tax returns will remain open for examination by the federal and state authorities for three and four years, respectively, from the date of utilization of any net operating loss or credits. Certain tax years are subject to foreign income tax examinations by tax authorities until the statute of limitations expire.

### 11. NET LOSS PER SHARE ATTRIBUTABLE TO COMMON STOCKHOLDERS

The following table sets forth the computation of basic and diluted net loss per share attributable to common stockholders (in millions, except share and per share data):

	Year Ended December 31,					
	2017			2018		2019
Numerator:						
Net loss	\$	(207)	\$	(208)	\$	(129)
Less: deemed dividend to redeemable convertible preferred						
stockholders		(40)		_		(7)
Net loss attributable to common stockholders	\$	(247)	\$	(208)	\$	(136)
Denominator:			<u>-</u>			
Weighted-average shares used in computing net loss per share						
attributable to common stockholders, basic and diluted	10,0	038,841	10,:	<u> 296,604</u>	10	,404,562
Net loss per share attributable to common stockholders, basic and diluted	\$	(24.60)	\$	(20.20)	\$	(13.07)

The following outstanding shares of potentially dilutive securities were excluded from the computation of diluted net loss per share because including them would have had an anti-dilutive effect:

	As of December 31,			
	2017	2018	2019	
Redeemable convertible preferred stock, all series	41,295,644	41,295,644	42,169,192	
Series B warrants.	986,640	986,640	986,640	
Warrants to purchase common stock	55,000	55,000	55,000	
Common stock options outstanding	8,542,607	8,112,859	8,032,296	
Restricted stock units outstanding	2,465,125	3,335,076	4,383,735	
Total	53,345,016	53,785,219	55,626,863	

### 12. UNAUDITED PRO FORMA NET LOSS PER SHARE

Unaudited pro forma basic net loss per share is computed to give effect to the automatic conversion of all outstanding shares of the Company's redeemable convertible preferred stock into common stock using the if-converted method as though the conversion had occurred as of the beginning of the period or the original date of issuance, if later, and the exercise of Series B warrants, resulting in the issuance of common stock. The pro forma share amounts also give effect to the RSUs subject to both service-based and liquidity event-related performance vesting conditions for which the service-based vesting condition had been satisfied as of on a weighted-average basis. As it relates to RSUs for which the service-based vesting condition was satisfied as of , the pro forma shares outstanding for the year ended includes the weighted-average issuance of shares of common stock.

The net loss used in computing unaudited pro forma basic net loss per share gives effect to remove losses resulting from the remeasurement of the Series B warrant liability. The net loss used in computing unaudited pro forma basic net loss per share does not give effect to the stock-based compensation expense associated with these RSUs. If the IPO had been completed on the Company would have recognized approximately of stock-based compensation related to these RSUs on the effective date. Unaudited pro forma diluted net loss per share is the same as the unaudited pro forma basic net loss per share for the period as the impact of any potentially dilutive securities was anti-dilutive.

The following table presents the calculation of pro forma basic and diluted net loss per share (in millions, except share and per share data):

	Year Ended December 31, 2019
Numerator:	
Net loss attributable to common stockholders	\$
Add: deemed dividend to redeemable convertible preferred stockholders	
Add: change in fair value of redeemable convertible preferred stock warrant liability	
Pro forma net loss attributable to common stockholders, basic and diluted	\$
Denominator:	
Weighted-average shares used in computing net loss per share attributable to common stockholders, basic and diluted	
Pro forma adjustment to reflect the automatic conversion of redeemable convertible preferred stock	
Pro forma adjustment to reflect the assumed exercise and conversion of Series B warrants	
Pro forma adjustment to reflect assumed vesting of RSUs with liquidity event-related performance vesting condition	
Weighted-average shares used in computing pro forma net loss per share attributable to common stockholders, basic and diluted	
Pro forma net loss attributable to common stockholders per share, basic and diluted	\$

### 13. GEOGRAPHICAL INFORMATION

The Company believes it is relevant to disclose marketplace geographical revenue information on both a demand basis, determined by the ship-to address of the user, and on a supply basis, determined by the location of the merchants' operations.

Marketplace revenue by geographic area based on the ship-to address of the user is as below (in millions):

		Year Ended December 31,								
	2	2017 <sup>(1)</sup>			2018 <sup>(2)</sup>			2019 <sup>(2)</sup>		
Europe	\$ 51	5 47%	\$	866	50%	\$	857	49%		
North America <sup>(3)</sup>	46	60 42%		660	38%		675	38%		
South America	4	2 4%		69	4%		89	5%		
Other	8	34 7%		127	8%		143	8%		
Total marketplace revenue	\$1,10	100%	\$1	,722	100%	\$1	1,764	100%		

Revenue has not been adjusted to Topic 606 and is reported under Topic 605.

Revenue has been adjusted for the impact of Topic 606.
United States accounted for \$402 million, \$563 million and \$562 million of marketplace revenue in 2017, 2018 and 2019, respectively.

China accounted for substantially all of marketplace and logistics revenue in 2017, 2018 and 2019, respectively, based on the location of the merchants' operations.

The Company's long-lived tangible assets, which consist of property and equipment, net and operating lease right-of-use assets, net, located in the United States were 93% and 94% of the total

long-lived tangible assets as of December 31, 2018 and 2019, respectively. The long-lived tangible assets outside the United States were located in Canada, China, and the Netherlands.

### 14. SUBSEQUENT EVENTS

The Company has evaluated subsequent events for recognition and measurement purposes through August 28, 2020, which is the date that the consolidated financial statements and accompanying notes were available for issuance.

### 2010 Stock Plan

On May 5, 2020, the Company's Board of Directors approved an increase in the shares reserved for issuance under the 2010 Stock Plan from 14,577,861 to 14,947,861 shares.

\*\*\*\*\*

F-35

**Shares** 

### ContextLogic Inc.

Class A Common Stock



Goldman Sachs & Co. LLC

J.P. Morgan

**BofA Securities** 

#### Part II

#### INFORMATION NOT REQUIRED IN PROSPECTUS

### Item 13. Other Expenses of Issuance and Distribution

The following table presents the costs and expenses, other than underwriting discounts and commissions, payable in connection with this offering. All amounts are estimates except the SEC registration fee, the FINRA filing fee and listing fee. Except as otherwise noted, all the expenses below will be paid by us.

\$ *
*
*
*
*
*
*
*
\$ *
\$

To be completed by amendment.

### Item 14. Indemnification of Directors and Officers

Section 145 of the Delaware General Corporation Law authorizes a court to award, or a corporation's board of directors to grant, indemnity to directors and officers under certain circumstances and subject to certain limitations. The terms of Section 145 of the Delaware General Corporation Law are sufficiently broad to permit indemnification under certain circumstances for liabilities, including reimbursement of expenses incurred, arising under the Securities Act.

As permitted by the Delaware General Corporation Law, our amended and restated certificate of incorporation and amended and restated bylaws contain provisions relating to the limitation of liability and indemnification of directors and officers. The amended and restated certificate of incorporation provides that our directors will not be personally liable to us or our stockholders for monetary damages for any breach of fiduciary duty as a director, except for liability:

- · for any breach of the director's duty of loyalty to us or our stockholders;
- for acts or omissions not in good faith or that involve intentional misconduct or a knowing violation of law;
- in respect of unlawful payments of dividends or unlawful stock repurchases or redemptions as provided in Section 174 of the Delaware General Corporation Law; or
- for any transaction from which the director derives any improper personal benefit.

Our amended and restated certificate of incorporation also provides that if Delaware law is amended after the approval by our stockholders of the certificate of incorporation to authorize corporate action further eliminating or limiting the personal liability of directors, then the liability of our directors will be eliminated or limited to the fullest extent permitted by Delaware law.

Our amended and restated bylaws provide that we will indemnify our directors and officers to the fullest extent permitted by Delaware law, as it now exists or may in the future be amended, against all expenses and liabilities reasonably incurred in connection with their service for or on our behalf. Our amended and restated bylaws provide that we shall advance the expenses incurred by a director or

officer in advance of the final disposition of an action or proceeding, and permit us to secure insurance on behalf of any director, officer, employee, or other enterprise agent for any liability arising out of his or her action in that capacity, whether or not Delaware law would otherwise permit indemnification.

We intend to enter into indemnification agreements with each of our directors and executive officers and certain other key employees, a form of which is attached as Exhibit 10.1. The form of agreement provides that we will indemnify each of our directors, executive officers and such other key employees against any and all expenses incurred by that director, executive officer, or other key employee because of his or her status as one of our directors, executive officers, or other key employees, to the fullest extent permitted by Delaware law, our restated certificate of incorporation and our amended and restated bylaws. In addition, the form agreement provides that, to the fullest extent permitted by Delaware law, we will advance all expenses incurred by our directors, executive officers and other key employees in connection with a legal proceeding.

Reference is made to the underwriting agreement contained in Exhibit 1.1 to this registration statement, indemnifying our directors and officers against limited liabilities. In addition, Section 1.9 of our amended and restated investors' rights agreement (the "IRA") contained in Exhibit 4.2 to this registration statement provides for indemnification of certain of our stockholders against liabilities described in our IRA.

We currently carry and intend to continue to carry liability insurance for our directors and officers.

### Item 15. Recent Sales of Unregistered Securities

Since January 1, 2017 we have issued the following unregistered securities (after giving effect to the conversion of our common stock into Class B common stock).

#### Preferred Stock Issuances

In April 2017, we issued 539,428 shares of our Series F redeemable convertible preferred stock in a subsequent closing at a price per share of \$64.88 per share to one accredited investor.

Between September 2017 and October 2017, we issued 1,687,319 shares of our Series G redeemable convertible preferred stock at a price per share of \$134.53 to 15 accredited investors.

In March 2019, we issued 943,548 shares of our Series H redeemable convertible preferred stock at a price per share of \$169.57 to two accredited investors.

### Plan-Related Issuances

From January 1, 2017 through July 31, 2020, we issued to our directors, officers, employees, consultants, and other service providers an aggregate of 1,090,780 shares of our Class B common stock at per share purchase prices ranging from \$0.01 to \$55.56 pursuant to exercises of options granted under our 2010 Stock Plan.

From January 1, 2017 through July 31, 2020, we granted to our directors, officers, employees, consultants, and other service providers an aggregate of 4,132,640 RSUs, to be settled in shares of our Class B common stock under our 2010 Stock Plan.

In April 2019, we issued to a former consultant an aggregate of 23,334 shares of our Class B Common Stock at a price per share of \$0.68 pursuant to a stock grant under our 2010 Stock Plan.

None of the foregoing transactions involved any underwriters, underwriting discounts or commissions, or any public offering. We believe that the offers, sales and issuances of the above securities were exempt from registration under the Securities Act by virtue of Section 4(a)(2) of the

Securities Act or Regulation D promulgated thereunder as transactions by an issuer not involving any public offering, or in reliance on Rule 701 promulgated under Section 3(b) of the Securities Act because the transactions were pursuant to compensatory benefit plans or contracts relating to compensation as provided under Rule 701. The recipients of the securities in each of these transactions represented their intentions to acquire the securities for investment only and not with a view to or for sale in connection with any distribution thereof, and appropriate legends were placed upon the stock certificates issued in these transactions. All recipients had adequate information about us or had adequate access, through their relationships with us, to information about us.

### Item 16. Exhibits and Financial Statement Schedules

### (a) Exhibits.

Exhibit Number	Description
1.1*	Form of Underwriting Agreement.
3.1	Restated Certificate of Incorporation of the Registrant, as amended, as currently in effect.
3.2*	Form of Amended and Restated Certificate of Incorporation of the Registrant, to be effective upon completion of this offering.
3.3	Bylaws of the Registrant, as currently in effect.
3.4*	Form of Amended and Restated Bylaws of the Registrant, to be effective upon completion of this offering.
4.1*	Form of Registrant's Class A common stock certificate.
4.2	Amended and Restated Investors' Rights Agreement, dated March 18, 2019 by and among the Registrant and the other parthereto.
4.3	Warrant to Purchase Series B Preferred Stock, issued to Formation8 Partners Fund I, L.P., dated August 1, 2016.
4.4	Warrant to Purchase Common Stock, issued to Silicon Valley Bank, dated May 13, 2014.
5.1*	Opinion of Gunderson Dettmer Stough Villeneuve Franklin & Hachigian, LLP.
10.1*	Form of Indemnification Agreement between the Registrant and each of its directors and executive officers.
10.2	2010 Stock Plan, as amended, and forms of agreements thereunder.
10.3*	2020 Equity Incentive Plan and form of agreements thereunder.
10.4*	2020 Employee Stock Purchase Plan.
10.5*	Offer Letter, dated , between the Company and Peter Szulczewski.
10.6*	Offer Letter, dated , between the Registrant and Rajat Bahri.
10.7*	Offer Letter, dated , between the Registrant and Devang Shah.
10.8*	Offer Letter, dated , between the Registrant and Thomas Chuang.
10.9*	Offer Letter, dated , between the Registrant and Pai Liu.
21.1*	Subsidiaries of the Registrant.
23.1*	Consent of Ernst & Young LLP, Independent Registered Public Accounting Firm.
23.2*	Consent of Gunderson Dettmer Stough Villeneuve Franklin & Hachigian, LLP (contained in Exhibit 5.1).
24.1	Power of Attorney (included on signature page).

parties

To be provided by amendment.

(b) **Financial Statement Schedules.** All schedules have been omitted because the information required to be presented in them is not applicable or is shown in the consolidated financial statements or related notes.

### Item 17. Undertakings

The undersigned registrant hereby undertakes to provide to the underwriter at the closing specified in the underwriting agreements certificates in such denominations and registered in such names as required by the underwriter to permit prompt delivery to each purchaser.

Insofar as indemnification for liabilities arising under the Securities Act may be permitted to directors, officers and controlling persons of the registrant pursuant to the foregoing provisions, or otherwise, the registrant has been advised that in the opinion of the SEC such indemnification is against public policy as expressed in the Act and is, therefore, unenforceable. In the event that a claim for indemnification against such liabilities (other than the payment by the registrant of expenses incurred or paid by a director, officer or controlling person of the registrant in the successful defense of any action, suit or proceeding) is asserted by such director, officer or controlling person in connection with the securities being registered, the registrant will, unless in the opinion of its counsel the matter has been settled by controlling precedent, submit to a court of appropriate jurisdiction the question whether such indemnification by it is against public policy as expressed in the Act and will be governed by the final adjudication of such issue.

The undersigned registrant hereby undertakes that:

- (1) For purposes of determining any liability under the Securities Act, the information omitted from the form of prospectus filed as part of this registration statement in reliance upon Rule 430A and contained in a form of prospectus filed by the registrant pursuant to Rule 424(b)(1) or (4) or 497(h) under the Securities Act shall be deemed to be part of this registration statement as of the time it was declared effective.
- (2) For the purpose of determining any liability under the Securities Act, each post-effective amendment that contains a form of prospectus shall be deemed to be a new registration statement relating to the securities offered therein, and the offering of such securities at that time shall be deemed to be the initial bona fide offering thereof.

### Confidential Treatment Requested by ContextLogic Inc. Pursuant to 17 C.F.R. Section 200.83

#### **SIGNATURES**

Pursuant to the requirements of the Securities Act of 1933, the registrant has duly caused this Registration Statement to be signed on its behalf by the undersigned, thereunto duly authorized, in the City of San Francisco, State of California, on this day of

CONTEXTLOGIC INC.

Rajat Bahri Chief Financial Officer

#### **POWER OF ATTORNEY**

KNOW ALL PERSONS BY THESE PRESENTS, that each person whose signature appears below hereby constitutes and appoints Peter Szulczewski, Rajat Bahri, and Devang Shah, and each of them, as his or her true and lawful attorney-in-fact and agent with full power of substitution, for him or her in any and all capacities, to sign any and all amendments to this registration statement (including post-effective amendments) and any registration statement related thereto filed pursuant to Rule 462(b) increasing the number of securities for which registration is sought, and to file the same, with all exhibits thereto and other documents in connection therewith, with the SEC, granting unto said attorney-in-fact and agent full power and authority to do and perform each and every act and thing requisite and necessary to be done in connection therewith, as fully for all intents and purposes as he or she might or could do in person, hereby ratifying and confirming all that said attorney-in-fact and agent, or his or her substitute, may lawfully do or cause to be done by virtue hereof.

Pursuant to the requirements of the Securities Act of 1933, as amended, this Registration Statement has been signed by the following persons in the capacities and on the dates indicated.

Signature	Title	Date
Peter Szulczewski	Chief Executive Officer and Chairperson (principal executive officer)	
Rajat Bahri	Chief Financial Officer (principal financial and accounting officer)	
Ari Emanuel	Director	
Joe Lonsdale	Director	
Tanzeen Syed	Director	
Stephanie Tilenius	Director	
Hans Tung	Director	

# AMENDED AND RESTATED CERTIFICATE OF INCORPORATION OF CONTEXTLOGIC INC.

## (Pursuant to Sections 242 and 245 of the General Corporation Law of the State of Delaware)

**ContextLogic Inc.**, a corporation organized and existing under and by virtue of the provisions of the General Corporation Law of the State of Delaware (the "General Corporation Law"),

#### **DOES HEREBY CERTIFY:**

**FIRST**: That the name of this corporation is **ContextLogic Inc.** and that this corporation was originally incorporated pursuant to the General Corporation Law on June 25, 2010 under the name **ContextLogic Inc.** 

**SECOND**: That the Board of Directors duly adopted resolutions proposing to amend and restate the Certificate of Incorporation of this corporation, declaring said amendment and restatement to be advisable and in the best interests of this corporation and its stockholders, and authorizing the appropriate officers of this corporation to solicit the consent of the stockholders therefor, which resolution setting forth the proposed amendment and restatement is as follows:

RESOLVED, that the Certificate of Incorporation of this corporation be amended and restated in its entirety as follows:

#### ARTICLE I

The name of this corporation is ContextLogic Inc.

#### ARTICLE II

The address of this corporation's registered office in the State of Delaware is 3500 South Dupont Highway in the City of Dover, County of Kent, Zip Code 19901. The name of this corporation's registered agent at such address is Incorporating Services, Ltd.

#### ARTICLE III

The nature of the business or purposes to be conducted or promoted is to engage in any lawful act or activity for which corporations may be organized under the General Corporation Law.

#### ARTICLE IV

A. <u>Authorization of Stock</u>. This corporation is authorized to issue two classes of stock to be designated, respectively, common stock and preferred stock. The total number of shares that this corporation is authorized to issue is 117,346,293. The total number of shares of

common stock authorized to be issued is 73,000,000, par value \$0.0001 per share (the "Common Stock"). The total number of shares of preferred stock authorized to be issued is 44,346,293, par value \$0.0001 per share (the "Preferred Stock"), 6,437,367 of which are designated as "Series A Preferred Stock," 6,883,083 of which are designated as "Series B Preferred Stock," 7,108,490 of which are designated as "Series C Preferred Stock," 5,519,533 of which are designated as "Series D Preferred Stock," 8,324,560 of which are designated as "Series E Preferred Stock", 6,321,932 of which are designated as "Series F Preferred Stock", 1,687,319 of which are designated as "Series G Preferred Stock" and 2,064,009 of which are designated as "Series H Preferred Stock."

B. <u>Rights, Preferences and Restrictions of Preferred Stock</u>. The rights, preferences, privileges and restrictions granted to and imposed on the Preferred Stock are as set forth below in this <u>Article IV(B)</u>.

#### 1. Dividend Provisions.

(a) The holders of shares of Preferred Stock shall be entitled to receive dividends, on a *pari passu* basis, out of any assets legally available therefor, prior and in preference to any declaration or payment of any dividend (payable other than in Common Stock or other securities and rights convertible into or entitling the holder thereof to receive, directly or indirectly, additional shares of Common Stock of this corporation) on the Common Stock of this corporation, at the applicable Dividend Rate (as defined below), payable when, as and if declared by the Board of Directors. Such dividends shall not be cumulative. The holders of the outstanding Preferred Stock can waive any dividend preference that such holders shall be entitled to receive under this Section I upon the affirmative vote or written consent of the holders of a majority of the voting power of shares of Preferred Stock then outstanding (voting together as a single class and not as separate series, and on an as-converted basis); provided however, if any such waiver is a waiver of a dividend preference for less than all series of Preferred Stock, the affirmative vote or written consent of the holders of a majority of the shares of each series of Preferred Stock for which such dividend preference is being waived shall be required. For purposes of this subsection I(a), "Dividend Rate" shall mean \$0.280464 per annum for each share of Series A Preferred Stock, \$0.316 per annum for each share of Series B Preferred Stock, \$0.502128 per annum for each share of Series C Preferred Stock, \$2.000168 per annum for each share of Series B Preferred Stock, \$% of the Original Issue Price of Series G Preferred Stock per annum for each share of Series F Preferred Stock per annum for each share of Series F Preferred Stock per annum for each share of Series H Preferred Stock per annum for each share of Series H Preferred Stock per annum for each share of Series H Preferred Stock per annum for each share of Series H Preferred Stock per annum for each share of Series H

(b) After payment of such dividends, any additional dividends or distributions shall be distributed among all holders of Common Stock and Preferred Stock in proportion to the number of shares of Common Stock that would be held by each such holder if all shares of Preferred Stock were converted to Common Stock at the then effective conversion rate.

#### 2. Liquidation Preference

(a) For purposes of this Amended and Restated Certificate of Incorporation, "Original Issue Price" shall mean \$3.5058 per share for each share of the Series A Preferred Stock, \$3.95 per share for each share of the Series B Preferred Stock, \$6.2766 per share for each share of Series C Preferred Stock, \$25.0021 per share for each share of Series D Preferred Stock, the original price per share paid to the corporation by check, wire transfer, cancellation of indebtedness or any combination of the foregoing for the Series E Preferred Stock in accordance with a written agreement with the corporation, a copy of which is maintained at the principal office of the corporation, setting forth the purchase price per share of such Series E Preferred Stock, the original price per share paid to the corporation by check, wire transfer, cancellation of indebtedness or any combination of the foregoing for the Series F Preferred Stock in accordance with a written agreement with the corporation, a copy of which is maintained at the principal office of the corporation, setting forth the purchase price per share of such Series F Preferred Stock, the original price per share paid to the corporation by check, wire transfer, cancellation of indebtedness or any combination of the foregoing for the Series G Preferred Stock in accordance with a written agreement with the corporation, a copy of which is maintained at the principal office of the corporation, setting forth the purchase price per share of such Series G Preferred Stock, and the original price per share paid to the corporation by check, wire transfer, cancellation of indebtedness or any combination of the foregoing for the Series H Preferred Stock in accordance with a written agreement with the corporation, a copy of which is maintained at the principal office of the corporation, setting forth the purchase price per share of such Series H Preferred Stock (each as adjusted for any stock splits, stock dividends, combinations, subdivisions, recapitalizations or the like with respect to such series of Preferred Stock). In the event of any Liquidation Event (as defined below), either voluntary or involuntary that is consummated, the holders of each series of Preferred Stock shall be entitled to receive, on a pari passu basis, out of the proceeds or assets of this corporation available for distribution to its stockholders (the "Proceeds"), prior to and in preference to any distribution of the Proceeds to the holders of Common Stock by reason of their ownership thereof, an amount per share equal to (i) in the case of shares of Series A Preferred Stock, Series B Preferred Stock, Series C Preferred Stock, Series D Preferred Stock, Series E Preferred Stock, Series F Preferred Stock, and Series G Preferred Stock (each, a "Prior Series"), the sum of the applicable Original Issue Price for such Prior Series, plus declared but unpaid dividends on such shares and (ii) in the case of shares of Series H Preferred Stock, the applicable Series H Preferred Stock Liquidation Amount (as defined in that certain Series H Preferred Stock Purchase Agreement, made by and among this corporation and certain other parties thereto including certain of this corporation's stockholders and which is dated as of March 18, 2019, and a copy of which is maintained at the principal office of the corporation (the "Series H Preferred Stock Purchase Agreement")), plus declared but unpaid dividends on such shares. If, upon the occurrence of such event, the Proceeds thus distributed among the holders of the Preferred Stock shall be insufficient to permit the payment to such holders of the full aforesaid preferential amounts, then the entire Proceeds legally available for distribution shall be distributed ratably among the holders of the Preferred Stock in proportion to the full preferential amount that each such holder is otherwise entitled to receive under this subsection

(b) Upon completion of the distribution required by <u>subsection 2(a)</u>, all of the remaining Proceeds available for distribution to stockholders shall be distributed among the holders of Common Stock pro rata based on the number of shares of Common Stock held by each.

(c) Notwithstanding the above, for purposes of determining the amount each holder of shares of Preferred Stock is entitled to receive with respect to a Liquidation Event, each such holder of shares of a series of Preferred Stock shall be deemed to have converted (regardless of whether such holder has actually converted) such holder's shares of such series into shares of Common Stock immediately prior to the Liquidation Event if, as a result of an actual conversion, such holder would receive, in the aggregate, an amount greater than the amount that would be distributed to such holder if such holder did not convert such series of Preferred Stock into shares of Common Stock. If any such holder shall be deemed to have converted shares of Preferred Stock into Common Stock pursuant to this <u>subsection 2(c)</u>, then such holder shall not be entitled to receive any distribution that would otherwise be made to holders of Preferred Stock that have not converted (or have not been deemed to have converted) into shares of Common Stock and shall instead be entitled to receive any distributions that such holder would be entitled to receive had such holder converted such shares of the applicable series of Preferred Stock into shares of Common Stock.

(d) (i) For purposes of this Section 2, a "Liquidation Event" shall mean (A) the closing of the sale, transfer or other disposition, in a single transaction or series of related transactions, of all or substantially all of this corporation's assets (including, without limitation, the sale or disposition (whether by merger, sale of securities or otherwise) of one or more subsidiaries of the corporation if substantially all of the assets of the corporation are held by such subsidiary or subsidiaries), (B) the consummation of a merger, reorganization or consolidation of this corporation with or into another entity or of a subsidiary of this corporation which is a constituent party and this corporation issues shares of its capital stock pursuant to such merger or consolidation (except a merger, reorganization or consolidation in which the holders of capital stock of this corporation immediately prior to such merger, reorganization or consolidation continue to hold at least 50% of the voting power of the capital stock of this corporation or the surviving or acquiring entity), (C) the closing of a transfer (whether by merger, reorganization, consolidation or otherwise), in one transaction or a series of related transactions, to a person or group of affiliated persons (other than an underwriter of this corporation's securities), of this corporation's securities if, after such closing, such person or group of affiliated persons would hold 50% or more of the outstanding voting stock of this corporation (or the surviving or acquiring entity) or (D) a liquidation, dissolution or winding up of this corporation; provided, however, that a transaction shall not constitute a Liquidation Event if its sole purpose is to change the state of this corporation's incorporation or to create a holding company that will be owned in substantially the same proportions by the persons who held this corporation's securities immediately prior to such transaction. Notwithstanding the prior sentence, the sale of shares of Series H Preferred Stock by this corporation in a financing transaction pursuant to the Series H Preferred Stock Purchase Agreement shall not be deemed a "Liquidation Event." The treatment of any particular transaction or series of related transactions as a Liquidation Event may be waived by the vote or written consent of the holders of (v) a majority of the voting power of outstanding Preferred Stock (voting together as a single class and not as separate series, and on an as-converted basis), (w) a majority of the outstanding shares of Series E Preferred Stock (voting as a separate series and on an as-converted basis), (x) the Series F Requisite Holders (as such term is defined in the Investors' Rights Agreement (as defined

below)), (y) the Series G Requisite Holders (as such term is defined in the Investors' Rights Agreement) and (z) a majority of the outstanding shares of Series H Preferred Stock (voting as a separate series and on an as-converted basis). "Investors' Rights Agreement" means that certain Amended and Restated Investors' Rights Agreement, as amended from time to time, made by and among this corporation and certain of this corporation's stockholders and which is dated on or about the Filing Date, and a copy of which is maintained at the principal office of the corporation.

- (ii) In any Liquidation Event, if Proceeds received by this corporation or its stockholders is other than cash, its value will be deemed its fair market value. Any securities shall be valued as follows:
  - (A) Securities not subject to investment letter or other similar restrictions on free marketability covered by (B) below:
  - (1) If traded on a securities exchange, the value shall be deemed to be the average of the closing prices of the securities on such exchange over the twenty (20) trading-day period ending three (3) trading days prior to the closing of the Liquidation Event;
  - (2) If actively traded over-the-counter, the value shall be deemed to be the average of the closing bid or sale prices (whichever is applicable) over the twenty (20) trading-day period ending three (3) trading days prior to the closing of the Liquidation Event; and
  - (3) If there is no active public market, the value shall be the fair market value thereof, as mutually determined in good faith by this corporation and the holders of a majority of the voting power of all then outstanding shares of Preferred Stock (voting together as a single class and not as separate series, and on an as-converted basis).
  - (B) The method of valuation of securities subject to investment letter or other restrictions on free marketability (other than restrictions arising solely by virtue of a stockholder's status as an affiliate or former affiliate) shall be to make an appropriate discount from the market value determined as above in  $\underline{\text{subsection 2}(\underline{d})(\underline{i}\underline{i})(\underline{A})}$ , clauses (1), (2) or (3) to reflect the approximate fair market value thereof, as mutually determined in good faith by this corporation and the holders of a majority of the voting power of all then outstanding shares of such Preferred Stock (voting together as a single class and not as separate series, and on an as-converted basis).

- (C) The foregoing methods for valuing non-cash consideration to be distributed in connection with a Liquidation Event shall, with the appropriate approval of the definitive agreements governing such Liquidation Event by the stockholders under the General Corporation Law and Section 6 of this Article IV(B), be superseded by the determination of such value set forth in the definitive agreements governing such Liquidation Event.
  - (iii) In the event the requirements of this Section 2 are not complied with, this corporation shall forthwith either:
- (A) cause the closing of such Liquidation Event to be postponed until such time as the requirements of this <u>Section 2</u> have been complied with; or
- (B) cancel such transaction, in which event the rights, preferences and privileges of the holders of the Preferred Stock shall revert to and be the same as such rights, preferences and privileges existing immediately prior to the date of the first notice referred to in <u>subsection 2(d)(iv)</u> hereof.
- (iv) This corporation shall give each holder of record of Preferred Stock written notice of such impending Liquidation Event not later than twenty (20) days prior to the stockholders' meeting called to approve such transaction, or twenty (20) days prior to the closing of such transaction, whichever is earlier, and shall also notify such holders in writing of the final approval of such transaction, which approval shall not be effective without first obtaining all required approvals pursuant to Section 6 of this Article IV(B). The first of such notices shall describe the material terms and conditions of the impending transaction and the provisions of this Section 2, and this corporation shall thereafter give such holders prompt notice of any material changes. The transaction shall in no event take place sooner than twenty (20) days after this corporation has given the first notice provided for herein or sooner than ten (10) days after this corporation has given notice of any material changes provided for herein; provided, however, that subject to compliance with the General Corporation Law such periods may be shortened or waived upon the written consent of the holders of Preferred Stock that represent a majority of the voting power of all then outstanding shares of such Preferred Stock (voting together as a single class and not as separate series, and on an as-converted basis).
- (e) This corporation shall not have the power to effect a Liquidation Event unless the agreement or plan of merger or consolidation for such transaction (the "Merger Agreement") provides that the consideration payable to the stockholders of this corporation shall be allocated among the holders of capital stock of this corporation in accordance with subsection 2(a).

In the event of a Liquidation Event in which proceeds of the Liquidation Event are received by this corporation or a subsidiary of this corporation, if this corporation does not effect a dissolution of this corporation under the General Corporation Law within ninety (90)

days after such Liquidation Event, then (i) this corporation shall send a written notice to each holder of record of Preferred Stock, no later than the ninetieth (90th) day after the Liquidation Event advising such holders of their right (and the requirements to be met to secure such right) pursuant to the terms of the following clause; (ii) to require the redemption of such shares of Preferred Stock, and (iii) if the holders of at least a majority of the voting power of (x) then outstanding shares Preferred Stock (voting together as a single class and not as separate series, and on an as-converted basis) or (y) then outstanding Series H Preferred Stock (in relation to the applicable liquidation amount payable in respect of the Series H Preferred Stock) (voting as a separate series and on an as-converted basis) so request in a written instrument delivered to this corporation not later than one hundred twenty (120) days after such Liquidation Event, this corporation shall use the consideration received by this corporation for such Liquidation Event (net of any retained liabilities associated with the assets sold or technology licensed, as determined in good faith by the Board of Directors), together with any other assets of this corporation available for distribution to its stockholders, all to the extent permitted by Delaware law governing distributions to stockholders (the "Available Proceeds"), on the one hundred fiftieth (150th) day after such Liquidation Event, to redeem all outstanding shares of Preferred Stock requested to be redeemed pursuant to the foregoing clause (iii)(x) or (iii)(y), as applicable, at an amount per share equal to the applicable liquidation amount set forth in subsection 2(a). Notwithstanding the foregoing, in the event of a redemption pursuant to the preceding sentence, if the Proceeds thus distributed among the holders of the applicable Preferred Stock shall be insufficient to permit the payment to such holders of the full preferential amounts, then the entire Proceeds legally available for distribution shall be distributed ratably among the holders of the applicable Preferred Stock in proportion to the full preferential amount that each such holder is otherwise entitled to receive under subsection 2(a). Unless otherwise approved by the holders of a majority of the voting power of then outstanding shares of Preferred Stock (voting together as a single class and not as separate series, and on an as-converted basis) or by the holders of a majority of the voting power of then outstanding shares of Series H Preferred Stock (voting as a separate series and on an as-converted basis), as applicable, prior to the distribution or redemption provided for in this subsection 2(e), this corporation shall not expend or dissipate the consideration received for such Liquidation Event, except to discharge expenses incurred in connection with such Liquidation Event.

- 3. <u>Redemption</u>. The Preferred Stock is not redeemable at the option of the holder thereof.
- 4. Conversion. The holders of the Preferred Stock shall have conversion rights as follows (the "Conversion Rights"):

(a) <u>Right to Convert</u>. Each share of Preferred Stock shall be convertible, at the option of the holder thereof, at any time after the date of issuance of such share, and without the payment of additional consideration by the holder thereof, at the office of this corporation or any transfer agent for such stock, into such number of fully paid and nonassessable shares of Common Stock as is determined by dividing the applicable Original Issue Price for such series by the applicable Conversion Price for such series (the conversion rate for a series of Preferred Stock into Common Stock is referred to herein as the "<u>Conversion Rate</u>" for such series), determined as hereafter provided, in effect on the date the certificate is surrendered for conversion. The initial "<u>Conversion Price</u>" per share for each series of Preferred Stock shall be the Original Issue Price applicable to such series; provided, however, that the Conversion Price for the Preferred Stock shall be subject to adjustment as set forth in <u>subsection 4(d)</u>.

(b) Automatic Conversion. Each share of Preferred Stock shall automatically be converted into shares of Common Stock at the Conversion Rate at the time in effect for such series of Preferred Stock immediately upon the closing of this corporation's sale of its Common Stock in a firm commitment underwritten public offering pursuant to a registration statement on Form S-1 under the Securities Act of 1933, as amended, reflecting a pre-money valuation of at least the Minimum Valuation (as defined in the Investors' Rights Agreement), with gross cash proceeds of no less than \$500,000,000 which complies with the provisions of Section 6.19 of the Series H Preferred Stock Purchase Agreement (the "Qualified Public Offering"). In addition, (v) each share of Series A Preferred Stock, Series B Preferred Stock, Series C Preferred Stock and Series D Preferred Stock (together, the "A-D Preferred Stock") shall automatically be converted into shares of Common Stock at the applicable Conversion Rate at the time in effect for each such series of the A-D Preferred Stock immediately upon the date, or the occurrence of an event, specified by vote or written consent or agreement of the holders of a majority of the then outstanding shares of the A-D Preferred Stock (voting together as a single class and not as a separate series, and on an as-converted basis), (w) each share of Series E Preferred Stock shall automatically be converted into shares of Common Stock at the Conversion Rate at the time in effect for such Series E Preferred Stock immediately upon the date, or the occurrence of an event, specified by vote or written consent or agreement of the holders of a majority of the then outstanding shares of the Series E Preferred Stock (voting as a separate series and on an as-converted basis), (x) each share of Series F Preferred Stock shall automatically be converted into shares of Common Stock at the Conversion Rate at the time in effect for such Series F Preferred Stock immediately upon the date, or the occurrence of an event, specified by vote or written consent or agreement of the Series F Requisite Holders, (y) each share of Series G Preferred Stock shall automatically be converted into shares of Common Stock at the Conversion Rate at the time in effect for such Series G Preferred Stock immediately upon the date, or the occurrence of an event, specified by vote or written consent or agreement of the Series G Requisite Holders and (z) each share of Series H Preferred Stock shall automatically be converted into shares of Common Stock at the Conversion Rate at the time in effect for such Series H Preferred Stock immediately upon the date, or the occurrence of an event, specified by vote or written consent or agreement of the holders of a majority of the outstanding shares of Series H Preferred Stock (voting as a separate series and on an as-converted basis).

(c) <u>Mechanics of Conversion</u>. Before any holder of Preferred Stock shall be entitled to voluntarily convert the same into shares of Common Stock, such holder shall surrender the certificate or certificates therefor, duly endorsed, at the office of this corporation or of any transfer agent for the Preferred Stock, and shall give written notice to this corporation at its principal corporate office, of the election to convert all or any number of the shares of Preferred Stock represented by such certificate or certificates, any event on which such conversion is contingent (if applicable) and the name or names in which the certificate or certificates for shares of Common Stock are to be issued. This corporation shall, as soon as practicable thereafter, issue and deliver at such office to such holder of Preferred Stock, or to the nominee or nominees of such holder, a certificate or certificates for the number of shares of Common Stock to which such holder shall be entitled as aforesaid and a certificate for the

number (if any) of the shares of Preferred Stock represented by the surrendered certificate or certificates that were not converted in Common Stock. Any declared but unpaid dividends for the Preferred Stock which is being converted shall be paid by the corporation at the same time as the declared but unpaid dividends are paid on the outstanding shares of such series of Preferred Stock. Such conversion shall be deemed to have been made immediately prior to the close of business on the date set forth for conversion in the written notice of the election to convert irrespective of the surrender of the shares of Preferred Stock to be converted, and the person or persons entitled to receive the shares of Common Stock issuable upon such conversion shall be treated for all purposes as the record holder or holders of such shares of Common Stock as of such date. If the conversion is in connection with an underwritten offering of securities registered pursuant to the Securities Act of 1933, as amended, the conversion may, at the option of any holder tendering Preferred Stock for conversion, be conditioned upon the closing with the underwriters of the sale of securities pursuant to such offering, in which event the persons entitled to receive the Common Stock upon conversion of the Preferred Stock shall not be deemed to have converted such Preferred Stock until immediately prior to the closing of such sale of securities. If the conversion is in connection with "automatic conversion" provisions of subsection 4(b) above, such conversion shall be deemed to have been made on the conversion date described in the stockholder consent approving such conversion, and the persons entitled to receive shares of Common Stock issuable upon such conversion shall be treated for all purposes as the record holders of such shares of Common Stock as of such date. Notwithstanding anything to the contrary in this Amended and Restated Certificate of Incorporation, a holder who wants to convert some or all of its shares of Preferred Stock into Common Stock (whether pursuant to subsection (a) or subsection (b) of this Section 4) must first send by electronic mail and overnight mail a written notice (the "Notice") to the corporation addressed to the Chief Executive Officer, at the corporation's United States headquarters; provided, however, that no Notice need be sent if the holder has verified that the conversion would not be subject to reporting requirements under the Hart-Scott-Rodino Antitrust Improvements Act of 1976 as amended (the "HSR Act"). The Notice must specify the number and class of shares that the holder wants to convert into shares of Common Stock. If any conversion (whether automatic or voluntary) would be subject to HSR Act reporting requirements, the conversion will not occur until the next business day after the applicable waiting period under the HSR Act has expired or been terminated. Any converted Preferred Stock shall be retired and cancelled and may not be reissued as shares of such series.

(d) <u>Conversion Price Adjustments of Preferred Stock for Certain Dilutive Issuances, Splits and Combinations</u>. The Conversion Price of the Preferred Stock shall be subject to adjustment from time to time as follows:

(1)

(A) If this corporation shall issue, on or after the date upon which this Amended and Restated Certificate of Incorporation is accepted for filing by the Secretary of State of the State of Delaware (the "Filing Date"), any Additional Stock (as defined below) without consideration or for a consideration per share less than the Conversion Price applicable to a series of Preferred Stock in effect immediately prior to the issuance of such Additional Stock, the Conversion Price for such

series in effect immediately prior to each such issuance shall forthwith (except as otherwise provided in this clause (i)) be adjusted to a price (calculated to the nearest one-thousandth of a cent) determined by multiplying such Conversion Price by a fraction, the numerator of which shall be the number of shares of Common Stock Outstanding (as defined below) immediately prior to such issuance plus the number of shares of Common Stock that the aggregate consideration received by this corporation for such issuance would purchase at such Conversion Price; and the denominator of which shall be the number of shares of Common Stock Outstanding (as defined below) immediately prior to such issuance plus the number of shares of such Additional Stock. For purposes of this Section 4(d)(i)(A), the term "Common Stock Outstanding" shall mean and include the following: (1) outstanding Common Stock, (2) Common Stock issuable upon conversion of outstanding Preferred Stock, (3) Common Stock issuable upon exercise of outstanding stock options and (4) Common Stock issuable upon exercise (and, in the case of warrants to purchase Preferred Stock, conversion) of outstanding warrants. Shares described in (1) through (4) above shall be included whether vested or unvested, whether contingent or non-contingent and whether exercisable or not yet exercisable. In the event that this corporation issues or sells, or is deemed to have issued or sold, shares of Additional Stock that results in an adjustment to a Conversion Price pursuant to the provisions of this Section 4(d) (the "First Dilutive Issuance"), and this corporation then issues or sells, or is deemed to have issued or sold, shares of Additional Stock in a subsequent issuance other than the First Dilutive Issuance that would result in further adjustment to a Conversion Price (a "Subsequent Dilutive Issuance") pursuant to the same instruments as the First Dilutive Issuance, then and in each such case, upon a Subsequent Dilutive Issuance, the applicable Conversion Price for each series of Preferred Stock shall be reduced to the applicable Conversion Price that would have been in effect had the First Dilutive Issuance and each Subsequent Dilutive Issuance all occurred on the closing date of the First Dilutive Issuance.

- (B) No adjustment of the Conversion Price for the Preferred Stock shall be made in an amount less than one-tenth of one cent per share. Except to the limited extent provided for in <u>subsections 4(d)(i)(E)(3)</u> and (E)(4), no adjustment of such Conversion Price pursuant to this <u>subsection 4(d)(i)</u> shall have the effect of increasing the Conversion Price above the Conversion Price in effect immediately prior to such adjustment.
- (C) In the case of the issuance of Additional Stock for cash, the consideration shall be deemed to be the amount of cash paid therefor before deducting any reasonable discounts, commissions or other expenses allowed, paid or incurred by this corporation for any underwriting or otherwise in connection with the issuance and sale thereof.

- (D) In the case of the issuance of the Additional Stock for a consideration in whole or in part other than cash, the consideration other than cash shall be deemed to be the fair market value thereof as determined in good faith by the Board of Directors, including a majority of the voting power of the Preferred Directors and the Remaining Director, collectively, irrespective of any accounting treatment.
- (E) In the case of the issuance of options to purchase or rights to subscribe for Common Stock, securities by their terms convertible into or exchangeable for Common Stock or options to purchase or rights to subscribe for such convertible or exchangeable securities, the following provisions shall apply for purposes of determining the number of shares of Additional Stock issued and the consideration paid therefor:
  - (1) The aggregate maximum number of shares of Common Stock deliverable upon exercise (assuming the satisfaction of any conditions to exercisability, including without limitation, the passage of time, but without taking into account potential anti-dilution adjustments) of such options to purchase or rights to subscribe for Common Stock shall be deemed to have been issued at the time such options or rights were issued and for a consideration equal to the consideration (determined in the manner provided in subsections  $4(\underline{d})(\underline{i})(\underline{C})$  and  $(\underline{d})(\underline{i})(\underline{D})$ , if any, received by this corporation upon the issuance of such options or rights plus the minimum exercise price provided in such options or rights (without taking into account potential anti-dilution adjustments) for the Common Stock covered thereby.
  - (2) The aggregate maximum number of shares of Common Stock deliverable upon conversion of, or in exchange (assuming the satisfaction of any conditions to convertibility or exchangeability, including, without limitation, the passage of time, but without taking into account potential anti-dilution adjustments) for, any such convertible or exchangeable securities or upon the exercise of options to purchase or rights to subscribe for such convertible or exchangeable securities and subsequent conversion or exchange thereof shall be deemed to have been issued at the time such securities were issued or such options or rights were issued and for a consideration equal to the consideration, if any, received by this corporation for any such securities and related options or rights (excluding any cash received on account of accrued interest or accrued dividends), plus the minimum additional consideration, if any, to be received by

this corporation (without taking into account potential anti-dilution adjustments) upon the conversion or exchange of such securities or the exercise of any related options or rights (the consideration in each case to be determined in the manner provided in subsections 4(d)(i)(C) and (d)(i)(D)).

- (3) In the event of any change in the number of shares of Common Stock deliverable or in the consideration payable to this corporation upon exercise of such options or rights or upon conversion of or in exchange for such convertible or exchangeable securities, the Conversion Price of the Preferred Stock, to the extent in any way affected by or computed using such options, rights or securities, shall be recomputed to reflect such change, but no further adjustment shall be made for the actual issuance of Common Stock or any payment of such consideration upon the exercise of any such options or rights or the conversion or exchange of such securities.
- (4) Upon the expiration of any such options or rights, the termination of any such rights to convert or exchange or the expiration of any options or rights related to such convertible or exchangeable securities, the Conversion Price of the Preferred Stock, to the extent in any way affected by or computed using such options, rights or securities or options or rights related to such securities, shall be recomputed to reflect the issuance of only the number of shares of Common Stock (and convertible or exchangeable securities that remain in effect) actually issued upon the exercise of such options or rights, upon the conversion or exchange of such securities or upon the exercise of the options or rights related to such securities.
- (5) The number of shares of Additional Stock deemed issued and the consideration deemed paid therefor pursuant to <u>subsections  $4(\underline{d})(\underline{i})(\underline{E})(\underline{1})$  and  $(\underline{2})$  shall be appropriately adjusted to reflect any change, termination or expiration of the type described in either <u>subsection  $4(\underline{d})(\underline{i})(\underline{E})(\underline{3})$  or  $(\underline{4})$ .</u></u>
- (ii) "Additional Stock" shall mean any shares of Common Stock issued (or deemed to have been issued pursuant to subsection  $\underline{4(\underline{d})(\underline{i})(\underline{E})}$ ) by this corporation on or after the Filing Date other than:
  - (A) Common Stock issued pursuant to a transaction described in <u>subsection 4(d)(iii)</u> hereof;

- (B) Common Stock issued or issuable to employees, directors, consultants and other service providers for the primary purpose of soliciting or retaining their services pursuant to plans or agreements approved by this corporation's Board of Directors;
  - (C) Common Stock issued pursuant to a Qualified Public Offering;
- (D) Common Stock issued pursuant to the conversion or exercise of convertible or exercisable securities outstanding on the Filing Date;
- (E) Common Stock issued as consideration in a bona fide business acquisition by this corporation, whether by merger, consolidation, sale of assets, sale or exchange of stock or otherwise which is approved by the Board of Directors;
- (F) Common Stock issued or deemed issued pursuant to subsection  $4(\underline{d})(\underline{i})(\underline{E})$  as a result of a decrease in the Conversion Price of any series of Preferred Stock resulting from the operation of Section  $4(\underline{d})$ ;
  - (G) Common Stock issued upon conversion of the Preferred Stock;
- (H) Common Stock issued pursuant to any equipment leasing arrangement or debt financing arrangement, which arrangement is approved by the Board of Directors and is primarily for non-equity financing purposes;
- (I) Common Stock issued to persons or entities with which this corporation has business relationships, provided such issuances are approved by the Board of Directors and are primarily for non-equity financing purposes; or
- (J) Common Stock that is issued, or approved to be issued (including approval on or prior to the Filing Date), with the unanimous approval of the Board of Directors of this corporation and the Board of Directors of this corporation specifically states that it shall not be Additional Stock.
- (iii) In the event this corporation should at any time or from time to time after the Filing Date fix a record date for the effectuation of a split or subdivision of the outstanding shares of Common Stock or the determination of holders of Common Stock entitled to receive a dividend or other distribution payable in additional shares of Common Stock or other securities or rights convertible into, or entitling the holder thereof to receive directly or indirectly, additional shares of Common Stock (hereinafter referred to as "Common Stock Equivalents") without payment of any consideration by such holder for the additional shares of Common Stock or the Common Stock

Equivalents (including the additional shares of Common Stock issuable upon conversion or exercise thereof), then, as of such record date (or the date of such dividend distribution, split or subdivision if no record date is fixed), the Conversion Price of the Preferred Stock shall be proportionately decreased so that the number of shares of Common Stock issuable on conversion of each share of such series shall be increased in proportion to such increase of the aggregate of shares of Common Stock outstanding and those issuable with respect to such Common Stock Equivalents with the number of shares issuable with respect to Common Stock Equivalents determined from time to time in the manner provided for deemed issuances in  $\underline{\text{subsection 4}(\underline{d})(\underline{i})(\underline{E})}$ .

- (iv) If the number of shares of Common Stock outstanding at any time after the Filing Date is decreased by a combination of the outstanding shares of Common Stock, then, following the record date of such combination, the Conversion Price for the Preferred Stock shall be appropriately increased so that the number of shares of Common Stock issuable on conversion of each share of such series shall be decreased in proportion to such decrease in outstanding shares.
- (e) Other Distributions. In the event this corporation shall declare a distribution payable in securities of other persons, evidences of indebtedness issued by this corporation or other persons, assets (excluding cash dividends) or options or rights not referred to in subsection  $\underline{4(\underline{d})(\underline{i}\underline{i}\underline{i})}$ , then, in each such case for the purpose of this subsection  $\underline{4(\underline{e})}$ , the holders of the Preferred Stock shall be entitled to a proportionate share of any such distribution as though they were the holders of the number of shares of Common Stock of this corporation into which their shares of Preferred Stock are convertible as of the record date fixed for the determination of the holders of Common Stock of this corporation entitled to receive such distribution.
- (f) <u>Recapitalizations</u>. If at any time or from time to time there shall be a reorganization, recapitalization, reclassification, consolidation or merger in which the Common Stock (but not the Preferred Stock) is converted into or exchanged for securities, cash, or other property (other than a subdivision, combination or merger or sale of assets transaction provided for elsewhere in this <u>Section 4</u> or in <u>Section 2</u>) provision shall be made so that the holders of the Preferred Stock shall thereafter be entitled to receive upon conversion of the Preferred Stock the number and kind of shares of stock or other securities, cash or property of this corporation or otherwise, to which a holder of Common Stock deliverable upon conversion would have been entitled on such recapitalization. In any such case, appropriate adjustment shall be made in the application of the provisions of this <u>Section 4</u> with respect to the rights of the holders of the Preferred Stock after the recapitalization to the end that the provisions of this <u>Section 4</u> (including adjustment of the Conversion Price then in effect and the number of shares purchasable upon conversion of the Preferred Stock) shall be applicable after that event as nearly equivalently as may be practicable.

#### (g) No Fractional Shares and Certificate as to Adjustments.

- (i) No fractional shares shall be issued upon the conversion of any share or shares of the Preferred Stock and the aggregate number of shares of Common Stock to be issued to particular stockholders, shall be rounded down to the nearest whole share and this corporation shall pay in cash the fair market value of any fractional shares as of the time when entitlement to receive such fractions is determined. Whether or not fractional shares would be issuable upon such conversion shall be determined on the basis of the total number of shares of Preferred Stock the holder is at the time converting into Common Stock and the number of shares of Common Stock issuable upon such conversion.
- (ii) Upon the occurrence of each adjustment or readjustment of the Conversion Price of Preferred Stock pursuant to this Section 4, this corporation, at its expense, shall promptly compute such adjustment or readjustment in accordance with the terms hereof and prepare and furnish to each holder of Preferred Stock a certificate setting forth such adjustment or readjustment and showing in detail the facts upon which such adjustment or readjustment is based. This corporation shall, upon the written request at any time of any holder of Preferred Stock, furnish or cause to be furnished to such holder a like certificate setting forth (A) such adjustment and readjustment, (B) the Conversion Price for such series of Preferred Stock at the time in effect, and (C) the number of shares of Common Stock and the amount, if any, of other property that at the time would be received upon the conversion of a share of Preferred Stock.
- (h) Notices of Record Date. In the event of (i) any taking by this corporation of a record of the holders of any class of securities for the purpose of determining the holders thereof who are entitled to receive any dividend (other than a cash dividend) or other distribution, or (ii) any capital reorganization of this corporation, any reclassification of Common Stock, or any Liquidation Event, this corporation shall mail to each holder of Preferred Stock, at least ten (10) days prior to the date specified therein, a notice specifying (x) the date on which any such record is to be taken for the purpose of such dividend or distribution, and the amount and character of such dividend or distribution, or (y) the effective date on which such reorganization, reclassification or Liquidation Event is proposed to take place, and the time, if any is to be fixed, as of which the holders of record of Common Stock (or such other capital stock or securities at the time issuable upon the conversion of the Preferred Stock) shall be entitled to exchange their shares of Common Stock (or such other capital stock or securities) for securities or other property deliverable upon such reorganization, reclassification, or Liquidation Event, and the amount per share and character of such exchange applicable to the Preferred Stock and the Common Stock.

(i) <u>Reservation of Stock Issuable Upon Conversion</u>. This corporation shall at all times reserve and keep available out of its authorized but unissued shares of Common Stock, solely for the purpose of effecting the conversion of the shares of the Preferred Stock, such number of its shares of Common Stock as shall from time to time be sufficient to effect the conversion of all outstanding shares of the Preferred Stock; and if at any time the number of authorized but unissued shares of Common Stock shall not be sufficient to

effect the conversion of all then outstanding shares of the Preferred Stock, in addition to such other remedies as shall be available to the holder of such Preferred Stock, this corporation will take such corporate action as may, in the opinion of its counsel, be necessary to increase its authorized but unissued shares of Common Stock to such number of shares as shall be sufficient for such purposes, including, without limitation, engaging in best efforts to obtain the requisite stockholder approval of any necessary amendment to this Amended and Restated Certificate of Incorporation.

(j) <u>Waiver of Adjustment to Conversion Price</u>. Notwithstanding anything herein to the contrary, any downward adjustment of the Conversion Price of any series of Preferred Stock may be waived, either prospectively or retroactively and either generally or in a particular instance, by the consent or vote of the holders of a majority of the outstanding shares of such series of Preferred Stock. Any such waiver shall bind all future holders of shares of such series of Preferred Stock.

#### 5. Voting Rights.

(a) <u>General Voting Rights</u>. The holder of each share of Preferred Stock shall have the right to one vote for each share of Common Stock into which such Preferred Stock could then be converted, and with respect to such vote, such holder shall have full voting rights and powers equal to the voting rights and powers of the holders of Common Stock, and shall be entitled, notwithstanding any provision hereof, to notice of any stockholders' meeting in accordance with the bylaws of this corporation (the "<u>Bylaws</u>"), and except as provided by law or in <u>subsection 5(b)</u> below with respect to the election of directors, shall be entitled to vote, together with holders of Common Stock, with respect to any question upon which holders of Common Stock have the right to vote. Fractional votes shall not, however, be permitted and any fractional voting rights available on an as-converted basis (after aggregating all shares into which shares of Preferred Stock held by each holder could be converted) shall be rounded to the nearest whole number (with one-half being rounded upward).

(b) <u>Voting for the Election of Directors</u>. As long as at least 1,000,000 shares of Series A Preferred Stock (as adjusted for any stock splits, stock dividends, combinations, subdivisions, recapitalizations or the like) remain outstanding, the holders of such shares of Series A Preferred Stock (voting as a separate series) shall be entitled to elect one director of this corporation at any election of directors (the "<u>Series A Director</u>"). As long as at least 1,000,000 shares of Series B Preferred Stock (as adjusted for any stock splits, stock dividends, combinations, subdivisions, recapitalizations or the like) remain outstanding, the holders of such shares of Series B Preferred Stock (voting as a separate series) shall be entitled to elect one director of this corporation at any election of directors (the "<u>Series B Director</u>"). As long as any shares of Series C Preferred Stock remain outstanding, the holders of such shares of Series C Preferred Stock (voting as a separate series) shall be entitled to elect one director of this corporation at any election of directors (the "<u>Series C Director</u>" and, together with the Series A Director and Series B Director, each, a "<u>Preferred Directors</u>"). The holders of outstanding Common Stock (voting as a separate series) shall be entitled to elect two (2) directors (each a "<u>Common Directors</u>") of this corporation at any election of directors. The holders of Series A Preferred Stock, Series B Preferred Stock, Series C Preferred Stock, and Common Stock (voting together as a single class and not as separate series, and on an as-converted basis) shall be entitled to elect one director of this corporation at any election of directors (the "<u>Remaining Director</u>").

(c) <u>Votes of Each Director; Quorum.</u> One Common Director ("<u>Common Director #1</u>"), in accordance with Section 141(d) of the General Corporation Law, shall be entitled to cast three (3) votes on each matter considered by the Board of Directors or any committee thereof (or any subcommittee of any committee). One Common Director ("<u>Common Director #2</u>"), in accordance with Section 141(d) of the General Corporation Law, shall be entitled to cast two (2) votes on each matter considered by the Board of Directors or any committee of any subcommittee of any subcommittee of any committee). Each other director shall be entitled to cast one vote on each matter considered by the Board of Directors or any committee thereof (or any subcommittee). Except as otherwise provided by applicable law, at all meetings of the Board of Directors 66.66% in voting power of the total number of directors authorized shall constitute a quorum for the transaction of business. Except as otherwise provided by applicable law, at all meetings of any committee of the Board of Directors (or any subcommittee of any such committee), a majority in voting power of the directors serving on such committee or subcommittee shall constitute a quorum for the transaction of business by such committee or subcommittee. For purposes of clarity, where there are six (6) authorized directors, by headcount, where one (1) such director is entitled to three (3) votes and one (1) such director is entitled to two (2) votes and where four (4) such directors are entitled to one (1) vote each, quorum shall only be achieved with the presence of six (6) director votes, regardless of whether any of such director seats is at such time vacant.

#### 6. Protective Provisions.

- (a) So long as at least 10,839,913 shares of Preferred Stock (as adjusted for any stock splits, stock dividends, combinations, subdivisions, recapitalizations or the like) remain outstanding, this corporation shall not, directly or indirectly (by amendment, merger, reorganization, consolidation, or otherwise), without (in addition to any other vote required by law or this Amended and Restated Certificate of Incorporation) first obtaining the approval (by vote or written consent, as provided by law) of the holders of a majority of the voting power of then outstanding shares of Preferred Stock (voting together as a single class and not as separate series, and on an as-converted basis):
  - (i) consummate a Liquidation Event or effect any other merger or consolidation of the corporation (whether consummated directly by the corporation or through a subsidiary);
  - (ii) amend, alter, repeal or waive any provision of this corporation's Amended and Restated Certificate of Incorporation or Bylaws;
  - (iii) authorize or issue any equity security (including any other security convertible into or exercisable for any such equity security) having a preference over, or being on a parity with, any series of Preferred Stock with respect to dividends, liquidation or redemption, other than the issuance of any authorized but unissued shares of Series H Preferred Stock designated in this Amended and Restated Certificate of Incorporation (including any security convertible into or exercisable for such shares of Preferred Stock);

- (iv) redeem, purchase or otherwise acquire (or pay into or set aside for a sinking fund for such purpose) any share or shares of Preferred Stock or Common Stock; <u>provided</u>, <u>however</u>, that this restriction shall not apply to the repurchase of shares of Common Stock from employees, officers, directors, consultants or other persons performing services for this corporation or any subsidiary pursuant to agreements under which this corporation has the option to repurchase such shares upon the occurrence of certain events, such as the termination of employment or service, or pursuant to a right of first refusal;
  - (v) change the authorized number of directors of this corporation;
- (vi) pay or declare any dividend on any shares of capital stock of this corporation other than (i) dividends payable on the Common Stock solely in the form of additional shares of Common Stock or (ii) dividends paid pursuant to Section 1(a) of this Article IV(B);
- (vii) effect an initial public offering of Common Stock (whether consummated directly by the corporation or through a subsidiary); or
- (viii) create, or, except as may already be the case on the Filing Date, hold capital stock in, any subsidiary that is not wholly owned (either directly or through one or more other subsidiaries) by this corporation, unless approved in advance (with an express reference to this provision) by the Board of Directors.
- (b) So long as any shares of Series E Preferred Stock are outstanding, this corporation shall not (by amendment, merger, reorganization, consolidation or otherwise) without (in addition to any other vote required by law or this Amended and Restated Certificate of Incorporation) first obtaining the approval (by vote or written consent, as provided by law) of the holders of a majority of the then outstanding shares of Series E Preferred Stock:
  - (i) amend, alter, repeal, waive or change the powers, preferences or rights of the shares of Series E Preferred Stock; or
  - (ii) increase or decrease (other than by redemption or conversion) the number of authorized shares of Series E Preferred Stock.

Notwithstanding anything to the contrary, and for the avoidance of doubt, the creation of any senior or *pari passu* security, in and of itself, shall not require the separate vote of the Series E Preferred Stock described in this Section 6(b).

(c) So long as any shares of Series F Preferred Stock are outstanding, this corporation shall not (by amendment, merger, reorganization, consolidation or otherwise) without (in addition to any other vote required by law or this Amended and Restated Certificate of Incorporation) first obtaining the approval (by vote or written consent, as provided by law) of the Series F Requisite Holders:

or

- (i) amend, alter, repeal, waive or change the powers, preferences or rights of the shares of Series F Preferred Stock;
- (ii) increase or decrease (other than by redemption or conversion) the number of authorized shares of Series F Preferred Stock;
- (iii) reclassify, alter or amend the Series G Preferred Stock, if such reclassification, alteration or amendment would render the Series G Preferred Stock senior to the Series F Preferred Stock in respect of any dividend, liquidation or redemption right, preference or privilege or (ii) reclassify, alter or amend any of the Prior Series, if such reclassification, alteration or amendment would render any of the Prior Series senior to or pari passu with the Series F Preferred Stock in respect of any dividend, liquidation or redemption right, preference or privilege.

Notwithstanding anything to the contrary, and for the avoidance of doubt, the creation of any senior or *pari passu* security, in and of itself, shall not require the separate vote of the Series F Preferred Stock described in this Section 6(g).

- (d) So long as any shares of Series G Preferred Stock are outstanding, this corporation shall not (by amendment, merger, reorganization, consolidation or otherwise) without (in addition to any other vote required by law or this Amended and Restated Certificate of Incorporation) first obtaining the approval (by vote or written consent, as provided by law) of the Series G Requisite Holders:
  - (i) amend, alter, repeal, waive or change the powers, preferences or rights of the shares of Series G Preferred Stock;
  - (ii) increase or decrease (other than by redemption or conversion) the number of authorized shares of Series G Preferred Stock; or
  - (iii) reclassify, alter or amend the Series F Preferred Stock, if such reclassification, alteration or amendment would render the Series F Preferred Stock senior to the Series G Preferred Stock in respect of any dividend, liquidation or redemption right, preference or privilege or (ii) reclassify, alter or amend any of the Prior Series, if such reclassification, alteration or amendment would render any of the Prior Series senior to or pari passu with the Series G Preferred Stock in respect of any dividend, liquidation or redemption right, preference or privilege.

Notwithstanding anything to the contrary, and for the avoidance of doubt, the creation of any senior or *pari passu* security, in and of itself, shall not require the separate vote of the Series G Preferred Stock described in this <u>Section 6(d)</u>.

- (e) So long as any shares of Series H Preferred Stock are outstanding, this corporation shall not (by amendment, merger, reorganization, consolidation or otherwise) without (in addition to any other vote required by law or this Amended and Restated Certificate of Incorporation) first obtaining the approval (by vote or written consent, as provided by law) of the holders of a majority of the outstanding shares of Series H Preferred Stock (voting as a separate series and on an as-converted basis):
  - (i) amend, alter, repeal, waive or change the powers, preferences or rights of the shares of Series H Preferred Stock (including, for the avoidance of doubt, any amendment, alteration, repeal, waiver or change to Sections 1(a), 2(a), 2(d), 2(e), 4(b), 4(j) or this 6(e) of Article IV(B));
  - (ii) increase or decrease (other than by redemption or conversion in accordance with this Amended and Restated Certificate of Incorporation) the number of authorized shares of Series H Preferred Stock; or
  - (iii) issue additional shares of Series H Preferred Stock unless such issuance is effected pursuant to the terms of the Series H Preferred Stock Purchase Agreement.

Notwithstanding anything to the contrary, and for the avoidance of doubt, the creation of any senior or *pari passu* security, in and of itself, shall not require the separate vote of the Series H Preferred Stock described in this <u>Section 6(e)</u>.

- 7. <u>Status of Converted Stock</u>. In the event any shares of Preferred Stock shall be converted pursuant to <u>Section 4</u> hereof, the shares so converted shall be cancelled and shall not be issuable by this corporation. The Amended and Restated Certificate of Incorporation of this corporation shall be appropriately amended to effect the corresponding reduction in this corporation's authorized capital stock.
- 8. Notices. Any notice required by the provisions of this Article IV(B) to be given to the holders of shares of Preferred Stock shall be deemed given (i) if deposited in the United States mail, postage prepaid, and addressed to each holder of record at his, her or its address appearing on the books of this corporation, (ii) if such notice is provided by electronic transmission in a manner permitted by Section 232 of the General Corporation Law, or (iii) if such notice is provided in another manner then permitted by the General Corporation Law.
- C. <u>Common Stock</u>. The rights, preferences, privileges and restrictions granted to and imposed on the Common Stock are as set forth below in this  $\underline{\text{Article IV}(C)}$ .
- 1. <u>Dividend Rights</u>. Subject to the prior rights of holders of all classes of stock at the time outstanding having prior rights as to dividends, the holders of the Common Stock shall be entitled to receive, when, as and if declared by the Board of Directors, out of any assets of this corporation legally available therefor, any dividends as may be declared from time to time by the Board of Directors.

- 2. <u>Liquidation Rights</u>. Upon the liquidation, dissolution or winding up of this corporation, the assets of this corporation shall be distributed as provided in <u>Section 2</u> of <u>Article IV(B)</u> hereof.
  - 3. Redemption. The Common Stock is not redeemable at the option of the holder.
- 4. <u>Voting Rights</u>. The holder of each share of Common Stock shall have the right to one vote for each such share, and shall be entitled to notice of any stockholders' meeting in accordance with the Bylaws, and shall be entitled to vote upon such matters and in such manner as may be provided by law; <u>provided</u>, <u>however</u>, that, except as otherwise required by law, holders of Common Stock, as such, shall not be entitled to vote on any amendment to this Amended and Restated Certificate of Incorporation that relates solely to the terms of one or more outstanding series of Preferred Stock if the holders of such affected series are entitled, either separately or together with the holders of one or more other such series, to vote thereon pursuant to this Amended and Restated Certificate of Incorporation or pursuant to the General Corporation Law. The number of authorized shares of Common Stock may be increased or decreased (but not below the number of shares thereof then outstanding) by the affirmative vote of the holders of a majority of the stock of this corporation entitled to vote, irrespective of the provisions of Section 242(b)(2) of the General Corporation Law.

#### ARTICLE V

Except as otherwise provided in this Amended and Restated Certificate of Incorporation, in furtherance and not in limitation of the powers conferred by statute, the Board of Directors is expressly authorized to make, repeal, alter, amend and rescind any or all of the Bylaws.

#### ARTICLE VI

Except as otherwise provided in any voting or stockholders agreement to which this corporation is a party, the number of directors of this corporation shall be determined in the manner set forth in the Bylaws.

#### ARTICLE VII

Elections of directors need not be by written ballot unless the Bylaws shall so provide.

#### ARTICLE VIII

Meetings of stockholders may be held within or without the State of Delaware, as the Bylaws may provide. The books of this corporation may be kept (subject to any provision contained in the statutes) outside the State of Delaware at such place or places as may be designated from time to time by the Board of Directors or in the Bylaws.

#### ARTICLE IX

A director of this corporation shall not be personally liable to this corporation or its stockholders for monetary damages for breach of fiduciary duty as a director, except for liability (i) for any breach of the director's duty of loyalty to this corporation or its stockholders, (ii) for acts or omissions not in good faith or that involve intentional misconduct or a knowing violation of law, (iii) under Section 174 of the General Corporation Law, or (iv) for any transaction from which the director derived any improper personal benefit. If the General Corporation Law is amended after approval by the stockholders of this Article IX to authorize corporate action further eliminating or limiting the personal liability of directors, then the liability of a director of this corporation shall be eliminated or limited to the fullest extent permitted by the General Corporation Law as so amended.

Any amendment, repeal or modification of the foregoing provisions of this <u>Article IX</u> by the stockholders of this corporation shall not adversely affect any right or protection of a director of this corporation existing at the time of, or increase the liability of any director of this corporation with respect to any acts or omissions of such director occurring prior to, such amendment, repeal or modification.

#### ARTICLE X

Except as otherwise provided in this Amended and Restated Certificate of Incorporation, this corporation reserves the right to amend, alter, change or repeal any provision contained in this Amended and Restated Certificate of Incorporation, in the manner now or hereafter prescribed by statute, and all rights conferred upon stockholders herein are granted subject to this reservation.

#### ARTICLE XI

1. Right to Indemnification of Directors and Officers. This corporation shall indemnify and hold harmless, to the fullest extent permitted by applicable law as it presently exists or may hereafter be amended, any person (an "Indemnified Person") who was or is made or is threatened to be made a party or is otherwise involved in any action, suit or proceeding, whether civil, criminal, administrative or investigative (a "Proceeding"), by reason of the fact that such person, or a person for whom such person is the legal representative, is or was a director or officer of this corporation or, while a director or officer of this corporation, is or was serving at the request of this corporation as a director, officer, employee or agent of another corporation or of a partnership, joint venture, limited liability company, trust, enterprise or nonprofit entity, including service with respect to employee benefit plans, against all liability and loss suffered and expenses (including attorneys' fees) reasonably incurred by such Indemnified Person in such Proceeding. Notwithstanding the preceding sentence, except as otherwise provided in Section 3 of this Article XI, this corporation shall be required to indemnify an Indemnified Person in connection with a Proceeding (or part thereof) commenced by such Indemnified Person only if the commencement of such Proceeding (or part thereof) by the Indemnified Person was authorized in advance by the Board of Directors.

- 2. <u>Prepayment of Expenses of Directors and Officers</u>. This corporation shall pay the expenses (including attorneys' fees) incurred by an Indemnified Person in defending any Proceeding in advance of its final disposition, <u>provided</u>, <u>however</u>, that, to the extent required by law, such payment of expenses in advance of the final disposition of the Proceeding shall be made only upon receipt of an undertaking by the Indemnified Person to repay all amounts advanced if it should be ultimately determined that the Indemnified Person is not entitled to be indemnified under this <u>Article XI</u> or otherwise.
- 3. <u>Claims by Directors and Officers</u>. If a claim for indemnification or advancement of expenses under this <u>Article XI</u> is not paid in full within thirty (30) days after a written claim therefor by the Indemnified Person has been received by this corporation, the Indemnified Person may file suit to recover the unpaid amount of such claim and, if successful in whole or in part, shall be entitled to be paid the expense of prosecuting such claim. In any such action this corporation shall have the burden of proving that the Indemnified Person is not entitled to the requested indemnification or advancement of expenses under applicable law.
- 4. <u>Indemnification of Employees and Agents.</u> This corporation may indemnify and advance expenses to any person who was or is made or is threatened to be made or is otherwise involved in any Proceeding by reason of the fact that such person, or a person for whom such person is the legal representative, is or was an employee or agent of this corporation or, while an employee or agent of this corporation, is or was serving at the request of this corporation as a director, officer, employee or agent of another corporation or of a partnership, joint venture, limited liability company, trust, enterprise or nonprofit entity, including service with respect to employee benefit plans, against all liability and loss suffered and expenses (including attorneys' fees) reasonably incurred by such person in connection with such Proceeding. The ultimate determination of entitlement to indemnification of persons who are non-director or officer employees or agents shall be made in such manner as is determined by the Board of Directors in its sole discretion. Notwithstanding the foregoing sentence, this corporation shall not be required to indemnify a person in connection with a Proceeding initiated by such person if the Proceeding was not authorized in advance by the Board of Directors.
- 5. <u>Advancement of Expenses of Employees and Agents</u>. This corporation may pay the expenses (including attorneys' fees) incurred by an employee or agent in defending any Proceeding in advance of its final disposition on such terms and conditions as may be determined by the Board of Directors.
- 6. <u>Non-Exclusivity of Rights</u>. The rights conferred on any person by this Article XI shall not be exclusive of any other rights which such person may have or hereafter acquire under any statute, provision of this Amended and Restated Certificate of Incorporation, the Bylaws, agreement, vote of stockholders or disinterested directors or otherwise.
- 7. Other Indemnification. This corporation's obligation, if any, to indemnify any person who was or is serving at its request as a director, officer or employee of another corporation, partnership, limited liability company, joint venture, trust, organization or other enterprise shall be reduced by any amount such person may collect as indemnification from such other corporation, partnership, limited liability company, joint venture, trust, organization or other enterprise.

8. <u>Insurance</u>. The Board of Directors may, to the full extent permitted by applicable law as it presently exists, or may hereafter be amended from time to time, authorize an appropriate officer or officers to purchase and maintain at this corporation's expense insurance: (a) to indemnify this corporation for any obligation which it incurs as a result of the indemnification of directors, officers and employees under the provisions of this <u>Article XI</u>; and (b) to indemnify or insure directors, officers and employees against liability in instances in which they may not otherwise be indemnified by this corporation under the provisions of this Article XI.

9. <u>Amendment or Repeal</u>. Any repeal or modification of the foregoing provisions of this <u>Article XI</u> shall not adversely affect any right or protection hereunder of any person in respect of any act or omission occurring prior to the time of such repeal or modification. The rights provided hereunder shall inure to the benefit of any Indemnified Person and such person's heirs, executors and administrators.

#### ARTICLE XII

To the extent one or more sections of any other state corporations code setting forth minimum requirements for this corporation's retained earnings and/or net assets are applicable to this corporation's repurchase of shares of Common Stock, such code sections shall not apply, to the greatest extent permitted by applicable law, in whole or in part with respect to repurchases by this corporation of its Common Stock from employees, officers, directors, advisors, consultants or other persons performing services for this corporation or any subsidiary pursuant to agreements under which this corporation has the right to repurchase such shares at cost upon the occurrence of certain events, such as the termination of employment. In the case of any such repurchases, distributions by this corporation may be made without regard to the "preferential dividends arrears amount" or any "preferential rights amount," as such terms may be defined in such other state's corporations code.

#### ARTICLE XIII

This corporation renounces, to the fullest extent permitted by law, any interest or expectancy of this corporation in, or in being offered an opportunity to participate in, any Excluded Opportunity. An "Excluded Opportunity" is any matter, transaction or interest that is presented to, or acquired, created or developed by, or which otherwise comes into the possession of (i) any director of this corporation who is not an employee of this corporation or any of its subsidiaries, or (ii) any holder of Series C Preferred Stock, Series D Preferred Stock, Series E Preferred Stock, Series F Preferred Stock, Series G Preferred Stock or Series H Preferred Stock or any partner, member, director, stockholder, employee or agent of any such holder, other than someone who is an employee of this corporation or any of its subsidiaries (collectively, "Covered Persons"), unless such matter, transaction or interest is presented to, or acquired, created or developed by, or otherwise comes into the possession of, a Covered Person expressly and solely in such Covered Person's capacity as a director of this corporation.

#### ARTICLE XIV

A. Forum Selection. Unless this corporation consents in writing to the selection of an alternative forum, the Court of Chancery of the State of Delaware (or, if the Court of Chancery does not have jurisdiction, the federal district court for the District of Delaware) shall, to the fullest extent permitted by law, be the sole and exclusive forum for (1) any derivative action or proceeding brought on behalf of this corporation, (2) any action asserting a claim of breach of a fiduciary duty owed by any director, officer or other employee of this corporation to this corporation or this corporation's stockholders, (3) any action arising pursuant to any provision of the General Corporation Law or this Amended and Restated Certificate of Incorporation or the Bylaws (as either may be amended from time to time), or (4) any action asserting a claim governed by the internal affairs doctrine. Any person or entity purchasing or otherwise acquiring or holding any interest in shares of capital stock of this corporation shall be deemed to have notice of and consented to the provisions of this <u>Article XIV</u>.

B. <u>Personal Jurisdiction</u>. If any action the subject matter of which is within the scope of Section A immediately above is filed in a court other than a court located within the State of Delaware (a "<u>Foreign Action</u>") in the name of any stockholder, such stockholder shall be deemed to have consented to (i) the personal jurisdiction of the state and federal courts located within the State of Delaware in connection with any action brought in any such court to enforce Section A immediately above (an "<u>FSC Enforcement Action</u>") and (ii) having service of process made upon such stockholder in any such FSC Enforcement Action by service upon such stockholder's counsel in the Foreign Action as agent for such stockholder.

\* \* \*

**THIRD**: The foregoing amendment and restatement was approved by the holders of the requisite number of shares of said corporation in accordance with Section 228 of the General Corporation Law.

**FOURTH**: That said Amended and Restated Certificate of Incorporation, which restates and integrates and further amends the provisions of this corporation's Amended and Restated Certificate of Incorporation, has been duly adopted in accordance with Sections 242 and 245 of the General Corporation Law.

IN WITNESS WHEREOF, this Amended and Restated Certificate of Incorporation has been executed by a duly authorized officer of this corporation on this  $18^{th}$  day of March, 2019.

/s/ Piotr Szulczewski

Piotr Szulczewski, President

#### AMENDED AND RESTATED

BYLAWS OF

CONTEXTLOGIC INC.

(A DELAWARE CORPORATION)

### TABLE OF CONTENTS

DTICI E	E I OFFICES	Pago
1.1	Registered Office	
1.1	Offices	1
		1
RTICLE	E II MEETINGS OF STOCKHOLDERS	1
2.1	Location	1
2.2	Timing	1
2.3	Notice of Meeting	1
2.4	Stockholders' Records	1
2.5	Special Meetings	2
2.6	Notice of Meeting	2
2.7	Business Transacted at Special Meeting	2
2.8	Quorum; Meeting Adjournment; Presence by Remote Means	2
2.9	Voting Thresholds	3
2.10		3
2.11	Action by Written Consent of Stockholders; Electronic Consent; Notice of Action	3
RTICLE	E III DIRECTORS	4
3.1	Authorized Directors	4
3.2	Vacancies	4
3.3	Board Authority	5
3.4	Location of Meetings	5
3.5	First Meeting	5
3.6	Regular Meetings	5
3.7	Special Meetings	5
3.8	Quorum	6
3.9	Action Without a Meeting	$\epsilon$
3.10	Telephonic Meetings	$\epsilon$
3.11	Committees	$\epsilon$
3.12	Minutes of Meetings	7
3.13		7
3.14	Removal of Directors	7
RTICLE	E IV NOTICES	7
4.1	Notice	7
4.2	Waiver of Notice	7
4.3	Electronic Notice	7
RTICLE	E V OFFICERS	۶
5.1	Required and Permitted Officers	8
5.2	Appointment of Required Officers	8
5.3	Appointment of Permitted Officers	8

i

5.4	Officer Compensation	9
5.5	Term of Office; Vacancies	9
5.6	Chairman Presides	9
5.7	Absence of Chairman	9
5.8	Powers of President	9
5.9	President's Signature Authority	9
5.10	Absence of President	9
5.11	Duties of Secretary	10
5.12	Duties of Assistant Secretary	10
5.13	Duties of Treasurer	10
5.14	Disbursements and Financial Reports	10
5.15	Treasurer's Bond	10
5.16	Duties of Assistant Treasurer	10
ARTICLE	VI CERTIFICATE OF STOCK	11
6.1	Stock Certificates	11
6.2	Facsimile Signatures	11
6.3	Lost Certificates	11
6.4	Transfer of Stock	12
6.5	Fixing a Record Date	12
6.6	Registered Stockholders	12
ARTICLE	VII GENERAL PROVISIONS	12
7.1	Dividends	12
7.2	Reserve for Dividends	12
7.3	Checks	12
7.4	Fiscal Year	13
7.5	Corporate Seal	13
7.6	Indemnification	13
7.7	Conflicts with Certificate of Incorporation	14
ARTICLE	VIII AMENDMENTS	14
ARTICLE	IX LOANS TO OFFICERS	14
ARTICLE	X RECORDS AND REPORTS	15

#### BYLAWS OF CONTEXTLOGIC INC.

#### ARTICLE I OFFICES

- 1.1 **Registered Office**. The registered office shall be in the City of Dover, County of Kent, State of Delaware.
- 1.2 **Offices**. The corporation may also have offices at such other places both within and without the State of Delaware as the Board of Directors may from time to time determine or the business of the corporation may require.

#### ARTICLE II MEETINGS OF STOCKHOLDERS

- 2.1 **Location**. All meetings of the stockholders for the election of directors shall be held in the County of San Francisco, State of California, at such place as may be fixed from time to time by the Board of Directors, or at such other place either within or without the State of Delaware as shall be designated from time to time by the Board of Directors and stated in the notice of the meeting; provided, however, that the Board of Directors may, in its sole discretion, determine that the meeting shall not be held at any place, but may instead be held solely by means of remote communication as authorized by Section 211 of the Delaware General Corporations Law ("DGCL"). Meetings of stockholders for any other purpose may be held at such time and place, if any, within or without the State of Delaware, as shall be stated in the notice of the meeting or in a duly executed waiver of notice thereof, or a waiver by electronic transmission by the person entitled to notice.
- 2.2 **Timing**. Annual meetings of stockholders, commencing with the year 2010, shall be held at such date and time as shall be designated from time to time by the Board of Directors and stated in the notice of the meeting, at which they shall elect by a plurality vote a Board of Directors, and transact such other business as may properly be brought before the meeting.
- 2.3 **Notice of Meeting**. Written notice of any stockholder meeting stating the place, if any, date and hour of the meeting, the means of remote communication, if any, by which stockholders and proxyholders may be deemed to be present in person and vote at such meeting, shall be given to each stockholder entitled to vote at such meeting not fewer than ten (10) nor more than sixty (60) days before the date of the meeting.
- 2.4 **Stockholders' Records**. The officer who has charge of the stock ledger of the corporation shall prepare and make, at least ten (10) days before every meeting of stockholders, a complete list of the stockholders entitled to vote at the meeting, arranged in alphabetical order, and showing the address (but not the electronic address or other electronic contact information) of each stockholder and the number of shares registered in the name of each

stockholder. Such list shall be open to the examination of any stockholder, for any purpose germane to the meeting for a period of at least 10 days prior to the meeting: (i) on a reasonably accessible electronic network, provided that the information required to gain access to such list is provided with the notice of the meeting, or (ii) during ordinary business hours, at the principal place of business of the corporation. In the event that the corporation determines to make the list available on an electronic network, the corporation may take reasonable steps to ensure that such information is available only to stockholders of the corporation. If the meeting is to be held at a place, then the list shall be produced and kept at the time and place of the meeting during the whole time thereof, and may be inspected by any stockholder who is present. If the meeting is to be held solely by means of remote communication, then the list shall also be open to the examination of any stockholder during the whole time of the meeting on a reasonably accessible electronic network, and the information required to access such list shall be provided with the notice of the meeting.

- 2.5 **Special Meetings**. Special meetings of the stockholders, for any purpose or purposes, unless otherwise prescribed by statute or by the certificate of incorporation, may be called by the president and shall be called by the president or secretary at the request in writing of a majority of the Board of Directors, or at the request in writing of stockholders owning at least fifty percent (50%) in amount of the entire capital stock of the corporation issued and outstanding and entitled to vote. Such request shall state the purpose or purposes of the proposed meeting.
- 2.6 **Notice of Meeting.** Written notice of a special meeting stating the place, date and hour of the meeting and the purpose or purposes for which the meeting is called, shall be given not fewer than ten (10) nor more than sixty (60) days before the date of the meeting, to each stockholder entitled to vote at such meeting. The means of remote communication, if any, by which stockholders and proxyholders may be deemed to be present in person and vote at such meeting shall also be provided in the notice.
- 2.7 **Business Transacted at Special Meeting**. Business transacted at any special meeting of stockholders shall be limited to the purposes stated in the notice.

#### 2.8 Quorum; Meeting Adjournment; Presence by Remote Means.

(a) Quorum; Meeting Adjournment. The holders of a majority of the stock issued and outstanding and entitled to vote thereat, present in person or represented by proxy, shall constitute a quorum at all meetings of the stockholders for the transaction of business except as otherwise provided by statute or by the certificate of incorporation. If, however, such quorum shall not be present or represented at any meeting of the stockholders, the stockholders entitled to vote thereat, present in person or represented by proxy, shall have power to adjourn the meeting from time to time, without notice other than announcement at the meeting, until a quorum shall be present or represented. At such adjourned meeting at which a quorum shall be present or represented, any business may be transacted that might have been transacted at the meeting as originally notified. If the adjournment is for more than thirty (30) days, or if after the adjournment a new record date is fixed for the adjourned meeting, a notice of the adjourned meeting shall be given to each stockholder of record entitled to vote at the meeting.

- (b) Presence by Remote Means. If authorized by the Board of Directors in its sole discretion, and subject to such guidelines and procedures as the Board of Directors may adopt, stockholders and proxyholders not physically present at a meeting of stockholders may, by means of remote communication:
  - (1) participate in a meeting of stockholders; and
- (2) be deemed present in person and vote at a meeting of stockholders whether such meeting is to be held at a designated place or solely by means of remote communication, provided that (i) the corporation shall implement reasonable measures to verify that each person deemed present and permitted to vote at the meeting by means of remote communication is a stockholder or proxyholder, (ii) the corporation shall implement reasonable measures to provide such stockholders and proxyholders a reasonable opportunity to participate in the meeting and to vote on matters submitted to the stockholders, including an opportunity to read or hear the proceedings of the meeting substantially concurrently with such proceedings, and (iii) if any stockholder or proxyholder votes or takes other action at the meeting by means of remote communication, a record of such vote or other action shall be maintained by the corporation.
- 2.9 **Voting Thresholds.** When a quorum is present at any meeting, the vote of the holders of a majority of the stock having voting power present in person or represented by proxy shall decide any question brought before such meeting, unless the question is one upon which by express provision of the statutes or of the certificate of incorporation, a different vote is required, in which case such express provision shall govern and control the decision of such question.
- 2.10 **Number of Votes Per Share.** Unless otherwise provided in the certificate of incorporation, each stockholder shall at every meeting of the stockholders be entitled to one vote by such stockholder or by proxy for each share of the capital stock having voting power held by such stockholder, but no proxy shall be voted on after three years from its date, unless the proxy provides for a longer period.

#### 2.11 Action by Written Consent of Stockholders; Electronic Consent; Notice of Action.

(a) Action by Written Consent of Stockholders. Unless otherwise provided by the certificate of incorporation, any action required or permitted to be taken at any annual or special meeting of the stockholders may be taken without a meeting, without prior notice and without a vote, if a consent in writing setting forth the action so taken, is signed in a manner permitted by law by the holders of outstanding stock having not less than the number of votes that would be necessary to authorize or take such action at a meeting at which all shares entitled to vote thereon were present and voted. Written stockholder consents shall bear the date of signature of each stockholder who signs the consent in the manner permitted by law and shall be delivered to the corporation as provided in subsection (b) below. No written consent shall be effective to take the action set forth therein unless, within sixty (60) days of the earliest dated consent delivered to the corporation in the manner provided above, written consents signed by a sufficient number of stockholders to take the action set forth therein are delivered to the corporation in the manner provided above.

(b) *Electronic Consent*. A telegram, cablegram or other electronic transmission consenting to an action to be taken and transmitted by a stockholder or proxyholder, or a person or persons authorized to act for a stockholder or proxyholder, shall be deemed to be written, signed and dated for the purposes of this section, provided that any such telegram, cablegram or other electronic transmission sets forth or is delivered with information from which the corporation can determine (1) that the telegram, cablegram or other electronic transmission was transmitted by the stockholder or proxyholder or by a person or persons authorized to act for the stockholder or proxyholder and (2) the date on which such stockholder or proxyholder or authorized person or persons transmitted such telegram, cablegram or electronic transmission is transmitted shall be deemed to be the date on which such consent was signed. No consent given by telegram, cablegram or other electronic transmission shall be deemed to have been delivered until such consent is reproduced in paper form and until such paper form is delivered to the corporation by delivery to its registered office in the State of Delaware, its principal place of business or an officer or agent of the corporation having custody of the book in which proceedings of meetings of stockholders are recorded. Delivery made to a corporation's registered office shall be made by hand or by certified or registered mail, return receipt requested. Notwithstanding the foregoing limitations on delivery, consents given by telegram, cablegram or other electronic transmission may be otherwise delivered to the principal place of business of the corporation or to an officer or agent of the corporation having custody of the book in which proceedings of meetings of stockholders are recorded if, to the extent and in the manner provided by resolution of the Board of Directors of the corporation.

(c) Notice of Action. Prompt notice of any action taken pursuant to this Section 2.11 shall be provided to the stockholders in accordance with Section 228(e) of the DGCL.

### ARTICLE III DIRECTORS

- 3.1 **Authorized Directors**. Subject to the corporation's certificate of incorporation, the total number of directors constituting the Board of Directors shall be determined by resolution of the Board of Directors or by the stockholders at the annual meeting of the stockholders, and each director elected shall hold office until his or her successor is elected and qualified. Directors need not be stockholders.
- 3.2 Vacancies. Unless otherwise provided in the corporation's certificate of incorporation, as it may be amended, vacancies and newly created directorships resulting from any increase in the authorized number of directors may be filled by a majority in voting power of the directors then in office, though less than a quorum, or by a sole remaining director, and the directors so chosen shall hold office until the next annual election and until their successors are duly elected and shall qualify, unless sooner displaced. If there are no directors in office, then an election of directors may be held in the manner provided by statute. If, at the time of filling any vacancy or any newly created directorship, the directors then in office shall constitute less than a

majority in voting power of the total number of directors authorized (as constituted immediately prior to any such increase), the Court of Chancery may, upon application of any stockholder or stockholders holding at least ten percent (10%) in voting power of the shares at the time outstanding having the right to vote for such directors, summarily order an election to be held to fill any such vacancies or newly created directorships, or to replace the directors chosen by the directors then in office.

- 3.3 **Board Authority**. The business of the corporation shall be managed by or under the direction of its Board of Directors, which may exercise all such powers of the corporation and do all such lawful acts and things as are not by statute or by the certificate of incorporation or by these bylaws directed or required to be exercised or done by the stockholders.
- 3.4 **Location of Meetings**. The Board of Directors of the corporation may hold meetings, both regular and special, either within or without the State of Delaware.
- 3.5 **First Meeting**. The first meeting of each newly elected Board of Directors shall be held at such time and place as shall be fixed by the vote of the stockholders at the annual meeting and no notice of such meeting shall be necessary to the newly elected directors in order to legally constitute the meeting, provided a quorum shall be present. In the event of the failure of the stockholders to fix the time or place of such first meeting of the newly elected Board of Directors, or in the event such meeting is not held at the time and place so fixed by the stockholders, the meeting may be held at such time and place as shall be specified in a notice given as hereinafter provided for special meetings of the Board of Directors, or as shall be specified in a written waiver signed by all of the directors.
- 3.6 **Regular Meetings**. Regular meetings of the Board of Directors may be held without notice at such time and at such place as shall from time to time be determined by the Board of Directors.
- 3.7 **Special Meetings**. Special meetings of the Board of Directors may be called by the president upon notice to each director; special meetings shall be called by the president or secretary in like manner and on like notice on the written request of two (2) directors unless the Board of Directors consists of only one director, in which case special meetings shall be called by the president or secretary in like manner and on like notice on the written request of the sole director. Notice of any special meeting shall be given to each director at his or her business or residence in writing, or by telegram, facsimile transmission, telephone communication or electronic transmission (provided, with respect to electronic transmission, that the director has consented to receive the form of transmission at the address to which it is directed). If mailed, such notice shall be deemed adequately delivered when deposited in the United States mails so addressed, with postage thereon prepaid, at least five (5) days before such meeting. If by telegram, such notice shall be deemed adequately delivered when the telegram is delivered to the telegraph company at least twenty-four (24) hours before such meeting. If by facsimile transmission or other electronic transmission, such notice shall be transmitted at least twenty-four (24) hours before such meeting. If by telephone, the notice shall be given at least twelve (12) hours prior to the time set for the meeting. Neither the business to be transacted at, nor the purpose of, any regular or special meeting of the Board of Directors need be specified in

the notice of such meeting, except for amendments to these Bylaws as provided under Section 8.1 of Article VIII hereof. A meeting may be held at any time without notice if all the directors are present (except as otherwise provided by law) or if those not present waive notice of the meeting in writing, either before or after such meeting.

- 3.8 **Quorum**. At all meetings of the Board of Directors 66.66% in voting power of the total number of directors authorized shall constitute a quorum for the transaction of business and any act of a majority in voting power of the directors present at any meeting at which there is a quorum shall be an act of the Board of Directors, except as may be otherwise specifically provided by statute or by the certificate of incorporation. If a quorum is not present at any meeting of the Board of Directors, the directors present thereat may, by a majority in voting power thereof, adjourn the meeting from time to time, without notice other than announcement at the meeting, until a quorum shall be present. For the avoidance of doubt, where there are six (6) authorized directors, by headcount, where one (1) such director is entitled to three (3) votes and one (1) such director is entitled to two (2) votes and where four (4) such directors are entitled to one (1) vote each, quorum shall only be achieved with the presence of six (6) director votes, regardless of whether any of such director seats is at such time vacant.
- 3.9 **Action Without a Meeting**. Unless otherwise restricted by the certificate of incorporation or these bylaws, any action required or permitted to be taken at any meeting of the Board of Directors or of any committee thereof may be taken without a meeting if all members of the Board of Directors or committee, as the case may be, consent thereto in writing or by electronic transmission, and the writing, writings, electronic transmission or transmissions are filed with the minutes of proceedings of the Board of Directors or committee.
- 3.10 **Telephonic Meetings**. Unless otherwise restricted by the certificate of incorporation or these bylaws, members of the Board of Directors or any committee designated by the Board of Directors may participate in a meeting of the Board of Directors or any committee, by means of conference telephone or other means of communication by which all persons participating in the meeting can hear each other, and such participation shall constitute presence in person at the meeting.
- 3.11 **Committees**. The Board of Directors may designate one or more committees, each committee to consist of one or more of the directors of the corporation. The Board of Directors may designate one or more directors as alternate members of any committee, who may replace any absent or disqualified member at any meeting of the committee.

In the absence or disqualification of a member of a committee, the member or members thereof present at any meeting and not disqualified from voting, whether or not he or she or they constitute a quorum, may unanimously appoint another member of the Board of Directors to act at the meeting in the place of any such absent or disqualified member.

Any such committee, to the extent provided in the resolution of the Board of Directors, shall have and may exercise all the powers and authority of the Board of Directors in the management of the business and affairs of the corporation, and may authorize the seal of the corporation to be affixed to all papers which may require it, but no such committee shall have the power or authority in reference to the following matters: (i) approving or adopting, or recommending to the stockholders, any action or matter expressly required by the DGCL to be submitted to stockholders for approval or (ii) adopting, amending or repealing any provision of these bylaws.

Except as otherwise provided by the certificate of incorporation or applicable law, at all meetings of any committee of the Board of Directors (or any subcommittee of any such committee), a majority in voting power of the directors serving on such committee or subcommittee shall constitute a quorum for the transaction of business by such committee or subcommittee, and the vote of a majority in voting power of the members of any committee or subcommittee present at a meeting at which a quorum is present shall be the act of the committee or subcommittee.

- 3.12 **Minutes of Meetings**. Each committee shall keep regular minutes of its meetings and report the same to the Board of Directors when required.
- 3.13 **Compensation of Directors**. Unless otherwise restricted by the certificate of incorporation or these bylaws, the Board of Directors shall have the authority to fix the compensation of directors. The directors may be paid their expenses, if any, of attendance at each meeting of the Board of Directors and may be paid a fixed sum for attendance at each meeting of the Board of Directors or a stated salary as director. No such payment shall preclude any director from serving the corporation in any other capacity and receiving compensation therefor. Members of special or standing committees may be allowed like compensation for attending committee meetings.
- 3.14 **Removal of Directors**. Unless otherwise provided by the certificate of incorporation or the DGCL, any director may be removed, with or without cause, by the holders of a majority in voting power of the shares entitled to vote for the election of such director.

## ARTICLE IV NOTICES

4.1 **Notice**. Unless otherwise provided in these bylaws, whenever, under the provisions of the statutes or of the certificate of incorporation or of these bylaws, notice is required to be given to any director or stockholder, it shall not be construed to mean personal notice, but such notice may be given in writing, by mail, addressed to such director or stockholder, at his or her address as it appears on the records of the corpo

ration, with postage thereon prepaid, and such notice shall be deemed to be given at the time when the same shall be deposited in the United States mail. Notice to directors may also be given by telegram.

4.2 **Waiver of Notice**. Whenever any notice is required to be given under the provisions of the statutes or of the certificate of incorporation or of these bylaws, a waiver thereof in writing, signed by the person or persons entitled to said notice, whether before or after the time stated therein, shall be deemed equivalent thereto.

#### 4.3 Electronic Notice.

(a) Electronic Transmission. Without limiting the manner by which notice otherwise may be given effectively to stockholders and directors, any notice to

stockholders or directors given by the corporation under any provision of the DGCL, the certificate of incorporation or these bylaws shall be effective if given by a form of electronic transmission consented to by the stockholder or director to whom the notice is given. Any such consent shall be revocable by the stockholder or director by written notice to the corporation. Any such consent shall be deemed revoked if (1) the corporation is unable to deliver by electronic transmission two consecutive notices given by the corporation in accordance with such consent and (2) such inability becomes known to the secretary or an assistant secretary of the corporation or to the transfer agent, or other person responsible for the giving of notice; provided, however, the inadvertent failure to treat such inability as a revocation shall not invalidate any meeting or other action.

- (b) Effective Date of Notice. Notice given pursuant to subsection (a) of this section shall be deemed given: (1) if by facsimile telecommunication, when directed to a number at which the stockholder or director has consented to receive notice; (2) if by electronic mail, when directed to an electronic mail address at which the stockholder or director has consented to receive notice; (3) if by a posting on an electronic network together with separate notice to the stockholder or director of such specific posting, upon the later of (i) such posting and (ii) the giving of such separate notice; and (4) if by any other form of electronic transmission, when directed to the stockholder or director. An affidavit of the secretary or an assistant secretary or of the transfer agent or other agent of the corporation that the notice has been given by a form of electronic transmission shall, in the absence of fraud, be prima facie evidence of the facts stated therein.
- (c) Form of Electronic Transmission. For purposes of these bylaws, "electronic transmission" means any form of communication, not directly involving the physical transmission of paper, that creates a record that may be retained, retrieved, and reviewed by a recipient thereof, and that may be directly reproduced in paper form by such a recipient through an automated process.

#### ARTICLE V OFFICERS

- 5.1 **Required and Permitted Officers**. The officers of the corporation shall be chosen by the Board of Directors and shall include a president, treasurer and a secretary. The Board of Directors may elect from among its members a Chairman of the Board and a Vice-Chairman of the Board. The Board of Directors may also choose one or more other officers as it shall deem necessary. Any number of offices may be held by the same person, unless the certificate of incorporation or these bylaws otherwise provide.
- 5.2 **Appointment of Required Officers**. The Board of Directors at its first meeting after each annual meeting of stockholders shall choose a president, a treasurer, and a secretary.
- 5.3 **Appointment of Permitted Officers**. The Board of Directors may appoint such other officers and agents as it shall deem necessary who shall hold their offices for such terms and shall exercise such powers and perform such duties as shall be determined from time to time by the Board of Directors.

- 5.4 Officer Compensation. The salaries of all officers and agents of the corporation shall be fixed by the Board of Directors.
- 5.5 **Term of Office; Vacancies**. The officers of the corporation shall hold office until their successors are chosen and qualify. Any officer elected or appointed by the Board of Directors may be removed at any time with or without cause by the affirmative vote of a majority of the Board of Directors. Any vacancy occurring in any office of the corporation shall be filled by the Board of Directors.

#### THE CHAIRMAN OF THE BOARD

- 5.6 **Chairman Presides**. The Chairman of the Board, if any, shall preside at all meetings of the Board of Directors and of the stockholders at which he or she shall be present. He or she shall have and may exercise such powers as are, from time to time, assigned to him or her by the Board of Directors and as may be provided by law.
- 5.7 **Absence of Chairman**. In the absence of the Chairman of the Board, the Vice-Chairman of the Board, if any, shall preside at all meetings of the Board of Directors and of the stockholders at which he or she shall be present. He or she shall have and may exercise such powers as are, from time to time, assigned to him or her by the Board of Directors and as may be provided by law.

#### THE PRESIDENT AND VICE-PRESIDENTS

- 5.8 **Powers of President**. Unless the Board of Directors appoints a chief executive officer, the president shall be the chief executive officer of the corporation; in the absence of the Chairman and Vice-Chairman of the Board he or she shall preside at all meetings of the stockholders and the Board of Directors; he or she shall have general and active management of the business of the corporation and shall see that all orders and resolutions of the Board of Directors are carried into effect.
- 5.9 **President's Signature Authority**. The president shall execute bonds, mortgages and other contracts requiring a seal, under the seal of the corporation, except where required or permitted by law to be otherwise signed and executed and except where the signing and execution thereof shall be expressly delegated by the Board of Directors to some other officer or agent of the corporation.
- 5.10 **Absence of President**. In the absence of the president or in the event of his or her inability or refusal to act, the vice-president, if any, (or in the event there be more than one vice-president, the vice-presidents in the order designated by the directors, or in the absence of any designation, then in the order of their election) shall perform the duties of the president, and when so acting, shall have all the powers of and be subject to all the restrictions upon the president. The vice-presidents shall perform such other duties and have such other powers as the Board of Directors may from time to time prescribe.

#### THE SECRETARY AND ASSISTANT SECRETARY

- 5.11 **Duties of Secretary**. The secretary shall attend all meetings of the Board of Directors and all meetings of the stockholders and record all the proceedings of the meetings of the corporation and of the Board of Directors in a book to be kept for that purpose and shall perform like duties for the standing committees when required. He or she shall give, or cause to be given, notice of all meetings of the stockholders and special meetings of the Board of Directors, and shall perform such other duties as may be prescribed by the Board of Directors or president, under whose supervision he or she shall be. He or she shall have custody of the corporate seal of the corporation and he or she, or an assistant secretary, shall have authority to affix the same to any instrument requiring it and when so affixed, it may be attested by his or her signature or by the signature of such assistant secretary. The Board of Directors may give general authority to any other officer to affix the seal of the corporation and to attest the affixing by his or her signature.
- 5.12 **Duties of Assistant Secretary**. The assistant secretary, or if there be more than one, the assistant secretaries in the order determined by the Board of Directors (or if there be no such determination, then in the order of their election) shall, in the absence of the secretary or in the event of his or her inability or refusal to act, perform the duties and exercise the powers of the secretary and shall perform such other duties and have such other powers as the Board of Directors may from time to time prescribe.

#### THE TREASURER AND ASSISTANT TREASURERS

- 5.13 **Duties of Treasurer**. The treasurer shall have the custody of the corporate funds and securities and shall keep full and accurate accounts of receipts and disbursements in books belonging to the corporation and shall deposit all moneys and other valuable effects in the name and to the credit of the corporation in such depositories as may be designated by the Board of Directors.
- 5.14 **Disbursements and Financial Reports**. He or she shall disburse the funds of the corporation as may be ordered by the Board of Directors, taking proper vouchers for such disbursements, and shall render to the president and the Board of Directors, at its regular meetings or when the Board of Directors so requires, an account of all his or her transactions as treasurer and of the financial condition of the corporation.
- 5.15 **Treasurer's Bond**. If required by the Board of Directors, the treasurer shall give the corporation a bond (which shall be renewed every six years) in such sum and with such surety or sureties as shall be satisfactory to the Board of Directors for the faithful performance of the duties of his or her office and for the restoration to the corporation, in case of his or her death, resignation, retirement or removal from office, of all books, papers, vouchers, money and other property of whatever kind in his or her possession or under his or her control belonging to the corporation.
- 5.16 **Duties of Assistant Treasurer**. The assistant treasurer, or if there shall be more than one, the assistant treasurers in the order determined by the Board of Directors (or if there be no such determination, then in the order of their election) shall, in the absence of the

treasurer or in the event of the treasurer's inability or refusal to act, perform the duties and exercise the powers of the treasurer and shall perform such other duties and have such other powers as the Board of Directors or these Bylaws may from time to time prescribe.

#### ARTICLE VI CERTIFICATE OF STOCK

6.1 **Stock Certificates**. Every holder of stock in the corporation shall be entitled to have a certificate, signed by or in the name of the corporation by, the Chairman or Vice-Chairman of the Board of Directors, or the president or a vice-president and the treasurer or an assistant treasurer, or the secretary or an assistant secretary of the corporation, certifying the number of shares owned by him or her in the corporation.

Certificates may be issued for partly paid shares and in such case upon the face or back of the certificates issued to represent any such partly paid shares, the total amount of the consideration to be paid therefor, and the amount paid thereon shall be specified.

If the corporation shall be authorized to issue more than one class of stock or more than one series of any class, the powers, designations, preferences and relative participating, optional or other special rights of each class of stock or series thereof and the qualification, limitations or restrictions of such preferences and/or rights shall be set forth in full or summarized on the face or back of the certificate which the corporation shall issue to represent such class or series of stock, provided that, except as otherwise provided in Section 202 of the DGCL, in lieu of the foregoing requirements, there may be set forth on the face or back of the certificate which the corporation shall issue to represent such class or series of stock, a statement that the corporation will furnish without charge to each stockholder who so requests the powers, designations, preferences and relative participating, optional or other special rights of each class of stock or series thereof and the qualifications, limitations or restrictions of such preferences and/or rights.

- 6.2 Facsimile Signatures. Any or all of the signatures on the certificate may be facsimile. In the event that any officer, transfer agent or registrar who has signed or whose facsimile signature has been placed upon a certificate shall have ceased to be such officer, transfer agent or registrar before such certificate is issued, the certificate may be issued by the corporation with the same effect as if such officer, transfer agent or registrar were still acting as such at the date of issue.
- 6.3 Lost Certificates. The Board of Directors may direct a new certificate or certificates to be issued in place of any certificate or certificates theretofore issued by the corporation alleged to have been lost, stolen or destroyed upon the making of an affidavit of that fact by the person claiming the certificate to be lost, stolen or destroyed. When authorizing such issuance of a new certificate or certificates, the Board of Directors may, in its discretion and as a condition precedent to the issuance, require the owner of such lost, stolen or destroyed certificate or certificates, or his or her legal representative, to advertise the same in such manner as it shall require and/or to give the corporation a bond in such sum as it may direct as indemnity against any claim that may be made against the corporation with respect to the certificate alleged to have been lost, stolen or destroyed.

- 6.4 **Transfer of Stock**. Upon surrender to the corporation or the transfer agent of the corporation of a certificate for shares duly endorsed or accompanied by proper evidence of succession, assignation or authority to transfer, it shall be the duty of the corporation to issue a new certificate to the person entitled thereto, cancel the old certificate and record the transaction upon its books.
- 6.5 Fixing a Record Date. In order that the corporation may determine the stockholders entitled to notice of or to vote at any meeting of stockholders or any adjournment thereof, or to express consent to corporate action in writing without a meeting, or entitled to receive payment of any dividend or other distribution or allotment of any rights, or entitled to exercise any rights in respect of any change, conversion or exchange of stock or for the purpose of any other lawful action, the Board of Directors may fix a record date which shall not be more than sixty (60) nor less than ten (10) days before the date of such meeting, nor more than sixty (60) days prior to any other action. A determination of stockholders of record entitled to notice of or to vote at a meeting of stockholders shall apply to any adjournment of the meeting; provided, however, that the Board of Directors may fix a new record date for the adjourned meeting.
- 6.6 **Registered Stockholders**. The corporation shall be entitled to recognize the exclusive right of a person registered on its books as the owner of shares to receive dividends, to vote as such owner, to hold liable for calls and assessments a person registered on its books as the owner of shares and shall not be bound to recognize any equitable or other claim to or interest in such share or shares on the part of any other person, whether or not it shall have express or other notice thereof, except as otherwise provided by the laws of Delaware.

#### ARTICLE VII GENERAL PROVISIONS

- 7.1 **Dividends**. Dividends upon the capital stock of the corporation, if any, subject to the provisions of the certificate of incorporation, may be declared by the Board of Directors at any regular or special meeting, pursuant to law. Dividends may be paid in cash, in property or in shares of the capital stock, subject to the provisions of the certificate of incorporation.
- 7.2 **Reserve for Dividends**. Before payment of any dividend, there may be set aside out of any funds of the corporation available for dividends such sum or sums as the directors from time to time, in their sole discretion, think proper as a reserve or reserves to meet contingencies, or for equalizing dividends, or for repairing or maintaining any property of the corporation, or for such other purposes as the directors think conducive to the interests of the corporation, and the directors may modify or abolish any such reserve in the manner in which it was created.
- 7.3 **Checks**. All checks or demands for money and notes of the corporation shall be signed by such officer or officers or such other person or persons as the Board of Directors may from time to time designate.

- 7.4 **Fiscal Year**. The fiscal year of the corporation shall be fixed by resolution of the Board of Directors.
- 7.5 **Corporate Seal**. The Board of Directors may adopt a corporate seal having inscribed thereon the name of the corporation, the year of its organization and the words "Corporate Seal, Delaware." The seal may be used by causing it or a facsimile thereof to be impressed or affixed or otherwise reproduced.
- 7.6 **Indemnification**. The corporation shall, to the fullest extent authorized under the laws of the State of Delaware, as those laws may be amended and supplemented from time to time, indemnify any director made, or threatened to be made, a party to an action or proceeding, whether criminal, civil, administrative or investigative, by reason of being a director of the corporation or a predecessor corporation or a director or officer of another corporation, if such person served in such position at the request of the corporation; provided, however, that the corporation shall indemnify any such director or officer in connection with a proceeding initiated by such director or officer only if such proceeding was authorized by the Board of Directors of the corporation. The indemnification provided for in this Section 7.6 shall: (i) not be deemed exclusive of any other rights to which those indemnified may be entitled under these bylaws, agreement or vote of stockholders or disinterested directors or otherwise, both as to action in their official capacities and as to action in another capacity while holding such office, (ii) continue as to a person who has ceased to be a director, and (iii) inure to the benefit of the heirs, executors and administrators of a person who has ceased to be a director. The corporation's obligation to provide indemnification under this Section 7.6 shall be offset to the extent of any other source of indemnification or any otherwise applicable insurance coverage under a policy maintained by the corporation or any other person.

Expenses incurred by a director of the corporation in defending a civil or criminal action, suit or proceeding by reason of the fact that he or she is or was a director of the corporation (or was serving at the corporation's request as a director or officer of another corporation) shall be paid by the corporation in advance of the final disposition of such action, suit or proceeding upon receipt of an undertaking by or on behalf of such director to repay such amount if it shall ultimately be determined that he or she is not entitled to be indemnified by the corporation as authorized by relevant sections of the DGCL. Notwithstanding the foregoing, the corporation shall not be required to advance such expenses to an agent who is a party to an action, suit or proceeding brought by the corporation and approved by a majority of the Board of Directors of the corporation that alleges willful misappropriation of corporate assets by such agent, disclosure of confidential information in violation of such agent's fiduciary or contractual obligations to the corporation or any other willful and deliberate breach in bad faith of such agent's duty to the corporation or its stockholders.

The foregoing provisions of this Section 7.6 shall be deemed to be a contract between the corporation and each director who serves in such capacity at any time while this bylaw is in effect, and any repeal or modification thereof shall not affect any rights or obligations then existing with respect to any state of facts then or theretofore existing or any action, suit or proceeding theretofore or thereafter brought based in whole or in part upon any such state of facts.

The Board of Directors in its sole discretion shall have power on behalf of the corporation to indemnify any person, other than a director, made a party to any action, suit or proceeding by reason of the fact that he or she, his or her testator or intestate, is or was an officer or employee of the corporation.

To assure indemnification under this Section 7.6 of all directors, officers and employees who are determined by the corporation or otherwise to be or to have been "fiduciaries" of any employee benefit plan of the corporation that may exist from time to time, Section 145 of the DGCL shall, for the purposes of this Section 7.6, be interpreted as follows: an "other enterprise" shall be deemed to include such an employee benefit plan, including without limitation, any plan of the corporation that is governed by the Act of Congress entitled "Employee Retirement Income Security Act of 1974," as amended from time to time; the corporation shall be deemed to have requested a person to serve the corporation for purposes of Section 145 of the DGCL, as administrator of an employee benefit plan where the performance by such person of his or her duties to the corporation also imposes duties on, or otherwise involves services by, such person to the plan or participants or beneficiaries of the plan; excise taxes assessed on a person with respect to an employee benefit plan pursuant to such Act of Congress shall be deemed "fines."

#### **CERTIFICATE OF INCORPORATION GOVERNS**

7.7 **Conflicts with Certificate of Incorporation**. In the event of any conflict between the provisions of the corporation's certificate of incorporation and these bylaws, the provisions of the certificate of incorporation shall govern.

#### ARTICLE VIII AMENDMENTS

8.1 These bylaws may be altered, amended or repealed, or new bylaws may be adopted by the stockholders or by the Board of Directors, when such power is conferred upon the Board of Directors by the certificate of incorporation at any regular meeting of the stockholders or of the Board of Directors or at any special meeting of the stockholders or of the Board of Directors if notice of such alteration, amendment, repeal or adoption of new bylaws be contained in the notice of such special meeting. If the power to adopt, amend or repeal bylaws is conferred upon the Board of Directors by the certificate of incorporation, it shall not divest or limit the power of the stockholders to adopt, amend or repeal bylaws.

#### ARTICLE IX LOANS TO OFFICERS

9.1 The corporation may lend money to, or guarantee any obligation of or otherwise assist any officer or other employee of the corporation or of its subsidiaries, including any officer or employee who is a director of the corporation or its subsidiaries, whenever, in the judgment of the Board of Directors, such loan, guarantee or assistance may reasonably be expected to benefit the corporation. The loan, guarantee or other assistance may be with or without interest and may be unsecured or secured in such manner as the Board of Directors shall approve, including, without limitation, a pledge of shares of stock of the corporation. Nothing in these bylaws shall be deemed to deny, limit or restrict the powers of guaranty or warranty of the corporation at common law or under any statute.

# ARTICLE X RECORDS AND REPORTS

10.1 The application and requirements of Section 1501 of the California General Corporation Law are hereby expressly waived to the fullest extent permitted thereunder.

# CERTIFICATE OF SECRETARY OF CONTEXTLOGIC INC.

The undersigned, Piotr Szulczewski, hereby certifies that he is the duly elected and acting Secretary of **CONTEXTLOGIC INC.**, a Delaware corporation (the "Corporation"), and that the Bylaws attached hereto constitute the Bylaws of said Corporation as duly adopted by Action by Unanimous Written Consent by the Board of Directors on March 14, 2019.

IN WITNESS WHEREOF, the undersigned has hereunto subscribed his name this 18th day of March, 2019.

/s/ Piotr Szulczewski

Piotr Szulczewski, Secretary

#### CONTEXTLOGIC INC.

# AMENDED AND RESTATED INVESTORS' RIGHTS AGREEMENT

March 18, 2019

#### TABLE OF CONTENTS

١.	Regis	tration Rights	Page 2
	1.1	Definitions	2
	1.2	Request for Registration	4
	1.3	Company Registration	6
	1.4	Form S-3 Registration; Shelf Take-Downs	7
	1.5	Obligations of the Company	10
	1.6	Information from Holder	12
	1.7	Expenses of Registration	12
	1.8	Delay of Registration	13
	1.9	Indemnification	13
	1.10	Reports Under the 1934 Act	15
	1.11	Assignment of Registration Rights	16
	1.12	Limitations on Subsequent Registration Rights	16
	1.13	"Market Stand-Off" Agreement	16
	1.14	Termination of Registration Rights	18
2.	Cove	nants of the Company	18
	2.1	Delivery of Financial Statements	18
	2.2	Inspection	19
	2.3	Termination of Information and Inspection Covenants	20
	2.4	Right of First Offer	20
	2.5	Proprietary Information and Inventions Agreements	22
	2.6	Successor Indemnification	22
	2.7	FCPA	22
	2.8	Confidentiality	22
	2.9	Termination of Certain Covenants	23
3.	Misco	ellaneous	23
	3.1	Successors and Assigns	23
	3.2	Governing Law	23
	3.3	Counterparts; Facsimile	23
	3.4	Titles and Subtitles	23

i

3.5	Notices	23
3.6	Expenses	24
3.7	Entire Agreement; Amendments and Waivers	24
3.8	Severability	25
3.9	Aggregation of Stock	25
3.10	Additional Investors	25
3.11	Dispute Resolution	25
3.12	Acknowledgment	26
3.13	Delays or Omissions	26
3.14	Specific Enforcement	26
3.15	Effect on Prior Agreement	26

### AMENDED AND RESTATED INVESTORS' RIGHTS AGREEMENT

THIS AMENDED AND RESTATED INVESTORS' RIGHTS AGREEMENT (the "Agreement") is made as of the 18th day of March 2019, by and among **CONTEXTLOGIC INC.**, a Delaware corporation (the "Company"), and the investors listed on Schedule A hereto, each of which is herein referred to as an "Investor" and collectively as the "Investors".

#### **RECITALS**

WHEREAS, certain of the Investors (the "Existing Investors") hold shares of the Company's Series A Preferred Stock, par value \$0.0001 per share (the "Series A Preferred Stock"), the Company's Series B Preferred Stock, par value \$0.0001 per share (the "Series B Preferred Stock, par value \$0.0001 per share (the "Series B Preferred Stock, par value \$0.0001 per share (the "Series D Preferred Stock, par value \$0.0001 per share (the "Series D Preferred Stock, par value \$0.0001 per share (the "Series B Preferred Stock, par value \$0.0001 per share (the "Series B Preferred Stock, par value \$0.0001 per share (the "Series G Preferred Stock, par value \$0.0001 per share

**WHEREAS**, the Prior Agreement may be amended, and any provision therein waived, with the consent of the Company and the holders of a majority of the outstanding Registrable Securities (as such term is defined in the Prior Agreement);

WHEREAS, the Existing Investors as holders of a majority of the outstanding Registrable Securities (as such term is defined in the Prior Agreement) desire to terminate the Prior Agreement and to accept the rights created pursuant hereto in lieu of the rights granted to them under the Prior Agreement; and

WHEREAS, certain Investors are parties to that certain Series H Preferred Stock Purchase Agreement, dated as of the date hereof (as it may be amended from time to time), by and among the Company and certain of the Investors (the "Series H Agreement"), which provides that as a condition to the closing of the sale of the Series H Preferred Stock, par value \$0.0001 per share (the "Series H Preferred Stock" and collectively with the Series A Preferred Stock, the Series B Preferred Stock, the Series C Preferred Stock, the Series E Preferred Stock, the Series F Preferred Stock and the Series G Preferred Stock, the "Preferred Stock"), this Agreement must be executed and delivered by such Investors, Existing Investors holding a majority of the outstanding Registrable Securities (as such term is defined in the Prior Agreement), and the Company.

**NOW, THEREFORE,** in consideration of the mutual promises and covenants set forth herein, the Company and the Existing Investors hereby agree that the Prior Agreement shall be superseded and replaced in its entirety by this Agreement, and the parties hereto further agree as follows:

- 1. Registration Rights. The Company covenants and agrees as follows:
  - 1.1 <u>Definitions</u>. For purposes of this Agreement:
    - (a) The term "1934 Act" means the Securities Exchange Act of 1934, as amended.
    - (b) The term "Act" means the Securities Act of 1933, as amended.
- (c) The term "Affiliate" means, with respect to any specified Person, any other Person who or which, directly or indirectly, controls, is controlled by, or is under common control with such specified Person, including, without limitation, any general partner, officer, director or manager of such person and any venture capital fund or other investment entity now or hereafter existing that is controlled by one or more general partners, managing members or managers of, or is under common investment management with, such Person; provided, however, that (i) each Wellington Investor shall be deemed to be an "Affiliate" of each other Wellington Investor, and (ii) an entity that is an "Affiliate" of a Wellington Investor shall not be deemed to be an "Affiliate" of any other Wellington Investor unless such entity is a Wellington Investor (and, for the avoidance of doubt, an "Affiliate" of such entity shall not be deemed an "Affiliate" of any Wellington Investor solely by virtue of being an "Affiliate" of such entity). For the avoidance of doubt, with respect to The Founders Fund V, LP, "Affiliate" shall include Founders Fund, LLC; Founders Fund V Management LLC; The Founders Fund V, LP; The Founders Fund V Principals Fund, LP; and The Founders Fund VI Deriver Fund VI Principals Fund, LP; and The Founders Fund VI Deriver Fund VI Principals Fund, LP; and The Founders Fund VI Entrepreneurs Fund VI Principals Fund, LP; and The Founders Fund VI Entrepreneurs Fund VI Principals Fund, LP; Peter Thiel, up to three (3) Founders Fund employee investment vehicles, or any partner or affiliate of any Permitted Founders Fund Entity, or any retirement accounts held on behalf of any such partner.
  - (d) The term "Board" means the Company's Board of Directors, as constituted from time to time.
- (e) The term "Form S-3" means such form under the Act as in effect on the date hereof or any registration form under the Act subsequently adopted by the SEC that permits inclusion or incorporation of substantial information by reference to other documents filed by the Company with the SEC.
  - (f) The term "Free Writing Prospectus" means a free-writing prospectus, as defined in Rule 405.
- (g) The term "General Atlantic" means General Atlantic (WI), L.P. or any of its Affiliates or permitted transferees that hold Registrable Securities.

- (h) The term "Holder" means any Person owning or having the right to acquire Registrable Securities or any assignee thereof in accordance with Section 1.11 hereof.
- (i) The term "<u>Initial Offering</u>" means the Company's first firm commitment underwritten public offering of its Common Stock under the Act.
  - (i) The term "Minimum Valuation" means \$8,750,000,000.
  - (k) The term "Person" means any individual, corporation, partnership, trust, limited liability company, association or other entity.
- (l) The terms "registered," "registered," and "registration" refer to a registration effected by preparing and filing a registration statement or similar document in compliance with the Act, and the declaration or ordering of effectiveness of such registration statement or document.
- (ii) any Common Stock acquired after the date of this Agreement in a secondary acquisition approved by the Board where the Board specifically references this Subsection (m)(ii) in its approval and states that such shares shall be Registrable Securities, and (iii) any Common Stock issued as (or issuable upon the conversion or exercise of any warrant, right or other security that is issued as) a dividend or other distribution with respect to, or in exchange for, or in replacement of, the shares referenced in (i) or (ii) above, excluding in all cases, however, any Registrable Securities sold by a Person in a transaction in which his rights under this Section 1 are not assigned. In addition, the number of shares of Registrable Securities outstanding shall equal the aggregate of the number of shares of Common Stock outstanding that are, and the number of shares of Common Stock issuable pursuant to then exercisable or convertible securities that are, Registrable Securities.
- (n) The term "Requisite Parties" means (i) the holders of a majority of the voting power of outstanding Preferred Stock and shares of Common Stock issued upon conversion of the Preferred Stock (voting together as a single class and not as separate series, and on an as-converted basis), (ii) the holders of a majority of the outstanding shares of Series E Preferred Stock (voting as a separate series and on an as-converted basis), (iii) the Series F Requisite Holders, (iv) the Series G Requisite Holders and (v) the Series H Requisite Holders.
- (o) The term "Restated Certificate" shall mean the Company's Amended and Restated Certificate of Incorporation, as amended and/or restated from time to time.
  - (p) The term "Rule 144" shall mean Rule 144 under the Act.
- (q) The term " $\underline{\text{Rule } 144(\underline{b})(\underline{1})(\underline{i})}$ " shall mean subsection (b)(1)(i) of Rule 144 under the Act as it applies to Persons who have held shares for more than one (1) year.
  - (r) The term "Rule 405" shall mean Rule 405 under the Act.

- (s) The term "SEC" shall mean the Securities and Exchange Commission.
- (t) The term "Series F Requisite Holders" means:
- (i) So long as Republic Technologies Pte. Ltd. owns at least 50% of the shares of Series F Preferred Stock originally issued to it, Republic Technologies Pte. Ltd.
- (ii) In the event Republic Technologies Pte. Ltd. owns less than 50% of the shares of Series F Preferred Stock originally issued to it, the holders of a majority of the then outstanding shares of Series F Preferred Stock (voting as a separate series).
- (u) The term "Series G New Investors" means Investors that, together with any of their respective Affiliates, did not purchase shares of Series A Preferred Stock, Series B Preferred Stock, Series C Preferred Stock, Series D Preferred Stock, Series E Preferred Stock or Series F Preferred Stock from the Company.

#### (v) The term "Series G Requisite Holders" means

- (i) So long as the Series G New Investors own at least 50% of the shares of Series G Preferred Stock originally issued to the Series G New Investors, (i) the holders of a majority of the then outstanding shares of Series G Preferred Stock held by the Series G New Investors and (ii) the holders of a majority of the then outstanding shares of Series G Preferred Stock (voting as a separate series).
- (ii) In the event the Series G New Investors own less than 50% of the shares of Series G Preferred Stock originally issued to the Series G New Investors, the holders of a majority of the then outstanding shares of Series G Preferred Stock (voting as a separate series).
- (w) The term "Series H Requisite Holders" means the holders of a majority of the then outstanding shares of Series H Preferred Stock (voting as a separate series and on an as-converted basis).
- (x) The term "Wellington" means Wellington Management Company LLP and any successor or affiliated investment advisor or subadvisor thereof to the Wellington Investors.
- (y) The term "<u>Wellington Investors</u>" means Investors, or permitted transferees of Registrable Securities held by Wellington Investors, that are advisory or subadvisory clients of Wellington.

#### 1.2 Request for Registration.

(a) Subject to the conditions of this Section 1.2, if the Company shall receive at any time after the earlier of (i) June 9, 2021 and (ii) one hundred eighty (180) days after the effective date of the Initial Offering, a written request from the Holders of fifty percent

(50%) or more of the Registrable Securities then outstanding (for purposes of this Section 1.2, the "<u>Initiating Holders</u>") that the Company file a registration statement under the Act covering the registration of Registrable Securities with an anticipated aggregate offering price, before underwriting discounts, commissions and fees, of at least \$15,000,000, then the Company shall, within twenty (20) days of the receipt thereof, give written notice of such request to all Holders, and subject to the limitations of this Section 1.2, use all commercially reasonable efforts to effect, as soon as practicable, the registration under the Act of all Registrable Securities that the Holders request to be registered in a written request received by the Company within twenty (20) days of the mailing of the Company's notice pursuant to this Section 1.2(a).

(b) If the Initiating Holders intend to distribute the Registrable Securities covered by their request by means of an underwriting, they shall so advise the Company as a part of their request made pursuant to this Section 1.2, and the Company shall include such information in the written notice referred to in Section 1.2(a). In such event the right of any Holder to include its Registrable Securities in such registration shall be conditioned upon such Holder's participation in such underwriting and the inclusion of such Holder's Registrable Securities in the underwriting (unless otherwise mutually agreed by a majority in interest of the Initiating Holders and such Holder) to the extent provided herein. All Holders proposing to distribute their securities through such underwriting shall enter into an underwriting agreement in customary form with the underwriter or underwriters selected for such underwriting by those Initiating Holders holding a majority of the Registrable Securities held by all Initiating Holders (which underwriter or underwriters shall be reasonably acceptable to the Company). Notwithstanding any other provision of this Section 1.2, if the underwriter advises the Company in writing that marketing factors require a limitation on the number of securities underwritten (including Registrable Securities), then the Company shall so advise all Holders of Registrable Securities that would otherwise be underwritten pursuant hereto, and the number of Registrable Securities held by all such Holders (including the Initiating Holders). In no event shall any Registrable Securities be excluded from such underwriting unless all other securities are first excluded. Any Registrable Securities excluded or withdrawn from such underwriting shall be withdrawn from the registration.

- (c) Notwithstanding the foregoing, the Company shall not be required to effect a registration pursuant to this Section 1.2:
- (i) in any particular jurisdiction in which the Company would be required to execute a general consent to service of process in effecting such registration, unless the Company is already subject to service in such jurisdiction and except as may be required under the Act; or
- (ii) after the Company has effected two (2) registrations pursuant to this Section 1.2, and such registrations have been declared or ordered effective; or
- (iii) during the period starting with the date sixty (60) days prior to the Company's good faith estimate of the date of the filing of and ending on a date one hundred eighty (180) days following the effective date of a Company-initiated registration

subject to Section 1.3 below, <u>provided</u> that the Company is actively employing in good faith all commercially reasonable efforts to cause such registration statement to become effective; or

(iv) if the Initiating Holders propose to dispose of Registrable Securities that may be registered on Form S-3 pursuant to Section 1.4 hereof; or

(v) if the Company shall furnish to Holders requesting a registration statement pursuant to this Section 1.2 a certificate signed by the Company's Chief Executive Officer or Chairman of the Board stating that in the good faith judgment of the Board, it would be seriously detrimental to the Company and its stockholders for such registration statement to be effected at such time, in which event the Company shall have the right to defer such filing for a period of not more than ninety (90) days after receipt of the request of the Initiating Holders, provided that such right shall be exercised by the Company not more than once in any twelve (12) month period and provided further that the Company shall not register any securities for the account of itself or any other stockholder during such ninety (90) day period (other than a registration relating solely to the sale of securities of participants in a Company stock plan, a registration relating to a corporate reorganization or transaction under Rule 145 of the Act, a registration on any form that does not include substantially the same information as would be required to be included in a registration statement covering the sale of the Registrable Securities, or a registration in which the only Common Stock being registered is Common Stock issuable upon conversion of debt securities that are also being registered).

#### 1.3 Company Registration.

(a) If (but without any obligation to do so) the Company proposes to register (including for this purpose a registration effected by the Company for stockholders other than the Holders) any of its stock or other securities under the Act in connection with the public offering of such securities (other than (i) a registration relating to a demand pursuant to Section 1.2, (ii) a registration relating solely to the sale of securities of participants in a Company stock plan, (iii) a registration relating to a corporate reorganization or transaction under Rule 145 of the Act, (iv) a registration on any form that does not include substantially the same information as would be required to be included in a registration statement covering the sale of the Registrable Securities, or (v) a registration in which the only Common Stock being registered is Common Stock issuable upon conversion of debt securities that are also being registered), the Company shall, at such time, promptly give each Holder written notice of such registration. Upon the written request of each Holder given within twenty (20) days after mailing of such notice by the Company in accordance with Section 3.5, the Company shall, subject to the provisions of Section 1.3(c), use all commercially reasonable efforts to cause to be registered under the Act all of the Registrable Securities that each such Holder requests to be registered.

(b) <u>Right to Terminate Registration</u>. The Company shall have the right to terminate or withdraw any registration initiated by it under this Section 1.3 prior to the effectiveness of such registration whether or not any Holder has elected to include securities in such registration. The expenses of such withdrawn registration shall be borne by the Company in accordance with Section 1.7 hereof.

(c) Underwriting Requirements. In connection with any offering involving an underwriting of shares of the Company's capital stock, the Company shall not be required under this Section 1.3 to include any of the Holders' securities in such underwriting unless they accept the terms of the underwriting as agreed upon between the Company and the underwriters selected by the Company (or by other Persons entitled to select the underwriters) and enter into an underwriting agreement in customary form with such underwriters, and then only in such quantity as the underwriters determine in their sole discretion will not jeopardize the success of the offering by the Company. If the total amount of securities, including Registrable Securities, requested by stockholders to be included in such offering exceeds the amount of securities sold other than by the Company that the underwriters determine in their sole discretion is compatible with the success of the offering, then the Company shall be required to include in the offering only that number of such securities, including Registrable Securities, that the underwriters determine in their sole discretion will not jeopardize the success of the offering. In no event shall any Registrable Securities be excluded from such offering unless all other stockholders' securities have been first excluded. In the event that the underwriters determine that less than all of the Registrable Securities requested to be registered can be included in such offering, then the Registrable Securities that are included in such offering shall be apportioned pro rata among the selling Holders based on the number of Registrable Securities held by all selling Holders or in such other proportions as shall mutually be agreed to by all such selling Holders. Notwithstanding the foregoing, in no event shall the amount of securities of the selling Holders included in the offering be reduced below thirty percent (30%) of the total amount of securities included in such offering, unless such offering is the Initial Offering, in which case the selling Holders may be excluded if the underwriters make the determination described above and no other stockholder's securities are included in such offering. For purposes of the preceding sentence concerning apportionment, for any selling stockholder that is a Holder of Registrable Securities and that is a venture capital fund, partnership, limited liability company or corporation, the affiliated venture capital funds, partners, retired partners, retired members and stockholders of such Holder, or the estates and family members of any such partners, members and retired partners and any trusts for the benefit of any of the foregoing Persons shall be deemed to be a single "selling Holder," and any pro rata reduction with respect to such "selling Holder" shall be based upon the aggregate amount of Registrable Securities owned by all such related entities and individuals.

Registrations effected pursuant to this Section 1.3 shall not be counted as requests for registration effected pursuant to Section 1.2 of this Agreement.

1.4 Form S-3 Registration; Shelf Take-Downs. After the Initial Offering and after such time as the Company is eligible to register securities on Form S-3, the Company shall use commercially reasonable efforts to qualify for registration on Form S-3 or any comparable or successor form, including a registration statement covering the sale or distribution of Registrable Securities from time to time by the Holders on a delayed or continuous basis pursuant to Rule 415 of the Act (a "Shelf Registration"). In case the Company shall receive from the Holders of Registrable Securities with an aggregate value of at least \$5,000,000 (for purposes of this Section 1.4, the "S-3 Initiating Holders") a written request or requests that the Company effect a registration on Form S-3, including a Shelf Registration, and any related qualification or compliance with respect to all or a part of the Registrable Securities owned by such Holder or Holders, the Company shall:

- (a) promptly give written notice of the proposed registration, and any related qualification or compliance, to all other Holders; and
- (b) use all commercially reasonable efforts to effect, as soon as practicable, such registration and all such qualifications and compliances as may be so requested and as would permit or facilitate the sale and distribution of all or such portion of such Holders' Registrable Securities as are specified in such request, together with all or such portion of the Registrable Securities of any other Holders joining in such request as are specified in a written request given within fifteen (15) days after receipt of such written notice from the Company, provided, however, that the Company shall not be obligated to effect any such registration, qualification or compliance or Underwritten Takedown or Shelf Take-Down, pursuant to this Section 1.4:
  - (i) if Form S-3 is not available for such offering by the Holders;
- (ii) if the Holders, together with the holders of any other securities of the Company entitled to inclusion in such registration or Underwritten Takedown or Shelf Take-Down, propose to sell Registrable Securities and such other securities (if any) at an aggregate price to the public (net of any underwriters' discounts or commissions) of less than \$5,000,000;
- (iii) if the Company shall furnish to all Holders requesting a registration statement or Underwritten Takedown or Shelf Take-Down pursuant to this Section 1.4 a certificate signed by the Company's Chief Executive Officer or Chairman of the Board stating that in the good faith judgment of the Board, it would be seriously detrimental to the Company and its stockholders for such registration statement or Underwritten Takedown or Shelf Take-Down to be effected at such time, in which event the Company shall have the right to defer such filing for a period of not more than ninety (90) days after receipt of the request of the S-3 Initiating Holders or the receipt of the Shelf Offering Request, provided that such right shall be exercised by the Company not more than twice in any twelve (12) month period and provided further that the Company shall not register any securities for the account of itself or any other stockholder during such ninety (90) day period (other than a registration relating solely to the sale of securities of participants in a Company stock plan, a registration relating to a corporate reorganization or transaction under Rule 145 of the Act, a registration on any form that does not include substantially the same information as would be required to be included in a registration statement covering the sale of the Registrable Securities, or a registration in which the only Common Stock being registered is Common Stock issuable upon conversion of debt that are also being registered);

(iv) if the Company has, within the twelve (12) month period preceding the date of such request, already effected one (1) registration or Underwritten Take-Down (not including Shelf Take-Downs that are not Underwritten Take-Downs) on Form S-3 pursuant to this Section 1.4;

- (v) in any particular jurisdiction in which the Company would be required to qualify to do business or to execute a general consent to service of process in effecting such registration, qualification or compliance; or
- (vi) if the Company, within thirty (30) days of receipt of the request of such S-3 Initiating Holders or the receipt of the Shelf Offering Request, gives notice of its bona fide intention to effect the filing of a registration statement with the SEC within one hundred twenty (120) days of receipt of such request (other than a registration effected solely to qualify an employee benefit plan or to effect a business combination pursuant to Rule 145), provided that the Company is actively employing in good faith all commercially reasonable efforts to cause such registration statement to become effective; or
- (vii) during the period starting with the date thirty (30) days prior to the Company's good faith estimate of the date of the filing of and ending on a date ninety (90) days following the effective date of a Company-initiated registration subject to Section 1.3 above, provided that the Company is actively employing in good faith all commercially reasonable efforts to cause such registration statement to become effective.
- (c) If the S-3 Initiating Holders intend to distribute the Registrable Securities covered by their request by means of an underwriting or intend for the registration to be a Shelf Registration, they shall so advise the Company as a part of their request made pursuant to this Section 1.4 and the Company shall include such information in the written notice referred to in Section 1.4(a). The provisions of Section 1.2(b) shall be applicable to such request (with the substitution of Section 1.4 for references to Section 1.2).
- (d) Subject to the foregoing, the Company shall file a registration statement covering the Registrable Securities and other securities so requested to be registered as soon as practicable after receipt of the request or requests of the S-3 Initiating Holders. Registrations effected pursuant to this Section 1.4 shall not be counted as requests for registration effected pursuant to Section 1.2.
- (e) In the event that a Shelf Registration is effective, the Holders shall have the right at any time or from time to time, on no more than two (2) occasions per year, to elect to sell pursuant to an offering (including an underwritten offering (an "<u>Underwritten Takedown</u>")) Registrable Securities available for sale pursuant to such Shelf Registration ("<u>Shelf Registrable Securities</u>"), so long as such Shelf Registration remains in effect and provided that the aggregate value of the offering is at least \$5,000,000. The Holders shall make such election by delivering to the Company a written request (a "<u>Shelf Offering Request</u>") for such offering specifying the number of Shelf Registrable Securities that such Holders desire to sell pursuant to such offering (the "<u>Shelf Take-Down</u>"). In the case of an Underwritten Takedown, as promptly as practicable, but no later than five (5) business days after receipt of a Shelf Offering Request, the Company shall give written notice (the "<u>Shelf Offering Notice</u>") of such Shelf Offering Request to all other holders of Shelf Registrable Securities. The Company shall include in such offering the Shelf Registrable Securities of any other Holder that shall have made a written request to the Company for inclusion in such Shelf Take-Down (which request shall specify the maximum number of Shelf Registrable Securities intended to be sold by such Holder) within ten (10) days after the receipt of the Shelf Offering Notice. The Company shall, as expeditiously as

possible (and in any event within ten (10) days after the receipt of a Shelf Offering Request, unless a longer period is agreed to by the Holders), use commercially reasonable efforts to facilitate such Shelf Take-Down. If the Holders request an Underwritten Takedown, then the Company shall have no obligation to effect such Underwritten Takedown if the Company has, within the twelve (12) month period preceding the date of such request, already effected one (1) Underwritten Takedown pursuant to this Section 1.4. The Holders shall be permitted to sell Registrable Securities pursuant to the Shelf Take-Down for a period of thirty (30) days from the date of the prospectus supplement relating to such Shelf Take-Down.

- (i) For any Shelf Take-Down that is conducted as an underwritten offering, the right of each Holder to have Registrable Securities included in a Shelf Take-Down shall be conditioned upon each Holder complying with the provisions of Section 1.2(b) hereof, as if such Shelf Take-Down were a registration pursuant to Section 1.2(a).
- (ii) Shelf Take-Downs effected pursuant to this Section 1.4 shall not be counted as requests for registration effected pursuant to Section 1.2 of this Agreement.
- 1.5 <u>Obligations of the Company</u>. Whenever required under this Section 1 to effect the registration of any Registrable Securities or effect a Shelf Take-Down, the Company shall, as expeditiously as reasonably possible:
- (a) in the case of effecting the registration of any Registrable Securities, prepare and file with the SEC a registration statement with respect to such Registrable Securities and use all commercially reasonable efforts to cause such registration statement to become effective, and, upon the request of the Holders of a majority of the Registrable Securities registered thereunder, keep such registration statement effective for a period of up to one hundred twenty (120) days or, if earlier, until the distribution contemplated in the Registration Statement has been completed; provided, however, that in the case of any registration of Registrable Securities on Form S-3 which are intended to be offered on a continuous or delayed basis, such one hundred twenty (120) day period shall be extended, if necessary, to keep the registration statement continuously effective, supplemented and amended to the extent necessary to ensure that it is available for sales of such Registrable Securities, and to ensure that it conforms with the requirements of this Agreement, the Act and the policies, rules and regulations of the SEC as announced from time to time, until the earlier of when (i) the Holders have sold all such Registrable Securities and (ii) such time as the registration statement expires;
- (b) prepare and file with the SEC such amendments and supplements to such registration statement and the prospectus used in connection with such registration statement as may be necessary to comply with the provisions of the Act with respect to the disposition of all securities covered by such registration statement and, in connection with any registration on Form S-3, use reasonable, diligent efforts to timely file all reports required under the 1934 Act in order to maintain the right to continue to use such Form S-3;

- (c) furnish to the Holders such number of copies of a prospectus, including a preliminary prospectus and any Free Writing Prospectus, in conformity with the requirements of the Act, and such other documents as they may reasonably request in order to facilitate the disposition of Registrable Securities owned by them;
- (d) use all commercially reasonable efforts to register and qualify the securities covered by such registration statement under such other securities or Blue Sky laws of such jurisdictions as shall be reasonably requested by the Holders, <u>provided</u> that the Company shall not be required in connection therewith or as a condition thereto to qualify to do business or to file a general consent to service of process in any such states or jurisdictions, unless the Company is already subject to service of process in such jurisdiction;
- (e) in the event of any underwritten public offering, enter into and perform its obligations under an underwriting agreement, in usual and customary form, with the managing underwriter of such offering;
- (f) notify each Holder of Registrable Securities covered by such registration statement at any time when a prospectus or Free Writing Prospectus (to the extent prepared by or on behalf of the Company) relating thereto is required to be delivered under the Act of the happening of any event as a result of which the prospectus included in such registration statement, as then in effect, includes an untrue statement of a material fact or omits to state a material fact required to be stated therein or necessary to make the statements therein not misleading in the light of the circumstances then existing, and, at the request of any such Holder, the Company will, as soon as reasonably practicable, file and furnish to all such Holders a supplement or amendment to such prospectus or Free Writing Prospectus (to the extent prepared by or on behalf of the Company) so that, as thereafter delivered to the purchasers of such Registrable Securities, such prospectus will not contain an untrue statement of a material fact or omit to state any fact necessary to make the statements therein not misleading in light of the circumstances under which they were made;
- (g) cause all such Registrable Securities registered pursuant to this Section 1 to be listed on a national exchange or trading system and on each securities exchange and trading system on which similar securities issued by the Company are then listed;
- (h) provide a transfer agent and registrar for all Registrable Securities registered pursuant to this Agreement and a CUSIP number for all such Registrable Securities, in each case not later than the effective date of such registration;
- (i) promptly make available for inspection by the selling Holders, any managing underwriter(s) participating in any disposition pursuant to such registration statement, and any attorney or accountant or other agent retained by any such underwriter or selected by the selling Holders, all financial and other records, pertinent corporate documents, and properties of the Company, and cause the Company's officers, directors, employees, and independent accountants to supply all information reasonably requested by any such seller, underwriter, attorney, accountant, or agent, in each case, as necessary or advisable to verify the accuracy of the information in such registration statement and to conduct appropriate due diligence in connection therewith;

- (j) notify each selling Holder, promptly after the Company receives notice thereof, of the time when such registration statement has been declared effective or a supplement to any prospectus forming a part of such registration statement has been filed; and
- (k) after such registration statement becomes effective, notify each selling Holder of any request by the SEC that the Company amend or supplement such registration statement or prospectus.

Notwithstanding the provisions of this Section 1, the Company shall be entitled to postpone or suspend, for a reasonable period of time, the filing, effectiveness or use of, or trading under, any registration statement if, in the good faith judgment of the Board, the Board determines that any such filing or the sale of any securities pursuant to such registration statement would:

- (i) materially impede, delay or interfere with any material pending or proposed financing, acquisition, corporate reorganization or other similar transaction involving the Company for which the Board has authorized negotiations;
- (ii) materially adversely impair the consummation of any pending or proposed material offering or sale of any class of securities by the Company; or
- (iii) require disclosure of material nonpublic information that, if disclosed at such time, would be materially harmful to the interests of the Company and its stockholders; <u>provided</u>, <u>however</u>, that during any such period all executive officers and directors of the Company are also prohibited from selling securities of the Company (or any security of any of the Company's subsidiaries or affiliates) other than through pre-existing trading plans pursuant to Rule 10b5-1 of the Exchange Act.

In the event of the suspension of effectiveness of any registration statement pursuant to this Section 1.5, the applicable time period during which such registration statement is to remain effective shall be extended by that number of days equal to the number of days the effectiveness of such registration statement was suspended.

- 1.6 <u>Information from Holder</u>. It shall be a condition precedent to the obligations of the Company to take any action pursuant to this Section 1 with respect to the Registrable Securities or Shelf Registrable Securities of any selling Holder that such Holder shall furnish to the Company such information regarding itself, the Registrable Securities held by it, and the intended method of disposition of such securities as shall be reasonably required to effect the registration of such Holder's Registrable Securities or to effect a Shelf Take-Down or an Underwritten Takedown.
- 1.7 Expenses of Registration. All expenses other than underwriting discounts and commissions incurred in connection with registrations, filings or qualifications pursuant to Sections 1.2, 1.3 and 1.4, including, without limitation, all registration, filing and qualification fees, printers' and accounting fees, fees and disbursements of counsel for the Company and the reasonable fees and disbursements of one counsel for the selling Holders (not to exceed \$50,000) shall be borne by the Company. Notwithstanding the foregoing, the Company shall not be required to pay for any expenses of any registration proceeding begun pursuant to Section 1.2 or

Section 1.4 if the registration request or Shelf Offering Request is subsequently withdrawn at the request of the Holders of a majority of the Registrable Securities to be registered or included in a the Shelf Take-Down (in which case all participating Holders shall bear such expenses pro rata based upon the number of Registrable Securities that were to be included in the withdrawn registration or Shelf Take-Down), unless, in the case of a registration requested under Section 1.2, the Holders of a majority of the Registrable Securities agree to forfeit their right to one demand registration pursuant to Section 1.2 or in the case of a Shelf Take-Down, such Holders agree to forfeit their right to a Shelf Take-Down for one year and provided, however, that if at the time of such withdrawal, the Holders have learned of a material adverse change in the condition, business or prospects of the Company from that known to the Holders at the time of their request and have withdrawn the request with reasonable promptness following disclosure by the Company of such material adverse change, then the Holders shall not be required to pay any of such expenses and shall retain their rights pursuant to Section 1.2 and 1.4.

- 1.8 <u>Delay of Registration</u>. No Holder shall have any right to obtain or seek an injunction restraining or otherwise delaying any such registration as the result of any controversy that might arise with respect to the interpretation or implementation of this Section 1.
  - 1.9 Indemnification. In the event any Registrable Securities are included in a registration statement under this Section 1:
- (a) To the extent permitted by law, the Company will indemnify and hold harmless each Holder, the partners, members, officers, directors and stockholders of each Holder, legal counsel and accountants for each Holder, any underwriter (as defined in the Act) for such Holder and each Person, if any, who controls such Holder or underwriter within the meaning of the Act or the 1934 Act, against any losses, claims, damages or liabilities (joint or several) to which they may become subject under the Act, the 1934 Act, any state securities laws or any rule or regulation promulgated under the Act, the 1934 Act or any state securities laws, insofar as such losses, claims, damages, or liabilities (or actions or proceedings, whether commenced or threatened, in respect thereof) arise out of or are based upon any of the following statements, omissions or violations (collectively a "Violation"): (i) any untrue statement or alleged untrue statement of a material fact contained in such registration statement, including any preliminary prospectus, final prospectus, or Free Writing Prospectus contained therein or any amendments or supplements thereto, any issuer information (as defined in Rule 433 of the Act) filed or required to be filed pursuant to Rule 433(d) under the Act or any other document incident to such registration prepared by or on behalf of the Company or used or referred to by the Company, (ii) the omission or alleged omission to state in such registration statement a material fact required to be stated therein, or necessary to make the statements therein not misleading or (iii) any violation or alleged violation by the Company of the Act, the 1934 Act, any state securities laws or any rule or regulation promulgated under the Act, the 1934 Act or any state securities laws, and the Company will reimburse each such Holder, underwriter, controlling Person or other aforementioned Person for any legal or other expenses reasonably incurred by them in connection with investigating or defending any such loss, claim, damage, liability or action as such expenses are incurred; provided, however, that the indemnity agreement contained in this subsection 1.9(a) shall not apply to amounts paid in settlement of any such loss, claim, damage, liability, action or proceeding if such settlement is effected without the consent of the Company (which consent shall not be unreasonably withheld), nor shall the

Company be liable in any such case for any such loss, claim, damage, liability, action or proceeding to the extent that it arises out of or is based upon a Violation that occurs in reliance upon and in conformity with written information furnished expressly for use in connection with such registration by any such Holder, underwriter, controlling Person or other aforementioned Person.

(b) To the extent permitted by law, each selling Holder, severally and not jointly, will indemnify and hold harmless the Company, each of its directors, each of its officers who has signed the registration statement, each Person, if any, who controls the Company within the meaning of the Act, legal counsel and accountants for the Company, any underwriter, any other Holder selling securities in such registration statement and any controlling Person of any such underwriter or other Holder, against any losses, claims, damages or liabilities (joint or several) to which any of the foregoing Persons may become subject, under the Act, the 1934 Act, any state securities laws or any rule or regulation promulgated under the Act, the 1934 Act or any state securities laws, insofar as such losses, claims, damages or liabilities (or actions or proceedings, whether commenced or threatened, in respect thereto) arise out of or are based upon any Violation, in each case to the extent (and only to the extent) that such Violation occurs in reliance upon and in conformity with written information furnished by such Holder expressly for use in connection with such registration; and each such Holder will reimburse any Person intended to be indemnified pursuant to this subsection 1.9(b) for any legal or other expenses reasonably incurred by such Person in connection with investigating or defending any such loss, claim, damage, liability, action or proceeding as such expenses are incurred; provided, however, that the indemnity agreement contained in this subsection 1.9(b) shall not apply to amounts paid in settlement of any such loss, claim, damage, liability, action or proceeding if such settlement is effected without the consent of the Holder (which consent shall not be unreasonably withheld), and provided that in no event shall any indemnity under this subsection 1.9(b) exceed the net proceeds from the offering received by such Holder.

(c) Promptly after receipt by an indemnified party under this Section 1.9 of notice of the commencement of any action or proceeding (including any governmental action or proceeding) for which a party may be entitled to indemnification, such indemnified party will, if a claim in respect thereof is to be made against any indemnifying party under this Section 1.9, deliver to the indemnifying party a written notice of the commencement thereof and the indemnifying party shall have the right to participate in and, to the extent the indemnifying party so desires, jointly with any other indemnifying party similarly noticed, to assume the defense thereof with counsel mutually satisfactory to the parties; provided, however, that an indemnified party (together with all other indemnified parties that may be represented without conflict by one counsel) shall have the right to retain one (1) separate counsel, with the fees and expenses to be paid by the indemnifying party, if representation of such indemnified party by the counsel retained by the indemnifying party would be inappropriate due to actual or potential differing interests between such indemnified party and any other party represented by such counsel in such proceeding. The failure to deliver written notice to the indemnifying party within a reasonable time of the commencement of any such action or proceeding, if materially prejudicial to its ability to defend such action or proceeding, shall relieve such indemnifying party of liability to the indemnified party under this Section 1.9 to the extent of such material prejudice, but the omission to so deliver written notice to the indemnifying party otherwise than under this Section 1.9.

No indemnifying party, in the defense of any such action or proceeding, shall, except with the consent of each indemnified party, consent to entry of judgment or enter into any settlement which does not include as an unconditional term thereof the giving by the claimant or plaintiff to such indemnified party of a release from all liability in respect to such action or proceeding.

- (d) If the indemnification provided for in this Section 1.9 is held by a non-appealable order from a court of competent jurisdiction to be unavailable to an indemnified party with respect to any loss, liability, claim, damage or expense referred to herein, then the indemnifying party, in lieu of indemnifying such indemnified party hereunder, shall contribute to the amount paid or payable by such indemnified party as a result of such loss, liability, claim, damage or expense in such proportion as is appropriate to reflect the relative fault of the indemnifying party on the one hand and the indemnified party on the other hand in connection with the statements or omissions that resulted in such loss, liability, claim, damage or expense, as well as any other relevant equitable considerations; provided, however, that (i) no contribution by any Holder, when combined with any amounts paid by such Holder pursuant to Section 1.9(b), shall exceed the net proceeds from the offering received by such Holder and (ii) no Person guilty of fraudulent misrepresentation (within the meaning of Section 11(f) of the Act) will be entitled to contribution from any Person who was not guilty of such fraudulent misrepresentation; and provided further that in no event shall a Holder's liability pursuant to this Section 1.9(d), when combined with the amounts paid or payable by such Holder pursuant to Section 1.9(b), exceed the proceeds from the offering received by such Holder (net of any expenses paid by such Holder). The relative fault of the indemnifying party and the indemnified party shall be determined by reference to, among other things, whether the untrue or alleged untrue statement of a material fact or the omission or alleged omission to state a material fact relates to information supplied by the indemnifying party or by the indemnified party and the parties' relative intent, knowledge, access to information and opportunity to correct or prevent such statement or omission.
- (e) Notwithstanding the foregoing, to the extent that the provisions on indemnification and contribution contained in the underwriting agreement entered into in connection with the underwritten public offering are in conflict with the foregoing provisions, the provisions in the underwriting agreement shall control.
- (f) The obligations of the Company and Holders under this Section 1.9 shall survive the completion of any offering of Registrable Securities in a registration statement under this Section 1 and otherwise shall survive the termination of this Agreement.
- 1.10 Reports Under the 1934 Act. With a view to making available to the Holders the benefits of Rule 144 and any other rule or regulation of the SEC that may at any time permit a Holder to sell securities of the Company to the public without registration or pursuant to a registration on Form S-3, the Company agrees to:
- (a) make and keep adequate and current public information available, as those terms are understood and defined in Rule 144, at all times after the effective date of the Initial Offering;

(b) file with the SEC in a timely manner all reports and other documents required of the Company under the Act and the 1934 Act;

and

- (c) furnish to any Holder, so long as the Holder owns any Registrable Securities, forthwith upon request (i) a written statement by the Company that it has complied with the reporting requirements of Rule 144 (at any time after ninety (90) days after the effective date of the first registration statement filed by the Company), the Act and the 1934 Act (at any time after it has become subject to such reporting requirements), or that it qualifies as a registrant whose securities may be resold pursuant to Form S-3 (at any time after it so qualifies), (ii) a copy of the most recent annual or quarterly report of the Company and such other reports and documents so filed by the Company, and (iii) such other information as may be reasonably requested to avail any Holder of any rule or regulation of the SEC that permits the selling of any such securities without registration or pursuant to such form.
- 1.11 <u>Assignment of Registration Rights.</u> The rights to cause the Company to register Registrable Securities pursuant to this Section 1 may be assigned (but only with all related obligations) by a Holder to a transferee or assignee of such securities that (i) is an Affiliate, subsidiary, parent, partner, limited partner, retired partner, member, equityholder, retired member or equityholder of a Holder, (ii) is a Holder's family member or trust established for the benefit of an individual Holder, or (iii) after such assignment or transfer, holds at least fifty thousand (50,000) shares of Registrable Securities (appropriately adjusted for any stock split, dividend, combination or other recapitalization), provided that: (a) the Company is, within a reasonable time after such transfer, furnished with written notice of the name and address of such transferee or assignee and the securities with respect to which such registration rights are being assigned; (b) such transferee or assignee agrees in writing to be bound by and subject to the terms and conditions of this Agreement, including, without limitation, the provisions of Section 1.13 below; and (c) such assignment shall be effective only if immediately following such transfer the further disposition of such securities by the transferee or assignee is restricted under the Act.
- 1.12 <u>Limitations on Subsequent Registration Rights.</u> From and after the date of this Agreement, the Company shall not, without the prior written consent of the Holders holding a majority of the Registrable Securities then held by all Holders, enter into any agreement with any holder or prospective holder of any securities of the Company that would allow such holder or prospective holder (a) to include any of such securities in any registration filed under Section 1.2, Section 1.3 or Section 1.4 hereof, unless under the terms of such agreement, such holder or prospective holder may include such securities in any such registration only to the extent that the inclusion of such securities will not reduce the amount of the Registrable Securities of the Holders that are included, or (b) to demand registration of their securities.

#### 1.13 "Market Stand-Off" Agreement.

(a) Each Holder hereby agrees that it will not, without the prior written consent of the managing underwriter, during the period commencing on the date of the final prospectus relating to the Company's Initial Offering and ending on the date specified by the Company and the managing underwriter (such period not to exceed one hundred eighty (180) days) (i) lend, offer, pledge, sell, contract to sell, sell any option or contract to purchase,

purchase any option or contract to sell, grant any option, right or warrant to purchase, or otherwise transfer or dispose of, directly or indirectly, any shares of Common Stock or any securities convertible into or exercisable or exchangeable for Common Stock held immediately prior to the effectiveness of the Registration Statement for such offering, or (ii) enter into any swap or other arrangement that transfers to another, in whole or in part, any of the economic consequences of ownership of the Common Stock, whether any such transaction described in clause (i) or (ii) above is to be settled by delivery of Common Stock or other securities, in cash or otherwise. The foregoing provisions of this Section 1.13 shall apply only to the Company's Initial Offering, shall not apply to the sale of any shares to an underwriter pursuant to an underwriting agreement, or to any shares purchased in the Initial Offering or in the secondary market following effectiveness of the registration statement relating to the Initial Offering, and shall only be applicable to the Holders if all officers, directors and greater than one percent (1%) stockholders of the Company enter into similar agreements. The underwriters in connection with the Company's Initial Offering are intended third-party beneficiaries of this Section 1.13 and shall have the right, power and authority to enforce the provisions hereof as though they were a party hereto. Each Holder further agrees to execute such agreements as may be reasonably requested by the underwriters in the Company's Initial Offering that are consistent with this Section 1.13 or that are necessary to give further effect thereto. Any discretionary waiver or termination of the restrictions of any or all of such agreements by the Company or the underwriters shall apply to all Holders subject to such agreements pro rata based on the number of shares subject to such agreements. In order to enforce the foregoing covenant, the Company may impose stop-transfer instructions with respect to the Registrable Securities of each Holder (and the shares or securities of every other Person subject to the foregoing restriction) until the end of such period. Notwithstanding the foregoing, if the Company is not an "Emerging Growth Company" as defined under the Jumpstart Our Business Startups (JOBS) Act of 2012 at the time of its Initial Offering, and (i) during the last seventeen (17) days of the one hundred eighty (180)-day restricted period, the Company issues an earnings release or material news or a material event relating to the Company occurs; or (ii) prior to the expiration of the one hundred eighty (180)-day restricted period, the Company announces that it will release earnings results during the sixteen (16)-day period beginning on the last day of the one hundred eighty (180)-day period, the restrictions imposed by this Section 1.13 shall continue to apply until the expiration of the eighteen (18)-day period beginning on the issuance of the earnings release or the occurrence of the material news or material event.

(b) Each Holder agrees that a legend reading substantially as follows shall be placed on all certificates representing all Registrable Securities of each Holder (and the shares or securities of every other Person subject to the restriction contained in this Section 1.13):

THE SECURITIES REPRESENTED BY THIS CERTIFICATE ARE SUBJECT TO A LOCK-UP PERIOD AFTER THE EFFECTIVE DATE OF THE ISSUER'S REGISTRATION STATEMENT FILED UNDER THE ACT, AS AMENDED, AS SET FORTH IN AN AGREEMENT BETWEEN THE COMPANY AND THE ORIGINAL HOLDER OF THESE SECURITIES, A COPY OF WHICH MAY BE OBTAINED AT THE ISSUER'S PRINCIPAL OFFICE. SUCH LOCK-UP PERIOD IS BINDING ON TRANSFEREES OF THESE SHARES.

1.14 <u>Termination of Registration Rights</u>. No Holder shall be entitled to exercise any right provided for in this Section 1 (including a Shelf Take-Down or an Underwritten Takedown) (a) after three (3) years following the consummation of the Initial Offering, (b) as to any Holder, such earlier time after the Initial Offering at which such Holder (i) can sell all shares held by it in compliance with Rule 144(b)(1)(i) and (ii) holds one percent (1%) or less of the Company's outstanding Common Stock and all Registrable Securities held by such Holder (together with any Affiliate of the Holder with whom such Holder must aggregate its sales under Rule 144) can be sold in any three (3) month period without registration in compliance with Rule 144 and without any limitations as to volume or manner of sale, or (c) after the consummation of a Liquidation Event, as that term is defined in the Restated Certificate. The Company shall be permitted to withdraw a Shelf Registration on the earlier to occur of the dates specified in clauses (a) and (b) in the previous sentence.

#### 2. Covenants of the Company.

#### 2.1 Delivery of Financial Statements.

- (a) The Company shall, upon request, deliver to (i) each Investor (or transferee of an Investor) that holds at least 2,000,000 shares of Registrable Securities (appropriately adjusted for any stock split, dividend, combination or other recapitalization), (ii) so long as the Wellington Investors collectively hold at least a majority of shares of Series G Preferred Stock originally acquired by the Wellington Investors, each Wellington Investor and (iii) so long as General Atlantic holds at least a majority of shares of Series H Preferred Stock acquired by General Atlantic at the Initial Closing (as such term is defined in the Series H Agreement), General Atlantic (each, a "Major Investor"):
- (i) as soon as practicable, but in any event within one hundred eighty (180) days after the end of each fiscal year of the Company, an income statement for such fiscal year, a balance sheet of the Company and statement of stockholders' equity as of the end of such year, and a statement of cash flows for such year, such year-end financial reports to be in reasonable detail, prepared in accordance with generally accepted accounting principles ("GAAP"), and audited and certified by independent public accountants of nationally recognized standing selected by the Company;
- (ii) as soon as practicable, but in any event within forty-five (45) days after the end of each of the first three (3) quarters of each fiscal year of the Company, (A) an unaudited income statement, statement of cash flows for such fiscal quarter and an unaudited balance sheet and a statement of stockholders' equity as of the end of such fiscal quarter, all prepared in accordance with GAAP (except that such financial statements may (x) be subject to normal year-end audit adjustments and (y) not contain all notes thereto that may be required in accordance with GAAP) and (B) a statement detailing the gross merchandise value calculations for the trailing twelve-month period measured as of the last day of a calendar quarter;
- (iii) within thirty (30) days of the end of each month, an unaudited income statement and statement of cash flows for such month, and an unaudited balance sheet as of the end of such month, all prepared in accordance with GAAP (except that such financial statements may (i) be subject to normal year-end audit adjustments and (ii) not contain all notes thereto that may be required in accordance with GAAP);

(iv) as soon as practicable, a capitalization table showing the fully-diluted capitalization of the Company and each holder of securities of the Company;

- (v) such other information relating to the financial condition, business or corporate affairs of the Company or any subsidiary of the Company as any Major Investor may from time to time reasonably request (including, an annual budget and operating plans, monthly reports of key performance indicators (KPIs), and such other report prepared for the Company's management and/or any Investors); provided, however, that the Company shall not be obligated under this subsection (v) or any other subsection of Section 2.1 to provide information that (A) it deems in good faith to be a trade secret or similar confidential information or (B) the disclosure of which would adversely affect the attorney-client privilege between the Company and its counsel.
- (b) If, for any period, the Company has any subsidiary whose accounts are consolidated with those of the Company, then in respect of such period the financial statements delivered pursuant to the foregoing sections shall be the consolidated and consolidating financial statements of the Company and all such consolidated subsidiaries.
- (c) Notwithstanding anything else in this Section 2.1 to the contrary, the Company may cease providing the information set forth in this Section 2.1 during the period starting with the date thirty (30) days before the Company's good-faith estimate of the date of filing of a registration statement if it reasonably concludes it must do so to comply with the SEC rules applicable to such registration statement and related offering; provided that the Company's covenants under this Section 2.1 shall be reinstated at such time as the Company is no longer actively employing its commercially reasonable efforts to cause such registration statement to become effective.

#### 2.2 Inspection.

- (a) The Company shall permit each Major Investor, at such Major Investor's expense, to visit and inspect the Company's and its subsidiaries' properties, to examine their books of account and records and to discuss the Company's and its subsidiaries' affairs, finances and accounts with their respective officers, all at such reasonable times as may be requested by the applicable Major Investor; provided, however, that the Company shall not be obligated pursuant to this Section 2.2 to provide access to any information that (A) it reasonably deems in good faith to be a trade secret or similar confidential information or (B) the disclosure of which would adversely affect the attorney-client privilege between the Company and its counsel.
- (b) Each Investor acknowledges and understands that, but for the waiver made herein, such Investor would be entitled, upon written demand under oath stating the purpose thereof, to inspect for any proper purpose, and to make copies and extracts from, the Company's stock ledger, a list of its stockholders, and its other books and records, and the books and records of subsidiaries of the Company, if any, under the circumstances and in the manner

provided in Section 220 of the General Corporation Law of Delaware (any and all such rights, and any and all such other rights of such Investor as may be provided for in Section 220, the "Inspection Rights"). In light of the foregoing, until the Initial Offering, each Investor hereby unconditionally and irrevocably waives the Inspection Rights, whether such Inspection Rights would be exercised or pursued directly or indirectly pursuant to Section 220 or otherwise, and covenants and agrees never to directly or indirectly commence, voluntarily aid in any way, prosecute, assign, transfer, or cause to be commenced any claim, action, cause of action, or other proceeding to pursue or exercise the Inspection Rights. The foregoing waiver applies to the Inspection Rights of each Investor in such Investor's capacity as a stockholder and shall not affect any rights of a director, in his or her capacity as such, under Section 220. The foregoing waiver shall not apply to any contractual inspection rights of an Investor under any written agreement with the Company.

- 2.3 <u>Termination of Information and Inspection Covenants.</u> The covenants set forth in Sections 2.1 and 2.2 shall terminate and be of no further force or effect upon the earlier to occur of (a) the consummation of the sale of securities pursuant to a registration statement filed by the Company under the Act in connection with the firm commitment underwritten offering of its securities to the general public, (b) when the Company first becomes subject to the periodic reporting requirements of Sections 12(g) or 15(d) of the 1934 Act, and (c) the consummation of a Liquidation Event, as that term is defined in the Restated Certificate.
- 2.4 <u>Right of First Offer</u>. Subject to the terms and conditions specified in this Section 2.4, the Company hereby grants to each Major Investor a right of first offer with respect to future sales by the Company of its Shares (as hereinafter defined). For purposes of this Section 2.4, the term "<u>Major Investor</u>" includes any general partners and Affiliates of a Major Investor. A Major Investor shall be entitled to apportion the right of first offer hereby granted it among itself and its partners and Affiliates in such proportions as it deems appropriate.

Each time the Company proposes to offer any shares of, or securities convertible into or exchangeable or exercisable for any shares of, its capital stock (<u>"Shares"</u>), the Company shall first make an offering of such Shares to each Major Investor in accordance with the following provisions:

- (a) The Company shall deliver a notice in accordance with Section 3.5 ("Notice") to the Major Investors stating (i) its bona fide intention to offer such Shares, (ii) the number of such Shares to be offered and (iii) the price and terms upon which it proposes to offer such Shares.
- (b) By written notification received by the Company within twenty (20) calendar days after the giving of Notice, each Major Investor may elect to purchase, at the price and on the terms specified in the Notice, up to that portion of such Shares that equals the proportion that the number of shares of Common Stock that are Registrable Securities issued and held by such Major Investor (assuming full conversion and exercise of all convertible and exercisable securities then outstanding) bears to the total number of shares of Common Stock of the Company then outstanding (assuming full conversion and exercise of all convertible and exercisable securities then outstanding). The Company shall promptly, in writing, inform each Major Investor that elects to purchase all the shares available to it (a "Fully-Exercising Investor")

of any other Major Investor's failure to do likewise. During the ten (10) day period commencing after such information is given to the Fully-Exercising Investors, each Fully-Exercising Investor may elect to purchase that portion of the Shares for which Major Investors were entitled to subscribe, but which were not subscribed for by the Major Investors, that is equal to the proportion that the number of shares of Registrable Securities issued and held by such Fully-Exercising Investor bears to the total number of shares of Common Stock issued and held, or issuable upon conversion of the Preferred Stock then held, by all Fully-Exercising Investors who wish to purchase some of the unsubscribed Shares.

- (c) If all Shares that Major Investors are entitled to obtain pursuant to subsection 2.4(b) are not elected to be obtained as provided in subsection 2.4(b) hereof, the Company may, during the ninety (90) day period following the expiration of the period provided in subsection 2.4(b) hereof, offer the remaining unsubscribed portion of such Shares to any Person or Persons at a price not less than that, and upon terms no more favorable to the offeree than those, specified in the Notice. If the Company does not enter into an agreement for the sale of the Shares within such period, or if such agreement is not consummated within sixty (60) days of the execution thereof, the right provided hereunder shall be deemed to be revived and such Shares shall not be offered unless first reoffered to the Major Investors in accordance herewith.
- (d) The right of first offer in this Section 2.4 shall not be applicable to (i) the issuance or sale of shares of Common Stock (or options therefor) (appropriately adjusted for any stock split, dividend, combination or other recapitalization) to employees, directors, consultants and other service providers for the primary purpose of soliciting or retaining their services pursuant to plans or agreements approved by the Board, (ii) the issuance of securities pursuant to a bona fide, firmly underwritten public offering of shares of Common Stock registered under the Act, (iii) the issuance of securities pursuant to the conversion or exercise of convertible or exercisable securities, (iv) the issuance of securities as consideration in connection with a bona fide business acquisition by the Company, whether by merger, consolidation, sale of assets, sale or exchange of stock or otherwise, which is approved by the Board, (v) the issuance and sale of Series H Preferred Stock and/or Common Stock pursuant to the Series H Agreement, (vi) the issuance of stock, warrants or other securities or rights to Persons or entities with which the Company has business relationships, provided such issuances are primarily for other than equity financing purposes, (vii) the issuance of stock, warrants or other securities or rights pursuant to any equipment leasing arrangement or debt financing arrangement, which arrangement is approved by the Board and is primarily for non-equity financing purposes or (viii) the issuance of securities that are specifically deemed not to be subject to the right of first offer in this Section 2.4 by the unanimous approval of the Board. In addition to the foregoing, the right of first offer in this Section 2.4 shall not be applicable with respect to any Major Investor in any subsequent offering of Shares if (A) at the time of such offering, the Major Investor is not an "accredited investor," as that term is then defined in Rule 501(a) of the Act and (B) such offering of Shares is otherwise being offered onl
- (e) The rights provided in this Section 2.4 may not be assigned or transferred by any Major Investor; <u>provided</u>, <u>however</u>, that a Major Investor that is an investment firm or vehicle may assign or transfer all or a portion of such rights to its general partners or Affiliates.

- (f) The covenants set forth in this Section 2.4 shall terminate and be of no further force or effect upon the consummation of the earlier of (i) the Company's sale of its Common Stock or other securities pursuant to Registration Statement under the Act (other than a registration statement relating either to the sale of securities to employees of the Company pursuant to its stock option, stock purchase or similar plan or a SEC Rule 145 transaction) and (ii) a Liquidation Event, as that term is defined in the Restated Certificate.
- 2.5 <u>Proprietary Information and Inventions Agreements</u>. The Company shall, and shall cause its subsidiaries to, require all employees, officers and consultants with access to confidential information to execute and deliver a Proprietary Information and Inventions Agreement (in the case of employees and officers), or a Consulting Agreement (in the case of consultants), in each case in substantially the form approved by the Board.
- 2.6 Successor Indemnification. If the Company or any of its successors or assignees consolidates with or merges into any other Person and is not the continuing or surviving corporation or entity of such consolidation or merger, then to the extent necessary, proper provision shall be made so that the successors and assignees of the Company assume the obligations of the Company with respect to indemnification of members of the Board as in effect immediately before such transaction, whether such obligations are contained in the Company's Bylaws, its Restated Certificate, or elsewhere, as the case may be.
- 2.7 FCPA. The Company represents that it shall not, and shall not permit any of its subsidiaries or affiliates or any of its or their respective directors, officers, managers, employees, independent contractors, representatives or agents to, promise, authorize or make any payment to, or otherwise contribute any item of value to, directly or indirectly, any Non-U.S. Official, in each case, in violation of the Foreign Corrupt Practices Act of 1977, as amended (the "FCPA"), the UK Bribery Act 2010 (the "UK Bribery Act") or any other applicable anti-bribery or anti-corruption law. The Company further represents that it shall, and shall cause each of its subsidiaries and affiliates to, cease all of its or their respective activities, as well as remediate any actions taken by the Company, its subsidiaries or affiliates, or any of their respective directors, officers, managers, employees, independent contractors, representatives or agents in violation of the FCPA, the UK Bribery Act or any other applicable anti-bribery or anti-corruption law.
- 2.8 Confidentiality. Each Investor agrees that such Investor will keep confidential and will not disclose, divulge, or use for any purpose (other than to monitor its investment in the Company) any confidential information obtained from the Company pursuant to the terms of this Agreement (including notice of the Company's intention to file a registration statement), unless such confidential information (a) is known or becomes known to the public in general (other than as a result of a breach of this Subsection 2.8 by such Investor), (b) is or has been independently developed or conceived by the Investor without use of the Company's confidential information, or (c) is or has been made known or disclosed to the Investor by a third party without a breach of any obligation of confidentiality such third party has to the Company; provided, however, that an Investor may disclose confidential information (i) to its attorneys, accountants, consultants, and other professionals to the extent necessary to obtain their services in connection with monitoring its investment in the Company; (ii) to any prospective purchaser of any Registrable Securities from such Investor, if such prospective purchaser agrees to be bound by the provisions of this Subsection 2.8; (iii) to (A) any current or prospective Affiliate,

partner, partner of a partner, member, stockholder, or wholly owned subsidiary of such Investor or (B) any prospective limited partner of an investment entity formed (or to be formed) after the date hereof that is an advisory or subadvisory client of Wellington, in each case, in the ordinary course of business, provided that such Investor informs such Person that such information is confidential and directs such Person to maintain the confidentiality of such information and that such Person is bound by binding confidentiality obligations; or (iv) as may otherwise be required by law, provided that the Investor promptly notifies the Company of such disclosure and takes reasonable steps to minimize the extent of any such required disclosure.

2.9 <u>Termination of Certain Covenants</u>. The covenants set forth in Section 2.5 and 2.7 shall terminate and be of no further force or effect upon the consummation of the earlier of (a) the Company's sale of its Common Stock or other securities pursuant to Registration Statement under the Act (other than a registration statement relating either to the sale of securities to employees of the Company pursuant to its stock option, stock purchase or similar plan or a transaction under Rule 145 of the Act) and (b) a Liquidation Event, as that term is defined in the Restated Certificate.

#### 3. Miscellaneous.

- 3.1 <u>Successors and Assigns</u>. Except as otherwise provided herein, the terms and conditions of this Agreement shall inure to the benefit of and be binding upon the respective successors and assigns of the parties (including transferees of any shares of Registrable Securities). Nothing in this Agreement, express or implied, is intended to confer upon any party other than the parties hereto or their respective successors and assigns any rights, remedies, obligations or liabilities under or by reason of this Agreement, except as expressly provided in this Agreement.
- 3.2 <u>Governing Law</u>. This Agreement shall be governed by and construed under the laws of the State of California as applied to agreements among California residents entered into and to be performed entirely within California.
- 3.3 <u>Counterparts; Facsimile</u>. This Agreement may be executed and delivered by facsimile or electronic signature and in two (2) or more counterparts, each of which shall be deemed an original, but all of which together shall constitute one (1) and the same instrument.
- 3.4 <u>Titles and Subtitles</u>. The titles and subtitles used in this Agreement are used for convenience only and are not to be considered in construing or interpreting this Agreement.
- 3.5 Notices. All notices and other communications given or made pursuant hereto shall be in writing and shall be deemed effectively given upon the earlier to occur of actual receipt or: (a) upon personal delivery to the party to be notified, (b) when sent by confirmed electronic mail or facsimile if sent during normal business hours of the recipient; if not, then on the next business day, (c) five (5) days after having been sent by registered or certified mail, return receipt requested, postage prepaid, (d) one (1) day after deposit with a nationally recognized overnight courier, specifying next day domestic delivery, with written verification of receipt or (e) three (3) days after deposit with an internationally recognized

overnight courier, specifying next day international delivery, with written verification of receipt. All notices and other communications shall be sent to the Company at One Sansome Street, 40th Floor, San Francisco, CA 94111, Attention: CEO and to the other parties at the addresses set forth in the Company's books and records (or at such other addresses as shall be specified by notice given in accordance with this Section 3.5).

3.6 Expenses. If any action at law or in equity is necessary to enforce or interpret the terms of this Agreement, the prevailing party shall be entitled to reasonable attorneys' fees, costs and necessary disbursements in addition to any other relief to which such party may be entitled.

3.7 Entire Agreement; Amendments and Waivers. This Agreement (including the Exhibits hereto, if any) constitutes the full and entire understanding and agreement among the parties with regard to the subjects hereof and thereof. Any term of this Agreement (except Section 1.1(g), Section 1.1(j), Section 1.1(n), Section 1.1(t), Section 1.1(u), Section 1.1(v), Section 1.1(w), Section 1.1(x) and Section 1.1(y), Section 2.1, Section 2.2, Section 2.3, and Section 2.4) may be amended and the observance of any term of this Agreement may be waived (either generally or in a particular instance and either retroactively or prospectively) only with the written consent of the Company and the Investors holding a majority of the Registrable Securities. The provisions of Section 1.1(n) and this sentence may be amended or waived (either generally or in a particular instance and either retroactively or prospectively) only with the written consent of the Company and the Requisite Parties. The provisions of Section 1.1(t) may be amended or waived (either generally or in a particular instance and either retroactively or prospectively) only with the written consent of the Company and the Series F Requisite Holders. The provisions of Sections 1.1(u) and 1.1(v) may be amended, terminated or waived (either generally or in a particular instance and either retroactively or prospectively) only with the written consent of the Company and the Series G Requisite Holders. The provisions of Sections 1.1(m) and 1.1(w) and this sentence may be amended or waived (either generally or in a particular instance and either retroactively or prospectively) only with the written consent of the Company and the Series H Requisite Holders. The provisions of Section 1.1(j) and this sentence may be amended, terminated or waived (either generally or in a particular instance and either retroactively or prospectively) only with the written consent of the Company, the Series F Requisite Holders, the Series G Requisite Holders and the Series H Requisite Holders. The provisions of Section 2.1, Section 2.2, Section 2.3 and Section 2.4 and this sentence may be amended, terminated or waived (either generally or in a particular instance and either retroactively or prospectively) only with the written consent of the Company and the Major Investors holding a majority of the Registrable Securities then held by all of the Major Investors; provided that, notwithstanding any waiver of the provisions of Section 2.4 with respect to a particular offering, if any of the Major Investor actually purchases Shares in such offering, then each Major Investor that did not consent to such waiver shall be permitted to participate in such offering on a pro rata basis with respect to the securities that are allocated for purchase by the Major Investor and in accordance with the other provisions (including notice and election periods) set forth in Section 2.4. The provisions of Section 1.1(c) (as it relates to a Wellington Investor), Section 1.1(x), Section 1.1(y) and the definition of "Major Investor" in Section 2.1(a) (as it relates to a Wellington Investor) may be amended or waived (either generally or in a particular instance and either retroactively or prospectively) only with the written consent of such Wellington Investor. The provisions of Section 1.1(g), the definition of "Major Investor" in

Section 2.1(a) (as it relates to General Atlantic) and this sentence may be amended or waived (either generally or in a particular instance and either retroactively or prospectively) only with the written consent of General Atlantic. Further, this Agreement may not be amended, and no provision hereof may be waived, in each case, in any way which would adversely affect the rights or obligations of a Major Investor hereunder in a manner disproportionate to any adverse effect such amendment or waiver would have on the rights of the other Major Investors hereunder, without also the written consent of such Major Investor. Any amendment or waiver effected in accordance with this paragraph shall be binding upon each holder of any Registrable Securities, each future holder of all such Registrable Securities and the Company.

- 3.8 <u>Severability</u>. Whenever possible, each provision of this Agreement shall be interpreted in such manner as to be effective and valid under applicable law, but if any provision of this Agreement shall be held to be prohibited by or invalid under applicable law, such provision shall be ineffective only to the extent of such prohibition or invalidity, without invalidating the remainder of such provision or the remaining provisions of this Agreement.
- 3.9 <u>Aggregation of Stock</u>. All shares of Registrable Securities held or acquired by a Holder's Affiliates or affiliated entities (including affiliated venture capital funds or venture capital funds under common investment management) or Persons shall be aggregated together for the purpose of determining the availability of any rights under this Agreement.
- 3.10 <u>Additional Investors</u>. Notwithstanding Section 3.7, no consent shall be necessary to add additional Investors as signatories to this Agreement, provided that such Investors have purchased Series H Preferred Stock pursuant to the subsequent closing provisions of Section 1.3 of the Series H Agreement.
- 3.11 <u>Dispute Resolution</u>. Any unresolved controversy or claim arising out of or relating to this Agreement, except as (a) otherwise provided in this Agreement, or (b) any such controversies or claims arising out of either party's intellectual property rights for which a provisional remedy or equitable relief is sought, shall be submitted to arbitration by one arbitrator mutually agreed upon by the parties, and if no agreement can be reached within thirty (30) days after names of potential arbitrators have been proposed by the American Arbitration Association (the "<u>AAA</u>"), then by one arbitrator having reasonable experience in corporate finance transactions of the type provided for in this Agreement and who is chosen by the AAA. The arbitration shall take place in San Francisco, California, in accordance with the AAA rules then in effect, and judgment upon any award rendered in such arbitration will be binding and may be entered in any court having jurisdiction thereof. There shall be limited discovery prior to the arbitration hearing as follows: (a) exchange of witness lists and copies of documentary evidence and documents relating to or arising out of the issues to be arbitrated, (b) depositions of all party witnesses and (c) such other depositions as may be allowed by the arbitrators upon a showing of good cause. Depositions shall be conducted in accordance with the California Code of Civil Procedure, the arbitrator shall be required to provide in writing to the parties the basis for the award or order of such arbitrator, and a court reporter shall record all hearings, with such record constituting the official transcript of such proceedings.

- 3.12 <u>Acknowledgment</u>. The Company acknowledges that the Investors are in the business of venture capital and/or private equity and other investing and therefore have general knowledge with respect to and review the business plans and related proprietary information of many enterprises, including enterprises which may have products or services which compete directly or indirectly with those of the Company. Nothing in this Agreement nor the receipt of confidential information shall preclude or in any way restrict the Investors from use of such general knowledge or investing or participating in any particular enterprise whether or not such enterprise has products or services which compete with those of the Company.
- 3.13 <u>Delays or Omissions</u>. No delay or omission to exercise any right, power or remedy accruing to any party under this Agreement, upon any breach or default of any other party under this Agreement, shall impair any such right, power or remedy of such non-breaching or non-defaulting party nor shall it be construed to be a waiver of any such breach or default, or an acquiescence thereto, or of any similar breach or default thereafter occurring; nor shall any waiver of any single breach or default be deemed a waiver of any other breach or default previously or thereafter occurring. Any waiver, permit, consent or approval of any kind or character on the part of any party of any breach or default under this Agreement, or any waiver on the part of any party of any provision or condition of this Agreement, must be in writing and shall be effective only to the extent specifically set forth in such writing.
- 3.14 <u>Specific Enforcement</u>. It is agreed and understood that monetary damages would not adequately compensate an injured party for the breach of this Agreement by any other party hereto, that this Agreement shall be specifically enforceable, and that any breach or threatened breach of this Agreement shall be the proper subject of a temporary or permanent injunction or restraining order. Further, each party hereto waives any claim or defense that there is an adequate remedy at law for such breach or threatened breach.
- 3.15 Effect on Prior Agreement. Upon the effectiveness of this Agreement, the Prior Agreement shall be superseded and replaced in its entirety by this Agreement and shall be of no further force or effect.

[Signature Page to Follow]

#### **COMPANY**

By: /s/ Piotr Szulczewski
Piotr Szulczewski, CEO

Address: 1 Sansome St. 40th Floor

San Francisco, CA 94104

written.

#### **INVESTOR:**

## GGV CAPITAL IV L.P.

By: GGV Capital IV L.L.C., its General Partner

By: /s/ Hans Tung

Hans Tung, Managing Director

## GGV CAPITAL IV ENTREPRENEURS FUND L.P.

By: GGV Capital IV L.L.C., its General Partner

By: /s/ Hans Tung

Hans Tung, Managing Director

## INVESTOR:

## GGV CAPITAL SELECT L.P.

By: GGV Capital Select L.L.C., its General Partner

By: GGV Management, L.L.C., its Manager

By: /s/ Hans Tung

Hans Tung, Manager

**INVESTORS:** 

THE FOUNDERS FUND V, LP

By: /s/ Brian Singerman Name: Brian Singerman Title: Managing Member

**INVESTORS:** 

THE FOUNDERS FUND V PRINCIPALS FUND, LP

By: /s/ Brian Singerman Name: Brian Singerman Title: Managing Member

**INVESTORS:** 

THE FOUNDERS FUND V ENTREPRENEURS FUND, LP

By: /s/ Brian Singerman Name: Brian Singerman Title: Managing Member

#### **INVESTORS:**

# FORMATION8 PARTNERS FUND I, L.P.

By Formation 8 GP, LLC Its General Partner

By: /s/ Joe Lonsdale
Name: Joe Lonsdale
Title: Managing Member

#### **INVESTORS:**

## CL SPV, L.P.

By Eight Partners VC GP I, LLC

Its General Partner

By: /s/ Joe Lonsdale
Name: Joe Lonsdale
Title: Managing Member

written.

#### **INVESTOR:**

# HADLEY HARBOR MASTER INVESTORS (CAYMAN) II L.P.

By: Wellington Management Company LLP, as investment

By: /s/ Emily D. Babalas

Name: Emily D. Babalas

Title: Managing Director and Counsel

Address for notice, which shall be set forth in the Company's books and records:

Wellington Management Company LLP Legal and Compliance [Address]

With a copy (which shall not constitute notice) to:

WilmerHale LLP [Address]

written.

#### INVESTOR:

# RAHUL RAVINDRA RAJ MEHTA AND PARUL MEHTA JTWROS

By: /s/ Rahul Mehta

Name: Rahul Mehta Title Managing Member

By: /s/ Parul Jtwros

Name: Parul Jtwros Title Managing Member

Address: [Address]

## **INVESTORS:**

## REDBYTE LIMITED

By: /s/ Despoina Zinonos
Name: Despoina Zinonos

Title: Director

**INVESTORS:** 

APOLETTO LIMITED

By: /s/ Elena Azarenko
Name: Elena Azarenko

Title: Director

[Address]

#### **INVESTORS:**

DST GLOBAL V, L.P.

By: DST Managers Limited Its: General Partner

By: /s/ Despoina Zinonos

Name: Despoina Zinonos

Title: Director

[Address]

**INVESTORS:** 

DST GLOBAL IV, L.P.

By: DST Managers Limited Its: General Partner

By: /s/ Despoina Zinonos

Name: Despoina Zinonos

Title: Director

[Address]

written.

#### **INVESTORS:**

## DST GLOBAL IV CO-INVEST, L.P.

By: DST Managers Limited Its: General Partner

By: /s/ Despoina Zinonos

Name: Despoina Zinonos

Title: Director

[Address]

written.

**INVESTORS:** 

DST TEAM FUND LIMITED

By: /s/ Despoina Zinonos

Name: Despoina Zinonos Title: Director

[Address]

**INVESTORS:** 

DST INVESTMENTS XI, L.P.

By: DST Managers Limited Its: General Partner

By: /s/ Despoina Zinonos

Name: Despoina Zinonos

Title: Director

[Address]

#### **INVESTORS:**

## DST INVESTMENTS XV, L.P.

By: DST Managers V Limited Its: General Partner

By: /s/ Despoina Zinonos

Name: Despoina Zinonos

Title: Director

[Address]

#### **INVESTORS:**

## DST INVESTMENTS XVI, L.P.

By: DST Managers V Limited Its: General Partner

By: /s/ Despoina Zinonos

Name: Despoina Zinonos

Title: Director

[Address]

#### **INVESTORS:**

## REPUBLIC TECHNOLOGIES PTE. LTD.

By: /s/ Ang Peng Huat
Name: Ang Peng Huat
Title: Authorized Signatory

#### **INVESTOR:**

#### GENERAL ATLANTIC (WI), L.P.

By: General Atlantic (SPV) GP, LLC, its general partner

By: General Atlantic LLC, its sole member

By: /s/ J. Frank Brown
Name: J. Frank Brown

Title: Managing Director

Address for notice, which shall be set forth in the Company's books and records:

[Address]

THIS WARRANT AND THE SECURITIES ISSUABLE UPON THE EXERCISE HEREOF HAVE NOT BEEN REGISTERED UNDER THE SECURITIES ACT OF 1933, AS AMENDED. THEY MAY NOT BE SOLD, OFFERED FOR SALE, PLEDGED, HYPOTHECATED, OR OTHERWISE TRANSFERRED EXCEPT PURSUANT TO AN EFFECTIVE REGISTRATION STATEMENT UNDER THE SECURITIES ACT OF 1933, AS AMENDED, OR AN OPINION OF COUNSEL SATISFACTORY TO THE COMPANY THAT REGISTRATION IS NOT REQUIRED UNDER SUCH ACT OR UNLESS SOLD PURSUANT TO RULE 144 UNDER SUCH ACT.

Date of Issuance August 1, 2016

# CONTEXTLOGIC INC. WARRANT TO PURCHASE SHARES OF SERIES B PREFERRED STOCK

For the purchase price of \$3,897,129.34, this Warrant is issued to Formation8 Partners Fund I, L.P. or its assigns (the "Holder") by **CONTEXTLOGIC INC.**, a Delaware corporation (the "Company").

#### 1. Purchase of Shares.

- (a) <u>Number of Shares.</u> Subject to the terms and conditions set forth herein, the Holder is entitled, upon surrender of this Warrant at the principal office of the Company (or at such other place as the Company shall notify the Holder in writing), to purchase from the Company up to 986,640 fully paid and nonassessable shares of the Company's Series B Preferred Stock, par value \$0.0001 per share (the "Preferred Stock").
- (b) Exercise Price. The exercise price for the shares of Preferred Stock issuable pursuant to this Section 1 (the "Shares") shall be \$0.0001 per share (the "Exercise Price"). The Shares and the Exercise Price shall be subject to adjustment pursuant to Section 8 hereof.
- 2. Exercise Period. This Warrant shall only be exercisable one day prior to the earliest to occur of (a) the consummation of the Company's sale of its Common Stock or other securities in the Company's first underwritten public offering pursuant to an effective registration statement under the Securities Act of 1933, as amended (the "Act") (other than a registration statement relating either to sale of securities to employees of the Company pursuant to its stock option, stock purchase or similar plan or a SEC Rule 145 transaction) (an "Initial Public Offering") and (b) the consummation of a Liquidation Event, as such term is defined in the Company's Amended and Restated Certificate of Incorporation on file with the Secretary of State of the State of Delaware. For purposes of this Warrant, any of the transactions described in subsection (b) shall be referred to herein as a "Corporate Transaction". In the event of an Initial Public Offering or Corporate Transaction, the Company shall notify the Holder at least three (3) days prior to the consummation of such Initial Public Offering or Corporate Transaction.

Notwithstanding anything to the contrary in this Warrant, if Holder fails to purchase a Note within sixty days of the Date of Issuance pursuant to Section 2 of the Settlement and Release Agreement between the Company and Holder of even date herewith, then this Warrant shall terminate and shall not be exercisable for any Shares.

#### 3. Method of Exercise.

- (a) While this Warrant remains outstanding and exercisable in accordance with Section 2 above, the Holder may exercise, in whole or in part, the purchase rights evidenced hereby. Such exercise shall be effected by:
- (i) the surrender of the Warrant, together with a duly executed copy of the Notice of Exercise attached hereto, to the Secretary of the Company at its principal office (or at such other place as the Company shall notify the Holder in writing); and
- (ii) except in the case of a Net Exercise (as defined in Section 4), the payment to the Company of an amount equal to the aggregate Exercise Price for the number of Shares being purchased.
- (b) Each exercise of this Warrant shall be deemed to have been effected immediately prior to the close of business on the day on which this Warrant is surrendered to the Company as provided in Section 3(a) above. At such time, the person or persons in whose name or names any certificate for the Shares shall be issuable upon such exercise as provided in Section 3(c) below shall be deemed to have become the holder or holders of record of the Shares represented by such certificate.
- (c) As soon as practicable after the exercise of this Warrant in whole or in part the Company at its expense will cause to be issued in the name of, and delivered to, the Holder, or as such Holder (upon payment by such Holder of any applicable transfer taxes) may direct:
  - (i) a certificate or certificates for the number of Shares to which such Holder shall be entitled, and
- (ii) in case such exercise is in part only, a new warrant or warrants (dated the date hereof) of like tenor, calling in the aggregate on the face or faces thereof for the number of Shares equal to the number of such Shares described in this Warrant minus the number of such Shares purchased by the Holder upon all exercises made in accordance with Section 3(a) above or Section 4 below.
- (d) Notwithstanding the provisions of Section 2, if the holder has not exercised this Warrant prior to the closing of a Corporate Transaction or an Initial Public Offering, this Warrant shall automatically be deemed to be exercised in full in the manner set forth in Section 4, without any further action on behalf of the Holder immediately prior to such closing.

4. <u>Net Exercise</u>. In lieu of exercising this Warrant for cash, the Holder may elect to receive shares equal to the value of this Warrant (or the portion thereof being exercised) by surrender of this Warrant at the principal office of the Company together with notice of such election (a "Net Exercise"). A Holder who Net Exercises shall have the rights described in Sections 3(b) and 3(c) hereof, and the Company shall issue to such Holder a number of Shares computed using the following formula:

$$X = \underline{Y(A - B)}$$

Where

X = The number of Shares to be issued to the Holder.

Y = The number of Shares purchasable under this Warrant or, if only a portion of the Warrant is being exercised, the portion of the Warrant being exercised (at the date of such calculation).

A = The fair market value of one (1) Share (at the date of such calculation).

B = The Exercise Price (as adjusted to the date of such calculation).

For purposes of this Section 4, the fair market value of a Share shall mean the average of the closing prices of the Shares (or equivalent shares of Common Stock underlying the Warrant) quoted in the over-the-counter market in which the Shares (or equivalent shares of Common Stock underlying the Warrant) are traded or the closing price quoted on any exchange or electronic securities market on which the Shares (or equivalent shares of Common Stock underlying the Warrant) are listed, whichever is applicable, as published in *The Wall Street Journal* for the thirty (30) trading days prior to the date of determination of fair market value (or such shorter period of time during which such Shares were traded over-the-counter or on such exchange). In the event that this Warrant is exercised pursuant to this Section 4 in connection with the Initial Public Offering, the fair market value per Share shall be the product of (a) the per share offering price to the public of the Initial Public Offering, and (b) the number of shares of Common Stock into which each Share is convertible at the time of such exercise. If the Shares are not traded on the over-the-counter market, an exchange or an electronic securities market, the fair market value shall be the price per Share that the Company could obtain from a willing buyer for Shares sold by the Company from authorized but unissued Shares, as such prices shall be determined in good faith by the Company's Board of Directors.

- 5. <u>Representations and Warranties of the Company.</u> In connection with the transactions provided for herein, the Company hereby represents and warrants to the Holder that:
- (a) <u>Organization, Good Standing, and Qualification.</u> The Company is a corporation duly organized, validly existing, and in good standing under the laws of the State of Delaware and has all requisite corporate power and authority to carry on its business as now conducted. The Company is duly qualified to transact business and is in good standing in each jurisdiction in which the failure to so qualify would have a material adverse effect on its business or properties.

- (b) <u>Authorization.</u> Except as may be limited by applicable bankruptcy, insolvency, reorganization or similar laws relating to or affecting the enforcement of creditors' rights, all corporate action has been taken on the part of the Company, its officers, directors, and stockholders necessary for the authorization, execution and delivery of this Warrant. The Company has taken all corporate action required to make all the obligations of the Company reflected in the provisions of this Warrant the valid and enforceable obligations they purport to be. The issuance of this Warrant will not be subject to preemptive rights of any stockholders of the Company. The Company has authorized sufficient shares of Preferred Stock to allow for the exercise of this Warrant.
- 6. <u>Representations and Warranties of the Holder.</u> In connection with the transactions provided for herein, the Holder hereby represents and warrants to the Company that:
- (a) <u>Authorization.</u> Holder represents that it has full power and authority to enter into this Warrant. This Warrant constitutes the Holder's valid and legally binding obligation, enforceable in accordance with its terms, except as may be limited by (i) applicable bankruptcy, insolvency, reorganization, or similar laws relating to or affecting the enforcement of creditors' rights and (ii) laws relating to the availability of specific performance, injunctive relief or other equitable remedies.
- (b) <u>Purchase Entirely for Own Account.</u> The Holder acknowledges that this Warrant is entered into by the Holder in reliance upon such Holder's representation to the Company that the Warrant and the Shares, and the Common Stock issuable upon conversion of the Shares, (collectively, the "Securities") will be acquired for investment for the Holder's own account, not as a nominee or agent, and not with a view to the resale or distribution of any part thereof, and that the Holder has no present intention of selling, granting any participation in or otherwise distributing the same. By acknowledging this Warrant, the Holder further represents that the Holder does not have any contract, undertaking, agreement, or arrangement with any person to sell, transfer or grant participations to such person or to any third person, with respect to the Securities.
- (c) <u>Disclosure of Information</u>. The Holder acknowledges that it has received all the information it considers necessary or appropriate for deciding whether to acquire the Securities. The Holder further represents that it has had an opportunity to ask questions and receive answers from the Company regarding the terms and conditions of the offering of the Securities.
- (d) <u>Investment Experience.</u> The Holder is an investor in securities of companies in the development stage and acknowledges that it is able to fend for itself, can bear the economic risk of its investment, and has such knowledge and experience in financial or business matters that it is capable of evaluating the merits and risks of the investment in the Securities. If other than an individual, the Holder also represents it has not been organized solely for the purpose of acquiring the Securities.

- (e) <u>Accredited Investor.</u> The Holder is an "accredited investor" within the meaning of Rule 501 of Regulation D, as presently in effect, as promulgated by the Securities and Exchange Commission (the "SEC") under the Act.
- (f) <u>Restricted Securities</u>. The Holder understands that the Securities are characterized as "restricted securities" under the federal securities laws inasmuch as they are being acquired from the Company in a transaction not involving a public offering and that under such laws and applicable regulations such securities may be resold without registration under the Act, only in certain limited circumstances. In this connection, the Holder represents that it is familiar with Rule 144, as presently in effect, as promulgated by the SEC under the Act ("Rule 144"), and understands the resale limitations imposed thereby and by the Act.
- (g) <u>Further Limitations on Disposition.</u> Without in any way limiting the representations set forth above, the Holder further agrees not to make any disposition of all or any portion of the Shares unless and until the transferee has agreed in writing for the benefit of the Company to be bound by the terms of this Warrant, including, without limitation, this Section 6, Section 21, and:
- (i) there is then in effect a registration statement under the Act covering such proposed disposition and such disposition is made in accordance with such registration statement;
- (ii) the Holder shall have notified the Company of the proposed disposition and shall have furnished the Company with a detailed statement of the circumstances surrounding the proposed disposition, and if reasonably requested by the Company, the Holder shall have furnished the Company with an opinion of counsel, reasonably satisfactory to the Company, that such disposition will not require registration of such shares under the Act. It is agreed that the Company will not require opinions of counsel for transactions made pursuant to Rule 144 except in extraordinary circumstances; or
- (iii) the Holder shall not make any disposition to any of the Company's competitors as such is reasonably determined by the Company.
  - (h) <u>Legends</u>. It is understood that the Securities may bear the following legend:

"THESE SECURITIES HAVE NOT BEEN REGISTERED UNDER THE SECURITIES ACT OF 1933, AS AMENDED. THEY MAY NOT BE SOLD, OFFERED FOR SALE, PLEDGED, HYPOTHECATED, OR OTHERWISE TRANSFERRED EXCEPT PURSUANT TO AN EFFECTIVE REGISTRATION STATEMENT UNDER THE SECURITIES ACT OF 1933, AS AMENDED, OR AN OPINION OF COUNSEL SATISFACTORY TO THE COMPANY THAT REGISTRATION IS NOT REQUIRED UNDER SUCH ACT OR UNLESS SOLD PURSUANT TO RULE 144 UNDER SUCH ACT."

- 7. State Commissioners of Corporations. THE SALE OF THE SECURITIES WHICH ARE THE SUBJECT OF THIS WARRANT HAS NOT BEEN QUALIFIED WITH THE COMMISSIONER OF CORPORATIONS OF THE STATE OF CALIFORNIA AND THE ISSUANCE OF SUCH SECURITIES OR THE PAYMENT OR RECEIPT OF ANY PART OF THE CONSIDERATION FOR SUCH SECURITIES PRIOR TO SUCH QUALIFICATION IS UNLAWFUL, UNLESS THE SALE OF SECURITIES IS EXEMPT FROM QUALIFICATION BY SECTION 25100, 25102 OR 25105 OF THE CALIFORNIA CORPORATIONS CODE. THE RIGHTS OF ALL PARTIES TO THIS WARRANT ARE EXPRESSLY CONDITIONED UPON SUCH QUALIFICATION BEING OBTAINED, UNLESS THE SALE IS SO EXEMPT.
- 8. <u>Adjustment of Exercise Price and Number of Shares.</u> The number and kind of Shares purchasable upon exercise of this Warrant and the Exercise Price shall be subject to adjustment from time to time as follows:
- (a) <u>Subdivisions</u>, <u>Combinations</u> and <u>Other Issuances</u>. If the Company shall at any time after the issuance but prior to the expiration of this Warrant subdivide its Preferred Stock, by split-up or otherwise, or combine its Preferred Stock, or issue additional shares of its Preferred Stock or Common Stock as a dividend with respect to any shares of its Preferred Stock, the number of Shares issuable on the exercise of this Warrant shall forthwith be proportionately increased in the case of a subdivision or stock dividend, or proportionately decreased in the case of a combination. Appropriate adjustments shall also be made to the Exercise Price payable per share, but the aggregate Exercise Price payable for the total number of Shares purchasable under this Warrant (as adjusted) shall remain the same. Any adjustment under this Section 8(a) shall become effective at the close of business on the date the subdivision or combination becomes effective, or as of the record date of such dividend, or in the event that no record date is fixed, upon the making of such dividend.
- (b) Reclassification, Reorganization and Consolidation. In case of any reclassification, capital reorganization or change in the capital stock of the Company (other than as a result of a subdivision, combination or stock dividend provided for in Section 8(a) above), then, as a condition of such reclassification, reorganization or change, lawful provision shall be made, and duly executed documents evidencing the same from the Company or its successor shall be delivered to the Holder, so that the Holder shall have the right at any time prior to the expiration of this Warrant to purchase, at a total price equal to that payable upon the exercise of this Warrant, the kind and amount of shares of stock and other securities or property receivable in connection with such reclassification, reorganization or change by a holder of the same number and type of securities as were purchasable as Shares by the Holder immediately prior to such reclassification, reorganization or change. In any such case appropriate provisions shall be made with respect to the rights and interest of the Holder so that the provisions hereof shall thereafter be applicable with respect to any shares of stock or other securities or property deliverable upon exercise hereof, and appropriate adjustments shall be made to the Exercise Price per Share payable hereunder, provided the aggregate Exercise Price shall remain the same.

- (c) <u>Notice of Adjustment</u>. When any adjustment is required to be made in the number or kind of shares purchasable upon exercise of the Warrant, or in the Exercise Price, the Company shall promptly notify the Holder of such event and of the number of Shares or other securities or property thereafter purchasable upon exercise of this Warrant.
- (d) <u>Conversion of Preferred Stock</u>. In the event that all outstanding shares of Preferred Stock are converted to Common Stock, or any other security, in accordance with the terms of the Company's Certificate of Incorporation in connection with the Company's Initial Public Offering, Corporate Transaction or other event, this Warrant shall become exercisable for Common Stock or such other security.
- 9. <u>No Fractional Shares or Scrip.</u> No fractional shares or scrip representing fractional shares shall be issued upon the exercise of this Warrant, but in lieu of such fractional shares the Company shall make a cash payment therefor on the basis of the Exercise Price then in effect.
- 10. No Stockholder Rights. Prior to exercise of this Warrant, the Holder shall not be entitled to any rights of a stockholder with respect to the Shares, including (without limitation) the right to vote such Shares, receive dividends or other distributions thereon, exercise preemptive rights or be notified of stockholder meetings, and, except as otherwise provided in this Warrant, such Holder shall not be entitled to any stockholder notice or other communication concerning the business or affairs of the Company.
- 11. <u>Transfer of Warrant.</u> Subject to compliance with applicable federal and state securities laws and any other contractual restrictions between the Company and the Holder contained herein, this Warrant and all rights hereunder are transferable in whole or in part by the Holder to any person or entity upon written notice to the Company. Within a reasonable time after the Company's receipt of an executed Assignment Form in the form attached hereto, the transfer shall be recorded on the books of the Company upon the surrender of this Warrant, properly endorsed, to the Company at its principal offices, and the payment to the Company of all transfer taxes and other governmental charges imposed on such transfer. In the event of a partial transfer, the Company shall issue to the new holders one (1) or more appropriate new warrants.
- 12. <u>Governing Law.</u> This Warrant shall be governed by and construed under the laws of the State of California as applied to agreements among California residents, made and to be performed entirely within the State of California.
- 13. <u>Successors and Assigns.</u> The terms and provisions of this Warrant shall inure to the benefit of, and be binding upon, the Company and the holders hereof and their respective successors and assigns.
- 14. Counterparts. This Warrant may be executed in two or more counterparts, each of which shall be deemed an original, but all of which together shall constitute one and the same instrument.
- 15. <u>Titles and Subtitles.</u> The titles and subtitles used in this Warrant are used for convenience only and are not to be considered in construing or interpreting this Warrant.

16. Notices. All notices and other communications given or made pursuant hereto shall be in writing and shall be deemed effectively given: (a) upon personal delivery to the party to be notified, (b) when sent by confirmed electronic mail or facsimile if sent during normal business hours of the recipient, and if not so confirmed, then on the next business day, (c) five (5) days after having been sent by registered or certified mail, return receipt requested, postage prepaid, or (d) one (1) day after deposit with a nationally recognized overnight courier, specifying next day delivery, with written verification of receipt. All communications shall be sent to the respective parties at the following addresses (or at such other addresses as shall be specified by notice given in accordance with this Section 16):

If to the Company:

#### CONTEXTLOGIC INC.

One Sansome Street, 40th Floor San Francisco, CA 94104

If to Holder:

At the address shown on the signature page hereto.

- 17. <u>Finder's Fee.</u> Each party represents that it neither is or will be obligated for any finder's fee or commission in connection with this transaction. The Holder agrees to indemnify and to hold harmless the Company from any liability for any commission or compensation in the nature of a finder's fee (and the costs and expenses of defending against such liability or asserted liability) for which the Holder or any of its officers, partners, employees or representatives is responsible. The Company agrees to indemnify and hold harmless the Holder from any liability for any commission or compensation in the nature of a finder's fee (and the costs and expenses of defending against such liability or asserted liability) for which the Company or any of its officers, employees or representatives is responsible.
- 18. Expenses. If any action at law or in equity is necessary to enforce or interpret the terms of this Warrant, the prevailing party shall be entitled to reasonable attorneys' fees, costs and necessary disbursements in addition to any other relief to which such party may be entitled.
- 19. Entire Agreement; Amendments and Waivers. This Warrant and any other documents delivered pursuant hereto constitute the full and entire understanding and agreement between the parties with regard to the subjects hereof and thereof. Nonetheless, any term of this Warrant may be amended and the observance of any term of this Warrant may be waived (either generally or in a particular instance and either retroactively or prospectively), with the written consent of the Company and the Holder; or if this Warrant has been assigned in part, by the holders or rights to purchase a majority of the shares originally issuable pursuant to this Warrant.

- 20. <u>Severability.</u> If any provision of this Warrant is held to be unenforceable under applicable law, such provision shall be excluded from this Warrant and the balance of the Warrant shall be interpreted as if such provision were so excluded and shall be enforceable in accordance with its terms.
- 21. "Market Stand-Off" Agreement. The Shares shall be subject to the "Market Stand-Off" Agreement contained in Section 1.13 of the Amended and Restated Investors' Rights Agreement dated as of February 3, 2015 (as it may be amended).

IN WITNESS WHEREOF, the parties have executed this Warrant as of the date first written above.

#### CONTEXTLOGIC INC.

By: /s/ Peter Szulczewski

Peter Szulczewski, CEO

Address: 1 Sansome St. 40th Floor

San Francisco, CA 94104

#### ACKNOWLEDGED AND AGREED:

#### FORMATION8 PARTNERS FUND I, L.P.

By Formation 8 GP, LLC Its General Partner

By: /s/ Joe Lonsdale

Name: Joe Lonsdale

Title: Managing Member

Address: [Address]

## NOTICE OF EXERCISE

## CONTEXTLOGIC INC.

Attention: Corporate Secretary

The unders	igned hereby elects to purchase, pursuant to the provisions of the Warrant, as follows:
	shares of Series B Preferred Stock pursuant to the terms of the attached Warrant, and tenders herewith payment cash of the Exercise Price of such Shares in full, together with all applicable transfer taxes, if any.
	Net Exercise the attached Warrant with respect to Shares.
The unders	igned hereby represents and warrants that Representations and Warranties in Section 6 hereof are true and correct as of the date hereof
	HOLDER:
Date:	By:
	Address:
Name in which s	nares should be registered:

#### **ASSIGNMENT FORM**

(To assign the foregoing Warrant, execute this form and supply required information. Do not use this form to purchase shares.)

FOR VALUE RECEIVED, the foregoing Warrant and all rights evidenced thereby are hereby assigned to

Name:	
_	(Please Print)
Address:	
_	(Please Print)
Dated:	
Holder's Signature:	
Holder's Address:	

**NOTE**: The signature to this Assignment Form must correspond with the name as it appears on the face of the Warrant. Officers of corporations and those acting in a fiduciary or other representative capacity should provide proper evidence of authority to assign the foregoing Warrant.

THIS WARRANT AND THE SHARES ISSUABLE HEREUNDER HAVE NOT BEEN REGISTERED UNDER THE SECURITIES ACT OF 1933, AS AMENDED (THE "ACT"), OR THE SECURITIES LAWS OF ANY STATE AND, EXCEPT AS SET FORTH IN SECTIONS 5.3 AND 5.4 BELOW, MAY NOT BE OFFERED, SOLD, PLEDGED OR OTHERWISE TRANSFERRED UNLESS AND UNTIL REGISTERED UNDER SAID ACT AND LAWS OR IN FORM AND SUBSTANCE SATISFACTORY TO THE COMPANY, SUCH OFFER, SALE, PLEDGE OR OTHER TRANSFER IS EXEMPT FROM SUCH REGISTRATION.

#### WARRANT TO PURCHASE COMMON STOCK

Company: CONTEXTLOGIC INC. (dba Wish)

Number of Shares of Common Stock: 30,000, plus all Additional Shares which Holder is entitled to purchase pursuant to Section 1.7

Warrant Price: \$1.49 per share Issue Date: May 13, 2014

Expiration Date: May 13, 2024 See also Section 5.1(b).

Credit Facility: This Warrant to Purchase Common Stock ("Warrant") is issued in connection with that certain Loan and Security Agreement of even date herewith between Silicon Valley Bank and the Company (the "Loan Agreement").

THIS WARRANT CERTIFIES THAT, for good and valuable consideration, SILICON VALLEY BANK (together with any successor or permitted assignee or transferee of this Warrant or of any shares issued upon exercise hereof, "Holder") is entitled to purchase the number of fully paid and non-assessable shares (the "Shares") of the above-stated common stock (the "Common Stock") of the above-named company (the "Company") at the above-stated Warrant Price, all as set forth above and as adjusted pursuant to Section 2 of this Warrant, subject to the provisions and upon the terms and conditions set forth in this Warrant. Reference is made to Section 5.4 of this Warrant whereby Silicon Valley Bank shall transfer this Warrant to its parent company, SVB Financial Group.

#### SECTION 1. EXERCISE.

1.1 Method of Exercise. Holder may at any time and from time to time exercise this Warrant, in whole or in part, by delivering to the Company the original of this Warrant together with a duly executed Notice of Exercise in substantially the form attached hereto as Appendix 1 and, unless Holder is exercising this Warrant pursuant to a cashless exercise set forth in Section 1.2, a check, wire transfer of same-day funds (to an account designated by the Company), or other form of payment acceptable to the Company for the aggregate Warrant Price for the Shares being purchased.

1.2 <u>Cashless Exercise</u>. On any exercise of this Warrant, in lieu of payment of the aggregate Warrant Price in the manner as specified in Section 1.1 above, but otherwise in accordance with the requirements of Section 1.1, Holder may elect to receive Shares equal to the value of this Warrant, or portion hereof as to which this Warrant is being exercised. Thereupon, the Company shall issue to the Holder such number of fully paid and non-assessable Shares as are computed using the following formula:

X = Y(A-B)/A

where:

X = the number of Shares to be issued to the Holder;

- Y = the number of Shares with respect to which this Warrant is being exercised (inclusive of the Shares surrendered to the Company in payment of the aggregate Warrant Price);
- A = the Fair Market Value (as determined pursuant to Section 1.3 below) of one Share; and
- B = the Warrant Price.
- 1.3 <u>Fair Market Value</u>. If the Company's Common Stock is then traded or quoted on a nationally recognized securities exchange, interdealer quotation system or over-the-counter market (a "<u>Trading Market</u>"), the fair market value of a Share shall be the closing price or last sale price of a share of Common Stock reported for the Business Day immediately before the date on which Holder delivers this Warrant together with its Notice of Exercise to the Company. If the Company's Common Stock is not traded in a Trading Market, the Board of Directors of the Company shall determine the fair market value of a Share in its reasonable good faith judgment.
- 1.4 <u>Delivery of Certificate and New Warrant</u>. Within a reasonable time after Holder exercises this Warrant in the manner set forth in Section 1.1 or 1.2 above, the Company shall deliver to Holder a certificate representing the Shares issued to Holder upon such exercise and, if this Warrant has not been fully exercised and has not expired, a new warrant of like tenor representing the Shares not so acquired.
- 1.5 Replacement of Warrant. On receipt of evidence reasonably satisfactory to the Company of the loss, theft, destruction or mutilation of this Warrant and, in the case of loss, theft or destruction, on delivery of an indemnity agreement reasonably satisfactory in form, substance and amount to the Company or, in the case of mutilation, on surrender of this Warrant to the Company for cancellation, the Company shall, within a reasonable time, execute and deliver to Holder, in lieu of this Warrant, a new warrant of like tenor and amount.

## 1.6 Treatment of Warrant Upon Acquisition of Company.

- (a) <u>Acquisition</u>. For the purpose of this Warrant, "<u>Acquisition</u>" means any transaction or series of related transactions involving: (i) the sale, lease, exclusive license, or other disposition of all or substantially all of the assets of the Company (ii) any merger or consolidation of the Company into or with another person or entity (other than a merger or consolidation effected exclusively to change the Company's domicile), or any other corporate reorganization, in which the stockholders of the Company in their capacity as such immediately prior to such merger, consolidation or reorganization, own less than a majority of the Company's (or the surviving or successor entity's) outstanding voting power immediately after such merger, consolidation or reorganization; or (iii) any sale or other transfer by the stockholders of the Company of shares representing at least a majority of the Company's then-total outstanding combined voting power.
- (b) <u>Treatment of Warrant at Acquisition</u>. In the event of an Acquisition in which the consideration to be received by the Company's stockholders consists solely of cash, solely of Marketable Securities or a combination of cash and Marketable Securities (a "<u>Cash/Public Acquisition</u>"), and the fair market value of one Share as determined in accordance with Section 1.3 above would be greater than the Warrant Price in effect on such date immediately prior to such Cash/Public Acquisition, and Holder has not exercised this Warrant pursuant to Section 1.1 above as to all Shares, then this Warrant shall automatically be deemed to be Cashless Exercised pursuant to

Section 1.2 above as to all Shares effective immediately prior to and contingent upon the consummation of a Cash/Public Acquisition. In connection with such Cashless Exercise, Holder shall be deemed to have restated each of the representations and warranties in Section 4 of the Warrant as the date thereof and the Company shall promptly notify the Holder of the number of Shares (or such other securities) issued upon exercise. In the event of a Cash/Public Acquisition where the fair market value of one Share as determined in accordance with Section 1.3 above would be less than the Warrant Price in effect immediately prior to such Cash/Public Acquisition, then this Warrant will expire immediately prior to the consummation of such Cash/Public Acquisition.

- (c) Upon the closing of any Acquisition other than a Cash/Public Acquisition defined above, the acquiring, surviving or successor entity shall assume the obligations of this Warrant, and this Warrant shall thereafter be exercisable for the same securities and/or other property as would have been paid for the Shares issuable upon exercise of the unexercised portion of this Warrant as if such Shares were outstanding on and as of the closing of such Acquisition, subject to further adjustment from time to time in accordance with the provisions of this Warrant.
- (d) As used in this Warrant, "Marketable Securities" means securities meeting all of the following requirements: (i) the issuer thereof is then subject to the reporting requirements of Section 13 or Section 15(d) of the Securities Exchange Act of 1934, as amended (the "Exchange Act"), and is then current in its filing of all required reports and other information under the Act and the Exchange Act; (ii) the class and series of shares or other security of the issuer that would be received by Holder in connection with the Acquisition were Holder to exercise this Warrant on or prior to the closing thereof is then traded in Trading Market, and (iii) following the closing of such Acquisition, Holder would not be restricted from publicly re-selling all of the issuer's shares and/or other securities that would be received by Holder in such Acquisition were Holder to exercise or convert this Warrant in full on or prior to the closing of such Acquisition, except to the extent that any such restriction (x) arises solely under federal or state securities laws, rules or regulations, and (y) does not extend beyond six (6) months from the closing of such Acquisition.
- 1.7 <u>Additional Shares</u>. Upon the funding of the first Growth Capital Advance (as defined in the Loan Agreement), this Warrant shall be exercisable, in addition to the number of Shares which this Warrant can otherwise be exercised for by Holder, for 10,000 additional Shares (the "<u>Initial Advance Shares</u>"), and upon the funding of Growth Capital Advances in an aggregate original principal amount in excess of \$2,500,000, this Warrant shall be exercisable, in addition to the number of Shares which this Warrant can otherwise be exercised for by Holder and the Initial Advance Shares, for 15,000 additional Shares (such additional Shares together with the Initial Advance Shares being called the "<u>Additional Shares</u>").

## SECTION 2. ADJUSTMENTS TO THE SHARES AND WARRANT PRICE.

2.1 Stock Dividends, Splits, Etc. If the Company declares or pays a dividend or distribution on the outstanding shares of the Common Stock payable in securities or property (other than cash), then upon exercise of this Warrant, for each Share acquired, Holder shall receive, without additional cost to Holder, the total number and kind of securities and property which Holder would have received had Holder owned the Shares of record as of the date the dividend or distribution occurred. If the Company subdivides the outstanding shares of the Common Stock by reclassification or otherwise into a greater number of shares, the number of Shares purchasable hereunder shall be proportionately increased and the Warrant Price shall be proportionately decreased. If the outstanding shares of the Common Stock are combined or consolidated, by reclassification or otherwise, into a lesser number of shares, the Warrant Price shall be proportionately increased and the number of Shares shall be proportionately decreased.

- 2.2 <u>Reclassification</u>, <u>Exchange</u>, <u>Combinations or Substitution</u>. Upon any event whereby all of the outstanding shares of the Common Stock are reclassified, exchanged, combined, substituted, or replaced for, into, with or by Company securities of a different class and/or series, then from and after the consummation of such event, this Warrant will be exercisable for the number, class and series of Company securities that Holder would have received had the Shares been outstanding on and as of the consummation of such event, and subject to further adjustment thereafter from time to time in accordance with the provisions of this Warrant. The provisions of this Section 2.2 shall similarly apply to successive reclassifications, exchanges, combinations substitutions, replacements or other similar events.
  - 2.3 Intentionally Omitted.
  - 2.4 Intentionally Omitted.
- 2.5 No Fractional Share. No fractional Share shall be issuable upon exercise of this Warrant and the number of Shares to be issued shall be rounded down to the nearest whole Share. If a fractional Share interest arises upon any exercise of the Warrant, the Company shall eliminate such fractional Share interest by paying Holder in cash the amount computed by multiplying the fractional interest by (i) the fair market value (as determined in accordance with Section 1.3 above) of a full Share, less (ii) the then-effective Warrant Price.
- 2.6 Notice/Certificate as to Adjustments. Upon each adjustment of the Warrant Price, Common Stock and/or number of Shares, the Company, at the Company's expense, shall notify Holder in writing within a reasonable time setting forth the adjustments to the Warrant Price, class and/or number of Shares and facts upon which such adjustment is based. The Company shall, upon written request from Holder, furnish Holder with a certificate of its Chief Financial Officer, including computations of such adjustment and the Warrant Price, class and number of Shares in effect upon the date of such adjustment.

## SECTION 3. REPRESENTATIONS AND COVENANTS OF THE COMPANY.

- 3.1 Representations and Warranties. The Company represents and warrants to, and agrees with, the Holder as follows:
- (a) The initial Warrant Price referenced on the first page of this Warrant is not greater than the price per share at which shares of Company Common Stock or options to purchase shares of Company Common Stock were last issued prior to the Issue Date hereof.
- (b) All Shares which may be issued upon the exercise of this Warrant, shall, upon issuance, be duly authorized, validly issued, fully paid and non-assessable, and free of any liens and encumbrances except for restrictions on transfer provided for herein or under applicable federal and state securities laws. The Company covenants that it shall at all times cause to be reserved and kept available out of its authorized and unissued capital stock such number of securities as will be sufficient to permit the exercise in full of this Warrant.
  - (c) The Company's capitalization table attached hereto as Schedule 1 is true and complete, in all material respects, as of the Issue Date.

- 3.2 Notice of Certain Events. If the Company proposes at any time to:
- (a) declare any dividend or distribution upon the outstanding shares of the Company's stock, whether in cash, property, stock, or other securities and whether or not a regular cash dividend;
- (b) offer for subscription or sale pro rata to the holders of the outstanding shares any additional shares of any class or series of the Company's stock (other than pursuant to contractual pre-emptive rights);
- (c) effect any reclassification, exchange, combination, substitution, reorganization or recapitalization of the outstanding shares of the Common Stock;
  - (d) effect an Acquisition or to liquidate, dissolve or wind up; or
- (e) effect its initial, underwritten offering and sale of its securities to the public pursuant to an effective registration statement under the Act (the "IPO");

then, in connection with each such event, the Company shall give Holder:

- (1) in the case of the matters referred to in (a) and (b) above, at least seven (7) Business Days prior written notice of the earlier to occur of the effective date thereof or the date on which a record will be taken for such dividend, distribution, or subscription rights (and specifying the date on which the holders of outstanding shares of the Common Stock will be entitled thereto) or for determining rights to vote, if any,
- (2) in the case of the matters referred to in (c) and (d) above at least seven (7) Business Days prior written notice of the date when the same will take place (and specifying the date on which the holders of outstanding shares of the Class will be entitled to exchange their shares for the securities or other property deliverable upon the occurrence of such event and such reasonable information as Holder may reasonably require regarding the treatment of this Warrant in connection with such event giving rise to the notice); and
- (3) with respect to the IPO, at least seven (7) Business Days prior written notice of the date on which the Company proposes to file its registration statement in connection therewith.

Company will also provide information requested by Holder that is reasonably necessary to enable Holder to comply with Holder's accounting or reporting requirements.

## SECTION 4. REPRESENTATIONS, WARRANTIES OF THE HOLDER.

The Holder represents and warrants to the Company as follows:

4.1 <u>Purchase for Own Account</u>. This Warrant and the Shares to be acquired upon exercise of this Warrant by Holder are being acquired for investment for Holder's account, not as a nominee or agent, and not with a view to the public resale or distribution within the meaning of the Act. Holder also represents that it has not been formed for the specific purpose of acquiring this Warrant or the Shares.

- 4.2 <u>Disclosure of Information</u>. Holder is aware of the Company's business affairs and financial condition and has received or has had full access to all the information it considers necessary or appropriate to make an informed investment decision with respect to the acquisition of this Warrant and its underlying securities. Holder further has had an opportunity to ask questions and receive answers from the Company regarding the terms and conditions of the offering of this Warrant and its underlying securities and to obtain additional information (to the extent the Company possessed such information or could acquire it without unreasonable effort or expense) necessary to verify any information furnished to Holder or to which Holder has access.
- 4.3 <u>Investment Experience</u>. Holder understands that the purchase of this Warrant and its underlying securities involves substantial risk. Holder has experience as an investor in securities of companies in the development stage and acknowledges that Holder can bear the economic risk of such Holder's investment in this Warrant and its underlying securities and has such knowledge and experience in financial or business matters that Holder is capable of evaluating the merits and risks of its investment in this Warrant and its underlying securities and/or has a preexisting personal or business relationship with the Company and certain of its officers, directors or controlling persons of a nature and duration that enables Holder to be aware of the character, business acumen and financial circumstances of such persons.
  - 4.4 Accredited Investor Status. Holder is an "accredited investor" within the meaning of Regulation D promulgated under the Act.
- 4.5 <u>The Act</u>. Holder understands that this Warrant and the Shares issuable upon exercise hereof have not been registered under the Act in reliance upon a specific exemption therefrom, which exemption depends upon, among other things, the bona fide nature of the Holder's investment intent as expressed herein. Holder understands that this Warrant and the Shares issued upon any exercise hereof must be held indefinitely unless subsequently registered under the Act and qualified under applicable state securities laws, or unless exemption from such registration and qualification are otherwise available. Holder is aware of the provisions of Rule 144 promulgated under the Act.
- 4.6 <u>Market Stand-off Agreement</u>. The Holder agrees that the Shares shall be subject to the Market Standoff provisions in Section 1.13 of the Amended and Restated Investors' Rights Agreement or similar agreement.
  - 4.7 No Voting Rights. Holder, as a Holder of this Warrant, will not have any voting rights until the exercise of this Warrant.

### SECTION 5. MISCELLANEOUS.

- 5.1 Term and Automatic Conversion Upon Expiration.
- (a) <u>Term.</u> Subject to the provisions of Section 1.6 above, this Warrant is exercisable in whole or in part at any time and from time to time on or before 6:00 PM, Pacific time, on the Expiration Date and shall be void thereafter.
- (b) <u>Automatic Cashless Exercise upon Expiration</u>. In the event that, upon the Expiration Date, the fair market value of one Share (or other security issuable upon the exercise hereof) as determined in accordance with Section 1.3 above is greater than the Warrant Price in effect on such date, then this Warrant shall automatically be deemed on and as of such date to be exercised pursuant to Section 1.2 above as to all Shares (or such other securities) for which it shall not previously have been exercised, and the Company shall, within a reasonable time, deliver a certificate representing the Shares (or such other securities) issued upon such exercise to Holder.

5.2 <u>Legends</u>. The Shares (and the securities issuable, directly or indirectly, upon conversion of the Shares, if any) shall be imprinted with a legend in substantially the following form:

THE SHARES EVIDENCED BY THIS CERTIFICATE HAVE NOT BEEN REGISTERED UNDER THE SECURITIES ACT OF 1933, AS AMENDED (THE "ACT"), OR THE SECURITIES LAWS OF ANY STATE AND, EXCEPT AS SET FORTH IN THAT CERTAIN WARRANT TO PURCHASE COMMON STOCK ISSUED BY THE ISSUER TO SILICON VALLEY BANK DATED MAY 13, 2014, MAY NOT BE OFFERED, SOLD, PLEDGED OR OTHERWISE TRANSFERRED UNLESS AND UNTIL REGISTERED UNDER SAID ACT AND LAWS OR IN FORM AND SUBSTANCE SATISFACTORY TO THE ISSUER, SUCH OFFER, SALE, PLEDGE OR OTHER TRANSFER IS EXEMPT FROM SUCH REGISTRATION.

- 5.3 Compliance with Securities Laws on Transfer. This Warrant and the Shares issuable upon exercise of this Warrant (and the securities issuable, directly or indirectly, upon conversion of the Shares, if any) may not be transferred or assigned in whole or in part except in compliance with applicable federal and state securities laws by the transferor and the transferee (including, without limitation, the delivery of investment representation letters and legal opinions reasonably satisfactory to the Company, as reasonably requested by the Company). The Company shall not require Holder to provide an opinion of counsel if the transfer is to SVB Financial Group (Silicon Valley Bank's parent company) or any other affiliate of Holder, provided that any such transferee is an "accredited investor" as defined in Regulation D promulgated under the Act. Additionally, the Company shall also not require an opinion of counsel if there is no material question as to the availability of Rule 144 promulgated under the Act.
- 5.4 <u>Transfer Procedure</u>. After receipt by Silicon Valley Bank of the executed Warrant, Silicon Valley Bank will transfer all of this Warrant to its parent company, SVB Financial Group. By its acceptance of this Warrant, SVB Financial Group hereby makes to the Company each of the representations and warranties set forth in Section 4 hereof and agrees to be bound by all of the terms and conditions of this Warrant as if the original Holder hereof. Subject to the provisions of Section 5.3 and upon providing the Company with written notice, SVB Financial Group and any subsequent Holder may transfer all or part of this Warrant or the Shares issuable upon exercise of this Warrant (or the securities issuable directly or indirectly, upon conversion of the Shares, if any) to any transferee, provided, however, in connection with any such transfer, SVB Financial Group or any subsequent Holder will give the Company notice of the portion of the Warrant being transferred with the name, address and taxpayer identification number of the transferee and Holder will surrender this Warrant to the Company for reissuance to the transferee(s) (and Holder if applicable); and provided further, that any subsequent transferee other than SVB Financial Group shall agree in writing with the Company to be bound by all of the terms and conditions of this Warrant. Notwithstanding any contrary provision herein, at all times prior to the IPO, Holder may not, without the Company's prior written consent, transfer this Warrant or any portion hereof, or any Shares issued upon any exercise hereof, or any shares or other securities issued upon any conversion of any Shares issued upon any exercise hereof, to any person or entity who directly competes with the Company, except in connection with an Acquisition of the Company by such a direct competitor.

5.5 Notices. All notices and other communications hereunder from the Company to the Holder, or vice versa, shall be deemed delivered and effective (i) when given personally, (ii) on the third (3rd) Business Day after being mailed by first-class registered or certified mail, postage prepaid, (iii) upon actual receipt if given by facsimile or electronic mail and such receipt is confirmed in writing by the recipient, or (iv) on the first Business Day following delivery to a reliable overnight courier service, courier fee prepaid, in any case at such address as may have been furnished to the Company or Holder, as the case may be, in writing by the Company or such Holder from time to time in accordance with the provisions of this Section 5.5. All notices to Holder shall be addressed as follows until the Company receives notice of a change of address in connection with a transfer or otherwise:

SVB Financial Group [Address]

Notice to the Company shall be addressed as follows until Holder receives notice of a change in address:

ContextLogic Inc. Attn: Chief Executive Officer 20 California Street, 4<sup>th</sup> Floor San Francisco, CA 94111

with a copy to: Gunderson Dettmer Stough Villeneuve Franklin & Hachigian, LLP

- 5.6 <u>Waiver</u>. This Warrant and any term hereof may be changed, waived, discharged or terminated (either generally or in a particular instance and either retroactively or prospectively) only by an instrument in writing signed by the party against which enforcement of such change, waiver, discharge or termination is sought.
- 5.7 Attorney's Fees. In the event of any dispute between the parties concerning the terms and provisions of this Warrant, the party prevailing in such dispute shall be entitled to collect from the other party all costs incurred in such dispute, including reasonable attorneys' fees.
- 5.8 <u>Counterparts; Facsimile/Electronic Signatures</u>. This Warrant may be executed in counterparts, all of which together shall constitute one and the same agreement. Any signature page delivered electronically or by facsimile shall be binding to the same extent as an original signature page with regards to any agreement subject to the terms hereof or any amendment thereto.
- 5.9 Governing Law. This Warrant shall be governed by and construed in accordance with the laws of the State of California, without giving effect to its principles regarding conflicts of law.

- 5.10 <u>Headings</u>. The headings in this Warrant are for purposes of reference only and shall not limit or otherwise affect the meaning of any provision of this Warrant.
  - 5.11 <u>Business Days</u>. "Business Day" is any day that is not a Saturday, Sunday or a day on which Silicon Valley Bank is closed.

[Remainder of page left blank intentionally]
[Signature page follows]

	N WITNESS WHEREOF, the parties have caused natives effective as of the Issue Date written above.	l this	Warrant	to	Purchase	Common	Stock	to	be	executed	by	their	duly	authorized
"COMF	ANY"													
CONTE	EXTLOGIC INC.													
By: <u>/s/</u>	Piotr Szulczewski													
Name:	Piotr Szulczewski (Print)													
Title:	Chief Executive Officer													
"HOLD	ER"													
SILICO	N VALLEY BANK													
By: <u>/s/</u>	Dan Hardman													
Name:	Dan Hardman (Print)													

Title: Vice President

## APPENDIX 1

# NOTICE OF EXERCISE

1. (the " <u>Co</u> shares as	The undersigned Holder hereby exercises its right purchase shares of the Common Stock of CONTEXTLOGIC INC. <a href="mainto:mpany">mpany</a> ") in accordance with the attached Warrant To Purchase Common Stock, and tenders payment of the aggregate Warrant Price for such follows:
[ ]	check in the amount of \$ payable to order of the Company enclosed herewith
[ ]	Wire transfer of immediately available funds to the Company's account
[ ]	Cashless Exercise pursuant to Section 1.2 of the Warrant
[ ]	Other [Describe]
	Holder's Name
	(Address)  By its execution below and for the benefit of the Company, Holder hereby restates each of the representations and warranties in Section 4 of the o Purchase Common Stock as of the date hereof.
	HOLDER:
	Ву:
	Name:
	Title:
	(Date):
	Appendix 1

# CONTEXTLOGIC INC. 2010 STOCK PLAN ADOPTED ON JULY 7, 2010

AMENDED AND RESTATED ON JANUARY 29, 2015

## TABLE OF CONTENTS

SECTIO	N 1. Establishment And Purpose	Page 1
SECTIO	N 2. Administration	1
	Committees of the Board of Directors	1
(b)	Authority of the Board of Directors	1
SECTIO	N 3. Eligibility	1
(a)	General Rule	1
(b)	Ten-Percent Stockholders	1
SECTIO	N 4. Stock Subject To Plan	2
(a)	Basic Limitation	2
(b)	Additional Shares	2
SECTIO	N 5. Terms And Conditions Of Awards Or Sales Of Shares	2
(a)	Stock Purchase Agreement	2
(b)	Duration of Offers and Nontransferability of Rights	2
(c)	Purchase Price	2
SECTIO	N 6. Terms And Conditions Of Options	2
(a)	Stock Option Agreement	2
(b)	Number of Shares	3
(c)	Exercise Price	3
(d)	Exercisability	3
(e)	Basic Term	3
(f)	Termination of Service (Except by Death)	3
(g)	Leaves of Absence	4
(h)	Death of Optionee	4
(i)	No Rights as a Stockholder	4
(j)	Modification, Extension and Assumption of Options	4
SECTIO	N 7. Payment For Shares	5
(a)	General Rule	5
(b)	Services Rendered	5
(c)	Promissory Note	5
(d)	Surrender of Stock	5
(e)	Exercise/Sale	5
(f)	Other Forms of Payment	5
SECTIO	N 8. Terms and Conditions of Restricted Stock Units	5
(a)	Restricted Stock Unit Agreement	5
(b)	Payment for Restricted Stock Units	6
(c)	Vesting Conditions	6
(d)	Voting and Dividend Rights	6

i

(e) Form and Time of Settlement of Restricted Stock Units	6
(f) Modification, Extension and Assumption of Restricted Stock Units	6
(g) Forfeiture	6
(h) Death of Recipient	6
(i) Creditors' Rights	7
SECTION 9. Adjustment Of Shares	7
(a) General	7
(b) Mergers and Consolidations	7
(c) Reservation of Rights	8
SECTION 10. Miscellaneous	9
(a) Securities Law Requirements	9
(b) No Retention Rights	9
(c) Transferability of Awards	9
(d) Conditions and Restrictions on Shares	9
(e) Tax Matters	9
SECTION 11. Duration and Amendments	10
(a) Term of the Plan	10
(b) Right to Amend or Terminate the Plan	10
(c) Effect of Amendment or Termination	11
SECTION 12. Definitions	11

#### CONTEXTLOGIC INC. 2010 STOCK PLAN

#### SECTION 1. ESTABLISHMENT AND PURPOSE.

The purpose of the Plan is to offer selected persons an opportunity to acquire a proprietary interest in the success of the Company, or to increase such interest, by acquiring Shares of the Company's Stock. The Plan provides for the direct award or sale of Shares, the grant of Options to purchase Shares and the grant of Restricted Stock Units. Options granted under the Plan may include Nonstatutory Options as well as ISOs intended to qualify under Section 422 of the Code.

Capitalized terms are defined in Section 13.

#### SECTION 2. ADMINISTRATION.

- (a) Committees of the Board of Directors. The Plan may be administered by one or more Committees. Each Committee shall consist of one or more members of the Board of Directors who have been appointed by the Board of Directors. Each Committee shall have such authority and be responsible for such functions as the Board of Directors has assigned to it. If no Committee has been appointed, the entire Board of Directors shall administer the Plan. Any reference to the Board of Directors in the Plan shall be construed as a reference to the Committee (if any) to whom the Board of Directors has assigned a particular function.
- **(b) Authority of the Board of Directors.** Subject to the provisions of the Plan, the Board of Directors shall have full authority and discretion to take any actions it deems necessary or advisable for the administration of the Plan. All decisions, interpretations and other actions of the Board of Directors shall be final and binding on all Participants deriving their rights from a Participant.

## **SECTION 3. ELIGIBILITY**.

- (a) General Rule. Only Employees, Outside Directors and Consultants shall be eligible for the grant of Nonstatutory Options, Restricted Stock Units or the direct award or sale of Shares. Only Employees shall be eligible for the grant of ISOs.
- **(b) Ten-Percent Stockholders.** A person who owns more than 10% of the total combined voting power of all classes of outstanding stock of the Company, its Parent or any of its Subsidiaries shall not be eligible for the grant of an ISO unless (i) the Exercise Price is at least 110% of the Fair Market Value of a Share on the date of grant and (ii) such ISO by its terms is not exercisable after the expiration of five years from the date of grant. For purposes of this Subsection (b), in determining stock ownership, the attribution rules of Section 424(d) of the Code shall be applied.

#### SECTION 4. STOCK SUBJECT TO PLAN.

(a) Basic Limitation. Not more than []1 Shares may be issued under the Plan (subject to Subsecti	on (b) below and
Section 9(a)). Up to 12,935,769 of these Shares may be issued upon the exercise of ISOs. The number of Shares that are subject to A	wards outstanding
at any time under the Plan shall not exceed the number of Shares that then remain available for issuance under the Plan. The Compa	ny, during the term
of the Plan, shall at all times reserve and keep available sufficient Shares to satisfy the requirements of the Plan. Shares offered und	er the Plan may be
authorized but unissued Shares or treasury Shares.	

**(b)** Additional Shares. In the event that Shares previously issued under the Plan are reacquired by the Company, such Shares shall be added to the number of Shares then available for issuance under the Plan. In the event that an outstanding Award expires or is canceled, the Shares allocable to the unexercised or unvested portion of such Award, as applicable, shall be added to the number of Shares then available for issuance under the Plan.

## SECTION 5. TERMS AND CONDITIONS OF AWARDS OR SALES OF SHARES.

- (a) Stock Purchase Agreement. Each award or sale of Shares under the Plan (other than upon exercise of an Option or settlement of Restricted Stock Units) shall be evidenced by a Stock Purchase Agreement between the Purchaser and the Company. Such award or sale shall be subject to all applicable terms and conditions of the Plan and may be subject to any other terms and conditions which are not inconsistent with the Plan and which the Board of Directors deems appropriate for inclusion in a Stock Purchase Agreement. The provisions of the various Stock Purchase Agreements entered into under the Plan need not be identical.
- **(b) Duration of Offers and Nontransferability of Rights**. Any right to purchase Shares under the Plan (other than an Option) shall automatically expire if not exercised by the Purchaser within 30 days after the grant of such right was communicated to the Purchaser by the Company. Such right shall not be transferable and shall be exercisable only by the Purchaser to whom such right was granted.
- (c) Purchase Price. The Board of Directors shall determine the Purchase Price of Shares to be offered under the Plan at its sole discretion. The Purchase Price shall be payable in a form described in Section 7.

#### SECTION 6. TERMS AND CONDITIONS OF OPTIONS.

(a) Stock Option Agreement. Each grant of an Option under the Plan shall be evidenced by a Stock Option Agreement between the Optionee and the Company. The Option shall be subject to all applicable terms and conditions of the Plan and may be subject to any other terms and conditions which are not inconsistent with the Plan and which the Board of Directors deems appropriate for inclusion in a Stock Option Agreement. The provisions of the various Stock Option Agreements entered into under the Plan need not be identical.

Please refer to Exhibit A for a schedule of the initial share reserve and any subsequent increases in the reserve.

- **(b) Number of Shares.** Each Stock Option Agreement shall specify the number of Shares that are subject to the Option and shall provide for the adjustment of such number in accordance with Section 9. The Stock Option Agreement shall also specify whether the Option is an ISO or a Nonstatutory Option.
- (c) Exercise Price. Each Stock Option Agreement shall specify the Exercise Price. The Exercise Price of any Option shall not be less than 100% of the Fair Market Value of a Share on the date of grant, and in the case of an ISO a higher percentage may be required by Section 3(b). Subject to the preceding sentence, the Exercise Price shall be determined by the Board of Directors at its sole discretion. The Exercise Price shall be payable in a form described in Section 7.
- (d) Exercisability. Each Stock Option Agreement shall specify the date when all or any installment of the Option is to become exercisable. No Option shall be exercisable unless the Optionee (i) has delivered an executed copy of the Stock Option Agreement to the Company or (ii) otherwise agrees to be bound by the terms of the Stock Option Agreement. The Board of Directors shall determine the exercisability provisions of the Stock Option Agreement at its sole discretion. All of an Optionee's Options shall become exercisable in full if Section 9(b)(iv) applies.
- (e) Basic Term. The Stock Option Agreement shall specify the term of the Option. The term shall not exceed 10 years from the date of grant, and in the case of an ISO a shorter term may be required by Section 3(b). Subject to the preceding sentence, the Board of Directors at its sole discretion shall determine when an Option is to expire.
- **(f) Termination of Service (Except by Death)**. If an Optionee's Service terminates for any reason other than the Optionee's death, then the Optionee's Options shall expire on the earliest of the following occasions:
  - (i) The expiration date determined pursuant to Subsection (e) above;
  - (ii) The date three months after the termination of the Optionee's Service for any reason other than Disability, or such later date as the Board of Directors may determine; or
  - (iii) The date six months after the termination of the Optionee's Service by reason of Disability, or such later date as the Board of Directors may determine.

The Optionee may exercise all or part of the Optionee's Options at any time before the expiration of such Options under the preceding sentence, but only to the extent that such Options had become exercisable before the Optionee's Service terminated (or became exercisable as a result of the termination) and the underlying Shares had vested before the Optionee's Service terminated (or vested as a result of the termination). The balance of such Options shall lapse

when the Optionee's Service terminates. In the event that the Optionee dies after the termination of the Optionee's Service but before the expiration of the Optionee's Options, all or part of such Options may be exercised (prior to expiration) by the executors or administrators of the Optionee's estate or by any person who has acquired such Options directly from the Optionee by beneficiary designation, bequest or inheritance, but only to the extent that such Options had become exercisable before the Optionee's Service terminated (or became exercisable as a result of the termination) and the underlying Shares had vested before the Optionee's Service terminated (or vested as a result of the termination).

- **(g)** Leaves of Absence. For purposes of Subsection (f) above, Service shall be deemed to continue while the Optionee is on a bona fide leave of absence, if such leave was approved by the Company in writing and if continued crediting of Service for this purpose is expressly required by the terms of such leave or by applicable law (as determined by the Company).
- **(h) Death of Optionee**. If an Optionee dies while the Optionee is in Service, then the Optionee's Options shall expire on the earlier of the following dates:
  - (i) The expiration date determined pursuant to Subsection (e) above; or
  - (ii) The date 12 months after the Optionee's death, or such later date as the Board of Directors may determine.

All or part of the Optionee's Options may be exercised at any time before the expiration of such Options under the preceding sentence by the executors or administrators of the Optionee's estate or by any person who has acquired such Options directly from the Optionee by beneficiary designation, bequest or inheritance, but only to the extent that such Options had become exercisable before the Optionee's death (or became exercisable as a result of the death) and the underlying Shares had vested before the Optionee's death (or vested as a result of the Optionee's death). The balance of such Options shall lapse when the Optionee dies.

- (i) No Rights as a Stockholder. An Optionee, or a transferee of an Optionee, shall have no rights as a stockholder with respect to any Shares covered by the Optionee's Option until such person becomes entitled to receive such Shares by filing a notice of exercise and paying the Exercise Price pursuant to the terms of such Option.
- (j) Modification, Extension and Assumption of Options. Within the limitations of the Plan, the Board of Directors may modify, extend or assume outstanding Options or may accept the cancellation of outstanding Options (whether granted by the Company or another issuer) in return for the grant of new Options for the same or a different number of Shares and at the same or a different Exercise Price. The foregoing notwithstanding, no modification of an Option shall, without the consent of the Optionee, impair the Optionee's rights or increase the Optionee's obligations under such Option.

#### SECTION 7. PAYMENT FOR SHARES.

- (a) General Rule. The entire Purchase Price or Exercise Price of Shares issued under the Plan shall be payable in cash or cash equivalents at the time when such Shares are purchased, except as otherwise provided in this Section 7.
- **(b) Services Rendered**. At the discretion of the Board of Directors, Shares may be awarded under the Plan in consideration of services rendered to the Company, a Parent or a Subsidiary prior to the Award.
- (c) Promissory Note. At the discretion of the Board of Directors, all or a portion of the Purchase Price or Exercise Price (as the case may be) of Shares issued under the Plan may be paid with a full-recourse promissory note. The Shares shall be pledged as security for payment of the principal amount of the promissory note and interest thereon. The interest rate payable under the terms of the promissory note shall not be less than the minimum rate (if any) required to avoid the imputation of additional interest under the Code. Subject to the foregoing, the Board of Directors (at its sole discretion) shall specify the term, interest rate, amortization requirements (if any) and other provisions of such note.
- (d) Surrender of Stock. At the discretion of the Board of Directors, all or any part of the Exercise Price may be paid by surrendering, or attesting to the ownership of, Shares that are already owned by the Optionee. Such Shares shall be surrendered to the Company in good form for transfer and shall be valued at their Fair Market Value as of the date when the Option is exercised.
- (e) Exercise/Sale. To the extent that a Stock Option Agreement so provides, and if Stock is publicly traded, all or part of the Exercise Price and any withholding taxes may be paid by the delivery (on a form prescribed by the Company) of an irrevocable direction to a securities broker approved by the Company to sell Shares and to deliver all or part of the sales proceeds to the Company.
- (f) Other Forms of Payment. To the extent that a Stock Purchase Agreement or Stock Option Agreement so provides, the Purchase Price or Exercise Price of Shares issued under the Plan may be paid in any other form permitted by the Delaware General Corporation Law, as amended.

#### SECTION 8. TERMS AND CONDITIONS OF RESTRICTED STOCK UNITS.

(a) Restricted Stock Unit Agreement. Each grant of Restricted Stock Units under the Plan shall be evidenced by a Restricted Stock Unit Agreement between the recipient and the Company. Restricted Stock Units granted under the Plan shall be subject to all applicable terms and conditions of the Plan and may be subject to any other terms and conditions that are not inconsistent with the Plan and which the Board of Directors deems appropriate for inclusion in a Restricted Stock Unit Agreement. The provisions of the various Restricted Stock Unit Agreements entered into under the Plan need not be identical.

- (b) Payment for Restricted Stock Units. No cash consideration shall be required of the recipient in connection with the grant of Restricted Stock Units.
- (c) Vesting Conditions. Restricted Stock Units may or may not be subject to vesting, as determined in the discretion of the Board of Directors. Vesting may occur, in full or in installments, upon the satisfaction of the vesting conditions specified in the Restricted Stock Unit Agreement, which may include continued Service with the Company or a Parent or Subsidiary, achievement of performance goals and/or such other criteria as the Board of Directors may determine. A Restricted Stock Unit Agreement may provide for accelerated vesting upon specified events.
- (d) Voting and Dividend Rights. The recipient of Restricted Stock Units shall have no voting rights. Prior to settlement or forfeiture, any Restricted Stock Unit granted under the Plan may, at the discretion of the Board of Directors, carry with it a right to dividend equivalents. Such right entitles the recipient to be credited with an amount equal to all cash dividends paid on one Share for each Restricted Stock Unit held by the recipient at the time the cash dividend is paid. Dividend equivalents may be made in the form of cash, in the form of Shares, or in a combination of both. Prior to distribution, any dividend equivalents that are not paid shall be subject to the same conditions and restrictions as the Restricted Stock Units to which they attach.
- (e) Form and Time of Settlement of Restricted Stock Units. Settlement of vested Restricted Stock Units may be made in the form of (i) cash, (ii) Shares or (iii) any combination of cash and Shares, as determined by the Board of Directors. The actual number of Restricted Stock Units eligible for settlement may be larger or smaller than the number included in the original award, based on predetermined performance factors. Vested Restricted Stock Units shall be settled in such manner and at such time(s) as specified in the Restricted Stock Unit Agreement. Until Restricted Stock Units are settled, the number of such Restricted Stock Units shall be subject to adjustment pursuant to Section 9.
- **(f) Modification, Extension and Assumption of Restricted Stock Units.** Within the limitations of the Plan, the Board of Directors may modify, extend or assume outstanding Restricted Stock Units. The foregoing notwithstanding, no modification of a Restricted Stock Unit shall, without the consent of the Participant, impair the Participant's rights or increase the Participant's obligations under such Restricted Stock Unit.
- **(g) Forfeiture**. Unless a Restricted Stock Unit Agreement provides otherwise, upon termination of the Participant's Service or upon such other time specified in the Restricted Stock Unit Agreement, any unvested Restricted Stock Units shall be forfeited to the Company. For this purpose, Service shall be deemed to continue while the Participant is on a bona fide leave of absence, if such leave was approved by the Company in writing and if continued crediting of Service for this purpose is expressly required by the terms of such leave or by applicable law (as determined by the Company).
- (h) Death of Recipient. Any Restricted Stock Units that become distributable after the recipient's death shall be distributed to the recipient's beneficiaries. Each recipient shall designate one or more beneficiaries for this purpose by filing the prescribed form with the Company, and may thereafter change such designation by filing the prescribed form with the Company at any time. If no beneficiary was designated or if no designated beneficiary survives the Participant, then any Restricted Stock Units that become payable after the Participant's death shall be distributed to his or her estate.

(i) Creditors' Rights. A holder of Restricted Stock Units shall have no rights other than those of a general creditor of the Company. Restricted Stock Units represent an unfunded and unsecured obligation of the Company, subject to the terms and conditions of the applicable Restricted Stock Unit Agreement.

#### SECTION 9. ADJUSTMENT OF SHARES.

- (a) General. In the event of a subdivision of the outstanding Stock, a declaration of a dividend payable in Shares, a combination or consolidation of the outstanding Stock into a lesser number of Shares, a reclassification, or any other increase or decrease in the number of issued shares of Stock effected without receipt of consideration by the Company, proportionate adjustments shall automatically be made in each of (i) the number of Shares available for future grants under Section 4, (ii) the number of Shares covered by each outstanding Option or Restricted Stock Unit and (iii) the Exercise Price under each outstanding Option. In the event of a declaration of an extraordinary dividend payable in a form other than Shares in an amount that has a material effect on the Fair Market Value of the Stock, a recapitalization, a spin-off, or a similar occurrence, the Board of Directors at its sole discretion may make appropriate adjustments in one or more of (i) the number of Shares available for future grants under Section 4, (ii) the number of Shares covered by each outstanding Option or Restricted Stock Unit or (iii) the Exercise Price under each outstanding Option; provided, however, that the Board of Directors shall in any event make such adjustments as may be required by Section 25102(o) of the California Corporations Code.
- **(b) Mergers and Consolidations**. In the event that the Company is a party to a merger or consolidation, all Shares acquired under the Plan and all Awards outstanding on the effective date of the transaction shall be treated in the manner described in the agreement of merger or consolidation, which need not treat all Awards in an identical manner. The agreement of merger or consolidation shall provide for one or more of the following with respect to each Award:
  - (i) The continuation of the outstanding Award by the Company (if the Company is the surviving corporation).
  - (ii) The assumption of the outstanding Award by the surviving corporation or its parent, provided that the assumption of an Option shall be in a manner that complies with Section 424(a) of the Code (whether or not the Option is an ISO).
  - (iii) The substitution by the surviving corporation or its parent of an equivalent award for the outstanding Award (including, but not limited to, an award to acquire the same consideration paid to the holders of Shares in the transaction), provided that the substitution of an Option shall be in a manner that complies with Section 424(a) of the Code (whether or not the Option is an ISO).

- (iv) Full exercisability of the Option and full vesting of the Shares subject to the Option, followed by the cancellation of the Option. The full exercisability of the Option and full vesting of the Shares subject to the Option may be contingent on the closing of such merger or consolidation. The Optionee shall be able to exercise the Option during a period of not less than five full business days preceding the closing date of such merger or consolidation, unless (A) a shorter period is required to permit a timely closing of such merger or consolidation and (B) such shorter period still offers the Optionee a reasonable opportunity to exercise the Option. Any exercise of the Option during such period may be contingent on the closing of such merger or consolidation.
- (v) The cancellation of the outstanding Award and a payment to the Participant with respect to each Share subject to the Award (including both vested and unvested Shares) as of the merger or consolidation equal to the excess of (A) the Fair Market Value of a Share as of the closing date of such merger or consolidation over (if applicable) (B) the per-Share Exercise Price of the Award (such excess, if any, the "Spread"). Such payment shall be made in the form of cash, cash equivalents, or securities of the surviving corporation or its parent with a Fair Market Value equal to the Spread. Subject to Section 409A of the Code, such payment may be made in installments and may be deferred until the date or dates when the Award would have become vested. Such payment may be subject to vesting based on the Participant's continuing Service, provided that the vesting schedule shall not be less favorable to the Participant than the schedule under which the Award would have vested. If the Spread applicable to an Award is zero or a negative number, then the Award may be cancelled without making a payment to the Participant. For purposes of this Paragraph (v), the Fair Market Value of any security shall be determined without regard to any vesting conditions that may apply to such security. In the event that a Restricted Stock Unit is subject to Section 409A of the Code, the payment described in this Section 9(b)(v) shall be made on the settlement date specified in the applicable Restricted Stock Unit Agreement, provided that settlement may be accelerated in accordance with Treasury Regulation 1.409A-3(j)(4).

Any action taken under this Section 9(b) must either preserve an Award's status as exempt from Section 409A of the Code or comply with Section 409A of the Code.

(c) Reservation of Rights. Except as provided in this Section 9, a Participant shall have no rights by reason of (i) any subdivision or consolidation of shares of stock of any class, (ii) the payment of any dividend or (iii) any other increase or decrease in the number of shares of stock of any class. Any issuance by the Company of shares of stock of any class, or securities convertible into shares of stock of any class, shall not affect, and no adjustment by reason thereof shall be made with respect to, the number of Shares subject to an Award or the Exercise Price of Shares subject to an Option. The grant of an Award pursuant to the Plan shall not affect in any way the right or power of the Company to make adjustments, reclassifications, reorganizations or changes to its capital or business structure, to merge or consolidate or to dissolve, liquidate, sell or transfer all or any part of its business or assets.

#### SECTION 10. MISCELLANEOUS.

- (a) Securities Law Requirements. Shares shall not be issued under the Plan unless the issuance and delivery of such Shares comply with (or are exempt from) all applicable requirements of law, including (without limitation) the Securities Act of 1933, as amended, the rules and regulations promulgated thereunder, state securities laws and regulations, and the regulations of any stock exchange or other securities market on which the Company's securities may then be traded.
- **(b)** No Retention Rights. Nothing in the Plan or in any right or Award granted under the Plan shall confer upon the Participant any right to continue in Service for any period of specific duration or interfere with or otherwise restrict in any way the rights of the Company (or any Parent or Subsidiary employing or retaining the Participant) or of the Participant, which rights are hereby expressly reserved by each, to terminate his or her Service at any time and for any reason, with or without cause.
- (c) Transferability of Awards. Awards shall be transferable by a Participant only by (a) beneficiary designation, (b) a will or (c) the laws of descent and distribution, except as provided in the next sentence. If the applicable Award Agreement so provides, an Award (other than an ISO) shall also be transferable by gift or domestic relations order to a Family Member of the Optionee. An ISO may be exercised during the lifetime of the Optionee only by the Optionee or by the Optionee's guardian or legal representative.
- (d) Conditions and Restrictions on Shares. Shares issued under the Plan shall be subject to such forfeiture conditions, rights of repurchase, rights of first refusal, other transfer restrictions and such other terms and conditions as the Board of Directors may determine. Such conditions and restrictions shall be set forth in the applicable Award Agreement and shall apply in addition to any restrictions that may apply to holders of Shares generally.

## (e) Tax Matters.

- (i) As a condition to the award, grant, issuance, vesting, purchase, exercise or transfer of any Award, or Shares issued pursuant to any Award, granted under this Plan, the Participant shall make such arrangements as the Board of Directors may require or permit for the satisfaction of any federal, state, local or foreign withholding tax obligations that may arise in connection with such event.
- (ii) Unless otherwise expressly set forth in an Award Agreement, it is intended that awards granted under the Plan shall be exempt from Section 409A of the Code, and any ambiguity in the terms of an Award Agreement and the Plan shall be interpreted consistently with this intent. To the extent an award is not exempt from Section 409A of the Code (any such award, a "409A Award"), any ambiguity in the terms of such award and the Plan shall be

interpreted in a manner that to the maximum extent permissible supports the award's compliance with the requirements of that statute. Notwithstanding anything to the contrary permitted under the Plan, in no event shall a modification of an Award not already subject to Section 409A of the Code be given effect if such modification would cause the Award to become subject to Section 409A of the Code unless the parties explicitly acknowledge and consent to the modification as one having that effect. A 409A Award shall be subject to such additional rules and requirements as specified by the Board of Directors from time to time in order for it to comply with the requirements of Section 409A of the Code. In this regard, if any amount under a 409A Award is payable upon a "separation from service" to an individual who is considered a "specified employee" (as each term is defined under Section 409A of the Code), then no such payment shall be made prior to the date that is the earlier of (i) six months and one day after the Participant's separation from service or (ii) the Participant's death, but only to the extent such delay is necessary to prevent such payment from being subject to Section 409A(a)(1). In addition, if a transaction subject to Section 9(b) constitutes a payment event with respect to any 409A Award, then the transaction with respect to such award must also constitute a "change in control event" as defined in Treasury Regulation Section 1.409A-3(i)(5) to the extent required by Section 409A of the Code.

(iii) Neither the Company nor any member of the Board of Directors shall have any liability to a Participant in the event an Award held by the Participant fails to achieve its intended characterization under applicable tax law.

## SECTION 11. DURATION AND AMENDMENTS.

- (a) Term of the Plan. The Plan, as set forth herein, shall become effective on the date of its adoption by the Board of Directors, subject to the approval of the Company's stockholders. If the stockholders fail to approve the Plan within 12 months after its adoption by the Board of Directors, then any grants, exercises or sales that have already occurred under the Plan shall be rescinded and no additional grants, exercises or sales shall thereafter be made under the Plan. The Plan shall terminate automatically 10 years after the later of (i) the date when the Board of Directors adopted the Plan or (ii) the date when the Board of Directors approved the most recent increase in the number of Shares reserved under Section 4 that was also approved by the Company's stockholders. The Plan may be terminated on any earlier date pursuant to Subsection (b) below.
- **(b) Right to Amend or Terminate the Plan.** The Board of Directors may amend, suspend or terminate the Plan at any time and for any reason; provided, however, that any amendment of the Plan shall be subject to the approval of the Company's stockholders if it (i) increases the number of Shares available for issuance under the Plan (except as provided in Section 9) or (ii) materially changes the class of persons who are eligible for the grant of ISOs. Stockholder approval shall not be required for any other amendment of the Plan. If the stockholders fail to approve an increase in the number of Shares reserved under Section 4 within 12 months after its adoption by the Board of Directors, then any grants, exercises or sales that have already occurred in reliance on such increase shall be rescinded and no additional grants, exercises or sales shall thereafter be made in reliance on such increase.

(c) Effect of Amendment or Termination. No Shares shall be issued or sold under the Plan after the termination thereof, except upon exercise of an Option or settlement of a Restricted Stock Unit granted prior to such termination. The termination of the Plan, or any amendment thereof, shall not affect any Share previously issued or any Award previously granted under the Plan.

#### **SECTION 12. DEFINITIONS.**

- (a) "Award" shall mean any award granted under the Plan, including an Option, Restricted Stock Unit or other right to acquire Shares under the Plan.
  - (b) "Award Agreement" means a Stock Option Agreement, Stock Purchase Agreement or Restricted Stock Unit Agreement.
  - (c) "Board of Directors" shall mean the Board of Directors of the Company, as constituted from time to time.
  - (d) "Code" shall mean the Internal Revenue Code of 1986, as amended.
  - (e) "Committee" shall mean a committee of the Board of Directors, as described in Section 2(a).
  - (f) "Company" shall mean ContextLogic Inc., a Delaware corporation.
- (g) "Consultant" shall mean a person who performs bona fide services for the Company, a Parent or a Subsidiary as a consultant or advisor, excluding Employees and Outside Directors.
- (h) "Disability" shall mean that the Optionee is unable to engage in any substantial gainful activity by reason of any medically determinable physical or mental impairment.
  - (i) "Employee" shall mean any individual who is a common-law employee of the Company, a Parent or a Subsidiary.
- (j) "Exercise Price" shall mean the amount for which one Share may be purchased upon exercise of an Option, as specified by the Board of Directors in the applicable Stock Option Agreement.
- (k) "Fair Market Value" shall mean the fair market value of a Share, as determined by the Board of Directors in accordance with applicable law. Such determination shall be conclusive and binding on all persons.

- (l) "Family Member" shall mean (i) any child, stepchild, grandchild, parent, stepparent, grandparent, spouse, former spouse, sibling, niece, nephew, mother-in-law, father-in-law, son-in-law, daughter-in-law, brother-in-law or sister-in-law, including adoptive relationships, (ii) any person sharing the Participant's household (other than a tenant or employee), (iii) a trust in which persons described in Clause (i) or (ii) have more than 50% of the beneficial interest, (iv) a foundation in which persons described in Clause (i) or (ii) or the Participant control the management of assets and (v) any other entity in which persons described in Clause (i) or (ii) or the Participant own more than 50% of the voting interests.
  - (m) "ISO" shall mean an employee incentive stock option described in Section 422(b) of the Code.
  - (n) "Nonstatutory Option" shall mean a stock option not described in Sections 422(b) or 423(b) of the Code.
  - (o) "Option" shall mean an ISO or Nonstatutory Option granted under the Plan and entitling the holder to purchase Shares.
  - (p) "Optionee" shall mean a person who holds an Option.
  - (q) "Outside Director" shall mean a member of the Board of Directors who is not an Employee.
- (r) "Parent" shall mean any corporation (other than the Company) in an unbroken chain of corporations ending with the Company, if each of the corporations other than the Company owns stock possessing 50% or more of the total combined voting power of all classes of stock in one of the other corporations in such chain. A corporation that attains the status of a Parent on a date after the adoption of the Plan shall be considered a Parent commencing as of such date.
  - (s) "Participant" shall mean an individual who holds an Award granted under the Plan.
  - (t) "Plan" shall mean this ContextLogic Inc. 2010 Stock Plan, as amended from time to time.
- (u) "Purchase Price" shall mean the consideration for which one Share may be acquired under the Plan (other than upon exercise of an Option), as specified by the Board of Directors.
- (v) "Purchaser" shall mean a person to whom the Board of Directors has offered the right to acquire Shares under the Plan (other than upon exercise of an Option or settlement of a Restricted Stock Unit).
  - (w) "Restricted Stock Unit" shall mean a bookkeeping entry representing the equivalent of one Share, as awarded under the Plan.
- (x) "Restricted Stock Unit Agreement" shall mean the agreement between the Company and the recipient of a Restricted Stock Unit that includes the terms, conditions and restrictions pertaining to such Restricted Stock Unit.

- (y) "Service" shall mean service as an Employee, Outside Director or Consultant.
- (z) "Share" shall mean one share of Stock, as adjusted in accordance with Section 9 (if applicable).
- (aa) "Stock" shall mean the Common Stock of the Company.
- (bb) "Stock Option Agreement" shall mean the agreement between the Company and an Optionee that contains the terms, conditions and restrictions pertaining to the Optionee's Option.
- (cc) "Stock Purchase Agreement" shall mean the agreement between the Company and a Purchaser who acquires Shares under the Plan that contains the terms, conditions and restrictions pertaining to the acquisition of such Shares.
- (dd) "Subsidiary" shall mean any corporation (other than the Company) in an unbroken chain of corporations beginning with the Company, if each of the corporations other than the last corporation in the unbroken chain owns stock possessing 50% or more of the total combined voting power of all classes of stock in one of the other corporations in such chain. A corporation that attains the status of a Subsidiary on a date after the adoption of the Plan shall be considered a Subsidiary commencing as of such date.

#### CONTEXTLOGIC INC.

#### AMENDED AND RESTATED 2010 STOCK PLAN

#### NOTICE OF RESTRICTED STOCK UNIT AWARD

You ("Participant") have been granted Restricted Stock Units ("RSUs") representing shares of Common Stock of ContextLogic Inc. (the "Company") on the following terms:

Name:	«Name»
Total Number of Stock Units Granted:	«Share Number»
Date of Grant:	«DateGrant»
Vesting Commencement Date:	«VestComDate»
Expiration Date:	«Expiration Date»
Vesting:	You will receive a benefit with respect to a RSU only if it vests. In or

You will receive a benefit with respect to a RSU only if it vests. In order for an RSU to vest, two vesting requirements must be satisfied on or before the Expiration Date specified above: (i) the Time-Based Requirement and (ii) the Liquidity Event Requirement. If both the Time-Based Requirement and the Liquidity Event Requirement are satisfied on or before the Expiration Date, the vesting date ("Vesting Date") of a RSU will be the first date upon which

both of those requirements are satisfied with respect to that particular RSU.

Time-Based Requirement: The Time-Based Requirement will be satisfied in installments as to [\_\_]% of the RSUs subject

to this award when you complete [\_\_] months of continuous Service and as to [\_\_]% for each month of continuous Service thereafter, measured from the Vesting Commencement Date set

forth above, subject to Section 2 of the Restricted Stock Unit Agreement.

Liquidity Event Requirement: The Liquidity Event Requirement will be satisfied (as to any then-outstanding RSUs that have

not theretofore been terminated pursuant to Section 2 of the Restricted Stock Unit Agreement)

on the earlier to occur of (i) an IPO, or (ii) a Sale Event (each, a "Liquidity Event").

Settlement:

Settlement of RSUs refers to the issuance of Shares (or, if applicable, cash) once the award is vested. If a RSU vests as a result of satisfaction of both applicable vesting requirements as described above, the Company will deliver one Share for that RSU at the time of settlement, unless at the time of settlement the Board of Directors, in its sole discretion, determines that settlement shall, in whole or in part, be in the form of cash, based on the then Fair Market Value of a Share. Settlement shall occur on or following the Vesting Date, but not later than two and one-half (2½) months following the end of the year in which the Vesting Date applicable to a RSU occurs (the last day of such two and one-half month period is referred to as the "Short Term Deferral End Date"). Notwithstanding the above, settlement of RSUs that become vested RSUs upon (i) a Sale Event will be made in Shares, unless otherwise specified in the definitive agreement for such Sale Event, or (ii) an IPO shall occur on the earlier of (a) the 185th day following the IPO Date or (b) the Short Term Deferral End Date.

By signing below, you and the Company agree that the RSUs are granted under and governed by the terms and conditions of the Company's Amended and Restated 2010 Stock Plan (the "Plan") and the Restricted Stock Unit Agreement, both of which are attached to and made a part of this document. Capitalized terms not otherwise defined herein shall have the meaning set forth in the Plan. You hereby acknowledge that the vesting of the RSUs pursuant to this Notice of Restricted Stock Unit Award is conditioned on the satisfaction of the Time-Based Requirement and the Liquidity Event Requirement on or before the Expiration Date. You shall have no right with respect to the RSUs to the extent a Liquidity Event does not occur on or before the Expiration Date (regardless of the extent to which the Time-Based Requirement is satisfied).

You further agree to accept by email all documents relating to the Company, the Plan or these RSUs and all other documents that the Company is
required to deliver to its security holders (including, without limitation, disclosures that may be required by the Securities and Exchange Commission).
You also agree that the Company may deliver these documents by posting them on a website maintained by the Company or by a third party under
contract with the Company. If the Company posts these documents on a website, it will notify you by email. You acknowledge that you may incur costs
in connection with electronic delivery, including the cost of accessing the internet and printing fees, and that an interruption of internet access may
interfere with your ability to access the documents.

PARTICIPANT:	CONTEXTLOGIC INC.	
	By:	
Address for Mailing Stock Certificate:	Title:	
		_

#### CONTEXTLOGIC INC.

#### AMENDED AND RESTATED 2010 STOCK PLAN

#### RESTRICTED STOCK UNIT AGREEMENT

#### SECTION 1. GRANT OF RESTRICTED STOCK UNITS.

- (a) **Grant**. On the terms and conditions set forth in the Notice of Restricted Stock Unit Award and this Agreement, the Company grants to you on the Date of Grant the number of RSUs set forth in the Notice of Restricted Stock Unit Award. Each RSU represents the right to receive one share of the Company's Common Stock on the terms and conditions set forth in this Agreement.
  - (b) Consideration. No payment is required for the RSUs that have been granted to you.
- (c) Nature of RSUs; No Rights As a Stockholder. The RSUs are mere bookkeeping entries and represent only the Company's unfunded and unsecured promise to issue Shares on a future date under specified conditions. As a holder of RSUs, you have no rights other than the rights of a general creditor of the Company. The RSUs carry neither voting rights nor rights to cash dividends. You have no rights as a stockholder of the Company unless and until the RSUs vest and are settled pursuant to Section 4.
- (d) **Stock Plan and Defined Terms**. The RSUs are granted pursuant to the Plan, a copy of which you acknowledge having received. The provisions of the Plan are incorporated into this Agreement by this reference. Capitalized terms are defined in Section 10 of this Agreement.

## **SECTION 2. VESTING.**

- (a) **Generally**. The RSUs vest in accordance with the vesting schedule set forth in the Notice of Restricted Stock Unit Award. You will receive a benefit with respect to a RSU only if both the Time-Based Requirement and the Liquidity Event Requirement are satisfied on or before the Expiration Date. The RSUs will not vest (in whole or in part) if only one (or if neither) of such requirements is satisfied on or before the Expiration Date.
- (b) **Termination of Service**. If your Service terminates for any reason, all RSUs as to which the Time-Based Requirement has not been satisfied as of your termination date shall automatically terminate and be cancelled. You will not satisfy the Time-Based Requirement for any additional RSUs after your Service has terminated, regardless of the reason for such termination. Upon such termination of Service, any RSUs as to which the Time-Based Requirement has been satisfied will (if the Liquidity Event Requirement has not yet been satisfied) remain outstanding until the first to occur of: (A) the satisfaction of the Liquidity Event Requirement, (B) the Expiration Date or (C) the third anniversary of your Service termination date. In case of any dispute as to whether your Service has terminated, the Board of Directors shall have sole discretion to determine whether such termination has occurred and the effective date of such termination.

- (c) **Expiration of RSUs**. If the Liquidity Event Requirement is not satisfied on or before the Expiration Date, all RSUs (regardless of whether or not, or to the extent which, the Time-Based Requirement had been satisfied as to such RSUs) shall automatically terminate upon such date. Upon a termination of one or more RSUs pursuant to this Section 2, you will have no further right with respect to such RSUs or the Shares previously allocated thereto.
- (d) **Part-Time Employment and Leaves of Absence**. If you commence working on a part-time basis, then the Company may adjust the Time-Based Requirement set forth in the Notice of Restricted Stock Unit Award. If you go on a leave of absence, then the Company may adjust the Time-Based Requirement set forth in the Notice of Restricted Stock Unit Award in accordance with the Company's leave of absence policy or the terms of such leave. Except as provided in the preceding sentence, Service shall be deemed to continue for any purpose under this Agreement while you are on a *bona fide* leave of absence, if (i) such leave was approved by the Company in writing and (ii) continued crediting of Service for such purpose is expressly required by the terms of such leave or by applicable law (as determined by the Company). Service shall be deemed to terminate when such leave ends, unless you immediately return to active work.

#### SECTION 3. RESTRICTIONS APPLICABLE TO RSUS.

- (a) **Restrictions on Transfer**. Except as otherwise provided in this Agreement, the RSUs and the rights and privileges conferred hereby shall not be sold, assigned, transferred, pledged, hypothecated, or otherwise disposed of by you prior to the settlement of the RSUs. However, you may designate a third party who, in the event of your death, shall thereafter be entitled to receive any distribution of Shares to which you were entitled at the time of your death pursuant to this Agreement by delivering a written beneficiary designation to the Company's headquarters on the prescribed form. If you deliver no such beneficiary designation or if your designated beneficiaries do not survive you, your estate will receive payments in respect of any vested RSUs.
- (b) Forfeiture of RSUs. In connection with the RSUs, the Company may be required to provide you with certain highly confidential information about the Company, including information regarding its financial condition and business prospects. Unauthorized disclosure of such information is prohibited under your existing Proprietary Information and Inventions Agreement with the Company and under Company policy, and you may be required to sign an additional nondisclosure agreement prior to receiving this type of information. In addition, unauthorized disclosure of the Company's confidential information could result in the immediate forfeiture of the RSUs, including vested RSUs as well as termination of your Service with the Company.

#### SECTION 4. SETTLEMENT OF RSUS.

- (a) **Settlement Date**. Upon a Vesting Date with respect to a particular RSU, the Company will deliver one Share for that RSU, unless at the time of settlement the Board of Directors, in its sole discretion, determines that settlement shall, in whole or in part, be in the form of cash, based on the then-Fair Market Value of a Share. Settlement shall occur on or following the Vesting Date, but not later than Short-Term Deferral End Date (as defined in the Notice of Restricted Stock Unit Grant). *Notwithstanding the above*, for RSUs that become vested upon a Sale Event, settlement will be made in Shares, unless otherwise specified in the definitive agreement for such Sale Event, and for RSUs that become vested upon an IPO, settlement shall occur on the earlier of (i) the 185th day following the IPO Date or (ii) the Short-Term Deferral End Date.
- (b) Form of Delivery. The form of any delivery of Shares (e.g., a stock certificate or electronic entry evidencing such shares) shall be determined by the Company.
- (c) **Legality of Issuance**. No Shares shall be issued to you upon settlement of the RSUs unless and until the Company has determined that (i) you and the Company have taken any actions required to register the Shares under the Securities Act or to perfect an exemption from the registration requirements thereof; (ii) any applicable listing requirement of any stock exchange or other securities market on which stock is listed has been satisfied; and (iii) any other applicable provision of federal, State or foreign law has been satisfied. The Company shall have no liability to issue Shares in respect of the RSUs unless it is able to do so in compliance with applicable law.

## **SECTION 5. TAXES.**

(a) Withholding Taxes. Upon the Vesting Date and/or settlement date for the RSUs, the Fair Market Value of the Shares is treated as income subject to withholding by the Company and/or the Parent or Subsidiary employing you (your "Employer") for the payment of all applicable federal, State, local and foreign income and employment withholding taxes which arise in connection with the vesting or settlement of the RSUs (the "Withholding Taxes"). No consideration will be paid to you in respect of this award unless you have made arrangements satisfactory to your Employer to satisfy the Withholding Taxes. To the extent that you fail to make such arrangements with respect to certain RSUs, then you will permanently forfeit such RSUs. At the discretion of the Company, these arrangements may include (i) withholding from other compensation or amounts that are owed to you by your Employer, (ii) payment in cash, (iii) if the Stock is publicly traded, payment from the proceeds of the sale of shares through a Company-approved broker, (iv) withholding a number of Shares that otherwise would be issued to you when the RSUs are settled with a Fair Market Value equal to the minimum statutory amount required to be withheld, or (v) any other method permitted by the Company. However, if you are a Company officer subject to Section 16 of the Exchange Act, then the Withholding Taxes will be satisfied pursuant to clause (iv) of the preceding sentence, unless otherwise determined in advance by the Board of Directors. If the Withholding Taxes are satisfied pursuant to clause (iv), you will be deemed to have been issued the full number of Shares subject to the RSUs and the Fair Market Value of the withheld Shares, determined as of the date when taxes otherwise would have been

withheld in cash, will be applied to the Withholding Taxes and such amount will be remitted to appropriate tax authorities by the Company or your Employer. The Company will not withhold fractional shares pursuant to clause (iv), so if the Withholding Taxes are satisfied pursuant to clause (iv), you hereby authorize the Company or your Employer to withhold the amount of any remaining Withholding Taxes from your wages or other cash compensation.

- (b) **Section 409A**. The settlement of the RSUs is intended to be exempt from the application of Code Section 409A pursuant to the "short-term deferral exemption" in Treasury Regulation 1.409A-1(b)(4) and shall be administered and interpreted in a manner that complies with such exemption. To the extent that any provision of this Agreement is ambiguous as to its exemption from Code Section 409A, the provision shall be read in such a manner so that all payments hereunder are exempt from Code Section 409A. Notwithstanding the foregoing, if this award of RSUs is interpreted as not being exempt from Code Section 409A, it shall be interpreted to comply with the requirement of Code Section 409A so that this award is not subject to additional tax or interest under Code Section 409A. In this regard, if this award is payable upon your "separation from service" within the meaning of Code Section 409A(a)(2)(A)(i) (a "Separation") and you are a "specified employee" of the Company or any affiliate thereof within the meaning of Code Section 409A(a)(2)(B)(i) on the day of your Separation, then no such payment shall be made prior to the date that is the earlier of (i) six months and one day after your Separation, or (ii) your death, but only to the extent such delay is necessary so that this award is not subject to additional tax or interest under Code Section 409A.
- (c) Acknowledgements. You acknowledge that there will be tax consequences upon vesting and/or settlement of the RSUs and/or disposition of the Shares, if any, received hereunder, and you should consult a tax advisor regarding your tax obligations prior to such event. You acknowledge that the Company is not providing any tax, legal, or financial advice, nor is the Company making any recommendations regarding your participation in the Plan or acquisition or sale of Shares subject to this award. You are hereby advised to consult with your own personal tax, legal, and financial advisors regarding your participation in the Plan. You further acknowledge that the Company (i) makes no representations or undertakings regarding the tax treatment of the award of RSUs, including, but not limited to the grant, vesting, or settlement of the RSUs, the subsequent sale of Shares acquired pursuant to such RSUs, and the receipt of any dividends; and (ii) does not commit to and is under no obligation to structure the terms of the grant of the RSUs to reduce or eliminate your tax liability or achieve any particular tax result. You agree that the Company does not have a duty to design or administer the RSUs, the Plan or its other compensation programs in a manner that minimizes your tax liability. You shall not make any claim against the Company or its Board of Directors, officers, or employees related to tax matters arising from this award or your other compensation.

#### SECTION 6. RIGHT OF FIRST REFUSAL.

(a) **Right of First Refusal**. In the event that you propose to sell, pledge or otherwise transfer to a third party any Shares acquired under this Agreement, or any interest in such Shares, the Company shall have the Right of First Refusal with respect to all (and not less than all) of such Shares. If you desire to transfer Shares acquired under this Agreement, you

must give a written Transfer Notice to the Company describing fully the proposed transfer, including the number of Shares proposed to be transferred, the proposed transfer price, the name and address of the proposed Transferee and proof satisfactory to the Company that the proposed sale or transfer will not violate any applicable federal, State or foreign securities laws. The Transfer Notice shall be signed both by you and by the proposed Transferee and must constitute a binding commitment of both parties to the transfer of the Shares. The Company shall have the right to purchase all, and not less than all, of the Shares on the terms of the proposal described in the Transfer Notice (subject, however, to any change in such terms permitted under Section 6(b) below) by delivery of a notice of exercise of the Right of First Refusal within 30 days after the date when the Transfer Notice was received by the Company.

- (b) **Transfer of Shares**. If the Company fails to exercise its Right of First Refusal within 30 days after the date when it received the Transfer Notice, you may, not later than 90 days following receipt of the Transfer Notice by the Company, conclude a transfer of the Shares subject to the Transfer Notice on the terms and conditions described in the Transfer Notice, provided that any such sale is made in compliance with applicable federal, State and foreign securities laws and not in violation of any other contractual restrictions to which you are bound. Any proposed transfer on terms and conditions different from those described in the Transfer Notice, as well as any subsequent proposed transfer by you, shall again be subject to the Right of First Refusal and shall require compliance with the procedure described in Section 6(a) above. If the Company exercises its Right of First Refusal, the parties shall consummate the sale of the Shares on the terms set forth in the Transfer Notice within 60 days after the date when the Company received the Transfer Notice (or within such longer period as may have been specified in the Transfer Notice); provided, however, that in the event the Transfer Notice provided that payment for the Shares was to be made in a form other than cash or cash equivalents paid at the time of transfer, the Company shall have the option of paying for the Shares with cash or cash equivalents equal to the present value of the consideration described in the Transfer Notice.
- (c) Additional or Exchanged Securities and Property. In the event of a merger or consolidation of the Company with or into another entity, any other corporate reorganization, a stock split, the declaration of a stock dividend, the declaration of an extraordinary dividend payable in a form other than stock, a spin-off, an adjustment in conversion ratio, a recapitalization or a similar transaction affecting the Company's outstanding securities, any securities or other property (including cash or cash equivalents) that are by reason of such transaction exchanged for, or distributed with respect to, any Shares subject to this Section 6 shall immediately be subject to the Right of First Refusal. Appropriate adjustments to reflect the exchange or distribution of such securities or property shall be made to the number and/or class of the Shares subject to this Section 6.
- (d) **Termination of Right of First Refusal**. Any other provision of this Section 6 notwithstanding, in the event that the Stock is readily tradable on an established securities market when you desire to transfer Shares, the Company shall have no Right of First Refusal, and you shall have no obligation to comply with the procedures prescribed by Sections 6(a) and 6(b) above.

- (e) **Permitted Transfers**. This Section 6 shall not apply to (i) a transfer by beneficiary designation, will or intestate succession or (ii) a transfer to one or more members of the your Immediate Family or to a trust established by you for the benefit of you and/or one or more members of your Immediate Family, provided in either case that the Transferee agrees in writing on a form prescribed by the Company to be bound by all provisions of this Agreement. If you transfer any Shares acquired under this Agreement, either under this Section 6(e) or after the Company has failed to exercise the Right of First Refusal, then this Agreement shall apply to the Transferee to the same extent as to you.
- (f) **Termination of Rights as Stockholder**. If the Company makes available, at the time and place and in the amount and form provided in this Agreement, the consideration for the Shares to be purchased in accordance with this Section 6, then after such time the person from whom such Shares are to be purchased shall no longer have any rights as a holder of such Shares (other than the right to receive payment of such consideration in accordance with this Agreement). Such Shares shall be deemed to have been purchased in accordance with the applicable provisions hereof, whether or not the certificate(s) therefor have been delivered as required by this Agreement.
- (g) **Assignment of Right of First Refusal**. The Board of Directors may freely assign the Company's Right of First Refusal, in whole or in part. Any person who accepts an assignment of the Right of First Refusal from the Company shall assume all of the Company's rights and obligations under this Section 6.

#### SECTION 7. RESTRICTIONS APPLICABLE TO SHARES.

- (a) Securities Law Restrictions. Regardless of whether the offering and sale of Shares under the Plan have been registered under the Securities Act or have been registered or qualified under the securities laws of any State, the Company, at its discretion, may impose restrictions upon the sale, pledge or other transfer of the Shares (including the placement of appropriate legends on stock certificates or the imposition of stop-transfer instructions) if, in the judgment of the Company, such restrictions are necessary or desirable in order to achieve compliance with the Securities Act, the securities laws of any State or any other law. You (or the beneficiary or your personal representative in the event of your death or incapacity, as the case may be) shall deliver to the Company any representations or other documents or assurances as the Company may deem necessary or reasonably desirable to ensure compliance with all applicable legal and regulatory requirements.
- (b) Market Stand-Off. In connection with any underwritten public offering by the Company of its equity securities pursuant to an effective registration statement filed under the Securities Act, including the Company's initial public offering, you or a Transferee shall not directly or indirectly sell, make any short sale of, loan, hypothecate, pledge, offer, grant or sell any option or other contract for the purchase of, purchase any option or other contract for the sale of, or otherwise dispose of or transfer, or agree to engage in any of the foregoing transactions with respect to, any Shares acquired under this Agreement without the prior written consent of the Company or its managing underwriter. Such restriction (the "Market Stand-Off") shall be in effect for such period of time following the date of the final prospectus for the offering as may

be requested by the Company or such underwriter. In no event, however, shall such period exceed 180 days plus such additional period as may reasonably be requested by the Company or such underwriter to accommodate regulatory restrictions on (i) the publication or other distribution of research reports or (ii) analyst recommendations and opinions, including (without limitation) the restrictions set forth in Rule 2711(f)(4) of the National Association of Securities Dealers and Rule 472(f)(4) of the New York Stock Exchange, as amended, or any similar successor rules. The Market Stand-Off shall in any event terminate two years after the date of the Company's initial public offering. In the event of the declaration of a stock dividend, a spin-off, a stock split, an adjustment in conversion ratio, a recapitalization or a similar transaction affecting the Company's outstanding securities without receipt of consideration, any new, substituted or additional securities which are by reason of such transaction distributed with respect to any Shares subject to the Market Stand-Off, or into which such Shares thereby become convertible, shall immediately be subject to the Market Stand-Off. In order to enforce the Market Stand-Off, the Company may impose stop-transfer instructions with respect to the Shares acquired under this Agreement until the end of the applicable stand-off period. The Company's underwriters shall be beneficiaries of the agreement set forth in this Section 7(b). This Section 7(b) shall not apply to Shares registered in the public offering under the Securities Act.

- (c) **Investment Intent at Grant**. You represent and agree that the Shares to be acquired upon settlement of the RSUs will be acquired for investment, and not with a view to the sale or distribution thereof.
- (d) **Investment Intent at Settlement**. In the event that the sale of Shares under the Plan is not registered under the Securities Act but an exemption is available that requires an investment representation or other representation, you shall represent and agree at the time of issuance that the Shares being acquired upon settlement of the RSUs are being acquired for investment, and not with a view to the sale or distribution thereof, and shall make such other representations as are deemed necessary or appropriate by the Company and its counsel.
- (e) **Rights of the Company**. The Company shall not be required to (i) transfer on its books any Shares that have been sold or transferred in contravention of this Agreement or (ii) treat as the owner of Shares, or otherwise to accord voting, dividend or liquidation rights to, any Transferee to whom the Shares have been transferred in contravention of this Agreement.
  - (f) Legends. All certificates evidencing the Shares issued under this Agreement shall bear the following legend:

"THE SHARES REPRESENTED HEREBY MAY NOT BE SOLD, ASSIGNED, TRANSFERRED, ENCUMBERED OR IN ANY MANNER DISPOSED OF, EXCEPT IN COMPLIANCE WITH THE TERMS OF A WRITTEN AGREEMENT BETWEEN THE COMPANY AND THE REGISTERED HOLDER OF THE SHARES (OR THE PREDECESSOR IN INTEREST TO THE SHARES). SUCH AGREEMENT GRANTS TO THE COMPANY CERTAIN RIGHTS OF FIRST REFUSAL UPON AN ATTEMPTED TRANSFER OF THE SHARES. THE SECRETARY OF THE COMPANY WILL UPON WRITTEN REQUEST FURNISH A COPY OF SUCH AGREEMENT TO THE HOLDER HEREOF WITHOUT CHARGE."

All certificates evidencing Shares issued under this Agreement in an unregistered transaction shall bear the following legend (and such other restrictive legends as are required or deemed advisable under the provisions of any applicable law):

"THE SHARES REPRESENTED HEREBY HAVE NOT BEEN REGISTERED UNDER THE SECURITIES ACT OF 1933, AS AMENDED, AND MAY NOT BE SOLD, PLEDGED, OR OTHERWISE TRANSFERRED WITHOUT AN EFFECTIVE REGISTRATION THEREOF UNDER SUCH ACT OR AN OPINION OF COUNSEL, SATISFACTORY TO THE COMPANY AND ITS COUNSEL, THAT SUCH REGISTRATION IS NOT REQUIRED."

If required by the authorities of any State in connection with the issuance of the Shares, the legend or legends required by such State authorities shall also be endorsed on all such certificates.

- (g) **Removal of Legends**. If, in the opinion of the Company and its counsel, any legend placed on a stock certificate representing Shares issued under this Agreement is no longer required, the holder of such certificate shall be entitled to exchange such certificate for a certificate representing the same number of Shares but without such legend.
- (h) **Administration**. Any determination by the Company and its counsel in connection with any of the matters set forth in this Section 7 shall be conclusive and binding on you and all other persons.

## SECTION 8. ADJUSTMENT OF SHARES.

In the event of any transaction described in Section 9(a) of the Plan, the terms of the RSUs (including, without limitation, the number and kind of shares subject to these RSUs) shall be subject to Section 9(a) of the Plan. In the event that the Company is party to certain corporate transactions, the RSUs shall be subject to Section 9(b) of the Plan, provided that any action taken must either preserve the exemption of the RSUs from Code Section 409A or comply with Code Section 409A. Any additional RSUs and any new, substituted or additional shares, cash or other property that become subject to this award as a result of any such transaction shall be subject to the same conditions and restrictions as applicable to the RSUs to which they relate.

## SECTION 9. MISCELLANEOUS PROVISIONS.

(a) **Successors and Assigns**. Except as otherwise expressly provided to the contrary, the provisions of this Agreement shall inure to the benefit of, and be binding upon, the Company and its successors and assigns and be binding upon you and your legal representatives, heirs, legatees, distributees, assigns and transferees by operation of law, whether or not any such person has become a party to this Agreement or has agreed in writing to join herein and to be bound by the terms, conditions and restrictions hereof.

- (b) No Retention Rights. Nothing in this Agreement or in the Plan shall confer upon you the right to remain in Service in any capacity or for any period of specific duration or interfere with or otherwise restrict in any way the rights of the Company (or any Parent or Subsidiary employing or retaining you) or you, which rights are hereby expressly reserved by each, to terminate your Service at any time and for any reason, with or without cause
- (c) **Notice**. Any notice required by the terms of this Agreement shall be given in writing. It shall be deemed effective upon (i) personal delivery, (ii) deposit with the United States Postal Service, by registered or certified mail, with postage and fees prepaid or (iii) deposit with a reputable overnight courier, with shipping charges prepaid. Notice shall be addressed to the Company at its principal executive office and to you at the address that you most recently provided to the Company in accordance with this Section 9(c).
- (d) Effect on Other Employee Benefit Plans. The value of the RSUs and the Shares issuable thereunder shall not be included as compensation, earnings, salaries, or other similar terms used when calculating benefits under any employee benefit plan (other than the Plan) sponsored by the Company or a Parent or Subsidiary, except as such plans otherwise expressly provide.
- (e) **Entire Agreement**. The Notice of Restricted Stock Unit Award, this Agreement and the Plan constitute the entire understanding between you and the Company regarding the subject matter hereof. They supersede any other agreements, representations or understandings (whether oral or written and whether express or implied) that relate to the subject matter hereof.
- (f) Choice of Law. This Agreement shall be governed by, and construed in accordance with, the laws of the State of Delaware, as such laws are applied to contracts entered into and performed in such State.

#### SECTION 10. DEFINITIONS.

- (a) "Agreement" shall mean this Restricted Stock Unit Agreement.
- (b) "Board of Directors" shall mean the Board of Directors of the Company, as constituted from time to time or, if a Committee has been appointed, such Committee.
  - (c) "Code" shall mean the Internal Revenue Code of 1986, as amended.
  - (d) "Committee" shall mean a committee of the Board of Directors, as described in Section 2 of the Plan.
  - (e) "Company" shall mean ContextLogic Inc., a Delaware corporation.
- (f) "Consultant" shall mean a person who performs bona fide services for the Company, a Parent or a Subsidiary as a consultant or advisor, excluding Employees and Outside Directors.

- (g) "Date of Grant" shall mean the date specified in the Notice of Restricted Stock Unit Award.
- (h) "Employee" shall mean any individual who is a common-law employee of the Company, a Parent or a Subsidiary.
- (i) "Exchange Act" shall mean the Securities Exchange Act of 1934, as amended.
- (j) "Expiration Date" shall mean the expiration date of the RSUs as set forth in the Notice of Restricted Stock Unit Award.
- (k) "Fair Market Value" shall mean the fair market value of a Share, as determined by the Board of Directors in good faith. Such determination shall be conclusive and binding on all persons.
- (l) "Immediate Family" shall mean any child, stepchild, grandchild, parent, stepparent, grandparent, spouse, sibling, mother-in-law, father-in-law, son-in-law, daughter-in-law, brother-in-law or sister-in-law and shall include adoptive relationships.
- (m) "IPO" shall mean the first firm commitment underwritten public offering pursuant to an effective registration statement on an established national or foreign securities exchange covering the offer and sale by the Company of its equity securities, as a result of or following which the Shares shall be publicly held, and "IPO Date" shall mean the date on which the IPO occurs.
  - (n) "Liquidity Event Requirement" shall mean the requirement that the Company complete an IPO or Sale Event.
  - (o) "Outside Director" shall mean a member of the Board of Directors who is not an Employee.
- (p) "Parent" shall mean any corporation (other than the Company) in an unbroken chain of corporations ending with the Company, if each of the corporations other than the Company owns stock possessing 50% or more of the total combined voting power of all classes of stock in one of the other corporations in such chain. A corporation that attains the status of a Parent on a date after the adoption of the Plan shall be considered a Parent commencing as of such date.
  - (q) "Plan" shall mean the ContextLogic Inc. 2010 Stock Plan, as amended from time to time.
  - (r) "Right of First Refusal" shall mean the Company's right of first refusal described in Section 6.
- (s) "RSUs" shall mean the Restricted Stock Units granted to you by the Company as set forth in the Notice of Restricted Stock Unit Award.

- (t) "Sale Event" means the consummation of the following transactions in which holders of Shares receive cash or marketable securities tradable on an established national or foreign securities exchange: (i) a sale of all or substantially all of the assets of the Company determined on a consolidated basis to an unrelated person or entity; (ii) a merger, reorganization, or consolidation involving the Company in which the shares of voting stock of the Company outstanding immediately prior to such transaction represent or are converted into or exchanged for securities of the surviving or resulting entity immediately upon completion of such transaction which represent less than 50% of the outstanding voting power of such surviving or resulting entity; or (iii) the acquisition of all or a majority of the outstanding voting stock of the Company in a single transaction or series of related transactions by a person or group of persons. For the avoidance of doubt, an initial public offering, any subsequent public offering, another capital raising event, and a merger effected solely to change the Company's domicile shall not constitute a "Sale Event." In addition, a transaction shall not constitute a Sale Event unless such transaction also qualifies as an event under Treasury Regulation Section 1.409A-3(i)(5)(v) (change in the ownership of a corporation), Treasury Regulation Section 1.409A-3(i)(5)(vii) (change in the ownership of a substantial portion of a corporation's assets).
  - (u) "Securities Act" shall mean the Securities Act of 1933, as amended.
  - (v) "Service" shall mean service as an Employee, Consultant or Outside Director.
  - (w) "Share" shall mean one share of Stock, as adjusted in accordance with Section 9 of the Plan (if applicable).
  - (x) "Stock" shall mean the Common Stock of the Company.
- (y) "Subsidiary" shall mean any corporation entity (other than the Company) in an unbroken chain or corporations beginning with the Company, if each of the corporations other than the last corporation in the unbroken chain owns stock possessing 50% or more of the total combined voting power of all classes of stock in one of the other corporations in such chain. A corporation that attains the status of a Subsidiary on a date after the adoption of the Plan shall be considered a Subsidiary commencing as of such date.
- (z) "Time-Based Requirement" shall mean the requirement to provide Service over the period of time set forth in the Notice of Restricted Stock Unit Award.
  - (aa) "Transferee" shall mean any person to whom you have directly or indirectly transferred any Shares acquired under this Agreement.
  - (bb) "Transfer Notice" shall mean the notice of a proposed transfer of Shares described in Section 6.

(cc) "Vesting Date" shall mean the first date on or before the Expiration Date upon which both the Time-Based Requirement and the Liquidity Event Requirement are satisfied.

# CONTEXTLOGIC, INC. 2010 STOCK PLAN

# NOTICE OF STOCK OPTION GRANT (INSTALLMENT EXERCISE)

The Optionee has been granted the following option to purchase shares of the Common Stock of ContextLogic, Inc.:

Name of Optionee:	«Name»
Total Number of Shares:	«TotalShares»
Type of Option:	«ISO» Incentive Stock Option (ISO)
	«NSO» Nonstatutory Stock Option (NSO)
Exercise Price per Share:	\$«PricePerShare»
Date of Grant:	«DateGrant»
Date Exercisable:	The Right of Repurchase shall lapse with respect to the first «Percent»% of the Shares subject to this option when the Optionee completes «CliffPeriod» months of continuous Service beginning with the Vesting Commencement Date set forth above. The Right of Repurchase shall lapse with respect to an additional «Fraction»% of the Shares subject to this option when the Optionee completes each month of continuous Service thereafter.
Vesting Commencement Date:	«VCD»
Expiration Date:	«ExpDate». This option expires earlier if the Optionee's Service terminates earlier, as provided in Section 6 of the Stock Option Agreement.
	at this option is granted under, and governed by the terms and conditions of, the 2010 Stock tents are attached to, and made a part of, this Notice of Stock Option Grant. <b>Section 13 of the gements of the Optionee</b> .
OPTIONEE:	CONTEXTLOGIC, INC.
	Ву:
	Peter Szulczewski, Chief Executive Officer

THE OPTION GRANTED PURSUANT TO THIS AGREEMENT AND THE SHARES ISSUABLE UPON THE EXERCISE THEREOF HAVE NOT BEEN REGISTERED UNDER THE SECURITIES ACT OF 1933, AS AMENDED, AND MAY NOT BE SOLD, PLEDGED, OR OTHERWISE TRANSFERRED WITHOUT AN EFFECTIVE REGISTRATION THEREOF UNDER SUCH ACT OR AN OPINION OF COUNSEL, SATISFACTORY TO THE COMPANY AND ITS COUNSEL, THAT SUCH REGISTRATION IS NOT REQUIRED.

# CONTEXTLOGIC, INC. 2010 STOCK PLAN: STOCK OPTION AGREEMENT

#### SECTION 1. GRANT OF OPTION.

- (a) **Option**. On the terms and conditions set forth in the Notice of Stock Option Grant and this Agreement, the Company grants to the Optionee on the Date of Grant the option to purchase at the Exercise Price the number of Shares set forth in the Notice of Stock Option Grant. The Exercise Price is agreed to be at least 100% of the Fair Market Value per Share on the Date of Grant (110% of Fair Market Value if Section 3(b) of the Plan applies). This option is intended to be an ISO or an NSO, as provided in the Notice of Stock Option Grant.
- (b) \$100,000 Limitation. Even if this option is designated as an ISO in the Notice of Stock Option Grant, it shall be deemed to be an NSO to the extent (and only to the extent) required by the \$100,000 annual limitation under Section 422(d) of the Code.
- (c) Stock Plan and Defined Terms. This option is granted pursuant to the Plan, a copy of which the Optionee acknowledges having received. The provisions of the Plan are incorporated into this Agreement by this reference. Capitalized terms are defined in Section 14 of this Agreement.

#### SECTION 2. RIGHT TO EXERCISE.

- (a) **Exercisability**. Subject to Subsection (b) below and the other conditions set forth in this Agreement, all or part of this option may be exercised prior to its expiration at the time or times set forth in the Notice of Stock Option Grant. In addition, this option shall become exercisable in full if Section 8(b)(iv) of the Plan applies.
- (b) **Stockholder Approval**. Any other provision of this Agreement notwithstanding, no portion of this option shall be exercisable at any time prior to the approval of the Plan by the Company's stockholders.

# SECTION 3. NO TRANSFER OR ASSIGNMENT OF OPTION.

Except as otherwise provided in this Agreement, this option and the rights and privileges conferred hereby shall not be sold, pledged or otherwise transferred (whether by operation of law or otherwise) and shall not be subject to sale under execution, attachment, levy or similar process.

#### SECTION 4. EXERCISE PROCEDURES.

- (a) **Notice of Exercise**. The Optionee or the Optionee's representative may exercise this option by giving written notice to the Company pursuant to Section 12(c). The notice shall specify the election to exercise this option, the number of Shares for which it is being exercised and the form of payment. The person exercising this option shall sign the notice. In the event that this option is being exercised by the representative of the Optionee, the notice shall be accompanied by proof (satisfactory to the Company) of the representative's right to exercise this option. The Optionee or the Optionee's representative shall deliver to the Company, at the time of giving the notice, payment in a form permissible under Section 5 for the full amount of the Purchase Price.
- (b) **Issuance of Shares**. After receiving a proper notice of exercise, the Company shall cause to be issued one or more certificates evidencing the Shares for which this option has been exercised. Such Shares shall be registered (i) in the name of the person exercising this option, (ii) in the names of such person and his or her spouse as community property or as joint tenants with the right of survivorship or (iii) with the Company's consent, in the name of a revocable trust. The Company shall cause such certificates to be delivered to or upon the order of the person exercising this option.
- (c) Withholding Taxes. In the event that the Company determines that it is required to withhold any tax as a result of the exercise of this option, the Optionee, as a condition to the exercise of this option, shall make arrangements satisfactory to the Company to enable it to satisfy all withholding requirements. The Optionee shall also make arrangements satisfactory to the Company to enable it to satisfy any withholding requirements that may arise in connection with the disposition of Shares purchased by exercising this option.

## SECTION 5. PAYMENT FOR STOCK.

- (a) Cash. All or part of the Purchase Price may be paid in cash or cash equivalents.
- (b) **Surrender of Stock**. At the discretion of the Board of Directors, all or any part of the Purchase Price may be paid by surrendering, or attesting to the ownership of, Shares that are already owned by the Optionee. Such Shares shall be surrendered to the Company in good form for transfer and shall be valued at their Fair Market Value as of the date when this option is exercised.
- (c) Exercise/Sale. All or part of the Purchase Price and any withholding taxes may be paid by the delivery (on a form prescribed by the Company) of an irrevocable direction to a securities broker approved by the Company to sell Shares and to deliver all or part of the sales proceeds to the Company. However, payment pursuant to this Subsection (c) shall be permitted only if (i) Stock then is publicly traded and (ii) such payment does not violate applicable law.

## SECTION 6. TERM AND EXPIRATION.

- (a) **Basic Term**. This option shall in any event expire on the expiration date set forth in the Notice of Stock Option Grant, which date is 10 years after the Date of Grant (five years after the Date of Grant if this option is designated as an ISO in the Notice of Stock Option Grant and Section 3(b) of the Plan applies).
- (b) **Termination of Service (Except by Death)**. If the Optionee's Service terminates for any reason other than death, then this option shall expire on the earliest of the following occasions:
  - (i) The expiration date determined pursuant to Subsection (a) above;
  - (ii) The date three months after the termination of the Optionee's Service for any reason other than Disability; or
  - (iii) The date six months after the termination of the Optionee's Service by reason of Disability.

The Optionee may exercise all or part of this option at any time before its expiration under the preceding sentence, but only to the extent that this option had become exercisable before the Optionee's Service terminated. When the Optionee's Service terminates, this option shall expire immediately with respect to the number of Shares for which this option is not yet exercisable. In the event that the Optionee dies after termination of Service but before the expiration of this option, all or part of this option may be exercised (prior to expiration) by the executors or administrators of the Optionee's estate or by any person who has acquired this option directly from the Optionee by beneficiary designation, bequest or inheritance, but only to the extent that this option had become exercisable before the Optionee's Service terminated.

- (c) **Death of the Optionee**. If the Optionee dies while in Service, then this option shall expire on the earlier of the following dates:
  - (i) The expiration date determined pursuant to Subsection (a) above; or
  - (ii) The date 12 months after the Optionee's death.

All or part of this option may be exercised at any time before its expiration under the preceding sentence by the executors or administrators of the Optionee's estate or by any person who has acquired this option directly from the Optionee by beneficiary designation, bequest or inheritance, but only to the extent that this option had become exercisable before the Optionee's death. When the Optionee dies, this option shall expire immediately with respect to the number of Shares for which this option is not yet exercisable.

(d) Part-Time Employment and Leaves of Absence. If the Optionee commences working on a part-time basis, then the Company may adjust the vesting schedule set forth in the Notice of Stock Option Grant in accordance with the Company's part-time work

policy or the terms of an agreement between the Optionee and the Company pertaining to his or her part-time schedule. If the Optionee goes on a leave of absence, then the Company may adjust the vesting schedule set forth in the Notice of Stock Option Grant in accordance with the Company's leave of absence policy or the terms of such leave. Except as provided in the preceding sentence, Service shall be deemed to continue for any purpose under this Agreement while the Optionee is on a *bona fide* leave of absence, if (i) such leave was approved by the Company in writing and (ii) continued crediting of Service for such purpose is expressly required by the terms of such leave or by applicable law (as determined by the Company). Service shall be deemed to terminate when such leave ends, unless the Optionee immediately returns to active work.

- (e) **Notice Concerning ISO Treatment**. Even if this option is designated as an ISO in the Notice of Stock Option Grant, it ceases to qualify for favorable tax treatment as an ISO to the extent that it is exercised:
  - (i) More than three months after the date when the Optionee ceases to be an Employee for any reason other than death or permanent and total disability (as defined in Section 22(e)(3) of the Code);
  - (ii) More than 12 months after the date when the Optionee ceases to be an Employee by reason of permanent and total disability (as defined in Section 22(e)(3) of the Code); or
  - (iii) More than three months after the date when the Optionee has been on a leave of absence for 90 days, unless the Optionee's reemployment rights following such leave were guaranteed by statute or by contract.

#### SECTION 7. RIGHT OF FIRST REFUSAL.

(a) **Right of First Refusal**. In the event that the Optionee proposes to sell, pledge or otherwise transfer to a third party any Shares acquired under this Agreement, or any interest in such Shares, the Company shall have the Right of First Refusal with respect to all (and not less than all) of such Shares. If the Optionee desires to transfer Shares acquired under this Agreement, the Optionee shall give a written Transfer Notice to the Company describing fully the proposed transfer, including the number of Shares proposed to be transferred, the proposed transfer price, the name and address of the proposed Transferee and proof satisfactory to the Company that the proposed sale or transfer will not violate any applicable federal, State or foreign securities laws. The Transfer Notice shall be signed both by the Optionee and by the proposed Transferee and must constitute a binding commitment of both parties to the transfer of the Shares. The Company shall have the right to purchase all, and not less than all, of the Shares on the terms of the proposal described in the Transfer Notice (subject, however, to any change in such terms permitted under Subsection (b) below) by delivery of a notice of exercise of the Right of First Refusal within 30 days after the date when the Transfer Notice was received by the Company.

- (b) **Transfer of Shares**. If the Company fails to exercise its Right of First Refusal within 30 days after the date when it received the Transfer Notice, the Optionee may, not later than 90 days following receipt of the Transfer Notice by the Company, conclude a transfer of the Shares subject to the Transfer Notice on the terms and conditions described in the Transfer Notice, provided that any such sale is made in compliance with applicable federal, State and foreign securities laws and not in violation of any other contractual restrictions to which the Optionee is bound. Any proposed transfer on terms and conditions different from those described in the Transfer Notice, as well as any subsequent proposed transfer by the Optionee, shall again be subject to the Right of First Refusal and shall require compliance with the procedure described in Subsection (a) above. If the Company exercises its Right of First Refusal, the parties shall consummate the sale of the Shares on the terms set forth in the Transfer Notice within 60 days after the date when the Company received the Transfer Notice (or within such longer period as may have been specified in the Transfer Notice); provided, however, that in the event the Transfer Notice provided that payment for the Shares was to be made in a form other than cash or cash equivalents paid at the time of transfer, the Company shall have the option of paying for the Shares with cash or cash equivalents equal to the present value of the consideration described in the Transfer Notice.
- (c) Additional or Exchanged Securities and Property. In the event of a merger or consolidation of the Company with or into another entity, any other corporate reorganization, a stock split, the declaration of a stock dividend, the declaration of an extraordinary dividend payable in a form other than stock, a spin-off, an adjustment in conversion ratio, a recapitalization or a similar transaction affecting the Company's outstanding securities, any securities or other property (including cash or cash equivalents) that are by reason of such transaction exchanged for, or distributed with respect to, any Shares subject to this Section 7 shall immediately be subject to the Right of First Refusal. Appropriate adjustments to reflect the exchange or distribution of such securities or property shall be made to the number and/or class of the Shares subject to this Section 7.
- (d) **Termination of Right of First Refusal**. Any other provision of this Section 7 notwithstanding, in the event that the Stock is readily tradable on an established securities market when the Optionee desires to transfer Shares, the Company shall have no Right of First Refusal, and the Optionee shall have no obligation to comply with the procedures prescribed by Subsections (a) and (b) above.
- (e) **Permitted Transfers**. This Section 7 shall not apply to (i) a transfer by beneficiary designation, will or intestate succession or (ii) a transfer to one or more members of the Optionee's Immediate Family or to a trust established by the Optionee for the benefit of the Optionee and/or one or more members of the Optionee's Immediate Family, provided in either case that the Transferee agrees in writing on a form prescribed by the Company to be bound by all provisions of this Agreement. If the Optionee transfers any Shares acquired under this Agreement, either under this Subsection (e) or after the Company has failed to exercise the Right of First Refusal, then this Agreement shall apply to the Transferee to the same extent as to the Optionee.
- (f) **Termination of Rights as Stockholder**. If the Company makes available, at the time and place and in the amount and form provided in this Agreement, the consideration for the Shares to be purchased in accordance with this Section 7, then after such time the person from whom such Shares are to be purchased shall no longer have any rights as a holder of such

Shares (other than the right to receive payment of such consideration in accordance with this Agreement). Such Shares shall be deemed to have been purchased in accordance with the applicable provisions hereof, whether or not the certificate(s) therefor have been delivered as required by this Agreement.

(g) **Assignment of Right of First Refusal**. The Board of Directors may freely assign the Company's Right of First Refusal, in whole or in part. Any person who accepts an assignment of the Right of First Refusal from the Company shall assume all of the Company's rights and obligations under this Section 7.

#### SECTION 8. LEGALITY OF INITIAL ISSUANCE.

No Shares shall be issued upon the exercise of this option unless and until the Company has determined that:

- (a) It and the Optionee have taken any actions required to register the Shares under the Securities Act or to perfect an exemption from the registration requirements thereof;
  - (b) Any applicable listing requirement of any stock exchange or other securities market on which Stock is listed has been satisfied; and
  - (c) Any other applicable provision of federal, State or foreign law has been satisfied.

## SECTION 9. NO REGISTRATION RIGHTS.

The Company may, but shall not be obligated to, register or qualify the sale of Shares under the Securities Act or any other applicable law. The Company shall not be obligated to take any affirmative action in order to cause the sale of Shares under this Agreement to comply with any law.

## SECTION 10. RESTRICTIONS ON TRANSFER OF SHARES.

- (a) **Securities Law Restrictions**. Regardless of whether the offering and sale of Shares under the Plan have been registered under the Securities Act or have been registered or qualified under the securities laws of any State, the Company at its discretion may impose restrictions upon the sale, pledge or other transfer of such Shares (including the placement of appropriate legends on stock certificates or the imposition of stop-transfer instructions) if, in the judgment of the Company, such restrictions are necessary or desirable in order to achieve compliance with the Securities Act, the securities laws of any State or any other law.
- (b) Market Stand-Off. In connection with any underwritten public offering by the Company of its equity securities pursuant to an effective registration statement filed under the Securities Act, including the Company's initial public offering, the Optionee or a Transferee shall not directly or indirectly sell, make any short sale of, loan, hypothecate, pledge, offer, grant or sell any option or other contract for the purchase of, purchase any option or other contract for the sale of, or otherwise dispose of or transfer, or agree to engage in any of the foregoing

transactions with respect to, any Shares acquired under this Agreement without the prior written consent of the Company or its managing underwriter. Such restriction (the "Market Stand-Off") shall be in effect for such period of time following the date of the final prospectus for the offering as may be requested by the Company or such underwriter. In no event, however, shall such period exceed 180 days plus such additional period as may reasonably be requested by the Company or such underwriter to accommodate regulatory restrictions on (i) the publication or other distribution of research reports or (ii) analyst recommendations and opinions, including (without limitation) the restrictions set forth in Rule 2711(f)(4) of the National Association of Securities Dealers and Rule 472(f)(4) of the New York Stock Exchange, as amended, or any similar successor rules. The Market Stand-Off shall in any event terminate two years after the date of the Company's initial public offering. In the event of the declaration of a stock dividend, a spin-off, a stock split, an adjustment in conversion ratio, a recapitalization or a similar transaction affecting the Company's outstanding securities without receipt of consideration, any new, substituted or additional securities which are by reason of such transaction distributed with respect to any Shares subject to the Market Stand-Off, or into which such Shares thereby become convertible, shall immediately be subject to the Market Stand-Off. In order to enforce the Market Stand-Off, the Company may impose stop-transfer instructions with respect to the Shares acquired under this Agreement until the end of the applicable stand-off period. The Company's underwriters shall be beneficiaries of the agreement set forth in this Subsection (b). This Subsection (b) shall not apply to Shares registered in the public offering under the Securities Act.

- (c) **Investment Intent at Grant**. The Optionee represents and agrees that the Shares to be acquired upon exercising this option will be acquired for investment, and not with a view to the sale or distribution thereof.
- (d) **Investment Intent at Exercise**. In the event that the sale of Shares under the Plan is not registered under the Securities Act but an exemption is available that requires an investment representation or other representation, the Optionee shall represent and agree at the time of exercise that the Shares being acquired upon exercising this option are being acquired for investment, and not with a view to the sale or distribution thereof, and shall make such other representations as are deemed necessary or appropriate by the Company and its counsel.
  - (e) Legends. All certificates evidencing Shares purchased under this Agreement shall bear the following legend:

"THE SHARES REPRESENTED HEREBY MAY NOT BE SOLD, ASSIGNED, TRANSFERRED, ENCUMBERED OR IN ANY MANNER DISPOSED OF, EXCEPT IN COMPLIANCE WITH THE TERMS OF A WRITTEN AGREEMENT BETWEEN THE COMPANY AND THE REGISTERED HOLDER OF THE SHARES (OR THE PREDECESSOR IN INTEREST TO THE SHARES). SUCH AGREEMENT GRANTS TO THE COMPANY CERTAIN RIGHTS OF FIRST REFUSAL UPON AN ATTEMPTED TRANSFER OF THE SHARES. THE SECRETARY OF THE COMPANY WILL UPON WRITTEN REQUEST FURNISH A COPY OF SUCH AGREEMENT TO THE HOLDER HEREOF WITHOUT CHARGE."

All certificates evidencing Shares purchased under this Agreement in an unregistered transaction shall bear the following legend (and such other restrictive legends as are required or deemed advisable under the provisions of any applicable law):

"THE SHARES REPRESENTED HEREBY HAVE NOT BEEN REGISTERED UNDER THE SECURITIES ACT OF 1933, AS AMENDED, AND MAY NOT BE SOLD, PLEDGED, OR OTHERWISE TRANSFERRED WITHOUT AN EFFECTIVE REGISTRATION THEREOF UNDER SUCH ACT OR AN OPINION OF COUNSEL, SATISFACTORY TO THE COMPANY AND ITS COUNSEL, THAT SUCH REGISTRATION IS NOT REQUIRED."

- (f) **Removal of Legends**. If, in the opinion of the Company and its counsel, any legend placed on a stock certificate representing Shares sold under this Agreement is no longer required, the holder of such certificate shall be entitled to exchange such certificate for a certificate representing the same number of Shares but without such legend.
- (g) **Administration**. Any determination by the Company and its counsel in connection with any of the matters set forth in this Section 10 shall be conclusive and binding on the Optionee and all other persons.

## SECTION 11. ADJUSTMENT OF SHARES.

In the event of any transaction described in Section 8(a) of the Plan, the terms of this option (including, without limitation, the number and kind of Shares subject to this option and the Exercise Price) shall be adjusted as set forth in Section 8(a) of the Plan. In the event that the Company is a party to a merger or consolidation, this option shall be subject to the agreement of merger or consolidation, as provided in Section 8(b) of the Plan.

# SECTION 12. MISCELLANEOUS PROVISIONS.

- (a) **Rights as a Stockholder**. Neither the Optionee nor the Optionee's representative shall have any rights as a stockholder with respect to any Shares subject to this option until the Optionee or the Optionee's representative becomes entitled to receive such Shares by filing a notice of exercise and paying the Purchase Price pursuant to Sections 4 and 5.
- (b) **No Retention Rights**. Nothing in this option or in the Plan shall confer upon the Optionee any right to continue in Service for any period of specific duration or interfere with or otherwise restrict in any way the rights of the Company (or any Parent or Subsidiary employing or retaining the Optionee) or of the Optionee, which rights are hereby expressly reserved by each, to terminate his or her Service at any time and for any reason, with or without cause.
- (c) **Notice**. Any notice required by the terms of this Agreement shall be given in writing. It shall be deemed effective upon (i) personal delivery, (ii) deposit with the United States Postal Service, by registered or certified mail, with postage and fees prepaid or (iii) deposit with Federal Express Corporation, with shipping charges prepaid. Notice shall be addressed to the Company at its principal executive office and to the Optionee at the address that he or she most recently provided to the Company in accordance with this Subsection (c).

- (d) **Entire Agreement**. The Notice of Stock Option Grant, this Agreement and the Plan constitute the entire contract between the parties hereto with regard to the subject matter hereof. They supersede any other agreements, representations or understandings (whether oral or written and whether express or implied) that relate to the subject matter hereof.
- (e) Choice of Law. This Agreement shall be governed by, and construed in accordance with, the laws of the State of Delaware, as such laws are applied to contracts entered into and performed in such State.

#### SECTION 13. ACKNOWLEDGEMENTS OF THE OPTIONEE.

- (a) **Tax Consequences**. The Optionee agrees that the Company does not have a duty to design or administer the Plan or its other compensation programs in a manner that minimizes the Optionee's tax liabilities. The Optionee shall not make any claim against the Company or its Board of Directors, officers or employees related to tax liabilities arising from this option or the Optionee's other compensation. In particular, the Optionee acknowledges that this option is exempt from Section 409A of the Code only if the Exercise Price is at least equal to the Fair Market Value per Share on the Date of Grant. Since Shares are not traded on an established securities market, the determination of their Fair Market Value is made by the Board of Directors or by an independent valuation firm retained by the Company. The Optionee acknowledges that there is no guarantee in either case that the Internal Revenue Service will agree with the valuation, and the Optionee shall not make any claim against the Company or its Board of Directors, officers or employees in the event that the Internal Revenue Service asserts that the valuation was too low.
- (b) **Electronic Delivery of Documents**. The Optionee agrees that the Company may deliver by email all documents relating to the Plan or this option (including, without limitation, a copy of the Plan) and all other documents that the Company is required to deliver to its security holders (including, without limitation, disclosures that may be required by the Securities and Exchange Commission). The Optionee also agrees that the Company may deliver these documents by posting them on a website maintained by the Company or by a third party under contract with the Company. If the Company posts these documents on a website, it shall notify the Optionee by email.

#### **SECTION 14. DEFINITIONS.**

- (a) "Agreement" shall mean this Stock Option Agreement.
- (b) "Board of Directors" shall mean the Board of Directors of the Company, as constituted from time to time or, if a Committee has been appointed, such Committee.
  - (c) "Code" shall mean the Internal Revenue Code of 1986, as amended.
  - (d) "Committee" shall mean a committee of the Board of Directors, as described in Section 2 of the Plan.
  - (e) "Company" shall mean ContextLogic, Inc., a Delaware corporation.

- (f) "Consultant" shall mean a person who performs bona fide services for the Company, a Parent or a Subsidiary as a consultant or advisor, excluding Employees and Outside Directors.
- (g) "Date of Grant" shall mean the date of grant specified in the Notice of Stock Option Grant, which date shall be the later of (i) the date on which the Board of Directors resolved to grant this option or (ii) the first day of the Optionee's Service.
- (h) "Disability" shall mean that the Optionee is unable to engage in any substantial gainful activity by reason of any medically determinable physical or mental impairment.
  - (i) "Employee" shall mean any individual who is a common-law employee of the Company, a Parent or a Subsidiary.
- (j) "Exercise Price" shall mean the amount for which one Share may be purchased upon exercise of this option, as specified in the Notice of Stock Option Grant.
- (k) "Fair Market Value" shall mean the fair market value of a Share, as determined by the Board of Directors in good faith. Such determination shall be conclusive and binding on all persons.
- (l) "Immediate Family" shall mean any child, stepchild, grandchild, parent, stepparent, grandparent, spouse, sibling, mother-in-law, father-in-law, son-in-law, daughter-in-law, brother-in-law or sister-in-law and shall include adoptive relationships.
  - (m) "ISO" shall mean an employee incentive stock option described in Section 422(b) of the Code.
  - (n) "Notice of Stock Option Grant" shall mean the document so entitled to which this Agreement is attached.
  - (o) "NSO" shall mean a stock option not described in Sections 422(b) or 423(b) of the Code.
  - (p) "Optionee" shall mean the person named in the Notice of Stock Option Grant.
  - (q) "Outside Director" shall mean a member of the Board of Directors who is not an Employee.
- (r) "Parent" shall mean any corporation (other than the Company) in an unbroken chain of corporations ending with the Company, if each of the corporations other than the Company owns stock possessing 50% or more of the total combined voting power of all classes of stock in one of the other corporations in such chain.
  - (s) "Plan" shall mean the ContextLogic, Inc. 2010 Stock Plan, as in effect on the Date of Grant.

- (t) "Purchase Price" shall mean the Exercise Price multiplied by the number of Shares with respect to which this option is being exercised.
  - (u) "Right of First Refusal" shall mean the Company's right of first refusal described in Section 7.
  - (v) "Securities Act" shall mean the Securities Act of 1933, as amended.
  - (w) "Service" shall mean service as an Employee, Outside Director or Consultant.
  - (x) "Share" shall mean one share of Stock, as adjusted in accordance with Section 8 of the Plan (if applicable).
  - (y) "Stock" shall mean the Common Stock of the Company.
- (z) "**Subsidiary**" shall mean any corporation (other than the Company) in an unbroken chain of corporations beginning with the Company, if each of the corporations other than the last corporation in the unbroken chain owns stock possessing 50% or more of the total combined voting power of all classes of stock in one of the other corporations in such chain.
- (aa) "Transferee" shall mean any person to whom the Optionee has directly or indirectly transferred any Share acquired under this Agreement.
  - (bb) "Transfer Notice" shall mean the notice of a proposed transfer of Shares described in Section 7.